KETENES II

KETENES II

Thomas T. Tidwell



Copyright © 2006 by John Wiley & Sons, Inc. All rights reserved.

Published by John Wiley & Sons, Inc., Hoboken, New Jersey. Published simultaneously in Canada.

No part of this publication may be reproduced, stored in a retrieval system, or transmitted in any form or by any means, electronic, mechanical, photocopying, recording, scanning, or otherwise, except as permitted under Section 107 or 108 of the 1976 United States Copyright Act, without either the prior written permission of the Publisher, or authorization through payment of the appropriate per-copy fee to the Copyright Clearance Center, Inc., 222 Rosewood Drive, Danvers, MA 01923, (978) 750-8400, fax (978) 750-4470, or on the web at www.copyright.com. Requests to the Publisher for permission should be addressed to the Permissions Department, John Wiley & Sons, Inc., 111 River Street, Hoboken, NJ 07030, (201) 748-6011, fax (201) 748-6008, or online at http://www.wiley.com/go/permission.

Limit of Liability/Disclaimer of Warranty: While the publisher and author have used their best efforts in preparing this book, they make no representations or warranties with respect to the accuracy or completeness of the contents of this book and specifically disclaim any implied warranties of merchantability or fitness for a particular purpose. No warranty may be created or extended by sales representatives or written sales materials. The advice and strategies contained herein may not be suitable for your situation. You should consult with a professional where appropriate. Neither the publisher nor author shall be liable for any loss of profit or any other commercial damages, including but not limited to special, incidental, consequential, or other damages.

For general information on our other products and services please contact our Customer Care Department within the United States at (800) 762-2974, outside the United States at (317) 572-3993 or fax (317) 572-4002.

Wiley also publishes its books in a variety of electronic formats. Some content that appears in print, however, may not be available in electronic format.

Library of Congress Cataloging-in-Publication Data:

Tidwell, Thomas T., 1939Ketenes/by Thomas T. Tidwell. – 2nd ed. p. cm.
Includes bibliographical references and index.
ISBN-13: 978-0-471-69282-9
ISBN-10: 0-471-69282-4
1. Ketenes. I. Title.
QD305. K2T53 2006
547'.436-dc22

2005013965

Printed in the United States of America

10 9 8 7 6 5 4 3 2 1

Dedicated to Sarah, for her inspiration and support

Contents

Pref	àce	xiii
Intr	oduction	1
	References	3
1	Structure, Bonding, and Thermochemistry of Ketenes	5
	1.1 Theoretical and Computational Studies of Ketenes	5
	1.1.1 Molecular and Electronic Structure and Energy	5
	References	8
	1.1.2 Theoretical and Computational Studies of	
	Ketene Reactions	9
	References	12
	1.1.3 Substituent Effects on Ketenes	13
	References	17
	1.2 Molecular Structure Determinations	18
	References	21
	1.3 Thermochemistry of Ketenes	22
	References	25
2	Spectroscopy and Physical Properties of Ketenes	27
	2.1 Nuclear Magnetic Resonance Spectroscopy	27
	References	33
	2.2 Ultraviolet Spectra, Photoelectron Spectra, and	
	Chiroptical Properties	35
	References	38
	2.3 Infrared Spectra and Raman Spectra	39
	References	46
	2.4 Dipole Moments	48
	References	49
	2.5 Mass Spectrometry and Gas Phase Ion Chemistry	49
	References	52
3	Preparation of Ketenes	55
	3.1 Ketenes from Ketene Dimers	55
	References	57

vii

3.2	Ketene	s from Carboxylic Acids and their Derivatives	57
	3.2.1	Ketenes from Carboxylic Acids and Anhydrides	57
		3.2.1.1 Ketenes from Carboxylic Acids	57
		References	60
		3.2.1.2 Ketenes from Acid Anhydrides	61
		References	64
	3.2.2	Ketenes from Acyl Halides and Activated Acids	65
	Referen	nces	74
	3.2.3	Ketenes from Esters	76
		3.2.3.1 <i>Ketenes from Ester Enolates</i>	76
		3.2.3.2 Ketenes by Ester Pyrolysis	81
		References	84
	3.2.4	Ketenes by Dehalogenation of α -Halo Carboxylic	
		Acid Derivatives	86
	Referei	nces	89
3.3	Ketene	s from Diazo Ketones (Wolff Rearrangements)	90
	3.3.1	Thermal Wolff Rearrangement	93
	3.3.2	Catalyzed Wolff Rearrangement	97
	3.3.3	Photochemical Wolff Rearrangement	104
	3.3.4	Other Routes to Ketocarbene Rearrangements	110
	Referen	nces	111
3.4	Ketene	s by Photochemical and Thermolytic Methods	114
	3.4.1	Ketenes from Cyclobutanones and Cyclobutenones	114
	Referen	nces	129
	3.4.2	Ketenes from Photolysis of Cycloalkanones	
		and Enones	132
	Referen	nces	141
	3.4.3	Ketenes from Cyclohexadienones and other	
		Cycloalkenones	142
	Referen	nces	151
	3.4.4	Ketenes from Dioxinones	153
	Referen	nces	155
	3.4.5	Ketenes by Thermolysis of Alkynyl Ethers	156
	Referen	nces	159
	3.4.6	Ketenes from other Thermolytic and Photochemical	
		Routes	160
	Referen	nces	169
3.5	Ketene	s from Metal Carbene Complexes	171
Refe	rences		179
3.6	Ketene	Formation from Cations and Free Radicals	181
Refe	rences		182
3.7	Ketene	s from Oxidation of Alkynes	183
Refe	rences		185
3.8	Other 1	Routes to Ketenes	186
Refe	rences		191

4	Тур	es of Ke	etenes	193
	4.1	Carbon	-Substituted Ketenes	193
		4.1.1	Alkylketenes	193
		Referer	nces	203
		4.1.2	Alkenylketenes	206
		Referer	nces	213
		4.1.3	Alkynyl- and Cyanoketenes	214
		Referer	nces	220
		4.1.4	Aryl- and Heteroarylketenes	221
		Referer	nces	234
		4.1.5	Cyclopropyl-, Cyclopropenyl-, and	
			Oxiranylketenes	237
		Referer	nces	243
		4.1.6	Acylketenes	244
		Referer	nces	259
		4.1.7	Imidoylketenes	263
		Referer	nces	267
		4.1.8	Cumulene-Substituted Ketenes	268
		Referer	nces	273
		4.1.9	Ketenes with Charged, Radical, or Carbenic	
			Side Chains	273
		Referer	nces	283
		4.1.10	Fulvenones and Cumulenones	284
			4.1.10.1 Triafulvenones	285
			4.1.10.2 Pentafulvenones	285
			References	292
		4.1.11	Oxoquinone Methides, Oxoxylylenes, and	
			Related Species	294
		Referer	nces	303
	4.2	Nitroge	en-Substituted Ketenes	305
	Refe	erences		311
	4.3	Oxyger	n-Substituted Ketenes	313
	Refe	erences		320
	4.4	Haloge	en-Substituted Ketenes	321
		4.4.1	Fluoroketenes and Fluoroalkylketenes	321
		Referer	nces	329
		4.4.2	Chloro- and Bromoketenes	331
		Referer	nces	340
	4.5	Silyl-,	Germyl- and Stannylketenes	342
	Refe	erences		358
	4.6	Phosph	orus-Substituted Ketenes	361
	Refe	erences		369
	4.7	Sulfur-	Substituted Ketenes	371
	Refe	erences		376

x CONTENTS

	4.8	Metal-	Substituted	1 Ketenes	377
		4.8.1	Ynols an	d Ynolates	377
		Refere	nces		382
		4.8.2	Boron-St	ubstituted Ketenes	384
		Refere	nces		386
		4.8.3	Other M	etal-Substituted Ketenes and Metal Ketenides	387
		Refere	nces		389
		4.8.4	Metal-Co	omplexed Ketenes	389
		Refere	nces		398
	4.9	Bisket	enes		400
	Refe	erences			422
	4.10	Keteny	l Radicals	, Anions, and Cations	425
	Refe	erences			427
	4.11	Cumul	enones		428
	Refe	erences			433
5	Rea	ctions o	f Ketenes		437
	5.1	Oxidat	ion and Re	eduction of Ketenes	437
	Refe	erences			440
	5.2	Photoc	hemical R	eactions	441
	Refe	erences			445
	5.3	Therm	olysis Rea	ctions	446
	Refe	erences			449
	5.4	Cycloa	ddition Re	eactions of Ketenes	449
	Refe	erences			452
		5.4.1	Intermol	ecular [2+2] Cycloaddition	453
			5.4.1.1	Dimerization of Ketenes	453
			Referenc	es	462
			5.4.1.2	Cycloadditions with Alkenes and Dienes	464
			Referenc	es	472
			5.4.1.3	Mechanism of Ketene [2+2]	
				Cycloadditions with Alkenes	474
			Referenc	es	485
			5.4.1.4	Cycloaddition of Ketenes with	
				Nucleophilic Alkenes	487
			Referenc	es	492
			5.4.1.5	Cycloaddition of Ketenes with Allenes	492
			Referenc	es	494
			5.4.1.6	Cycloaddition of Ketenes with Alkynes	495
			Referenc	es	499
			5.4.1.7	Cycloaddition of Ketenes with Imines	500
			Referenc	es	510
		5.4.2	[3+2] C	ycloadditions of Ketenes	513
		Refere	nces		515

	5.4.3	[4+2] Cy	cloadditions	516
	Referen	ces		520
	5.4.4	Intramole	cular Cycloadditions of Ketenes	520
	Referen	ces		525
	5.4.5	Cycloadd	ition of Ketenes with Carbonyl Groups	525
	Referen	ces		531
5.5	Nucleop	philic Add	ition to Ketenes	533
	5.5.1	Mechanis	ms of Hydration of Ketenes	533
	Referen	ces		541
	5.5.2	Mechanis	ms of Amination of Ketenes	543
	Referen	ces		551
	5.5.3	Nucleoph	ilic Addition to Ketenes: Preparative Aspects	552
		5.5.3.1	Hydride Reduction	552
		Reference	S	553
		5.5.3.2	Reaction of Oxygen Nucleophiles with Ketenes	554
		Reference	S	559
		5.5.3.3	Reaction of Nitrogen Nucleophiles and	
			Electrophiles with Ketenes	560
		Reference	S	567
		5.5.3.4	Carbon and Silicon Nucleophiles	568
		Reference	S	575
	5.5.4	Wittig-Ty	pe Reactions	576
	Referen	ces		579
5.6	Electro	philic Add	ition to Ketones	580
	5.6.1	Protonatio	on of Ketenes	580
	Referen	ces		585
	5.6.2	Oxygenat	ion of Ketenes	587
	Referen	ces		591
5.7	Radical	Reactions	of Ketenes	592
Refe	rences			598
5.8	Polyme	rization of	Ketenes	600
Refe	rences			602
5.9	Stereos	electivity i	n Ketene Reactions	603
Refe	rences			615
5.10	Other A	Additions to	o Ketenes	618
	5.10.1	Reaction	with Diazomethanes	618
	5.10.2	Reaction	with Sulfur Dioxide	621
	5.10.3	Hydrogen	ation of Ketenes	621
	Referen	ces		621
5.11	Ketene	Reactions	Using Polymer Supports	622
Refe	rences			625

Preface

It was a century ago that Hermann Staudinger introduced ketenes to the world, and almost 50 years ago that I first encountered these beguiling reactive intermediates in my first course in organic chemistry, using a textbook written by the inimitable Louis Fieser. Ketenes are remarkable for the variety of ways in which they can be prepared, for their high chemical reactivity, and for the diverse range of useful products from their reactions. Ketenes have long been renown for their unique reactivity in [2+2] cycloaddition reactions giving a ready source of four membered ring compounds, their distinctive characteristic of undergoing facile dimerization, and the high stereoselectivity shown in many of their reactions.

The first edition of this book appeared 10 years ago, and even though ketene chemistry seemed at a high point at that time it was predicted "... in 2005, the opportunities for research will be even greater than they are today." As better methodology has been devised for ketene preparation and new applications have been found for this family the interest in their chemistry has steadily increased, and the developments over the past decade have exceeded my greatest expectations. This new edition incorporates 1,000 new references, some published in 2005, and to keep the text to a reasonable size just as many have had been dropped. The format of the previous work has proven to be useful and is retained, with 5 principal sections "Structure, Bonding, and Thermochemistry", Spectroscopy and Physical Properties", Preparation", Types of Ketenes", and "Reactions". The text has been completely rewritten with incorporation of new material, and all of the chemical equations and structural formulae have been redrawn in a uniform style. The extensive lists of known ketenes have not been continued because of space restraints, and because of the present capability of searching these online.

Much has been learned, but there is even a greater amount still to be discovered. Advances in experimental techniques have made many classes of ketenes into conveniently observable intermediates, but other challenges remain, particularly in the observation of highly reactive ketenes by methods such as NMR. Several classes of ketenes calculated to be stable are still unknown experimentally, and provide a continuing challenge. Many surprising new developments may be confidently awaited in the next decade.

Thanks are due to the many enterprising chemists around the world who have provided the fascinating results which are reported here. I am particularly grateful to Peter Lillya, who graciously read the entire manuscript, and to Pat Woodcock for her invaluable help with the writing over many years. I apologize for all errors and omissions, for which I am solely responsible, and am grateful to those including

xiv PREFACE

reviewers of the first edition who have pointed out mistakes and provided ideas. My own contributions would have been impossible without my delightful and skillful coworkers and collaborators, especially Annette Allen, Huda Henry-Riyad, Nanyan Fu, Shinjiro Kobayashi, Andrei Fedorov, Katayoun Najafian, Kuangsen Sung, Wen-wei Huang, Mike Fenwick, Adel Rafai Far, Pat Moore, Romeo Marra, Jim Colomvakos, Ian Egle, Paul Renton, and many others.

FRENCH BEACH

British Columbia, July 2005

Introduction

The first ketene to be prepared and characterized was diphenylketene (1). It was made by Hermann Staudinger in 1905, while he was an instructor (*Unterrichts-Assistent*) at the Kaiser Wilhelms-Universität in Strasbourg (then part of Germany), in the Organic Department of Johannes Thiele.¹ The preparation involved the reaction of 2-chorodiphenylacetyl chloride with zinc (equation 1) and, as recounted by Staudinger in his scientific autobiography,² was inspired by the preparation of the stable triphenylmethyl radical by Gomberg. Dimethylketene (3)³ and dibenzopentafulvenone (4)⁴ were soon prepared by the same method.



Shortly thereafter, N. T. M. Wilsmore at University College, London, prepared the parent ketene $CH_2=C=O(5)$ from the pyrolysis of acetic anhydride or acetone using a hot platinum wire.⁵ Staudinger and Klever then prepared **5** from bromoacetyl bromide and zinc, and engaged in a brisk discussion with Wilsmore as to the priority for the discovery and as to whether ketene is best represented by $CH_2=C=O$ or $HC\equiv COH$, a possibility that had been mentioned by Wilsmore. This was decided in favor of Wilsmore's priority and the former structure.^{7,8}

Staudinger was then successively "Associate Professor"² at the Technische Hochschule in Karlsruhe (1907–1912), Professor at the ETH in Zürich (1912–1926), and Professor in Freiburg (1926–1951). He summarized his early studies on ketenes in a monograph in 1912⁹ and published further studies on ketenes through 1925. He then devoted his efforts to the study of macromolecules and was recognized for this work by the award of the Nobel Prize in Chemistry in 1953.

Ketenes II, by Thomas T. Tidwell

Copyright © 2006 John Wiley & Sons, Inc.

2 INTRODUCTION

Wilsmore had been at the ETH in Zürich, and left there in 1903 for University College, London, where he subsequently became Assistant Professor. In 1912 he was appointed to the Chair of Chemistry at the new University of Western Australia and, owing to the demands of setting up a new department, made no more contributions to ketene chemistry.¹⁰

Edgar Wedekind missed the opportunity to discover ketenes while in Tübingen in 1901, when he treated Ph₂CHCOCl (**6**) with *n*-Pr₃N, noted the formation of *n*-Pr₃NHCl, and proposed the formation of intermediate **7** (equation 2), which is equivalent to Ph₂C=C=O (**2**).¹¹ However, Wedekind equivocated in his claim¹² and did not isolate the ketene, and this achievement was left for Staudinger in 1905.

$$\begin{array}{ccc} Ph & O & \xrightarrow{n-Pr_3N} & Ph_2\dot{C}-\dot{C}=O \\ & & & & & & \\ 6 & & & & 7 \end{array}$$

Ludwig Wolff in 1902 also generated ketenes as unobserved intermediates in what became known as the Wolff rearrangement,¹³ but did not identify the intermediate as a ketene or adopt a ketene structure until 1912,¹⁴ by which time ketenes were well-recognized intermediates. Wolff's experiment involved refluxing the diazo ketone **8** (formulated at that time as having a cyclic structure) in water, forming the ketene **9**, which led to the unobserved acid **10**, which underwent decarboxylation to give **11** (equation 3).¹³ Wolff recognized that rearrangement had occurred and that the acid **10** was a key intermediate, but rather than ketene formation, he proposed that water addition occurred, giving **12**, which underwent a double migration forming **10**, and cited the analogy of the pinacol and benzilic acid rearrangements (equation 4).¹³



The utility of ketenes in both laboratory and industrial practice was quickly recognized, and these species have been extensively utilized, including as intermediates in the manufacture of acetic acid and acetic anhydride, pharmaceutical intermediates, integrated circuits for computer chips, and paper coatings. Their study has attracted many of the sharpest minds in chemistry, among them Nobel

laureates in addition to Staudinger including R. B. Woodward, E. J. Corey, Roald Hoffmann, R. G. W. Norrish, George Porter, K. Fukui, L. Ruzicka, V. Prelog, O. Diels, R. W. Holley, G. Natta, W. N. Lipscomb, A. Zewail, John Polanyi, Derek Barton, John Pople, R. F. Curl, Jr., F. S. Rowland, and George Olah.

Following publications of Staudinger's book,⁹ there have been many reviews on ketenes, including reviews on specific topics.^{15–31} Among these is a volume devoted to a detailed survey of the preparation of ketenes and their application in synthesis.¹⁵ A more detailed description of the origins of ketene chemistry has been presented.³²

REFERENCES

- 1. Staudinger, H. Chem. Ber. 1905, 38, 1735-1739.
- 2. Staudinger, H. From Organic Chemistry to Macromolecules; Wiley, New York, 1970.
- 3. Staudinger, H.; Klever, H. Chem. Ber. 1906, 39, 968-971.
- 4. Staudinger, H. Chem. Ber. 1906, 39, 3062-3067.
- 5. Wilsmore, N. T. M. J. Chem. Soc. 1907, 91, 1938-1941.
- 6. Staudinger, H.; Klever, H. Chem. Ber. 1908 41, 594-600.
- 7. Wilsmore, N. T. M. Chem. Ber. 1907, 91, 1938-1941.
- 8. Staudinger, H.; Klever, H. Chem. Ber. 1908 41, 1516-1517.
- 9. Staudinger, H. Die Ketene; Verlag Enke, Stuttgart, 1912.
- 10. Obituary Notice of N. T. M Wilsmore. J. Chem. Soc. 1941, 59-60.
- 11. Wedekind, E. Chem. Ber. 1901, 34, 2070-2077.
- 12. Wedekind, E. Liebigs Ann. Chem. 1902, 323, 246-257.
- 13. Wolff, L. Justus Liebigs Ann. Chem. 1902, 325, 129-195.
- 14. Wolff, L. Justus Liebigs Ann. Chem. 1912, 394, 23-59.
- 15. In Science of Synthesis; Bellus, D.; Danheiser, R., Eds.; Thieme, Stuttgart, 2006, Vol. 23, all.
- 16. Hanford, W. E.; Sauer, J. C. Org. React. 1946, 3, 108-140.
- Mitzel, T. M. In *Encyclopedia of Reagents for Organic Synthesis*; Paquette, L. A., Ed.; Wiley, New York, 1995, pp. 2929–2932.
- Leahy, J. W. In *Encyclopedia of Reagents for Organic Synthesis*; Paquette, L. A., Ed.; Wiley, New York, 1995, pp. 1714–1719.
- Kappe, T. In *Encyclopedia of Reagents for Organic Synthesis*; Paquette, L. A., Ed.; Wiley, New York, 1995, pp. 1098–1100.
- Harrowven, D. C.; Dennison, S. T. In *Comprehensive Organic Functional Group Trans*formations; Pattenden, G., Ed.; Pergamon Press, 1995, Vol. 3, pp. 525–545.
- Temperley, C. M. In Comprehensive Organic Functional Group Transformations II; Katritzky, A. R.; Taylor, R. J. K., Eds.; Elsevier, Amsterdam, 2005, Vol. 3, pp. 573–603.
- 22. Miller, R.; Abaecherli, C.; Said, A. In *Ullman's Encyclopedia of Industrial Chemistry*, 6th ed.; 2002, Vol. A15, pp. 63–75.
- 23. Yranzo, G. I.; Elguero, J.; Flammang, R.; Wentrup, C. Eur. J. Org. Chem. 2001, 2209–2220.

4 INTRODUCTION

- 24. Roberts, J. C. In *Paper Chemistry*; 2nd ed. Roberts, J. C. Ed.; Chapman and Hall, 1996, pp. 140–160.
- 25. Roberts, J. D.; Sharts, C. M. Org. React. 1962, 12, 1-6.
- Ghosez, L.; O'Donnell, M. J. In *Pericyclic Reactions*; Marchand, A. P.; Lehr, R. E., Eds.; Academic Press, New York, 1977, pp. 79–140.
- 27. Ulrich, H. Cycloaddition Reactions of Heterocumulenes; Academic Press, New York, 1967, pp. 38–109.
- 28. Luknitskii, F. I.; Vovsi, B. A. Russian Chem. Revs. 1969, 38, 487-494.
- 29. Brady, W. T. Tetrahedron, 1981, 37, 2949-2966.
- 30. Hyatt, J. A.; Raynolds, P. W. Org. React. 1994, 45, 159-646.
- 31. Ghosez, L.; Marchand-Brynaert, J. In *Comprehensive Organic Synthesis*; Trost, B. M., Ed.; Pergamon Press, New York, 1991, Vol. 5, pp. 85–122.
- 32. Tidwell, T. T. Angew. Chem. Int. Ed. 2005, 44, 3875-3883.

1 Structure, Bonding, and Thermochemistry of Ketenes

1.1 THEORETICAL AND COMPUTATIONAL STUDIES OF KETENES

Computational methods are increasingly used not only for the elucidation of the structures, energies, and properties of ketenes, but also as an adjunct to a wide variety of ketene studies. Consequently, the use of computational studies appears throughout this book and only a few of the more fundamental studies devoted to computation of ketene properties are noted here, while others are included in appropriate sections throughout the book.

A comprehensive review of theoretical studies on ketene appeared in 1980,¹ and this and other earlier studies^{2,3} will not be repeated here. These provide the fundamental basis for understanding the properties of ketenes. The emphasis is now on the application of computational studies for understanding the properties of different types of ketenes and the myriad reactions that they undergo.

1.1.1 Molecular and Electronic Structure and Energy

The unique properties of ketenes derive from their cumulene structure, with the highest occupied molecular orbital (HOMO) perpendicular to the ketene plane and the lowest unoccupied molecular orbital (LUMO) in the ketene plane (Figure 1.1).⁴ These reveal that ketene has substantial negative charge on oxygen and C_2 , while there is positive charge on C_1 (the carbonyl carbon) in the ketene plane. As a consequence, electrophiles are expected to attack ketenes at oxygen or perpendicular to the ketene plane at C_2 , while nucleophiles approach in the ketene plane at C_1 .



The important resonance structures for ketene are **1a–c**, and these predict the negative charge on C_2 and the positive charge on C_1 . These resonance structures were used by Hannay and Smyth⁵ to explain the dipole moment of ketene (1.45

Ketenes II, by Thomas T. Tidwell

Copyright © 2006 John Wiley & Sons, Inc.



Figure 1.1 Frontier molecular orbitals of ketene (reproduced from reference 4 by permission of the American Chemical Society).

Debye [D]), which is less than those of formaldehyde (2.27 D) or acrolein (3.04 D), suggesting negative charge buildup on C_2 and the possibility of less negative charge on oxygen than in most aldehydes and ketones (Section 2.4). An alternative interpretation by Walsh⁶ was that the sp hybridization of C_1 reduces the negative charge on oxygen, but this did not explain the propensity for electrophilic attack at C_2 .



High-level *ab initio* calculations of the electronic states of ketene^{7–10} have been reported. Calculated vertical excitation energies of ketene agree well with experimental values.⁷ The calculation of the molecular structure and vibrational infrared (IR) spectrum of ketene, as calculated by post-Hartree-Fock and density functional theory methods, show that good agreement with experimental values can be obtained by either method.⁹ *Ab initio* calculations of the electronic structures of $CH_2=C=O$ and C_3O_2 give the dipole polarizabilities of these molecules and the results provide a basis for predicting experimental properties.¹⁰

The structures and stabilities of the C_2H_2O isomers ketene (1), oxirene (2), formylmethylene (3), and hydroxyacetylene (4) and their interconversion by the Wolff rearrangement have been the subject of many computational studies (equation 1).^{11–17} Investigations using (AM1) were particularly concerned with the possibility of competing hydrogen rearrangement¹⁸ and the effects of substituents in aryl derivatives.¹⁹ Calculations of the isomerization of vibrationally excited ketene via oxirene (equation 1) agree with experimental results.²⁰ This reaction has also been studied by density functional theory (DFT) methods, including the role of formylcarbene.²¹ The nature of the stationary point of oxirene has been examined using a variety of density functional methods, and the results with the largest basis sets agree with previous high-level correlated methods that this species is an energy minimum.^{22,23} This problem is discussed further in Section 3.3.

The dissociation of ketene into CH_2 and CO by photochemical or thermal methods (equation 2) is a classical problem in chemistry, particularly regarding the energy distribution and electronic states of the products (Sections 5.2, 5.3).^{24,25}

$$\begin{array}{c} H \\ C = C = 0 \end{array} \xrightarrow{\Delta \text{ or } hv} CH_2 + CO \\ H \\ 1 \end{array}$$

$$(2)$$

Examples of recent studies of ketenes include calculations of the structures, conformations and IR spectra of 5-oxo-2,4-pentadienal (**5**), and it was concluded that of the previously claimed structures, conformer **5a** does not exist as an energy minimum and that only **5b–d** occur.²⁶ Structures of the cyclized forms **6** and α -pyrone (**7**) were also calculated, as well as that for the transition state for the degenerate rearrangement of **5b** by hydrogen migration. Structures of hydroxycumulenones **8** were also calculated.²⁶ Structures, conformations, and vibrational spectra have been calculated for the haloacyl and acetyl ketenes **9**.^{27,28} The dipole polarizabilities of ketene and carbon suboxide (C₃O₂) have also been calculated by *ab initio* methods.¹⁰



The moleculecular structure of vinylketene (10) has been of interest, and the C–C rotational barrier has also been determined by computations (equation 3).²⁹ The dipole moment and ¹³C NMR (nuclear magnetic resonance) spectrum (Section 4.1.2) of 10 have been interpreted as showing the importance of electron delocalization to C₄, as shown in resonance structure 10a.

тт

$$H \xrightarrow{H} C = 0 \xrightarrow{H} H \xrightarrow{H} C = 0 \xrightarrow{H} H \xrightarrow{H} C = 0 \xrightarrow{H} H \xrightarrow{H} C \equiv 0^{+} (3)$$

Molecular mechanics force fields for studies of ketenes have also been presented,³⁰ and the use of natural valence coordinates for ketene computations have been tested.³¹

REFERENCES FOR SECTION 1.1.1

- 1. Dykstra, C. E.; Schaefer, H. F., III. In *Chemistry of Ketenes, Allenes, and Related Compounds*; Patai, S., Ed; Wiley, New York, 1980, pp 1–44.
- 2. Del Bene, J. E. J. Am. Chem. Soc. 1972, 94, 3713-3718.
- 3. Dykstra, C. E.; Schaefer, H. F., III. J. Am. Chem. Soc. 1976, 98, 2689-2695.
- 4. Cossio, F. P.; Ugalde, J. M.; Lopez, X.; Lecea, B. Palomo, C. J. Am. Chem. Soc. 1993, 115, 995–1004.
- 5. Hannay, N. B.; Smyth, C. P. J. Am. Chem. Soc. 1946, 68, 1357-1360.
- 6. Walsh, A. D. J. Am. Chem. Soc. 1946, 68, 2408-2409.
- 7. Szalay, P. G.; Császár, A. G.; Nemes, L. J. Chem. Phys. 1996, 105, 1034-1045.
- 8. Allen, W. D.; Schaefer, H. F., III. J. Chem. Phys. 1986, 84, 2212-2225.
- 9. Kwiatkowski, J. S.; Leszczynski, J. J. Mol. Struct. (Theochem.) 1995, 342, 43-49.
- 10. Stout, J. M.; Dykstra, C. E. J. Am. Chem. Soc. 1995, 117, 5127-5132.
- 11. Tanaka, K.; Yoshimine, M. J. Am. Chem. Soc. 1980, 102, 7655-7662.
- 12. Yoshimine, M. J. Chem. Phys. 1989, 90, 378-385.
- 13. Dykstra, C. E. J. Chem. Phys. 1978, 68, 4244-4247.
- Bouma, W. J.; Nobes, R. H.; Radom. L.; Woodward, C. E. J. Org. Chem. 1982, 47, 1869–1875.
- Scott, A. P.; Nobes, R. H.; Schaeffer, H. F., III; Radom. L. J. Am. Chem. Soc. 1994, 116, 10159–10164.
- Vacek, G.; Galbraith, J. M.; Yamaguchi, Y.; Schaeffer, H. F., III; Nobes, R. H.; Scott, A. P.; Radom. L. J. Phys. Chem. 1994, 98, 8660–8665.
- Wang, F.; Winnik, M. A.; Peterson, M. R.; Csizmadia, I. G. J. Mol. Struct. (Theochem.) 1991, 232, 203–210.
- Enriquez, F.; Lopez-Herrera, F. J.; Quirante, J. J.; Sarabia, F. Theor. Chim. Acta 1996, 94, 13–22.
- 19. Kim, C. K.; Lee, I. Bull. Korean Chem. Soc. 1997, 18, 395-401.
- 20. Gezelter, J. D.; Miller, W. H. J. Chem. Phys. 1995, 103, 7868-7876.
- 21. Wilson, P. J.; Tozer, D. J. Chem. Phys. Lett. 2002, 352, 540-544.
- 22. Mawhinney, R. C.; Goddard, J. D. J. Mol. Struct. 2003, 629, 263-270.
- 23. Delamere, C.; Jakins, C.; Lewars, E. J. Mol. Struct. 2002, 593, 79-91.
- 24. Yamabe, S.; Morokuma, K. J. Am. Chem. Soc. 1978, 100, 7551-7556.
- 25. Allen, W. D.; Schaefer, H. F., III. J. Chem. Phys. 1988, 89, 329-344.
- 26. Birney, D. M. J. Org. Chem. 1996, 61, 243-251.
- 27. Badawi, H. M.; Förner, W.; Al-Saadi, A. J. Mol. Struct. 2001, 561, 103-119.
- 28. Badawi, H. M.; Al-Saadi, A.; Förner, W. Spectrochim. Acta. A 2002, 58, 33-53.
- 29. Badawi, H. M.; Förner, W.; Al-Saadi, A. J. Mol. Struct. 2001, 535, 183-197.

30. Stewart, E. L.; Bowen, J. P. J. Comput. Chem. 1992, 13, 1125-1137.

31. Fogarasi, G.; Zhou, X.; Taylor, P. W.; Pulay, P. J. Am. Chem. Soc. 1992, 114, 8192-8201.

1.1.2 Theoretical and Computational Studies of Ketene Reactions

The mechanisms and energetics of many different reactions of ketenes have been analyzed by computational and theoretical methods and provide useful insights into the understanding of ketene reactivity. Many of these studies are included in later sections dealing with the preparation and properties of ketenes, but a brief survey of the approaches to some of the most common ketene reactions is presented here.

The dimerization of ketene (1) has been studied using *ab initio* molecular dynamics calculations, 1–5 which lead to the conclusion in the most recent calculation¹ that the β -lactone dimer **3** is the most stable, (equation 1),¹ with a free energy change of –5.5 kcal/mol, while formation of 1,3-cyclobutanone **2** has a free energy change of –4.5 kcal/mol. Previous calculations predicted a lower barrier for forming **3**,^{2–5} in agreement with experiment. Dynamics calculations, however, predicted a lower barrier for formation of **2**, and appear to be less reliable in this regard.¹

$$\overset{H}{\underset{H}{\longrightarrow}} c=0 \longrightarrow \overset{O}{\underset{O}{\longrightarrow}} + \overset{O}{\underset{CH_{2}}{\longrightarrow}} 0 \qquad (1)$$

The equilibrium between cyclobuten-3,4-dione 4 and 1,2-bisketene 5, which is shown by theory and experiment to have a twisted, almost perpendicular conformation, is considered in Section 4.9 (equation 2). Ring opening of cyclobutenone 6 to vinylketene 7 (equation 3) is discussed in Section 5.4.4, as well as the opening of the corresponding aza- and oxa-cyclobutenones to form imidoylketene (8) and formylketene (9), respectively.



Computations using DFT have been used to examine the [4+2] and [2+2] cycloadditions of vinylketene (7), imidoylketene (8), and formylketene (9) with formaldimine (10).⁶ For the [2+2] cycloaddition, both stepwise mechanisms involving zwitterionic intermediates and transition states for concerted reactions were located, and it was suggested that multiple transition states may be involved. For the [4+2] cycloadditions, both stepwise and concerted pseudopericyclic pathways were also calculated.⁶

$$\begin{array}{cccc} CH_2 & HN = & O = \\ \hline C = O & C = O & CH_2 = NH \\ \hline 7 & 8 & 9 & 10 \end{array}$$

A theoretical study of the conversion of acetyl cyanide, acetyl isocyanide and acetyl fluoride (**11a–c**) to ketene and other products has also been reported (equation 4).^{7,8}

$$CH_3 \xrightarrow{O} \xrightarrow{-HX} \xrightarrow{H} C=0$$
11 (X = CN, NC, F) 1
(4)

A DFT computational study of the [2+2] cycloaddition of ketene with CH₂=CH₂, CH₂=O, and (C₅H₅)Re(O)₂=O discusses the role of orbital interactions, electrostatics, and repulsions due to the Pauli principle on the activation energies.⁹

Protonation of ketenes and bisketenes has been the subject of a number of studies.^{10–15} Protonation of ketene (**1**) at C₂ forming the acylium ion **12** is strongly favored relative to protonation at C₁ forming α -formyl carbocation **13**, but for substituted ketenes **14**, strongly electron-donating substituents R enhance the possibility of formation of **13** (equation 5).^{13,15} Protonation at oxygen forming **15** is not as favorable as protonation at C₂ in the cases studied. For vinylketene protonation at C₄ forming the allylic cation **16** is the lowest energy pathway (equation 6).¹¹

 $= \underbrace{C=0}_{H^*} \xrightarrow{H^*}_{CH_3} \xrightarrow{CH_3}_{C=0} \xrightarrow{CH_3}_{C=0} \xrightarrow{CH_3}_{C=0} (6)$

Proton abstraction from ketene (1) forming the ynolate anion 17 has been the subject of computational study, and the linear structure for 17 with negative charge

delocalized between oxygen and C_2 is favored over a bent structure **18** that could have negative charge concentrated on C_2 (equation 7).^{16,17}

$$\stackrel{H}{\underset{H}{\longrightarrow}} C=O \xrightarrow[-BH^+]{B} H-C\equiv C-O^- H^{\overline{C}=C=O}$$

$$1 \qquad 17 \qquad 18$$

$$(7)$$

The hydration of ketene has been the subject of numerous computational studies, and reaction with two water molecules through a cyclic transition state **19** forming the acid enol **20** is favored over addition of a single H₂O molecule or addition of two H₂O molecules to the C–C double bond, forming acetic acid directly (equation 8).^{18–21} This is the accepted mechanism for hydration of aldehydes and ketones forming hydrates in both experimental and theoretical studies.^{22–25}



Reaction of formylketene $(21)^{26}$ with H₂O was predicted to occur with coordination of the H₂O to the formyl oxygen through pseudopericyclic transition state **22** forming **23** (equation 9).^{26,27} Imidoylketene **24** was found to behave similarly²⁷



The reactions of LiH, and LiCH₃ with ketene are predicted to occur through initial lithium coordination to the ketenyl oxygen and then bond formation to the carbonyl carbon.²⁸ The addition of the lithium enolate of acetaldehyde was predicted to occur similarly, with C–O bond formation giving **25**, even though C–C bond formation gives a more stable product (equation 10).²⁹ This result was confirmed experimentally (Section 5.5.2.2)

The structure of mesitylphenylketene **27** has been calculated at the HF/3–21G level and has been found to have a twisted structure of the mesityl perpendicular

to the ketene plane, with the bond distances (**27a**) and bond angles (**27b**) shown.³⁰ The transition state for addition of MeLi was also calculated, and the preferential attack from the side of the mesityl group was explained by the perpendicular conformation adopted by this substituent to minimize steric interactions.³⁰



Other theoretical studies include the possible formation of metal-complexed ketenes from vinyl carbene complexes (Section 3.5),³¹ radical cyclization of ketenes (Section 5.7),³² and the reaction of ketenes with diazomethanes (Section 5.10).³³

The structure of the complex of ketene with $Cr(CO)_4OH_2$ was calculated using the B3LYP(6-31)G⁺) basis set for C, H, and O, and a double (ζ) basis set for Cr (Figure 3.1, Section 3.5).¹⁵

REFERENCES FOR SECTION 1.1.2

- 1. Kelly, E.; Seth, M.; Ziegler, T. J. Phys. Chem. A 2004, 108, 2167-2180.
- 2. Seidl, E. T.; Schaefer, H. F. J. Am. Chem. Soc. 1991, 113, 5195-5200.
- 3. Seidl, E. T.; Schaefer, H. F. J. Am. Chem. Soc. 1990, 112, 1493-1499.
- 4. Salzner, U.; Bachrach, S. M. J. Am. Chem. Soc. 1994, 116, 6850-6855.
- 5. Seidl, E. T.; Schaefer, H. F. J. Phys. Chem. 1992, 96, 657-661.
- 6. Zhou, C.; Birney, D. M. J. Am. Chem. Soc. 2002, 124, 5231-5241.
- 7. Sumathi, R.; Nguyen, M. T. J. Phys. Chem. A 1998, 102, 412-421.
- 8. Nguyen, L. T.; Sumathi, R.; Nguyen, M. T. Phys. Chem. Chem. Phys. 1999, 1, 1013–1024.
- 9. Deubel, D. V. J. Phys. Chem. A 2002, 106, 431-437.
- 10. Lien, M. H.; Hopkinson, A. C. J. Org. Chem. 1988, 53, 2150-2154.
- Leung-Toung, R.; Peterson, M. R.; Tidwell, T. T.; Csizmadia, I. G. J. Mol. Struct. 1989, 183, 319–330.
- 12. Nobes, R. H.; Bouma, W. J.; Radom, L. J. Am. Chem. Soc. 1983, 105, 309-314.
- Armitage, M. A.; Higgins, M. J.; Lewars, E. G.; March, R. E. J. Am. Chem. Soc. 1980, 102, 5064–5068.
- Allen, A. D.; Ma, J.; McAllister, M. A.; Tidwell, T. T.; Zhao, D.-C. J. Chem. Soc., Perkin Trans. 2, 1995, 847–851.
- 15. White, J. C.; Cave, R. J.; Davidson, E. R. J. Am. Chem. Soc. 1988, 110, 6308-6314.
- 16. Smith, B. J.; Radom, L.; Kresge, A. J. J. Am. Chem. Soc. 1989, 111, 8297-8299.

- 17. Hayes, R. N.; Sheldon, J. C.; Bowie, J. H. Aust. J. Chem. 1985, 38, 355-362.
- 18. Skanke, P. N.; J. Phys. Chem. 1992, 96, 8065-8069.
- 19. Nguyen, M. T.; Raspoet, G. Can. J. Chem. 1999, 77, 817-829.
- 20. Guthrie, J. P. Can. J. Chem. 1999, 77, 934-943.
- 21. Cannizzaro, C. E.; Houk, K. N. J. Am. Chem. Soc. 2004, 126, 10992-11008.
- 22. Bell, R. P. Adv. Phys. Org. Chem. 1966, 3, 1-29.
- 23. Wolfe, S.; Kim, C.-K.; Yang, K.; Weinberg, N.; Shi, Z. J. Am. Chem. Soc. 1995, 117, 4240–4260.
- 24. Guthrie, J. P.; Pitchko, V. J. Am. Chem. Soc. 2000, 122, 5520-5528.
- 25. Guthrie, J. P. J. Am. Chem. Soc. 2000, 122, 5529-5538.
- 26. Allen, A. D.; McAllister, M. A.; Tidwell, T. T. Tetrahedron Lett. 1993, 34, 1095-1098.
- 27. Ham, S.; Birney, D. M. J. Org. Chem. 1996, 61, 3962-3968.
- 28. Leung-Toung, R.; Schleyer, P. v. R.; Tidwell, T. T., unpublished results.
- 29. Leung-Toung, R.; Tidwell, T. T. J. Am. Chem. Soc. 1990, 112, 1042-1048.
- 30. Yamataka, H.; Aleksiuk, O.; Biali, S. E.; Rappoport, Z. J. Am. Chem. Soc. 1996, 118, 12580–12587.
- Arrieta, A.; Cossío, F. P.; Fernández, I.; Gómez-Gallego, M.; Leccea, B.; Mancheño, M. J.; Sierra, M. A. J. Am. Chem. Soc. 2000, 122, 11509–11510.
- 32. Yamamoto, Y.; Ohno, M.; Eguchi, S. J. Org. Chem. 1996, 61, 9264-9271.
- 33. Sorensen, T. S.; Sun, F. J. Am. Chem. Soc. 1997, 119, 11327-11328.

1.1.3 Substituent Effects on Ketenes

The effects of substituents on the stability of ketenes were studied by *ab initio* calculations of the structures and energies of the ketenes and isodesmic comparison with alkenes (equation 1).^{1–3} The energy change (kcal/mol) in the conversion of a substituted ketene **1** and propene (**2**) to methylketene **3** and the substituted alkene **4** was taken as the stabilization energy (*SE*) of that substituent on ketenes (Table 1.1). The choice of CH₃ as the standard of comparison is arbitrary, but emphasizes that by this method many alkyl, alkenyl, and aryl groups (Me, cyclopropyl, CF₃, vinyl, phenyl) are calculated to have similar effects, and the *SE* values are small. The results are consistent using HF/3–21G, HF/6–31G*, and MP2/6–31G* level calculations, which produces confidence that the conclusions will not be affected by higher-level calculations.

From a qualitative viewpoint, the *SE* values (Table 1.1) are consistent with experience, for example in the known stability of silyl-substituted ketenes and the high reactivity of haloketenes, especially fluoroketenes. This suggested a correlation with substituent electronegativity, and indeed, a reasonable correlation

R	ΔE (MP2)	ΔE (HF)	$\chi_{\rm BE} a$
Н	3.9	3.3	2.20
Li	30.1	27.9	1.00
BeH	19.6	18.1	1.47
BH ₂	18.7	16.8	1.93
CH ₃	0.0	0.0	2.56
NH ₂	-9.3	-7.2	3.10
OH	-14.5	-14.2	3.64
F	-16.9	-17.2	4.00
Na	34.3	29.2	1.00
MgH	23.2	21.9	1.33
AlH ₂	20.1	18.7	1.62
SiH ₃	11.5	10.9	1.91
PH ₂	6.3	6.3	2.17
SH	0.2	1.3	2.63
Cl	-7.7	-7.5	3.05
CF ₃	—	-0.1	2.68
<i>c</i> -Pr	—	-1.8	2.56
$CH=CH_2^{b}$	—	-0.2	2.61
$CH=O^{c}$	5.4	3.6	2.60
Ph	—	0.9	2.58
CO ₂ H	—	4.7	2.66
C≡CH	0.5	0.2	2.66
CN	0.6	-0.4	2.69
$CH=C=CH_2$		-1.6	
CH=C=O	—	-5.3^{d}	2.58
N=O	—	-0.1	3.06
NO ₂	1.9	-3.2	3.22
$N \equiv C^{e}$	—	-8.5	3.30

TABLE 1.1 Comparative Calculated Stabilization Energies (kcal/mol) (MP2/6-31G*// MP2/6-31G* and HF/6-31G*//HF/6-31G*) for Ketenes RCH=C=O for the Isodesmic Reaction of Equation $1^{1,3}$

^aGroup electronegativity from refs. 4, and 5, except Pauling electronegativity for H. ^bTranscoid.

^cCiscoid.

^dFor the process (CH=C=O)₂ + (CH₂=CH)₂ \rightarrow 2 CH₂=CHCH=C=O.

^eIsocyano.

(Figure 1.2) was found with the group electronegativities χ_{BE} provided by Boyd, Boyd, and Edgecombe (equation 2).^{4,5} Thus, ketenes are expected to be stabilized by electropositive substituents, which are capable of σ - π electron donation from the R–C₂ bond to the carbonyl group, and destabilized by electronegative groups. Electronegative atoms such as O, N, and the halogens also have lone pairs of electrons, which have a repulsive interaction with C₂ of ketenes, which bears a high negative charge, as found by the computations, and as seen in the remarkably high field ¹³C



Figure 1.2 Plot of MP2/6–31G* calculated stabilization energies for substutued ketene versus substituent group electronegativities (X_{BE}).

NMR shifts of this carbon (Section 2.1), and as expected by the resonance structure **1c**.

$$SE_{(MP2)} = -17.0_{\chi BE} + 46.4 \quad (r = 0.97)$$

$$\overset{R}{\xrightarrow{}}_{H} C_{2} = C_{1} = 0 \quad \longleftrightarrow \quad \overset{R}{\xrightarrow{}}_{H} \stackrel{c}{\xrightarrow{}}_{C} - 0^{-} \quad \longleftrightarrow \quad \overset{R}{\xrightarrow{}}_{H} C \equiv 0^{+}$$

$$1a \qquad 1b \qquad 1c \qquad (2)$$

As shown in Figure 1.2, there are some deviations from the correlation, and those for the substituents Na, BH₂ and CH=O are all significantly above the correlation line (obs: 34.3, 18.7, and 5.4, respectively, calculated from equation 2: 29.4, 13, and 2.2, respectively), indicating greater stabilization than expected from these substituents on the basis of their electronegativities. The extra stabilization of sodioketene (**5**) can be understood as arising from the bridged structure **5** calculated for this ketene (bond distances in Å shown), while boranylketene (**6**) and formylketene (**7**) are stabilized by conjugation.



16 STRUCTURE, BONDING, AND THERMOCHEMISTRY OF KETENES

The nonplanar geometries of hydroxyketene (8) and aminoketene (9) are another structural feature revealed by the computations.^{1,3} These ketenes with n- π donor groups adopt these conformations to minimize repulsive interactions between the π -electrons concentrated at C₂ and the lone pairs. The suggestion of a bonding interaction between the nitrogen lone pair and the carbonyl carbon in 9 was discounted, as the calculated bond order between these atoms is negative.¹



The stabilization of silylketenes **10** and others with electropositive substituents may be ascribed to hyperconjugative electron donation, as shown in **10a** (Section 4.5). This is an example of the β -silicon effect, as observed in β -silyl carbocations.⁶



A limited study using computations at the G3(MP2)//B3LYP level for an isodesmic comparison of ketenes with substituted methanes was carried out, which included disubstituted ketenes (equation 3).⁷ The results are given in Table 1.2. This treatment compares the substituent effects for ketenes to those for substituted methanes and also includes substituent effects on disubstituted ketenes.

Because of the different references used, the results in Tables 1.1 and 1.2 are not directly comparable. The magnitudes of the ΔE values are larger for the comparisons in Table 1.1, and this is evidently due to the difference of the substituent effects in RCH=CH₂ as compared to RCH₃. In Table 1.1 the magnitude of the F/OH and F/Cl differences are 2.4 and 11.6 kcal/mol, respectively, compared to 6.4 and 6.2 kcal/mol, respectively, in Table 1.2. Thus, while the magnitude of the substituent effects is F > OH > Cl for both sets, there are quantitative differences. More striking are the effects in the disubstituted ketenes (Table 1.2), which, rather

Ketene	ΔE (kcal/mol)
HOCH=C=O	-5.6
MeOCH=C=O	-6.1
ClCH=C=O	-5.8
ClC(OH) = C = O	-4.3
ClC(OMe)=C=O	-3.7
FCH=C=O	-12.0
FC(OH) = C = O	-5.4
FC(OMe) = C = O	-5.4

TABLE 1.2 Calculated Stabilization Energies (kcal/ mol) (G3MP2//B3LYP) for Ketenes $RR^1CH=C=O$ for the Isodesmic Reaction of Equation 3^7

than being additive, are actually less for the disubstitued cases. This suggests that stabilizing effects are present when there are two electronegative substituents on C_2 of ketene. Further study to test the generality of this effect is warranted.

Substituent effects in ketenylketenes are considered in Section 4.9 and those in allenylketenes in Section 4.18.⁸ Computational studies of antiaromatic destabilization in trifulvenone **14** and heptafulvenone **16**, and of aromatic stabilization of pentafulvenone **15**, are discussed in Section 4.1.10.^{9,10}



REFERENCES FOR SECTION 1.1.3

- 1. Gong, L.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1991, 113, 6021-6028.
- Allen, A. D.; Andraos, J.; Kresge, A. J.; McAllister, M. A.; Tudwell, T. T. J. Am. Chem. Soc. 1992, 114, 1878–1879.
- 3. McAllister, M. A.; Tidwell, T. T. J. Org. Chem. 1994, 59, 4506-4515.
- 4. Boyd, R. J.; Edgecombe, K. E. J. Am. Chem. Soc. 1988, 110, 4182-4186.
- 5. Boyd, R. J.; Boyd, S. L. J. Am. Chem. Soc. 1992, 114, 1652-1655.
- 6. Brook, M. A. Silicon in Organic, Organometallic, and Polymer Chemistry; Wiley, New York, 2000.
- 7. Scott, A. P.; Platz, M. S.; Radom, L. J. Am. Chem. Soc. 2001, 123, 6069-6076.
- Huang, W.; Fang, D.; Temple, K.; Tidwell, T. T. J. Am. Chem. Soc. 1997, 119, 2832–2838.
- 9. McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1992, 114, 5362-5368.

 Najafian, K.; Schleyer, P. v. R.; Tidwell, T. T. Chem. Biomol. Chem. 2003, 1, 3410– 3417.

1.2 MOLECULAR STRUCTURE DETERMINATIONS

Molecular structures of many ketenes¹ have been examined experimentally by microwave,^{2–15} electron diffraction,^{16–18} and X-ray techniques,^{19–27} and many more by computations at various theoretical levels. Experimental data for some simple ketenes are given in Table 1.3, with calculated values for comparison.

Structures have been determined using microwave spectroscopy for $CH_2=C=O$ (1),²⁻⁶ NCCH=C=O (2),⁷ FCH=C=O (3),^{8,9} ClCH=C=O (4),¹⁰ CH₃CH=C=O (5),¹¹ and BrCH=C=O (6).¹² The microwave spectrum of vinylketene (7) was measured, but only the dipole moment was obtained from the data.¹³ The microwave spectrum of $(CH_3)_2C=C=O$ (8) was measured and the structure obtained with the assumption of the same C=O bond length as in CH₃CH=C=O.¹⁴ The ketene Cl₂C=C=O (9) had a negligible dipole moment and the microwave spectrum could not be measured,¹⁵ but the structure was determined using electron diffraction.¹⁶ A molecular structure for thioketene (CH₂=C=S, 10) was also determined.¹⁷ The experimental bond distances, bond angles, and dipole moments obtained are summarized in Table 1.3, along with comparative computational results.

Gas phase pyrolysis of the acyl chloride with appropriate isotopic labeling at 750° C gave cyanoketene (2) with ²H, ¹⁵N, ¹³N, carbonyl ¹³C, or ¹⁸O labeling for determination of the molecular structure by microwave spectroscopy.⁷ The molecule had a dipole moment of 3.542(15) D, and the molecular structure was in reasonable agreement with the calculated structure (Table 1.3). See also Section 4.1.3.



The gas phase molecular structure of $(CF_3S)_2C=C=O$ (11) has been determined by electron diffraction and displays C₂ symmetry, with the CF₃–S bonds perpendicular above and below the molecular plane.¹⁸ This geometry is that predicted by molecular orbital calculations, and minimizes repulsion between the lone pairs on sulfur and the electron-rich HOMO at C₂ of the ketene.



ents	
Mom	
ipole	
und D	
Jeg), {	
gles (I	
nd Ang	
s, Bon	
stance	
id Di	
s) Boi	
othese	
Parei	
1G*)(
IF/6-3	
G*//E	
F/6-31	
IH) be	
culate	
d Cal	
ıtal ar	
rimer	2
Expe	P ¹ C.1
1.3	noc D
ABLE	. Kata
E	č

of Ketene	S KK ⁻ C2										
R	\mathbb{R}^{1}	C=0	$C_1 = C_2$	C_2-R_1	C_2 -R	OC_1C_2	RC_1C_2	RC_2R_1	$R_1C_2C_1$	μ (Debye)	Ref
Н	Η	1.164	1.3165	1.0800	1.0800	180.0	119.01	121.98	119.01	1.41	ю
		1.1626	1.3147	1.0905	1.0905	180.0	118.27	123.46	118.27		0
		(1.145)	(1.306)	(1.075)		(180.0)	(119.3)	(121.4)	(119.3)	(1.63)	22
		(1.1681)	(1.3166)	(1.0745)	(1.0745)	(180.0)	(119.4)	(121.2)	(119.38)		2^a
н	Η	1.167	1.317	1.080	1.356	178.0	119.5	118.2	122.3	1.29	8,9
		(1.147)	(1.308)	(1.067)	(1.339)	(177.9)	(120.1)	(117.4)	(122.5)	(1.59)	22
		(1.181)	(1.323)	(1.074)	(1.363)	(178.1)	(120.0)	(117.7)	(122.3)		$^{q}6$
CI	Η	1.161	1.316	1.082	1.726	180.0	120.9	119.3	119.8	1.2	10
		(1.142)	(1.308)	(1.068)	(1.731)	(179.2)	(120.5)	(118.7)	(120.8)	(1.39)	22
Br	Η	1.162	1.316	1.082	1.880	180.0	118.5	120.6	120.9		12
CH_3	Η	1.171	1.306	1.083^c	1.518	180.5	122.6^{d}	123.7	113.7	1.79	11
		(1.149)	(1.305)	(1.074)	(1.512)	(180.0)	(123.2)	(120.8)	(116.0)	(1.99)	22
NC	Η	1.148	1.334		1.434	176.5	118.5	119.9		3.542(15)	٢
		1.133	1.321	(1.072)	(1.424)	(178.7)		(121.3)	(120.6)	(3.72)	22
CH=CH ₂	Н									0.97	12
										(1.20)	22
CI	ū	1.160	1.299	1.726	1.726	180.0	120.4	119.2	120.4		16
		(1.161)	(1.306)	(1.726)	(1.726)	(180.0)	(120.7)	(118.6)	(120.7)	(0.07)	16a
CH_3	CH_3	1.171	1.300	1.514		180.0	120.6	118.8	120.6		14
CF_3S	CF_3S	1.18	1.33					127.3			18
^a MP3/6–31 ^b MP2/4–31 ^c For CH ₃ C	G**. G**. 7-H 1.083 (((1.084), 1.11.									
"H-C-H 1	09.9, 108.8;	H-C-C ₂ 111.	.1 (111.4).								

Structures have been obtained by X-ray studies of dimesitylketene (12) and bis(3,5-dibromomesityl)ketene (13).¹⁹ Both structures have propeller conformations of the aryl groups, and in 12 these are nonequivalent, with dihedral angles of 48.8° and 56.8° , while in 13 these are equivalent, with dihedral angles of 56.8° . The structure of mesitylphenylketene 14 has been calculated at the HF/3–21G level, with the indicated bond distances and bond angles (parentheses).²⁰



Structures determined by X-ray for the aliphatic ketene **15**,²¹ the alkenylketene **16** with a chromiun tricarbonyl complexed to the side chain,²² and the tetraketene **17**²³ have also been reported. There are systematic differences in the ketenyl C=C bond lengths determined by X-ray, which are all between 1.25 and 1.29 Å, and those obtained by microwave spectroscopy or electron diffraction reported in Table 1.3, which are all greater than 1.30 Å, and those calculated at the MP2/6–31G* level, which are all also greater than 1.30 Å, with the exception of those to the highly electropositive Li, Na, and MgH substituents, none of which are known experimentally.^{24,25}



The X-ray structure of ketene **18**, derived from the dimerization of dipivaloylketene,²⁶ revealed that the ketenyl C=O and C=C bond lengths of 1.13 and 1.35 Å are distinctly shorter and longer, respectively, than the average of those of four previous ketene X-ray structure determinations of 1.18 and 1.28 Å, respectively, but are very similar to recently determined values of 1.14 and 1.34 Å for a (metaloxy)silylketene **19**.²⁷ The shorter C=O and longer C=C bonds for **18** are consistent with acyl conjugation of the ketene as proposed for other acylketenes (**20**; see also Section 4.1.4), but the C–C bond length of the silylketene was not shortened, as could be expected by resonance structure **21**.



A molecular structure of $CH_3CH=C=O$ (5) calculated using the MP2/6–31G level (parentheses) shows no better agreement with experimental values than do previous HF/6–31G* calculations [brackets].²⁹

[1.305] (1.301) (1.321) (1.302) (1.312) (1.306) (1.312) (1.306) (1.512) (1.509) (1.518) (C=C=O H I.171) (1.184) (1.1

Carbon suboxide C_3O_2 (O=C=C=C=O, **22**) has a quasilinear structure with a very small energy barrier for bending around the central carbon.^{29–31} The structure of crystalline C_3O_2 has been determined by X-ray.³²

Other molecular structures of ketenes have been determined by X-ray for the silyl stabilized ketene **23**,³³ and for a cobalt-complexed ketene.³⁴



REFERENCES FOR SECTION 1.2

- Runge, W. In *The Chemistry of Ketenes, Allenes, and Related Compounds;* Patai, S., Ed.; Wiley, New York, 1980, p.45–95.
- Brown, R. D.; Godfrey, P. D.; McNaughton, D.; Pierlot, A. P.; Taylor, W. H. J. Mol. Spec. 1990, 140, 340–352.
- 3. Duncan, J. L.; Munro, B. J. Mol. Struct. 1987, 161, 311-319.
- 4. Johnson, H. R.; Strandberg, M. W. P. J. Chem. Phys. 1952, 20, 687-695.
- 5. Cox, A. P.; Thomas, L. F.; Sheridan, J. Spectrochim. Acta 1959, 15, 542-543.
- 6. Moore, C. B.; Pimentel, G. C. J. Chem. Phys. 1963, 38, 2816-2829.

22 STRUCTURE, BONDING, AND THERMOCHEMISTRY OF KETENES

- 7. Hahn, M.; Bodenseh, H.-K.; Ferner, M. J. Mol. Spectrosc. 2004, 223, 138-147.
- 8. Brown, R. D.; Godfrey, P. D.; Wiedenmann, K. H. J. Mol. Spec. 1989, 136, 241-249.
- 9. Brown, R. D.; Godfrey, P. D.; Kleibömer, B. Chem. Phys. 1986, 105, 301-305.
- 10. Gerry, M. C. L.; Lewis-Bevan, W.; Westwood, N. P. C. J. Chem. Phys. 1983, 79, 4655-4663.
- Bak, B.; Christiansen, J. J.; Kunstmann, K.; Nygaard, L.; Rastrup-Andersen, J. J. Chem. Phys. 1966, 45, 883–887.
- 12. Westwood, N. P. C.; Lewis-Bevan, W.; Gerry, M. C. L. J. Mol. Spectrosc. 1989, 136, 93-104.
- 13. Brown, R. D.; Godfrey, P. D.; Woodruff, M. Aust. J. Chem. 1979, 32, 2103-2109.
- 14. Nair, K. P. R.; Rudolph, H. D.; Dreizler, H. J. Mol. Spectrosc. 1973, 48, 571-591.
- 15. Gerry, M. C. L.; Lewis-Bevan, W.; Westwood, N. P. C. Can. J. Chem. 1985, 63, 676-677.
- Rozsondai, B.; Tremmel, J.; Hargittai, I.; Khabashesku, V. N.; Kagramanov, N. D.; Nefedov, O. M. J. Am. Chem. Soc. 1989, 111, 2845-2849.
- 16a. Davidovics, G.; Monnier, M.; Allouche, A. Chem. Phys. 1991, 395-403.
- 17. Duncan, J. L.; Jarman, C. N. Struct. Chem. 1990, 1, 195-199.
- Korn, M.; Mack, H.-G.; Praas, H.-W.; Vedova, C. O. D.; Oberhammer, H. J. Mol. Struct. 1995, 352/353, 145–151.
- 19. Biali, S. E.; Gozin, M.; Rappoport, Z. J. Phys. Org. Chem. 1989, 2, 271-280.
- 20. Yamataka, H.; Aleksiuk, O.; Biali, S. E.; Rappoport, Z. J. Am. Chem. Soc. 1996, 118, 12580-12587.
- 21. Schaumann, E.; Harto, S.; Adiwidjaja, G. Chem. Ber. 1979, 112, 2698-2708.
- 22. Schubert, U.; Dötz, K. H.; Cryst. Struct. Commun. 1979, 8, 989-994.
- 23. Allen, A. D.; Lough, A. J.; Tidwell, T. T. Chem. Commun. 1996, 2171-2172.
- 24. Gong, L.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1991, 113, 6021-6028.
- 25. McAllister, M. A.; Tidwell, T. T. J. Org. Chem. 1994, 59, 4506-4515.
- 26. Kappe, C. O.; Evans, R. A.; Kennard, C. H. L.; Wentrup, C. J. Am. Chem. Soc. 1991, 113, 4234-4237.
- Radu, N. S.; Engeler, M. P.; Gerlach, C. P.; Tilley, T. D.; Rheingold, A. L. J. Am. Chem. Soc. 1995, 117, 3621–3622.
- 28. McKee, M. L.; Radom, L. Org. Mass. Spec. 1993, 28, 1238-1244.
- 29. Jensen, P.; Johns, J. W. C. J. Mol. Spectrosc. 1986, 118, 248-266.
- 30. Ohshuma, Y.; Yamamoto, K.; Satoshi, K. Acta Chem. Scand. Ser. A 1988, 42, 307-317.
- 31. Koput, J. Chem. Phys. Lett. 2000, 320, 237-244.
- 32. Ellern, A.; Drews, T.; Seppelt, K. Z. Anorg. Allg. Chem. 2001, 627, 73-76.
- 33. Hörnig, H.; Walther, E.; Schubert, U. Organometallics 1985, 4, 1905-1906.
- 34. Wulff, W. D.; Gilbertson, S. R.; Springer, J. P. J. Am. Chem. Soc. 1986, 108, 520-522.

1.3 THERMOCHEMISTRY OF KETENES

The determination of the thermodynamic properties of ketenes has been of continuing interest,¹⁻¹⁴ and these properties have been obtained by both experimental and computational methods. There have been significant differences in values from different sources, but resolution of these differences has apparently been achieved. Preferred values from the most recent reports for ketene, methylketene, and dimethylketene are given in Table 1.4, and a compilation of some of the other existing data is presented in Table 1.5.¹⁻²¹ The computational value for

Ketene	$\Delta H^{\rm o}{}_{\rm f}$	PA (kJ/mol) [kcal/mol]	GB (kJ/mol) [kcal/mol]	Ref
$CH_2=C=O$	-54±5 [-12.9]	822.9 ± 3.4 [195]	790.1±1.5 [188]	2, 5, 22
CH ₃ CH=C=O	$-66.9 [-16.0]^{21}$	839.8±3.6 [200.7]	809.6 ± 3.3 [193.5]	1, 2, 5, 21
		845.4 ± 4.8		7
$(CH_3)_2C = C = O$	-92	855 [204]		4, 6

 TABLE 1.4 Thermodynamic Values for Simple Ketenes (kJ/mol; kcal/mol in brackets)

 TABLE 1.5 Experimental and Theoretical Heats of Formation and Proton Affinities

 (kJ/mol; kcal/mol in brackets)

	ΔH_{f298}^{o}	PA	Ref
CH ₂ =C=O	-52		6
	-54 ± 5 [-12.9]	822.9 ± 3.4 [195]	2, 5
	-47.7(-11.4)		15
	(-14.7)		16
	-100 ± 10		18
$CH_2 = C = O^{+\bullet}$	875 [209]		4,6
$CH_3C(+)=O$	658		6
$CH_3CH=C=O$	-68 [-16.3]		4,6
	$-97{\pm}5$		1, 2
		845 ± 4.8	7
		834.1	20
	-95.5 [-22.8]		5
	-86.6 [-20.6]	853	12
	-65.1 ± 10		3
	-105 [-25]		17, 19
$CH_3CH=C=O^{\pm \bullet}$	795 [190]		6
	797.0		3, 4
$CH_3CH_2C(+)=0$	618 [148]		6
$(CH_3)_2C=C=O$	-92 [22]	855 (204)	4, 6
	-137 ± 5 [-32.7]		5
	-155 [-37]		19
$(CH_3)_2C=C=O^{+\bullet}$	726 (174)		4
	723 [173]		6
$(CH_3)_2CHC(\pm)=0$	583 [139]		6
NCCH=C=O	$-100.4 \ [-24 \pm 5]$		10
$NCCH_2C(+)=O$	828 [198±3]		10
$NCCH_2C(\bullet)=O$	133.8 [32 \pm 1]		10
NCCH=C= $O^{+\bullet}$	$1061 \ [253 \pm 3]$		10
PhCH=C=O	25.1 [6] (est)		19
$Ph_2C=C=O$	121 [29] (est)		19
$CH_2 = CHCH = C = O$	16.7 [4]		14
$CH_2 = CHCH = C = O^{+\bullet}$	816 [195]		14

	$\Delta H_{\rm f}^{298}$	ΔS^{298}		ΔH_f^{298}	ΔS^{298}
Ketene	(kcal/mol)	(cal/mol K)	Ketene	(kcal/mol)	(cal/mol K)
$CH_2 = C = O(1)$	-11.34	57.47	MeOC(OH)=C=O (17)	-68.94	80.20
MeCH=C=O(2)	-15.14	68.20	MeC(OH) = C = O(18)	-44.90	75.73
EtCH=C=O(3)	-20.67	77.61	MeC(OMe) = C = O(19)	-40.15	84.34
i-PrCH=C=O (4)	-28.57	84.05	EtC(OH)=C=O (20)	-50.42	84.85
<i>t</i> -BuCH=C=O (5)	-37.75	87.66	<i>i</i> -PrC(OH)=C=O (21)	-58.29	91.63
$Me_2C = C = O(6)$	-21.73	74.92	<i>t</i> -BuC(OH)=C=O (22)	-68.88	95.04
MeCEt=C=O(7)	-27.97	85.84	EtC(OMe)=C=O(23)	-45.77	94.10
i-PrCMe=C=O (8)	-35.76	92.13	i-PrC(OMe)=C=O (24)	-54.08	100.17
t-BuCMe=C=O (9)	-44.87	95.68	t-BuC(OMe)=C=O (25)	-64.68	103.36
$Et_2C = C = O(10)$	-34.66	93.66	$(HOCH_2)CH=C=O(26)$	-51.60	77.65
i-PrCEt=C=O (11)	-42.52	101.41	$(HOCH_2)CMe=C=O(27)$	-58.53	85.64
t-BuCEt=C=O (12)	-51.76	104.35	(HOCHMe)CH=C=O (28)	-61.18	85.32
HOCH=C=O (13)	-35.82	67.30	(HOCHMe)CMe=C=O (29)	-68.83	94.02
MeOCH=C=O(14)	-31.33	75.68	$(HOCMe_2)CH=C=O(30)$	-72.47	88.83
$(HO)_2C = C = O(15)$	-73.22	70.23	$(HOCMe_2)CMe=C=O(31)$	-79.94	97.06
$(MeO)_2C = C = O$ (16)	-67.74	86.97			
0)		$0 \ge C \bigvee_{K=0}^{Me} C \ge 0$		
C C C C	-15.18	85 48		-41 51	108 11
32	15.10	05.10	Me	11.51	100.11
			39		
			o.=0		
⁰ [°] C _° C [°] C)				
× F Ma	-22.85	95.48		-10.03	76.91
33			C. 0		
55			40		
0)		Me C ⁼⁰		
C C	20.00	100 (0	Ť	17.04	04.55
Me Me	-30.98	102.69	^ℕ C _№	-17.06	86.75
34			1		
			41		
Me			Me C ⁼⁰		
$^{\circ}$ $^{\circ}$ $^{\circ}$ $^{\circ}$ $^{\circ}$ $^{\circ}$ $^{\circ}$ $^{\circ}$)	04.48		24.12	04.42
	-23.04	94.40	Me	-24.12	94.42
35			42		
			-0		
O _{≥C} Me)		HOVC		
	-31.73	102.13		-41.32	85.70
Me			`C _{₹0}		
36			43		

 TABLE 1.6
 CBS-Q
 Calculated
 $\Delta H_{\rm f}^{298}$ and
 ΔS^{298} for Ketenes and Bisketenes

Ketene	$\Delta H_{\rm f}^{298}$ (kcal/mol)	ΔS^{298} (cal/mol K)	Ketene	$\Delta H_{\rm f}^{298}$ (kcal/mol)	ΔS^{298} (cal/mol K)
$0 \xrightarrow{Me}_{Me} C^{-C}$ $Me Me$ 37	-39.88	110.27	HO C = O C	-48.89	94.76
^O ≈C 38	-32.93	97.07	HO C C C C C C C C C C C C C C C C C C C	-72.35	92.25

 TABLE 1.6
 (Continued)

 $\Delta H_{\rm f}^0$ of -92 kJ/mol^{4,6} for dimethylketene appears preferable to the experimental value of -137 kJ/mol,⁵ as the latter value depends on reference values for other cations and a recently revised experimental value of $\Delta H^{\rm o}_{\rm f}$ for the propionyl cation necessitated a revision of the proton affinity for methylketene.⁷ The trend in the values for the three ketenes also appears reasonable.

Based on an analysis of literature data, a revised value for the enthalpy of formation of ketene $\Delta H_{\rm f}^0 = -100\pm10$ kJ mol⁻¹ was proposed.¹⁸ This differs significantly from the previously accepted value of $\Delta H_{\rm f}^0 = -48$ kJ mol⁻¹, and further confirmation of this value appears warranted.

Heats of formation (kcal/mol values in parentheses) have been reported for the following species: NCCH=C=O (24 \pm 5), NCCH₂C(+)=O (198 \pm 3), NCCH₂C(\bullet)=O (32 \pm 1), and NCCH=C=O^{+ \bullet} (253 \pm 3).^{7,10}

Sumathi and Green¹¹ have used MP2/6–31G*//MP2/6–31G* calculations to determine the heats of formation of 45 fairly simple monoketenes and bisketenes, as shown in Table 1.6. The calculated values were considered to be as accurate as the available experimental values and were used to derive group equivalents for the estimation of $\Delta H_{\rm f}$ for other species.

Proton affinities (kJ/mol) of ketenes RCH=C=O for the groups R have been found by CBS-QB3 calculations as follows: H (820), CH₃ (844), NH₂ (917), OH (823), and F (793).²² These results reflect the combined effect of the substituent on the stability of the ketene and on the cation RCH₂C(+)=O such that the least favorable protonation for R = F was attributed to the great instability of the cation, FCHC(+)=O, while the most favorable process was for R = NH₂, where the initial product was the dissociating complex H₂NCH⁺ ••• C=O.²²

REFERENCES FOR SECTION 1.3

1. Bouchoux, G.; Salpin, J.-Y. J. Phys. Chem. 1996, 100, 16555-16560.

2. Bouchoux, G.; Salpin, J.-Y. Rapid Commun. Mass Spectrom. 1999, 13, 932-936.
26 STRUCTURE, BONDING, AND THERMOCHEMISTRY OF KETENES

- 3. McKee, M. L.; Radom, L. Org. Mass Spectrom. 1993, 28, 1238-1244.
- 4. Scott, A. P.; Radom, L. Int. J. Mass. Spectrom. Ion Proc. 1997, 160, 73-81.
- 5. Aubry, C.; Holmes, J. L.; Terlouw, J. K. J. Phys. Chem. A 1997, 101, 5958-5961.
- 6. Nguyen, M. T.; Nguyen, H. M. T. Chem. Phys. Lett. 1999, 300, 346-350.
- 7. Harvey, Z. A.; Traeger, J. C. J. Mass Spectrom. 2004, 39, 802-807.
- 8. Golovin, A. V.; Ponomarev, D. A.; Takhistov, V. V. J. Mol. Struct. 2000, 524, 259-281.
- 9. Ruscic, B.; Litorja, M.; Asher, R. L. J. Phys. Chem. A 1999, 103, 8625-8633.
- Holmes, J. L.; Mayer, P. M.; Vasseur, M.; Burgers, P. C. J. Phys. Chem. 1993, 97, 4865–4870.
- 11. Sumathi, R.; Green, W. H., Jr. J. Phys. Chem. A 2002, 106, 7939-7949.
- 12. Traeger, J. C. Org. Mass. Spectrom. 1985, 20, 223-227.
- Turacek, F.; Drinkwater, D. E.; McLafferty, F. W. J. Am. Chem. Soc. 1990, 112, 5892–5893.
- 14. Terlouw, J. K.; Burgers, P. C.; Holmes, J. L. J. Am. Chem. Soc. 1979, 101, 225-226.
- 15. Nuttall, R. L.; Laufer, A. H.; Kilday, M. V. J. Chem. Thermodynam. 1971, 3, 167-174.
- 16. Rice, F. O.; Greenberg, J. J. Am. Chem. Soc. 1934, 56, 2268-2270.
- 17. Lias, S. G.; Bartmess, J. E.; Liebman, J. F.; Holmes, J. L.; Levin, R. D.; Mallard, W. G. J. *Phys. Chem. Ref. Data* **1988**, *17*, Suppl. 1.
- Takhistov, V. V.; Ponomarev, D. A.; Golovin, A. V. Russ, J. Gen. Chem., 2003, 73, 1774–1776.
- Deming, R. L.; Wulff, C. A. In *The Chemistry of Ketenes, Allenes, and Related Com*pounds; Patai, S., Ed.; Wiley, New York, 1980, pp. 155–164.
- 20. Hunter, E. P. L.; Lias, S. G. J. Phys. Chem. Ref. Data 1998, 27, 413-656.
- 21. Traeger, J. C. Int. J. Mass Spectrom. 2000, 194, 261-267.
- 22. Ervasti, H. K.; Burgers, P. C.; Ruttink, P. J. A. Eur. J. Mass. Spectrom. 2004, 10, 791–799.

2 Spectroscopy and Physical Properties of Ketenes

The spectroscopy of ketenes has been discussed in previous reviews,^{1,2} but as study of these species has gained momentum and experimental techniques have improved, there has been a great increase in the available data. Ketenes possess very distinctive IR absorptions near 2100 cm^{-1} and equally characteristic ¹³C NMR chemical shifts for C₂ at remarkably high fields; these are very useful for providing definitive identification of these species.

2.1 NUCLEAR MAGNETIC RESONANCE SPECTROSCOPY

The ¹H chemical shift of ketene (1, CH₂=C=O) at δ 2.43–2.46 is at a remarkably high field for olefinic protons, and this may be attributed to the high electron density at C₂.^{3–5} The large value for the geminal coupling constant $J_{\text{H-H}'}$ compared to that for CH₂=CH₂ (2.2 Hz) was attributed to favorable hyperconjugative interactions of the C–H bonds in ketene with the carbonyl π bond.³ A compilation of ¹H chemical shifts and coupling constants for ketenes is given in Table 2.1.

Many diarylketenes have been isolated and characterized, and an extensive set of ¹³C and ¹⁷O NMR data is available for these compounds, as reported in Table 2.2.¹¹ The measurement of ¹⁷O shifts at natural abundance is more convenient for ketenes than for many other carbonyl compounds. The ¹³C chemical shifts are in the range of δ 178 to 203 ppm for the carbonyl carbon C₁ and 33–48 ppm for C₂. The very high field absorption for C₂ is characteristic of all ketenes and is attributed to the high negative charge density at this carbon, as confirmed by computations and illustrated by the characteristic ketenyl resonance structures **1a–c**. The C₂ shifts of these diaryl derivatives are at somewhat lower fields than for aliphatic and silyl substituted ketenes, which is the usual effect of α -aryl groups.

Ketenes II, by Thomas T. Tidwell

Copyright © 2006 John Wiley & Sons, Inc.

R	\mathbb{R}^1	δ (¹ H)	J (Hz)	Other δ , J (Hz)	Ref
Н	Н	2.43 ^b	15.8 (H, H')		3
Н	Η	2.46			4
Н	D	2.40	2.42 (H, D)		3
Me	Η	2.67	7.4	1.56 (CH ₃ , $J = 7.4$)	4, 5
Et	Η	2.80	6.9	2.04 (CH ₂), 1.06 (CH ₃ , $J = 7.4$)	4, 5
<i>i</i> -Pr	Н	2.82	5.7	2.49 (CH), 1.07 (CH ₃ , <i>J</i> = 6.7)	4, 5
<i>t</i> -Bu	Н	2.81		1.19 (t-Bu)	4, 5
Me ₃ Si	Н	1.65			10
EtHSi	Н	1.65		3.0 (HSi-CH)	10
(CD ₃) _e Si	Η	1.70			10
t-BuMe ₂ Si	Η	1.60			6
t-BuPh ₂ Si	Н	2.07			6
Cl ₃ Si	Η	2.84			10
CH ₃ CH=CH	Η	3.95, 4.10 (E, Z)		9	
CH ₂ =CH	Η	4.09	10	4.66 (<i>Z</i>), 4.95 (<i>E</i>), 6.06 (vinyl,	4, 5, 8
CH ₂ =CH	Me	1.76 (Me)		J = 11.0, 18) 4.76, 4.77, 6.30 (vinyl	4, 5
Me	Me	1.58, 1.65 ^c (Me)		J = 11, 16.6	4, 5, 10a

TABLE 2.1 ¹H NMR Chemical Shifts and Coupling Constants for Ketenes RR¹C=C=O

^{*a*}CDCl₃ unless otherwise noted. ^{*b*}85% in TMS. ^{*c*}CCl₄, ref. 10a.

TABLE 2.2¹³C and ¹⁷O NMR Chemical Shifts and Ketenyl IR StretchingFrequencies for Ar¹ArC=C=O (data from ref. 11 unless otherwise noted)

		δ ¹³	С		
Ar ¹	Ar	C ₁	C ₂	$\delta^{17}O$	$IR (cm^{-1})$
4-Me ₂ NC ₆ H ₄	Ph	202.6	45.8	340	2088
4-MeOC ₆ H ₄	Ph	202.0	45.7	339	$2093 (2096)^d$
4-MeC ₆ H ₄	Ph	201.7	46.4	340	2092
Ph	Ph	201.2	47.0	340	$2094 (2100)^d$
4-Br C ₆ H ₄	Ph	200.0	46.4	340	2095
4-Cl C ₆ H ₄	Ph	200.1	46.2	341	2096
$4-O_2C_6H_4$	Ph	195.7	48.0	340	$2100 (2109)^d$
MesBr ₂ ^{<i>a</i>}	MesBr ₂ ^a	178.7	55.8	310	2104
Mes ^b	Mes ^b	191.4	38.9	310	2096
C ₆ Me ₅	C_6Me_5	191.3	41.9	301	2096
Tip ^c	Tip ^c	191.5	33.4	292	2095

^a2,4,6-Me₃-3,5-Br₂C₆.

^b2,4,6-Me₃C₆H₂.

^c2,4,6-(*i*-Pr)₃C₆H₂.

^{*d*}Data for $Ar_{2}^{1}C=C=O$ (ref. 17).

The data in Table 2.2 reveal that for ketenes 4-XC₆H₄CPh=C=O, substituents X have rather small effects on the shifts of C₁ and C₂, although there is a correlation of the shift of C₁ with σ_p for the groups X, r = 0.995.¹¹ The ¹⁷O shifts are unaffected by the substituents X.

For the ketenes with highly substituted aryl groups there are significant changes in both the ¹³C and ¹⁷O shifts, and these are proposed to result from steric interactions.¹¹ The mechanism of this interaction was proposed to be a repulsive interaction between the bulky ortho substituents and the lone pairs on oxygen, which favored resonance structure **1c**, with increased electron density on C₂, giving rise to the higher field absorption at C₂ and the lower field for ¹⁷O in these cases.

The three diarylketenes in Table 2.1 with bulky hydrocarbon substituents have ${}^{13}C$ shifts for both C_1 and C_2 that are at a higher field than for the less crowded compounds. This may be due to a twisted aryl ring in these compounds, as shown in the calculated structure for meistylphenyketene **2**.¹² This effect could arise from a lack of conjugation or the effect of the nearby methyl groups. For the compound with two 2,4,6-Me₃Br₂Ph substituents C_1 is at an even higher field, while C_2 is at the lowest field of the group.



Results for other ketenes are collected in Table 2.3, and include a number of silyl substituted ketenes. Because of their high stability these compounds can usually be isolated, which permits convenient measurement of their spectra. Results for the ¹³C, ¹⁷O, and ²⁹Si NMR of silyketenes show a trend toward higher field shifts for C₁ and ¹⁷O, and lower field shifts for ²⁹Si, compared to other silyl compounds bonded to sp² carbons (Table 2.3).^{13,14} These results are consistent with "neutral hyperconjugation," as shown for resonance structures **A** and **B**, with a shift in electron density from the Si–C₂ bond to C₁ and O. Similar effects are seen in the ¹⁷O and ²⁹Si spectra of ketones. Substituents RMe₂Si have similar effects on the C₁ and ¹⁷O shifts for R = Me, *t*-Bu, Ph, but the group R does affect the ²⁹Si shifts.



The calculated chemical shifts of Me₃SiCH=C=O (**3**) and (CH₃)₂C=C=O (**4**) obtained using the (IGLO) method, basis set II, with MP2/6–31G* optimized geometries, are shown below and gave reasonable agreement with the experimental values, shown in parentheses.¹³ In particular, the calculations reproduce the large

TABLE 2.3 13 C, 17 O, and	²⁹ Si NMR Chemi	ical Shifts in CD	Cl ₃ of Ketenes and IR	Stretching Frequ	uencies of Reference Con	spunodu
Ketene	δ^{13} C C ₁	\mathbf{C}_2	$\delta^{17}O$	δ^{29} Si	v (cm ^{-1})	Ref
$CH_2 = C = 0$	194.0	2.5			2133, 2142, 2151 ^a	44
MeCH=C=0	200.0	10.9				44
EtCH=C=0	200.0	18.6				44
(2-Bu)CH=C=O	200.2	31.3				25
$Me_2C=C=0$	204.9^{a}	24.2	329^{b}			5, 44
MeEtC=C=0	206.1	28.2				44
<i>t</i> -Bu ₂ C=C=O	203.1	51.9	330.6 (331.5)		2083	45, 46
t-BuC(CO ₂ Me)=C=O	191.8	50.0				42
t-BuC(COBu-t)=C=0	196.5	53.4				42
$(1-Ad)_2C=C=O$	203.4	53.0			2035	18
$(CH_2)_2 C = C = 0$	206.1					
PhCMe=C=0	205.6	33.8				44
PhCEt=C=0	205.6	42.1				44
$Ph_2C=C=O$	201.3	47.0	340		2100^{17}	7, 11
CHCI=C=0	166.2	70.1				43, 43a
CMeCl=C=0	168.9	85.0				43, 43a
$CBr_2 = C = 0$	178.6	98.5				7
CH ₂ =CHCH=C=0	200.2	28.6				8
$t-Bu_2P(S)C(Ac)=C=O$	170.2	9.5			2100	40
$(CF_3S)_2C=C=0$	171.8	18.8			2139	39
$(C_6F_5S)_2C=C=0$	172.6	31.4			2135	47
$(CF_3Se)_2C=C=0$	169.3	124.9			2127	37
F ₅ SCMe=C=O	196.8	72.4			2149	38
$(CF_3SO_2)_2C=C=0$	181.0	13.5			2139	37
Me ₃ SiCH=C=0	179.2	-0.1	255.0	-0.2	2112	10
PhMe ₂ SiCH=C=0	178.9	-0.7	256.1	-5.4	2114	20
Me ₃ SiCPh=C=0	182.5			-0.6		13
<i>t</i> -BuMe ₂ SiCH=C=O	180.0	-4.6				9
t-BuPh ₂ SiCH=C=O	178.5	-3.6				9

$(Me_3SiC=C=0)_2$	181.8	5.6	$269.2 (269.0)^a$	3.2		13
$(t-BuMe_2SiC=C=O)_2$	182.2	4.3	270.5	10.6		13
(PhMe ₂ SiC=C=O) ₂	180.9	5.9	270	-2.4	2080	13
Me ₂ SiCH ₂ CO ₂ H				3.8		13
Me ₃ SiCH ₂ CO ₂ Et				3.2^c		13
Me ₃ SiCH ₂ COCH ₃	206.8		536.7	1.6		13
<i>t</i> -BuMe ₂ Si						
β	202.0		499 1	-8 4		13
<i>t</i> -BuMe ₂ Si			1	-		1
t-BuMe ₂ Si0						
3 	202.1			0.4		13
<i>t</i> -BuMe ₂ Si						
Me ₂ C=O (acetone)	206.5		$565.3 (571)^a$			13
MeaSiCH=CH,				-6.6		13
Me ₃ SiC ₆ H ₅				-4.5		13
t-BuMe ₂ SiCH=CH ₂				0.9		13
$CH_2 = CHC(SiMe_3) = C = O$	183.7	22.3			2085	48
Me ₂ Si(CH=C=O) ₂	178.8	0.9	260.1	-3.3	2108, 2129	14
MeSi(CH=C=0) ₃	178.0	1.8	263.9	-7.9	2108, 2136	14
Me ₅ Si ₂ CH=C=O	178.7	-2.4	252.2	-18.3, -18.8	2108	19
(SiMe ₂ CH=C=0) ₂	178.2	-2.2	253.6	-18.6	2108, 2052	19
(CH ₂ SiMe ₂ CH=C=O) ₂	179.0	-1.6	253.8	3.1	2115, 2052	19
$(Me_3Si)_2C=C=O$	1.668	1.7	211.2	-1.5	2085	19, 23
$1,4-C_6H_4(SiMe_2CH=C=O)_2$	178.9	-0.7	256.6	-5.4	2115	21



TABLE 2.3 (Continue)	(pən						
R ¹	\mathbb{R}^{1}	C1	C_2	C3	C_4		
Me ₃ Si	hh	178.8	7.9	33.5	202.2	2093	25, 26
Me ₃ Si	4-MeOC ₆ H ₄	178.4	7.4	31.9	204.2	2094	21, 36
Me ₃ Si	$4-MeC_6H_4$	178.3	7.1	32.6	203.1	2094	21, 36
Me ₃ Si	$4-AcC_6H_4$	177.5	7.0	34.2	198.1	2100	21, 36
<i>i</i> -Pr ₃ Si	Ph	178	11.9	33.5	201.6	2092	26, 36
Me ₃ Si	Me	180.6	10.7	18.9	204.2	2101	25, 26, 36
Me ₃ Si、O							
R∕∕CO ₂ R ³							
	${ m R}_3$	c ₁	C_2	C3	C_4		
	4-MeOC ₆ H ₄	179.8	46.4			2089	21
	Me	181.3	9.8			2085	27
	Et	180.9	10.8			2091	27
	<i>i</i> -Pr	181.0	11.0			2084	27
0		181.4	4.4		3.4, 6.8	2089	22
∕ √	iMe ₃	181.7	5.6				
-CH2Me2Si							
	/						
D							

 a CH₃CN solvent. ^bSolid, argon matrix, and vapor, respectively. effects of Si substitution on the chemical shifts for ${}^{13}C_1$ ($\Delta\delta$ calc 39.6, obs 25.7 ppm) and ${}^{17}O$ ($\Delta\delta$ calc 78.7, obs 74 ppm).

$$\begin{array}{c} (-0.1) - 16.9 \\ (CH_{3})_3 Si & -6.0 & (-0.2) \\ (1.65) & 1.45 \\ H & \uparrow \\ 175.4 & (179.2) \end{array} \begin{array}{c} (10.1) & 9.6 \\ (24.2) & 26.9 \\ CH_3 & \uparrow \\ 215.0 & (204.9) \end{array}$$

A continuing challenge is the measurement of NMR spectra of highly reactive, short-lived ketenes. There has been considerable progress in the measurement of IR spectra of such species, using either time-resolved IR (TRIR), matrix isolation, or measurements in dilute solutions with dry solvents, but NMR studies have not been as successful. There is a report of the generation of PhCH=C=O and measurement of the ¹³C shift for C₂, but this requires confirmation.¹⁵

The ¹H and ¹³C NMR spectra of the cationic diacylketene **5** show C₁ and C₂ at 169.7 and 78.0 Hz, respectively; $J_{CC} = 98.0$ Hz, with the IR absorption at 2200 cm⁻¹.¹⁶



Solid-state ${}^{13}C$ NMR spectra were measured for $CH_2=C=O$ and $O=C=C=C=O.{}^{16a}$

REFERENCES FOR SECTION 2.1

- 1. Munson, J. W. In *The Chemistry of Ketenes, Allenes, and related Compounds;* Patai, S., Ed.; Wiley, New York, 1980, pp. 165–188.
- 2. Runge, W. Prog. Phys. Org. Chem. 1981, 13, 315-484.
- 3. Allred, E. L.; Grant, D. M.; Goodlett, W. J. Am. Chem. Soc. 1965, 87, 673-674.
- 4. Masters, A. P.; Sorensen, T. S.; Ziegler, T. J. Org. Chem. 1986, 51, 3558-3559.
- 5. Sorensen, T. S; unpublished results privately communicated.
- 6. Valentí, Pericas, M. A.; Serratosa, F. J. Org. Chem. 1990, 55, 395-397.
- 7. Duddeck, H.; Praas, H-W. Magn. Reson. Chem. 1993, 31, 182-184.
- Trahanovsky, W. S.; Surber, B. W.; Wilkes, M. C.; Preckel, M. M. J. Am. Chem. Soc. 1982, 104, 6779–6780.
- 9. Schiess, P.; Radimerski, P. Helv. Chim. Acta 1974, 57, 2583-2597.
- Shchukovskaya, L. L.; Kol'tsov, A. I.; Lazarev, A. N.; Pal'chik, R. I. Dokl. Akad. Nauk 1968, 892–895. Engl. Transl. 1968, 318–320.
- 10a. Ogata, Y.; Adachi, K. J. Org. Chem. 1982, 47, 1182-1184.
- 11. Cerioni, G.; Plumitallo, A.; Frey, J.; Rappoport, Z. Magn. Res. Chem. 1995, 33, 669-673.

- Yamataka, H.; Aleksiuk, O.; Biali, S. E.; Rappoport, Z. J. Am. Chem. Soc. 1996, 118, 12580–12587.
- 13. Allen, A. D.; Egle, I.; Janoschek, R.; Liu, H. W.; Ma, J.; Marra, R. M.; Tidwell, T. T. *Chem. Lett.* **1996**, 45–46.
- 14. Sung, K.; Tidwell, T. T. J. Am. Chem. Soc. 1996, 118, 2768-2769.
- Solladié-Cavallo, A.; Liptaj, T.; Schmitt, M.; Solgadi, A. *Tetrahedron Lett.* 2002, 43, 415–408.
- Csihony, S.; Bodor, A.; Rohonczy, J.; Horvath, I. T. J. Chem. Soc. Perkin Trans. 1 2002, 2861–2865.
- 16a. Beeler, A. J.; Orendt, A. M.; Grant, D. M.; Cutts, P. W.; Michl, J.; Zilm, K. W.; Downing, J. W.; Facelli, J. C.; Schindler, M. S.; Kutzelnigg, W. J. Am. Chem. Soc. 1984, 106, 7672–7676.
- 17. Melzer, A.; Jenny, E. F. Tetrahedron Lett. 1968, 4503-4506.
- Reetz, M. T.; Schwellnus, K.; Hübner, F.; Massa, W.; Schmidt, R. E. Chem. Ber. 1983, 116, 3708–3724.
- 19. Sung, K.; Tidwell, T. T. Organometallics 1997, 16, 78-85.
- 20. Liu, R.; Marra, R. M.; Tidwell, T. T. J. Org. Chem. 1996, 61, 6227-6232.
- 21. Liu, R.; Tidwell, T. T. J. Chem. Soc. Perkin Trans. 2, 1996, 2757-2762.
- 22. Allen, A. D.; Lough, A. J.; Tidwell, T. T. Chem. Commun. 1996, 2171-2172.
- 23. Uhlig, W.; Tzschach, A. Z. Chem. 1988, 28, 409-410.
- 24. Liu, R.; Tidwell, T. T. Can. J. Chem. 1995, 73, 1818-1822.
- 25. Allen, A. D.; Lai, W.-Y.; Ma, J.; Tidwell, T. T. J. Am. Chem. Soc. 1994, 116, 2625-2626.
- 26. Allen, A. D.; Ji, R.; Lai, W.-Y.; Ma, J.; Tidwell, T. T. Heteroatom. Chem. 1994, 5, 235-244.
- 27. Zhao, D.-C.; Allen, A. D.; Tidwell, T. T. J. Am. Chem. Soc. 1993, 115, 10097-10103.
- 28. Allen, A. D.; Baigrie, L. M.; Gong, L.; Tidwell, T. T. Can. J. Chem. 1991, 69, 138-145.
- Muchall, H. M.; Werstiuk, N. H.; Ma, J.; Tidwell, T. T.; Sung, K. Can. J. Chem. 1997, 75, 1851–1861.
- 30. Werstiuk, N. H.; Muchall, H. M.; Ma, J.; Liu, M. T. H. Can. J. Chem. **1998**, 76, 1162–1173.
- 31. Shustov, G. V.; Sun, F.; Sorensen, T. S.; Rauk, A. J. Org. Chem. 1998, 63, 661-669.
- Shustov, G. V.; Sun, F.; Sorensen, T. S.; Rauk, A. J. Chem. Soc. Perkin Trans. 2 1999, 1035–1041.
- Grishin, Y. K.; Ponomarev, S. V.; Lebedev, S. A. Zh. Org. Khim. 1974, 10, 404–405. Engl. Transl. 1973, 10, 402.
- 34. Egle, I.; Lai, W.-Y.; Moore, P.; Renton, P.; Tidwell, T. T.; Zhao, D.-C. J. Org. Chem. 1997, 61, 18–25.
- Wagner, B. D.; Arnold, B. R.; Brown, G. W.; Lusztyk, J. J. Am. Chem. Soc. 1998, 120, 1827–1834.
- Allen, A. D.; Colomvakos, J. D.; Diederich, F.; Egle I.; Hao, X.; Liu, R.; Lusztyk, J.; Ma, J.; McAllister, M. A.; Rubin, Y.; Sung, K.; Tidwell, T. T.; Wagner, B. D. *J. Am. Chem. Soc.* **1997**, *119*, 12125–12130.
- 37. Haas, A.; Praas, H.-W. Chem. Ber. 1992, 125, 571-579.
- 38. Bittner, J.; Seppelt, K. Chem. Ber. 1990, 123, 2187-2190.
- 39. Haas, A.; Lieb, M.; Praas, H.-W. J. Fluorine Chem. 1989, 44, 329-337.

- Lukashev, N. V.; Fil'chikov, A. A.; Zhichkin, P. E.; Kazankova, M. A.; Beletskaya, I. P. Zh. Obshch. Khim. 1991, 61, 1014–1016. J. Gen. Chem. (Engl. Transl.) 1991, 61, 920–921.
- 41. Allen, A. D.; Cheng, B.; Fenwick, M. H.; Huang, W.; Missiha, S. Tahmassebi, D.; Tidwell, T. T. Org. Lett., **1999**, *1*, 693–696.
- 42. Leung-Toung, R.; Wentrup, C. Tetrahedron 1992, 48, 7641-7654.
- 43. Lindner, E.; Steinwand, M.; Hoehne, S. Angew. Chem. Int. Ed. Engl. 1982, 21, 355-356.
- 43a. Lindner, E.; Merkle, R. D.; Mayer, H. A. Chem. Ber. 1986, 119, 645-658.
 - 44. Firl, J.; Runge, W. Z. Naturforsch. 1974, 29B, 393–398; Angew. Chem. Int. Ed. 1973, 12, 668–669.
 - 45. Hofman, P.; Perez, L. A.; Kain, I. Synthesis 1986, 43-44.
 - 46. Olah, G.; Wu, A.; Farooq, O. Synthesis 1989, 568.
 - 47. Haas, A.; Praas, H.-W.; Radau, G. J. Fluorine Chem. 1994, 67, 257-260.
 - 48. Danheiser, R. L.; Sard, H. J. Org. Chem. 1980, 4810-4812.

2.2 ULTRAVIOLET SPECTRA, PHOTOELECTRON SPECTRA, AND CHIROPTICAL PROPERTIES

The ultraviolet (UV) spectrum of ketene was first measured in 1924¹ and has been reviewed.² This showed the C=C $\pi \rightarrow \pi^*$ transition at 183 nm,³ the C=O $\pi \rightarrow \pi^*$ transition at 215 nm ($\epsilon = 80$),⁴ and the n $\rightarrow \pi^*$ transition at 325 nm ($\epsilon = 10$).^{5,6} The UV spectrum of ketene has been recorded using a continuously tunable light source of synchrotron radiation.⁷ The electronic states in the 140 to 220 nm cluster region have a predominantly Rydberg character and were described by coupled methods for excitation energies.⁷ The absorption spectrum in this region was calculated from first principles based on a vibronic model and compared to the experimental spectrum.⁸

The UV spectra of a variety of ketenes and polyketenes are given in Table 2.4. The series of ketenes 1-3 with one, two, and three ketenyl groups, respectively, show UV absorption with maxima in the range 300 to 308 nm, with no significant effect of the position on the number of ketenyl groups attached to silicon but an increase in the extinction coefficient as the number of ketenyl groups increases.

Gas phase UV spectra showing the maxima have been reported for ketene (330 nm), methylketene (365 nm), ethylketene (355 nm), and dimethylketene (370 nm).²⁵ The vibrational progression of $CH_2=C=O$ suggests that the lowest singlet state is less than 61 kcal/mol above the ground state, while the energy of the lowest triplet in less than 55 kcal/mol.²⁶

The photoelectron spectrum of bisketene **4** was found to be in good agreement with calculations for the twisted conformation shown.²⁷ Photoelectron spectra of **1-3** and other silylated mono- and bisketenes were also determined.^{27,28}



	λ_{max} (nm)	$\epsilon (cm^2 mol^{-1})$	Ref
CH2=C=0	183, 215, 325	(-), 80, (10)	1–6
$Me_2C=C=O$	370		9
$Et_2C=C=O$	378	13.3	9
$t-Bu_2C=C=O$	225, 360	1,200, 11.6	10
t-BuCH=C=O	360	13.2	10
t-BuC(CO ₂ Et)=C=O	228		10
PhCH=C=O	360	13.2	10
$4-O_2NC_6H_4CH=C=O$	495		35
$Ph_2C=C=O$	350, 405, 399	(-), 310	11, 12
$(EtS)_2CH=C=O$	288	6,000	13
c-PrCPh=C=O	248, 395 (CH ₂ Cl ₂)		19
PnCPh=C=O	260, 385 (CH ₂ Cl ₂)		19
c-HxCPh=C=O	264, 388 (CH ₂ Cl ₂)		19
$c-\Pr(Bu-t)=C=O$	240, 373 (CH ₂ Cl ₂)		19
$c - \Pr_2 C = C = O$	280, 376 (CH ₂ Cl ₂)		19
	~280		14
CN CN	13, 323 sh	15, 500, 12, 500	15
Ph C ⁼⁰	221 324	14 000 11 500	15
Ph	221, 324	14,000, 11,000	15
Ph CN	212 207	1 200 20	10
$Me_3SICH=C=O(1)$	213, 307	1,300, 30	10
t-Bulle ₂ SiCH=C=O	292	33 200_110	10, 20
$PhMe_2SiCH=C=0$	254, 509	200, 110	24
$Me_2SI(CH=C=O)_2(2)$	300 207	88 170	20
$M_{2}S(CH=C=0)_{3}(5)$	207	170	20
$MeSiC(CH=CH_2)=C=O$	255	9,000	34 21
$(\text{We}_2\text{SI})_2\text{C}=\text{C}=\text{O}$	213, 278 217	1,000, 27	21
$\frac{1}{1} \frac{1}{1} \frac{1}$	317	210 610	21 21
$(CH SiMe CH = C = O)_2$	300	220	21 21
(C_{12}) C_{2} C_{12}	307	220	21
0			

TABLE 2.4UV Spectra of Ketenes



	λ_{max} (nm)	$\epsilon (cm^2 mol^{-1})$	Ref
Me ₃ Si, Me ₃ Si (4)	325, 376	250, 107	16
t-BuMe ₂ Si, t-BuMe ₂ Si	326, 400	260, 80	16
Me ₃ Si, Ph	257		17, 18
<i>i</i> –Pr ₃ Si, Ph	260		17, 18
Me_3Si , 4- $MeOC_6H_4$	257		22
Me_3Si , 4- MeC_6H_4	257		22
Me ₃ Si, 4-MeCOC ₆ H ₄	289		22
PhMe ₂ Si, PhMe ₂ Si	210 (sh)	16,000	23
$O = C \xrightarrow{C} O \xrightarrow{Me_3Si} C = O$	280		23

TABLE 2.4 (Cont

Thermolysis of the diazoacetophenones **5** at 500 °C gave the arylketenes **6** (equation 1), whose vertical ionization potentials (IP) were measured by photoelectron spectroscopy in the gas phase, and gave good agreement with values calculated at the B3LYP/6-31+G(d)//HF/6-31G(d) level.²⁹ It was proposed that the HOMO of these ketenes was destabilized by the presence of π -donor substituents at C–4 of the phenyl ring, thus leading to a lowered IP.²⁹

 $4-RC_6H_4 \xrightarrow{O} N_2 \xrightarrow{\Delta} 4-RC_6H_4 \xrightarrow{4-RC_6H_4} C=0$ (1)

	IP (eV)	
R	PE	Calc
Н	8.23	7.93
CH ₃	8.22	7.68
CH ₃ O	7.91	7.36
Cl	8.29	7.94

Other measurements of photoelectron spectra of ketenes include CH_2 =CRCH =C=O (R = H, CO₂H),³⁰ haloketenes, and others.³¹

The conformational behavior of ketenes CH₃CHRCH=C=O (7, R = H, Me, and EtO) was calculated at the B3LYP/6-31/+G**//B3LYP/6-31G* level, and chiroptical properties were calculated at the CIS/6-31+G* level.³² The calculations indicated that the conformation with the C=O bond eclipsed by an α -hydrogen was of lowest energy, and the barriers for rotation around the C–C bond were 1.3 to 1.6 kcal/mol. The CD spectrum of (*S*)–**8** was measured at –13°C in CH₂Cl₂/pentane ($\lambda_{max} = 363 \text{ nm}$) ¹H NMR δ 0.81 (t, 3, *J* = 7.3), 0.98 (d, 3, *J* = 6.7), 1.05–1.20 and 1.30–1.40 (ea 1, m), 2.13 (m, 1), and 2.75 (1, d, *J* = 6.9); ¹³C NMR: δ 11.4 (CH₃), 21.4 (CH₃), 22.9 (CH), 28.2 (CH), 31.3 (CH₂), 200.2 (C=O). The effect of the alkyl substituent on the n- π * Cotton effect sign obeyed a reverse octant rule, and there was agreement between the calculated and experimental chiroptical properties.³² The effects of alkyl substituents on the chiroptical properties of bicyclo[2.2.1]heptylideneketenes **9** have also been examined.³³



REFERENCES FOR SECTION 2.2

- 1. Lardy, G. C. J. Chim. Phys. 1924, 21, 353-385.
- 2. Rabalais, J.; McDonald, J. M.; Scherr, V.; McGlynn, S. P. Chem. Revs. 1971, 71, 73–108.
- 3. Price, W. C.; Teegan, J. P.; Walsh, A. D. J. Chem. Soc. 1951, 920-926.
- 4. Knox, K.; Norrish, R. G. W.; Porter, G. J. Chem. Soc. 1952, 1477-1486.
- 5. Runge, W. Prog. Phys. Org. Chem. 1981, 13, 315-484.
- 6. Kirmse, W. Carbene Chemistry, 2nd ed.; Academic Press, New York, 1971, p. 9.
- 7. Chiang, S.-Y.; Bahou, M.; Wu, Y.-J.; Lee, Y. -P. J. Chem. Phys. 2002, 117, 4306-4316.
- 8. Nooijen, M. Int. J. Quant. Chem. 2003, 95, 768-783.
- 9. Holyroyd, R. A; Blacet, F. E. J. Am. Chem. Soc. 1957, 79, 4830-4834.
- 10. Allen, A. D., University of Toronto, unpublished results.
- Nadzhimutdinov, S. H.; Slovokhotova, N. A.; Kargin, V. A. Zh. Fiz. Khim. 1966, 40, 893–897. Chem. Abstr. 1966, 65, 4851h.
- 12. Huisgen, R.; Feiler, L. A.; Otto, P. Chem. Ber. 1969, 102, 3444-3459.

- 13. Inoue, S.; Hori, T. Bull. Chem. Soc. Jpn. 1983, 56, 171-174.
- 14. Winnik, M. A.; Wang, F.; Nivaggioli, T.; Hruska, Z.; Fukumura, H.; Masuhara, H. J. Am. Chem. Soc. **1991**, *113*, 9702–9704.
- 15. Hobson, J. D.; Al Holly, M. M.; Malpass, J. R. Chem. Commun. 1968, 764-766.
- 16. Zhao, D.-C.; Allen, A. D.; Tidwell, T. T. J. Am. Chem. Soc. 1993, 115, 10097-10103.
- Allen, A. D.; Lai, W.-Y.; Ma, J.; Tidwell, T. T. J. Am. Chem. Soc. 1994, 116, 2625–2626.
- Allen, A. D.; Ji, R.; Lai, W.-Y.; Ma, J.; Tidwell, T. T. *Heteroatom. Chem.* 1994, 5, 235–244.
- 19. Allen, A. D.; Baigrie, L. M.; Gong, L.; Tidwell, T. T. Can. J. Chem. 1991, 69, 138-145.
- 20. Sung, K.; Tidwell, T. T. J. Am. Chem. Soc. 1996, 118, 2768-2769.
- 21. Sung, K.; Tidwell, T. T. Organometallics 1997, 16, 78-85.
- 22. Liu, R.; Tidwell, T. T. J. Chem. Soc. Perkin Trans. 2, 1996, 2757-2762.
- 23. Liu, R.; Tidwell, T. T. Can. J. Chem. 1995, 73, 1818-1822.
- 24. Liu, R.; Marra, R. M.; Tidwell, T. T. J. Org. Chem. 1996, 61, 6227-6232.
- 25. Hatakeyama, S.; Honda, S.; Washida, N.; Akimoto, H. Bull. Chem. Soc. Jpn. 1985, 58, 2157–2162.
- 26. Laufer, A. H.; Keller, R. A. J. Am. Chem. Soc. 1971, 93, 61-63.
- 27. Werstiuk, N. H.; Ma, J.; McAllister, M. A.; Tidwell, T. T.; Zhao, D. J. Chem. Soc. Faraday Trans. **1994**, *90*, 3383–3390.
- Muchall, H. M.; Werstiuk, N. H.; Ma, J.; Tidwell, T. T.; Sung, K. Can. J. Chem. 1997, 75, 1851–1861.
- Werstiuk, N. H.; Muchall, H. M.; Ma, J.; Liu, M. T. H. Can. J. Chem. 1998, 76, 1162– 1173.
- Chuburu, F.; Lacombe, S.; Pfister-Guillouzo, G.; Ben Cheik, A.; Chuche, J.; Pommelet, J. C. J. Am. Chem. Soc. 1991, 113, 1954–1960.
- 31. Bock, H.; Solouki, B. Angew. Chem. Int. Ed. Engl. 1981, 20, 427-444.
- 32. Shustov, G. V.; Sun, F.; Sorensen, T. S.; Rauk, A. J. Org. Chem. 1998, 63, 661-669.
- 33. Shustov, G. V.; Sun, F.; Sorensen, T. S.; Rauk, A. J. Chem. Soc., Perkin Trans. 2 1999, 1035–1041.
- 34. Danheiser, R. L.; Sard, H. J. Org. Chem. 1980, 45, 4810-4812.
- 35. Broxton, T. J.; Duddy, N. W. J. Org. Chem. 1981, 46, 1186-1191.

2.3 INFRARED SPECTRA AND RAMAN SPECTRA

The IR spectrum of ketene has been the subject of continuing interest for a long time,^{1–9} and the principal band assignments for $CH_2=C=O$, CHD=C=O, and $CD_2=C=O$ have been determined (Table 2.5).^{2,4} Further investigations have included the combination bands and the vibrational and rotational spectrum.

Measurements of the vibrational spectrum and dipole moment of ketene have appeared, including the $v_2 + v_3$ and $v_2 + v_4$ bands at 3526 to 3528 cm⁻¹ and at 3257 to 3277 cm⁻¹, respectively, from IR-UV double resonance photolysis.^{5a}

Mode	$CH_2=C=O^4$	$CH_2=C=O^{4a}$	$CD_2 = C = O^2$
C-H (D) stretch	3071	3070.4	3115
C=O stretch	2152	2152.6	2150
CH ₂ deformation	1388	1386.6	1293
C=C stretch	1118	1116.0	1022
C-H (D) stretch (antisymmetric)	3166	3165.4	2297
CH ₂ rocking	977	977.8	815
CH ₂ wagging	588	587.3	
C=C=O bending	438	439.0	
C=C=O bending (out of plane)	528	528.4	515

 TABLE 2.5
 IR Absorption Band Assignments for Ketenes (cm⁻¹)

The IR spectra of ketenes showed a dependence of the position of the ketenyl band upon the medium, and for $CH_2=C=O$ this appears at 2151, 2142, and 2133 cm⁻¹ in the vapor, the argon matrix, and the solid, respectively.⁶ The Raman absorption of ketene was observed at 2049 cm⁻¹.⁷

The four fundamental IR bands for ketene below 1000 cm⁻¹ have been obtained free from Coriolis contributions.^{8,9} A detailed investigation of the IR spectrum of gaseous methylketene has also been reported.¹⁰

The ketenyl IR bands for ketenes $4-XC_6H_4CPh=C=O$ (Table 2.2, Section 2.1) are correlated with the σ_p substituent parameters ($r^2 = 0.978$).¹¹ The increase in the frequency is 7 cm^{-1} for X changing from MeO to NO₂, and for the disubstituted series $(4-XC_6H_4)_2=C=O^{11}$ this change is 13 cm^{-1} , showing an additivity of the effect. These trends are consistent with the proposal that electron acceptor substituents stabilize the resonance structure **1a** and lead to an increase in the frequency.



A correlation derived from the IR frequencies of just five ketenes $R_2C=C=O$ (R = H, Cl, CH₃, CN, CH₃) was proposed between the frequencies and the *F* (field) and *R* (resonance) parameters of the substituents (equation 1).¹² This correlation indicated that the ketene frequency v was increased by both effects.

$$v(cm^{-1}) = 24F + 30R + 2141 \tag{1}$$

A later correlation of computed IR frequencies found a modest correlation with a coefficient of 0.87 of the observed frequencies with the *F* parameter but

no dependence on R (equation 2).¹³

$$v(cm^{-1}) = 88F + 2141 \tag{2}$$

IR spectra of a variety of ketenes¹⁴ and bisketenes¹⁵ formed by laser flash photolysis have been measured by TRIR spectroscopy, as well as many by conventional spectroscopy. Some of these are collected in Tables 2. Significant medium effects on these absorptions, including variations with solvent, have been observed.⁶

The ketenylidenecycloalkanes 2-4 all show doublet absorption of the ketenyl band in the IR, a result attributable to Fermi resonance.^{16,18,29}



Ketene **5** labeled with ¹³C in the carbonyl carbon displays four different IR bands at 2068, 2074, 2081, and 2089 cm⁻¹ in an argon matrix at 8 K, attributed to the conformations **5a–d**, respectively.¹⁷ As noted in Section 1.1.1 (ref. 26), it has been suggested that **5a** is not an energy minimum.



The IR spectra of $CH_2=C=O$, $CH_3CH=C=O$, and $C_2H_5CH=C=O$ complexes in argon matrices are reported in Table 2.6 and were assigned to the structure **6** involving coordination of C_2 .^{18a-c} In xenon for $CH_2=C=O$ an additional band at 2134 cm⁻¹ was assigned to structure **7**.^{18a-c}

$$\begin{array}{c} CI \\ H \\ H'' \\ C = C = O \\ H \\ 6 \\ 7 \end{array}$$

Ketene	ν	Medium	Ref
CH2=C=0	2151	gas phase	6
	2164	gas phase	5b
	2141	argon matrix	6
	2133	solid	6
CHF=C=O	2145	argon matrix	30
	2148	gas phase	30
	2148	argon matrix	30
CHCl=C=O	2151	argon matrix	30
	2150	gas phase	27
	2157	gas phase	30
CHBr=C=O	2148	argon matrix	30
CHI=C=O	2145	argon matrix	30
$CCl_2 = C = O$	2160	gas phase	33
2	2158	matrix	32
	2155	matrix	28
$CF_2 = C = O$	2162	argon matrix	28a.b
$CH_2 = C = O \bullet HCl$	2145	argon matrix	18a–c
2	2142, 2134	xenon matrix	18a-c
MeCH=C=O	2140	gas phase	5b
	2130	solution	25
$MeCH=C=O \bullet HCl$	2127	argon matrix	30
EtCH=C=O	2134	gas phase	5b
$Me_2C=C=0$	2134	gas phase	5b
t = BuCH = C = O	2108	THF	60
$t - Bu_0 C = C = 0$	2090	film	63
$t = \operatorname{Bu}_2 C(\operatorname{Pr}_i) = C = O$	2090	CCL	59
t - BuC(Pr-c) = C = O	2090		59
$c - Pr_0 C = C = 0$	2106		59
$M_{e}OCH = C = O$	2100	matrix	16a
FtOCH-C-O	2120	solution	70
	2120	isooctana	20
	2250	ras phase	65
$Ph_C = C = O$	2004	CHCl-	64
PhCMe=C=O	2094	THE	60
PhCEt = C = O	2135	CCI	60
PhC(Pr, i) = C = O	2130		50
PhC(Pr, c) = C = O	2090		59
PhC(Pu, t) = C = 0	2102		50
FIC(Bu-i) = C = 0	2097	iconstance	39
	2110	Isooctane	29
E,Z-MECH=CHCH=C=O	2118		07
$Cn_2 = CnC(SiMe_3) = C = O$	2085	solution	24
PIIC = C C C C	2131	isooctane	29
$(NC)_2C=C=O$	21/5	matrix	32
$(CF_3)_2C=C=O$	2196	matrix	34
$(EtS)_2C=C=O$	2160	solution	35
$(t-BuS)_2C=C=O$	2170	solution	35

 TABLE 2.6 IR Stretching Frequencies of Ketenes (cm⁻¹)

Ketene	ν	Medium	Ref
$(ArS)_2C=C=O$	2190	solution	35
F ₅ SCH=C=O	2177	film	31
$F_5SCMe=C=O$	2149	film	31
$F_5SCCl=C=O$	2175	vapor	31
$F_5SCBr=C=O$	2169	vapor	31
CH2=CHCH=C=O	2118	toluene	22
CH ₂ =CHCH=C=O	2130	gas phase	22a
Me ₃ SiCH=C=O	2112	solution	36
$(CD_3)_3SiCH=C=O$	2122	gas phase	36
$(Me_3Si)_2C=C=O$	2085	solution	36
Et ₃ SiC=C=O	2115	solution	36
Cl ₃ SiCH=C=O	2165	gas phase	36
t-BuMe ₂ SiCH=C=O	2110	CCl ₄	37
t-BuPh ₂ SiCH=C=O	2110	CCl ₄	37
$EtO_2CCMe=C=O$	2137	gas phase	38
MeCOCMe=C=O	2121	gas phase	38
MeCOCH=C=0	2137	gas phase	38
t-BuCOCH=C=O	2140	gas phase	38
t-BuCOCH=C=O	2140	gas phase	38
$(C_{c}C_{c})_{c}C=C=0$	2130	gus phuse	39
$(C_{0}E_{1})_{2}C=C=0$	2130	CCL	40
CH = C = 0	2023	matrix	41
$Ph_{2}P-C-C-O$	2025	KBr	42
~~~C ⁼⁰	2111	toluene	22
C ⁼⁰	2117	toluene	22
C=0	2101	toluene	21
C=0	2101, 2103	toluene, matrix	22, 52
	2120	isooctane	20
C=0	2110	matrix	4
Ph" C=0	2121	isooctane	20
X c=0	2119	isooctane	20, 44

 TABLE 2.6 (Continued)

Ketene	ν	Medium	Ref
	2126	isooctane	54
R = (H, Ph, c-Pr)	2119–2121	isooctane	20
0 ~ C ²⁰	2117, 2129	matrix, $c$ -C ₆ H ₁₂	17
C=0	2119, 2130	solution, matrix	18, 45
C=0	2125, 2145	matrix	46
C=0	2098, 2150	solution	16
C=0	2106, 2122	hexane	29
C=0	2100	CH ₂ Cl ₂	68
C=0	2106	CH ₂ Cl ₂	69
C=0	2100	THF	60
C=0	2080 ⁶¹ , 2115 ⁶²	film	61, 62
^O ≈C≫[] _n ⊂ ^{≠O}	2113–2114	hexane	55
C=0 C=0	2115	toluene	19

**TABLE 2.6** (Continued)

Ketene	ν	Medium	Ref
0 [±] C [*] C [*] 0	2116, 2122	isooctane	19
0 ⁵ C ²⁰ C ²⁰	2115	hexane	66
$Me_{3}Si + C^{zO}$ $Me_{3}Si + C_{0}$	2084	CDCl ₃	48
$Ph C^{20}$ $Ph C_{0}$	2100, 2112	matrix	49
$Me + C^{=0}$ $Me + C_{0}$	2096, 2138	matrix	49
	2077, 2138	CH ₃ CN	50
	2128 (4); 2123 (3) 2123 (anti-2); 2132 (syn-	2)	53, 53a
С=0	2133	matrix	56
0 0 →= c=0	2154, 2176	matrix	57
	2124	matrix	56
	2135	CH ₃ CN	51

**TABLE 2.6** (Continued)

#### **REFERENCES FOR SECTION 2.3**

- 1. Drayton, L. G.; Thompson, H. W. J. Chem. Soc. 1948, 1416-1419.
- 2. Arendale, W. F.; Fletcher, W. H. J. Chem. Phys. 1957, 26, 793-797.
- 3. Duncan, J. L.; Ferguson, A. M.; Harper, J.; Tonge, K. H. J. Mol. Spec. 1987, 125, 196-213.
- 4. Leszcznynski, J.; Kwiatowski, J. S. Chem. Phys. Lett. 1993, 201, 79-83.
- 4a. Ni, C.-K.; Wade, E. A.; Ashikhmin, M. V.; Moore, C. B. J. Mol. Spectrosc. 1996, 177, 285-293.
- Johns, J. W. C.; Nemes, L.; Yamada, K. M. T.; Wang, T. Y.; Domenech, J. L.; Santos, J.; Cancio, P.; Bermejo, D.; Ortigoso, J.; Escribano, R. J. Mol. Spectrosc. 1992, 156, 501–503.
- 5a. Hinze, R.; Zerbe-Foese, H.; Doose, J.; Guarnieri, A. J. Mol. Spectrosc. 1996, 176,133–138.
- 5b. Hatakeyama, S.; Honda, S.; Washida, N.; Akimoto, H. *Bull. Chem. Soc. Jpn.* **1985**, *58*, 2157–2162.
- 5c. Rowland, B.; Winter, P. R.; Ellison, G. B.; Radziszewski, J. G.; Hess, W. P. J. Phys. Chem. A 1999, 103, 9654–9670.
- 6. Moore, C. B.; Pimentel, G. C. J. Chem. Phys. 1963, 38, 2816-2829.
- 7. Kopper, H. Z. Phys. Chem. 1936, B34, 396-401.
- Nemes, L.; Luckhaus, D.; Quack, M.; Johns, J. W. C. J. Mol. Struct. 2000, 517–518, 217–226.
- 9. Gruebele, M.; Johns, J. W. C.; Nemes, L. J. Mol. Spectrosc. 1999, 198, 376-380.
- 10. Winther, F.; Meyer, S.; Nicolaisen, F. M. J. Mol. Struct. 2002, 611, 9-22
- 11. Melzer, A.; Jenny, E. F. Tetrahedron Lett. 1968, 4503-4506.
- 12. Gano, J. E.; Jacobson, R. H.; Wettach, R. H. Angew. Chem. Int. Ed. Engl. 1983, 22, 165–166.
- 13. McAllister, M. A.; Tidwell, T. T. Can. J. Chem. 1994, 72, 882-887.
- Wagner, B. D.; Arnold, B. R.; Brown, G. S.; Lusztyk, J. J. Am. Chem. Soc. 1998, 120, 1827–1834.
- Allen, A. D.; Colomvakos, J. D.; Diederich, F.; Egle I.; Hao, X.; Liu, R.; Lusztyk, J.; McAllister, M. A.; Rubin, Y.; Sung, K.; Tidwell, T. T.; Wagner, B. D. J. Am. Chem. Soc. 1997, 119, 12125–12130.
- 16. Leung-Toung, R.; Wentrup, C. J. Org. Chem. 1992, 57, 4850-4855.
- 16a. Leung-Toung, R.; Wentrup, C. Tetrahedron 1992, 48, 7641-7654.
- 17. Huang, B. S.; Pong, R. G. S.; Laureni, J.; Krantz, A. J. Am. Chem. Soc. 1977, 99, 4154–4156.
- 17a. Arnold, B. R.; Brown, C. E.; Lusztyk, J. J. Am. Chem. Soc. 1993, 115, 1576-1577.
- 18. Allen, A. D.; Porter, J.; Tahmassebi, D.; Tidwell, T. T. J. Org. Chem. 2001, 66, 7420-7426.
- 18a. Pietri, N.; Chiavassa, T.; Allouche, A.; Aycard, J.-P. J. Phys. Chem. A 1997, 101, 1093–1098.
- Piétri, N.; Gaudel-Siri, A.; Couturier-Tamburelli, I.; Pons, J.-M.; Aycard, J.-P. J. Phys. Chem. A 2005,
- 18c. Sumathi, R.; Chandra, A. K. Chem. Phys. Lett. 1997, 271, 287-295.
- 19. Allen, A. D.; Rangwala, H.; Saidi, K.; Tidwell, T. T.; Wang, J. *Russ. Chem. Bull.* 2001, 2130–2133.
- Allen, A. D.; Fenwick, M. H.; Henry-Riyad, H.; Tidwell, T. T. J. Org. Chem. 2001, 66, 5759–5763.

- 21. Saidi, K.; Tidwell, T. T. Arkivoc 2001, (xii), 100-107.
- 22. Tidwell, T. T.; Fenwick, M. T. Eur. J. Org. Chem. 2001, 3415-3419.
- 22a. Brown, R. D.; Godfrey, P. D.; Woodruff, M. Austral. J. Chem. 1979, 32, 2103-2109.
- 23. Krügerke, T.; Seppelt, K.. Chem. Ber. 1988, 121, 1977-1981.
- 24. Danheiser, R. L.; Sard, H. J. Org. Chem. 1980, 45, 4810-4812.
- 25. McCarney, C. C.; Ward, R. S. J. Chem. Soc. Perkin Trans. 1 1975, 1600-1603.
- 26. Mayr, H. Angew. Chem. Int. Ed. 1975, 14, 500-501
- 27. Colbourne, D.; Westwood, N. P. C. J. Chem. Soc., Perkin Trans. 2, 1985, 2049–2054.
- 28. Torres, M.; Ribo, J.; Clement, A.; Strausz, O. P. Nouv. J. Chim. 1981, 5, 351-352.
- 28a. Kötting, C.; Sander, W.; Senzlober, M.; Bürger, H. Chem Eur. J. 1998, 4, 1611-1615.
- 28b. Sander, W.; Kötting, C. Chem. Eur. J. 1999, 5, 24-28.
  - Allen, A. D.; Cheng, B.; Fenwick, M. H.; Huang, W.; Missiha, S. Tahmassebi, D.; Tidwell, T. T. Org. Lett. 1999, 1, 693–696.
  - 30. Russell, D. K.; Wesendrup, R. J. Chem. Phys. 2002, 117, 5252-5256.
  - 31. Bittner, J.; Seppelt, K. Chem. Ber. 1990, 123, 2187-2190.
  - 32. Davidovics, G.; Monnier, M.; Allouche, A. Chem. Phys. 1991, 150, 395-403.
  - 33. Gerry, M. C. L.; Lewis-Bevan, W.; Westwood, N. P. C. Can. J. Chem. 1985, 63, 676–677.
  - 34. Mal'tsev, A. K.; Zuev, P. S.; Nefedov, O. M. Izv. Akad. Nauk SSSR, Ser. Khim. 1985, 957–958. Engl. Transl. 1985, 876.
  - 35. Inoue, S.; Hori, T. Bull. Chem. Soc. Jpn. 1983, 56, 171-174.
  - 36. Shchukovskaya, L. L.; Kol'tsov, A. A.; Lazarev, A. N.; Pal'chik, R. I. Dokl. Akad. Nauk SSSR 1968, 179, 892–895. Engl. Transl. 1968, 179, 318–320.
  - 37. Valentí, E.; Pericas, M. A.; Serratosa, F. J. Org. Chem. 1990, 55, 395-397.
  - 38. Emerson, D. W.; Titus, R. L.; Gonzalez, R. M. J. Org. Chem. 1991, 56, 5301-5307.
  - 39. O'Neill, P.; Hegarty, A. F. J. Org. Chem. 1992, 57, 4421-4426.
  - 40. Lubenets, E. G.; Gerasimova, T. N.; Barkhash, V. A. Zh. Org. Khim. 1972, 8, 654; Engl. Transl. 1972, 8, 663.
  - 41. Unfried, K. G.; Curl. R. F. J. Mol. Spect. 1991, 150, 86-98.
  - 42. Bestmann, H. J.; Sandmeier, D. Chem. Ber. 1980, 113, 274-277.
  - 43. Meier, G.; Hoppe, M.; Lanz, K.; Reisenauer, H. P. Tetrahedron Lett. 1984, 25, 5645–5648.
  - 44. Agosta, W. C.; Smith, A. B., III. J. Am. Chem. Soc. 1971, 93, 5513-5520.
  - 45. Baird, M. S.; Dunkin, I. R.; Hacker, N.; Poliakoff, M.; Turner, J. J. J. Am. Chem. Soc. 1981, 103, 5190–5195.
  - 46. Baxter, G. J.; Brown, R. F. C.; Eastwood, F. W.; Harrington, K. J. *Tetrahedron Lett.* **1975**, 4283–4284.
  - 47. Gano, J. E.; Jacob, E. J. Spectrochim. Acta 1987, 43A, 1023-1025.
  - 48. Zhao, D.; Tidwell, T. T. J. Am. Chem. Soc. 1993, 114, 10980-10981.
  - 49. Chapman, O. L.; McIntosh, C. L.; Barber, L. L. Chem. Commun. 1971, 1162-1163.
  - 50. Mosandl, T.; Wentrup, C. J. Org. Chem. 1993, 58, 747-749.
  - 51. Liu, R. C.-Y.; Lusztyk, J.; McAllister, M. A.; Tidwell, T. T.; Wagner, B. D. J. Am. Chem. Soc. **1998**, 120, 6247–6251.

- 52. McMahon, R. J.; Chapman, O. L. J. Am. Chem. Soc. 1986. 108, 1713-1714.
- Acton, A. W.; Allen, A. D.; Antunes, L. M.; Federov, A. V.; Najafian, K.; Tidwell, T. T.; Wagner, B. D. J. Am. Chem. Soc. 2002, 124, 13790–13794.
- 53a. Kuhn, A.; Plüg, C.; Wentrup, C. J. Am. Chem. Soc. 2000, 122, 1945-1948.
- 54. Henry-Riyad, H.; Tidwell, T. T. Can. J. Chem. 2003, 81, 697-704.
- 55. Allen, A. D.; Fenwick, M. H.; Jabri, A.; Rangwala, H.; Saidi, K.; Tidwell, T. T. Org. Lett. 2001, 3, 4095–4098.
- 56. Friermuth, B.; Wentrup, C. J. Org. Chem. 1991, 56, 2286-2289.
- 57. Chapman, O. L.; Miller, M. D.; Pitzenberger, S. M. J. Am. Chem. Soc. 1987, 109, 6867–6868.
- 58. Wagner, B. D.; Arnold, B. R.; Brown, G. S.; Lusztyk, J. J. Am. Chem. Soc. 1998, 120, 1827–1834.
- 59. Allen, A. D.; Baigrie, L. M.; Gong, L; Tidwell, T. T. Can. J. Chem. 1991, 69, 138-145.
- 60. Baigrie, L. M.; Seikaly, H. R.; Tidwell, T. T. J. Am. Chem. Soc. 1985, 107, 5391-5396.
- 61. Naef, F.; Decorzant, R. Tetrahedron 1986, 42, 3245-3250.
- 62. Wuest, J. D.; Madonik, A. M.; Gordon, D. C. J. Org. Chem. 1977, 42, 2111-2113.
- 63. Back, T. G.; Barton, D. H. R.; Britten-Kelly, M. R.; Guziec, F. S., Jr. J. Chem. Soc., Perkin Trans. 1 1976, 2079–2089.
- 64. Cerioni, G.; Plumitallo, A.; Frey, J.; Rappoport, Z. Magn. Res. Chem. 1995, 33, 669–673.
- 65. Sammynaiken, R.; Westwood, N. P. C. J. Chem. Soc. Perkin Trans. 2 1989, 1987–1992.
- 66. Allen, A. D. unpublished results at the University of Toronto.
- 67. Schiess, P.; Radimerski, P. Helv. Chim. Acta 1974, 57, 2583-2597.
- 68. Krapcho, A. P.; Abegaz, B. J. Org. Chem. 1974, 39, 2251-2255.
- 69. Shustov, G. V.; Sun, F.; Sorensen, T. S.; Rauk, A. J. Chem. Soc., Perkin Trans. 2 1999, 1035–1041.
- 70. Krantz, A. J. Chem. Soc., Chem. Commun. 1973, 670-671.

#### 2.4 DIPOLE MOMENTS

The dipole moment of ketene was reported in 1946 as 1.45 D,¹ and the fact that this was less than the values for formaldehyde (2.27 D) and acrolein (3.04 D) provided evidence for the resonance structures of ketene, with high negative charge at C₂.¹ A highly precise value for the dipole moment of ketene of 1.42215(28) was obtained by a molecular beam electric resonance,² and microwave spectroscopy gave a value of 1.42429.³ Dipole moments were also obtained for the three lowest-lying vibrational states v₅, v₆, v₉ of ketene as 1.37408, 1.40433, and 1.42545 D, respectively.³

Dipole moments of many ketenes have been obtained computationally.⁴ The ketene  $CCl_2=C=O$  did not give a microwave spectrum, a result attributed to the absence of a dipole moment. A dipole moment was also measured for  $CH_2=CHCH=CH=C=O$  (Section 4.1.2). Dipole moments of 2.7 and 1.8 D were measured for the bisketene 4 and Me₃SiCH=C=O, respectively, confirming

the twisted structure of  $4.^{6}$ 



#### **REFERENCES FOR SECTION 2.4**

- 1. Hannay, N. B.; Smyth, C. P. J. Am. Chem. Soc. 1946, 68, 2408-2409.
- 2. Fabricant, B.; Krieger, D.; Muenter, J. S. J. Chem. Phys. 1977, 67, 1576-1586.
- 3. Hinze, R.; Zerbe-Foese, H.; Doose, J.; Guarnieri, A. J. Mol. Spect. 1996, 176, 133-138.
- 4. Gong, L.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1991, 113, 6021-6028.
- 5. Gerry, M. C. L.; Lewis-Bevan, W.; Westwood, N. P. C. Can. J. Chem. 1985, 63, 676-677.
- Werstiuk, N. H.; Ma, J.; McAllister, M. A.; Tidwell, T. T. J. Chem. Soc., Faraday Trans. 1994, 90, 3383–3390.

#### 2.5 MASS SPECTROMETRY AND GAS PHASE ION CHEMISTRY

The mass spectrometry of ketenes has been reviewed.¹ Ketenes are conveniently prepared in the gas phase by thermolytic dehydrochlorination of acyl chlorides, and an extensive set of alkylketenes **2** was generated quantitatively by flash vacuum thermolysis of the acyl chlorides **1** at 650 °C with loss of HCl (equation 1), and their mass spectra were measured (Table 2.7).²

TABLE 2.7 El	ectron Impact	Mass Spectra	of Ketenes from	Acyl	Chlorides (	(Ref 2	.)
--------------	---------------	--------------	-----------------	------	-------------	--------	----

Ketene	Principal Fragment Ions (relative abundance in parentheses)
$CH_2 = C = O$	42 (M ⁺ , 100), 41 (M ⁺ –H, 25)
MeCH=C=O	56 (M ⁺ , 100), 55 (M ⁺ –H, 21), 29 (MH ⁺ –CO, 25)
EtCH=C=O	70 (M ⁺ , 100), 55 (M ⁺ –CH ₃ , 97), 41 (M ⁺ –COH, 48)
n-PrCH=C=O	84 (M ⁺ , 34), 55 (M ⁺ $-C_2H_5$ , 100)
<i>n</i> -BuCH=C=O	98 (M ⁺ , 21), 55 (M ⁺ $-C_3H_7$ , 100)
<i>n</i> -PnCH=C=O	112 (M ⁺ , 25), 55 (M ⁺ $-C_4H_9$ , 100)
i-PrCH=C=O	84 (M ⁺ , 75), 69 (M ⁺ –CH ₃ , 80), 55 (M ⁺ –C ₂ H ₅ , 36) 41 (C ₃ H ₅ ⁺ , 100)
s-BuCH=C=O	98 (M ⁺ , 35), 69 (M ⁺ $-C_2H_5$ , 100), 55 (M ⁺ $-C_3H_7$ , 24)
t-BuCH=C=O	98 (M ⁺ , 52), 83 (M ⁺ –CH ₃ , 100), 55 (M ⁺ –C ₃ H ₇ , 80)
$Me_2C=C=O$	70 (M ⁺ , 100), 41 (C ₃ H ₅ ⁺ or C ₂ HO ⁺ , 74)
<i>n</i> -PrCMe=C=O	98 (M ⁺ , 46), 69 (M ⁺ $-C_2H_5$ , 52), 41 (M ⁺ $-C_3H_7$ , 100)
EtCMe=C=O	84 (M ⁺ , 50), 69 (M ⁺ – CH ₃ , 16), 55 (M ⁺ – C ₂ H ₅ , 53), 41 (C ₃ H ₅ +, 100)
$Et_2C=C=O$	98 (M ⁺ , 51), 83 (M ⁺ – CH ₃ , 19), 69 (M ⁺ – Et, 6), 55 (M ⁺ – C ₃ H ₇ , 100)

#### 50 SPECTROSCOPY AND PHYSICAL PROPERTIES OF KETENES

A characteristic feature of the mass spectra is the appearance of substantial molecular ion peaks  $M^+$ , which is due to the stabilization of the radical ions **3** formed, which have substantial acylium ion character (equation 2), as also noted by the ease of electron transfer from ketenes in solution (Section 5.1). The main fragmentation pathways of alkyl ketenes, as seen in Table 2.7, are  $\beta$ -cleavage giving conjugated cations **3**, with preferential loss of the larger R group, followed by further loss of CO forming vinyl cations **4** (equation 2).



Many stabilized silyl-substituted ketenes have been isolated and found to give parent ions by (EI) mass spectrometry.^{3–7} Loss of CO and of Me₃Si⁺ are prominent processes observed.^{3–7} Loss of alkyl groups from silicon in ketenes was also observed; thus, the parent ion from PhMe₂SiCH=C=O results from loss of CH₃.^{3–7} Mass spectra of some silylated bisketenes have also been reported.^{6,7}

Because of their stabilized acylium ion structures, ketene radical cations are frequently formed by fragmentation of various precursors, and this has applications for structure analysis. The electrospray mass spectrum of peptide **6** functionalized with a peroxycarbamate function on the lysine sidechain proceeded by formation of aminyl radical **7** (equation 3), which abstracted hydrogen forming **8**, which gave cleavage to the ketenyl fragment **9** (equation 4).⁸ This process was proposed as part of a strategy for directed fragmentation by free radical processes as a simplified method for protein analysis. Fragmentation of diacyl glycerophosphoethanolamine under low-energy, collision-activated dissociation with neutral loss of ketene was charge driven.⁹



The 70 eV mass spectrum of vinylketene CH₂=CHCH=C=O (10) gave a major parent ion 11 with a relative abundance of 67% compared to the fragment  $C_2H_4^+$  (equation 5).¹⁰ The ion 11 was also formed in the mass spectra of Z-crotonic acid 12 and of cyclohexenone 13 (equation 6).¹¹



The mass spectrum of the ketene PhOCH₂CH=C=O (14) generated by dehydrochlorination of the acyl chloride was observed and gave the same ion  $14^+$  formed by electron impact–induced Wolff rearrangement of 15 (equation 7).¹²



The kinetic energy release in the metastable fragmentation of  $CH_2=C=O^{13,14}$  and the photoionization mass spectra of  $CH_2=C=O$  and  $CD_2=C=O^{15}$  have been studied. Ketene complexes with ionic species in the gas phase have been observed, including the complex **17** formed from **16** (equation 8).¹⁶

$$CH_3 \xrightarrow{O} O^+ \longrightarrow H_H = C = O - - - H - CH_2 C \equiv O^+$$

$$I6 \qquad I7 \qquad (8)$$

Ketene and alkoxide/alcohol ion pairs in the gas phase formed hydrogen-bonded ynolate ions **18** (equation 9).¹⁷ The ketene radical cation **19** reacted with  $NH_3$  in a selected ion flow tube (SIFT) by two reaction channels, including formation of the ketenyl radical **20** (equation 10).^{18,19}

The ketene radical cation **19** was formed from the acetone radical ion,²⁰ and formed complexes detected in the gas phase with  $H_2O^{21}$  and with acetone.²² The reaction of **19** with ketene formed the distonic ion **20** (equation 11).²³

$$\begin{array}{cccccccccccccc}
H & H & H & C = 0 & \stackrel{-CO}{\longrightarrow} & H & C^{CH_2} & (11) \\
\end{array}$$
19 20

The ketene complex  $CH_2=NH_2^+\bullet\bullet\bullet CH_2=C=O$  (21) was formed in the 70 eV mass spectrum of  $CH_3CONH(CH_2)_2CH_3$ .²⁴ A complex of ketene with protonated acetone formed by rearrangement was also detected.²⁵ The molecular ion of  $CH_3CH(OH)CH_2CO_2Me$  also formed a ketene complex.²⁶ The radical cation of hydroxyketene 22 was formed by dissociative electron ionization of glycolic acid, and formed the ions 23 and 24, which fragmented, with formation of both CO and  $CH_2=O^{+\bullet}$  (equation 12).²⁷



#### **REFERENCES FOR SECTION 2.5**

- 1. Schwarz, H.; Köppel, C. In *The Chemistry of Ketenes, Allenes, and Related Compounds*; Patai, S., Ed.; Wiley, New York, 1980; pp. 189–222.
- 2. Marquestiau, A.; Flammang, R.; Pauwels, P. Org. Mass. Spectrosc. 1983, 18, 547-552.
- 3. Dalton, A. M.; Zhang, Y.; Davie, C. P.; Danheiser, R. L. Org. Lett. 2002, 4, 2465-2468.
- 4. Allen, A. D.; Tidwell, T. T. Can. J. Chem. 1999, 77, 802-805.
- 5. Liu, R.; Marra, R. M.; Tidwell, T. T. J. Org. Chem. 1996, 61, 6227-6232.
- 6. Sung, K.; Tidwell, T. T. Organometallics 1997, 16, 78-85.
- 7. Sung, K.; Tidwell, T. T. J. Am. Chem. Soc. 1996, 118, 2768-2769.
- Masterson, D. S.; Yin, H.; Chacon, A.; Hachey, D. L.; Norris, J. L.; Porter, N. A. J. Am. Chem. Soc. 2004, 126, 720–721.
- 9. Hsu, F.-F.; Turk, J. J. Am. Soc. Mass. Spectrom. 2000, 11, 892-899.
- 10. Terlouw, J. K.; Burgers, P. C.; Holmes, J. L. J. Am. Chem. Soc. 1979, 101, 225-226.
- 11. Holmes, J. L.; Terlouw, J. K.; Vijfhuizen, P. C.; A'Campo, C. Org. Mass. Spectrom. 1979, 14, 204–212.
- 12. Selva, A.; Ferrario, F.; Saba, A. Org. Mass. Spectrom. 1987, 22, 189-196.
- 13. Lorquet, J. C. Int. J. Mass Spectrom. Ion Phys. 1981, 38, 343-350.
- 14. Stockbauer, R. Int. J. Mass Spectrom. Ion Phys. 1977, 25, 401-410.

- 15. Vogt, J.; Williamson, A. D.; Beauchamp, J. L. J. Am. Chem. Soc. 1978, 100, 3478–3483.
- Tortajada, J.; Berthomieu, D.; Morizur, J.-P.; Audier, H.-E. J. Am. Chem. Soc. 1992, 114, 10874–10880.
- 17. Hayes, R. N.; Sheldon, J. C.; Bowie, J. H. Austr. J. Chem. 1985, 38, 355-362.
- 18. Iraqi, M.; Lifshitz, C.; Reuben, B. G. J. Phys. Chem. 1991, 95, 7742-7746.
- Drewello, T.; Heinrich, N.; Maas, W. P. M.; Nibbering, N. M. M.; Weiske, T.; Schwarz, H. J. Am. Chem. Soc. 1987, 109, 4810–4818.
- 20. Heinrich, N.; Louage, F.; Lifshitz, C.; Schwarz, H. J. Am. Chem. Soc. 1988, 110, 8183–8192.
- 21. Heinrich, N.; Schwarz, H. Int. J. Mass Spectrom. Ion Proc. 1987, 79, 295-310.
- 22. Stirk, K. M.; Kenttämaa, H. I. Org. Mass. Spectrom. 1992, 27, 1153-1154.
- 23. Smith, R. L.; Franklin, R. L.; Stirk, K. M.; Kenttämaa, H. I. J. Am. Chem. Soc. 1993, 115, 10348–10355.
- Burgers, P. C.; Kingsmill, C. A.; McGibbon, G. A.; Terlouw, J. K. Org. Mass. Spectrom. 1992, 27, 398–405.
- 25. Van Baar, B. L. M.; Terlouw, J. K.; Akkok, S.; Zummack, W.; Weiske, T.; Schwarz, H. *Chimia*, **1988**, *42*, 226–229.
- 26. Morizur, J.-P.; Chapon, E.; Berthomieu, D. Org. Mass. Spectrom. 1991, 26, 1119–1120.
- 27. Polce, M. J.; Song, W.; Cerda, B. A.; Wesdemiotis, C. Eur. J. Mass Spectrom. 2000, 6, 121–130.

# **3** Preparation of Ketenes

#### 3.1 KETENES FROM KETENE DIMERS

Reviews of the chemistry of the ketene dimer have appeared.^{1,2} Pyrolysis of ketene dimer (1) provides a simple synthesis of ketene (2) (equation 1).³ The reaction of ozone with (1) gave the elusive malonic anhydride (3), and this formed ketene by loss of carbon dioxide (equation 2).^{4,5}



For preparation of norbornylideneketene (8), dimer 6 of ketene 5 was first made by treatment of 2-norbornenyl carbonyl chloride (4) with triethylamine (equation 3).⁶ Hydrogenation gave (7), and flash vacuum pyrolysis at 590–600 °C gave ketene 8 (IR 2110 cm⁻¹), which was trapped at -78 °C (equation 4). Reaction of ketene 8 with diazomethane is discussed in Section 5.10.1. Dimethylketene has also been prepared by pyrolysis of the dimer.⁷



Ketenes II, by Thomas T. Tidwell

Copyright © 2006 John Wiley & Sons, Inc.



Photolysis of *cis* or *trans* cyclobutanedione **9** was suggested to give the diradical **10**, which formed only 5-12% yields of trapping products of ketene **11**, along with ketones **12** and **13** (equations 5, 6).⁸ Photolysis of some dimers of dialkylketenes including **14** also gave the ketenes in yields of 20-30%.^{9,10}



Photolysis of the cyclobutanedione (16) in  $CH_2Cl_2$  gave pentamethyleneketene (17), as identified by IR absorption at 2100 cm⁻¹ (equation 8).¹¹ Photolysis in MeOH gave the ester 3 in 20% yield (equation 8).¹¹ Tetramethyleneketene was formed in a similar fashion, but ketene formation was less efficient for smaller and larger ring sizes.¹¹



#### **REFERENCES FOR SECTION 3.1**

- Clemens, R. J.; Witzeman, J. S. In Acetic Acid and Its Derivatives; Agreda, V. H.; Zoeller, J. R., Eds.; Dekker, New York, 1993, pp. 173–224.
- 2. Clemens, R. J. Chem. Rev. 1986, 86, 241-318.
- 3. Andreades, S.; Carlson, H. D. Org. Synth., 1973, 5, 679-684.
- 4. Perrin, C. L.; Arrhenius, T. J. Am. Chem. Soc. 1978, 100, 5249-5251.
- 5. Perrin, C. L.; Magde, D.; Berens, S. J.; Roque, J. J. Am. Chem. Soc. 1980, 45, 1705-1706.
- 6. Cordes, M. H. J.; Berson, J. A. J. Am. Chem. Soc. 1996, 118, 6241-6251.
- 7. Moore, H. W.; Wilbur, D. S. J. Org. Chem. 1980, 45, 4483-4491.
- 8. Zang, H.; Neckers, D. C. J. Org. Chem. 1999, 64, 2103-2106.
- Turro, N. J.; Leermakers, P. A.; Wilson, H. R.; Neckers, D. C.; Byers, G. W.; Vesley, G. F. J. Am. Chem. Soc. 1965, 87, 2613–2619.
- 10. Haller, I.; Srinivasan, R. J. Am. Chem. Soc. 1965, 87, 1144-1145, 2619.
- 11. Krapcho, A. P.; Abegaz, B. J. Org. Chem. 1974, 39, 2251-2255.

## **3.2 KETENES FROM CARBOXYLIC ACIDS AND THEIR DERIVATIVES**

#### 3.2.1 Ketenes from Carboxylic Acids and Anhydrides

**3.2.1.1** *Ketenes from Carboxylic Acids* The direct preparation of ketene by the pyrolysis of acetic acid is not practical for laboratory use but has been used on an industrial scale (equation 1),¹ and the kinetics of acetic acid thermolysis forming ketene and other products have been reported.² The pilot-scale preparation of ketene by pyrolysis of acetic acid has been studied and enhanced by model-aided optimization.³

$$\begin{array}{c} O \\ CH_{3} \\ OH \\ H_{2}O \\ H_{2}O \\ H \\ H \\ H \\ H \\ H \\ H \end{array} C=O$$
(1)

Many other procedures have been studied to effect this transformation. Catalytic dehydration of acetic acid to ketene on metal oxides can form ketene,^{4,5} and the C₃ to C₅ carboxylic acids have been dehydrated with a catalyst to the corresponding ketenes over a packed bed of high surface area SiO₂ above 700 K,^{6,7} The catalytic decomposition of acetic acid at -50 °C gave ketene, which was used to acetylate glycol monoethers.⁸ The IR multiphoton dissociation of acetic acid gave ketene and water by one reaction channel, while decarboxylation to CH₄ and CO₂ accounted for roughly 35% of the total dissociation.⁹ However, thermal and catalytic dehydration do not usually provide convenient laboratory preparations of ketene.

In the catalytic hydrogenation of acetic acid over zirconia to produce acetaldehyde, the formation of acetone was also observed, and it was proposed that the reaction involved removal of an  $\alpha$ -hydrogen from the acetic acid, forming ketene, which could transfer CH₂ to acetic acid, forming acetone.¹⁰

Dehydration of acetic acid on  $UO_2$  single crystals yields ketene,^{11,12} and acetic acid and isobutyric acid on functionalized silica monoliths form ketene and dimethylketene, respectively.¹³ Thermal decomposition of thulium acetate hydrate [Tm(OAc)₃•4H₂O] forms ketene, acetic acetate, acetone, and methane.¹⁴

The reaction of bromoacetic acid on  $TiO_2$  single crystal surfaces results in the formation of ketene and the ketene dimers **2** and **3** (equation 2).¹⁵ Acetic acid on  $TiO_2$  was found to form ketene near 290 °C.¹⁶



On a laboratory scale, the formation of ketenes from carboxylic acids may be achieved in certain cases. The very stable ketenes  $t-Bu_2C=C=O$  (4) and Me₃SiCH=C=O (5) were prepared by the direct dehydration of the carboxylic acids using dicyclohexyl carbodimide (DCC) (equation 3).¹⁷

$$\stackrel{t-\mathrm{Bu}}{\longrightarrow} \operatorname{CO}_{2}\mathrm{H} \xrightarrow{\mathrm{DCC}} \stackrel{t-\mathrm{Bu}}{\longrightarrow} \operatorname{C=O} _{t-\mathrm{Bu}}$$
(3)  
$$4 (70\%)$$

Reaction of carboxylic acids **5** with strongly electron-withdrawing  $\alpha$ -substituents with DCC in the presence of crowded alcohols rapidly gave high yields of esters at room temperature via ketenes **6** (equation 4).^{18,19} The formation of ketenes was implicated by the incorporation of deuterium when *t*-BuOD was used and by the formation of [4 + 2] cycloaddition of adducts of ketenes with DCC in some cases.^{18,19} Competitive reactions showed that MeOH, EtOH, and *c*-PrOH reacted with (EtO)₂POCH=C=O (**6a**) at almost equal rates, while *t*-BuOH was 15 times less reactive. Ketenes **6** generated in this way reacted selectively with methanol in preference to more acidic alcohols or many phenols.²⁰

$$R \xrightarrow{O}_{OH} \xrightarrow{DCC}_{CH_3CN} \xrightarrow{R}_{H} C=O \xrightarrow{r-BuOH} R \xrightarrow{O}_{OBu-t} (4)$$
5 6  $R = (EtO)_2 PO (a), EtCO_2 (b), NC (c), 4-CH_3C_6H_4SO_2 (d)$ 

Triphosgene was used to react with carboxylic acids **5** to generate ketenes **7**, and in the presence of imines in dichloromethane at -40 °C it provided an efficient preparation of  $\beta$ -lactams **8** (equation 5).^{21,22}



Mukaiyama's reagent **9** (2-choro-1-methylpyridinium tosylate or iodide)²³ effects the in situ dehydration of carboxylic acids to ketenes, and was suggested to react with the acids by formation of 2-pyridinium carboxylate esters, which gave ketenes and *N*-methylpyridone **10** (equation 6).²³



Optimized conditions using Mukaiyama's reagent for generation of ketene **13** from the optically active glycine-derived substrate **11** in the presence of imines **12** led to  $\beta$ -lactams **14** in yields of  $\geq 90\%$  and selectivities for the *cis*-product of 90 to 99% (equation 7).²⁴ These reactions were carried out under mild conditions with all the reagents present together, implicating the formation of ketene intermediates.²⁴ Mukaiyama's reagent has also been utilized on a solid support²⁵ as well as in solution^{26,27} for ketene formation for reaction with imines forming  $\beta$ -lactams, and **9** was used for ketene generation for intramolecular cyclization.²⁸



Cyanuric chloride  $(15)^{29,30}$  also served as a dehdrating agent for formation of a ketene or ketene equivalent, as in the reaction of the acid 16 forming

phenylthioketene (17), which underwent [2 + 2] cycloaddition with the imine benzylideneaniline forming  $\beta$ -lactam 18 (equations 8, 9).^{29,30}



#### **REFERENCES FOR SECTION 3.2.1.1**

- Wagner, F. S. Kirk-Othmer Encyclopedia of Chemical Technology, 3rd ed.; Wiley, New York, 1980, Vol 1, p. 155. Abaecherli, C.; Miller, R. J. Kirk-Othmer Encyclopedia of Chemical Technology, ed; Wiley, New York, 1995,
- 2. Mackie, J. C.; Doolan, K. R. Int. J. Chem. Kinet. 1984, 16, 525-541.
- 3. Bruns, M; Baurmeister, J.; Schafer, T.; Schuppert, A. *Nachr. Chem., Tech. Lab.* **1997**, *45*, 483–485.
- 4. Martinez, R.; Huff, M. C.; Barteau, M. A. J. Catal. 2004, 222, 404-409.
- 5. Barteau, M. A. Chem. Rev. 1996, 96, 1413-1430.
- Libby, M. C.; Watson, P. C.; Barteau, M. A. Ind. Eng. Chem. Res. 1994, 33, 2904– 2912.
- 7. Watson, M. A.; Libby, M. C.; Barteau, M. A. U.S. Patent 5,475,144, 1995.
- Schnurpfeil, D.; Wiegner, J.-P.; Hillmann, H.; Lauche, H.; Schultz, H.-P. Ger. Offen. DE 19532697 A1 6, 1997. *Chem. Abs.* 1997, *126*, 211830.
- 9. Longfellow, C. A.; Lee, Y. T. J. Phys. Chem. 1995, 99, 15532-15537.
- Pestman, R.; van Duijne, A.; Pieterse, J. A. Z.; Ponec, V. J. Mol. Cat. A, Chemical 1995, 103, 175–180.
- 11. Chong, S. V.; Idriss, H. J. Vacuum Sci. Technol. A 2000, 18, 1900-1905.
- 12. Chong, S. V.; Idriss, H. J. Vacuum Sci. Technol. A 2001, 19, 1933-1937.
- 13. Martinez, R.; Huff, M. C.; Barteau, M. A. Appl. Catal. A 2000, 200, 79-88.
- Hussein, G. A. M.; Mekhemer, G. A. H.; Balboul, B. A. A. Phys. Chem. Chem. Phys. 2000, 2, 2033–2038.

- 15. Titheridge, D. J.; Wilson, J. N.; Idriss, H. Res. Chem. Intermed. 2003, 29, 553-574.
- 16. Tanner, R. E.; Liang, Y.; Altman, E. I. Surface Sci. 2002, 506, 251-271.
- 17. Olah, G. A.; Wu, A.; Farooq, O. Synthesis 1989, 568.
- 18. Nahmany, M.; Melman, A. Org. Lett. 2001, 3, 3733-3735.
- 19. Shelkov, R.; Nahmany, M.; Melman, A. J. Org. Chem. 2002, 67, 8975-8982.
- 20. Shelkov, R.; Nahmany, M.; Melman, A. Org. Biomol. Chem. 2004, 397-401.
- Krishnaswamy, D.; Govande, V. V.; Gumaste, V. K.; Bhawal, B. M.; Deshmukh, A. R. A. S. *Tetrahedron* 2002, 58, 2215–2225.
- 22. Krishnaswamy, D.; Bhawal, B. M.; Deshmukh, A. R. A. S. *Tetrahedron Lett.* **2000**, *41*, 417–419.
- Armstrong, A. Encyclopedia of Reagents for Organic Synthesis; Wiley, New York, 1995, pp. 1174–1175.
- 24. Matsui, S.; Hashimoto, Y.; Saigo, K. Synthesis 1998, 1161-1166.
- 25. Donati, D.; Morelli, C.; Porcheddu, A.; Taddei, M. J. Org. Chem. 2004, 69, 9316-9318.
- 26. Amin, S. G.; Glazer, R. D.; Manhas, M. S. Synthesis, 1979, 210-213.
- 27. Georg, G. I.; Mashava, P. M.; Guan, X. Tetrahedron Lett. 1991, 32, 581-584.
- 28. Funk, R. L.; Abelman, M. M.; Jellison, K. M. Synlett 1989, 36-37.
- 29. van der Veen, J. M.; Bari, S. S.; Krishnan, L.; Manhas, M. S.; Bose, A. K. *J. Org. Chem.* **1989**, *54*, 5758–5762.
- 30. Manhas, M. S.; Bose, A. K.; Khajavi, Synthesis 1981, 209-211.

*3.2.1.2 Ketenes from Acid Anhydrides* Pyrolysis of acetic anhydride using a platinum wire was used by Wilsmore in the first preparation of ketene (equation 1).^{1,2} Laboratory procedures using this method are available.³

$$CH_{3} \xrightarrow{O} O CH_{3} \xrightarrow{500-510 \circ C} H \xrightarrow{H} C=O$$

$$H$$

$$1 (26\%)$$

$$(1)$$

IR laser-powered pyrolysis (IRLPP) of acetic anhydride with SF₆ as a sensitizer has been used to generate CH₂=C=O for direct IR observation.⁴ The kinetics of the thermal unimolecular conversion of acetic anhydride to ketene and acetic acid were measured between 750 and 980 K and give Arrhenius parameters  $A = 11.8 \text{ s}^{-1}$ and  $E_a = 138 \text{ kJ/mol}$ , with  $k = 10^{12.2} (\text{exp} - 145 \text{ kJ/mol}^{-1}/\text{RT})\text{s}^{-1}$ .⁴ Ab initio calculations favored a six-membered cyclic transition state with a barrier of 156 kJ/mol at the MP4 (SDTQ) level (equation 2).⁵ Ketene formation from acetic anhydride with zeolite (BEA) in the presence of anisole gave 4-methoxyacetophenone.⁶

$$\begin{array}{c} 0 & 0 \\ CH_3 & \longrightarrow \end{array} \xrightarrow{\Lambda} \left[ \begin{array}{c} 0 & -\zeta \\ H & 0 \\ -\zeta \\ 0 \end{array} \right] \xrightarrow{H} H = C = 0 + CH_3CO_2H \quad (2)$$

#### 62 PREPARATION OF KETENES

Methylketene (2) has been generated effectively from pyrolysis of propionic anhydride and has been obtained as a green solution usable for further reactions.⁷ Trapping of the pyrolysis vapors in a solution of a catalytic amount of quinidine in (THF) or CH₂Cl₂ at -78 °C gave the dimer **3** with 99% ee (equation 3), and this could be reacted with amine nucleophiles to form chiral amides.⁸

Carboxylic acid **4** reacted with sodium acetate in refluxing acetic acid (Perkin conditions) to form the acid anhydride, which yielded ketene **5**, which gave intramolecular [2 + 2] cycloaddition with the pendent alkenyl group forming cyclobutanone **6** (equation 4).⁹⁻¹¹ The  $\beta$ -lactone **8** was formed from the ketoketene **7** generated under these conditions, and underwent decarboxylation (equation 5).¹²



The base-induced formation of the ketene 10 from acid anhydride 9, forming 11, is a mechanistic pathway proposed in a total synthesis of furaquinocins (equation 6).¹³


Ketenes such as **12** have been formed from malonic acids by conversion to mixed anhydrides with trifluoroacetic anhydride and thermolysis (equation 7).¹⁴ Reaction of carboxylic acid **13** with trifluoroacetic anhydride formed the mixed anhydride **14** (equation 8). This led to the dienylketene **15**, which cyclized to phenol **16** (equation 9).¹⁵ Anhydrides from carboxylic acids and ethyl chloroformate reacted similarly.¹⁵



Mixed tosylate anhydrides have also been used to generate ketenes from carboxylic acids, as in the conversion of acid **17** with toluenesulfonyl chloride forming a mixed anhydride **18** (equation 10), which was converted to ketene **19**, which reacts by intramolecular [2 + 2] cycloaddition, forming cyclobutanone **20** (equation 11).^{16–18} Mixed anhydrides **21** of cyclohexanecarboxylic acid with sulfonic acids gave cyclohexylideneketene (**22**) on pyrolysis (equation 12).¹⁹ Similarly mixed anhydrides of CH₃SO₃H with carboxylic acids are reported to form ketenes on pyrolysis at 130 °C²⁰ or on reaction with Et₃N at 0 °C.²¹



#### 64 PREPARATION OF KETENES

Reactions of carboxylic acids with phosphoryl chlorides allowed the in situ generation of ketenes,^{22–27} which evidently formed from intermediate anhydrides, as in the conversion of the acid to anhydride **23**, forming ketene **24** (equation 13), which reacted by [2+2] cycloaddition with imines, forming  $\beta$ -lactams **25** (equation 14).²⁵



Photolysis of the mixed anhydride **26** formed ketene **22**, which was captured by methanol forming the ester **27** in a 70% yield (equation 15).²⁸



## **REFERENCES FOR SECTION 3.2.1.2**

- 1. Wilsmore, N. T. M. J. Chem. Soc. 1907, 91, 1938-1941.
- 2. Peytral, E. J. M. French Patent 722,477. Chem. Abstr. 1932, 24, 4063.
- 3. Fisher, G. J.; Maclean, A. F.; Schnizer, A. W. J. Org. Chem. 1953, 18, 1055-1057.
- 4. Hussein, G. A. M.; Mekhemer, G. A. H.; Balboul, B. A. A. Phys. Chem. Chem. Phys. 2000, 2, 2033–2038.
- Akao, M.; Saito, K.; Okada, K.; Takahashi, O; Tabayashi, K. Ber. Bunsen-Ges. Phys. Chem. 1996, 100, 1237–1241.
- 6. Bonati, M. L. M.; Joyner, R. W.; Stockenhuber, M. Catalysis Today, 2003, 81, 653-658.
- 7. Jenkins, A. D. J. Chem. Soc. 1952, 2563-2568.
- 8. Calter, M. A.; Guo, X.; Liao, W. Org. Lett. 2001, 3, 1499-1501.
- 9. Beereboom, J. J. J. Am. Chem. Soc. 1963, 85, 3525-3526.
- 10. Brady, W. T.; Gu, Y. J. Heterocyclic Chem. 1988, 25, 969-971.
- 11. Marotta, E.; Medici, M.; Righi, P.; Rosini, G. J. Org. Chem. 1994, 59, 7529-7531.
- 12. Brady, W. T.; Gu, Y. J. Org. Chem. 1988, 53, 1353-1356.

- Saito, T.; Suzuki, T.; Morimoto, M.; Akiyama, C.; Ochiai, T.; Takeuchi, K.; Matsumoto, T.; Suzuki, K. J. Am. Chem. Soc. 1998, 120, 11633–11644.
- 14. Duckworth, A. C. J. Org. Chem. 1962, 27, 3146-3148.
- 15. Serra, S.; Fuganti, C.; Moro, A. J. Org. Chem. 2001, 66, 7883-7888.
- 16. Brady, W. T.; Marchand, A. P.; Giang, Y. F.; Wu, A.-H. Synthesis 1987, 395-396.
- Brady, W. T.; Giang, Y. F.; Marchand, A. P.; Wu, A. J. Org. Chem. 1987, 52, 3457– 3461.
- 18. Corey, E. J.; Rao, K. S. Tetrahedron Lett. 1991, 32, 4623-4626.
- 19. Tempesti, E.; Giuffre, L.; Sioli, G.; Fornaroli, M.; Airoldi, G. J. Chem. Soc. Perkin Trans. 1 1974, 771–773.
- 20. Karger, M. H.; Mazur, Y. J. Org. Chem. 1971, 36, 528-531.
- 21. Karger, M. H.; Mazur, Y. J. Org. Chem. 1971, 36, 532-540.
- 22. Arrieta, A.; Lecea, B.; Palomo, C. J. Chem. Soc., Chem. Commun. 1987, 1743-1744.
- 23. Arrieta, A.; Lecea, B.; Cossió, F. P.; Palomo, C. J. Org. Chem. 1988, 53, 3784– 3791.
- 24. Arrieta, A.; Cossio, F. P.; Palomo, C. Tetrahedron. 1985, 41, 1703-1712.
- 25. Cossio, F. P; Ganboa, I.; Garcia, J. M.; Lecea, B.; Palomo, C. *Tetrahedron Lett.* **1987**, *28*, 1945–1948.
- 26. Arrieta, A.; Lecea, B.; Palomo, C. J. Chem. Soc., Perkin Trans. 1 1987, 845-850.
- 27. Palomo, C.; Cossio, F. P.; Cuevas, C.; Lecea, B.; Mielgo, A.; Román, P.; Luque, A.; Martinez-Ripoll, J. Am. Chem. Soc. 1992, 114, 9360–9369.
- 28. Penn, J. H.; Owens, W. H. J. Am. Chem. Soc. 1993, 115, 82-86.

## 3.2.2 Ketenes from Acyl Halides and Activated Acids

The preparation and isolation of ketenes by dehydrohalogenation was confirmed by Staudinger in 1911,¹ although Wedekind had reacted diphenylacetyl chloride with *n*-Pr₃N in 1901 but failed to isolate the ketene.² This method has usually been successful for the isolation of ketenes only when these are stabilized by the presence of two bulky or aryl groups, as in the examples of diphenylketene (**2**, equation 1),³ di*-tert*-butylketene (**4**, equation 2),⁴ cycloheptylideneketene **6** (equation 3),⁵ and 1,3,3-trimethylnorbornylidene-ketene (**8**, equation 4).⁶ Ketene **4** was also prepared by dehydrochlorination at 80 °C with *n*-Bu₃N, IR 2090 cm⁻¹ (Section 4.1.1).^{6a}

$$\begin{array}{cccc} Ph & O & \xrightarrow{Et_3N, Et_2O, 0 \circ C} & Ph \\ Ph & Cl & & Ph \\ & & & Ph \end{array} C=O \\ & & & & 1 \end{array}$$
(1)

$$\begin{array}{cccc} t - Bu & O & \xrightarrow{Et_3N} & t - Bu \\ t - Bu & Cl & \xrightarrow{ultrasound} & t - Bu \\ 3 & 4 & 6 (86\%) \end{array}$$
(2)



Dehydrochlorination was also successful in the preparation of the very stable trimethylsilyketene (9, equation 5).⁷

$$\begin{array}{ccc} Me_{3}Si & & & Me_{3}Si \\ & & & \\ Cl & & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\$$

Other diarylketenes prepared by dehydrochlorination that have been isolated include bis(pentamethylphenyl)ketene,⁸ ditipylketene (tiptyl is 2,4,6-triisopropylphenyl),^{9,10} and bis(pentachlorophenylketene.¹¹ Alkyl(aryl)ketenes have also been isolated, including methyl(phenyl)ketene,¹² ethyl(phenyl)ketene,¹³ and ethyl(2-tolyl)ketene (**10**, equation 6).¹⁴

$$\begin{array}{c} 2\text{-Tol} \\ \underset{\text{Et}}{\overset{\text{O}}{\longrightarrow}} Cl & \xrightarrow{\text{Me}_{2}\text{NEt}} & 2\text{-Tol} \\ \underset{\text{Et}}{\overset{\text{O}}{\longrightarrow}} C=0 \\ \underset{\text{Et}}{\overset{\text{O}}{\longrightarrow}} C=0 \end{array}$$
(6)

The isolation of more reactive ketenes prepared by dehydrohalogenation may be difficult because of other reactions, such as dimerization. There is evidence that these reactions involve acylammonium intermediates and that the dehydrohalogenation step is reversible. Thus, in the reaction of isobutyryl chloride (**11**) with triethylamine at -60 °C, an IR absorption at 1810 cm⁻¹ ascribed to the acylammonium salt **12** was observed, and a signal at  $\delta$  1.4 in the ¹H NMR was assigned as the ketene **13** (equation 7).^{15,16}

$$\begin{array}{cccc}
CH_3 & O & E_{t_3N} & CH_3 & O & CH_3 & O \\
CH_3 & Cl & -60 \,^{\circ}C & CH_3 &$$

In the preparation of diphenylketene (2, equation 1) by dehydrochlorination in refluxing chloroform, the by-product 14 was formed to the extent of 6-9%

(equations 8, 9).¹⁷ The mechanism proposed for the formation of **14** involved initial hydride transfer to **2**, forming the corresponding enolate.¹⁷



In esterifications using acyl chlorides RCOCl ( $R = CH_3$ , ClCH₂, PhOCH₂, and PhCH₂) with CH₃OD catalyzed by Et₃N ketene formation was shown to be competitive with direct addition in product formation.¹⁸ The reaction of CH₃CHClCOCl with Et₃N followed a complex path and was proposed to involve formation of CH₃CCl=C=O (**15**), which added Et₃N, forming the zwitterion **16**, but the ketene was not observed even using stopped flow NMR (equation 10).¹⁹ The formation of dimers of chloroketene and dichloroketene as unobserved products in Et₃N-catalyzed dehydrochlorination reactions is discussed in Section 4.4.2.

In the reaction of the chiral acyl chloride **17** in the presence of imine **19** catalyzed by *i*-Pr₂NEt, a strong band in the IR was observed at  $2110 \text{ cm}^{-1}$  due to ketene **18** (equation 11).¹⁸ When the reaction was monitored by IR at -22 °C the appearance of the ketene and the slower formation of the  $\beta$ -lactam **20** were observed, and the kinetics were fit to the reaction of equation 11.²⁰



Reactive ketenes generated by dehydrochlorination are frequently trapped in situ with other reagents, and such processes can be convenient, giving high yields of products with high selectivity. An example is the dehydrochlorination of **21** forming 3-chloropropylketene **22**, which underwent [2 + 2] cycloaddition with the

dihydropyrrole 23, forming the cyclobutanone 24 (equation 12).²¹



Efficient intramolecular [2+2] cycloadditions with pendant alkenyl groups also occur with reactive ketenes generated by dehydrochlorination. Cyclization of ketene 25 formed in this way proceeded in only a 3% yield with R=H, but for the more activated double bond with  $R=CH_3$  the yield of 26 was 80% (equation 13).²² The more reactive chloro-substituted ketene 27 gave a 58%yield of 28 when generated by activation of the acid with tosyl chloride and triethylaminc (equation 14).²³ This methodology was applied in synthesis using ketene **29** for the preparation of a precursor **30** to retigeranic acid (equation 15).²⁴ These reactions are relevant to the mechanism of ketene [2+2] cycloadditions, as they are geometrically restrained from a concerted  $[\pi 2_s + \pi 2_a]$  pathway, and a  $[\pi 2_a + \pi 2_a]$  pathway is forbidden, so a stepwise pathway is evidently followed.^{25,26}



 $\alpha$ -Aryl- $\alpha$ -cyanomethylketenes **32** generated by dehydrochlorination of the acyl chlorides **31** in the presence of either *R*- or *S*-*N*-phenylpantolactam (**33**) formed the cyano esters **34** in nearly quantitative yields with diastereometric ratios up to 93:7 (equation 16).²⁷



There have been significant improvements in the generation of ketene solutions by dehydrochlorination using the combination of a stochiometric base that reacts with the HCl generated and a catalytic amount of a trialkylamine that serves as a kinetic shuttle base.^{28–31} The amine rapidly effects the dehydrochlorination and then transfers the proton to the stronger base, with regeneration of the catalytic base (Scheme 3.1). Bases that were screened included bis(1,8-dimethylamino)-naphthalene (**30**), NaH with 15-crown-5, K₂CO₃, and the polymer bound base BEMP, which is a triaminophosphonamide imine bound to a polymer support. The use of chiral tertiary amines permits catalytic asymmetric reactions of the ketenes.



Of the bases, BEMP is insoluble and has the advantage that the ketene can be produced in solution by dehydrochlorination, either by passing the solution of the acyl chloride through a column containing BEMP or by stirring the two together and filtering off the ketene solution. However,  $K_2CO_3$  and NaH are much cheaper, and for generation of ketenes such as phenyketene (**31**) react by a shuttle mechanism with dehydrohalogenation of the acyl chloride **29** by the amine  $R_3N$ , forming **31** in solution (Scheme 3.1).^{28,29} The catalyst  $R_3N$  acts as a shuttle base in toluene



Scheme 3.1 Shuttle mechanism for formation of phenylketene (31) by dehydrochlorination using NaH and  $R_3N$  a shuttle base.

solution, forming the initial  $R_3NH^+Cl^-$  salt, which transfers the proton to NaH, forming H₂ and regenerating  $R_3N$ . As described in Section 4.1.4, this methodology has been adapted to the generation of ketenes that were directly observable by IR. Other developments include the use of bases on solid supports²⁹ and insoluble  $K_2CO_3$  for dehydrochlorination forming ketenes.^{28–31}

When chiral amines such as benzoylquinine (BQ) are used as the shuttle base, these can catalyze asymmetric reactions of the ketene, as in the reaction of benzyloxyketene (**32**) with the imine **33**, forming the  $\beta$ -lactam **34** in 60% yield, 99% ee, and 8/1 *cis/trans* ratio (equation 17).³¹ When 1,8-bis(dimethylamino)naphthalene (**30**) is used, the deprotonation of the acyl chloride may be reversible, and in some cases the reaction may go though an acylammonium ion and not a free ketene.³¹



The use of other chiral alkaloid bases as catalyts also permitted asymmetric reactions of ketenes generated by dehydrochlorination of acyl chlorides **35**. In a careful study, the reaction of propionyl chloride (**35a**) with diisopropylethylamine (DIPEA) in CH₂Cl₂ at room temperature in the presence of trimethylsilylquinine (**37**) gave methylketene (**36a**), which formed the dimer **39a** in 79% yield and 94% *ee* (*R*) (equation 18).³² Kinetic studies were consistent with rate-limiting ketene formation, with the stereoselectivity determined by complexation of the ketene with the chiral base forming **38**. Other ketenes **36** reacted similarly. Chiral ketene dimers **39** were shown to be valuable intermediates in a variety of synthetic procedures.³²



Ketene (40) generated by dehydrobromination of acetyl bromide in the presence of 10 mol % catalyst 41 reacted with acetaldehyde in situ and gave (*R*)-3-methylpropiolactone (42) in 92% yield, >98% *ee* (equation 19).³³ This intermediate was utilized in a total synthesis of the pharmacologically active natural product (–)-laulimalide.

$$CH_{3}C''_{Br} \xrightarrow{i \cdot Pr_{2}NEt}_{41, -78 \circ C} \xrightarrow{H}_{H} = C = O \xrightarrow{Me}_{Me} \xrightarrow{O}_{O} \xrightarrow{i \cdot Pr \cdot M}_{Tf} \xrightarrow{Bn}_{N} \xrightarrow{Pr \cdot i}_{Me} (19)$$

The reaction of the acyl chlorides derived from a series of acrylic acids with Et₃N in CH₂Cl₂ at -20 °C gave ketenes CH₂=C(CH₃)CH=C=O (**43**), CH₃CH=CHCH=C=O (**44**), EtCH=CHCH=C=O (**45**), CH₂=CHCH=CMe=C=O (**46**), CH₃CH=CHCH=CMe=C=O (**47**), CH₂=CHCBr=C=O (**48**), CH₂=CMeCBr=C=O (**49**), and **50**, which upon addition of benzyl alcohol led to isolation of benzyl esters with a predominance of the  $\beta$ , $\gamma$ -unsaturated isomers (equation 20).³⁴ A similar reaction is shown in Section 4.1.2. Reaction of **51** with Et₃N in the presence of alcohols led to significant amounts of the ketene **52**, as evidenced by the formation of deconjugated esters **53** (equation 21).³⁵



Photolyses of acetyl chloride,^{36–38} acetyl bromide,³⁹ propionyl chloride,^{40,41} and butanoyl chloride⁴² in solid argon gave absorptions assigned to molecular complexes of the ketene with HCl or HBr. The IR spectra of CH₂=C=O (**40**) in these complexes were practically unchanged from that of the free ketene, but the HCl and HBr stretching frequencies were significantly lower than those of the monomeric species in solid argon (see Section 2.3).^{36–38} The propionyl chloride dissociation follows a concerted electrocyclic pathway, giving a T-shaped ketene-HCl complex.⁴⁰ The known reaction of CH₃C(•)O and Cl• radicals to give ketene (**40**) was examined theoretically and proposed to involve intermediate formation of acetyl chloride, which formed CH₂=C=O and HCl (equation 22).⁴¹

$$CH_3 - \overset{0}{C'} \xrightarrow{CI} CH_3 \overset{0}{\swarrow} \xrightarrow{-HCI} \overset{H}{\underset{H}{\longrightarrow}} C = 0$$
 (22)

Photolysis of cyclopropanecarbonyl chloride (54) in an argon matrix at 15 K gave cyclopropylideneketene (55) and HCl, and it was proposed that upon further photolysis, decarbonylation to the complex of the carbone 56 and HCl was formed

(equation 23).^{42,43} Photolysis of matrix-isolated acrylyl chloride gave ketene **57** by a 1,3-chlorine migration (equation 24).⁴⁴

$$\bigvee_{H}^{COCl} \xrightarrow{hv}_{\lambda \ge 327 \text{ nm}} \bigvee_{\Delta \ge 327 \text{ nm}} \bigvee_{C=0} + HCl \xrightarrow{hv}_{\lambda \ge 327 \text{ nm}} \bigvee_{C:} + HCl + CO$$
(23)  
54 55 56  

$$\bigvee_{CH_2}^{Cl} \to 0 \xrightarrow{hv}_{\lambda \ge 310 \text{ nm}} \xrightarrow{CICH_2}_{H} C=0$$
(24)  
57

Thermolysis of malonyl dichloride at 50 °C gave chloroformylketene (**58**), for which two conformers were observed by IR in rare gas matrices (equation 25).⁴⁵ Further photolysis of **58** gave the complex **59** of  $C_3O_2$  with HCl, and the barriers for the reaction in the ground state and the first excited state were calculated (equation 25).⁴⁵ The gas phase IR laser photolysis of CH₃COCl with SbF₆ leads to ketene formation, but in the presence of W(CO)₆ radical abstraction of Cl leading to CH₃C(•)=O occurs.⁴⁶

An unusual formation of ketene **61** from AlCl₃-catalyzed rearrangement of cyclopropyl acyl chloride **60** leading to formation of naphthol **62** has been proposed (equations 26, 27).⁴⁷ Inversion of configuration occurs upon treatment of the acid **63** with oxalyl chloride and the indole **64**. This was attributed to intervention of the ketene **65** or ketene iminium ion **66**, which formed the observed amide **67** (equation 28).⁴⁸ These reactions, possibly involving chloride loss and acylium ion formation, share some mechanistic characteristics with dehydrochlorination.





The addition of ammonia to 2-sulfonyl substituted propenoyl chloride **68** led to  $\beta$ -lactama, and was proposed to involve formation of ketene **69** by an additionelimination sequence, although this were not detected directly (equation 30).⁴⁹ Stereoselectivity in the amine addition and formation of the resultant  $\beta$ -lactam was achieved with a chiral Lewis acid catalyst.⁵⁰



Thermolysis of 1-acetylbenzotriazole **70** with carbodimides **72** at 210 °C formed quinoline derivatives **73** by a process proposed to occur with formation of ethylketene (**71**) by an elimination process followed by reaction forming **73**, which reacted further with an isocyanate derived from **72** (equations 31, 32).⁵¹





Reaction of arylacetyl benzotriazoles **75** with NaH was proposed to form arylketenes **76**, which gave dimers **77** that yielded 1,3-diarylacetones **78** upon hydrolysis and decarboxylation (equation 33).⁵²



#### **REFERENCES FOR SECTION 3.2.2**

- 1. Staudinger, H. Chem. Ber. 1911, 44, 1619-1623.
- 2. Wedekind, E. Chem. Ber. 1901, 34, 2070-2077.
- Taylor, E. C.; McKillop, A.; Hawks, G. H. In *Organic Syntheses*; Noland, W. E., Ed.; Wiley, New York, 1988, Coll. Vol. VI, pp. 549–551.
- 4. Olah, G. A.; Wu, A.-H.; Farooq, O. Synthesis 1989, 566-567.
- 5. Hodous, B. L.; Ruble, J. C.; Fu, G. C. J. Am. Chem. Soc. 1999, 121, 2637-2638.
- 6. Kirmse, W.; Spaleck, W. Angew. Chem. Int. Ed. Engl. 1981, 20, 776-777.
- 6a. Back, T. G.; Barton, D. H. R.; Britten-Kelly, M. R.; Guziec, F. S., Jr. J. Chem. Soc., Perkin 1 1976, 2079–2089.
  - Kostyuk, A. S.; Boyadzhan, Zh. G.; Zaitseva, G. S.; Sergeev, V. N.; Savel'eva, N. I.; Baukov, Yu. I.; Lutsenko, I. F. *Zh. Obshch. Khim.* **1979**, *49*, 1543–1542, *Engl. Transl.* 1346–1353.
- Allen, B. M.; Hegarty, A. F.; O'Neill, P.; Nguyen, M. T. J. Chem. Soc., Perkin Trans. 2 1992, 927–934.
- 9. Frey, J.; Rappoport, Z. J. Am. Chem. Soc. 1996, 118, 3994-3995.
- 10. Frey, J.; Rappoport, Z. J. Am. Chem. Soc. 1996, 118, 5182-5191.
- 11. O'Neill, P.; Hegarty, A. F. J. Org. Chem. 1992, 57, 4421-4426.
- 12. Pracejus, H.; Wallura, G. J. Prakt. Chem. 1963, 19, 33-36.
- 13. Kresze, G.; Runge, W.; Ruch, E. Liebigs Ann. 1972, 756, 112-127.
- 14. Hodous, B. L.; Fu, G. C. J. Am. Chem. Soc. 2002, 124, 10006-10007.
- 15. Brady, W. T.; Lloyd, R. M. J. Org. Chem. 1980, 45, 2336-2339.
- 16. Brady, W. T.; Schreubel, G. A. J. Am. Chem. Soc. 1973, 95, 7447-7449.
- 17. Abe, N.; Osaki, I.; Kojima, S.; Matsuda, H.; Sugihara, Y.; Kakehi, A. J. Chem. Soc., Perkin Trans. 1996, 1, 2351–2356.
- 18. Truce, W. E.; Bailey, P. S., Jr. J. Org. Chem. 1969, 34, 1341-1345.
- 19. Cocivera, M.; Effio, A. J. Org. Chem. 1980, 45, 415-420.
- Lynch, J. E.; Riseman, S. M.; Laswell, W. L.; Tschaen, D. M.; Volante, R. P.; Smith, G. B.; Shinkai, I. J. Org. Chem. 1989, 54, 3792–3796.
- 21. de Faria, A. R.; Salvador, E. L.; Correia, C. R. D. J. Org. Chem. 2002, 67, 3651-3661.

- 22. Marko, I.; Ronsmans, B.; Hesbain-Frisque, A.; Dumas, S.; Ghosez, L.; Ernst, B.; Greuter, H. J. Am. Chem. Soc. 1985, 107, 2192–2194.
- 23. Corey, E. J.; Rao, K. S. Tetrahedron Lett. 1991, 32, 4623-4626.
- 24. Corey, E. J.; Desai, M. C.; Engler, T. A. J. Am. Chem. Soc. 1985, 107, 4339-4341.
- 25. Snider, B. B.; Kulkarni, Y. S. J. Org. Chem. 1987, 52, 307-310.
- 26. Corey, E. J.; Desai, M. C. Tetrahedron Lett. 1985, 26, 3535-3538.
- 27. Camps, P.; Munoz-Torrero, D.; Sánchez, L. *Tetrahedron Asymm.* 2004, 15, 311–321.
- France, S.; Wack, H.; Taggi, A. E.; Hafez, A. M.; Wagerle, T. R.; Shah, M. H.; Dusich, C. L.; Lectka, T. J. Am. Chem. Soc. 2004, 126, 4245–4255.
- 29. Hafez, A. M.; Taggi, A. E.; Dudding, T.; Lectka, T. J. Am. Chem. Soc. 2001, 123, 10853–10859.
- France, S.; Weatherwax, A.; Taggi, A. E.; Leckta, T. Acc. Chem. Res. 2004, 37, 592–600.
- Taggi, A. E.; Hafez, A. M.; Wack, H.; Young, B.; Ferraris, D.; Lectka, T. J. Am. Chem. Soc. 2002, 124, 6626–6635.
- 31a. France, S.; Shah, M. H.; Weatherwax, A.; Wack, H.; Roth, J. P.; Leckta, T. J. Am. Chem. Soc. 2005, 127, 1206–1215.
- 32. Calter, M. A.; Orr, R. K.; Song, W. Org. Lett. 2003, 5, 4745-4748.
- 33. Nelson, S. G.; Cheung, W. S.; Kassick, A. J.; Hilfiker, M. A. J. Am. Chem. Soc. 2002, 124, 13654–13655.
- 34. Cardillo, G.; De Simone, A.; Mingardi, A.; Tomasini, C. Synlett **1995**, 1131–1132.
- 35. Hoye, T. R.; Magee, A. S.; Trumper, W. S. Synth. Commun. 1982, 12, 183-187.
- 36. Kogure, N.; Ono, T.; Suzuki, E.; Watari, F. J. Mol. Struct. 1993, 296, 1-4.
- Pettersen, M. V.; Schriver-Mazzuoli, L.; Shriver, A.; Chaquin, P.; Lasson, E. Chem. Phys. 1996, 204, 115–127.
- Winter, P. R.; Rowland, B.; Hess, W. P.; Radziszewski, J. G.; Nimlos, M. R.; Ellison, G. B. J. Phys. Chem. A **1998**, 102, 3238–3248.
- Kogure, N.; Hatakeyama, R.; Suzuki, E.; Watari, F. J. Mol. Struct. 1993, 299, 105– 109.
- 40. Rowland, B.; Winter, P. R.; Ellison, G. B.; Radziszewski, J. E.; Hess, W. P. J. Phys. Chem. A **1999**, 103, 965–970.
- 41. Sumathi, R.; Chandra, A. K. Chem. Phys. Lett. 1997, 271, 287.
- 42. Sumathi, R.; Nguyen, M. T. J. Phys. Chem. A 1998, 102, 8150-8156.
- Monnier, M.; Allouche, A.; Verlaque, P.; Aycard, J.-P. J. Phys. Chem. 1995, 99, 5977– 5985.
- 44. Piétri, N.; Monnier, M.; Aycard, J.-P. J. Org. Chem. 1998, 63, 2462-2468.
- Piétri, N.; Chiavassa, T.; Allouche, A.; Rajzmann, M.; Aycard, J.-P. J. Phys. Chem. 1996, 100, 7034–7041.
- 46. Allen, G. R.; Renner, N. D.; Russell, D. K. Chem. Commun. 1998, 703-704.
- 47. Nishii, Y.; Tanabe, Y. J. Chem. Soc. Perkin 1 1997, 477-486.
- 48. Grieco, P. A.; Kaufman, M. D. J. Org. Chem. 1999, 64, 7586-7593.

- 49. Zhou, F.; Rosen, J.; Zebrowski-Young, J. M.; Freihammer, P. M; Detty, M. R.; Lachicotte, R. J. J. Org. Chem. **1998**, 63, 5403–5412.
- 50. Zhou, F.; Detty, M. R.; Lachicotte, R. J. Tetrahedron Lett. 1999, 40, 585-588.
- 51. Katritzky, A. R.; Huang, T.-B.; Voronkov, M. V.; Steel, P. J. J. Org. Chem. 2000, 65, 8069–8973.
- 52. Katritzky, A. R.; Soleiman, M.; Yang, B. Heteroatom. Chem. 1996, 7, 365-367.

## 3.2.3 Ketenes from Esters

**3.2.3.1** Ketenes from Ester Enolates Ester hydrolysis by the E1cB mechanism is a well-documented process in which a base removes a proton adjacent to the carbonyl group in an ester 1, forming ester enolate 2, and good alkoxy or aryloxy leaving groups depart, forming ketene intermediates 3, which in water form acids 4 (equation 1).^{1–5} This reaction has been studied computationally for ethyl acetate with a comparison of the free energy of reaction in water and in DMSO; the calculated  $\Delta G^*_{sol}$  value was only 6.0 and 8.4 kcal/mol greater for the E1cB mechanism (equation 1), respectively, compared to the B_{AC}2 mechanism of base attack at the carbonyl carbon.⁶ It was expected that in crowded esters where access to the carbonyl group was hindered, the E1cB process could become dominant.⁶



There is good evidence for this mechanism in the hydrolysis of aryl ethyl malonates **5**,¹ *o*-nitrophenyl cyanoacetate (**6**),² and aryl acetoacetates **7**, which form carboethoxy-, cyano-, and acetylketenes, respectively.³ Evidence for this mechanism includes linear correlations of log*k* for hydrolysis of **7** with the *pK_a* of the phenolate leaving group,³ an enhancement of the reaction by bulky substituents on the aryl leaving group,^{7,8} and positive activation volumes, characteristic of dissociative processes.^{9,10}



The reactions of aryl arylacetates **8** in CH₃CN with amines R₂NH were interpreted as proceeding through the formation of arylketenes **9** by the one-step E2 mechanism (equation 2).^{11,12} For 4-nitrophenyl 4-nitrophenylacetate promoted by R₂NH/R₂NH₂⁺ buffers in 70% CH₃CN/H₂O, the reaction was interpreted as involving concurrent E2 and E1cb mechanisms. Reactions of Ph₂CHCO₂Ar¹ were

interpreted as involving similar processes.^{13,14}



Reactions of aryl 4-nitrophenylacetates **8** (Ar = 4-O₂NC₆H₄) with OH⁻ in 80% DMSO/H₂O proceeded through formation of carbanion intermediates **11**, which were observed by UV and which formed 4-nitrophenylketene **12**, as evidenced by the correlation of the rate constants for carbanion decay with the  $\sigma^-$  constants of the aryl substituents with  $\rho = 4.4$  (equation 3).¹⁵ The UV absorption of **12** was observed with  $\lambda_{max} = 495$  nm, and this was used to monitor the kinetics of hydration of **12** in this solution.¹⁵ Similar mechanisms have been proposed for hydrolysis of **8** in 20% H₂O/CH₃CN.¹⁶



Conjugate addition of  $OH^-$  to arylacrylates **13** generated carbanion intermediates **14**, leading to E1cB ester formation of ketenes **15** (equation 4).¹⁷



Basic hydrolysis of aryl 4-hydroxybenzoate esters **16** occurred by deprotonation of the phenolic hydroxyl, forming the oxoquinone methide **17** (equation 5).^{18,19} Several homologues of this system were studied,^{20–22} and the doubly vinylogous dinitrobenzoate esters **17a** and **17b** were proposed to react in basic solution, forming oxoquinone methides **17c** and **17d** (equation 5a).^{23,24}





Aryl fluorene-9-carboxylates **18** with electron-withdrawing aryl substituents (R = 3- or 4-NO₂, 4-CN, 4-Ac, 3- or 4-Cl) hydrolyzed by the E1cB mechanism, but for poorer leaving groups reacted by hydroxide attack at carbonyl carbon (equation 6).²⁵ The effects of micelles on the reactivity of **18** were also examined.²⁶



Crowded 2,6-di-*tert*-butyl-4-tolyl (BHT) esters of carboxylic acids (**20**) were deprotonated by *n*-BuLi forming ester enolates which expelled the aryloxide anion, forming ketenes **21**, which were captured by excess *n*-BuLi, forming ketone enolates **22** (equation 7).^{27,28} The product enolates were captured by aldehydes to form aldols or by Me₃SiCl to give silyl enol ethers.^{27,28}



The particularly stable bis(trimethylsilyl)ketene **24** was isolated from reaction of the ester **23** with LDA in a 60% yield (equation 8).²⁹



The ketene **26** formed from the BHT ester **25** by conjugate addition (*cf.* equation 4) was used to form the allene **27** by a ketene Wittig reaction (equation 9).³⁰ Reaction of BHT arylacetates **28** with *n*-BuLi formed enolates that gave the ketenes **29**, and upon addition of trimethyl phosphonoacetate ion in the presence of  $\text{Sn}^{2+}$  or

 $Zn^{2+}$  these gave allenes **30** (equation 10).³¹ Ketenes generated from BHT esters were also trapped with stannyllithium reagents.³²



The formation of phenylketene PhCH=C=O by the dissociation of the enolate of PhCH₂CO₂Pr-*i* is reported in Section 4.1.4, including measurement of the ¹³C NMR spectrum, but this report requires confirmation.

Asymmetric Michael addition to chiral binaphthyl cinnamate esters **31** gave ketenes **32**, which were trapped by Me₂CuLi to give optically active ketones **33** with high selectivity (equation 11).^{33,34} Phenyl esters PhCRCO₂Ph reacted with LDA and ZnCl₂ to give ketenes that were enantioselectively converted to allenes by a chiral phosphonate.³⁵



Additions of organolithiums to 4-methoxy-2,6-di-*tert*-butyl esters of naphthoic acids in the presence of chiral ligands resulted in asymmetric addition to the ring with ketene formation.³⁶ Reduction gave dihydronaphthalenes.³⁷

Enolates of methoxymethyl esters **34** were proposed to fragment to ketenes **35**, formaldehyde, and methoxide, which recombined to give the aldol product **36** (equation 12).³⁸ With an  $\beta$ -bromo ester **37** the ketene **38** formed the glycidic ester **39** (equation 13).³⁸



In an example of the three-phase test for reaction intermediates, the polymerbound ester 40 with  $Et_3N$  in dioxane formed ketene 41, which gave acylation of a polymer-bound amine (equation 14).³⁹



Birch reduction of the benzamide 43 was suggested to proceed through the amide enolate 44, which formed the ketenyl anion 45, leading to benzaldehyde 46 (equation 15).⁴⁰



Reaction of cyclohexanone enamines **46a** with  $\alpha$ , $\beta$ -unsaturated thio esters and seleno esters leading to bicyclo[3.3.1]nonane-2,9-diones **46d** may have involved [3,3]-sigmatropic rearrangements of intermediate *N*-acylated enamines **46b** forming ketenes **46c**, which cyclized further and were hydrolyzed to the final products (equation 16).⁴¹ This was analogous to the reactions of acyl chlorides previously

observed.42,43



A claim of formation of ketenes **47** as discrete intermediates by elimination from the corresponding esters (equation 16a)⁴⁴ requires independent confirmation, as these ketenes have eluded other efforts at preparation in solution and by every indication would be quite reactive.

$$R \xrightarrow{O}_{OCH_3} \xrightarrow{P_4O_{10}} \xrightarrow{R}_{H} C = O$$

$$H$$

$$47 \quad R = NO_2, R = CN (3b)$$

$$(16a)$$

Treatment of aminonicotinic esters with base was proposed to give acylketenes that cyclized to 1,8-naphthopyridin-2-ones.⁴⁵ Acetoacetate esters of thiols, including *S*-acetoacetyl coenzyme A, were found to hydrolyze by the E1cB mechanism, implicating the formation of ketene intermediates.^{46,46a}

**3.2.3.2** *Ketenes by Ester Pyrolysis* Pyrolysis of isopropenyl esters was used to generate ketenes with long hydrocarbon chains through a mechanism involving the ester enol, as in the pyrolysis through the pericyclic transition state **48** forming *n*-hexadecylketene **49**, which was obtained as the dimer **50** (equation 17).⁴⁷ Isopropenyl acetoacetate (**51**) formed acetylketene (**52**) by a pericyclic process and was captured by cyclohexanone as the dioxinone **53** or the dimer **54** (equation 18).⁴⁸





Pyrolysis of alkyl acetocetates **55** formed ketene **52** and the alcohol, and evidently proceeded through a pericyclic mechanism involving the ester enol **56** (equation 19).^{49,50} Evidence for the pericyclic mechanisms comes from kinetic results in Table 3.1, which show significant accelerations in reactivity for the isopropenyl and *tert*-butyl esters.^{49,50} The acceleration for the *tert*-butyl ester evidently arose from a steric effect favoring the dissociative mechanism.



Pyrolysis of the trimethylsilyl ether of dimethyl malonate was also proposed to react through a pericyclic transition state, and formed carbomethoxyketene **57** (equation 20).⁵¹ The acylketene **58** was proposed to be generated by thermolysis of the ester in alcohols and captured as the product of ester interchange (equation 21).^{52,53}



TABLE 3.1 Relative Rates of Thermolysis of Acetoacetates  $CH_3COCH_2CO_2R$  in *n*-BuOH at 91.7 °C⁵⁰

R	k _{rel}	R	k _{rel}	R	k _{rel}
Me Et	1.0 1.1	<i>i</i> -Pr <i>t</i> -Bu	1.4 17	Isopropenyl ⁴⁹	7.3



The titanium isoproposide complex of  $CH_3COCH_2CO_2Bu$ -*n* underwent thermal decomposition to form acetylketene with deposition of TiO₂ as a thin film.⁵⁴

Plasma dissociation of vinyl and saturated esters was found to give ketenes as detected by FTIR in the vapor, with bands near  $2130 \text{ cm}^{-1.55}$  Methyl crotonate (**59**) reacted by a pericyclic mechanism, forming ethylketene **60** (equation 22), while methyl isobutyrate (**61**) reacted by a radical mechanism, forming methylketene **63** (equation 23) and ketene **64** (equation 24).⁵⁵



Heating of phenyl acetate in the gas phase over molecular halide clusters of Nb, Mo, Ta, and W at 250  $^{\circ}$ C led to the formation of phenol and ketene (equation 25).⁵⁶

$$CH_{3} \xrightarrow{O} \xrightarrow{O} \xrightarrow{250 \circ C} \xrightarrow{H} C=O$$

$$(25)$$

$$64$$

Gas phase reaction of phosphonium ions  $(MeO)_2PO^+$  with aliphatic esters formed ketene (64).⁵⁷ This reaction was modeled by B3LYP/6-31G(d,p) calculations for methyl acetate–forming complex 65, which formed the ketene complex

**66** (equation 26).⁵⁷



Pyrolysis of ester **67** gave ketene **68**, which cyclized to 3-azabicyclo[3.2.0] heptenones **69** (equation 27).⁵⁸



Esterolysis melt polymerization using  $CH_3CO_2ArCO_2CH_3$  suffers from color problems due to the formation of ketene. These problems can be avoided by using  $PhCO_2ArCO_2CH_3$  for the preparation of polyesters.⁵⁹

#### **REFERENCES FOR SECTIONS 3.2.3.1 AND 3.2.3.2**

- 1. Holmquist, B.; Bruice, T. C. J. Am. Chem. Soc. 1969, 91, 2993-3002.
- 2. Holmquist, B.; Bruice, T. C. J. Am. Chem. Soc. 1969, 91, 3003-3009.
- 3. Pratt, R. F.; Bruice, T. C. J. Am. Chem. Soc. 1970, 92, 5956-5964.
- 4. Williams, A.; Douglas, K. T. Chem. Rev. 1975, 75, 627-649.
- Satchell, D. P. N.; Satchell, R. S. In Chemistry of Functional Groups: The Chemistry of Acid Derivatives. Supplement B; Patai, S., Ed; Wiley, New York, 1992, Vol. 2, pp. 747– 802.
- 6. Pliego, J. R., Jr.; Riveros, J. M. J. Phys. Chem. A 2004, 108, 2520-2526.
- 7. Inoue, M.; Bruice, T. C. J. Org. Chem. 1983, 48, 3559-3561.
- 8. Inoue, M.; Bruice, T. C. J. Org. Chem. 1986, 51, 959-963.
- 9. Isaacs, N. S.; Najem, T. S. Can. J. Chem. 1986, 64, 1140-1144.
- 10. Isaacs, N. S.; Najem, T. S. J. Chem. Soc. Perkin Trans. 2 1988, 557-562.
- 11. Cho, B. R.; Kim, Y. K.; Yoon, C.-O. M. J. Am. Chem. Soc. 1997, 119, 691-697.
- 12. Cho, B. R.; Kim, N. S.; Kim, Y. K.; Son, K. H. J. Chem. Soc. Perkin Trans. 2, 2000, 1419–1423.
- Cho, B. R.; Kim, Y. K.; Seung, Y. J.; Kim, J. C.; Pyun, S. Y. J. Org. Chem. 2000, 65, 1239–1242.

- Cho, B. R.; Jeong, H. C.; Seung, Y. J.; Pyun, S. Y. J. Org. Chem. 2002, 67, 5232– 5238.
- 15. Broxton, T. J.; Duddy, N. W. J. Org. Chem. 1981, 46, 1186-1191.
- Cheong, D. Y.; Yoh, S. D.; Choi, J. H.; Shim, K. T. J. Korean Chem Soc. 1992, 36, 446– 452.
- 17. Inoue, M.; Bruice, T. C. J. Am. Chem. Soc. 1982, 104, 1644-1653.
- Cevasco, G.; Guanti, G.; Hopkins, A. R.; Thea, S.; Williams, A. J. Org. Chem. 1985, 50, 479–484.
- 19. Cevasco, G.; Thea, S. Gazz. Chim. Ital. 1992, 122, 199-200.
- 20. Cevasco, G.; Thea, S. J. Org. Chem. 1995, 60, 70-73.
- 21. Cevasco, G.; Thea, S. J. Org. Chem. 1999, 64, 5422-5426.
- 22. Cevasco, G.; Vigo, D.; Thea, S. Org. Lett. 1999, 1, 1165-1167.
- 23. Cevasco, G.; Vigo, D.; Thea, S. J. Org. Chem. 2000, 65, 7833-7838.
- 24. Cevasco, G.; Vigo, D.; Thea, S. J. Org. Chem. 2001, 66, 7685-7690.
- 25. Alborz, M.; Douglas, K. T. J. Chem. Soc., Perkin Trans. 2, 1982, 331-339.
- 26. Correia, V. R.; Cuccovia, I. M.; Stelmo, M.; Chaimovich, H. J. Am. Chem. Soc. 1992, 114, 2144–2146.
- 27. Häner, R.; Laube, T.; Seebach, D. J. Am. Chem. Soc. 1985, 107, 5396-5403.
- 28. Seebach, D.; Amstutz, R.; Laube, T.; Schweiser, W. B.; Dunitz, J. D. J. Am. Chem. Soc. **1985**, *107*, 5403–5409.
- 29. Sullivan, D. F.; Woodbury, R. P.; Rathke, M. W. J. Org. Chem. 1977, 42, 2038-2039.
- 30. Tanaka, K.; Otsubo, K.; Fuji, K. Synlett 1995, 933-934.
- 31. Tanaka, K.; Otsubo, K.; Fuji, K. Tetrahedron Lett. 1995, 36, 9513-9514.
- 32. Barbero, A.; Pulido, F. J. Synlett 2001, 827-829.
- Tanaka, F.; Node, M.; Tanaka, K.; Mizuchi, M.; Hosoi, S.; Nakayama, M.; Taga, T.; Fuji, K. J. Am. Chem. Soc. 1995, 117, 12159–12171.
- Fuji, K.; Yang, X.-S.; Tanaka, K.; Asakawa, N.; Hao, X.-J. *Tetrahedron Lett.* 1996, 37, 7373–7376.
- 35. Yamazaki, J.; Watanabe, T.; Tanaka, K. Tetrahedron Asym. 2001, 12, 669-675.
- 36. Tomioka, K.; Shindo, M.; Koga, K. J. Org. Chem. 1990, 55, 2276-2277.
- 37. Tomioka, K.; Shindo, M.; Koga, K. Tetrahedron Lett. 1993, 34, 681-682.
- 38. Schultz, A. G.; Berger, M. H. J. Org. Chem. 1976, 41, 585-586.
- 39. Rebek, J.; Brown, D.; Zimmerman, S. J. Am. Chem. Soc. 1975, 97, 454-455.
- 40. Schultz, A. G. Chem Commun. 1999, 1263-1271.
- 41. Byeon, C.-H.; Hart, D. J.; Lai, C.-S.; Unch, J. Synlett 2000, 119-121.
- 42. Hickmott, P. W.; Ahmed, M. G.; Ahmed, S. A.; Wood, S.; Kapon, M. J. Chem. Soc. Perkin Trans. 1 1985, 2559–2571.
- 43. Hickmott, P. W. S. Afr. J. Chem. 1989, 42, 17-19.
- 44. Zavlin, P. M.; Efremov, D. A.; Essentseva, N. S.; Tebby, J. C. *Russ. J. Gen. Chem.* **1995**, 65, 529–532.
- Zografos, A. L.; Mitsos, C. A.; Igglessi-Markopoulou, O. J. Org. Chem. 2001, 66, 4413– 4415.
- 46. Yaggi, N. F.; Douglas, K. T. J. Am. Chem. Soc. 1977, 99, 4844-4845.

- 46a. Douglas, K. T.; Alborz, M.; Rullo, G. R.; Yaggi, N. F. J. Chem. Soc. Chem. Commun. 1982, 245–246.
- 47. Rothman, E. S. J. Am. Oil Chem. Soc. 1968, 45, 189-193.
- 48. Clemens, R. J.; Witzeman, J. S. J. Am. Chem. Soc. 1989, 111, 2186-2193.
- 49. Witzeman, J. S. Tetrahedron Lett. 1990, 31, 1401–1404.
- 50. Witzeman, J. S.; Nottingham, W. D. J. Org. Chem. 1991, 56, 1723-1728.
- 51. Jullien, J.; Pechine, J. M.; Perez, F. Tetrahedron Lett. 1983, 24, 5525-5526.
- 52. Emerson, D. W.; Titus, R. L.; Gonzáles, R. M. J. Org. Chem. 1990, 55, 3572-3576.
- Emerson, D. W.; Titus, R. L.; Gonzáles, R. M. J. Org. Chem. 1991, 56, 5301– 5307.
- 54. Bhakta, R.; Thomas, R.; Hipler, F.; Bettinger, H. F.; Müller, J.; Ehrhart, P.; Devi, A. J. *Mater. Chem.* **2004**, *14*, 3231–3238.
- 55. Pan, Y. V.; Denton, D. D. J. Appl. Poly. Sci. 1999, 73, 17-27.
- 56. Chihara, T.; Kamiguchi, S. Chem Lett. 2002, 70-71.
- 57. Leclerc, E.; Taphanel, M.-H.; Morizur, J.-P. J. Mass Spectrom. 2002, 37, 903-909.
- 58. Arya, F.; Bouquant, J.; Chuche, J. Tetrahedron Lett. 1986, 27, 1913–1914.
- 59. Choi, W.-S.; Padias, A. B.; Hall, H. K., Jr. J. Poly. Sci. A 2000, 38, 3586-3595.

#### 3.2.4 Ketenes by Dehalogenation of α-Halo Carboxylic Acid Derivatives

The initial preparation of diphenylketene (1) by Staudinger in 1905 involved the reduction of  $\alpha$ -bromodiphenylacetyl bromide (2) with zinc (equation 1).^{1,2} This procedure has been extensively utilized ever since, and has also been used for the preparation of 1,1- and 2,2-dinaphthylketenes.³

Improved yields for the preparation of methyl- and ethylketene were obtained by carrying out the zinc dehalogenation in THF with codistillation of the ketene and solvent under reduced pressure.⁴ Removal of the ketene solution from the precipitated zinc salts has also been achieved by using syringe transfer, as in the preparation of a THF solution of ketene **4** from **3** (equation 2).⁵

$$COBr \xrightarrow{Zn} COBr \xrightarrow{THF} C=0$$

$$3 \qquad 4 (IR 2100 \text{ cm}^{-1})$$

$$(2)$$

In the formation of chloroketenes by dehalogenation for in situ [2 + 2] cycloadditions, complexation of the zinc salts with POCl₃^{6,7} or dimethoxyethane⁸ prior to workup has been recommended. This preparation of the highly reactive dichloroketene (**6**) for in situ trapping by the dehalogenation of trichloroacetyl chloride (**5**) has been described in *Organic Syntheses*.^{7,8} Ultrasound served to activate zinc for formation of dichloroketene.⁹ This has been utilized in cycloadditions with alkynes to give cyclobutenones **4**, which may be hydrolyzed to cyclobutenediones (equation 3).⁹

Various procedures for zinc activation have been utilized,  $^{10-12}$  and activation using ultrasound  13  or simple heating  14  was found to be simple and quite effective.



The metal ions  $Mn(CO)_5^-$  and  $Cr(CO)_4NO^-$  have proved to be very useful in the generation of small quantities of reactive ketenes at low temperatures and have permitted the measurement of ¹H NMR spectra of sensitive ketenes such as **8** (equation 4).^{15–17} The use of Zn-Ag couple with the  $\alpha$ -bromoacyl chloride **9** was utilized for the preparation of cyclopropylideneketene **10**, which formed the dimer **11** (equation 5).^{18–20} Analogues of **10** were prepared by the same procedure.^{18–20}



An alternative procedure for debromination of  $\alpha$ -bromodiphenylacetyl bromide (2) was treatment with triphenylphosphine, which gave diphenylketene (1, equation 6).²¹ Chloroketene (12) was obtained by the reaction of dichloroacetyl chloride with Me₃SiPPh₂ at low temperature, forming the acylphosphine, followed by heating and trapping of the ketene at -80 °C (equation 7).²² Reaction of 13 with Ph₃P gave dibromoketene (14), which was trapped with cyclopentadiene by [2 + 2]

cycloaddition, forming **15** (equation 8).²³

$$\begin{array}{cccc} Ph & O & PPh_{3} & Ph \\ Br & & & \\ Ph & Br & & \\ 20.5-22.5 \,^{\circ}C & Ph \\ \end{array} \begin{array}{c} Ph & \\ Ph & \\ Ph & \\ Ph & \\ 1 & \\ \end{array} \begin{array}{c} Ph & \\ Ph &$$

Formation of methylketene (17) by debromination of  $\alpha$ -bromopropionyl bromide (16) in THF, with distillation of the product at room temperature and 110 Torr as it formed and treatment of the solution with quinidine at -78 °C, gave stereoselective formation of the dimer 18 as a solution in THF (equation 9).^{24,25} The dimer was a useful synthem for further synthetic transformations.

Dimethylketene (20) was prepared by debromination of 2-bromoisobutyryl bromide (19) with zinc in ethyl acetate and collected by distillation with heating at 300 torr as a solution in ethyl acetate (equation 9).²⁶

DI

Electrochemical reduction of  $\alpha$ -bromodiphenylacetyl bromide (2) formed diphenylketene (1) in almost quantitative yield, and in the presence of elemental sulfur this reaction was utilized in the synthesis of [1,3]oxathiolan-5-one **21**, proposed to result from reaction of **1** with S²⁻ (equation 11).²⁷

Thermolyis of aluminium salts of iodoacetic acid was reported to form ketene (**22**), which was trapped in low yield as the ester (equation 12).²⁸

## **REFERENCES FOR SECTION 3.2.4**

- 1. Staudinger, H. Chem. Ber. 1905, 38, 1735-1739.
- Hanford, W. E.; Sauer, J. C. Org. React.; Wiley, New York, 1946, Coll. Vol. 3, pp. 108– 140.
- 3. Wuest, J. D. Tetrahedron, 1980, 36, 2291-2296.
- 4. McCarney, C. C.; Ward, R. S. J. Chem. Soc., Perkin Trans. 1, 1975, 1600-1603.
- 5. Baigrie, L. M.; Seikaly, H. R.; Tidwell, T. T. J. Am. Chem. Soc. 1985, 107, 5391-5396.
- 6. Hassner, A.; Krepski, L. R. J. Org. Chem. 1978, 43, 2879-2881.
- 7. Depres, J.; Greene, A. E. Org. Synth. 1989 68, 41-48.
- 8. Danheiser, R. L.; Savariar, S.; Cha, D. D. Org. Synth. 1989, 68, 32-40.
- 9. Parker, M. S. A.; Rizzo, C. J. Synth. Commun. 1995, 25, 2781-2789.
- 10. Reike, R. Acc. Chem. Res. 1977, 10, 301-306.
- 11. Erdik, E. Tetrahedron 1987, 43, 2203-2212.
- 12. Tsuda, K.; Ohki, E.; Nozoe, S. J. Org. Chem. 1963, 28, 783-785.
- 13. Mehta, G.; Rao, H. S. P. Synth. Commun. 1985, 15, 991-1000.
- 14. Stenstrøm, Y. Synth. Commun. 1992, 22, 2801-2810.
- 15. Masters, A. P.; Sorensen, T. S.; Ziegler, T. J. Org. Chem. 1986, 51, 3558-3559.
- 16. Masters, A. P.; Sorensen, T. S.; Tran, P. M. J. Org. Chem. 1987, 52, 1449-1502.
- 17. Masters, A. P.; Sorensen, T. S. Tetrahedron Lett. 1989, 5869-5872.
- 18. Wulferding, A.; Wartchow, R.; Hoffmann, H. M. R. Synlett 1992, 22, 476-479.
- Hoffmann, H. M. R.; Geschwinder, P. M.; Hollwege, H.-P.; Walenta, A. *Helv. Chim.* Acta 1988, 71, 1930–1936.
- Hoffmann, H. M. R.; Walenta, A.; Eggert, U.; Schomberg, D. Angew. Chem. Int. Ed. Engl. 1985, 24, 607.
- 21. Darling, S. D.; Kidwell. R. L. J. Org. Chem. 1968, 33, 3974-3975.
- 22. Lindner, E.; Steinwand, M.; Hoehne, S. Angew. Chem. Int. Ed. 1982, 21, 355-356.
- 22a. Lindner, E.; Steinwand, M.; Hoehne, S. Chem. Ber. 1982, 115, 2181-2191.
- 23. Okada, T.; Okawara, R. Tetrahedron Lett. 1971, 2801-2802.
- 24. Calter, M. A. J. Org. Chem. 1996, 61, 8006-8007.
- 25. Calter, M. A.; Guo, X. J. Org. Chem. 1998, 63, 5308-5309.
- 26. Smith, C. W.; Norton, D. G. Org. Synth. Coll. Vol. IV 1963, 348-350.
- 27. Lozano, J. I.; Barba, F. Tetrahedron 1996, 52, 1259-1266.
- Stepovik, L. P.; Kazakina, S. V.; Martynova, I. M. Russ. J. Gen. Chem. 2000, 70, 1371– 1377.

# **3.3 KETENES FROM DIAZO KETONES** (WOLFF REARRANGEMENTS)

Wolff reported the first examples of the rearrangement of diazo ketones in  $1902^1$  but did not formulate the reaction as involving a ketene intermediate until later,² after this had been put forward by Schröter.³ This reaction has proven to be of great value and mechanistic interest and has been frequently reviewed.^{4–13} The reaction of diazo ketones 1 can occur thermally, by metal-catalyzed reaction, or photochemically, and in some cases forms a keto carbene intermediate 2 as a precursor to ketene 3 (equation 1).



The electronic structure of diazo ketones may be represented by the resonance structures **4a–d** shown in Scheme 3.2, in which electron donation from the diazo group to the carbonyl group plays an important role. Structure **4d** is of negligible importance. These structures stabilize diazo ketones compared to other diazoal-kanes, and predict the observed double bond character and restricted rotation in the central C-C bond between the two groups.

The effect of substituents on the structure and energy of diazo compounds and of diazirines has been studied by *ab initio* molecular orbital calculations.^{14,15} These confirm the stabilizing effect of carbonyl groups on the diazo group.

The two conformers of diazoacetone (5) have been observed by ¹H NMR, and there is a barrier for intercoversion of *anti*-5 to *syn*-5 of 15.5 kcal/mol, with a 92:8 preference for *syn*-5 at -40 °C (equation 2).¹⁶ For HCOCHN₂ the *syn/anti* ratio is 69:31.¹⁶ If the coplanarity and conjugation of the  $\pi$ -system is restricted by bulky groups (equation 3)^{17,18} or by twisting of a ring system (equation 4),¹⁹ then the substrates are much more labile and nitrogen loss occurs at a lower temperature. However, if the migrating group cannot occupy a position antiperiplanar to the departing nitrogen, then migration is inhibited, and methyl migration without ketene formation is dominant (equation 3).¹⁷ However, upon photolysis in the gas phase, the yield of the ketene **7** is 40%.¹⁸ The twisted seven-membered ring **9** promoted



Scheme 3.2 Resonance structures for diazo ketones.

thermal Wolff rearrangement even at 20 °C, and the ketene **10** was observed by IR, with a reported frequency of  $2200 \text{ cm}^{-1}$  and was hydrolyzed to the acid in 48% overall yield (equation 4).¹⁹ Similarly, photolysis of **12** formed the ketene **13** in 96% yield, while thermolysis gave the ketene in 85% yield (equation 5).²⁰



The mechanism of the Wolff rearrangement raises subtle questions regarding the intervention of possible reactive intermediates, and these have been pursued through increasingly sophisticated experimental and theoretical means. Possible pathways for the ketene **3** (Scheme 3.3) include equilibration of *syn-***1** with *anti-***1** 



Scheme 3.3 Some possible steps in the Wolff rearrangement.

and initial loss of N₂ from *anti*-1 forming singlet keto carbene **2a**, which may undergo spin inversion to a triplet carbene **2b**, which can reform **2a** or undergo some other reaction such as hydrogen abstraction. The singlet carbene **2a** can undergo Wolff rearrangement to the ketene **3**, which can also be formed in a concerted reaction from *syn*-1. The intervention of an oxirene intermediate **14** has been found to occur in some cases, as demonstrated for R = Ph,  $R^1 = H$  by isotope labeling studies using ²H, ¹³C, and ¹⁸O.²¹ For  $R = CF_3$ ,  $R^1 = H$  migration forming the ketene was demonstrated, whereas for  $R = CF_3$ ,  $R^1 = CO_2Et$  a triplet carbene was formed and no Wolff rearrangement occurred.²²

Besides ketene formation, other possible reactions of the singlet keto carbene 2a are addition of a solvent, such as methanol, and addition to a carbon-carbon double bond forming a cyclopropane. In the case of diazocamphor (15), reversible photochemical rearrangement of the diazo group to the diazirine 16 has been observed, as well as intramolecular insertion by a concerted pathway or from the intermediate carbene 17 forming the tricylanone 18 (equation 6).²³ In ethanol the ketene 19 resulting from ring contraction by Wolff rearrangement formed the *exo*-ester 20 by proton transfer to the least hindered side (equation 7).²³



Singlet carbene **22** is formed by laser flash photolysis of diazoacetaldehyde (**21**) in  $CH_2Cl_2$  and has a lifetime determined as 0.15 to 0.73 ns, leading to ketene (**23**) (equation 8).²⁴ Other computational studies of the Wolff rearrangement are noted in Section 1.1.1.

$$H \xrightarrow{O}_{H} N_{2} \xrightarrow{hv}_{H} H \xrightarrow{O}_{H} H \xrightarrow{H}_{H} C=0 \qquad (8)$$
21 22 23

Photolysis of methyl  $\alpha$ -diazo-(2-naphthyl)acetate (24) forming the carbene 25 and ketene 26 has been studied by three different groups.^{25–27} Included were both solution phase and matrix isolation studies, and rearrangement of the carbene

gave the ketene, with an IR band at 2096 cm⁻¹ (equation 9). Detection of IR bands from both the singlet and triplet states of the spin-equilibrated carbene allowed a direct experimental estimate of the singlet/triplet energy gap in solution at ambient temperature.²⁶ The singlet carbene was preferentially stabilized in more polar solvents.²⁸



The intermediates generated from a series of aryl diazomethyl ketones 27 were compared to those from photolysis of arylhydroxycyclopropenones 28 by trapping with pyridine (equation 10).²⁹ The trapped species were identical, and since ketenes 29 were known to be formed in the latter reaction, it was concluded that these were also formed from 27 within the duration of the laser pulse (20 ns), so that any  $\alpha$ -carbonylcarbene intermediates from 27 must have lifetimes shorter than this.



## 3.3.1 Thermal Wolff Rearrangement

Diphenylketene has been prepared by thermal Wolff rearrangement (**32**, equation 11),³⁰ but with rare exceptions such as **13**,²⁰ thermal reactions are not useful for the preparation and isolation of ketenes from diazo ketones because of the occurrence of reaction between the ketenes and residual diazo ketone.^{31–33} Even fluore-nylideneketene (**33**) has apparently not been made preparatively by thermal Wolff rearrangement. Heating diazoacetophenone (**34**) led to the adduct **36** from combination of unreacted **34** with the derived phenylketene (**35**) (equation 12).³¹ The dimer **37** from oxidation of **36** was also formed.³¹

$$Ph \xrightarrow{0}_{N_2} Ph \xrightarrow{100-110 \circ C}_{benzene} Ph \xrightarrow{0}_{Ph} C=0 \qquad (11)$$

$$31 \qquad 32 (64\%) \qquad 33$$



The kinetics of thermal Wolff rearrangements forming substituted diarylketenes **40** from diazo ketones **38** and **39** (equation 13) have been measured, as shown in Table 3.2.³⁴ The effect of substituents X on the rates of thermal Wolff rearrangement of **38** were correlated by Hammett  $\sigma$  constants with a value of  $\rho$  of 0.75, while for **39** the rates were correlated by  $\sigma^+$  constants with a value of  $\rho^+$  of -1.49.³⁴ The diminished reactivity with electron donor substituents in **38** may have arisen from ground state stabilization due to electron donor substituents in **39** could have been due to weakening of the C-N bond, with negative charge on the N₂ moiety. The combination of substituent effects in the disubstituted diazo ketones **41** forming ketenes **42** produced substantial rate differences (equation 14 Table 3.3).³⁵

X ( <b>38</b> )	$k_{\rm rel}$	X ( <b>39</b> )	$k_{\rm rel}$
MeO	0.73	Me	3.04
Et	0.76	Н	1.0
Me	0.80	F	1.92
Н	1.0	Cl	0.74
F	1.29	Br	0.76
Cl	1.56	$NO_2$	0.071
Br	1.66		

TABLE 3.2 Relative Rates of Decomposition of 38 and 39 in *n*-Butyl Ether at 70  $^{\circ}C^{34}$ 

TABLE 3.3 Relative Rates of Decomposition of 41,³⁵ 43, and 45^{36,37}

X	Y	$k_{\rm rel}(41)^a$	Ar	R	$k_{\rm rel}(43)^b$
NO ₂	MeO	220	4-Tol	Me	26.7
MeO	MeO	40	4-Tol	<i>i</i> -Pr	45.6
Н	Н	1.0	4-Tol	<i>t</i> -Bu	101
$NO_2$	$NO_2$	0.04	Mes ^c	<i>t</i> -Bu	306
MeO	$NO_2$	0.026	45		3.9
Н	NO ₂	0.022			

^{*a*}In 1,2-dichloroethane.

^bIn dioxane, 101 °C.

^cMes is 2,4,6-Me₃C₆H₂.

Increasing steric bulk enhanced reactivity in the aryl alkyl diazo ketones **43**, forming ketenes **44** (equation 15, Table 3.3), as was observed for twisted alkyl systems, while **45**, which formed ketene **46**, was stabilized by efficient conjugation between the diazo group and the carbonyls (equation 16).^{36,37}



Other kinetic studies of Wolff rearrangement showed that phenyl groups were more activating than acyl groups for decomposition.^{38,39}

Microwave-assisted Wolff rearrangement of diazo ketones gave significantly higher yields of ketene-derived products than did thermolysis.⁴⁰ Polar solvents were found to promote ketene formation, as the Z-conformers that lead efficiently to rearrangement have higher dipole moments than the *E*-conformers, so the former are favored in polar solvents. Even cyclic diazo ketones such as diazocamphor (15) that favor insertion forming 18 on thermolysis over formation of ketene 19 captured by benzylamine forming 47 gave much higher yields of capture of Wolff rearrangement products with microwaves than with thermolysis. This was attributed to the diazo ketone grouping adopting a conformation with a higher dipole moment aligned with the electric field of the microwaves, which was more favorable for

Wolff rearrangement (equation 17).⁴⁰



Flash vacuum pyrolysis at 260 to 400 °C of diazo ketone **48** with a tethered cyclohexenyl group formed ketene **49**, which led by [2 + 2] cycloaddition to **50** with a high degree of stereoselectivity (equation 18,19).^{41–43} This method was also utilized for synthesis of optically active  $\beta$ -lactams.⁴¹



Thermal decomposition of phosphonate-substituted diazo ketones **51** produced competition between formation of ketene **52** followed by intramolecular attack by nitrogen on the ketene-forming zwitterion **53** (equation 20), and capture of an initial carbene **54** by nitrogen to form ylide **55** in yields from 0 to 45% (equation 21).^{44,45} Intramolecular hydrogen transfer was proposed to occur for ketene **56**, leading to the intermediate **57**, which formed **58** (equation 22).^{45–47} 2-Diazo- $\beta$ -ketophosphonates **59** with Rh(II) catalysts were proposed to give ketenes **60**, which formed  $\gamma$ -lactones **61** (equation 23).⁴⁸





Isoxazole-substituted ketene **62** generated by Wolff rearrangement cyclized to fused derivatives **63** (equation 24).^{45,47} These are also discussed in Section 4.6.



#### 3.3.2 Catalyzed Wolff Rearrangement

Wolff rearrangement was enhanced by metal catalysts, particularly silver salts, as in the formation of diazo ketone **64** forming 1-naphthylketene (**65**), which is captured by the solvent ethanol, forming **66** (equation 25).⁴⁹ It has been proposed that in the presence of amines, silver nanoclusters are formed from silver benzoate that under these conditions are effective catalysts. This was confirmed by experiments using

preformed silver nanoclusters with diazo ketone 67 in the generation of ketene 68 forming acid **69** (equation 26).⁵⁰



Diazomalonate esters with attached silver or copper ions introduced into the gas phase by electrospray ionization and subjected to collisional activation undergo Wolff rearrangement to unobserved ketene intermediates 70 followed by decarbonylation (equation 27).^{51,52}



Rhodium catalysts with ligands from strong carboxylic acids promoted Wolff rearrangements of 71 to ketene 72 captured by allyl alcohol forming esters 73 in modest yields (equation 28) in competition with intramolecular reaction of 71 with the allyl ether moiety to give 74 and intramolecular C-H insertion to give 75 (equation 29).⁵³ Thus, Rh₂(O₂CCF₃)₄ gave 73, 74, and 75 in a 22:50:28 ratio, where 73 was derived from the ketene 72, while with  $Rh_2(OAc)_4$  the corresponding ratio was 0:60:40.⁵³


Reaction of Rh₂(OAc)₄ with the  $\alpha$ -diazo ketones such as **76** led in some cases to products proposed to arise from initial carbenoid reaction forming **77** followed by vinylogous Wolff rearrangement to ketene **78** captured by methanol as the ester **79** (equation 30).⁵⁴ This was proposed not to involve rearrangement of an intermediate keto carbene, but rather insertion of the diazo carbon into a  $\gamma$ -alkenyl bond forming a bicyclo[2.1.0] pentane that underwent fragmentation to a ketene **78**, which is captured by water in 60% yield (equation 30).⁵⁵ Copper catalysis with CuSO₄, Cu(OTf)₂, or Cu(AcAc)₂ has also been used to promote the vinylogous Wolff rearrangement.⁵⁶ Photolysis of **76** gave the normal Wolff product in 92% yield, with only 8% of the **79**. Other examples of the vinylogous Wolff rearrangement are given in Section 3.4.2. Rhodium-catalyzed Wolff rearrangement of diazoquinolinediones **80** formed ketenes **81**, followed by hydration and decarboxylation to give oxindoles **82** in 40–81% yields (equation 31).⁵⁶ Rhodium-catalyzed cleavage of **83** into an aryl aldehyde and ketene (**23**) was a significant competing reaction during an attempted intramolecular cyclopropanation (equation 32).⁵⁷









Arylalkenylketene **85** from rhodium octanoate catalyzed Wolff rearrangement of **84** cyclized to form naphthol **86** (equation 33).⁵⁸ A similar cyclization is reported in Section 5.4.4.



The catalyzed Wolff rearrangement of **87** proceeded through the unobserved ketene **88** (equation 34), which reacted with *N*-hydroxysuccinimide forming **89** (equation 35).⁵⁹ Ultrasound has been found to accelerate the silver benzoate–catalyzed Arndt-Eistert reaction of aryl and alkyl diazo ketones in methanol.⁶⁰ It was also found that diazo ketones derived from amino acids underwent Wolff rearrangement followed by cycloaddition with imines using microwave irradiation.⁶¹ Even imines not substituted with aryl groups gave satisfactory reactions.



Ultrasonic irradiation of the vinylcyclopropyl diazo ketone **90** forming the ketene **91** was followed by a ketene-Cope reaction forming **92** in 96% yield (equation 36).^{62,63} Several related examples were studied, and some similar reactions of cyclopropylketenes are shown in Section 4.1.5.



The silver benzoate–catalyzed Arndt-Eistert reaction has been used for the preparation of  $\beta$ -amino acids through the reaction of diazo ketones **93** to give the ketene intermediates **94** (equation 37).⁶⁴ In all cases there was complete retention of stereochemistry at the migrating carbon in the Wolff rearrangement,⁶⁴ but for  $R^1 = Ph \ 10\%$  of the enantiomer was formed during preparation of **93**.⁶⁴ β-Amino-ketenes generated in this way reacted selectively with the less hindered hydroxy group of 3-methyl-1,3-dihydroxybutane,⁶⁵ and also showed selectivity in the acylation of 2-aminoethanol, isopropylidene-*D*-xylofuranose, sucrose, thymidine, and adenosine derivatives.⁶⁵ Cyclization of ketene **95** formed by a Wolff rearrangement led to proline precursor **96** (equation 38).⁶⁶ A similar cyclization was observed for ketenes derived from Ag⁺-catalyzed Wolff rearrangement of RCH(NHTs)-CH₂COCHN₂.⁶⁶ The Ag⁺-catalyzed Wolff rearrangement⁶⁷ proved useful in the Arndt-Eistert synthesis of *N*–Fmoc-β-homoamino acids.^{68,69} The reaction was promoted by ultrasound⁷⁰ and could be used in polymer-supported synthesis.⁷¹



R¹, R²: Me, BnO or *t*-BuO (**a**); Me, BnO or *t*-BuO (**b**); Bn, *t*-BuO (**c**); *t*-Bu, BnO (**d**); *t*-BuOCH₂, *t*-BuO (**e**); Me, Z-Ala (**f**); *i*-Pr, BnO (**g**)



Ketenes **98** generated by  $Ag^+$  or photochemical Wolff rearrangement of diazo ketones **97** (equation 39) reacted with 5'-amino-substituted oligonucleotides **99** to give oligoionucleopeptides (equation 40).⁷² These reactions were also successful with polymer-supported oligonucleotides. Wolff rearrangement of the di-*O*-cyclohexylidene-*D*-arabino-hexulose diazo ketone **101** induced by  $Ag_2O$  gave ketene **102**, which formed the hexenic acid derivative **103** (equation 41).⁷³





1-Diazo-3,3-dimethylalkyn-2-ones such as **104** with silver (I) catalyst did not give Wolff rearrangement, but instead gave intramolecular 1,3-dipolar cycloaddition leading to bicyclic pyrazoles **105** (equation 42).⁷⁴ In the absence of the geminal dimethyl groups, normal Arndt-Eistert products were observed upon reaction in EtOH.⁷⁴



Reaction of the diazo ketone **106** with  $Rh_2(OAc)_4$  in the presence of MeOH was proposed to proceed by formation of the oxonium ylide **107**, which opened to ketene **108**, which was trapped by MeOH to give the ester (equation 43).⁷⁵ This non-Wolff rearrangement was suggested to be of a hitherto unknown type. Dirhodium(II)-catalyzed reactions of secondary benzylic and allylic diazoacetates **109** may form ketones and ketene **(23)** by a non-Wolff mechanism (equation 44).⁷⁶ The formation of ketene was established by trapping with 2,5-dimethyl aniline.



The reaction of acyl chlorides with trimethylsilyldiazomethane followed by thermolysis in the presence of alcohols gives Arndt-Eistert products without the need to use diazomethane.⁷⁷ Carboxylic acids activated as mixed anhydrides by reaction with ethyl chloroformate reacted with Me₃SiCHN₂ gave diazo ketones, and one of these was used in an Arndt-Eistert reaction catalyzed by Ag ion.⁷⁸ Reaction of carboxylic acids with DCC and Me₃SiCHN₂ gave diazoketones, but yields did not exceed 50%.⁷⁹ The Rh(II) octanoate-catalyzed reaction of Et₃Si-substituted diazo ketones provides an unusual route to silylketenes **110** in yields of 12–80% (equation 45).⁷⁹

 $Ph \xrightarrow{O}_{N_2} SiEt_3 \xrightarrow{Rh(O_2CR)_2} \xrightarrow{Et_3Si}_{Ph} C=O$  (45) 110 (33%)

The AgO₂CPh-catalyzed reaction of **111** in CH₃OH gave products derived from the intramolecular capture of the ketocarbene **112**, as well as the capture by methanol of the ketene **113** from Wolff rearrangement (equations 46, 47).⁸⁰ The extremely hindered diazo ketone **115** was prepared from an acyl mesylate and gave Ag₂O-catalyzed Arndt-Eistert homologation via ketene **116** forming acid **117** in 35% yield based on the carboxylic precursor to **115** (equation 48) that was crucial to the synthesis of CP molecules.^{81,82}





Diazo esters **118** reacted with rhodium II acetate in the presence of water to form  $\beta$ , $\gamma$ -unsaturated esters **120** via ketenes **119** that underwent hydration and decarbox-ylation (equation 49).⁸³



2-Diazothiol esters **121** reacted in rhodium-catalyzed processes to form thiosubstituted ketenes **122** that were captured in [2+2] cycloaddition reactions with alkenes, dienes, imines, and alkynes (equation 50).⁸⁴



Cycloalkyl diazo ketones **124** reacted with  $Rh_2(O_2CCF_3)_4$  to form ketenederived products (dimers, 30–68%, and carboxylic acids, 12–35%), along with variable amounts of cyclization products.⁸⁵



### 3.3.3 Photochemical Wolff Rearrangement

Horner and coworkers demonstrated that photolysis of diazo ketones gave efficient ketene formation,^{86,87} but this method is not efficient for ketene isolation because of the possibility of photodecomposition of the formed ketene upon prolonged photolysis. Also, in concentrated solutions the ketenes can react with residual diazo ketone. Thus, while photolysis of neat diazopinacolone (**126**) formed *tert*-butylketene (**127**), this reacted with residual **126** to give lactone **128** in 62% yield (equation 52),⁸⁸ but in the presence of alcohols the derived esters were obtained. These problems were overcome in the preparation of **13** (equation 5).²⁰



Photolysis of the stereoisomeric 7,7-diaryldiazo-norbornenyl ketones **129** gave both normal Wolff rearrangements forming ketenes **130** leading to esters **131** (equation 53) and vinylogous Wolff rearrangements forming ketenes **133** (equation 54).⁸⁹ In the formation of **131** (Ar¹ = 2-pyridyl) there was a 2/1 preference for *endo* protonation, showing that the 2-pyridyl group did not give major assistance for *exo* protonation.⁸⁹



Photolysis of acyl diazo ketones **135** with alkenyl side chains sensitized with benzophenone in CH₃OH led to both cyclization products **137** arising from triplet ketocarbenes (**136a**) (equation 55) and Wolff rearrangement products from ketenes **138** resulting from conversion of the triplet ketocarbene to the singlet **136b** (equation 56).⁹⁰ For four CH₂ groups in the side chain the yield of cyclization product was 20% (equation 55), with 32% of ester **137b** from the ketene, and 37% insertion product (equation 56). With two and three CH₂ groups in the side chain, cyclization yields were 88 and 83%, respectively.⁹⁰



Photolysis of the diazo ketone **139** resulted not only in the formation of ketene **140** (which was trapped as the pyridine ylide **141**) via a singlet pathway (equation 57), but also in the formation of a short-lived triplet of **139** that decayed back to the ground state.⁹¹



Thermal Wolff rearrangement induced with a  $CO_2$  laser of **142** was proposed to give the oxygen-substituted ketene **143**, which underwent facile decarbonylation and eventual formation of **146** (equation 58).⁹² Analysis of the photochemical reaction of **142** by TRIR spectroscopy showed that ketene **145** was produced faster than the time resolution of the instrumentation, and led to the hypothesis that **145** was formed directly from **142** without the intervention of carbene intermediate **147** (equation 59).⁹³ Some of **147** was formed in competition with **143** but was not a precursor to **143**.



Photochemical Wolff rearrangement of **148** in a matrix was inhibited by the presence of CO or O₂, which react with the carbene **149** to form the ketene **150** with an IR band at 2166 cm⁻¹ or a carbonyl oxide, respectively, instead of the ketene **151** (equation 60).⁹⁴ In solution, photolysis of **148** gave dimers of **151**, and the ketene was also trapped by [2 + 2] cycloaddition with cyclopentadiene.⁹⁵



Photolysis of diazo Meldrum's acid (152) near the absorption maximum at 248 nm gave mainly Wolff rearrangement to ketene 154, which formed ester 155 in methanol (equation 61).⁹⁶ Photolysis at 355 nm of 152, however, gave almost exclusively formation of the diazirine 153 (equation 61).⁹⁶ Thermolysis of 153 resulted in reformation of 152, whereas photolysis of 153 gave some 152 but mainly proceeded through a diacylcarbene leading to Wolff rearrangment. Photolysis of 152 in poly(methyl methacrylate) using picosecond IR detection led to formation within 20 ps of the keto ketene 154, as detected by its IR absorption centered at  $2161 \text{ cm}^{-1}$  (equation 61).^{97,98} This ketene was observed by picosecond IR spectroscopy at room temperature^{97,98} and is similar to photoresist materials discussed in Section 4.1.10.



Photolysis of the bis(diazo)cyclohexanetetraone **156** in an Ar matrix at 17 K gave the diazo ketene **157**, with IR bands at 2165 and 2131 cm⁻¹. Upon further photolysis, this gave the cyclic diacylbisketene **158**, with IR bands at 2101 and 1747 cm⁻¹, and on further photolysis this gave carbon suboxide ( $C_3O_2$ , **159**) (equation 62).^{99–101}



Photolysis of the bis(diazo) diketone **160**, which has two nonequivalent diazo ketone groupings, with light of >350 nm in an Ar matrix at 12 K, gave rise to a product with IR bands at 2142, 2088, 1690, and 1381 cm⁻¹ that was assigned the structure **161** on the basis of the PM3 calculated IR frequencies of 2145 (C=N₂), 2061 (C=C=O), 1722 (C=O), and 1382 (indanone in-plane deformation) cm⁻¹ (equation 63). Upon further photolysis this gave a band at 2152 cm⁻¹ assigned to the bis(ketene) **162** (equation 63).¹⁰² Irradiation of **160** in the matrix with light of >420 nm gave rise to IR bands at 2136, 2070, 1695, and 1468 cm⁻¹ that were assigned to the structure **163** on the basis of the PM3 calculated IR frequencies of 2133 (C=N₂), 2056 (C=C=O), 1743 (C=O), and 1466 (5-ring in-plane deformation). Further photolysis led to bands at 2116 and 1780 cm⁻¹ assigned to

**164** (equation 64). The molecular orbital analysis of the PM3 calculations suggested the participation of higher excited states in these photoreactions.¹⁰²



Photolysis of **160** in benzene-MeOH led to products **166** and **167**, which were assigned to result from **165** and **163**, respectively (equations 65, 66). The yield of **165** was greatly favored when light of >420 nm was used.¹⁰³



Photolysis of the diazoketone **168** in a matrix gave an IR absorption at  $2117 \text{ cm}^{-1}$  due to ketene **169**, although preparative experiments in solution in the presence of CH₃OH or Me₂NH gave products attributed to cleavage of the intermediate keto carbene (equation 67).^{104,105} The structure of ketene **169** was calculated at the MP2/6-31G* level.¹⁰⁵



168

169 (IR 2117 cm⁻¹)

Heating of **170** with trapping of the volatile products in solid argon led to **171**, which upon photolysis gave the ene-yne ketene **172**, identified by its IR band at  $2128 \text{ cm}^{-1}$  (equation 68).¹⁰⁶ Photolysis of **173** also formed **172**, and deuterium labeling experiments favored a pathway through ketene **174** (equation 69).¹⁰⁶ The conformation shown for **174** was calculated to be the most favorable.



Photolysis of diazo ketone **175** in methanol forming the ketene **176**, which was captured as the methyl ester **177**, was achieved by near-IR photolysis with two-photon absorption inducing the Wolff rearrangement (equation 70).¹⁰⁷ Diazo ketone **175** was irradiated with an ultrafast pulsed laser operating at 800 nm, a wavelength with no spectral overlap with that of **175**. The yield of **177** as a function of the irradiation time was fit by an equation derived for two-photon photolysis, and the yield was also fit by an equation for two-photon absorption. This system was also utilized for two-photon photolithography.¹⁰⁷



Long chain diazo ketones on water surfaces underwent Wolff rearrangement to give ketenes that undergo competitive hydration and dimerization.¹⁰⁸ Irradiation of **178** on a thin film with an electron beam induced formation of **179**,¹⁰⁹ and this may involve an initial ionization step analogous to mass spectrometry (equation 71).^{110–112} Relative migratory aptitudes of substituents in diazo diketones **180** forming ketenes **181** have been compared for thermal photochemical and electron impact–induced Wolff rearrangement.¹¹²





Photolysis of **182** in argon at 14 K gave carbene **183**, which was observed by IR and UV and reacted with CO to give ketene **184** (equation 73).¹¹³ Photolysis of **184** also appeared to give a low yield of ketene **185**, as identified by the IR band at 2148 cm⁻¹ (equation 73).¹¹³ Calculations have been used to evaluate barriers to Wolff rearrangements in carbonyl carbenes, and the barriers are usually correlated with the exothermicities of the reactions.¹¹⁴



## 3.3.4 Other Routes to Ketocarbene Rearrangements

Aryliodonium ylides provided another route to the keto-carbene intermediates involved in the Wolff rearrangement. Refluxing **186** in  $CH_2Cl_2$  with an equimolar amount of an amine was proposed to lead through the diacyl ketene **187**, forming amide **188** (equation 74).¹¹⁵



Acylketene **189** generated in a matrix by Wolff rearrangement also underwent decarbonylation upon further photolysis and gave Wolff rearrangement leading to ketene **190** (equation 75).¹¹⁶ Other examples of Wolff rearrangements from photolysis of ketoketenes were demonstrated.¹¹⁶



## **REFERENCES TO SECTIONS 3.3 AND 3.3.1–3.3.4**

- 1. Wolff, L. Liebigs Ann. Chem. 1902, 325, 129-195.
- 2. Wolff, L. Liebigs Ann. Chem. 1912, 394, 23-59.
- 3. Schröter, G. Chem. Ber. 1909, 41, 2336-2349.
- 4. Kirmse, W. Eur. J. Org. Chem. 2002, 2193-2256.
- Zollinger, H. Diazo Chemistry II; VCH Publishers, Weinheim, 1995, Chap. 8.6, pp. 344– 358.
- Toscano, J. P. In Advances in Carbene Chemistry; Brinker, U. H., Ed; JAI Press, Greenwich, CT, 1998, Vol. 2, pp. 215–244.
- 7. Regitz, G.; Maas, G. Diazo Compounds; Academic Press, Orlando, FL, 1986.
- 8. Bachmann, W. E.; Struve, W. S. Org. React. 1942, 1, 38-62.
- 9. Maas, G. *Methoden der Organischen Chemie*, 4th ed, Vol. 19b, *Carbene (Carbenoid)*; Regitz, M., Ed.; Thieme, Stuttgart, 1989, pp. 1231–1259.
- Gill, G. B. In *Comprehensive Organic Synthesis*; Pattenden, G., Ed; Pergamon, Oxford, 1991, Vol 3, pp. 887–912.
- 11. Ando, W. In *The Chemistry of Diazonium and Diazo Groups*; Patai, S. Ed.; New York, 1978, Vol. I, pp. 341–487.
- Eistert, B.; Regitz, M.; Heck, G.; Schwall, H. In *Methoden der organischen Chemie*, 3rd ed., Vol. 10/4, *Carbene (Carbenoid)*; Regitz, M., Ed.; Thieme, Stuttgart, 1989, pp. 473– 893.
- 13. Meier, H.; Zeller, K.-P. Angew, Chemie. Int. Ed. 1975, 14, 32-43.
- 14. McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1992, 114, 6553-6555.
- 15. McAllister, M. A.; Tidwell, T. T. J. Org. Chem. 1994, 59, 4506-4515.
- 16. Kaplan, F.; Meloy, G. K. J. Am. Chem. Soc. 1966, 88, 950-956.
- 17. Newman, M. S.; Arkell, A. J. Org. Chem. 1959, 24, 385-387.
- Marfisi, C.; Verlaque, P.; Davidovics, G.; Pourcin, J.; Pizzala, L.; Aycard, J.-P.; Bodot, H. J. Org. Chem. 1983, 533–537.
- de Groot, A.; Boerma, J. A.; de Valk, J.; Wynberg, J. J. Org. Chem. 1968, 33, 4025–4029.
- 20. Kaplan, F.; Mitchell, M. L. Tetrahedron Lett. 1971, 88, 759-762.
- 21. Haiss, P.; Zeller, K.-P. Z. Naturforsch. 2003, 58b, 595-605.
- 22. Haiss, P.; Zeller, K.-P. Org. Biomol. Chem. 2003, 1, 2556-2558.
- 23. Rau, H.; Bokel, M. J. Photochem. Photobiol. A. 1990, 53, 311-322.
- 24. Toscano, J. P.; Platz, M. S.; Nikolaev, V.; Cao, Y.; Zimmt, M. B. J. Am. Chem. Soc. **1996**, 118, 3527–3528.
- Zhu, Z.; Bally, T.; Stracener, L. L.; McMahon, R. J. J. Am. Chem. Soc. 1999, 121, 2863– 2874.
- 26. Wang, Y.; Yuzawa, T.; Hamaguchi, H.; Toscano, J. P. J. Am. Chem. Soc. 1999, 121, 2875–2882.
- 27. Wang, J.; Likhotvorik, I.; Platz, M. S. J. Am. Chem. Soc. 1999, 121, 2883-2890.
- 28. Wang, Y.; Hadad, C. M.; Toscano, J. P. J. Am. Chem. Soc. 2002, 124, 1761-1767.
- 29. Chiang, Y.; Kresge, A. J.; Popik, V. V. J. Am. Chem. Soc. 1999, 121, 5930-5932.

- Smith, L. I.; Hoehn, H. H. Organic Syntheses, Coll. Vol. III; Hornung, E. C., ed.; Wiley, New York, 1955, pp. 356–358.
- 31. Yates, P.; Clark, T. J. Tetrahedron Lett. 1961, 435-439.
- 32. Huisgen, R.; Binsch, G.; Ghosez, L. Chem. Ber. 1964, 97, 2628-2639.
- 33. Yates, P.; Abrams, G. D.; Betts, M. J.; Goldstein, S. Can. J. Chem. 1971, 49, 2850-2860.
- 34. Jugelt, W.; Schmidt, W. Tetrahedron 1969, 25, 969-984.
- 35. Melzler, Jenny, E. F. Tetrahedron Lett. 1968, 4503-4506.
- 36. Nikolaev, V. A.; Popik, V. V. Tetrahedron Lett. 1992, 33, 4483-4486.
- 37. Popik, V. V.; Nikolaev, V. A. J Chem Soc. Perkin Trans. 2 1993, 1791-1793.
- 38. Regitz, M.; Bartz, W. Chem. Ber. 1970, 103, 1477-1485.
- 39. Bartz, W.; Regitz, M. Chem. Ber. 1970, 103, 1463-1476.
- Sudrik, S. G.; Chavan, S. P.; Chandrakumar, K. R. S.; Pal, S.; Date, S. K.; Chavan, S. P.; Sonawane, H. R. J. Org. Chem. 2002, 67, 1574–1579.
- 41. Sugimura, T.; Tei, T.; Okuyama, T. Tetrahedron Lett. 2003, 44, 3115-3117.
- 42. Fujita, M.; Matsushima, H.; Sugimura, T.; Tai, A.; Okuyama, T. J. Am. Chem. Soc. 2001, 123, 2946–2956.
- 43. Sugimura, T. Chem. Eur. J. 2004, 1185–1192.
- 44. Chantegrel, B.; Deshayes, C.; Faure, R. Tetrahedron Lett. 1995, 36, 7859-7862.
- 45. Leóst, F.; Chantegrel, B.; Deshayes, S. Tetrahedron 1997, 53, 7557-7576.
- 46. Leóst, F.; Chantegrel, B.; Deshayes, S. Tetrahedron 1998, 54, 6457-6474.
- 47. Chen, Y. P.; Chantegrel, B.; Deshayes, C. Heterocycles 1995, 41, 175-186.
- 48. Dayoub, W.; Diab, Y.; Doutheau, A. Tetrahedron Lett. 2001, 42, 8455-8457.
- 49. Lee, V.; Newman, M. S. In *Organic Syntheses*; Noland, W. E., Ed.; Wiley, New York, 1988, Coll. Vol. VI, pp. 613–615.
- Sudrik, S. G.; Maddanimath, T.; Chaki, N. K.; Chavan, S. P.; Chavan, S. P.; Sonawane, H. R.; Vijayamohanan, K. Org. Lett. 2003, 5, 2355–2358.
- 51. Julian, R. R.; May, J. A.; Stoltz, B. M.; Beauchamp, J. L. J. Am. Chem. Soc. 2003, 125, 4478–4486.
- 52. Julian, R. R.; May, J. A.; Stoltz, B. M.; Beauchamp, J. C. Angew. Chem. Int. Ed. 2003, 42, 1012–1015.
- 53. Pirrung, M. C.; Morehead, A. T., Jr. J. Am. Chem. Soc. 1994, 116, 8991-9000.
- 54. Yong, K.; Salim, M.; Capretta, A. J. Org. Chem. 1998, 63, 9828-9833.
- 55. Smith, A. B., III; Toder, B. H.; Branca, S. J. J. Am. Chem. Soc. 1984, 106, 3995-4001.
- 56. Lee, Y. R.; Suk, J. Y.; Kim, B. S. Tetrahedron Lett. 1999, 40, 8219-8221.
- 57. Doyle, M. P.; Hu, W.; Chapman, B.; Marnett, A. B.; Peterson, C. S.; Vitale, J. P.; Stanley, S. A. J. Am. Chem. Soc. 2000, 122, 5718–5728.
- Karady, S.; Amato, J. S.; Reamer, R. A.; Weinstock, L. M. *Tetrahdron Lett.* 1996, 37, 8277–8280.
- 59. Blagbrough, I. S.; Moya, E.; Walford, S. P. Tetrahedron Lett. 1996, 37, 551-554.
- Winum, J.-Y.; Kamal, M.; Leydet, A.; Roque, J.-P.; Montero, J.-L. *Tetrahedron Lett.* 1996, *37*, 1781–1782.
- 61. Linder, M. R.; Podlech, J. Org. Lett. 2001, 3, 1849-1851.
- 62. Sarpong, R.; Su, J. T.; Stoltz, B. M. J. Am. Chem. Soc. 2003, 125, 13624-13625.

- Su, J. T.; Sarpong, R.; Stoltz, B. M; Goddard, W. A., III. J. Am. Chem. Soc. 2004, 126, 24–25.
- 64. Podlech, J.; Seebach, D. Liebigs Ann. 1995, 1217-1228.
- 65. Guibourdenche, C.; Podlech, J.; Seebach, D. Liebigs Ann. 1996, 1121-1129.
- 66. Dieterich, P.; Young, D. W. Tetrahedron Lett. 1993, 34, 5455-5458.
- 67. Wang, J.; Hou, Y. J. Chem. Soc., Perkin Trans. 1 1998, 1919–1923.
- Leggio, A.; Liguori, A.; Procopio, A.; Sindona, G. J. Chem. Soc. Perkin Trans. 1, 1997, 1969–1971.
- 69. Ananda, K.; Gopi, N. H.; Babu, V. V. S. Indian J. Chem. Sec. B 2001, 40, 790-795.
- 70. Müller, A.; Vogt, C.; Sewald, N. Synthesis 1998, 837-841.
- 71. Guichard, G.; Abele, S.; Seebach, D. Helv. Chim. Acta 1998, 81, 187-206.
- 72. Guibordenche, C.; Seebach, D.; Natt, F. Helv. Chim. Acta 1997, 80, 1-13.
- Glebova, Z. I.; Kornilov, V. I.; Nazarova, S. I.; Zhadanov, Y. A. Russ. J. Gen. Chem. 2000, 70, 1906–1909.
- 74. Kende, A. S.; Journet, M. Tetrahedron Lett. 1995, 36, 3087-3090.
- 75. Oku, A.; Ohki, S.; Yoshida, T.; Kimura, K. Chem. Commun. 1996, 1077-1078.
- 76. Doyle, M. P.; Dyatkin, A. B.; Autry, C. L. J. Chem. Soc. Perkin Trans. 1 1995, 619-621.
- 77. Aoyama, T.; Shioiri, T. Chem. Pharm. Bull. 1981, 29, 3249-3255.
- 78. Cesar, J.; Dolenc, M. S. Tetrahedron Lett. 2001, 42, 7099-7102.
- 79. Marsden, S. P.; Pang, W.-K. Chem. Commun. 1999, 1199-1200.
- Schantl, J. G.; Rettenbacher, A. S.; Wurst, K. Angew. Chem. Int. Ed. 1998, 37, 2229– 2231.
- Nicolaou, K.; Baran, P. S.; Zhong, Y.-L.; Choi, H.-S.; Yoon, W. H.; He, Y.; Fong, K. C. Angew. Chem. Int. Ed. 1999, 38, 1669–1675.
- Nicolaou, K.; Baran, P. S.; Zhong, Y.-L.; Fong, K. C.; He, Y.; Yoon, W. H.; Choi, H.-S. Angew. Chem. Int. Ed. 1999, 38, 1676–1678.
- 83. Taylor, E. C.; Davies, H. M. L. Tetrahedron Lett. 1983, 24, 5453-5456.
- 84. Lawlor, M. D.; Lee, T. W.; Danheiser, R. L. J. Org. Chem. 2000, 65, 4375-4384.
- Ceccherelli, P.; Curini, M.; Marcotullio, M. C.; Pisani, E.; Rosati, O. *Tetrahedron* 1997, 53, 8501–8506.
- 86. Horner, L.; Spiestchka, A.; Gross, A. Liebigs Ann. 1951, 573, 17-30.
- 87. Kirmse, W.; Horner, L. Chem. Ber. 1956, 89, 2759-2765.
- 88. Wiberg, K. B.; Hutton, T. W. J. Am. Chem. Soc., 1954, 76, 5367-5371.
- 89. Zimmerman, H. E.; Wang, P. Can. J. Chem. 2003, 81, 517-524.
- 90. Fien, J.; Kirmse, W. Angew. Chem. Int. Ed. 1998, 37, 2232-2234.
- 91. Ortica, F.; Pohlers, G.; Scaiano, J. C.; Cameron, J. F.; Zampini, A. Org. Lett. 2000, 2, 1357–1360.
- Werstiuk, N. H.; Ma, J.; Roy, C. D.; Kresge, A. J.; Jefferson, E. Can. J. Chem. 1995, 73, 1738–1740.
- 93. Wang, Y.; Toscano, J. P. J. Am. Chem. Soc. 2000, 122, 4512-4513.
- 94. McCluskey, A.; Dunkin, I. R. Aust. J. Chem. 1995, 48, 1107-1116.
- Pomerantz, M.; Levanon, M.; Gu, X.; Dias, H. V. R. *Tetrahedron* 1997, 53, 10019– 10028.

- Bogdanova, A.; Popik, V. V. J. Am. Chem. Soc. 2003, 125, 1456–1457; 14153–14162; 2004, 126, 11293–11302.
- 97. Lippert, T.; Koskelo, A.; Stoutland, P. O. J. Am. Chem. Soc. 1996, 118, 1551-1552.
- 98. Lippert, T.; Stoutland, P. O. Appl. Surf. Sci. 1997, 109/110, 43-47.
- 99. Maier, G.; Reisenauer, H. P.; Balli, H.; Brandt, W.; Janoschek, R. Angew. Chem. Int. Ed. Engl. 1990, 29, 905–908.
- 100. Maier, G. Pure Appl. Chem. 1991, 63, 275-282.
- 101. Ohana, T.; Yabe, A. Chem. Lett. 1996, 325-326.
- 102. Murata, S.; Kobayashi, J.; Kongou, C.; Miyata, M.; Matsushita, T.; Tomioka, H. J. Am. Chem. Soc. **1998**, 120, 9088–9089.
- 103. Murata, S.; Kobayashi, J.; Kongou, C.; Miyata, M.; Matsushita, T.; Tomioka, H. J. Org. Chem. 2000, 65, 6082–6092.
- 104. Wiberg, K. B.; Snoonian, J. R.; Lahti, P. M. Tetrahedron Lett. 1996, 37, 8285-8288.
- 105. Wiberg, K. B.; Snoonian, J. R. J. Org. Chem. 1998, 63, 1390-1401.
- 106. Sander, W.; Albers, R.; Komnick, P.; Wandel, H. Liebigs Ann./Recueil. 1997, 901– 905.
- 107. Urdabayev, N. K.; Popik, V. V. J. Am. Chem. Soc. 2004, 126, 4058-4059.
- 108. Holden, D. A.; Ringsdorf, H.; Haubs, M. J. Am. Chem. Soc. 1984, 106, 4531– 4536.
- 109. Pacansky, J.; Coufal, H. J. Am. Chem. Soc. 1980, 102, 410-412.
- 110. De Jongh, D. C.; Van Fossen, R. Y.; Dusold, L. R.; Cava, M. P. Org. Mass. Spec. **1970**, 3, 31.
- 111. Zeller, K.-P.; Meier, H.; Müller, E. Liebigs Ann. 1971, 749, 178-182.
- 112. Zeller, K.-P.; Meier, H.; Müller, E. Tetrahedron 1972, 28, 5831-5838.
- 113. Likhotvorik, I.; Zhu, Z.; Tae, E. L.; Tippmann, E.; Hill, B. T.; Platz, M. S. J. Am. Chem. Soc. **2001**, *123*, 6061–6068.
- 114. Scott, A. P.; Platz, M. S.; Radom, L. J. Am. Chem. Soc. 2001, 123, 6069-6076.
- 115. Malamidou-Xenikaki, E.; Spyroudis, S.; Tsanakopoulou, M. J. Org. Chem. 2003, 68, 5627–5631.
- 116. Leung-Toung, R.; Wentrup, C. J. Org. Chem. 1992, 57, 4850-4858.

# 3.4 KETENES BY PHOTOCHEMICAL AND THERMOLYTIC METHODS

#### 3.4.1 Ketenes from Cyclobutanones and Cyclobutenones

The ring openings of cyclobutanones to form ketenes¹ and of cyclobutenones to form vinylketenes^{1,2} have been the subject of reviews and play an important part in the chemistry of cyclobutanes.^{3–6}

The photochemical ring opening of cyclobutanone (1) is a classic problem in chemistry,^{7,8} and the details of the Norrish type 1 photocleavage leading to

diradical **2**, and the subsequent cleavage to either ketene (**3**) and ethylene (equation 1) or cyclopropane and carbon monoxide (CO), have been evaluated using femtosecond time-resolved mass spectrometry.⁹ Further computational studies of the photochemical cycloreversion of cyclobutanone to form ketene and ethylene, as well as the competing decarbonylation giving cyclopropane and propene and formation of the oxocarbene, have been carried out (equation 2).¹⁰



Thermolysis of the substituted cyclobutanone **5** gave stereoselective formation of *Z*-butene and ketene (equation 3),¹¹ while ring opening of the acyl cyclobutanone **6** gave complete conversion to the  $\delta$ -lactone **7** (equation 4).¹² Vinylcyclobutanones gave ring-expanded enones in 45–81% yields at 145 to 190 °C,¹³ but the intermediacy of ketenes in these reactions has not been proven.



Strained cyclobutanones cleaved more readily to ketenes, and in the case of bicyclo [1.1.1]pentan-2-one 8 formed ketene 9 (equation 5),¹⁴ while bicyclo[2.2.0]-hexanone 10 formed ketene 11 (equation 6).^{15,16}



The kinetics of decomposition of substituted cyclobutanones in the gas phase from 192 to 285 °C have been measured, and as in the case of **12** these cleaved in two directions giving two sets of products (equation 7).^{17–21} The results were interpreted as implicating two zwitterionic transition states, **13** and **14**, resembling the  $[\pi 2_s + \pi 2_a]$  cycloaddition (Section 5.4.1) for ketene formation.



Thermal and photochemical cleavage of cyclobutanones may proceed in different directions, as with 2,2,3-triphenylcyclobutanone (**15**), which formed diphenylketene in the thermal reaction but ketene upon photolysis (equation 8).²² The two reaction paths were proposed to involve isomeric biradicals **16** or **17**, respectively,²² but zwitterions cannot be excluded.

Photolysis of the cyclobutanone **18** gave the ketene **19**, which was trapped as the ester **20**, which was used in a synthesis of dihydrocleavamine (equation 9).²³ Treatment the diazo ketone **21** with  $Rh_2(OAc)_4$  resulted in the formation of the acid **23** derived from the allylketene **22** in a reaction that may have involved a bicyclo[2.1.0]pentanone intermediate (equation 10).²⁴ This is an example of the viny-logous Wolff rearrangement (Section 3.3.2). Dichlorocyclobutenones **25** prepared by dichloroketene cycloaddition undergo photoelimination, forming dichloroalkenes in 54–65% yields and  $CH_2=C=O$  (equation 11).²⁵





Photolysis of cyclobutanone **26** in MeOH occurred with ring opening to the vinylketene **27** as well as ring expansion to the oxocarbene **28** (equation 12), and these were captured as the MeOH addition products **29** and **30**, respectively (equation 13).²⁶

25

3

24



Thermolysis of the diazo ketone **31** with PhCO₂Ag formed not only the normal ketene **32** trapped as the ester **33** (equation 14), but also the bicyclo[2.1.0]pentan-2-one **34**, which was an intermediate in the vinylogous Wolff rearrangement (Section 3.3.2), and opened to the ketene **35** (equation 15).²⁷





The hydroxycubane **37** reacted by homoketonization, forming the cyclobutanone **38**, which opened to the ketene **39** (equation 16).²⁸



Thermal ring opening of 2,4,4-triphenylcyclobutenone **40** was found to give vinylketene **41**, which cyclized, forming naphthol **42** (equation 17).²⁸ This procedure has been found to be quite general and has been widely utilized in synthesis (Smith-Hoehn reaction).^{2,4} The substrate **40** was itself prepared by diphenylketene cycloaddition with phenylacetylene.²⁹



The ring opening of cyclobutenone (**43**), forming vinylketene (**44**), was studied computationally, and was found to be essentially thermoneutral, with a barrier of 21 kcal/mol (equation 18).³⁰ It was shown that optically active cyclobutenone **45** underwent racemization on heating in CHCl₃ at 100 °C at rates that could be conveniently measured (equation 19).³¹ The reaction involved reversible ring opening to the vinylketene **46**, which was trapped with EtOH, forming **47** (equation 19).³¹ Further investigation showed that thermolysis of **45** gave a preference for formation of *Z*-**46**, but upon prolonged reaction racemization occurred.³² Photolysis of **45** 





The torquoselectivity in the thermal ring opening of cyclobutenones **48** has been examined by theoretical means and proposed to occur with outward rotation of electron donor groups to form vinylketenes **49** (equation 20).³³



Studies of solvent effects on the cyclobutenone ring opening of **50** ( $\mathbf{R} = \mathbf{Me}$ ,  $\mathbf{R}^1 = \mathbf{H}$ ) reveal that these were rather small (Table 3.4), with somewhat slower rates in solvents of greater polarity, as measured by the  $E_T$  values. This suggests that the rather polar enone starting material was stabilized in the polar solvent (equation 21).^{34,35} In alcoholic solvents, **51** was trapped as the ester, but cyclized to the naphthol in nonnucleophilic solvents.^{34,35}



TABLE 3.4Solvent Effects on Ring Opening ofCyclobutenone 50

-		
Solvent	$E_{\mathrm{T}}$	$k \times 10^4$ , s ⁻¹ (55 °C)
Cyclohexane	31.2	5.36
Benzene	34.6	4.14
EtOAc	38.1	2.53
DMF	43.8	1.75
HOAc	51.1	1.83
EtOH	51.9	2.01
MeOH	55.5	1.78

R	$\mathbb{R}^1$	$\mathbb{R}^2$	$k \times 10^{6} ({ m s}^{-1}) \ 80 \ {}^{\circ}{ m C}$
Me	Me	Me	1.1
Me	Me	Ph	90
Me	Et	Ph	33
Me	<i>i</i> -Pr	Ph	12.7
Me	Ph	Ph	2,430
Et	Ph	Ph	3,710
Н	Ph	Ph	20,400

TABLE 3.5 Substituent Effects on Cyclobutenone Ring Opening

 $k \times 10^{6} \,(\text{s}^{-1}) \,80 \,^{\circ}\text{C}$ 



Changing the substituents on the cyclobutenones **51** has major effects on the reactivity in ring opening (equation 22), with a difference of a factor of  $2 \times 10^4$  between the first and last entries (Table 3.5).³⁵



Other examples of these reactions are given in Sections 3.5 and 5.4.4.

Heating of **53** gave the ketene **54**, which formed the naphthol **55** (equation 23).³⁷ The vinylcyclobutenone **56** formed the ketene **57**, which gave hydroquinone **58**, used in the synthesis of echinochrome A (equation 24).³⁸ This reaction has also been used in the synthesis of naphthoquinones³⁹ and in heterocyclic analogues for conversion of diethyl squarate **59** to the furan-containing ketene **60** to form **61** (equation 25).⁴⁰ The analogous reaction of vinylcyclobutenone **62** proceeded via the ketene **63** to give **64**, which was converted to the antiparasitic monoterpene espintanol (equation 26).⁴¹





Thermal rearrangement of cyclobutenone **66**, formed from the cyclobutenedione **65**, formed the vinylketene **67**, which cyclized and was oxidized by air to the naphthoquinone **68** (equation 27).⁴² Conversion of **68** to **69** and thermolysis gave the new vinylketene **70** (equation 28), which cyclized to phenanthrene derivative **71** (equation 29).⁴² This methodology was also adapted for the preparation of naphthoquinone-substituted porphyrins.⁴³



0 71 (18%)





НÒ

ö

Thermolysis of the cyclobutenone 72 (R = TBDMS,  $R^1 = TIPS$ ) gave the vinylketene 73 (equation 30), which underwent cyclization and, after air oxidation of the product and removal of the protecting groups, gave furaquinocin E (74) (equation 31).⁴⁴



Cycloaddition of ketene 76, formed by a photochemical Wolff rearrangement of diazo ketone 75, with alkynes 77 gave cyclobutenones 78 (equation 32), which on continued photolysis gave the ketenes **79**, which cyclized to **80** (equation 33).⁴⁵ Other polycyclic natural products and highly substituted naphthols were prepared

using this methodology.46,46a



Oxidation of hydroxycyclobutenones **81** (R=Me, *n*-Bu, CH₂COPh, Ph, 2-furyl) with lead tetraacetate was proposed to form oxy radicals **82**, which upon ring opening gave acyl radical intermediates **83** (equation 34) that led to furanones **84a** and **84b** (equation 35).⁴⁷ The structures of the intermediate acyl radicals **83** were calculated using (UHF/PM3) methods.⁴⁷



The ethenylcyclobutenone **85** underwent thermal ring opening forming divinylketene **86**, which was proposed to undergo migration of sulfur to form **89** (equations 36, 37).⁴⁸ The vinylcyclobutenone **90** formed the vinyl ene-yne ketene **91**, which gave 92 (equation 38).⁴⁹



Ring opening of alkynylcyclobutenones **93** formed ketenes **94**, which cyclized, forming quinones **95** after oxidation (equation 39).⁵⁰ This was adapted using ²H- and ¹³C-labeled propynyl units to form labeled **95**, which was converted to ubiquinones.⁵⁰ Thermolysis of alkynylcyclobutenones **96** proceeded through alkynylketenes **97**, which cyclized by [2 + 2] cycloaddition to bicyclo[3.2.0]heptenones **98** (equation 40).⁵¹



Ring opening of **99** gave ketene **100**, which reacted by ring closure on the pyran ring, forming **101** after deprotection and oxidation (equation 41).⁵² The alkynylcy-clobutenone **102** formed ene-yne ketene **103** (equation 42), which upon cyclization gave diradical **104**, which formed **105** by a radical cyclization and intramolecular hydrogen atom abstraction (equation 43).⁵³



Other cyclizations of ene-yne ketenes were proposed for the synthesis of dihydroisoquinoline **108** from **106** through the intermediate ketene **107** followed by oxidation (equation 44),⁵³ and for the formation of dihydrobenzophenanthridine



111 by ring opening of 109 to the ketene 110 (equation 45).⁵⁴

Ring opening of **112** gave ketene **113**, which formed **114** by electrocyclization on the pyridine ring (equation 46).⁵⁵ The aminoalkyl group in **116** reacted by an intramolecular nucleophilic addition forming dihydropyridone **117** (equation 47).⁵⁶ The in situ–generated 4-acylcyclobutenone **118** (equation 48) gave ring opening to ketene **119**, which formed  $\gamma$ -pyrone **120** (equation 49).⁵⁷ The dihydropyridine-substituted ketene **122** gave electrocyclization to dihydroquinoline **123** (equation 50), which was converted to a quinolinylquinone.⁵⁸





Allenylcyclobutenone **124** gave give torquoselective³³ ring opening with outward rotation of the OH group to form the ene-allenyl ketene **125**, that cyclized to a transient quinone methide that tautomerized to **126** (equation 51).⁵⁹



The 3-acyl-4-aminocyclobutenone 127 reacted by ring opening to ketene 128, which cyclized to 129 (equation 52).⁶⁰ The 4-dimethylaminophenyl-substituted

derivative **130** was proposed to form ketene **131**, which cyclized by an ionization and recombination of acetate ion to furan derivative **132** (equation 54).⁶⁰ Related cyclizations onto furan, thiophene, and pyrrole rings have also been observed,  61,62  including a xanthone synthesis.⁶²







4-Allyl-4-ethoxycyclobutenone **133** on thermolysis gave the vinylketene **134**, which formed bicyclo[3.2.0]heptanone **135** through an intramolecular [2 + 2] cycloaddition reaction (equation 55).^{63,64} Related substrates give similar cyclizations.^{65–68} 2-Vinyl-4-allylcyclobutenone **136** formed divinylketene **137**, which gave vinyl bicyclo[3.2.0]heptenone **138** (equation 56).⁶⁷ 2-Dienyl-substituted cyclobutenone **139** gave ketene **140**, which cyclized to give phenol **141**, which formed benzofuran **142** with acid (equation 57).⁶⁸





#### **REFERENCES FOR SECTION 3.4.1**

- 1. Lee-Ruff, E. Adv. Strain Org. Chem. 1991, 1, 167-213.
- 2. Moore, H. W.; Yerxa, B. R. Chemtracts: Org. Chem. 1992, 5, 273-313
- 3. Bellus, D.; Ernst, B. Angew. Chem. Int. Ed. Engl. 1988, 27, 797-827.
- 4. Wang, K. K. Chem. Rev. 1996, 96, 207-222.
- 5. Moore, H. W.; Yerxa, B. R. Adv. Strain Org. Chem. 1995, 4, 81-162.
- Durst, T.; Breau, L. In *Comprehensive Organic Synthesis*, B. M. Trost, Ed., Vol. 5, Chap. 6.1, pp. 675–697, Elsevier (1991).
- 7. Benson, S. W.; Kistiakowsky, G. B. J. Am. Chem. Soc. 1942, 64, 80-86.
- Das, M. N.; Kern, F.; Coyle, T. D.; Walters, W. D. J. Am. Chem. Soc. 1954, 76, 6271– 6274.
- 9. Diau, E. W.-G.; Köffing, C.; Zewail, A. H. Chem. Phys. Chem. 2001, 294-309.
- 10. Chen, Y.; Ye, S. Int. J. Quant. Chem. 2004, 97, 725-735.
- 11. Carless, H. A.; Lee, E. K. C. J. Am. Chem. Soc. 1970, 92, 4482.
- 12. Huet, F.; Lechevallier, A.; Conia, J.-M. Chem. Lett. 1981, 1515–1519.
- 13. Huston, R.; Rey, M.; Dreiding, A. S. Helv. Chim. Acta 1982, 65, 451-461.
- 14. Sponsler, M. B.; Dougherty, D. A. J. Org. Chem. 1984, 49, 4978-4984.
- 15. McDonald, R. N; Reineke, C. E. Tetrahedron Lett. 1966, 2739.
- 16. McDonald, R. N.; Reineke, C. E. J. Org. Chem. 1967, 32, 1888-1893.

### **130** PREPARATION OF KETENES

- 17. Egger, K. W. J. Am. Chem. Soc. 1973, 95, 1745-1750.
- 18. Cocks, A. T.; Egger, K. W. J. Chem. Soc. Perkin Trans. 2 1973, 835-838.
- 19. Egger, K. W.; Cocks, A. T. J. Chem. Soc. Perkin Trans. 2 1972, 211-214.
- 20. McGee, T. H.; Schleifer, A.; Egger, K. W. J. Phys. Chem. 1972, 76, 963-967.
- 21. Schiess, P.; Fünfschilling, P. Tetrahedron Lett. 1967, 5191-5194.
- 22. Kende, A. Tetrahedron Lett. 1967, 2661-2666.
- 23. Kanada, R. M.; Ogasawara, K. Tetrahedron Lett. 2001, 42, 7311-7313.
- 24. Shi, G.-q.; Cai, W.-l. J. Chem. Soc. Perkin 1 1996, 2337-2338.
- 25. Ramnauth, J.; Lee-Ruff, E. Can. J. Chem. 1999, 77, 1245-1248.
- Davies, H. G.; Rahman, S. S.; Roberts, S. M.; Wakefield, B. J.; Winders, J. A. J. Chem. Soc. Perkin Trans. 1 1987, 85–89.
- 27. Canet, J.-L.; Fadel, A.; Salaün, J. J. Org. Chem. 1992, 57, 3463-3473.
- Unpublished work of Horman, R. E., cited in Eaton, P. E.; Yip, Y. C. J. Am. Chem. Soc. 1991, 113, 7692–7697.
- 29. Smith, L. I.; Hoehn, H. H. J. Am. Chem. Soc. 1939, 61, 2619-2624.
- Nguyen, M. T.; Ha, T.-K.; More O'Ferrall, R. A. J. Org. Chem. 1990, 55, 3251– 3256.
- 31. Jenny, E. F.; Roberts, J. D. J. Am. Chem. Soc. 1956, 78, 2005-2009.
- 32. Baldwin, J. E.; McDaniel, M. C. J. Am. Chem. Soc. 1968, 90, 6118-6124.
- Niwayama, S.; Kallel, E. A.; Sheu, C.; Houk, K. N. J. Org. Chem. 1996, 61, 2517– 2522.
- 34. Mayr, H.; Huisgen, R. J. Chem. Soc. Chem. Commun. 1976, 57-58.
- 35. Huisgen, R.; Mayr, H. J. Chem. Soc. Chem. Commun. 1976, 55-56.
- 36. Mayr, H. Angew. Chem. Int. Ed. 1975, 14, 500-501.
- 37. Schmidt, A. H.; Kircher, G.; Maus, S.; Bach, H. J. Org. Chem. 1996, 61, 2085-2094.
- 38. Peña-Cabrera, E.; Liebeskind, L. S. J. Org. Chem. 2002, 67, 1689–1691.
- Onofrey, T. J.; Gomez, D.; Winters, M.; Moore, H. W. J. Org. Chem. 1997, 62, 5658– 5659.
- 40. Perri, S. T.; Rice, P.; Moore, H. W. Org. Synth. 1990, 69, 220-225.
- 41. Tomooka, C. S.; Liu, H.; Moore, H. W. J. Org. Chem. 1996, 61, 6009-6012.
- 42. Koo, S.; Liebeskind, L. S. J. Am. Chem. Soc. 1995, 117, 3389-3404.
- 43. Shi, X.; Liebeskind, L. S. J. Org. Chem. 2000, 65, 1665-1661.
- 44. Trost, B. M.; Thiel, O. R.; Tsui, H.-C. J. Am. Chem. Soc. 2002, 124, 11616-11617.
- 45. Danheiser, R. L.; Casebier, D. S.; Huboux, A. H. J. Org. Chem. 1994, 59, 4844-4848.
- 46. Danheiser, R. L.; Casebier, D. S.; Firooznia, F. J. Org. Chem. 1995, 60, 8341-8350.
- 46a. Turnbull, P.; Moore, H. W. J. Org. Chem. 1995, 60, 644-649.
- 47. Yamamoto, Y.; Ohno, M.; Eguchi, S. J. Am. Chem. Soc. 1995, 117, 9653-9661.
- 46a. Ohno, M.; Yamamoto, Y.; Eguchi, S. Synlett 1998, 1167-1174.
- 48. Lee, K. H.; Moore, H. W. J. Org. Chem. 1995, 60, 735-738.
- Liu, H.; Tomooka, C. S.; Xu, S. L.; Yerxa, B. R.; Sullivan, R. W.; Moore, H. W. Org. Synth. 1998, 76, 189–198.

- Boullais, C.; Rannou, C.; Réveillere, E.; Mioskowski, C. *Eur. J. Org. Chem.* 2000, 723–727.
- 51. Verma, S. K.; Fleischer, E. B.; Moore, H. W. J. Org. Chem. 2000, 65, 8564-8573.
- 52. Winters, M. P.; Stranberg, M.; Moore, H. W. J. Org. Chem. 1994, 59, 7572-7574.
- 53. Xiong, Y.; Xia, H.; Moore, H. W. J. Org. Chem. 1995, 60, 6460-6467.
- 53a. Wipf, P.; Hopkins, C. R. J. Org. Chem. 1999, 64, 6881-6887.
- 54. Hergueta, A. R.; Moore, H. W. J. Org. Chem. 1999, 64, 5979-5983.
- 55. Birchler, A. G.; Liu, F.; Liebeskind, L. S. J. Org. Chem. 1994, 59, 7737-7745.
- 56. Zhang, S.; Liebeskind, L. S. J. Org. Chem. 1999, 64, 4042-4049.
- 57. Mingo, P.; Zhang, S.; Liebeskind, L. S. J. Org. Chem. 1999, 64, 2145-2148.
- 58. Zhang, D.; Llorente, I.; Liebeskind, L. J. Org. Chem. 1997, 62, 4330-4338.
- 59. Taing, M.; Moore, H. W. J. Org. Chem. 1996, 61, 329-340.
- 60. Sun, L.; Liebeskind, L. S. J. Org. Chem. 1995, 60, 8194-8203.
- 61. Sun, L.; Liebeskind, L. S. Tetrahedron Lett. 1997, 38, 3663-3666.
- 62. Sun, L.; Liebeskind, L. S. J. Am. Chem. Soc. 1996, 118, 12473-12474.
- 63. Yamamoto, Y.; Ohno, M.; Eguchi, S. Chem. Lett. 1995, 525-526.
- 64. Yamamoto, Y.; Ohno, M.; Eguchi, S. Bull. Chem. Soc. Jpn. 1996, 69, 1353-1362.
- 65. MacDougall, J. M.; Moore, H. W. J. Org. Chem. 1997, 62, 4554-4556.
- MacDougall, J. M.; Turnbull, P.; Verma, S. K.; Moore, H. W. J. Org. Chem. 1997, 62, 3792–3793.
- MacDougall, J. M.; Santora, V. J.; Verma, S. K.; Turnbull, P.; Hernandez, C. R.; Moore, H. W. J. Org. Chem. 1998, 63, 6905–6913.
- 68. MacDougall, J. M.; Moore, H. W. J. Org. Chem. 1999, 64, 7445-7450.
- 69. Turnbull, P.; Heileman, M. J.; Moore, H. W. J. Org. Chem. 1996, 61, 2584-2585.
- 70. Tiedemann, R.; Turnbull, P.; Moore, H. W. J. Org. Chem. 1999, 64, 4030-4031.
- 71. Heileman, M. J.; Moore, H. W. Tetrahedron Lett. 1998, 39, 3643-3646.
- 72. Ohno, M.; Yamamoto, Y.; Eguchi, S. Synlett 1998, 1167-1174.
- 73. Ohno, M.; Noda, M.; Yamamoto, Y.; Eguchi, S. J. Org. Chem. 1999, 64, 707-712.
- 74.
- 75. Petasis, N. A.; Fu, D.-K. Synlett 1996, 155-156.
- 76. Sturino, C. F.; Doussot, P.; Paquette, L. A. Tetrahedron 1997, 53, 8913-8926.
- 77. Heileman, M. J.; Tiedemann, R.; Moore, H. W. J. Am. Chem. Soc. 1998, 120, 3801-3802.
- Tiedemann, R.; Heileman, M. J.; Moore, H. W.; Schaumann, E. J. Org. Chem. 1999, 64, 2170–2171.
- 78a. Liu, F.; Liebeskind, L. S. J. Org. Chem. 1998, 63, 2835-2844.
- 79. Goudgaon, N. M.; Shi, J.; Schinazi, R. F. Tetrahedron Lett. 1998, 39, 1869-1872.
- 80. Smith, A. B. III; Kozmin, S. A.; Paone, D. V. J. Am. Chem. Soc. 1999, 121, 7423-7424.
- Smith, A. B. III; Adams, C. M.; Kozmin, S. A.; Paone, D. V. J. Am. Chem. Soc. 2001, 125, 5925–2537.
- Dudley, G. B.; Takaki, K. S.; Cha, D. D.; Danheiser, R. L. Org. Lett. 2000, 2, 3407– 3410.

## 3.4.2 Ketenes from Photolysis of Cycloalkanones and Enones

Norrish Type I cleavage of cycloalkanones forms diradicals that can form an aldehyde or a ketene by intramoleclar hydrogen transfer,^{1–7} and cyclohexanone (1) gave diradical 2 with low quantum efficiency that by hydrogen atom transfer formed *n*-butylketene (3), together with *n*-5-hexenal (4) in a ratio of 39:61 (equation 1).⁴ Cyclopentanones gave low yields of ketenes upon photolysis compared to cylohexanones.^{1–3} 3,5-Dimethylcyclohexanone (5) formed diradical 6, where the transition state 7 to form ketene 8 was favored over 9 for aldehyde formation due to steric interactions in the latter (equation 2).⁵



β-Silyl substituents have been proposed to enhance Norrish Type I cleavage of cyclic ketones **10** due to silicon stabilization of the resulting carbon radicals **11**, resulting in increased formation of ketenes **12**, which are captured by solvent (equation 4).^{8–11} In substrates **10** with R = tert-butyl, enhancement of ring cleavage was observed for the equatorial R₃Si group.¹¹



Bicyclic ketone 13 underwent hydrogen transfer resulting in efficient formation of ketene 15 from diradical 14 to the complete exclusion of aldehyde formation, which was hindered for steric reasons (equation 5).^{12,13} The camphor analogue 16 formed the diradical 17, which was similarly excluded from aldehyde formation, and gave ketene 18 captured as the ester 19 (equation 6).¹⁴ Verbinone (20) upon

photolysis gave ring opening to the diradical **21**, which formed the alkenylketene **22** identified by the IR absorption at  $2120 \text{ cm}^{-1}$  and by capture as the methyl ester **23** (equation 7).¹⁵



Photolysis of bicyclo[2.2.2]hexadienone **24** resulted in cycloreversion forming dimethylketene (**25**, equation 8).¹⁶ Dibenzo derivatives of **24**^{17–19} and a dihydro analogue²⁰ upon photolysis lead to several different types of ketenes.^{17–19} Photolysis of the bicyclic keto sulfide **26** in *t*-BuOH gave predominantly Norrish Type I cleavage forming the biradical **27**, in which hydrogen atom abstraction to give the ketene **28** is facile, followed by ester formation (equation 9).²¹



### **134** PREPARATION OF KETENES

Photolysis of the epoxy bicyclo[2.2.1]hetan-7-one **30** formed the diradical **31**, which gave ketene-aldehyde **32**, identified by IR absorption at 2117 cm⁻¹ (equation 10) and which was captured by MeOH forming ester **33**.²² The cyclic acetal **34** upon photolysis was proposed to react in an electrocyclic process forming the ketene **35**, which was captured with MeOH to give **36** in 73% yield (equation 11).²³



Photolysis of cyclopenten-4-one (**37**) gave a ketene intermediate with an IR band at  $2121-2126 \text{ cm}^{-1}$  in an argon matrix and  $2066-2210 \text{ cm}^{-1}$  in the gas phase, tentatively identified as **38a** or **38b** (equation 12).²⁴



Photolysis of 4,4-diphenylcyclopentenone **39** led to the ketene **41**, observed by IR in a process proposed to occur by di- $\pi$ -methane rearrangement to **41**, with capture of the ketene by methanol as the ester **42** (equation 13).²⁵ Photolysis of 5,5-diphenylcyclopent-2-en-1-one **43** gave cyclopropylketene **45**, identified by the IR absorption at 2110 cm⁻¹, and capture as the ester (equation 14).²⁶ Tetramethyleneketene **48** was identified by its IR absorption at 2100 cm⁻¹ as the product from photolysis of the cyclopentenyl ketone **46** via fragmentation of the diradical **47** (equation 15).²⁷ Methanol treatment gave the methyl ester in 25% yield.²⁷


Solid phase photolysis of crystalline cyclopropanocyclopentenone **49** led to ketene **50**, as observed by NMR, and shown by its capture with nucleophiles and its cyclization to **51** on long standing (equation 16).²⁸ Photolysis of **52** gave ketene **53**, as evidenced by an IR band at 2091 cm⁻¹ at -196 °C and cyclization to **54** (equation 17).²⁹ Photolysis of **55** formed **56**, captured by CH₃OH as the ester **57**, in a conversion assigned as a concerted [2 + 2 + 2 + 2] cycloreaction (equation 18).³⁰





Photolysis of **58** gave ketene **59**, which was characterized by an IR band at  $2105 \text{ cm}^{-1}$ , capture by diethylamine, and rearrangement to **60** with intramolecular MeO migration (equation 19).³¹



Photolysis of 5-methyl-1,5-hexadiene-3-one (**61**) gave cyclization to the diradical **62**, which opened to the 3-butenylketene **63** (the vinylogous Wolff rearrangement, Section 3.3.2), captured by MeOH as the ester **64** in 27% yield (equation 20).³² There was competitive formation of the bicyclo[2.1.1]hexan-2one **65** in 43% yield from **61**.³² The use of deuterium labeling showed that in the conversion of **61**-*d*₂, in addition to the 1,6-electrocyclization (equation 20), at  $\lambda$  313 nm there was also  $\alpha$ -cleavage leading to the radical pair **66**, which gave recombination to both **67** and **68** (equation 21).³³ Studies with triplet sensitizers indicated that  $\alpha$ -cleavage to **66** was a singlet process and that cyclization to **65** occurred from the triplet state of the ketone.³⁴



Photolysis of Z-dibenzoylethylene **69** in methanol is proposed to involve phenyl migration via **70**, forming ketene **71**, which was trapped with MeOH, giving the

ester **72** (equation 22).^{35–38} Photochemical rearrangement at 10 K of tetrabenzoylethylene **73**, forming **74**, was followed by bond rotation forming the conformer **75**, which cyclized to **76** (equation 23).³⁹



In a further study of the regioselectivity of the carbon-to-oxygen photo-phenyl migration in Z-1,2-dibenzoylethylenes to give alkenylketenes, it was found that photolysis of **77** in MeOH gave ketene **78** (equation 24), as evidenced by the ester **79** formed by MeOH trapping, while irradiation in the solid state gave ketene **80**, as evidenced by the IR band at  $2103 \text{ cm}^{-1}$  (equation 24) and trapping with MeOH.⁴⁰ Irradiation of **77** in acetone was proposed to give both **78** and **80** and the latter underwent further acetone-sensitized isomerization of the double bond to form the new ketene **81**, which cyclized to **82** (equation 25).⁴⁰





Photolysis of **83** in an Ar matrix led to the cyclopropenylketene **84**, as characterized by the IR band at  $2105 \text{ cm}^{-1}$ , and further photolysis resulted in decarbonylation and formation of the vinylallene **85** (equation 26).⁴¹ Photolysis of the pyrone **86** gave some formation of ketene **87**, as evidenced by an IR band at  $2120 \text{ cm}^{-1}$  (equation 27).⁴¹ Photolysis of **88** in an argon matrix gave a species tentatively identified as **89**, with an IR band at  $2130 \text{ cm}^{-1}$  (equation 28).⁴²



Photolysis of the 2,5-(*trans*-4-octenyl)-bridged p-benzoquinone **90** led to formation of the ketene **91**, detected by an IR band at  $2110 \text{ cm}^{-1}$  and capture as the ester **92** with MeOH (equation 29).⁴³



Photolysis of **93** gave a formal [4+2] cycloreversion to the ketene **94** with expulsion of formaldehyde and capture of the ketene by MeOH as the ester **95** (equation 30).⁴⁴



Photolysis of **96** was reported to form ketene **97**, which was trapped by Me₂NH, forming **98** along with other products (equation 31).⁴⁵ The photolysis of **99** was proposed to lead to a ketene intermediate **100** (equation 32) that cyclized, forming both a lactone and a spirocyclopentenone.⁴⁶



A possible mechanism for the photochemical rearrangement of lumiphotosantonin (101) to photosantonic acid (104) involved the ketene-carbene 102 (equation 33),⁴⁷ and upon photolysis at -190 °C the IR of ketene 103 at 2113 cm⁻¹ was observed.⁴⁸ The photolysis of umbellulone (105) at -190 °C gave rise to an IR band at 2113 cm⁻¹ attributed to the ketene 106 (equation 34), which was trapped with CH₃OH. A similar ketene-carbene route was suggested for this reaction.⁴⁸ Concerted processes forming 103 and 106 directly without the intermediacy of the ketene carbenes were also considered possible.⁴⁸



Photolysis of the cyclopentenone **107** in CH₃OH led to two methyl esters resulting from trapping of ketenes **108** and **109** in 7 and 34% yields, respectively (equation 35).⁴⁹ The structures for possible intermediates in these reaction pathways were suggested.⁴⁹



Photolysis of hydroxycyclopentenone **110** gave lactone **112**, which was proposed to form from (hydroxycyclopropyl)ketene **111** (equation 36).⁵⁰ The ketenes could not be trapped by  $CH_3OH$  and it was proposed that the cyclizations rates were even faster than solvent attack.



## **REFERENCES FOR SECTION 3.4.2**

- Wagner, P. J. In *Rearrangements in Ground and Excited States*: de Mayo, P., Ed.; Academic Press, New York, 1980, Vol. 3, pp. 381–444.
- 2. Yates, P. Pure Appl. Chem. 1968, 16, 93-113.
- 3. Weiss, D. S. Org. Photochem. 1981, 5, 347-420.
- 4. Coyle, J. D. J. Chem. Soc. B 1971, 1736-1740.
- 5. Wagner, P. J.; Spoerke, R. W. J. Am. Chem. Soc. 1969, 91, 4437-4440.
- 6. Agosta, W. C.; Schreiber, W. L. J. Am. Chem. Soc. 1971, 93, 3947-3952.
- Dalton, J. C.; Dawes, K.; Turro, N. J.; Weiss, D. S.; Barltrop, J. A.; Coyle, J. D. J. Am. Chem. Soc. 1971, 93, 7213–7221.
- Hwu, J. R.; Gilbert, B. A.; Lin, L. C.; Liaw, B. R. J. Chem. Soc. Chem. Commun. 1990, 161–163.
- 9. Hwu, J. R.; Patel, H. V. Synlett 1995, 989-996.
- Hwu, J. R.; Chen, B.-L.; Huang, L. W.; Tang, T.-H. J. Chem. Soc. Chem. Commun. 1995, 299–300.
- Hwu, J. R.; Chen, B.-L.; Lin, C.-F.; Murr, B. L. J. Organomet. Chem. 2003, 686, 198– 201.
- 12. Agosta, W. C.; Wolff, S. J. Am. Chem. Soc. 1976, 98, 4182-4188.
- 13. Agosta, W. C.; Wolff, S. J. Am. Chem. Soc. 1976, 98, 4316-4317.
- 14. Meinwald, J.; Schneider, R. A.; Thomas, A. F. J. Am. Chem. Soc. 1967, 89, 70-73.
- 15. Erman, W. F. J. Am. Chem. Soc. 1967, 89, 3828-3841.
- 16. Murray, R. K., Jr.; Hart, H. Tetrahedron Lett. 1968, 4995-4998.
- 17. Hart, H.; Dean, D. L.; Buchanan, D. N. J. Am. Chem. Soc. 1973, 95, 6294-6301.
- 18. Iwamura, H.; Yoshimura, K. J. Am. Chem. Soc. 1974, 96, 2652-2654.
- 19. Ripoll, J. L. Tetrahedron 1977, 33, 389-391.
- 20. Yang, M.; Lu, S.; Rao, C. P.; Tsai, Y.; Liao, C. J. Org. Chem. 2003, 68, 6543-6553.
- 21. Muraoka, O.; Zheng, B.-Z.; Nishimura, M.; Tanabe, G. J. Chem. Soc. Perkin 2, 1996, 2265–2270.
- 22. Ayral-Kaloustian, S.; Wolff, S.; Agosta, W. C. J. Org. Chem. 1978, 43, 3314–3319.
- 23. Padwa, A.; Carter, S. P.; Nimmesgern, H.; Stull, P. D. J. Am. Chem. Soc. 1988, 110, 2894–2900.
- 24. Aycard, J.-P.; Synaly, D.; Bodot, H. Spectrosc. Lett. 1997, 30, 1325–1333.
- 25. Zimmerman, H. E.; Little, R. D. J. Am. Chem. Soc. 1974, 96, 4623-4630.
- 26. Agosta, W. C.; Smith, A. B., III. J. Am. Chem. Soc. 1971, 93, 5513-5520.
- 27. Smith, A. B., III; Agosta, W. C. J. Am. Chem. Soc. 1973, 95, 1961-1968.
- 28. Zimmerman, H. E.; Sebek, P. J. Am. Chem. Soc. 1997, 119, 3677-3690.
- 29. Dannenberg, W.; Lemmer, D.; Perst, H. Tetrahedron Lett. 1974, 2133-2136.
- 30. Baldwin, J. E.; Krueger, S. M. J. Am. Chem. Soc. 1969, 91, 2396-2397.
- 31. Nitta, M.; Omata, A.; Namiki, M. Chem. Lett. 1982, 227-230.
- 32. Gibson, T. W.; Erman, W. F. J. Org. Chem. 1972, 37, 1148-1154.
- 33. Matlin, A. R.; Wolff, S.; Agosta, W. C. Tetrahedron Lett. 1983, 24, 2961–2964.

- 34. Dauben, W. G.; Cogen, J. M.; Ganzer, G. A.; Behar, V. J. Am. Chem. Soc. 1991, 113, 5817–5824.
- 35. Griffin, G. W.; O'Connell, E. J. J. Am. Chem. Soc. 1962, 84, 4148-4149.
- 36. Zimmerman, H. E.; Dürr, H. G.; Lewis, R. G.; Bram, S. J. Am. Chem. Soc. 1962, 84, 4149–4150.
- Zimmerman, H. E.; Dürr, H. G.; Lewis, R. G.; Givens, R. S. J. Am. Chem. Soc. 1967, 89, 1863–1874.
- 38. Barik, R.; Bhattacharyya, K.; Das, P. K.; George, M. V. J. Org. Chem. 1986, 51, 3420-3428.
- 39. Rubin, M. B.; Sander, W. W. Tetrahedron Lett. 1987, 28, 5137-5140.
- 40. Maji, D.; Singh, R.; Mostafa, G.; Ray, S.; Lahiri, S. J. Org. Chem. 1996, 61, 5165-5168.
- 41. Maier, G.; Reisenauer, H. P. Chem. Ber. 1981, 114, 3916-3921.
- 42. LeBlanc, B. F.; Sheridan, R. S. J. Am. Chem. Soc. 1988, 110, 7250-7252.
- 43. Miyake, M.; Tsuji, T.; Furusaki, A.; Nishida, S. Chem. Lett. 1988, 47-50.
- 44. Er, E.; Margaretha, P. Helv. Chim. Acta 1994, 77, 904-908.
- Takahashi, K.; Mikami, E.; Takase, K.; Takahashi, T. *Tetrahedron Lett.* **1988**, *29*, 5933– 5936.
- 46. Paquette, L. A.; Gallou, F.; Zhao, Z.; Young, D. G.; Liu, J.; Yang, J.; Friedrich, D. J. Am. *Chem. Soc.* **2000**, *122*, 9610–9620.
- 47. van Tamelen, E. E.; Levin, S. H.; Brenner, G.; Wolinsky, J.; Aldrich, P. E. J. Am. Chem Soc. **1959**, *81*, 1666–1678.
- 48. Barber, L.; Chapman, O. L.; Lassila, J. D. J. Am. Chem. Soc. 1968, 90, 5933-5934.
- 49. Ohkita, M.; Nishida, S.; Tsuji, T. J. Am. Chem. Soc. 1999, 121, 4589-4597.
- 50. Gowda, G.; McMurry, T. B. H. J. Chem. Soc. Perkin Trans. 1 1979, 274-278.

# 3.4.3 Ketenes from Cyclohexadienones and Other Cycloalkenones

The photochemical ring opening of cyclohexadienones (Barton-Quinkert reaction) was first reported in  $1960^1$  and provides a facile synthesis of complex molecules from readily available starting materials, even with the use of incandescent light.^{2–6} Cyclohexadienones 1 absorb light at long wavelengths and undergo reversible ring opening to dienylketenes 2 and 3, which can be observed spectroscopically and intercepted by nucleophiles such as alcohols or amines forming 4 (equations 1, 2).⁶ Photochemical interconversion of the dienylketenes 2 and 3 may also occur.





The energetics of the interconversion of phenol (5), 2,4-cyclohexadienone (6), and Z- and E-1,3,5-cyclohexatrien-1-one (7) have been studied computationally⁷⁻¹¹ at the B3LYP/6-31G//B3LYP/6-31G level (Scheme 3.4).¹¹



Scheme 3.4 Photoisomers of phenol and relative energies (kcal/mol).

Computations at the (U)B3LYP/6–31G* level of the interconversion of cyclohexadienone and dienylketenes through ring opening and closure indicate that whereas the parent cyclohexadienone **6** has a high barrier for forming the less stable dienylketene Z-**7** (equation 3), conversion of **8** to phenylvinylketene **9** is significantly more facile due to the gain in aromaticity (equation 4).^{9,10} Cyclization of Z-**7** to the ketenylidenecyclopentenes **10** has also been examined.¹¹



The reaction of benzene with oxygen atoms produced from ozone photolysis in an argon matrix at 12 K produces phenol (5), benzene oxide (11), 2,4-cyclohexadienone (6), and the ketene *E*- 7 (equation 5).¹¹ The ketene was identified by a series of IR bands from 2116 to 2133 cm⁻¹, and the conformation shown of the *E*isomer was calculated to be the most stable. It was assumed that the initially formed *Z*-7 would be photoisomerized to the more stable *E*-7.¹¹ The relative energies of phenol, **6**, and ketenes Z-**7** and E-**7** were calculated as 0, 18.9, 36.8 and 35.4 kcal/ mol Scheme 3.4.¹¹ Similar oxidations were carried out with toluene, *m*- and *p*-xylene and mesitylene and ketene products were observed by IR.¹² The structures of these were assigned based on comparison of the observed with the calculated IR frequencies, including the examination of ¹⁸O isotopomers.¹²



Photolysis of benzene oxide  $(11)^{13}$  or the diketone  $12^{14}$  formed the parent cyclohexadienone **6** and this was converted to dienylketene Z-**7** on further photolysis (equation 6).^{13,14} The ketene Z-**7** was identified by its IR absorption at 2112 cm⁻¹, the UV spectrum, and capture by EtOH.^{13,14} The substituted dienylketene **13** was generated by photolysis and trapped with MeOD, forming the ester (equation 7).¹⁵



Dienylketene **15** from photolysis of cyclohexadienone **14** cyclized to bicyclo[3.1.0]hex-3-en-2-ones **16**, which underwent thermal ring opening back to **15** and **14** (equation 8).¹⁶ Some isomeric analogues were also studied.¹⁶ The formation of **19** from photolysis of **17** was similarly proposed to involve the intermediacy of the dienylketene **18** (equation 9).¹⁷ Photolysis of **20** gave ketene **21** (IR 2100 cm⁻¹), which thermally reformed **20** together with **22** or was captured by amines, which were much more efficient in this process than alcohols (equation 10).¹⁸





1-Azidocycloheptatrienone **23** upon photolysis formed (4-cyanobutadienyl)ketene **24**, which cyclized to 6-cyanocyclohexadienone **25** (equation 11).^{19,20} The ketene **24** was identified by its IR absorption at  $2120 \text{ cm}^{-1}$ , the UV spectrum, and capture by nucleophilic solvents. The half-life of **24** was 4 min at 22 °C and the major product was **26**, derived by acylation by **24** of the phenol from **25** (equation 11).^{19,20} The 2,4,6-trimethyl analogue of **23** gave the corresponding dienylketene, which cyclized faster than **24** by a rate factor estimated as  $10^{3}$ .^{19,20}



Laser photolysis of **27a** and of **27b** with 266 nm light gave transient species assigned as the cyclohexadienones **28a,b** (equation 12).²¹ Photolyses of these transients with a 308 nm laser gave new transient species with absorption maxima at 270 nm and 330 nm, assigned as the dienylketenes **29a** and **29b**, respectively.²¹ The longer-wavelength absorption of **29b** was attributed to conjugation of the acetyl substituent with the dienylketene.²¹



Photolysis of the cyclohexadienone **30** (R = H) with visible light in the presence of amines was proposed to form dienylketenes **31**, which then gave amides **32** in yields of 87–95% as *cis/trans* mixtures that were not separated (equation 13).²² In a competition experiment, photolysis of **30** in the presence of *n*-BuNH₂ (1 equiv) and EtOH (5 equiv) gave amide and ester products in a 9.5:1 ratio, showing the much greater nucleophilicity of the amine. It was suggested that dienylketenes generated in this way would be useful for reactions with amine functions in enzymes.²² Photolysis of **30** (R = SPh) with visible light in the presence of amines also gave amide products **32**.²³



Photolysis of the water-soluble cyclohexadienones **33** in the presence of amino acids in aqueous solution was carried out at pH 10 so that free amino groups were present and resulted in selective capture of the ketenes **34** by the nucleophiles (equation 14).^{24,25} Reaction of **34** with diamines gave the bisamides from reaction of two equivalents of the ketene.²⁶



Photolysis of chiral cyclohexadienone **35** gave the dienylketenes **36** ( $\lambda_{max}$  305 nm,  $\varepsilon$  5000), which either reverted to **35** or reacted with imine **37** in [2+2] cycloaddition reactions, forming  $\beta$ -lactam **38** (equation 15), or with nucleophiles (RNH₂, MeOH, EtOH), forming amide **41** (equation 16) or esters.²⁷ Two IR bands at 2102 and 2114 cm⁻¹ were observed, suggesting that both the *E* and *Z* isomers of the ketene **36** were present. The ¹H NMR spectra of these species were observed and capture products from both were obtained. With excess cyclohexylamine there is a pseudo first order reaction of **36** assigned to formation of a zwitterionic intermediate **39** ( $\lambda_{max}$  380 nm), which formed amide enol **40** and the amide **41** ( $\lambda_{max}$  460 nm) (equation 16).²⁷





The ketone **42** underwent photochemical conversion to the dienylketene **43** as two conformational isomers, with IR absorptions at 2104 and 2118 cm⁻¹, respectively (equation 17).²⁸ Reaction of **43** with cyclohexylamine was proposed to form a zwitterionic intermediate **44**, which was converted to an enol amide **45** that formed amide **46** by a cyclohexylamine-assisted process (equations 17, 18) and **48** by an intramolecular reaction (equation 19).²⁸











Photolysis of azidoalkylcyclohexendione **49** with 366 nm light gave dienylketene **50**, as evidenced by capture by MeOH as the ester **51** (equation 20).²⁹



Gas phase chemical ionization of protonated anisole formed ionized phenol radical cation ( $5^+$ ), but rearrangement of this to the radical cations of cyclohexadienone **6** and of butadienylketene **7** was not observed (equation 21).³⁰ For the radical cation species the relative energies of phenol, cyclohexadiene and dienylketene *E*-**7** are calculated to be 0, 35 and 29 kcal/mol, respectively. Thus, the relative stabilities of cyclohexadienone and the dienylketene are reversed between the neutral species (Scheme 3.4) and the radical cation.

$$\overset{\bullet}{\longrightarrow} \overset{\bullet}{\longrightarrow} \overset{\bullet}{\to} \overset{\bullet$$

Laser photolysis of 3,5-di-*tert*-butyl-1,2-benzoquinone **52** in hexane, carbon tetrachloride, or acetonitrile gave rise to bisketene **53**, which was identified by its IR absorption at 2173 cm⁻¹ and a UV absorption near 320 nm (equation 22).^{31,32} Upon continued photolysis **53** underwent decarbonylation to give the cyclopentadienone **54** (equation 22). Kinetic analysis led to values of  $5 \times 10^{-4}$  s⁻¹ for the rate constants *k* and  $k_2$  and of  $2 \times 10^{-5}$  s⁻¹ for  $k_{-1}$ . The addition of EtOH after 30 min photolysis led to immediate disappearance of **53** and formation of ester **55a** in the presence of a small amount of EtOH and of ester **55b** if a large amount of EtOH was added (equation 23). The formation of **54** from **53** may involve initial decarbonylation or possibly a zwitterion **56**.





Photolysis of 2,3,5,6-tetra(trimethylsilyl)-1,4-benzoquinone (**57**) gave the ketene **59** (IR 2089 cm⁻¹) as pale yellow crystals in 21% yield (equation 24).³³ A zwitterionic intermediate **58** may have been formed.



Photolysis of 6-imino-2,4-cyclohexadien-1-one (**60**) in an Ar matrix at 10 K in solid Ar gave the ketene **63**, identified by the ketenyl and keteniminyl IR absorptions at 2114 and 2031 cm⁻¹, respectively, which agreed with calculated absorption positions (equation 25).³⁴ Photolysis of the azidophenol **61** led to **62**, which gave a hydrogen shift to also form **60** (equation 25).^{34,35}



Thermolysis of *o*-quinone methide **64** at 760 to 1110 K gave benzene and fulvene and while the formation of the intermediate **65** with possible ketene character in this process was considered, a concerted pathway was favored (equation 26).³⁶



Photolysis of 2,4,6-cyclooctatrienone **66** gave the hexatrienylketene **67**, identified by its capture with  $CH_3OH^{37}$  and by the IR band at 2113 cm⁻¹ (equation 27).³⁸ The alkynyl-substituted cyclohexadienone **68** in MeOH gave the diene-yne ketene **69**, which was trapped as the methyl ester from cyclization of **70** (equation 28).³⁹



The gas phase collisionally activated dissociation (CAD) of *O*-deprotonated resorcinol and 2-, 4- and 5-methylresorcinols gave fragment ions consistent with O to C proton migration and ring opening to alkenylketene intermediates such as **71** that rearranged to the ketene **72**, which cleaved further to form species such as vinylketene (**73**) and the ynolate **74** (equations 29, 30).^{40,41}



The photosolvolysis of the 6-acetoxycyclohexa-2,4-dienone **75** was proposed to involve formation of the benzene oxide **76**, which opened to the oxepin **77** (equation 31), which gave the ketene **78** leading to the final product, **79** (equation 32).⁴²



Photolysis of bridged cyclohept-2,4-dien-1-ones gave cyclopropylketenes identified by their IR bands in the range  $2112-2120 \text{ cm}^{-1}$  and their ¹H NMR signals in the range  $\delta$  2.62–3.33 at –80 °C in CD₂Cl₂ for the proton at C₂. Upon reaction with MeOH, these gave the corresponding ester in yields of 70–100%.⁴³

Many examples of cyclization of metal-complexed dienylketenes are given in Section 3.5. See also Section 4.1.4

#### **REFERENCES FOR SECTION 3.4.3**

- 1. Barton, D. H. R.; Quinkert, G. J. Chem. Soc. 1960, 1-9.
- 2. Quinkert, G. Angew. Chem. Int. Ed. 1965, 4, 211-222.
- 3. Quinkert, G. Angew. Chem. Int. Ed. 1975, 14, 790-800.
- 4. Waring, A. J. Adv. Alicyclic Chem. 1966, 1, 129-256.
- Schultz, A. G. In CRC Handbook of Organic Photochemistry and Photobiology; Horspool, W. M.; Song, P.-S., Eds.; CRC Press, Boca Raton, FL, 1995, pp. 728– 736.
- Quinkert, G.; Kleiner, E.; Freitag, B.-J.; Glenneberg, J.; Billhardt, U.-M.; Cech, F.; Schmieder, K. R.; Schudok, C.; Steinmetzer, H-C.; Bats, J. W.; Zimmermann, G.; Dürner, G.; Rehm, D.; Paulus, E. F. *Helv. Chim. Acta* 1986, 69, 469–537.
- 7. Zhu, L.; Bozzelli, J. W. J. Phys. Chem. A 2003, 107, 3696-3703.
- 8. Zora, M.; Sahpaz, F.; Ozarslan, E. J. Mol. Struct. 2002, 589-590, 111-123.
- 9. Zora, M. J. Org. Chem. 2004, 69, 1940-1947.
- Allen, A. D.; Fenwick, M. F.; Givehchi, B.; Henry-Riyad, H.; Nikolaev, V. A.; Shikova, E. A.; Tahmassebi, D.; Tidwell, T. T.; Wang, S. J. Org. Chem. 2001, 66, 5759–5765.
- 11. Parker, J. K.; Davis, S. R. J. Am. Chem. Soc. 1999, 121, 4271-4277.
- 12. Parker, J. K.; Davis, S. R. J. Phys. Chem. A 2000, 104, 4108-4114.
- Jerina, D. M.; Witkop, B.; McIntosh, C. L.; Chapman, O. L. J. Am. Chem. Soc. 1974, 96, 5578–5580.

### **152** PREPARATION OF KETENES

- 14. Capponi, M.; Gut, I.; Wirz, J. Angew. Chem. Int. Ed. Engl. 1986, 25, 344-345.
- 15. Baldwin, J. E.; McDaniel, M. C. J. Am. Chem. Soc. 1968, 90, 6118-6124.
- 16. Dannenberg, W.; Perst, H. Liebigs Ann. 1975, 1873-1894.
- Swenton, J. S.; Saurborn, E.; Srinivasan, R.; Sonntag, F. I. J. Am. Chem. Soc. 1968, 90, 2990–2991.
- 18. Griffiths, J.; Hart, H. J. Am. Chem. Soc. 1968, 90, 3297-3298.
- 19. Hobson, J. D.; Al Holly, M. M.; Malpass, J. R. Chem. Commun. 1968, 764-766.
- 20. Hobson, J. D.; Malpass, J. R. J. Chem. Soc. C 1967, 1645-1648.
- 21. Jiménez, M. C.; Miranda, M. A.; Scaiano, J. C.; Tormos, R. Chem. Commun. 1997, 1487–1488.
- 22. Barton, D. H. R.; Chung, S. K.; Kwon, T. W. Tetrahedron Lett. 1996, 37, 3631–3634.
- 23. Kwon, T. W.; Kim, Y. M.; Song, S. J.; Kwon, Y. U.; Chung, S. K. Biorg. Med. Chem. Lett. 2000, 10, 1551–1553.
- 24. Barton, D. H. R.; Kwon, T.; Taylor, D. K.; Tajbakhsh, M. Biorg. Med. Chem. 1995, 3, 79–84.
- 25. Kim, Y. M.; Kwon, T. W.; Chung, S. K.; Barton, D. H. R. *Biorg. Med. Chem. Lett.* **1999**, 9, 1175–1178.
- 26. Kim, Y. M.; Kwon, T. W.; Chung, S. K. Molecules 2000, 5, 961–966.
- Quinkert, G.; Scherer, S.; Reichert, D.; Nestler, H.-P.; Wennemers, H.; Ebel, A.; Urbahns, K.; Wagner, K.; Michaelis, K.-P.; Wiech, G.; Prescher, G.; Bronstert, B.; Freitag, B.-J.; Wicke, I.; Lisch, D.; Belik, P.; Crecelius, T.; Hörstermann, D.; Zimmermann, G.; Bats, J. W.; Dürner, G.; Rehm, D. *Helv. Chim. Acta* 1997, *80*, 1683–1772.
- 28. Quinkert, G.; Englert, H.; Cech, F.; Stegk A.; Haupt, E.; Leibfritz, D.; Rehm, D. *Chem. Ber.* **1979**, *112*, 310–348.
- 29. Schultz, A. G.; Staib, R. R.; Eng, K. K. J. Org. Chem. 1987, 52, 2968-2972.
- 30. Le, H. T.; Flammang, R.; Gerbaux, P.; Bouchoux, G.; Nguyen, M. T. J. Phys. Chem. A **2001**, *105*, 11582–11592.
- 31. Alkenings, B.; Bettermann, H.; Dasting, I.; Schroers, H.-J. *Spectrochim Acta* **1993**, *49A*, 315–320.
- 32. Alkenings, B.; Bettermann, H.; Dasting, I.; Schroers, H. J. Vib. Spectrosc. 1993, 5, 43–49.
- Tsutsui, S.; Sakamoto, K.; Ebata, K.; Kabuto, C.; Sakurai, H. Bull. Chem. Soc. Jpn. 2002, 75, 2571–2577.
- 34. Morawietz, J.; Sander, W. Liebigs Ann. 1996, 2029-2037.
- 35. Tomioka, H.; Matsushita, T.; Murata, S.; Koseki, S. Liebigs Ann. 1996, 1971–1980.
- 36. Dorrestijn, E.; Pugin, R.; Nogales, M. V. C.; Mulder, P. J. Org. Chem. 1997, 62, 4804–4810.
- 37. Buchi, G.; Burgess, E. M. J. Am. Chem. Soc. 1962, 84, 3104-3109.
- 38. Barber, L. L.; Chapman, O. L.; Lassila, J. D. J. Am. Chem. Soc. 1969, 91, 531-533.
- Fernández-Zertuche, M.; Hernández-Lamoneda, R.; Ramírez-Solís, A. J. Org. Chem. 2000, 65, 5207–5211.
- Binkley, R. W.; Dillow, G. W.; Flechtner, T. W.; Winnik, W.; Tevesz, M. J. S. Org. Mass. Spectrom. 1994, 29, 491–495.

- 41. Flechtner, T. W.; Winnik, B.; Winnik, W.; Tevesz, M. J. S. J. Mass. Spectrom. 1996, 31, 377–382.
- 42. Eley, K. L; Crowley, P. J.; Bugg, T. D. H. J. Org. Chem. 2001, 66, 2091-2097.
- 43. Chapman, O. L.; Kane, M.; Lassila,, J. D.; Loeschen, R. L.; Wright, H. E. J. Am. Chem. Soc. **1969**, *91*, 6856–6858.

# 3.4.4 Ketenes from Dioxinones

The chemistry of dioxinone 1, a convenient precursor for acetylketene 2, has been the subject of a short review.¹ Dioxinone 1 is a product of the reaction of diketene and acetone, and forms acetylketene (2) and acetone on thermolysis (equation 1). Even though the preparation of diketene in acetone was used on a large scale in industry, the formation of 1 was not recognized for a long time. The structure of 1 was elucidated in a collaboration involving Michael F. Carroll of the A. Boake Roberts Company and Alfred Bader of Aldrich Chemical Company using UV and IR analysis and later NMR,^{2,3} and was one of the early offerings of the Aldrich Company.



Dioxinones cleave on thermolysis or photolysis, with formation of an acylketene and a ketone.⁴ Pyrolysis of **1** at 180–240 °C and trapping of the product in an argon matrix at 5–12 K showed the presence of **2** and acetone by IR (equation 2).⁵ Ketenyl bands at 2135 and 2142 cm⁻¹ suggested the presence of both the *s*-*cis* and *s*-*trans* conformations of **2**.⁵ Dimerization of **2** proceeded by a [4 + 2] cycloaddition to form **3** (equation 2).⁵ The stereochemistry of dioxinone formation has been studied by computational methods.⁶



### **154** PREPARATION OF KETENES

A variety of substituted acylketenes have been prepared from dioxinones, including halo derivatives **5** from 5-halodioxinones **4**, and these were trapped with dimethylcyanamide to give halooxazinones **6** (equation 3).^{7,8} It was noted that the chloroketene **5b** formed more readily than the fluoro compound, and this is consistent with the known destabilization of ketenes by fluorine (Section 4.4.1). Trifluoromethyldioxinones **7** gave the  $\alpha$ -trifluoromethylketenes **8**, which were trapped by benzyl alcohol (equation 4).⁹



Dioxinones **10** with side chains bearing nucleophilic groups form acylketenes **11**, which cyclize by intramolecular attack (equation 5).¹⁰ This reaction was also used to prepare eight-membered rings¹¹ and other medium ring-sized lactones.^{12,13}



Treatment of various Meldrum acid derivatives **13** with diazomethane gave 6methoxydioxinones **14**, which on thermolysis gave acylketenes **15** at 20–50 °C, as shown by trapping with *t*-BuOH and by observation of the characteristic IR absorptions at 2130–2149 cm⁻¹ (equation 6).¹⁴ These ketenes underwent gradual dimerization. Silylation of **17** gave **18**, which upon heating at 50 °C without solvent gave the persistent ketene **19**, with IR absorption at 2125 cm⁻¹ (equation 7).¹⁴





Thermolysis of acyldioxinone **20**, forming diacetylketene **21** in the presence of phenylisocyanate, gave [4 + 2] cycloaddition forming **22** (equation 8).¹⁵



Photolysis of the dioxinone 23 with 248 nm light provided a convenient source of the ketene 24 for measurement of the rate of hydration (equation 9).¹⁶ Thermolysis of 25 gave the ketene 26, which was hydrolyzed to the carboxylic acid that lost CO₂, forming 27 (equation 10).¹⁷ The ketene was also trapped with *t*-BuOH.¹⁷



## **REFERENCES FOR SECTION 3.4.4**

- Clemens, R. J. In *Encyclopedia of Reagents for Organic Synthesis*; Paquette, L. A., Ed.; Wiley, New York, 1995, pp. 5205–5207.
- 2. Carroll, M. F.; Bader, A. R. J. Am. Chem. Soc. 1952, 74, 6305; 1953, 75, 5400-5402.
- 3. Bader, A. R.; Gutowsky, H. S.; Heeschen, J. P. J. Org. Chem. 1956, 21, 821-822.
- 4. Kaneko, C.; Sato, M.; Sakaki, J.; Abe, Y. J. Heterocyclic Chem. 1990, 27, 25-30.
- 5. Clemens, R. J.; Witzeman, J. S. J. Am. Chem. Soc. 1989, 111, 2186-2193.

- Shumway, W.; Ham, S.; Moer, J.; Whittlesey, B. R.; Birney, D. M. J. Org. Chem. 2000, 65, 7731–7739.
- 7. Sato, M.; Kaneko, C.; Iwaoka, T.; Kobayashi, Y.; Iida, T. *J. Chem. Soc., Chem. Commun.* **1991**, 699–700.
- 8. Iwaoka, T.; Murohashi, T.; Sato, M.; Kaneko, C. Synthesis, 1992, 977-981.
- 9. Iwaoka, T.; Sato, M.; Kaneko, C. J. Chem. Soc. Chem. Commun. 1991, 1241-1242.
- 10. Boeckman, R. K., Jr.; Pruitt, J. R. J. Am. Chem. Soc. 1989, 111, 8286-8288.
- 11. Petasis, N. A.; Patane, M. A. J. Chem. Soc. Chem. Commun. 1990, 836-837.
- 12. Sakaki, J.; Sugita., Y.; Sato, M.; Kaneko., C. Tetrahedron 1991, 47, 6197-6214.
- Sato, M.; Sakaki, J.; Sugita., Y.; Yasuda, S.; Sakoda, H.; Kaneko, C. *Tetrahedron* 1991, 47, 5689–5708.
- 14. Sato, M.; Ban, H.; Kaneko, C. Tetrahedron Lett. 1997, 38, 6689-6692.
- 15. Little, R. D.; Russu, W. A. J. Org. Chem. 2000, 65, 896-8099.
- Chang, J. A.; Kresge, A. J.; Nikolaev, V. A.; Popik, V. V. J. Am. Chem. Soc. 2003, 125, 6478–6784.
- 17. Snyder, C. A.; Selegue, J. P.; Dosunmu, E.; Tice, N. C.; Parkin, S. J. Org. Chem. 2003, 68, 7455–7459.

# 3.4.5 Ketenes by Thermolysis of Alkynyl Ethers

Thermolysis of alkynyl ethers provides a mild route for the preparation of a variety of alkynyl ethers. This procedure was discovered in 1954 by Ficini, who found that ethyl alkynyl ethers **1** on thermolysis underwent loss of ethylene to form active acylating agents.¹ These were originally proposed to be alkynyl alcohols,¹ but on further investigation they were shown to be ketenes **3**.^{2–4} On the basis of deuterium isotope effect studies, the reactivity order *t*-BuO > *i*-PrO > EtO for alkoxy ethynyl ethers,⁵ and other studies, the mechanism of the reaction is considered to involve concerted hydrogen atom transfer through a 6-membered ring transition state **2** (equation 1).⁵

*tert*-Butyl alkynyl ethers **4** underwent ketene formation at temperatures as low as 40 °C⁶⁻⁸ and provided a route to ketene **5** itself, captured with amines (equation 2).⁶⁻¹⁰ Thermally stable ketenes such as trimethylsilylketene **7** have been prepared and isolated by this procedure (equation 3),^{7,7a} or the ketenes were trapped by reactions with nucleophiles or in [2 + 2] cycloadditions. Reaction of the ketene with the alkynyl ether precursor was often facile,⁸ and this represents a severe limitation to this method, but if as in the case of silylketenes the products are stable

neat, the reactions can be conducted using preparative gas chromatography and the products collected from the outlet after reaction in the chromatograph.

$$H \xrightarrow{CH_3}_{\substack{I \\ CH_3}} \xrightarrow{40 \circ C}_{-CH_2 = C(CH_3)_2} \xrightarrow{H}_{H} C = O \xrightarrow{RR^1 NH}_{CH_3} \xrightarrow{O}_{NRR^1} (2)$$

$$4 \qquad 5 \qquad (87-99\%)$$

$$Me_3Si \xrightarrow{OCH_2CH_3} \xrightarrow{A}_{-CH_2 = CH_2} \xrightarrow{Me_3Si}_{C=O} C = O \qquad (3)$$

This method has been used for the preparation of acetylketene (9) by the thermolysis of ethyl 3-oxo-1-butynyl ether (8), which formed the dimer in 77% yield (equation 4).¹¹ Vinylketene (11) was obtained by the thermolysis of but-3-en-1-ynyl ether (11) and was observed in the gas phase (equation 5).¹² Acylketenes with substituents Me₃Si, Me₃Ge, and Ph₂P were also prepared by alkynyl ether thermolysis.^{12a}



Reaction of alkynyl ether **12** with trimethylsilyl iodide gave trimethylsilylketene **13** (equation 6).^{13,14} The thermal rearrangement of trimethylsilyl alkynyl ether **14**, forming ketenes **16**, has been the subject of a computational study, and a mechanism with a concerted 1,3-silyl shift through the four-membered ring transition state **15** was presented, with retention of configuation at silicon (equation 7).¹⁵



Ketene 18 formed by thermolysis of the alkynyl ether 17 with a functionalized side chain bearing a pendant hydroxyl group cyclized intramolecularly with

formation of macrocyclic lactone **19** (equation 8).¹⁶ Tri-*n*-butylamine (10 equiv) was added to facilitate cyclization.



Photolysis of cyclohexyl ethynyl ether (**20**) in methanol gave a mixture of cyclohexanol (30–40%) and 32% methyl cyclohexyl acetate (**23**).¹⁷ The corresponding reaction of the alkynyl ether from (+)-2-octanol gave 40% racemization in formation of the rearranged ester, and a mechanism involving the intermediacy of a radial pair **21** that formed cyclohexylketene (**22**) was proposed (equation 9).¹⁷



Gas phase thermolysis of ethoxyalkynylsilanes has proven useful for the preparation of a variety of silylated ketenes and polyketenes,^{18–20} including **25** from the dialkynyldisilane **24** (equation 10).²⁰



Thermolysis of enediynyl ethyl ether **26** gave ene-yne ketene **27**, which cyclized to diradical intermediate **28** (equation 11), which formed **29a–d** (equation 12).²¹ Other routes to ene-yne ketenes are discussed in Section 3.4.1, and use of an alky-nyl ether to form an alkynylketene is considered in Section 4.1.3.





Mass spectrometric analysis of aryl ethoxy, *i*-propoxy, and *t*-butoxycyclopropenones (**30**) led to loss of the elements of CO and alkenes and the observation of ions with ratios m/z corresponding to the corresponding hydroxyalkynes or the isomeric arylketenes **32** formed from intermediate ethynyl ethers **31** (equation 13).²²



# **REFERENCES FOR SECTION 3.4.5**

- 1. Ficini, J. Bull. Soc. Chim. Fr. 1954, 1367-1371.
- 2. Nieuwenhuis, J.; Arens, J. F. Recl. Trav. Chim. Pays Bas, 1958, 77, 761-768.
- Van Daalen, J. J.; Kraak, A.; Arens, J. F. Recl. Trav. Chim. Pays Bas 1961, 80, 810– 818.
- Brandsma, L.; Bos, H. J. T.; Arens, J. F. In *Chemisty of the Acetylenes*; Viehe, H. G., Ed.; Marcel Dekker, New York, 1969, pp. 808–810.
- Moyano, A.; Pericás, M. A.; Serratosa, F.; Valentí, E. J. Org. Chem. 1987, 52, 5532– 5538.
- 6. Valentí, E.; Pericas, M. A.; Serratosa, F.; Mana, D. J. Chem. Res. (S) 1990, 118-119.
- 7. Valentí, E.; Pericás, M. A.; Serratosa, F. J. Org. Chem. 1990, 55, 395-397.
- 7a. Ruden, R. A. J. Org. Chem. 1974, 24, 3607-3608.
- 8. Serratosa, F. Acc. Chem. Res. 1983, 16, 170-176.
- 9. Pericas, M. A.; Serratosa, F.; Valentî, E. Synthesis 1985, 1118-1121.
- 10. Pericas, M. A.; Serratosa, F.; Valentî, E. Tetrahedron 1987, 43, 2311-2316.
- 11. Hyatt, J. A.; Feldman, P. L.; Clemens, R. J. J. Org. Chem. 1984, 49, 5105-5108.
- 12. Terlouw, J. K.; Burgers, P. C.; Holmes, J. L. J. Am. Chem. Soc. 1979, 101, 225-226.
- 12a. Lukashev, N. V.; Fil'chikov, A. A.; Kazankova, M. A.; Beletskaya, I. P. *Heteroatom. Chem.* **1993**, *4*, 403–407.
- 13. Sakurai, H.; Shirahata, A.; Sasaki, K.; Hosomi, A. Synthesis 1979, 740-741.
- Efimova, I. V.; Kazanokova, M. A.; Lutsenko, I. F. Zh. Obshch. Khim. 1985, 55, 1647– 1649. Engl. Transl. 1985, 55, 1465–1466.
- Oblin, M.; Fotiadu, F.; Rajzmann, M.; Pons, J.-M. J. Chem. Soc. Perkin Trans. 2, 1997, 1621–1623.
- 16. Liang, L.; Ramaseshan, M.; MaGee, D. I. Tetrahedron 1993, 49, 2159-2168.

- 17. Smith, B. A.; Callinan, A. J.; Swenton, J. S. Tetrahedron Lett. 1994, 35, 2283-2286.
- 18. Sung, K.; Tidwell, T. T. J. Am. Chem. Soc. 1996, 118, 2768-2769.
- 19. Liu, R.; Marra, R. M.; Tidwell, T. T. J. Org. Chem. 1996, 61, 6227-6232.
- 20. Sung, K.; Tidwell, T. T. Organometallics 1997, 16, 78-85.
- 21. Tarli, A.; Wang, K. K. J. Org. Chem. 1997, 62, 8841-8847.
- 22. Weidner, C. H.; Wadsworth, D. H.; Knop, C. S.; Oyefesso, A. I.; Hafer, B. L.; Hartman, R. J.; Mehlenbacher, R. C.; Hogan, S. C. *J. Org. Chem.* **1994**, *59*, 4319–4322.

# 3.4.6 Ketenes from Other Thermolytic and Photochemical Routes

Ketenes are formed by a variety of other thermal and photochemical methods of unique character, and a selection of these are included in this section. Ketene itself is prepared by pyrolysis of acetone (Section 4.1). Thermal ring opening of (-)-chrysanthenone (1) caused racemization upon heating to 81 °C, and the intermediacy of the ketene **2** was inferred from the capture as the ester **3** upon refluxing **1** in MeOH (equation 1).^{1–4} This example is a special case of cyclobutanone ring opening (Section 3.4.1).



The tricyclic ketone **4** opened the cyclopropane ring above 50 °C, giving cyclobutenylketene **5**, as evidenced by trapping experiments (equation 2).⁵ Irradiation of **6a** gave diketone **6** at room temperature, which showed an IR absorption attributed to ketene **7** present at equilibrium, which formed **8** on heating to 180 °C (equation 3).⁶ With MeOH, **6** was converted to **9** and **10** at room temperature (equation 4).⁶ The failure of **7** to react with MeOH was attributed to the steric crowding present, as observed in similar examples.⁶





Migration of alkylthio or arylthio groups upon heating of allenyl thio esters 11 led via vinylketenes 12 to isolable cyclobutenones 13 (equation 5).⁷



A thorough investigation of the thermolysis of Meldrum's acid derivative 14 at 70 °C showed that acylketene 15 was formed and trapped by amine 16, forming the intermediate 17 used for drug development with 98% conversion (equation 6).⁸ Generation of 15 in the presence of imines resulted in the formation of [2 + 2] or [4 + 2] cycloaddition products.⁸ Heating of the Meldrum's acid derivative 18 in the presence of menthol at 180 °C was proposed to lead to ketene 19, which formed the ester 20 in 90% yield (equation 7).⁹ Thermolysis of Meldrum's acid derivative 21 formed ketene 22, together with the cumulenone PhNHCH=C= C=O, as intermediates observed in a separate low-temperature experiment, and these cyclized to 23 (equation 8).¹⁰ Other reactions of Meldum's acid derivatives are given in Sections 4.1.6 and 5.4.1.7.



### **162** PREPARATION OF KETENES

Norbornadiene gave [4 + 2] cycloaddition with 1,3,4-oxadiazin-6-ones **23** forming adducts **24**, which lost N₂ at room temperature to give acylketenes **26**, presumably via intermediates **25** (equations 9, 10).^{11,12} The ketenes reacted by thermal loss of cyclopentadiene to form pyrones **27** (equation 10).



Photolysis of 2-pyrone (**28**, R = H) in a glass at 90 K formed the ketene **29**, identified by its IR absorption at 2128 cm⁻¹ (equation 11) and its conversion to the ester with MeOH.¹³ Thermolysis at 550–625 °C of **28** (R = Me, Br) resulted in conversion to the 3-substituted isomer **31**, while for R = CO₂Et **28** isomerized to **29**, and for R =²H a mixture of **28** and **31** were formed.^{13,14} It was proposed that the isomerization occurs by hydrogen transfer interconverting **29** and **30** (equation 11).^{13–16} Flash photolysis of **28** (R = H) gave **29**, IR 2128 cm⁻¹, with a half-life for reversion to **28** of 2.9 µs at room temperature.¹⁶



Photolysis of **32** labeled with ¹³C at the carbonyl carbon gave the ketene **33**, which at 8 K in an Ar matrix gave IR absorption at 2068, 2074, 2081, and 2089 cm⁻¹ attributed to the four conformers **33a–d** (equation 12).¹⁷ As noted in Section 1.1.1 (ref. 26), it was concluded that **33d** is not an energy minimum structure. Upon further photolysis hydrogen transfer occurred, resulting in scrambling off the label to the aldehydic carbon, and with new ketene IR absorptions at 2121, 2129, 2133, and 2139 cm^{-1.¹⁷} With prolonged photolysis the ketenyl IR

absorption disappeared.



Similar photolysis with light of  $\lambda > 285$  nm of  $\alpha$ -pyrone **32** in an argon matrix produced IR bands at 2108, 2124, 2130, and 2135 cm⁻¹ assigned to two conformers of the *Z*-isomer of ketene **33**, and further irradiation showed conversion to the *E*-isomer with observed bands at 2122 and 2127 cm⁻¹.¹⁸ The bands were assigned with the aid of B3LYP/6-311++G** computations. Photolysis of the 4,6-dimethyl isomer **34** with light of  $\lambda > 315$  nm gave valence isomerization to the Dewar bicyclo [2.2.0] isomer **35** as the major product to the extent of 90%, with only 5% of the ketene **36**, identified by its characteristic IR absorption at 2128.9 cm⁻¹ (equation 13).¹⁸



Thermolysis at 550–625 °C of the 5-methyl and 5-bromo pyrones (**37**, R = Me, Br) led to the isomeric 3-pyrones **39**,¹³ while the 3-carboethoxypyrone isomerized to the 5-isomer (equation 14).¹⁴ It was proposed that the isomerizations occur by hydrogen shifts via the isomeric ketenes **38** to form the more stable products.^{13,14,20} The bicyclo[2.2.0] isomer of **37** (R=H) lost CO₂ upon photolysis to form cyclobutadiene.²¹



Oxazinone 40 (equation 15)²² and pyrimidones 42 (equation 16)²³ gave similar photochemical ring opening, forming ketenes 41 and 43, respectively.





Thermolysis of furandione **44** formed dibenzoylketene **45**, trapped at low temperature and identified by the IR absorption at 2140 cm⁻¹ at -196 °C (equation 17).^{24–26} Upon warming, the products **47** and **48** were isolated, evidently formed from the dimer **46** (equation 18).



Dehydrochlorination of 4-chloropyrazolin-5-one **49** was proposed to form 1,2diazacyclopentadienone **50**, which gave diazo ketene **51**, leading to bimane **52** (equation 19).²⁷ Thermal rearrangement of **52** to the *anti*-isomer **55** was proposed to involve initial formation of ketenes **53** and **54** (equation 19).²⁷ An alternative route to **50** was also reported.²⁸



Photolysis of triazolines **56** at 366 nm gave tricyclic ketones **57**, which upon photolysis through pyrex formed pyrrole-substituted ketenes **58**, as evidenced by capture with MeOH and the formation of ketene dimers (equation 21).^{29–31}



Pathways for the pseudopericyclic decarbonylations of furandione **59** leading to ketene **60** and of **61** leading to ketene **62** have been calculated (equations 22, 23).³² A detailed computational study, including changes in magnetic susceptibility and anisotropy to estimate the associated aromatization, indicated that pseudopericyclic reactions have at least one disconnection in the cyclic array of overlapping orbitals.³³ Photolysis of **63** in the presence of acetone was proposed to give an initial triplet that led to unobserved ketene **64**, which was trapped by alcohols or underwent decarbonylation (equation 17).³⁴



$$\left(\begin{array}{c} & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & &$$



Determination by X-ray crystallography of the structures of **65–67** that undergo cheleotropic fragmentation forming ketenes, as in equation 23, showed that all three are planar and have bond elongations as expected for incipient pseudopericyclic fragmention, in accord with the structure correlation principle.³⁵ Compound **67** was the least reactive and, as expected, showed the lowest degree of distortion.³⁵



Reaction of diphenylcyclopropenone (**69**) with **68** was proposed to result in formation of ketene **71** from **70** (equation 25).³⁶ Ketene formation has been proposed for reactions of other nucleophiles with **69**.³⁷



Gas phase IR irradiation of fluorooxirane (**72**) gave ketene (**73**) and HF (equation 26).^{38,39} Photolysis of biacetyl on Ag(111) with 193 nm light formed ketene (**73**) from biacetyl in a process by cleavage to  $CH_3C(\cdot)=O$  and loss of H• that was competitive with decarbonylation (equation 27).^{40,41} Electron activation of acetone on silver also produced ketene.⁴¹

$$\begin{array}{cccc} F & & & H \\ & & & & H \\ & & & H \\ \hline & & & H \\ \hline & & & & H \end{array} C = 0$$
 (26)

$$\overset{\text{CH}_3}{\underset{\text{O}}{\longrightarrow}} \overset{\text{O}}{\underset{\text{CH}_3}{\longleftarrow}} \overset{hv, \text{Ag}}{\underset{\text{193 nm}}{\longrightarrow}} \overset{\text{H}}{\underset{\text{H}}{\longrightarrow}} \text{C=O}$$

$$(27)$$

Irradiation of the tetraketone **74** in an argon matrix at 10 K gave an IR band at 2128 cm⁻¹assigned to ketene **75** (equation 28).⁴² The ketene **75** reacted further with **74** to give the tricyclic dimer **76** (equation 28).⁴² Ketene **75** generated independently by dehydrochlorination of the acyl chloride gave the dimer **77** (equation 28a).⁴²



Photolysis of the sulfoxide and selenoxide derivatives **78** gave ketenes **79**, which were observed by IR and trapped with alcohols and amines (equation 29).⁴³



Photolysis at 193 nm⁴⁴ or pyrolysis⁴⁵ of furan gave ketene (**73**) and acetylene by one pathway, as well as other products (equation 30). Pyrolysis of **80** give oxiranylidene **81**, which was directly observed in an argon matrix and formed  $CH_2=C=O$  (equation 31).⁴⁶

$$\begin{array}{c} & & \stackrel{193 \text{ nm}}{\longrightarrow} & \stackrel{H}{\longrightarrow} C=0 + CH \equiv CH + CH_3C \equiv CH + HC(\bullet)O + C_3H_3 \bullet + CO \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & \\ & & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ &$$

Photochemical isomerization of acrolein (82) to methylketene (83) by 193 nm light has been proposed^{47–49} and examined by calculations.⁵⁰ Further photolysis of 83 to CH₃CH: and CO (equation 32) may compete with formation of C₂H₃ and CHO from 82.⁴⁷ *E*-Crotonaldehyde (84) upon photolysis in the gas phase formed ethylketene (85), identified by its IR absorption at  $2132 \text{ cm}^{-1}$  (equation 33).^{51,52} Photolysis of hydroxyacetone giving successive radical elimination via 86 forming ketene was studied computationally (equation 34).⁵³



$$\begin{array}{ccc} CH_3 & & & \\ & & & \\ H & & & \\ H & & & \\ H &$$

$$HO \xrightarrow{O}_{CH_3} \xrightarrow{hv}_{-CH_3^*} HO \xrightarrow{O}_{C^*} \xrightarrow{-HO^*} \overset{H}{\overset{H}{\longrightarrow}} C=O \qquad (34)$$

Photolysis of *N*-acetylpropanamide (**87**) in  $CH_3OD$  gave methyl propanoate containing one deuterium atom, thus implicating the formation of methylketene (**83**, equation 35).⁵⁴



Photolysis of (2,4-dimethylbenzoyl)formate esters **88** resulted in formation of carbonyl compounds and hydroxyketenes **90** (equation 36).⁵⁶ The hydroxyketenes were proposed to tautomerize to aldehydes **91** that were photolyzed further to benz-aldehydes (equation 37).⁵⁶



Gas phase collisional activation of deprotonated peptides 92 was proposed to give ketenes 93 (equation 38).⁵⁷



Hydration of 1,1-diethynyl diene **94** gave the dienone **96**, proposed to result from formal ketene elimination from the intermediate **95** (equation 39).⁵⁸



Pyrolysis of **97** at 420  $^{\circ}$ C gave **99**, proposed to form via ketene **98** in a multistep reaction (equation 40).⁵⁹



Carbon atoms reacted with butanal at -196 °C to form *n*-propylketene (**100**), as evidenced by the isolation of pentanoic acid after addition of H₂O (equation 41).⁶⁰ The reaction pathway was studied computationally.⁶⁰

 $CH_{3}(CH_{2})_{3}CH=O \xrightarrow[-196\°C]{C} \xrightarrow{CH_{3}CH_{2}CH_{2}} H \xrightarrow{C=O} \xrightarrow{H_{2}O} CH_{3}(CH_{2})_{4}CO_{2}H \qquad (41)$ 100

Diphenylketene **101** was formed from the reaction of diphenyldiazomethane with Ni(CO)₄ and trapped as the ester in 35–74% yields (equation 42).⁶¹



## **REFERENCES FOR SECTION 3.4.6**

- 1. Erman, W. F. J. Am. Chem. Soc. 1967, 89, 3828-3841.
- 2. Erman, W. F. J. Am. Chem. Soc. 1969, 91, 779-780.
- 3. Erman, W. F.; Wenkert, E.; Jeffs, P. W. J. Org Chem. 1969, 34, 2196-2203.
- 4. Gibson, T. W.; Erman, W. F. J. Org. Chem. 1972, 37, 1148-1154.
- 5. Ho, C.-Y.; Bond, T. F. J. Am. Chem. Soc. 1974, 96, 7355-7356.
- 6. Hart, H.; Shih, E. J. Org. Chem. 1976, 41, 3377-3381.
- 7. Himbert, G.; Fink, D. Z. Naturforsch. 1994, 49b, 542-550.
- Xu, F.; Armstrong, J. D., III; Zhou, G. X.; Simmons, B.; Hughes, D.; Ge, Z.; Grabowski, E. J. J. J. Am. Chem. Soc. 2004, 126, 13002–13009.
- Davis, F. A.; Reddy, G. V.; Chen, B.-C.; Kumar, A.; Haque, M. S. J. Org. Chem. 1995, 60, 6148–6153.
- 10. Gaber, A. E.-A. M.; McNab, H. Synthesis 2001, 2059-2074.
- Christl, M.; Bodenschatz, G.; Feineis, E.; Hegmann, J.; Hüttner, G.; Mertelmeyer, S.; Schätzlein, K. *Liebigs Ann.* 1996, 853–861.

- Christl, M.; Bodenschatz, G.; Feineis, E.; Hegmann, J.; Hüttner, G.; Mertelmeyer, S.; Schätzlein, K.; Schwarz, H. J. Prakt. Chem. 1995, 337, 659–674.
- 13. Pirkle, W. H.; Seto, H.; Turner, W. V. J. Am. Chem. Soc. 1970, 92, 6984-6985.
- Huang, B.-S.; Pong, R. G. S.; Laureni, J.; Krantz, A. J. Am. Chem. Soc. 1977, 99, 4154– 4156.
- Chapman, O. L.; McIntosh, C. L.; Pacansky, J. J. Am. Chem. Soc. 1973, 95, 244– 246.
- 16. Arnold, B. R.; Brown, C. E.; Lusztyk, J. J. Am. Chem. Soc. 1993, 115, 1576-1577.
- 17. Arnold, B. R.; Radziszewski, J. G.; Campion, A.; Perry, S. S.; Michl, J. J. Am. Chem. Soc. **1991**, 113, 692–694.
- 18. Breda, S.; Reva, I.; Lapinski, L.; Fausto, R. Phys. Chem. Chem. Phys. 2004, 6, 929-937.
- Breda, S.; Lapinski, L.; Reva, I.; Fausto, R. Photochem. Photobiol. A: Chemistry 2004, 162, 139–151.
- Huang, B.-S.; Pong, R. G. S.; Lauréni, J.; Krantz, A. J. Am. Chem. Soc. 1977, 99, 4153– 4154.
- 21. Engel, C. R.; DeKrassny, A. F.; Belanger, A.; Dionne, E. Can. J. Chem. 1973, 51, 3263–3271.
- 22. Maier, G.; Schäfer, U. Liebigs Ann. Chem. 1980, 798-813.
- 23. Lapinski, L.; Nowak, M.-J.; Les, A.; Adamowicz, L. J. Am. Chem. Soc. 1994, 116, 1461–1467.
- Wentrup, C.; Winter, H.-W.; Gross, G.; Netsch, K.-P.; Kollenz, G.; Ott, W.; Biedermann, A. G. Angew. Chem. Int. Ed. Engl. 1984, 23, 800–802.
- 25. Ziegler, E.; Kollenz, G.; Ott, W. Synthesis, 1973, 679-680.
- Kolesnikova, O. N.; Livantsova, L. I.; Shurov, S. N.; Zaitseva, G. S.; Andreichikov, Yu. S. Zh. Obshch. Khim. 1990, 60, 467–468. J. Gen. Chem., (Engl. Transl.) 1990, 60, 406–407.
- 27. Kosower, E. M.; Pazhenchevsky, B. J. Am. Chem. Soc. 1980, 102, 4983-4993.
- 27a. Kosower, E. M.; Pazhenchevsky, B. J. Am. Chem. Soc. 1978, 100, 6516-6518.
- 28. Allen, A.; Anselme, J.-P. Tetrahedron Lett. 1996, 37, 5039-5040.
- 29. Schultz, A. G.; Sha, C.-K. J. Org. Chem. 1980, 45, 2040-2041.
- Schultz, A. G.; Dittami, J. P.; Myong, S. O.; Sha, C.-K. J. Am. Chem. Soc. 1983, 105, 3273–3279.
- 31. Schultz, A. G.; Staib, R. R.; Eng, K. K. J. Org. Chem. 1987, 52, 2968-2972.
- 32. Birney, D. M.; Ham, S.; Unruh, G. R. J. Am. Chem. Soc. 1997, 119, 4509-4517.
- Rodríquez-Otero, J.; Cabaleiro-Lago, E. M.; Hermida-Ramón, J. M.; Peña-Gallego, A. J. Org. Chem. 2003, 68, 8823–8830.
- 34. Ang, K. H.; Prager, R. H. Aust. J. Chem. 1998, 51, 483-489.
- 35. Wei, H.-X.; Zhou, C.; Ham, S.; White, J. M.; Birney, D. M. Org. Lett. 2004, 6, 4289–4292.
- Shimizu, H.; Ozawa, M.; Matsuda, T.; Ikedo, K.; Kataoka, T.; Hori, M.; Kobayashi, K.; Tada, Y. J. Chem. Soc., Perkin Trans. 1 1994, 1709–1717.
- 37. Kascheres, A.; Kacheres, C.; Braga, A. C. H. J. Org. Chem. 1993, 58, 1702-1703.
- Hollenstein, H.; Luckhaus, D.; Pochert, J.; Quack, M.; Seyfang, G. Angew. Chem., Int. Ed. Engl. 1997, 36, 140–143.
- 39. Ha, T.-K.; Pochert, J.; Quack, M. J. Phys. Chem. A. 1998, 102, 5241-5252.
- 40. Pylant, E. D.; Junker, K. H.; Szulczewski, G.; Hubbard, M. J.; White, J. M. J. Phys. Chem. B 1997, 101, 4803–4809.
- 41. Sparks, S. C.; Szabo, A.; Szulczewski, G. J.; Junker, K.; White, J. M. J. Phys. Chem. B 1997, 101, 8315–8323.
- 42. Rubin, M. B.; Etinger, M.; Kapon, M.; Krochmal, E. C., Jr.; Monosov, R.; Wierlacher, S.; Sander, W. J. Org. Chem. **1998**, *63*, 480–488.
- 43. Kobayashi, K.; Shinhara, S.; Moriyama, M.; Fujii, T.; Horn, E.; Yabe, A.; Furukawa, N. *Tetrahedron Lett.* **1999**, *40*, 5211–5213.
- 44. Sorkhabi, O; Qi, F.; Rizvi, A. H.; Suits, A. G. J. Chem. Phys. 1999, 111, 100-107.
- 45. Sendt, K.; Bacskay, G. B.; Mackie, J. C. J. Phys. Chem. A 2000, 104, 1861-1875.
- 46. Maier, G.; Reisenauer, H. P.; Cibulka, M. Angew. Chem. Int. Ed. 1999, 38, 105-108.
- 47. Umstead, M. E.; Shortridge, R. G.; Lin, M. C. J. Phys. Chem. 1978, 82, 1455-1460.
- 48. Fujimoto, G. T.; Umstead, M. E.; Lin, M. C. J. Chem. Phys. 1985, 82, 3042-3044.
- 49. Jen, S.-H.; Chen, I.-C. J. Chem. Phys. 1999, 111, 8448-8453.
- 50. Fang, W.-H. J. Am. Chem. Soc. 1999, 121, 8376-8384.
- 51. Coomber, J. W.; Pitts, J. N., Jr.; Schrock, R. R. Chem. Commun. 1968, 190-191.
- 52. Coomber, J. W.; Pitts, J. N., Jr. J. Am. Chem. Soc. 1969, 91, 4955-4960.
- 53. Wu, Y.; Xie, D. Q.; Xue, Y. J. Comput. Chem. 2003, 24, 931-938.
- 54. Fu, T. Y.; Scheffer, J. R.; Trotter, J. Tetrahedron Lett. 1994, 35, 3235-3238.
- 55. Sakamoto, M.; Tohnishi, M.; Fujita, T.; Watanabe, S. J. Chem. Soc., Perkin Trans. 1, 1991, 347.
- 56. Pirrung, M. C.; Tepper, R. J. J. Org. Chem. 1995, 60, 2461-2465.
- 57. Marzluff, E. M.; Campbell, S.; Rodgers, M. T.; Beauchamp, J. L. J. Am. Chem. Soc. 1994, 116, 6947–6948.
- 58. Hopf, H.; Kampen, J.; Bubenitschek, P.; Jones, P. G. Eur. J. Org. Chem. 2002, 1708–1721.
- 59. Batsanov, A. S.; Brooke, G. M.; Holling, D.; Kenwright, A. M. Chem. Commun. 1999, 1549–1550; J. Chem. Soc., Perkin Trans. 1 2000, 1731–1734.
- Dewar, M. J. S.; Nelson, D. J.; Shevlin, P. B.; Biesiada, K. A. J. Am. Chem. Soc. 1981, 103, 2802–2807.
- 61. Rüchardt, C.; Schrauzer, G. N. Chem. Ber. 1960, 93, 1840-1848.

## 3.5 KETENES FROM METAL CARBENE COMPLEXES

The discovery of the thermal reaction of chromium carbonyl carbene complexe **1** with alkynes leading to the formation of alkenylketene chromium complex **2**, which cyclized to naphthol complex **3**, was made by Dötz in 1975 (equation 1).¹ This process (Dötz reaction) was proposed to proceed through complex **2** on the basis of the reaction kinetics and the isolation of complexed products  $3^{.2-5}$  The cyclization resembles a metal-complexed version of the cyclization of alkenylketenes generated from arylcyclobutenones (Smith-Hoehn reaction, Section 3.4.1), but the

alkenylketene is assembled in situ (equation 1). This methodology has been frequently reviewed.⁴⁻¹¹



Reaction of the aminocarbene complex **4** with diphenylacetylene gave the zwitterion **6** with  $Cr(CO)_3$  complexed to the phenyl, and the structure of this complex was determined by X-ray (equation 3).¹² Related complexes were obtained in yields of 48–80%.¹²



The cobalt complex **7** reacted with 3-hexyne at 25 °C to form the alkenylketene complex **8**, whose structure was determined by X-ray and which was oxidized by Ce(IV) to furanone **9** (equation 3).¹³ These reactions also took place with molyb-denum and tungsten complexes.



A simplified mechanism has been presented¹⁴ in which metathesis of the alkyne and a vinyl chromium complex 10 via complex 11 gives complex 12 (equation 4), which inserts CO to form the dienylketene complex 13, which cyclizes to 14 (equation 5).



Density functional theoretical study of the reaction of aryl chromium carbene complex **15** with ethyne leading to complex **16**, which cyclized to naphthol **17** (equation 6), was used to elucidate the steps in the reaction.¹⁵ However, other recent calculations have favored an alternative pathway reaction scheme in which complex **18** formed **19**, which cyclized to **20**, and the CO insertion occurred without the formation of a ketene complex (equation 7).^{16,17}



It was also found that photolysis of chromium carbene complexes with visible light gave species formulated as chromium complexes with ketenes, and that these species undergo typical ketene reactions including additions of nucleophiles, and [2 + 2] cycloadditions with imines forming  $\beta$ -lactams, and with alkenes forming cyclobutanones.^{10,11} Thus, photolysis of aminocarbene chromium-complexes **21** with chiral auxiliaries on the nitrogen led to intermediate chromium complexed ketenes **22** that reacted with alcohols in an asymmetric synthesis of amino acid esters **23** (equation 8).¹¹ The mechanism has been studied computationally.^{11a}



Irradiation with visible light of the complex **24** formed the ketene complex **25** in the presence of imine **26** and gave the  $\beta$ -lactam **27** as a *cis/trans* mixture (equation 9).¹⁸ A DFT study using B3LYP/6–31G* level computations found structures for the carbonylation of (OC)₅Cr=CH₂ forming the complex (OC)₄Cr•CH₂=C=O, which formed a new complex with the imine CH₂ = NH (Figure 3.1), and then the transition state for [2 + 2] cycloaddition.¹⁸





**Figure 3.1** DFT-computed structures for reaction of ketene complex  $(OC)_4Cr$ •CH₂=C=O with imine CH₂=NH (reproduced from ref. 18 with permission of the American Chemical Society).

Chromium-complexed ketene **29** from carbene complex **28** reacted with aldehydes, forming  $\beta$ -lactones **30** with significant *syn* selectivity (equation 10).¹⁹ Electron-rich aldehydes (R = 4-anis, 2-furyl; 3-methyl-2-butenyl) gave alkenes **31** that resulted from decarboxylation of the  $\beta$ -lactones.¹⁹



The UV photolysis of amine-substituted chromium carbene complexes in matrices at 12 K and with flash photolysis at room temperature resulted in CO loss, but ketene complexes were not observed.²⁰ This led to the conclusion that these complexes were not primary photoproducts but were formed in subsequent steps. Photolysis of the complex MeOCMe=Cr(CO)₅ (**24**) in argon and nitrogen matrices also showed no evidence of the formation of metal ketene complexes.²¹

Photolysis of the complex 32 formed the ketene complex 43, which cyclized to the naphthol 34 (equation 11).²²



Reaction of complex **35** with the chiral alkyne **36** led to the ketene complex **37**, which gave the cyclohexadienone **38** with 92:8 *de* at the newly formed quaternary carbon (equation 12).²³



Experimental studies indicated that there was equilibration of regioisomeric vinyl carbene intermediates in benzannulation reactions.²⁴ The intermediates **39** and **40** formed from chromium carbene complex photolysis have been isolated

and characterized at low temperature (equation 13) and led to typical phenolic products of Dötz reactions (equation 14).²⁵



The reaction of carbene complex **28** with diynes led to mixtures of cyclobutenones **43** and naphthols **44** at 60 °C (equation 15), and heating of **43** at 80 °C also gave **44** (equation 16).²⁶ This latter process evidently involved the vinylketene **45**, suggesting that the formation of **44** from **43** may be metal assisted (equation 16).



Reaction of hydroxyalkynylalkane **46** with chromium carbene complex **47** gave lactone **49**, which was used in natural product synthesis (equation 17).²⁷ Propargylic alcohols gave thermal reactions with chromium carbene complexes **50** to form  $\beta$ -lactones **52** in processes suggested to involve ketene complexes **51** (equation 18).²⁸ This reaction was found to be assisted by sonication.^{29,30}





Chromium complex **53** and related derivatives prepared from the corresponding diazo compounds reacted with 1-hexyne via ketene complexes **54** to form **55** (equation 19).³¹ Reaction of **56** with bis(trimethylsilyl)acetylene gave the ketene **57** with an IR band at 2077 cm⁻¹, isolated in 22% yield as the methyl ester (equation 20).³²



Chromium carbene complex **58** with a pyrrole substituent gave similar benzannulation reactions with alkynes, as well as ketene [2 + 2] cycloaddition with cyclohexadiene, forming **59** together with an unusual [4 + 2] reaction across a ketene carbonyl forming **60** (equation 21).³³



3-(2-Vinyl)indolylcarbene complexes **62** gave ketene complexes **63**, which formed carbazole derivatives **64** used in the synthesis of carbazoquinocin (equation 22).³⁴



Photolysis of chromium carbene complex **65** in the presence of optically active ene carbamates proceeded through stereoselective [2 + 2] cycloaddition of ketene **66** to give cyclobutanone **67**, which was converted to a butenolide (equation 23).³⁵



Intramolecular [2 + 2] cycloaddition of ketene intermediates **69** from chromium carbene complex **68** formed the bicyclo[2.1.1]hexane derivative **70** (equation 24).³⁶ Reaction between carbene complex **24** and diene-yne **71** gave ketene **72**, which formed bicyclo[3.1.1[heptan-6-ones **73** as a possible taxane A-ring synthon, in a 2:1 ratio with a cyclobutenone from [2 + 2] cycloaddition (equation 25).³⁶



Vinylcyclopropyl carbene complex **74** gave a vinylcyclopropane rearrangement of the intermediate cyclopropylketene complex **75** to form cylopentenone **76** (equation 26).³⁶ A Dötz benzannulation reaction of a vinylcarbene complex was used as a step toward the the synthesis of kendomycin.³⁸ Carbene complex **77** with an alkenyl side chain bearing an electron withdrawing group formed ketenes **78**, which gave efficient cyclization to phenols **79** (equation 27).³⁹



#### **REFERENCES FOR SECTION 3.5**

- 1. Dötz, K. H. Angew. Chem. Int. Ed. Engl. 1975, 14, 644-645.
- 2. Fischer, H.; Mühlemeimer, J.; Märkl, R.; Dötz, K. H. Chem. Ber. 1982, 115, 1355-1362.
- 3. Dötz, K. H. New J. Chem. 1990, 14, 433-445.
- Dötz, K. H. In Advances In Metal Carbene Chemistry: Schubert, U., Ed; Kluwer: Boston, 1989: pp 199–210.
- 5. Dötz, K. H.; Tomuschat, P. Chem. Soc. Rev. 1999, 218, 187-198.
- Wulff, W. D.; Tang, P. C.; Chan, K.-S.; McCallum, J. S.; Yang, D. C.; Gilbertson, S. R. *Tetrahedron* 1985, 41, 5813–5832.
- 7. Wulff, W. D. Adv. Metal-Org. Chem. 1988, 1, 209-393.
- 8. Wulff, W. D. *Comprehensive Organic Synthesis*: Trost, B. M.; Fleming, I. Eds. Pergamon Press: New York, 1991: Vol. 5, pp 1065–1113.
- 9. Hsung, R. P.; Wulff, W. D.; Challener, C. A. Synthesis, 1996, 773-789.
- 10. Hegedus, L. S. Tetrahedron 1997, 53, 4105-4128.
- 11. Hegedus, L. S. Acc. Chem. Res. 1995, 28, 299-305.
- 11a. Fernandez, I.; Sierra, M. A.; Gômez-Gallego, M.; Mancheño, M. J.; Cossio, F. P. Chem. Eur. J. 2005, 11, 5988–5996.
- 12. Chelain, E.; Goumont, R.; Hamon, L.; Parlier, A.; Rudler, M.; Rudler, H.; Daran, J.; Daran, J.-C.; Vaissermann. J. *J. Am. Chem. Soc.* **1992**, *114*, 8088–8098.
- 13. Wulff, W. D.; Gilbertson, S. R.; Springer, J. P. J. Am. Chem. Soc. 1986, 108, 520–522.
- McCallum, J. S.; Kunng, F.-A.; Gilbertson, S. R.; Wuulff, W. D.; Organometallics 1988, 7, 2346–2360.

- Gleichmann, M. M.; Dötz, K. H.; Hess, B. A. J. Am. Chem. Soc. 1996, 118, 10551– 10560.
- 16. Torrent, M.; Duran, M.; Sola, M. Chem. Commun. 1998, 999-1000.
- 17. Torrent, M.; Duran, M.; Sola, M. J. Am. Chem. Soc. 1999, 121, 1309-1316.
- Arrieta, A.; Cossío, F. P.; Fernandez, I.; Gómez-Gallego, M.; Lecea, B.; Mancheño, M. I.; Sierra, M. A. J. Am. Chem. Soc. 2000, 122, 11509–11510.
- 19. Merlic, D. A.; Doroh, B. C. J. Org. Chem. 2003, 68, 6056-6059.
- Doyle, K. O.; Gallagher, M. L.; Pryce, M. T.; Rooney, A. D. J. Organomet. Chem. 2001, 617–618, 269–279.
- 21. Gallagher, M. L.; Greene, J. B.; Rooney, A. D. Organometallics 1997, 16, 260-268.
- 22. Merlic, C. A.; Xu, D. J. Am. Chem. Soc. 1991, 113, 7418-7420.
- Hsung, R. P.; Quinn, J. F.; Weisenberg, B. A.; Wulff, W. D.; Yap, G. P. A.; Rheingold, A. L. Chem. Commun. 1997, 615–616.
- 24. Waters, M.; Bos, M. E.; Wulff, W. D. J. Am. Chem. Soc. 1999, 121, 6403-6413.
- Barluenga, J.; Aznar, F.; Gutiérrez, I.; Martín, A.; García-Granda, S.; Llorca-Baragaño, M. A. J. Am. Chem. Soc. 2000, 122, 1314–1324.
- 26. Bao, J.; Wulff, W. D.; Fumo, M. J.; Grant, E. B.; Heller, D. P.; Whitcomb, M. C.; Yeung, S.-M.. J. Am. Chem. Soc. **1996**, 118, 2166–2181.
- 27. Ishibashi, T.; Ochifuji, N.; Mori, M. Tetrahedron Lett. 1996, 37, 6165-6168.
- Harrity, J. P. A.; Heron, N. M.; Kerr, W. J.; McKendry, S.; Middlemiss, D.; Scott, J. S. Synlett 1996, 1184–1186.
- Caldwell, J. J.; Herrity, J. P. A.; Heron, N. M.; Kerr, W. J.; McKendry, S.; Middlemiss, D. *Tetrahedron Lett.* **1999**, *40*, 3481–3484.
- 30. Caldwell, J. J.; Kerr, W. J.; McKendry, S. Tetrahdron Lett. 1999, 40, 3485-3486.
- 31. Dötz, K. H.; Pfeiffer, J. J. Chem. Soc., Chem. Commun. 1996, 895-896.
- 32. Pfeiffer, J.; Nieger, M.; Dötz, K. H. Chem. Eur. J. 1998, 4, 1843-1851.
- 33. Merino, I.; Hegedus, L. S. Organometallics 1995, 14, 2522-2531.
- 34. Rawat, M.; Wulff, W. D. Org. Lett. 2004, 6, 329-332.
- 35. Kedar, T. E.; Miller, M. W.; Hegedus, L. S. J. Org. Chem. 1996, 61, 6121-6126.
- 36. Moser, W. H.; Hegedus, L. S. J. Am. Chem. Soc. 1996, 118, 7873-7879.
- 37. Jiang, W.; Fuertes, M. J.; Wulff, W. D. Tetrahedron 2000, 56, 2183-2194.
- 38. White, J. D.; Smits, H. Org. Lett. 2005, 7, 235-238.
- Barluenga, J.; Lopez, L. A.; Martínez, S.; Tomas, M. J. Org. Chem. 1998, 63, 7588– 7589.
- 40. Dumas, S.; Lastra, E.; Hegedus, L. S. J. Am. Chem. Soc. 1995, 117, 3368-3379.
- 41. Hsiao, Y.; Hegedus, L. S. J. Org. Chem. 1997, 62, 3586-3591.
- 42. Puntener, K.; Hellman, M. D.; Kuester, E.; Hegedus, L. S. J. Org. Chem. 2000, 65, 8301–8306.
- 43. Bouaucheau, C.; Parlier, A.; Rudler, H. J. Org. Chem. 1997, 62, 7247-7259.
- 44. Leese, T.; Dötz, K. H. Chem. Ber. 1996, 129, 623-631.
- 45. Merlic, C. A.; McInnes, D. M.; You, Y. Tetrahedron Lett. 1997, 38, 6787-6790.
- 46. Quinn, J. F.; Bos, M. E.; Wulff, W. D. Org. Lett. 1999, 1, 161-164.
- 47. Barulenga, J.; Aznar, F.; Palomero, M. A.; Barluenga, S. Org. Lett. 1999, 1, 541-543.

- 48. Rudler, H.; Parlier, A.; Bezennine-Lafollée, Vaissermann, J. Eur. J. Org. Chem. 1999, 2825–2833.
- 49. Martin-Vaca, B.; Rudler, H. J. Chem. Soc. Perkin 1 1997, 3119-3121.
- 49a. Hsung, R. P.; Xu, Y.-C.; Wulff, W. D. Tetrahedron Lett. 1995, 36, 8159-8162.
- 50. Weyershausen, B.; Dötz, K. H. Eur. J. Org. Chem. 1998, 1739-1742.
- 51. Wang, H.; Wulff, W. D. J. Am. Chem. Soc. 1998, 120, 10573-10574.
- 52. Wang, H.; Wulff, W. D.; Rheingold, A. L. J. Am. Chem. Soc. 2000, 122, 9862-9863.
- 53. Wang, H.; Huang, J.; Wulff, W. D.; Rheingold, A. L. J. Am. Chem. Soc. 2003, 125, 8980–8981.
- 54. Herndon, J. W.; Wang, H. J. Org. Chem. 1998, 63, 4562-4563.
- 55. Barluenga, J.; López, S.; Trabanco, A. A.; Fernández-Acebes, A.; Flórez, J. J Am. Chem. Soc. 2000, 122, 8145–8154.
- 56. Dötz, K. H.; Klumpe, M.; Nieger, M. Chem. Eur. J. 1999, 5, 691-695.
- 57. Gilbert, A. M.; Miller, R.; Wulff, W. D. Tetrahedron 1999, 55, 1607-1630.
- 58. Vorogushin, A. V.; Wulff, W. D.; Hansen, H.-J. Org. Lett. 2001, 3, 2641-2644.
- 59. Vorogushin, A. V.; Wulff, W. D.; Hansen, H.-J. J. Am. Chem. Soc. 2002, 124, 6512-6513.
- 60. Tumer, S. U.; Herndon, J. W.; McMullen, L. A. J. Am. Chem. Soc. 1992, 114, 8394–8404.
- 61. Herndon, J. W.; Zhu, J. Org. Lett. 1999, 1, 15-18.
- 62. Herndon, J. W.; Zhu, J. Tetrahedron.Lett. 1998, 39, 7443-7446.
- 63. Barluenga, J.; Aznard, F.; Palomero, M. A. Angew. Chem. Int. 2000, 39, 4346–4348.
- 64. Merlic, C. A.; Baur, A.; Aldrich, C. C. J. Am. Chem. Soc. 2000, 122, 7398-7399.
- 65. Merlic, C. A.; Aldrich, C. C.; Albaneze-Walker, J.; Saghatelian, A.; Mammen, J. J. *Org. Chem.* **2001**, *66*, 1297–1309.
- 66. Jackson, T. J.; Herndon, J. W. Tetrahedron 2001, 57, 3859-3868.
- 67. Jiang, M. X.-W.; Rawat, M.; Wulff, W. D. J. Am. Chem. Soc. 2004, 126, 5970-5971.

# 3.6 KETENE FORMATION FROM CATIONS AND FREE RADICALS

Ketenes are reactive toward acids, and their formation as recognizable intermediates in the presence of acids is rare. An exception occurs where intermediates with acylium ion character are formed, as in the reaction of methylmalonic acid or the dimethyl ester in SbF₅-FSO₃H forming **1**, as identified by the IR absorption at 2150 cm⁻¹ and the ¹H NMR signal at  $\delta$  3.05 (equation 1).¹



Acetyl chloride with AlCl₃ in CD₂Cl₂ formed protonated diacetylketene tetrachloroaluminate (2), which was characterized by  $\overline{IR}$  absorption at 2100 cm⁻¹ and by ¹H and ¹³C NMR (equation 2).² Polyketenes are formed from acyl halides under similar conditions in reactions proposed to involve an equilibrium between ketene and acylium ions (Section 5.6).³

$$CH_{3}C'_{Cl} \xrightarrow{AlCl_{3}} H \xrightarrow{P} O = CH_{3} \xrightarrow{P} C= O \xrightarrow{CH_{3}} H \xrightarrow{O} \xrightarrow{CH_{3}} C= O \xrightarrow{CH_{3}} C= O \xrightarrow{P} H \xrightarrow{O} \xrightarrow{CH_{3}} C= O \xrightarrow{CH_{3}} C$$

Gas phase dissociation of the malonate-derived cation 3 led to the ion-neutral complex 4 of ketene and acetylium ion (equation 3).⁴

Ketenes are quite reactive toward free radicals, but there are reports of ketene formation from radical processes. Thus, kinetic measurements for the formation of ketene from the reaction of Cl- with acetyl radical 5 were obtained by IR spectroscopy, as well as the rate constant for Cl- addition to ketene forming chloroacetvl radical **6** (equation 4).⁵ The vinyloxy radical **7** forms  $CH_3$ • and ketene by competitive processes (equation 5).⁶ Other examples of ketene formation from  $CH_3C(\bullet) = O(5)$  are given in Section 3.4.6.

$$CH_{3}-C^{\bullet} \xrightarrow{CI_{\bullet}} \overset{H}{\underset{-HCl}{\longrightarrow}} C=0 \xrightarrow{CI_{\bullet}} \overset{CI_{\bullet}}{\underset{-C}{\longrightarrow}} \overset{O}{\underset{-C}{\longrightarrow}} CH_{2}CI_{\bullet}$$
(4)

$$5 \qquad 6$$

$$CH_{3^{\bullet}} + CO \qquad H \qquad H \qquad H \qquad H \qquad H \qquad C=O + H^{\bullet} \qquad (5)$$

$$7$$

#### **REFERENCES FOR SECTION 3.6**

5

- 1. Conrow, K.; Morris, D. L. J. Chem. Soc., Chem. Commun. 1973, 5-6.
- 2. Csihony, S.; Bodor, A.; Rohonczy, J.; Horvath, I. T. J. Chem. Soc. Perkin Trans. 1, 2002, 2861-2865.
- 3. Olah, G. A.; Zadok, E.; Edler, R.; Adamson, D. H.; Kasha, W.; Prakash, G. K. S. J. Am. Chem. Soc. 1989, 111, 9123-9124.
- 4. Chamot-Rooke, J.; Amekraz, B.; Tortajada, J.; Mourgues, P.; Audier, H. E. J. Mass. Spectrom. 1997, 32, 779-780.

- 5. Maricq, M. M.; Ball, I. C.; Straccia, A. M.; Szente, J. J. Int. J. Chem. Kinet. 1997, 29, 421–429.
- 6. Osborn, D. L.; Choi, H.; Neumark, D. M. Adv. Chem. Phys. 1997, 101, 729-744.

#### 3.7 KETENES FROM OXIDATION OF ALKYNES

Ketene (4) is known to be formed by the oxidation of acetylene in a reaction important in studies of combustion, for which many steps are involved.¹ A DFT computational study examined this process and included the sequence of oxygen atom insertion into a C–H bond of acetylene forming ethynol (1), as well as addition to the triple bond forming oxirene (2) and formylcarbene (3), and the rearrangement of these species to ketene (equation 1).¹

$$H \xrightarrow{O} H \xrightarrow{O} H \xrightarrow{O} H + H \xrightarrow{O} H \xrightarrow{$$

Oxidation of arylalkynes with peracids was postulated to form ketenes, and reaction of 2-deuterobiphenylylacetylene **5** with *m*-chlorobenzoic acid in methanol gave the methyl ester **8**, consistent with formation of the oxirene **6** and the ketene **7** (equation 2).²

$$ArC \equiv CD \xrightarrow{ArCO_{3}H} Ar \xrightarrow{O} D \xrightarrow{Ar} D = C = O \xrightarrow{MeOH} D \xrightarrow{Ar} CO_{2}Me$$
(2)  
5 (Ar = PhC₆H₄) 6 7 8

Peracids oxidized phenylacetylene and diphenylacetylene to mixtures including ketene-derived products.^{3,4} Similar reactions of cycloalkynes including cyclooctyne **9** gave products derived from keto carbene **10** and rearrangement to ketene **11**, which was further oxidized to cycloheptanone **12** (equation 3).⁵



The oxidation of alkynes **13** by  $H_2O_2$  catalyzed by methylrhenium trioxide (CH₃ReO₃) formed  $\alpha$ -diketones and carboxylic acids from reactions interpreted as involving oxirene intermediates **14** that underwent partial rearrangement via ketocarbenes to ketenes (equation 4).⁶ Alkyne oxidation with the HOF•CH₃CN complex also gave products indicative of intermediate oxirene formation.⁷

$$R-C\equiv C-R \xrightarrow{H_2O_2} R \xrightarrow{O} R \xrightarrow{R} R \xrightarrow{R} R \xrightarrow{C} C=C \xrightarrow{(0)} R \xrightarrow{R} R \xrightarrow{R} C=O \qquad (4)$$
13

#### **184** PREPARATION OF KETENES

Incubation of thymine hydroxylase with 5-ethynyluracil (15) gave 5-(carboxymethyl)uracil (18) and uracil-5-acetylglycine 19 and resulted in inactivation of the protein by covalent modification.^{8,9} The formation of 18 and 19 was proposed to result from formation of an oxirene 16 and a ketene 17 (equations 5, 6). The protein inactivation resulted from reaction with the ketocarbene.



Arylpropynes ArC $\equiv$ CCH₃ **20** acted as mechanism-based inhibitors of cytochrome P450-dependent monooxygenase enzymes, and were proposed to function by formation of oxirenes **21** that formed ketenes **22** by 1,2-methyl migration, as evidenced by the isolation of 2-arylpropionic acids (equation 7),¹⁰ as had been proposed earlier.¹¹



The reaction of alkynes **23** with oxygen atoms in the gas phase proceeded by formation of ketocarbenes that formed excited ketenes **24**, which then underwent decarbonylation, forming carbenes **25** (equation 8).¹² The photo-reaction of ethyne and propyne complexes with NO₂ in solid argon led to transfer of an oxygen atom and formation of  $CH_2=C=O$  and  $CH_3CH=C=O$ , respectively, followed by decarbonylation, as monitored by FTIR.¹³ Gas phase reaction of  $NO_2^+$  with acetylene formed *O*-nitrosonium ion ketene complex **26**, which dissociated to ketene (equation 9).¹⁴ Several pathways to this product were examined computationally.

$$RC \equiv CR_1 \xrightarrow{O} \underset{R}{\overset{O}{\longrightarrow}} \underset{R}{\overset{R_1}{\underset{R}{\longrightarrow}}} = C = O \xrightarrow{R_1} \underset{R}{\overset{R_1}{\underset{R}{\longrightarrow}}} C :$$

$$23 \qquad 24 \qquad 25 \qquad (8)$$

$$HC \equiv CH \xrightarrow{NO_2^*} O \xrightarrow{N} O \xrightarrow{H} H \xrightarrow{H} C = O \cdots \stackrel{+}{N} O \xrightarrow{-NO^*} H \xrightarrow{H} C = O \qquad (9)$$

Acetylene reacted with  $CrCl_2O_2$  in a matrix, and IR bands were observed at 2105 and 2195 cm⁻¹, and assigned to the end-on ketene complex **27** and the more stable ketene complex **28** with the C=C bond (equation 10).¹⁵

$$HC \equiv CH \xrightarrow{CrCl_2O_2} H \xrightarrow{H} \xrightarrow{CrCl_2O} C \equiv O = --CrCl_2O + H \xrightarrow{H} \xrightarrow{CrCl_2O} C \equiv O$$
(10)  
27 (IR 2105 cm⁻¹) 28 (IR 2195 cm⁻¹)

Computational studies of possible products  $C_{60}O_n$  that conceivably could be formed by the oxidation of the fullerene  $C_{60}$  included pentaketene  $C_{60}O_5$  and hexaketene  $C_{60}O_6$ .¹⁶ These were not among the more stable isomers but  $C_{60}O_5$  was esthetically interesting, as illustrated for oxidation of the pentabenzocyclopentyl unit fragment of  $C_{60}$  **29** forming **30** (equation 11).¹⁶



#### **REFERENCES FOR SECTION 3.7**

- 1. Girard, Y.; Chaquin, P. A. J. Phys. Chem. A 2003, 107, 10462-10470.
- 2. Ortiz de Montellano, P. R.; Kunze, K. L. J. Am. Chem. Soc. 1980, 102, 7373-7375.
- 3. Stille, J. K.; Whitehurst, D. D. J. Am. Chem. Soc. 1964, 86, 4871-4876.
- 4. McDonald, R. N.; Schwab, P. A. J. Am. Chem. Soc. 1964, 86, 4866-4871.
- 5. Concannon, P. W.; Ciabattoni, J. J. Am. Chem. Soc. 1973, 95, 3284-3289.
- 6. Zhu, Z.; Espenson, J. H. J. Org. Chem. 1995, 60, 7728-7732.
- 7. Dayan, S.; Ben-David, I.; Rozen, S. J. Org. Chem. 2000, 65, 8816-8818.
- 8. Lai, M.-T.; Wu, W.; Stubbe, J. J. Am. Chem. Soc. 1995, 117, 5023-5030.
- 9. Thornburg, L. D.; Stubbe, J. Biochemistry, 1993, 32, 14034.
- Foroozesh, M.; Primrose, G.; Guo, Z.; Bell, L. C.; Alworth, W. L.; Guengerich, F. P. Chem. Res. Toxicol. 1997, 10, 91–102.

- 11. Ortiz de Montellano, P. R.; Kunze, K. L. Arch. Biochem. Biophys. 1981, 209, 710-712.
- 12. Xing, G.; Huang, X.; Wang, X.; Bersohn, R. J. Chem. Phys. 1996, 105, 488-495.
- 13. Harrison, J. A.; Frei, H. J. Phys. Chem. 1994, 98, 12142-12151.
- Bernardi, F.; Cacace, F.; de Petris, G.; Pepi, F.; Rossi, I.; Troianc, A. Chem. Eur. J. 2000, 537–544.
- 15. Ault, B. S. J. Phys. Chem. A 2004, 108, 5537-5543.
- 16. Kepert, D. L.; Clare, B. W. Inorg. Chim. Acta 2002, 327, 41-53.

#### 3.8 OTHER ROUTES TO KETENES

Ketene has been proposed as an intermediate in the H-ZSM-5 zeolite-catalyzed conversion of methanol to gasoline.¹ The mechanism put forward proceeded with generation of CO, which reacted with activated methanol, forming acylium ion **1**, which was reversibly deprotonated (equation 1).¹ Further conversion to higher acylium ions and ketenes occurred succesively, resulting in the formation of higher hydrocarbons. Ketene was also implicated by trapping as an intermediate in the conversion of synthesis gas to ethanol using Rh-Ti/SiO₂.^{2,3} Reaction of CH₂=CHCH=O on a Pt surface also formed some CH₂=C=O, together with some propene.^{4,5} Metal oxide–catalyzed conversion of acetic acid to acetone was suggested to involve ketene formation on the catalyst surface.^{5a}

$$CH_{3}OH_{2}^{+} \xrightarrow{CO} CH_{3}\dot{C}=O \xrightarrow{-H^{+}} H \xrightarrow{H} C=O \xrightarrow{CH_{3}OH_{2}^{+}} CH_{3}CH_{2}\dot{C}=O \longrightarrow etc.$$
(1)

Ketene formation from the reaction of 1,3-thiazolidine **3** with CO in the presence of chloro(1,5-cyclooctadiene)rhodium(I) dimer and KI was shown to give elimination of phenylketene (**4**) in 8–10% yields, as evidenced by the observation of a ketene band at  $2163 \text{ cm}^{-1}$  in the IR, and capture as the ester with MeOH (equation 2).⁶



The formation of norbornylideneketene **5** during the ring-opening polymerization of norbornene using  $W(CO)_3Cl_2AsPh_3)_2$  was demonstrated by trapping with EtOD (equation 3).⁷



Ynamine **6** reacted with CO₂ and was proposed to form the unsaturated  $\beta$ -lactone **7**, which gave the acylketene **8** resulting from [2 + 2] cycloaddition, and reaction with a second molecule of **6** gave the observed product **8a** (equation 4).^{8–10} Reactions of bis(ynamines) gave polymerization to poly(4-pyrones).⁸



Formation of ketene **11** by addition of the carbene C=C=O (**10**) formed by photolysis of  $C_3O_2$  (**9**) to norbornadiene has been suggested (equation 5).¹¹ No direct evidence for **11** was obtained, and further photolysis led to formation of a  $C_8H_8$  hydrocarbon.¹¹ Other reactions of **10** forming ketenes are discussed in Section 4.11.

$$0=C=C=C=0 \xrightarrow{hv} C=C=0 \xrightarrow{hv} C^{-0}$$
(5)  
9 10 11

The possibility of formation of ketene (2) by  $\alpha$ -elimination from  $\alpha$ -bromoenol phosphates 12 by a mechanism-based inhibitor of ribonuclease A¹² and from 13 by inhibition of phophotriesterase has been considered (equation 6).¹³



The  $\alpha$ -elimination of methoxide from  $\alpha$ -bromo ester enolate **14** forming ketene **15** was shown to occur by capture of the deuterated ester **16**, although to a small extent (equation 7).¹⁴ The reaction of **17** with dimethoxycarbene to form **19** was

proposed to involve formation of a ketene **18** by methoxide rearrangement (equation 8).¹⁵ Gas phase  $\alpha$ -elimination by collision-induced dissociation of **20** to form methylketene **21** has been observed (equation 9).¹⁶



Reaction of the aldehyde 22 with concentrated HCl has been proposed to form the ketene 23 as an intermediate observed by ¹H NMR in THF through an addition/ elimination sequence, leading to the ketone 24 (equation 10)¹⁷ However, such an elimination appears unusual, and other mechanisms are possible, so further proof of the intermediacy of ketene 23 is required.



Photolysis of **25** in a matrix gives  $Me_2C=C=O$  (**27**), possibly through the oxirene **26** (equation 11).¹⁸



The formation of  $CH_2=C=O(2)$  in the pyrolysis and oxidation of  $CH_4^{19}$  and of  $CH_2=CH_2^{20}$  has also been considered. Studies on oxidation of  $CH_4/CH_2=CH_2$  mixtures using FTIR and UV detection, and including a reevaluation of previous results, gave strong evidence for ketene formation by the IR band observed at 2152 cm⁻¹, and also indicated that the formation of ketene was usually underestimated due to loss of ketene in sampling.²¹ Ketene formation was also implicated in the oxidations of alkenes and alkynes, including combustion reactions. For example, ketene has been proposed as an intermediate in the reaction of acetylene with ozone,^{22–24} and reaction of ethylene with oxygen atoms could form the enolic radical **28**, which gave ketene (equation 12).

$$CH_2=CH_2 \xrightarrow{O} \cdot CH_2CH=O \xrightarrow{-H} H \xrightarrow{H} C=O$$
(12)  
28 2

Reaction of isoxazol-5(2*H*)-ones **29** with bases was proposed to form ketene intermediate **30** (equation 13), which reacted with amines or cyclized and was trapped with  $CH_3I$  (equation 14).²⁵



Reaction of **31** with CO, giving a possible ketene intermediate **32**, was proposed in the formation of **33** (equation 15).²⁶



Thermolysis of diphenylcyclopropenone **34** gave a dimer with the probable structure **36**, which may have formed by ring opening to carbene-substituted ketene **35**, which reacted with **34** to form **36** (equation 16).²⁷ Addition of phenyllithium to **34** led to the formation of ketene **37** upon acidification, as evidenced by observation of the IR absorption at  $2100 \text{ cm}^{-1}$ , and isolation of the acid **38** (equation 17).²⁸ Reaction of **34** with Ph₃P gave ketene **39**, identified by formation of the methyl ester **40** (equation 18).²⁹



Phenylketene (4) was generated in EtOH by a Wolff-type rearrangement of the sulfur ylide 41 and isolated in 26% yield as the ethyl ester (equation 19).³⁰ Reaction of *n*-butanal with carbon atoms and hydrolysis gave some pentanoic acid, indicative of formation of *n*-propylketene 42 by insertion into the C=O bond (equation 20).³¹

$$\begin{array}{cccc} & & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & & \\ & & & & & \\ & & & & & & \\ & & & & & \\ & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & &$$

#### **REFERENCES FOR SECTION 3.8**

- 1. Jackson, J. E.; Bertsch, F. M. J. Am. Chem. Soc. 1990, 112, 9085-9092.
- Wang, H.; Liu, J.; Fu, J.; Cai, J.; Zhang, H.; Cai, Q. J. Nat. Gas Chem. 1993, 2, 13–18. Chem. Abstr. 1993, 119, 48820v.
- 3. Wang, H.; Liu, J.; Fu, J.; Wan, H.; Tsai, K. Catal. Lett. 1992, 12, 87-96.
- 4. de Jesús, J. C.; Zaera, F. J. Mol. Cat. A. Chem. 1999, 138, 237-240.
- 5. de Jesús, J. C.; Zaera, F. Surface Sci. 1999, 430, 99-115.
- Pestman, R.; Koster, R. M.; van Duijne, A.; Pieterese, J. A. Z.; Ponec, V. J. Catal. 1997, 168, 265–272.
- 6. Khumtaveeporn, K.; Alper, H. J. Am. Chem. Soc. 1994, 116, 5662-5666.
- 7. Bencze, L.; Kraut-Vass, A.; Prókai, L. J. Chem. Soc. Chem. Commun. 1985, 911-912.
- 8. Tsuda, T.; Hokazono, H.; Toyota, K. J. Chem. Soc. Chem. Commun. 1995, 2417-2418.
- 9. Ficini, J.; Pouliquen, J. Tetrahedron Lett. 1972, 1131-1134.
- 10. Ficini, J.; Pouliquen, J. J. Am. Chem. Soc. 1971, 93, 3295-3297.
- 11. Klumpp, G. W.; van Dijk, P. M. Recl. Trav. Chim. Pays-Bas. 1971, 90, 381-384.
- 12. Stowell, J. K.; Widlanski, T. S.; Kutateladze, T. G.; Raines, R. T. J. Org. Chem. 1995, 60, 6930–6936.
- 13. Hong, S.-B.; Mullins, L. S.; Shim, H.; Raushel, F. M. Biochemistry 1997, 36, 9022–9028.
- 14. Maryanoff, C. A.; Sorgi, K. L.; Zientek, A. M. J. Org. Chem. 1994, 59, 237-239.
- 15. de Meijere, A.; Kozhushkov, S. I.; Yufit, D. S.; Boese, R.; Haumann, T.; Pole, D. L.; Sharma, P. K.; Warkentin, J. *Liebigs Ann.* **1996**, 601–612.
- Chyall, L. J.; Brickhouse, M. D.; Schnute, M. E.; Squires, R. R. J. Am. Chem. Soc. 1994, 116, 8681–8690.
- 17. Ogawa, H.; Taketugu, Y.; Imoto, T.; Taniguchi, Y.; Kato, H. *Tetrahedron Lett.* **1979**, 3457–3460.
- 18. Maier, G.; Sayrac, T.; Reisenauer, H. P. Chem. Ber. 1982, 115, 2202-2213.
- 19. Hidaka, Y.; Sato, K.; Henmi, Y.; Tanaka, H.; Inami, K. Combust. Flame 1999, 118, 340-358.
- 20. Hidaka, Y.; Nishimori, T.; Sato, K.; Henmi, Y.; Okuda, R.; Inami, K.; Higashihara, *Combust. Flame* **1999**, *117*, 755–776.
- 21. Scire, J. J., Jr.; Klotz, S. D.; Dryer, F. L. Zeit. Phys. Chem. 2001, 215, 1011-1023
- 22. Cremer, D.; Crehuet, R.; Anglada, J. J. Am. Chem. Soc. 2001, 123, 6127-6141.
- 23. Morton, M. L.; Szpunar, D. E.; Butler, L. J. J. Chem. Phys. 2001, 113, 204-216.
- 24. Su, H.; Bersohn, R. J. Chem. Phys. 2001, 113, 217-228.
- 25. Prager R. H.; Williams, C. M. Heterocycles 1999, 51, 3013-3030.
- 26. Gil-Rubio, J.; Weberndörfer, B.; Werner, H. Angew. Chem. Int. Ed. 2000, 39, 786–789.
- Breslow, R.; Eicher, T.; Krebs, A.; Peterson, R. A.; Posner, J. J. Am. Chem. Soc. 1965, 87, 1320–1325.
- 28. Ciabbattoni, J.; Kocienski, P. J.; Melloni, G. Tetrahedron Lett. 1969, 1883-1887.
- 29. Hamada, A.; Takizawa, T. Tetrahedron Lett. 1972, 1849-1850.
- 30. Trost, B. M. J. Am. Chem. Soc. 1966, 88, 1587-1588.
- Dewar, M. J. S.; Nelson, D. J.; Shevlin, P. B.; Biesiada, K. A. J. Am. Chem. Soc. 1981, 103, 2802–2807.

# **4** Types of Ketenes

## 4.1 CARBON-SUBSTITUTED KETENES

#### 4.1.1 Alkylketenes

The first report of an alkylketene was the preparation of dimethylketene (1) by Staudinger in his second publication on ketenes, which utilized the dehalogenation of  $\alpha$ -bromoisobutyryl bromide (equation 1).¹ However, as acknowledged by Staudinger in this report, **1** had probably already been generated, but not identified, in studies by Wedekind and Weisswange, as reported in an unpublished dissertation in 1904.¹ In this study the reaction of isobutyryl chloride with triethylamine formed the dimer **2**.

$$\begin{array}{cccc} CH_3 & O & & CH_3 \\ Br & & & \\ CH_3 & Br & & CH_3 \\ & & & CH_3 \\ \end{array} C=O \tag{1}$$

$$\begin{array}{cccc} CH_3 & O & \xrightarrow{Et_3N} & CH_3 \\ & & & & \\ CH_3 & Cl & & & \\ & & & CH_3 \end{array} \xrightarrow{CH_3} C=O & \longrightarrow \begin{array}{cccc} CH_3 & & & \\ CH_3 & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & & \\ & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & &$$

The parent ketene  $CH_2=C=O(3)$  is included in the alkylketene family, and the thermodynamic stability of ketene and methylketene (4) compared to the corresponding alkenes was found by the isodesmic comparison of equation  $3^{2,3}$ . This indicated some thermodynamic destabilization compared to the alkene for methylketene (4), but although both ketene and methylketene dimerize readily, they do have some stability in solution and even briefly as neat liquids. This is in contrast to arylketenes such as phenylketene, which dimerize much more rapidly.

Ketenes II, by Thomas T. Tidwell

Copyright © 2006 John Wiley & Sons, Inc.

Ketene (3) is a noxious gas that was first prepared by Wilsmore in 1907⁴ from the pyrolysis of acetic anhydride with a hot platinum wire through a likely cyclic transition state, and this method is still used today (equation 4).⁵ Ketene was then prepared by Staudinger and Klever in low yield by the zinc dehydrohalogenation of bromoacetyl bromide with zinc.⁶ There was some discussion as to whether ketene had the CH₂=C=O structure or the alkynol structure HC≡COH, but this was soon decided in favor of the former.^{7,8} The structure of ketene has been determined both by computations⁹⁻¹¹ and by microwave spectroscopy.^{11,12} The preparation, properties, and uses of ketene have been reviewed.^{13,14}

$$(CH_{3}CO)_{2}O \xrightarrow{500-510 \circ C} \begin{bmatrix} O^{--H} \\ CH_{3}-C \\ & C \\ O^{--C} \\ & O \end{bmatrix} \xrightarrow{H} C=O + CH_{3}CO_{2}H$$
(4)

The ¹H NMR signal of ketene was reported as  $\delta$  2.47 (neat),  $J_{\text{H-H}'} = 15.8 \text{ Hz}^{15}$ and  $\delta$  2.46 (CDCl₃).¹⁶ The large value for  $J_{\text{H-H}'}$  compared to that for CH₂=CH₂ (2.2 Hz) was attributed to favorable hyperconjugative interactions of the C–H bonds in ketene with the carbonyl  $\pi$  bond.¹⁵ The ¹³C NMR shifts are  $\delta$  194.0 for C₁ (the carbonyl carbon) and 2.5 for C₂,  $J_{\text{C-H}} = 171.5 \text{ Hz}.^{17}$  The finding that the dipole moment of 1.45 D was less than those of formaldehyde and acrolein (CH₂=CHCH=O) of 2.27 and 3.04 D, respectively, led to the proposal of the importance of the resonance structure **3a** for ketene, with negative charge on C₂.^{18,19} The remarkably high field ¹³C chemical shift of this carbon gives vivid evidence of this effect. The distinctive ketenyl IR band was also found to vary with the medium, ranging from 2151, 2142, and 2133 cm⁻¹ in the vapor, the argon matrix, and the solid, respectively.²⁰ The absorption of ketene in the Raman spectrum was observed at 2150 cm⁻¹.²¹ The UV spectrum displayed the C=C  $\pi \rightarrow \pi^*$  band at 183 nm, the C=O  $\pi \rightarrow \pi^*$  band at 215 nm, and the n  $\rightarrow \pi^*$ band at 325 nm.²²



On standing ketene forms the  $\beta$ -lactone dimer **5**, known as diketene, as the major product (equation 5). The debate about the structure of this material lasted for more than 40 years, and was finally settled as the structure **5** by a combination of chemical studies,²³ electron diffraction,²⁴ microwave spectroscopy,²⁵ and X-ray crystallography.^{26,27} The  $\beta$ -lactone dimer **5** is a valuable chemical intermediate.^{28,29} The dimerization was examined computationally,^{30–32} and the diketone dimer **6** was calculated to be 1 kcal/mol more stable than the  $\beta$ -lactone **5**,³⁰ but as noted below, a later computation reversed this order. The unknown vinyl ether dimer **7** was predicted to be 32 kcal/mol less stable but capable of existence.³⁰ The diketone





Dimerization in the absence of catalysts forms the isomeric dimers **5** and **6** in a 95:5 ratio.³⁴ This process has been studied further using *ab initio* molecular dynamics calculations, which led to the conclusion that the formation of the  $\beta$ -lactone dimer **5** is the result of thermodynamic control and that the formation of 1,3-cyclobutanone **6** has a lower barrier (equation 5).³⁵ Formation of the  $\beta$ -lactone dimer **5** has a calculated barrier of 31.5 kcal/mol and a free energy change of -5.5 kcal/mol, while formation of 1,3-cyclobutanedione **6** has a barrier of 26.4 kcal/ mol and a free energy change of -4.5 kcal/mol.³⁵ This is contrary to the results of previous calculations, which predicted a lower barrier for forming **5** and greater stability of **6** compared to **5**.^{30,36} In view of the known preference for formation of **5** it appears this is more stable, and has a lower barrier to its formation.

All ketenes react with nucleophiles, and the volatile parent is especially obnoxious and toxic.^{13,14,37} This compound should be regarded as highly dangerous and handled with care in a well-ventilated fume hood.

Pyrolysis of acetone is an alternative laboratory procedure for the preparation of ketene, with heating of the acetone with burners in a glass tube^{38,39} or with a metal filament (Hurd lamp).⁴⁰ This process occurs through a free radical chain mechanism involving the radicals **9a** and **9b** (equations 7, 8).

$$\begin{array}{c} O \\ CH_3 \\ CH_3$$

$$\begin{array}{c} O \\ CH_{3} \\ CH_{3} \\ CH_{3} \\ CH_{4} \\ CH_{4} \\ CH_{4} \\ CH_{3} \\ CH_{2} \\ CH_{2} \\ H \\ CH_{2} \\ H \\ CH_{2} \\ H \\ CH_{3} \\ CH_{2} \\ H \\ CH_{3} \\ CH_$$

Alkylketenes usually have a light green or yellow color, and for *tert*-butylketene (10) this may be ascribed to the UV  $\lambda_{max}$  at 360 nm due to the  $n-\pi^*$  transition of the carbonyl.²² The C=C absorption of alkylketenes is strong near 220 nm.^{40a}

CI

Monoalkylketenes typically undergo dimerization, and both methylene- $\beta$ -lactone dimers and 1,3-diketone dimers are known. For methylketene (**11**) the structures of the lactone dimer **12** and the enolized diketone dimer **13** were established in 1950 by Woodward and Small (equation 9).^{40b} Lactone dimers are formed by dehydrochlorination of acyl chlorides with triethylamine (equation 10),^{41–44} and it has been suggested⁴⁵ that their formation is catalyzed by triethylamine hydrochloride, whereas the uncatalyzed reaction leads to 1,3-diketone dimers. Dimerization of methylketene (**11**)^{46–48} with a chiral catalyst has been carried out with high stereoselectivity (Section 5.9).⁴⁸ The  $\beta$ -lactone dimers made from long-chain fatty acids have widespread application in paper sizing.¹³

Cyclopropylideneketene (14) is a highly reactive species that has been observed in a solid argon matrix.⁴⁹ Photolysis of cyclopropanecarbonyl chloride in the matrix showed the presence of 14 by IR absorption at 2135 and 2154 cm⁻¹, and further photolysis caused dissociation to the carbene and CO, which reverted to the ketene thermally (equation 11).⁴⁹ Photolysis of the anthracene derivative 15 gave 14 in high yield (equation 11a).⁵⁰ Dehydrohalogenation with highly reduced metals also gave 14, which was not observed but formed the dimer 15 (equation 12).⁵¹ Dehalogenation of 16 with Zn/Ag gave tetramethylcyclopropylideneketene 17, characterized as the dimer 18 (equation 13).^{52,53}





In contrast to the highly reactive 14, di-*tert*-butylketene (19) is stable indefinitely as a neat liquid⁵⁴ and undergoes hydration rather slowly in pure water (equation 14),⁵⁵ although the reaction is catalyzed by acid.⁵⁶ The difference in the C–C–C bond angles of 19 and 14 is reflected in the difference in their ketenyl IR spectra, at 2092 cm⁻¹ in the matrix⁵⁷ and 2090 cm⁻¹ (film)⁵⁸ for 19 and at 2135 and 2154 cm⁻¹ for 14.⁴⁹ The low frequency for 19 is similar to that for octamethyleneketene (20) of 2095 cm⁻¹ in CCl₄.⁵⁹ However, while 19 is apparently unreactive in air, 20 is readily oxidized to the ketone (equation 15).⁵⁹



Di-*tert*-butylketene has been prepared by ultrasound-assisted dehydrochlorination of the acyl chloride with  $\text{Et}_3\text{N}$ ,⁶⁰ dehydrochlorination with *n*-Bu₃N (equation 16),⁵⁸ dehalogenation of the  $\alpha$ -chloroacyl bromide,⁶¹ and dehydration of the acid with dicyclohexylcarbodiimide in a procedure also used to form di-1-adamantylketene **21** (equation 17).⁶² Although preparation of **19** by the photochemical Wolff rearrangement in solution was inefficient, this was successful in the gas phase, in 40% yield, with the IR absorption at 2100 cm⁻¹ (equation 24, vide infra).⁶³ This ketene has served as a valuable substrate for spectroscopic studies and for measurement of reactivity, including addition of electrophiles,⁵⁶ nucleophiles,⁵⁵ and an imine in [2 + 2] formation of a  $\beta$ -lactam.⁶⁴

$$\overset{t-\mathrm{Bu}}{\longrightarrow} \overset{\mathrm{O}}{\underset{\mathrm{Cl}}{\longrightarrow}} + n \cdot \mathrm{Bu}_{3}\mathrm{N} \xrightarrow{\mathrm{80 \, ^{\circ}\mathrm{C}}} \overset{t-\mathrm{Bu}}{\longrightarrow} \overset{t-\mathrm{Bu}}{\underset{t-\mathrm{Bu}}{\longrightarrow}} \mathrm{C=O}$$
(16)

$$\begin{array}{cccc}
R & \xrightarrow{DCC} & R & \xrightarrow{DC} & C=O \\
R & & 19, R = t-Bu (70\%) \\
& 21, R = 1-Adamantyl (74\%)
\end{array}$$
(17)

Di-*tert*-butylketene (19) reacted with perfluoroalkyl radicals at  $C_1$ , giving radicals that from their ESR spectra are interpreted as having perpendicular geometries 22 (equation 18).⁶⁵ Adducts of hydrogen atoms derived from HI adding to both  $C_1(23)$  and  $C_2$  (24) of the ketene were also observed.⁶⁵ The radical 23 had a planar structure, and in 24 the C-H bond at  $C_2$  was perpendicular to the singly occupied orbital (equation 18).⁶⁵

$$\begin{array}{c} t - Bu \\ t - Bu \end{array} = C = O \xrightarrow[CF_3]{R_f = C_6F_5} t - Bu \xrightarrow[CF_3]{O} \\ 19 \end{array} \xrightarrow{t - Bu \xrightarrow{r} t - Bu$$

Reaction of ozone with di-*tert*-butylketene (**19**) at -78 °C gave a product in solution assigned as the  $\alpha$ -lactone **25** on the basis of the ¹H NMR signal at  $\delta$  1.2 and the formation of polyester **26** upon warming (equation 19).⁶⁶



The persistent bicyclic ketene 27 upon photolysis in hydrocarbon solvents gave the tricyclane 29, evidently via decarbonylation to the carbene 28 and intramolecular insertion (equation 20).⁶⁷ Photolysis of 27 in CH₃OD at -60 °C formed 29 and the deuterated aldehyde 32 in a 1:1 ratio. The formation of 32 implicated photoprotonation of the ketene on the carbonyl carbon forming 30 and rearrangement to the carbocation 31 (equation 21).⁶⁷



Ketene **27** was prepared by dehydrochlorination, as described in Section 3.2.2,⁶⁷ as well as by dehalogenation (equation 22).⁶⁸ The crowded camphorketene (**33**)⁶⁷ and **34** (equation 23)⁶⁹ have also been prepared by zinc dehalogenation. The latter ketene was reacted with phenyllithium and gave stereoselective addition from the least hindered side, forming **35** (equation 23).⁶⁹



The low yield of di-*tert*-butylketene from thermolysis of diazo ketone **36** in which methyl migration forming the alkene **37** predominated arises from the unfavorable geometry for Wolff rearrangement in the conformation **36** in which the potential migrating group is *syn* to the departing N₂ (equation 24),⁷⁰ in contrast to the efficient formation of **39** from **38**, in which the migrating group is *anti* (equation 25).⁷¹ Gas phase photolysis of *tert*-butyl *tert*-butyldiazomethyl ketone **36** gave di-*tert*-butylketene **19** in an improved 40% yield, a result attributed to different confomational populations in the gas phase, but the efficiency of product formation was still only 1/300th that of formation of **39**.⁶³



Thermal Wolff rearrangement of the thiadiazocycloheptanone **40** formed the ketene **41**, which is persistent at 20 °C, and was identified by its ketenyl IR absorption (equation 26).⁷² The lability of the diazothiacycloheptanone was ascribed to twisting of the ring, which prevented conjugation in the diazo ketone grouping.



 $\alpha$ -Ketoketenes may undergo decarbonylation, and matrix photochemical decarbonylation at 12 K of  $\alpha$ -ketoketene **42** generated an  $\alpha$ -keto carbene **43**, which underwent Wolff rearrangement to cyclobutylideneketene **44** (equation 27).⁵⁷ The ketene was identified by the IR doublet at 2098 and 2150 cm⁻¹ and was captured with methanol at 100–140 K, giving methyl cyclobutanecarboxylate.⁵⁷ Ketene **44** was also generated by Wolff rearrangement of diazocyclopentanone and by pyrolytic dehydrochlorination.⁵⁷

Microwave irradiation was found to enhance the efficiency of thermal Wolff rearrangement, forming a variety of alkyl- and arylketenes.⁷³ Thus, diazocamphor (**45**)^{74–76} or pinene-derived diazo ketones⁷⁷ underwent Wolff rearrangement upon photolysis,⁷⁵ but thermally or with transition metal catalysis **45** formed the tricyclanone **46** (equation 28).⁷³ However, with microwave irradiation in the presence of benzylamine, **45** underwent Wolff rearrangement to ketene **47**, which yielded **48** (equation 29).⁷³



Photolysis of the 7,7-diarylbicyclo[2.2.1]hept-2-enes **49** gave normal Wolff rearrangement forming the ketenes **50**, which led to esters **51** from capture by MeOH in 34% yield from either isomer of **49**, and a preference for formation of *exo*-**51** (equation 30).⁷⁸ A second pathway was the vinylogous Wolff rearrangement via the transition state **52** forming ketenes **53**, which gave esters **54**, as well as some tricylane product (equation 31).⁷⁸





Photolysis of the diazo ketone **55** gave a 95% crude yield of an extraordinarily stable ketene **56**, IR 2115 cm⁻¹, UV  $\lambda_{max}$  (cyclohexane), 222 nm ( $\epsilon$  1260), that was stable to extraction with 1 *M* KOH but was hydrolyzed to the acid by heating with aqueous HClO₄ (equation 32).⁷⁹



The adamantylketenes **57** and **58** were prepared by dehydrochlorination of the corresponding acyl chlorides, and **59** by dehalogenation of the  $\alpha$ -bromo acyl chloride with zinc, and their ketenyl IR absorptions were observed in solution.⁸⁰ Derivatives of **57–59** were prepared, including the dimer of **59**.⁸⁰



Thermal intramolecular cyclization of ketene **60** generated from the acyl chloride gave cyclobutanone **61** (equation 33), which after chloride was exchanged for bromide, and then was converted via radical cyclization to **62**, followed by ring expansion to bicyclic ketone **63** (equation 34).^{80a}





Methylketene (4) generated by dehydrochlorination reacted with allyl morpholines 64, forming amides 66 with high selectivity (equations 35, 36).^{81,81a} The reactions were envisaged as proceeding through a ketene aza-Claisen reaction through 65, and were successful with catalysis by  $5-10 \mod\%$  Yb(OTf)₃, AlCl₃, Ti(OPr-*i*)₂Cl₂, and TiCl₄•(THF)₂.



Direct photolysis of diazo ketone **67** in the presence of ethyl vinyl ether gave cyclohexylketene **68**, which reacted by [2 + 2] cycloaddition to form the cyclobutanone **69** (equation 37), whereas metal-mediated reactions of **67** gave cyclopropyl ketone products not derived from ketenes.⁸²



Dimethylketene (1) generated by dehydrochlorination gave [2+2] cycloaddition with cyclopentadiene to form the intermediate **70** used in leukotriene synthesis (equation 38).⁸³

$$\stackrel{\text{Me}}{\longrightarrow} C=O + \bigoplus \stackrel{\text{CHCl}_3}{\longrightarrow} \bigwedge \stackrel{O}{\longrightarrow} Me$$

$$1 \qquad 70 (77\%) \stackrel{\text{Me}}{\longrightarrow} Me$$

$$(38)$$

Photolysis of diazo ketone **71** as a monolayer on a gold surface upon photolysis gave the ketene **72**, which was trapped by  $CH_3OH$ , leading to an IR absorption indicating formation of the ester (equation 39).⁸⁴

$$HS \begin{pmatrix} 0 \\ 1 \\ 5 \\ 71 \end{pmatrix} \stackrel{N_2}{\longrightarrow} HS \begin{pmatrix} 0 \\ 1 \\ 5 \\ 72 \end{pmatrix} \stackrel{N_2}{\longrightarrow} (39)$$

Photochemical Wolff rearrangement of the diazo ketone **73** adsorbed on a gold surface through the sulfur atoms gave ketene **74**, which was captured by water as the acid **75** (equation 40).⁸⁵ The ketene was also captured as the amide with *n*-octa-decylamine and as a photoluminescent ester by capture with 1-hydroxymethylpyr-ene.⁸⁵



#### **REFERENCES FOR SECTION 4.1.1**

- 1. Staudinger, H.; Klever, H. Chem. Ber. 1906, 39, 968-971.
- 2. Gong, L.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1991, 113, 6021-6028.
- 3. McAllister, M. A.; Tidwell, T. T. J. Org. Chem. 1994, 59, 4506-4515.
- 4. Wilsmore, N. T. M. J. Chem. Soc. 1907, 91, 1938-1941.
- 5. Fisher, G. J.; MacLean, A. F.; Schnizer, A. W. J. Org. Chem. 1953, 18, 1055-1057.
- 6. Staudinger, H.; Klever, H. W. Chem. Ber. 1908, 41, 594-600.
- 7. Wilsmore, N. T. M.; Stewart, A. W. Chem. Ber. 1908, 41, 1025-1027.
- 8. Staudinger, H.; Klever, H. W. Chem. Ber. 1908, 41, 1516-1517.
- 9. Gezelter, J. D.; Miller, W. H. J. Chem. Phys. 1995, 103, 7868-7876.
- 10. Wilson, P. J.; Tozer, D. J. Chem. Phys. Lett. 2002, 352, 540-544.
- Brown, R. D.; Godfrey, P. D.; McNaughton, D.; Pierlot, A. P.; Taylor, W. H. J. Mol. Spec. 1990, 140, 340–352.
- 12. Duncan, J. L.; Munro, B. J. Mol. Struct. 1987, 161, 311-319.
- Miller, R.; Abaecherli, C.; Said, A.; Jackson, B. Ullmann's Encyclopedia of Industrial Chemistry, 6th ed.; 2003, Vol. 18, pp. 717–732.
- 14. Mitzel, T. M., Encyl. Reagents Org. Synth., Paquette, L., Ed.; pp. 2929-2932.
- 15. Allred, E. L.; Grant, D. M.; Goodlett, W. J. Am. Chem. Soc. 1965, 87, 673-674.
- 16. Masters, A. P.; Sorensen, T. S.; Ziegler, T. J. Org. Chem. 1986, 51 3558-3559.
- Firl, J.; Runge, W. Z. Naturforsch. 1974, 29b, 393–398. Angew. Chem., Int Ed. Engl. 1973, 12, 668–669.

- 18. Hannay, N. B.; Smyth, C. P. J. Am. Chem. Soc. 1946, 68, 1357-1360.
- 19. Price, W. C.; Teegan, J. P.; Walsh, A. D. J. Chem. Soc. 1951, 920-926.
- 20. Moore, C. B.; Pimentel, G. C. J. Chem. Phys. 1963, 38, 2816-2829.
- 21. Kopper, H. Zeit. Phys. Chem. 1936, B34, 396-401.
- 22. Rabalais, J. W.; McDonald, J. M.; Scherr, V.; McGlynn, S. P. Chem. Revs. 1971, 71, 73–108.
- 23. Johnson, J. R.; Shiner, V. J., Jr. J. Am. Chem. Soc. 1953, 75, 1350-1355.
- 24. Bregman, J.; Bauer, S. H. J. Am. Chem. Soc. 1955, 77, 1955-1965.
- 25. Monig, F.; Dreizler, H.; Rudolph, H. D. Zeit. Naturforshch. A, 1967, 22A, 1471-1473.
- 26. Katz, L.; Lipscomb, W. N. J. Org. Chem. 1952, 17, 515-517.
- 27. Kay, M. C.; Katz, L. Acta Cryst. 1958, 11, 897-898.
- 28. Clemens, R. J. Chem. Revs. 1986, 86, 241-318.
- 29. Kato, T. Acc. Chem. Res. 1974, 7, 265-271.
- 30. Seidl, E. T.; Schaeffer, H. F., III. J. Am. Chem. Soc. 1990, 112, 1493-1499.
- 31. Seidl, E. T.; Schaeffer, H. F., III. J. Am. Chem. Soc. 1991, 113, 5195-5200.
- Schaad, L. J.; Gutman, I.; Hess, B. A., Jr.; Hu, J. J. Am. Chem. Soc. 1991, 113, 5200–5203.
- 33. Van Daalen, J. J.; Kraak, A.; Arens, J. F. Recl. Trav. Chim. Pays Bas 1961, 80, 810–818.
- 34. Tenud, L.; Weilmann, M.; Dallwigk, E. Helv. Chim. Acta 1977, 60, 975-977.
- 35. Kelly, E.; Seth, M.; Ziegler, T. J. Phys. Chem. A 2004, 108, 2167-2180.
- 36. Salzner, U.; Bachrach, S. M. J. Am. Chem. Soc. 1994, 116, 6850-6855.
- 37. Wooster, H. A; Lushbaugh, C. C.; Redeman, C. E. J. Am. Chem. Soc. 1946, 68, 2743.
- 38. Hurd, C. D. J. Am. Chem. Soc. 1923, 45, 3095-3101.
- 39. Hurd, C. D. Org. Synth., Col. Vol. I, 1941, 330-334.
- 40. Williams, J. W.; Hurd, C. D. J. Org. Chem. 1940, 5, 122-125.
- 40a. Holroyd, R. A.; Blacet, F. E. J. Am. Chem. Soc. 1957, 79, 4830-4834.
- 40b. Woodward, R. B.; Small, G., Jr. J. Am. Chem. Soc. 1950, 72, 1297-1304.
- 41. Sauer, J. C. J. Am. Chem. Soc. 1947, 69, 2444-2448.
- 42. Hurd, C. D.; Blanchard, C. A. J. Am. Chem. Soc. 1950, 72, 1461-1462.
- 43. Hanford, W. E.; Sauer, J. C. Org. React. 1946, 3, 108-140.
- Sauer, J. C. Organic Syntheses. Coll. Vol. IV, Rabjohn, N., Ed.; Wiley, New York, pp. 560–563.
- 45. Farnum, D. G.; Johnson, J. R.; Hess, R. E.; Marshall, T. B.; Webster, B. J. Am. Chem. Soc. 1965, 87, 5191–5197.
- 46. Cortez, G. S.; Oh, S. H.; Romo, D. Synthesis 2001, 1731-1736.
- 47. Jenkins, A. D. J. Chem. Soc. 1952, 2563-2568.
- 48. Orr, R. K.; Calter, M. A. Tetrahedron 2003, 59, 3545-3565.
- Monnier, M.; Allouche, A.; Verlaque, P.; Aycard, J.-P. J. Phys. Chem. 1995, 99, 5977–5985.
- 50. Ripoll, J. L. Tetrahedron, 1977, 33, 389-391.
- 51. Masters, A. P.; Sorensen, T. S. Tetrahedron Lett. 1989, 30, 5869-5872.

- 52. Hoffmann, H. M. R.; Walenta, A.; Eggert, U.; Schomberg, D. Angew. Chem. Int. Ed. Engl. 1985, 24, 607–608.
- 53. Hoffmann, H. M. R.; Geschwinder, P. M.; Hollwege, H.-P.; Walenta, A. *Helv. Chem.* Acta **1988**, *71*, 1930–1936.
- 54. Newman, M. S.; Arkell, A.; Funkunga, T. J. Am. Chem. Soc. 1960, 82, 2498-2501.
- 55. Allen, A. D.; Tidwell, T. T. J. Am. Chem. Soc. 1987, 109, 2774-2700.
- 56. Kabir, S. H.; Seikaly, H. R.; Tidwell, T. T. J. Am. Chem. Soc. 1979, 101, 1059-1060.
- 57. Leung-Toung, R.; Wentrup, C. J. Org. Chem. 1992, 57, 4850-4858.
- Back, T. G.; Barton, D. H. R.; Britten-Kelly, M. R.; Guziec, F. S., Jr. J. Chem. Soc., Perkin Trans. 1 1976, 2079–2089.
- 59. Concannon, P. W.; Ciabattoni, J. J. Am. Chem. Soc. 1973, 95, 3284-3289.
- 60. Olah, G.; Wu, A.; Farooq, O. Synthesis 1989, 566-567.
- 61. Hofman, P.; Moya, L. A. P.; Kain, I., Synthesis 1986, 43-44.
- 62. Olah, G.; Wu, A.; Farooq, O. Synthesis 1989, 568.
- Marfisi, C.; Verlaque, P.; Davidovics, G.; Pourcin, J.; Pizzala, L.; Aycard, J.-P., Bodot, H. J. Org. Chem. 1983, 48, 533–537.
- 64. Schaumann, E. Chem. Ber. 1976, 109, 906-921.
- 65. Malatesta, V.; Forrest, D.; Ingold, K. U. J. Phys. Chem. 1978, 82, 2370-2373.
- 66. Wheland, R.; Bartlett, P. D. J. Am. Chem. Soc. 1970, 92, 6057-6058.
- 67. Kirmse, W.; Spaleck, W. Angew. Chem. Int. Ed. Engl. 1981, 20, 776-777.
- Shustov, G. V.; Sun, F.; Sorensen, T. S.; Rauk, A. J. Chem. Soc. Perkin Trans. 2 1999, 1035–1041.
- 69. Baigrie, L. M.; Seikaly, H. R.; Tidwell, T. T. J. Am. Chem. Soc. 1985, 107, 5391-5396.
- 70. Newman, M. S.; Arkell, A. J. Org. Chem. 1959, 24, 385-387.
- 71. Kaplan, F.; Mitchell, M. L. Tetrahedron Lett. 1979, 759-762.
- 72. de Groot, A.; Boerma, J.; de Valk, J.; Wynberg, H. J. Org. Chem. 1968, 33, 4025-4029.
- Sudrik, S. G.; Chavan, S. P.; Chandrakumar, K. R. S.; Pal, S.; Date, S. K.; Chavan, S. P.; Sonawane, H. R. J. Org. Chem. 2002, 67, 1574–1579.
- 74. Horner, L.; Spietschka, E. Chem. Ber. 1955, 88, 934-939.
- 75. Rau, H.; Bokel, M. J. Photochem. Photobiol. A: Chem. 1990, 53, 311-322.
- 76. Meinwald, J.; Lewis, A.; Gassman, P. G. J. Am. Chem. Soc. 1962, 84, 977-983.
- 77. Meinwald, J.; Gassman, P. G. J. Am. Chem. Soc. 1960, 82, 2857-2863.
- 78. Zimmerman, H. E.; Wang, P. Can. J. Chem. 2003, 81, 517-524.
- 79. Dauben, W. G.; Friedrich, L. E. J. Org. Chem. 1972, 37, 241-250.
- 80. Strating, J.; Scharp, J.; Wynberg, H. Recl. Trav. Chem. Pays-Bas 1970, 89, 23-31.
- 80a. Zhang, W.; Collins, M. R.; Mahmood, K.; Dowd, P. Tetrahedron Lett. 1995, 36, 2729–2732.
- 81. Yoon, T. P.; Dong, V. M.; MacMillan, D. W. C. J. Am. Chem. Soc. 1999, 121, 9726–9727.
- 81a. Dong, V. M.; MacMillan, D. W. C. J. Am. Chem. Soc. 2001, 123, 2448-2449.
- 82. Lund, E. A.; Kennedy, I. A.; Fallis, A. G. Can. J. Chem. 1996, 74, 2401-2412.
- 83. Rey, M.; Roberts, S.; Dieffenbacher, A.; Dreiding, A. S. *Helv. Chim. Acta* **1970**, *53*, 417–432.
- 84. Jocys, G. J.; Workentin, M. S. Chem. Commun. 1999, 839-840.

#### 206 TYPES OF KETENES

 Hu, J.; Liu, Y.; Khemtong, C.; El Khoury, J. M.; McAfoos, T. J.; Taschner, I. S. *Langmuir* 2004, 20, 4933–4938.

#### 4.1.2 Alkenylketenes

Vinylketene (1) is the parent alkenylketene, and was generated by pyrolysis of vinylacetic acid or crotonic acid.¹ The dipole moment measured in the gas phase indicated that the *anti* conformation was favored (equation 1).² Computations confirmed that this was more stable than the *syn* conformation by 1.7 kcal/mol.^{3–5} The gas phase photoelectron spectrum of **1** has also been measured,¹ as well as the ¹³C NMR spectrum of the ketene formed by pyrolysis and collected at -196 °C in CS₂/ CDCl₃.⁶ This showed at -70 °C the ¹³C shift of the CH₂ at  $\delta$  109, consistent with negative charge delocalization to the terminal CH₂, as shown in **1a**.⁶

The vinylketene **1** and *E*-(1,3-butadienyl)ketene *E*-**2**⁷ have been generated by dehydrochlorination of the acyl chlorides and directly observed by IR.⁷ The same procedure gave the bis(alkenyl)ketene **3**⁸ using 1,8-bis(dimethylamino)-naphthalene as a stoichiometric base and Et₃N as a catalytic shuttle base (see Section 3.2.2).⁹ *Z*-(1,3-Butadienyl)ketene *Z*-**2** has been formed by photolysis of 2,4-cyclohexadienone **4** and identified by its IR¹⁰ and UV^{10,11} spectra (equation 2).



Flash vacuum pyrolysis of the hexadienoyl chloride **5** was reported to form an equilibrating mixture of E/Z-**2** (equation 3), as evidenced by the formation of phenol (**6**) and the ether **7** (equation 4).¹²





Computations at the MP2/6-31G^{*}//MP2/6-31G^{*} level using an isodesmic energy comparison are consistent with essentially equal stabilization of vinylketene (1) in comparison to methylketene (8), and with essentially equal conjugative stabilization in comparison to that of 1,3-butadiene (equation 5).^{3–5} The vinyl substituent also enhances the reactivity of alkenylketenes, which are quite reactive toward many reagents. Trimethylsilyl substituents stabilize silylketenes, as they do other ketenes, and  $9^{13}$  and  $10^{14}$  are long-lived. The crowded ketene 11 was isolable and resistant to dimerization, but was reactive toward electrophiles and nucleophiles.¹⁵



Dehydrochlorination of crotonyl chloride by triethylamine formed vinylketene **1**, which was trapped in situ by cyclopentadiene in a [2 + 2] cycloaddition forming the bicyclo[3.2.0]heptene **12** (equation 6).^{16,17} In the absence of trapping agents, dimerization to 2-pyrones **13** occurred (equation 7).^{16,17}


The reaction of  $\beta$ , $\beta$ -dimethylacrylyl chloride (14) with triethylamine in CH₂Cl₂ at -78 °C was proposed to lead to the unobserved alkenylketene 15, which was trapped in situ by geraniol, forming 16 (equation 8).¹⁸ A similar reaction is shown in Section 3.2.2. The optically active and long-lived alkenylketene 18 was prepared by dehydrohalogenation of 17 (equation 9).¹⁹ Chiroptical properties of ketenes are discussed in Section 2.2.



Photochemical Wolff rearrangement of diazo ketones **19** gave a general route to the formation of  $\alpha$ -trialkylsilylalkenylketenes **20** (equation 10).^{20–22} The reactions of **20** with diazoalkanes yielded cyclopentenones **21** (equation 10).²⁰ Alkenes and alkynes reacted by [4 + 2] cycloadditions with **22** formed in this way to give cyclohexenones **23** and phenols **24**, respectively (equations 11, 12)²¹ Imines reacted with **22** by [4 + 2] cycloadditions to give  $\delta$ -lactams **25** (equation 13).²² Other reactions of **20** and some analogues are discussed in Section 4.5.









Additional reactions for formation of vinylketenes are discussed elsewhere, including Wolff rearrangements (Section 3.3), cyclobutenone and cyclohexadienone ring openings (Sections 3.4.1 and 3.4.3, respectively), and reactions of metal carbene complexes (Section 3.5). Ene reactions of dienylaldehyde **26** formed **27** (equation 14²³ and Sections 3.4.6 and 5.4.4), and thermolysis of Meldrum's acid derivative **28** formed **29**, which gave the cumulenone **30** (equation 15), which was converted through the ynol **31** to vinylketene **32** (equation 16).²⁴ The parent **1** was also formed from the gas phase thermolysis of ene-yne ethyl ether **33** (equation 17),²⁵ and vinylketenes were reported to be formed in the flash vacuum pyrolysis of spiro[2,3]hexan-4-ones based on the products from trapping with MeOH.²⁶



$$CH_2 = CH - C \equiv C - OEt \xrightarrow{\Delta} CH_2 = C = O$$

$$(17)$$

The barriers to rearrangements of vinylketenes **34** to acylallenes **36** have been calculated by *ab initio* methods, and the thermal migration of methoxy through transition state **35** has been shown experimentally (equation 18).^{27,28} Allenyl amide **37** rearranged similarly, together with partial elimination upon flash vacuum thermolysis, to form a mixture of alkenylketenes **1** and **38** (equation 19).²⁸ The unobserved vinylketene **39** generated by a similar route rearranged to the observed **40** (equation 20).²⁹





Calculations of the conformations of the 2-pyranone (**41**) photolysis product 5-oxo-2,4-pentadienal (**42**) and its cyclization to **43** indicated that the conformer **42a** was not a stable structure, but would close without a barrier to **41** (equation 21).³⁰ Pyrolysis of the 2-oxopyran-5-carboxaldehyde **44** led to an equilibrium mixture of the 2-pyridones **44**, **47**, and **49** involving the acylketenes **45**, **46**, **46a**, and **48** as unobserved intermediates (equations 22, 23).³¹



The cyclobutenedione **50** was proposed to react with RCLi(CN)OTBDMS, forming **51**, which underwent ring opening to the vinyl ketene **52** followed by cyclization to the 2-pyranone **53** (equations 24, 25).³² Other examples of cyclobutenone ring openings forming vinylketenes are given in Section 3.4.1.





Matrix photolysis of the diazo ketone **54** gave the carbene **55** and the strained cyclopropene **56**, which on further photolysis gave the ene-yne ketene **57**, as identified by the matrix IR spectrum (equation 26).^{33,34} The ring closures of the ene-yne ketene **57** through the Schmittel and Myers-Saito pathways, forming the diradical intermediates **58** and **59**, respectively, (see Section 3.4.1) have been studied computationally (equation 27).³⁵



Flash vacuum pyrolysis (FVP) of the ester **60** was proposed to give the ketene **61**, which cyclized to **62** (equation 28).³⁶



Vinylketene **1** generated by dehydrohalogenation gave [4+2] cycloaddition reactions with 1,3-diaza-1,3-butadienes, forming **63** (equation 29).³⁷ Wolff rearrangement gave **64**, which reacted by a [4+2] cycloaddition pathway with an eneamine, forming **65** (equation 30).³⁸





Thermolysis of **66** was proposed to give the ketene **67**, which underwent cyclization and dehydrogenation to give the observed **68** (equation 31).³⁹ Photolysis of the silylated *para*-benzoquinone **69** gave the ketene **70** as a yellow solid, IR 2089 cm⁻¹ (equation 32).⁴⁰





Carbonylation of the triflate **71** using  $Pd(PPh_3)_4$  was proposed to form the ketene **72**, which cyclized to the cyclobutenone **73** (equation 33).⁴¹ Heating **73** in MeOH gave the ring-opened ester in 80% yield.



Reaction of the ene-yne-substituted epoxide **74** with the ruthenium catalyst TpRuPPh₃(CH₃CN)PF₆ [Tp=tris(1-pyrazolyl)borate] reacted through the Ru⁺-coordinated dienylketene **75**, which cyclized to the dienone **76** (equation 34).⁴² Several similar examples were also reported.⁴² This method is discussed further in Section 4.8.4.



Examples of thermolysis of substituted analogues of vinylcyclobutenone 77 forming divinylketene 78 (equation 35) are shown in Section 3.4.1.



## **REFERENCES FOR SECTION 4.1.2**

- 1. Mohmand, S.; Hirabayashi, T.; Bock, H. Chem. Ber. 1981, 114, 2609-2621.
- 2. Brown, R. D.; Godfrey, P. D.; Woodruff, M. Austr. J. Chem. 1979, 32, 2103-2109.
- 3. Gong, L.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1991, 113, 6021-6028.
- Allen, A. D.; Andraos, J.; Kresge, A. J.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1992, 114, 1848–1879.
- 5. McAllister, M. A.; Tidwell, T. T. J. Org. Chem. 1994, 59, 4506-4515.
- Trahanovsky, W. S.; Surber, B. W.; Wilkes, M. C.; Preckel, M. M. J. Am. Chem. Soc. 1982, 104, 6779–6781.
- 7. Tidwell, T. T.; Fenwick, M. T. Eur. J. Org. Chem. 2001, 3415-3419.
- 8. Saidi, K.; Tidwell, T. T. Arkivoc 2001, (xii) 100-107.
- Taggi, A. E.; Hafez, A. M.; Wack, H.; Young, B.; Ferraris, D.; Lectka, T. J. Am. Chem. Soc. 2002, 124, 6626–6635.
- Jerina, D. M.; Witkop, B.; McIntosh, C. L.; Chapman, O. L. J. Am. Chem. Soc. 1974, 96, 5578–5580.
- 11. Capponi, M.; Gut, I.; Wirz, J. Angew. Chem. Int. Ed. Engl. 1986, 25, 344-345.
- 12. Dehmlow, E. V.; Slopianka, M. Angew. Chem. Int. Ed. Engl. 1979, 18, 170.
- 13. Danheiser, R. L.; Sard, H. J. Org. Chem. 1980, 45, 4810-4812.
- 14. Tang, P. C.; Wulff, W. D. J. Am. Chem. Soc. 1984, 106, 1132-1133.
- 15. Allen, A. D.; Stevenson, A.; Tidwell, T. T. J. Org. Chem. 1989, 54, 2843-2848.
- 16. Jackson, D. A.; Rey, M.; Dreiding, A. S. Helv. Chim. Acta 1983, 66, 2330-2344.
- 17. Jackson, D. A.; Rey, M.; Dreiding, A. S. Tetrahedron Lett. 1983, 24, 4817-4820.
- 18. Corey, E. J.; Roberts, B. E.; Dixon, B. R. J. Am. Chem. Soc. 1995, 117, 193-196.
- 19. Chapuis, C.; Brauchli, R. Helv. Chim. Acta 1993, 76, 2070-2088.
- Loebach, J. L.; Bennett, D. M.; Danheiser, R. L. J. Org. Chem. 1998, 63, 8380– 8389.
- 21. Loebach, J. L.; Bennett, D. M.; Danheiser, R. L. J. Am. Chem. Soc. 1998, 120, 9690–9691.
- 22. Bennett, D. M.; Okamoto, I.; Danheiser, R. L. Org. Lett. 1999, 1, 641-644.

- 23. Schiess, ; Rademerski, P. Angew. Chem. Int. Ed. 1972, 11, 288–289. Helv. Chim. Acta 1974 57, 2583–2597.
- 24. Wentrup, C.; Lorencak, P. J. Am. Chem. Soc. 1988, 110, 1880-1883.
- 25. Terlouw, J. K.; Burgers, P. C.; Holmes, J. L. J. Am. Chem. Soc. 1979, 101, 225-226.
- 26. Rousseau, G.; Bloch, R.; Le Perchec, P.; Conia, J. M. J. Chem. Soc., Chem. Commun. 1973, 795–796.
- 27. Bibas, H.; Wong, M. W.; Wentrup, C. J. Am. Chem. Soc. 1995, 117, 9582-9583.
- 28. Bibas, H.; Wong, M. W.; Wentrup, C. Chem. Eur. J. 1997, 3, 237-248.
- 29. Bibas, H.; Koch, R.; Wentrup, C. J. Org. Chem. 1998, 63, 2619-2626.
- 30. Birney, D. M. J. Org. Chem. 1996, 61, 243-251.
- 31. Cartwright, G. A.; Gould, R. O.; McNab, H. Chem. Commun. 1997, 1293-1294.
- 32. Mingo, P. A.; Zhang, S.; Liebeskind, L. S. J. Org. Chem. 1999, 64, 2145-2148.
- 33. Bucher, G.; Sander, W. J. Org. Chem. 1992, 57, 1346-1351.
- 34. Sander, W.; Bucher, G.; Reichel, F.; Cremer, D. J. Am. Chem. Soc. 1991, 113, 5311-5322.
- Musch, P. W.; Remenyi, C.; Helten, H.; Engels, B. J. Am. Chem. Soc. 2002, 124, 1823– 1828.
- 36. McNab, H.; Thornley, C. J. Chem. soc. Perkin Trans. 1 1997, 2203-2209.
- 37. Sharma, A. K.; Mahajan, M. P. Tetrahedron 1997, 53, 13841-13854.
- 38. Collomb, D.; Doutheau, A. Tetrahedron Lett. 1997, 38, 1397-1398.
- Aliev, Z. G.; Krasnykh, O. P.; Maslivets, A. N.; Andreichikov, Y. S.; Atovmyan, L. O. *Izv. Akad. Nauk Ser. Khim.* 1997, 563–565; *Russ. Chem. Bull.* 1997, 46, 543–546.
- 40. Sakamoto, K.; Tsutsui, S.; Ebata, K.; Kabuto, C.; Sakurai, H. Chem. Lett. 2000, 226-227.
- 41. Toivola, R. J.; Savilampi, S. K.; Koskinen, A. M. P. Tetrahedron Lett. 2000, 41, 6207–6210.
- 42. Ming-Yuan, L.; Madhushaw, R. J.; Liu, R.-S. J. Org. Chem. 2004, 69, 7700-7704.

# 4.1.3 Alkynyl- and Cyanoketenes

The alkynyl substituent was calculated at the MP2/6-31G*//MP2/6-31G* level to have almost exactly the same stabilization energy as the methyl group.^{1–4} Known examples of alkynylketenes are quite reactive, and apparently no long-lived example has been reported, although these should be accessible.



Phenylethynylketene (6) formed by photochemical Wolff rearrangement of the diazo ketone 5 using flash photolysis was detected by TRIR spectroscopy by the characteristic absorption at 2132 cm⁻¹ in CH₃CN,⁵ and by the use of conventional photolysis and detection in isooctane at 2131 cm⁻¹ (equation 2).⁶ The ketene 6 was trapped by the stable free radical TEMPO (TO•), forming the adduct 8 (equa-

tion 2).⁶ Upon photolysis of **6** decarbonylation occurred, leading to the carbene **9**, which formed 1-phenylpropyne, phenylallene, and benzylacetylene (equation 3).⁷



Dehydrochlorination of the acyl chlorides **10** in the presence of cyclopentadiene led to generation of the alkynylketenes **11–15**, which gave in situ [2 + 2] cycloaddition forming the adducts **16**, with the alkynyl group preferentially in the *exo* position as the only observed products (equation 4).^{4,8} The smaller group preferentially occupied the *exo* position in these reactions, and this result illustrated the small steric requirements of the carbon-carbon triple bond.



Heating of the 1-*tert*-butoxy-1,3-butadiyne **18** gave the macrolide **20** through intramolecular capture of the intermediate alkynylketene **19** (equation 5).⁹ Analogous products were prepared by the same procedure,⁹ and another similar procedure is given in Section 3.4.5.



Thermolysis of the bicyclo[2.2.2]octadiene **21** in xylene at 138 °C was proposed to form alkynyl(trimethylsilyloxy)ketene **22**, which reacted with 1-phenylpropyne by [2 + 2] cycloaddition, forming cyclobutenone **23** (equation 6).¹⁰ Under these reaction conditions **23** underwent ring opening, forming ene-yne ketene **24**, which cyclized to cyclopentene **25** and benzoquinone **26** (equation 7) via diradical intermediates **27** and **28**, respectively.¹⁰ Formation of another ene-yne ketene is shown in Section 4.1.2.



Alkynylcyanoketenes **30** were generated by thermolysis of 2,5-dialkyl-3,6-diazidoquinones **29** and reacted with alkenes by [2+2] cycloaddition (equation 8).^{11,12} There was retention of the alkene stereochemistry, and a 4:1 preference for the *cis* orientation of the alkynyl group and the adjacent ethyl substituent.¹²



The cyano group in cyanoketene (**32**) was calculated at the MP2/ $6-31G^*//MP2/$ 6-31G* level by isodesmic comparison, as in equation 1, to have a small stabilizing energy of -0.4 kcal/mol relative to the methyl group.^{1–4} Cyanoketenes are, however, quite reactive, an attribute that arises from the small size of the substituent and the strong conjugative stabilization of transition states for reaction.

Cyanoketene (**32**) was prepared by pyrolysis of cyanoacetyl chloride (**33**) (equation 9), and the molecular structure was determined by microwave spectroscopy, showing a dipole moment of 1.542 *D* (Section 1.2).¹³ The photoelectron spectrum of **32** was also measured.^{13a} This ketene was shown to be short-lived in the condensed phase, contradicting a claim of the isolation of **32** as a liquid.¹⁴ Cyanoketene was prepared from the reaction of cyanoacetylene with oxygen in an ice matrix and identified by the IR bands at 2224 (CN) and 2166 (C=C=O) cm⁻¹, which agreed with the calculated spectrum.¹⁵



The cumulene **35** was prepared by pyrolysis of **34**, and rearranged to cyanoketene (**32**), which in the gas phase gave preferential protonation on nitrogen forming **36** (equation 9).¹⁶



Flash vacuum thermolysis (FVT) of various cyanoacetic acid derivatives also yielded **32**, as identified by its strong IR absorption at 2163 cm⁻¹ in an Ar matrix.¹⁷ Similarly, FVT of Meldrum's acid derivatives **37** gave **32**, mainly from the iminopropadienone **38** (equation 11).¹⁷ FVP of **39** with matrix isolation gave **32**, with the ketenyl IR band at 2162.8 cm⁻¹ and the CN band at 2240.2 cm⁻¹ (equation 12).¹⁸ Photolysis of **32** in the matrix with 185 or 193 nm light gave decarbonylation and formation of cyanocarbene, and upon photolysis with 254 nm light this reformed **32** (equation 12).¹⁸





The heats of formation of **32** and its radical cation have been determined as  $24\pm5$  and  $253\pm3$  kcal/mol, respectively, and the heat of formation of the radical cation of (NC)₂C=C=O (**40**) has been estimated as 300 kcal/mol.¹⁹

It was proposed that the basic hydrolysis of 2-nitrophenyl cyanoacetate (41) formed 32 on the basis of the enhanced rate of reaction compared to that expected for normal ester hydrolysis (equation 13).²⁰

$$\begin{array}{cccc} & & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & & \\ & & & & \\ & & & & \\ & & & & & \\ & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\$$

Chlorocyanoketene (43) was generated from pyrolysis of furanone 42 and reacted with 3-hexyne, forming cyclobutenone 44 (equation 14).^{21–23} This gave ring opening even at ambient temperatures, forming vinylketene 45 (equation 14), which reacted by [4 + 2] cycloaddition with 1-ethoxypropyne to form cyclobutenone 46, which opened to vinylketene 47 followed by cyclization to cyclohexadienone 48 (equation 15).²³ Cyclohexene reacted with 43 by [2 + 2] cycloaddition (equation 16).²³



Pyrolysis of azidoquinones **49** gave a general route via cyanocylopentenediones **50** to form substituted cyanoketenes **51** (equation 17).^{24–28} *tert*-Butylcyanoketene (**51a**, R=*t*-Bu) was obtained in benzene solution in 95% yield, and while not isolated was characterized by the IR bands at 2130 (C=C=O) and 2220 (CN) cm⁻¹ and the ¹H NMR singlet at  $\delta$  0.75.²⁵ The reactions of **43** and of *tert*-butylcyanoketene (**51a**) with styrene oxide gave keteneacetals **52** (equation 18).²⁹



*tert*-Butylcyanoketene (**51a**) reacted with arylalkenes (aryl = phenyl, 1- and 2naphthyl, anthracenyl) and gave cyclobutanones **53** with *cis*-stereochemistry of the *tert*-butyl with the phenyl and naphthyl groups but *trans* for *tert*-butyl with anthracenyl (equation 19).³⁰ The reaction was proposed to be concerted in the former two cases but stepwise in the latter.



Dicyanoketene (40) was prepared by thermolysis of dicyanodiazidoquinone  $54^{28}$  and from FVP at 500 °C of methyl dicyanoacetate (55).³¹ When trapped in an argon matrix, it was identified by the IR absorption at 2175 cm⁻¹ (equation 20).³¹ The synthetic utility of 40 was also explored.²⁸



### **REFERENCES FOR SECTION 4.1.3**

- 1. Gong, L.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1991, 113, 6021-6028.
- Allen, A. D.; Andraos, J.; Kresge, A. J.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1992, 114, 1848–1879.
- 3. McAllister, M. A.; Tidwell, T. T. J. Org. Chem. 1994, 59, 4506-4515.
- 4. Allen, A. D.; Gong, L.; Tidwell, T. T. J. Am. Chem. Soc. 1990, 112, 6396-6397.
- Wagner, B. D.; Arnold, B. R.; Brown, G. S.; Lusztyk, J. J. Am. Chem. Soc. 1998, 120, 1827–1834.
- Allen, A. D.; Cheng, B.; Fenwick, M. H.; Huang, W.; Missiha, S.; Tahmassebi, D.; Tidwell, T. T. Org. Lett. 1999, 1, 693–696.
- 7. Selvarajan, R.; Boyer, J. H. J. Org. Chem. 1971, 36, 1679-1682.
- 8. Leyi Gong, unpublished results at the University of Toronto.
- 9. Magriotis, P. A.; Vourloumis, D.; Scott, M. E.; Tarli, A. *Tetrahedron Lett.* 1993, 34, 2071–2074.
- Fernández-Zertuche, M.; López-Cortina, S.; Meza-Aviña, M. E.; Ordóñez; M.; Ramírez-Solís, A. Arkivoc 2003 (xi) 89–99.
- 11. Pollart, D. J.; Moore, H. W. J. Org. Chem. 1989, 54, 5444-5448.
- 12. Nguyen, N. V.; Chow, K.; Moore, H. W. J. Org. Chem. 1987, 52, 1315-1319.
- 13. Hahn, M.; Bodenseh, H.-K.; Ferner, M. J. Mol. Spectrosc. 2004, 223, 138-147.
- 13a. Bock, H.; Hirabayashi, T.; Mohmand, S. Chem. Ber. 1981, 114, 2595-2608.
- 14. Efremov, D. A.; Zavlin, P. M.; Essentseva, N. S..; Tebby, J. C. J. Chem. Soc. Perkin Trans. 1 1994, 3163–3168.
- 15. Borget, F.; Chiavassa, T.; Aycard, J.-P. Chem. Phys. Lett. 2001, 348, 425-432.
- Flammang, R.; Haverbeke, Y. V.; Wong, M. W.; Rühmann, A.; Wentrup, C. J. Phys. Chem. 1994, 98, 4814–4820.
- Moloney, D. W. J.; Wong, M. W.; Flammang, R.; Wentrup, C. J. Org. Chem. 1997, 62, 4240–4247.
- 18. Maier, G.; Reisenauer, H. P.; Rademacher, K. Chem. Eur. J. 1998, 4, 1957–1963.
- 19. Holmes, J. L.; Mayer, P. M.; Vasseur, M.; Burgers, P. C. J. Phys. Chem. 1993, 97, 4865–4870.
- 20. Holmquist, B.; Bruice, T. C. J. Am. Chem. Soc. 1969, 91, 3003-3009.
- 21. Moore, H. W.; Hernandez, L.; Sing, A. J. Am. Chem. Soc. 1976, 98, 3728-3730.
- 22. Fishbein, P. L.; Moore, H. W. J. Org. Chem. 1985, 50, 3226-3228.
- 23. Fishbein, P. L.; Moore, H. W. Org. Synth. 1990, 68, 205-211.
- 24. Moore, H. W.; Gheorghiu, M. G. Chem. Soc. Rev. 1981, 10, 289-328.
- Weyler, W., Jr.; Duncan, W. G.; Liewen, M. B.; Moore, H. W. Organic Syntheses, Coll. Vol. VI; Noland, W. E., Ed.; Wiley, New York, 1988, pp. 210–215.
- 26. Moore, H. W. Acc. Chem. Res. 1979, 12, 125-132.
- 27. Moore, H. W.; Decker, O. H. W. Chem. Rev. 1986, 86, 821-830.
- 28. Neidlein, R.; Leidholdt, R. Chem. Ber. 1986, 119, 844-849.
- 29. Danila, M. G.; Doana, M. I.; Filip, P. Rev. Roum. Chim. 2002, 47, 279-286.
- 30. Gheorghiu, M.; Mihai, M.; Parvulescu, L. Rev. Roum. Chim. 1995, 40, 653-660.
- 31. Gano, J. E.; Jacobson, R. H.; Wettach, R. H. Angew. Chem. Int. Ed. Engl. 1983, 22, 165–166.

## 4.1.4 Aryl- and Heteroarylketenes

Diphenylketene (1) was the first ketene to be isolated and characterized, by Staudinger in 1905,¹ by the zinc reduction of  $\alpha$ -chorodiphenylacetyl chloride (equation 1), and has been the most extensively utilized ketene. The fact that diphenylketene was isolable and relatively stable is evidently due to steric factors, as the bulky aryl groups hinder dimerization. The computed isodesmic comparison of equation  $2^{2-4}$  indicated that phenylketene is only slightly stabilized relative to methylketene. The stability of ketenes was correlated with the electronegativity of the substituents. Ketenes are stabilized by electropositive groups, and by conjugative  $\pi$ -electron acceptors, and destabilized by electronegative substituents,²⁻⁴ and for phenyl the sum of these effects is near that of methyl. Ketenes **1** and **2** are highly activated by the phenyl groups and give facile reactions with nucleophiles and in cycloaddition reactions.

Phenylketene was prepared in 1911 from  $\alpha$ -chlorophenylacetyl chloride by zinc dehalogenation and was detected by its color in solution.⁵ The ketene was trapped with *n*-BuNH₂ (equation 3)⁵ but polymerized upon attempted isolation by removal of the solvent. Both **1** and **2** were prepared in the gas phase by dehalogenation for measurement of their photoelectron spectra, and their conformations were determined by computation.⁵a

Procedures are reported in *Organic Syntheses* for the preparation of diphenylketene (1) by thermal Wolff rearrangement of the diazo ketone (equation 4)⁶ and by dehydrochlorination of diphenylacetyl chloride (equation 5).⁷ The latter procedure was also used to prepare 1 ¹³C labeled in both carbons from ¹³CO₂, in 22% yield.^{7a} When dehydrochlorination was carried out in refluxing chloroform, the byproduct **6** was formed to the extent of 6–9% (equations 6, 7).⁸ The mechanism proposed for the formation of **6** involved initial hydride transfer to **1**, forming the corresponding enolate.⁸



The original preparation of **1** was by reductive dehalogenation,^{1,9} and this procedure has also been used for the preparation of di-(2-naphthyl)ketene **7** (equation 8).¹⁰ Diphenylketene (**1**) has also been prepared by dehalogenation of  $\alpha$ -bromodiphenylacetyl chloride with triphenylphosphine (equation 9).¹¹ The pyrolysis of diphenylketene acetal **8a** also gave **1** (equation 10).¹² Bis(pentafluorophenyl)ketene (**8b**) was prepared by direct dehydration of the acid (equation 11).¹³

$$2-\text{Naphthyl} \qquad O \qquad Zn-Cu \qquad 2-\text{Naphthyl} \qquad Cl \qquad Et_2O \qquad 2-\text{Naphthyl} \qquad Cl \qquad Cl \qquad Et_2O \qquad 2-\text{Naphthyl} \qquad C=O \qquad (8)$$

$$\begin{array}{ccc}
Ph & OMe & \xrightarrow{190 \,^{\circ}C} & Ph \\
Ph & OSiMe_3 & Ph \\
\mathbf{8a} & \mathbf{1} (85\%)
\end{array} (10)$$

$$\begin{array}{cccc}
C_6F_5 & \bigcirc & & P_{2}O_5 & C_6F_5 \\
C_6F_5 & \bigcirc & & & & C_6F_5 \\
& & & & & C_6F_5 \\
& & & & & & Bb (42\%)
\end{array}$$
(11)

Three dimers of diphenylketene have been reported.¹⁴ The 1,3-cyclobutanedione dimer **9a** was obtained from **1** and quinoline, while dimerization induced by sodium methoxide gave the  $\beta$ -lactone dimer **9b**, and heating of **1** at 100 °C in benzoyl chloride formed **9c**.¹⁴



The generation of phenylketene (2) by dehydrochlorination with triethylamine in ether led to formation of the acylated dimer **10** (equation 12).¹⁵ This reaction evidently proceeded through formation of the 1,3-diketo dimer **10a**, followed by acylation of the enol. The enol has been obtained by basic hydrolysis of **10**.¹⁵ The  $\beta$ -lactone dimer **10b** of phenylketene was isolated when the ketene was prepared by zinc dechlorination (equation 13).¹⁶



Solutions of phenylketene (2) for synthetic applications have been conveniently obtained by the protocol of Lectka, in which phenylacetyl chloride was reacted with a tertiary amine such as  $Et_3N$  or a chiral amine such as benzoylquinine as a kinetic shuttle base and stoichiometric bases such as NaH with 15-crown-6 or K₂CO₃ that reacted irreversibly (Scheme 4.1).^{17,18} Chiral bases also acted as catalysts for stereoselective addition to the ketene, and with benzoylquinine together with In(OTF)₃ as an electrophilic cocatalyst, a [2 + 2] cycloaddition with imine **11** 



Scheme 4.1 Generation of phenylketene by dehydrochlorination with a shuttle base and NaH $\bullet$ 15-crown-5.

gave the  $\beta$ -lactam product **12** in 95% yield, 98% *ee*, and 60/1 *dr* (equation 14).¹⁷ The mechanism of the catalysis has been examined in detail.^{17a} 1,8-Bis(dimethyl-amino)naphthalene (**13**) has also been used as a stoichiometric base, but may give reversible deprotonation forming phenylketene, so the ketene may not be involved in the product-forming step.¹⁷

Reaction of **2** generated as in Scheme 4.1 with **14** gave stereoselective chlorination (equation 15). The procedure was also carried out with other bases, including sodium bicarbonate and polymer-bound triaminophosphonamide.¹⁸



Phenylketene has also been generated from phenylethynol (**17**), which was formed by the photochemical decarbonylation of hydroxycyclopropenone **16** (equation 16).^{19,20} The rearrangement was monitored by both TRIR spectroscopy¹⁹ and UV spectroscopy.²⁰ This method was also utilized for the formation of other aryl-ketenes whose kinetics of hydration to the carboxylic acids were monitored.²¹ Phenylynolate (**18**) was formed by a procedure involving LiCHBr₂ addition to ethyl benzoate (equation 17) and reacted with ethanol to give the ester **19** by way of the ketene **2** (equation 18).^{22–24} Other routes to **17** and **18** are given in Section 4.8.1.^{25,26}

$$\stackrel{\text{Ph}}{\longrightarrow} \stackrel{\text{OH}}{\longrightarrow} \stackrel{hv}{\longrightarrow} \text{Ph-C=C-OH} \stackrel{\text{Ph}}{\longrightarrow} \stackrel{\text{Ph}}{\longrightarrow} C=O$$
(16)

PhC=COLi 
$$\xrightarrow{H^+}$$
  $\xrightarrow{Ph}$   $\xrightarrow{EtOH}$  PhCH₂CO₂Et (18)  
18 2 19 (71%)

Phenylketene (2) and other monoarylketenes  $RC_6H_4CH=C=O$  20 (R = 2-Me, 2-Cl, 2-I, 2-Br, 4-Me, 4-MeO) were generated by microwave-assisted Wolff rearrangement and captured with benzylamine (equation 19).²⁷ Generation of monoarylketenes from thermal Wolff rearrangement in the absence of trapping agents was not satisfactory because of the reaction of the ketene formed with the residual diazo ketone (equation 20).²⁸ An *Organic Syntheses* procedure describes the silver benzoate–catalyzed generation of 1-naphthylketene (22) and capture

with ethanol (equation 21).²⁹



Diphenylketene (1) reacted with ethoxyacetylene to give a mixture of the [2 + 2] cycloaddition product 23 and the azulene derivative 24 (equation 22).^{10,30,31} The cyclobutenone 23 was characterized as the naphthol 26 (equation 23) formed by thermal rearrangement of 23 via the intermediate ketene 25 (Smith-Hoehn reaction; see Section 3.4.1), while 24 was proposed to result from a net [2 + 2 + 2] cycloaddition forming the intermediate **27** from attack on the ketene and cyclization onto the aromatic ring.



Dibenzofulvenone (28) is an analogue of diphenylketene that was prepared by Staudinger using dehalogenation by zinc reduction (equation 24).³² Other examples of fulvenones are described in Section 4.1.10. Another diphenylketene analogue is **30**, which was formed by the photochemical Wolff rearrangement of diazo ketone **29** in a matrix and identified by the IR absorption at 2127 cm⁻¹ (equation 25).^{33,34} However, ketene **30** was not observed as a product from photolysis or thermolysis

of **29** in solution.^{34,35} Diazulenylketene (**30a**) was prepared by thermal Wolff rearrangement of the diazo ketone and trapped with MeOH (equation 25a).³⁶



Photolysis of phenyldiazomethane in an argon matrix doped with CO led to formation of a small amount of phenylketene (**2**), as identified by its IR band at  $2115 \text{ cm}^{-1}$  (equation 26).³⁷ 2-Naphthylcarbene reacted similarly with CO and formed 2-naphthylketene, with an IR band at  $2121 \text{ cm}^{-1}$ .³⁸ Phenylketene has been generated by Wolff rearrangement as a relatively long-lived intermediate and observed by IR in isooctane solution.^{39,40}

$$\stackrel{\text{Ph}}{=} N_2 \xrightarrow{h\nu} \text{PhCH} \xrightarrow{\text{CO}} \stackrel{\text{Ph}}{=} C=0$$
(26)

The ester enolate **31** has been reported to form PhCH=C=O (equation 27) and NMR signals attributed to this species were reported,⁴¹ but confirmation of these results is desirable.

$$\begin{array}{ccc} Ph & OLi & Ph \\ & & & \\ OPr-i & & \\ 31 & 2 \end{array}$$

$$(27)$$

Photolysis at 403 nm of the diazo ketone **32** in an argon matrix at 10 K led to the ketene **33**, identified by its strong ketene IR bands at 2130, 2126, and 2121 cm⁻¹

(equation 28).⁴² Bands were also observed for the cyclopropenyl ketone **34**, which on further photolysis gave the carbene **35**, which itself formed the fulvenone **36** on photolysis at 260 nm (equation 29).⁴²



Photolysis of 3-diazo-4-oxocoumarin **37** and a variety of ring-substituted derivatives gave ketenes **38** studied for photolithographic applications (equation 30).⁴³ The reactivity of **38** in thin novolac films was monitored by the disappearance of the ketenyl IR absorptions near 2142 cm⁻¹ to determine the reactivity with the phenolic groups in the resin. The kinetics were modeled using fractal mathematics, which are useful for studies in polymer systems. The reactivities increased with increasing electron-withdrawing power of substituents on the aryl rings, with retardations due to bulky groups near the ketenyl moiety, but an acceleration by a proximate benzyloxy group, which may have attracted the nucleophile. Recent developments in the theory of the mechanism of action of novolac-diazonaphthoquinone resists have been reviewed (see also Section 4.1.10.2).⁴⁴



The mixed ketenes **39–43** with highly substituted aryl groups were prepared by dehydrochlorination of the acyl chlorides.^{45–56} Thermolysis of **44** led to the isochromene **45** by hydride transfer (equation 31), and at a rate 100 times faster than the corresponding reaction of Mes₂C=C=O (**46**).⁵⁰ The reaction of **46** with amines led to amide enols observable by NMR.^{48,50} The kinetics of the addition of alcohols catalyzed by pyridine and bridged pyridines to Mes₂C=C=O C=O have been measured.⁴⁶



Arylalkylketenes have been readily generated, usually from derivatives of  $\alpha$ -arylalkanoic acids by the dehydrochlorination or dehalogenation routes. They are less reactive than arylaldoketenes, and can usually be isolated, but are still sensitive to moisture and react readily with many nucleophilic reagents. Arylalkylketenes **47** dimerized on heating to give cyclobutane-1,3-diones **48**, with an increasing proportion of the *cis* product with increasing size of the alkyl group (equation 32).⁵⁷ The preference for formation of the more crowded 4-membered ring product is a common feature of ketene cycloaddition chemistry, and is explained as resulting from perpendicular approach of the reacting species with the larger substituents in the least crowded position in the transition state, which, however, leads to more crowded products.



Arylalkylketenes have been prepared by dehydrohalogenation of 2-alkyl-2-arylacetyl chlorides using  $\text{Et}_3\text{N}$ ,⁵⁸ but **49a–c** were preparing using  $\text{EtMe}_2\text{N}$  (equation 33).^{59–61} These ketenes may usually be isolated but have sometimes been generated for reaction in situ for use in Wittig reactions.^{62,63}

$$\begin{array}{ccc} Ar \\ R \\ R \\ Cl \\ 47 \\ \end{array} \xrightarrow{Me_2NEt} & Ar \\ THF \\ R \\ R \\ C=O \\ R \\ R \\ C=O \\ 2-Anis, Me (49b); \\ N-Me-indolyl, Bn (49c) \\ \end{array}$$
(33)

Intramolecular cyclization of aryl ketenes with remote double bonds provided a useful route to fused cyclobutanones.⁶⁴ Thus, dehydrohalogenation of 2-arylhept-6enoyl chlorides (**50**) generated unobserved 2-aryl-2-pent-4-enylketenes **51**, which gave [2+2] cyclizations forming bicyclo[3.2.0]heptanones **52** (equation 34).⁶⁴



2,6-Di-*tert*-butyl-4-methylphenyl (BHT) esters of  $\alpha$ -alkylarylacetates **53** were converted by *n*-BuLi in THF in the presence of ZnCl₂ or SnCl₂ to the corresponding ester enolates, which lost aryloxide anion to form ketenes **54** (equation 35).⁶⁵ The ketenes reacted with added lithium trimethyl phosphonoacetate **55**, forming allenes **56** through ketene Horner-Wadsworth-Emmons reactions (equation 36).⁶⁵



Reaction of 2,6-di-*tert*-butyl-4-methoxyphenyl (BHA) esters of 1- and 2naphthalenecarboxylates **57** with organolithiums proceeded by addition to the naphthalene ring and elimination, giving ketenes **58** (equation 37).⁶⁶ The ketenes were converted in situ by lithium triethylborohydride to aldehyde enolates that were protonated or alkylated, followed by NaBH₄ reduction forming **59** (equation 38).⁶⁶



Zinc dehalogenation of PhCBr(*t*-Bu)COBr (**60**) was used to form phenyl-(*tert*-butyl)ketene (**61**), which was isolated and characterized (equation 39).⁶⁷ The kinetics of hydration of this ketene were measured, and it was found to be less reactive in this nucleophilic attack by a factor of  $1.4 \times 10^4$  compared to phenylketene.⁶⁷

$$\begin{array}{cccc} Ph & O & Zn & Ph \\ \hline t-Bu & & \\ Br & Cl & \\ \hline & THF, reflux & \\ 60 & 61 (59\%) \end{array} C=O$$
(39)

The formation of arylalkylketenes by Wolff rearrangement requires the preparation of unsymmetrical diazo ketones, usually from aryl carboxylic acids and substituted diazomethanes or from cyclic ketones. This method was used to form phenyl(methyl)ketene **47a** (equation 40) and 1-ketenylideneindane **62**, which underwent [2 + 2] cycloaddition with alkynyl ethers (equation 41).⁶⁸ Because of the susceptibility of ketenes to photochemical decarbonylation, these reactions are usually done with in situ capture of the ketenes.



Treatment of 2-phenyl-2-alkylmalonic acids with trifluoroacetic anhydride gave phenylalkylketenes, and these were isolated by distillation (equation 42).^{69,70}

2-Arylcyclobutenone **63**, prepared by addition of the aryllithium to the cyclobutenedione, underwent thermal ring opening to give the unobserved arylvinylketene **64**, which cyclized, forming naphthoquinone **65** after air oxidation (Smith-Hoehn reaction; equation 43).^{71,72}



Phenyl(methyl)ketene (**47a**) gave [2 + 2] cycloaddition with a variety of alkenes with a preference between 1 and 2 for *endo* stereochemistry of the phenyl, but for cycloaddition with cyclopentadiene forming **66** the preference was greater than 20:1 (equation 44).⁷³



Photochemical Wolff rearrangement of diazo ketone **67** formed the ketene **68**, which cyclized via the diradical intermediate **69**, leading to the naphthol **70** (equation 45).⁷⁴



Addition of *n*-butyllithium to phenyl(ethyl)ketene (**47b**) took place with in-plane attack of the nucleophile on the least hindered side of the ketene opposite the aryl group forming the enolate **71**, which reacted with Me₃SiCl, forming *Z*-**72** (equation 46).⁷⁵



Electron transfer oxygenation of arylmethylketenes **73** was suggested to lead to  $\alpha$ -lactone intermediates **74**, which reacted with another molecule of **73** to form succinic anhydrides (equation 47).⁷⁶ Ring opening of **74** to a zwitterion could also be involved.

$$\begin{array}{c} Ar \\ \searrow \\ CH_3 \\ 73 \end{array} \xrightarrow{R_3N} \left[ \begin{array}{c} Ar \\ \bigcirc \\ CH_3 \end{array} \right] \xrightarrow{O} \\ O \\ CH_3 \\ O \\ (5-57\%) \end{array} \right]$$
(47)

Wolff rearrangement of pyridyl diazo ketone **75** as a monolayer on a platinum surface gave 3-pyridyl(methyl)ketene **76** bound to the surface, as identified by the ketenyl IR absorption band at 2111 cm⁻¹ (equation 48).⁷⁷ 4-Pyridyl(methyl)ketene **77** was formed in the same way. Photolysis of the diazo ketone **78** in an argon matrix at 10 K gave the ylide **79**, as identified by the IR spectrum which agreed with the DFT calculated spectrum, and which rearranged to 2-pyridylketene **80** upon further photolysis with light with  $\lambda > 320$  nm (equation 49).⁷⁸ Wolff rearrangement induced by FVP at 750 °C with trapping in an argon matrix at 10 K gave **80**, as identified by the IR bands at 2123 (*anti*-conformer) and 2132 cm⁻¹ (*syn*-conformer).⁷⁸ The 2-, 3-, and 4-pyridyl ketenes **80**, **81**, and **82**, respectively, were formed by Wolff rearrangements in solution⁷⁹ and identified using TRIR spectroscopy by their absorption bands at 2127, 2125, and 2123 cm⁻¹, respectively. Their reactivity with *n*-BuNH₂ and with H₂O is considered in Section 5.5.2.3. Other preparations of pyridylketenes have been reported.^{80–83}



Flash vacuum thermolysis (FVT) of the quinolizine-4-one **83** led to the formation of 2-pyridylcyanoketene (**84**) as well as  $CH_2=C=O$  (equation 50).⁸² These were trapped in Ar at 20 K and identified by their ketenyl IR bands at 2149 cm⁻¹ (*s*-Z-**84**) and 2141 cm⁻¹, respectively.⁸² Similar thermolysis of the triazole **85** gave 2-pyridylmethoxyketene (**86**), IR 2114 cm⁻¹, which upon photolysis formed CO and 2-acetylpyridine (equation 51).⁸² Photolysis of **85** in the matrix gave 2-pyridylketene (**80**) by the possible intermediacy of the diazo ester (equation 52).⁸³



Furandiones **87** underwent decarbonylation at 138–140 °C, forming quinoxalinylketenes **88** (equation 53), which gave [4 + 2] dimerization with rearrangement, forming **89**,^{84,85} or reacted with ketones by [4 + 2] cycloadditions.



Reaction of ferrocenylacetic acid **90** with PhOP(O)Cl₂ and Et₃N or of ferrocenylacetyl chloride **91** with Et₃N led to the formation of ferrocenylketene **92**, as evidenced by the in situ capture with imines forming  $\beta$ -lactams **93** (equation 54).⁸⁶ Photolysis of the diazo ketone **94** gave **92**, as observed by IR, and this was trapped by *n*-BuNH₂ or by the stable nitroxyl radical TEMPO (TO•), forming **95** and **96**, respectively.⁸⁷ Ferrocenylketenes have also been generated as unobserved intermediates in ring contraction of ferrocenyl diazo ketones.⁸⁸



Aryl(trialkyl)silylketenes **97** have been generated by Wolff rearrangements, isolated as long-lived species by chromatography in yields of 52–89%, and found to react with trimethylsilyldiazomethane to yield indanone derivatives **98** (equation 55).⁸⁹ These reactions were proposed to proceed with initial [2 + 1] cycloaddition, as shown in Section 4.5.



### **REFERENCES FOR SECTION 4.1.4**

- 1. Staudinger, H. Chem. Ber. 1905, 38, 1735-1739.
- 2. Gong, L.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1991, 113, 6021-6028.
- Allen, A. D.; Andraos, J.; Kresge, A. J.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1992, 114, 1848–1879.
- 4. McAllister, M. A.; Tidwell, T. T. J. Org. Chem. 1994, 59, 4506-4515.
- 5. Staudinger, H. Chem. Ber. 1911, 44, 533-543.
- 5a. Sammynaiken, R.; Westwood, N. P. C. J. Chem. Soc. Perkin Trans. 2 1989, 1987–1992.
- Smith, L. I.; Hoehn, H. H. In Organic Syntheses; Hornung, E. C., Ed.; Wiley, New York, 1955 Coll. Vol. III, pp. 356–359.

- Taylor, E. C.; McKillop, A.; Hawks, G. H. In *Organic Syntheses*; Noland, W. E., Ed.; Wiley, New York, 1988, Coll. Vol. VI, pp. 549–551.
- 7a. Anderson, J. C.; Broughton, S. Synthesis 2001, 2379-2380.
- Abe, N.; Osaki, I.; Kojima, S.; Matsuda, H.; Sugihara, Y.; Kakehi, A. J. Chem. Soc. Perkin Trans. 1 1996, 2351–2356.
- 9. Hanford, W. E.; Sauer, J. C. Org. React. 1946, 3, 108-140.
- 10. Wuest, J. D. Tetrahedron 1980, 36, 2291-2296.
- 11. Darling, S. D.; Kidwell, R. L. J. Org. Chem. 1968, 33, 3974-3975.
- 12. Ainsworth, C.; Chen, F.; Kuo, Y.-N. J. Organomet. Chem. 1972, 46, 59-71.
- Lubenets, E. G.; Gerasimova, T. N.; Barkhash, V. A. Zh. Org. Khim. 1972, 8, 654; Russ. J. Org. Chem. (Engl. Transl.) 1972, 8, 663.
- 14. Das, H.; Kooyman, E. C. Recl. Trav. Chim. Pays-Bas 1965, 84, 965-978.
- Farnum, D. G.; Johnson, J. R.; Hess, R. E.; Marshall, T. B.; Webster, B. J. Am. Chem. Soc. 1965, 87, 5191–5197.
- 16. Baldwin, J. E.; Roberts, J. D. J. Am Chem. Soc. 1963, 85, 2444-2445.
- France, S.; Shah, M. H.; Weatherwax, A.; Wack, H.; Roth, J. P.; Lectka, T. J. Am. Chem. Soc. 2005, 127, 1206–1215.
- 17a. Taggi, A. E.; Hafez, A. M.; Wack, H.; Young, B.; Ferraris, D.; Lectka, T. J. Am. Chem. Soc. 2002, 124, 6626–6635.
- France, S.; Wack, H.; Taggi, A. E.; Hafez, A. M.; Wagerle, T. R.; Shah, M. H.; Dusich, C. L.; Lectka, T. J. Am. Chem. Soc. 2004, 126, 4245–4255.
- Wagner, B. D.; Zgierski, M. Z.; Lusztyk, J. J. Am. Chem. Soc. 1994, 116, 6433– 6434.
- Chiang, Y.; Kresge, A. J.; Hochstrasser, R.; Wirz, J. J. Am. Chem. Soc. 1989, 111, 2355–2357.
- 21. Chiang, Y.; Kresge, A. J.; Popik, V. V. J. Am. Chem. Soc. 1995, 117, 9165-9171.
- 22. Kowalski, C. J.; Haque, M. S.; Fields, K. W. J. Am. Chem. Soc. 1985, 107, 1429-1430.
- 23. Kowalski, C. J.; Reddy, R. E. J. Org. Chem. 1992, 57, 7194-7208.
- 24. Reddy, R. E.; Kowalski, C. J. Org. Synth. 1992, 71, 146-157.
- Julia, M.; Jalmes, V. P.-S.; Plé, K.; Verpeaux, J.-N.; Hollingworth, G. Bull. Soc. Chim. Fr. 1996, 133, 15–24.
- 26. Hoppe, I.; Schöllkopf, U. Liebigs Ann. 1979, 219-226.
- Sudrik, S. G.; Chavan, S. P.; Chandrakumar, K. R. S.; Pal, S; Date, S. K.; Chavan. S. P.; Sonawane, H. R. J. Org. Chem. 2002, 67, 1574–1579.
- 28. Takebayashi, M.; Ibata, T. Bull. Chem. Soc. Jpn. 1968, 41, 1700-1707.
- 29. Lee, V.; Newman, M. S. In *Organic Syntheses*; Noland, W. E. Ed.; Wiley, New York, 1988, Coll. Vol. VI, pp. 613–615.
- 30. Teufel, H.; Jenny, E. F. Tetrahedron Lett. 1971, 1769-1772.
- Druey, J.; Jenny, E. F.; Schenker, K.; Woodward, R. B. *Helv. Chim. Acta* 1962, 45, 600–610.
- 31a. Jenny, E. F.; Schenker, K.; Woodward, R. B. Angew. Chem. 1961, 73, 756-757.
- 32. Staudinger, H. Chem. Ber. 1906, 39, 3062-3067.
- 33. Maier, G.; Reisenauer, H. P.; Sayrac, T. Chem. Ber. 1982, 115, 2192-2201.

- McMahon, R. J.; Chapman, O. L.; Hayes, R. A.; Hess, T. C.; Krimmer, H.-P. J. Am. Chem. Soc. 1985, 107, 7597–7606.
- 35. Blair, P. A.; Chang, S.-J.; Schechter, H. J. Org. Chem. 2004, 69, 7123-7133.
- Ito, S.; Hamazaki, H.; Hirasawa, Y.; Ohta, A.; Fujimori, K.; Tanaka, N.; Shiro, M. *Heterocycles* 2004, 62, 463–478.
- McMahon, R. J.; Abelt, C. J.; Chapman, O. L.; Johnson, J. W.; Kreil, C. L.; LeRoux, J.-P.; Mooring, A. M.; West, P. R. J. Am. Chem. Soc. 1987, 109, 2456–2469.
- 38. Albrecht, S. W.; McMahon, R. J. J. Am. Chem. Soc. 1993, 115, 855-859.
- Allen, A. D.; Cheng, B.; Fenwick, M. H.; Givehchi, B.; Henry-Riyad, H.; Nikolaev, V. A.; Shikova, E. A; Tahmassebi, D.; Tidwell, T. T.; Wang, S. J. Org. Chem. 2001, 66, 2611–2617.
- Allen, A. D.; Cheng, B.; Fenwick, M. H.; Huang, W.; Missiha, S.; Tahmassebi, D.; Tidwell, T. T. Org. Lett. 1999, 1, 693–696.
- Solladié-Cavallo, A.; Liptaj, T.; Schmitt, M.; Solgadi, A. Tetrahedron Lett. 2002, 43, 408–415.
- 42. Komnick, P.; Sander, W. Liebigs Ann. 1996, 7-9.
- Leeson, M. J.; Yueh, W.; Tattersall, P. I.; Pawloski, A.; Grayson, S. M.; Willson, C. G. Chem. Mater. 2004, 16, 1763–1769.
- 44. Reiser, A.; Huang, J. P.; He, X.; Yeh, T. F.; Jha, S.; Shih, H. Y.; Kim, M. S.; Han, Y. K.; Yan, K. *Eur. Polym. J.* **2002**, *38*, 619–629.
- 45. Rochlin, E.; Rappoport, Z. J. Org. Chem. 1994, 59, 3857-3870.
- 46. Schyja, W.; Peterson, S.; Lüning, U. Liebigs Ann. 1996, 2099-2105
- 47. Sigalov, M.; Rappoport, Z. J. Chem. Soc., Perkin Trans. 2 1997, 1911–1912.
- 48. Frey, J.; Rappoport, Z. J. Am. Chem. Soc. 1996, 118, 3994-3995.
- 49. Frey, J.; Rappoport, Z. J. Am. Chem. Soc. 1996, 118, 5182-5191.
- 49a. Frey, J.; Rappoport, Z. J. Am. Chem. Soc. 1995, 117, 1161-1162.
  - 50. Rappoport, Z.; Frey, J.; Sigalov, M.; Rochlin, E. Pure Appl. Chem. 1997, 69, 1933–1940.
  - 51. O'Neill, P.; Hegarty, A. F. J. Org. Chem. 1992, 57, 4421-4426.
  - 52. O'Neill, P.; Hegarty, A. F. J. Chem. Soc., Chem. Commum. 1987, 744-745.
  - Allen, B. M.; Hegarty, A. F.; O'Neill, P.; Nguyen, M. T. J. Chem. Soc. Perkin Trans. 2 1992, 927–934.
  - 54. Hegarty, A. F.; O'Neill, P. Synthesis. 1993, 606-610.
  - 55. Biali, S. E.; Gozin, M.; Rappoport, Z. J. Phys. Org. Chem. 1989, 2, 271-280.
  - 56. Fuson, R. C.; Armstrong, L. J.; Shenk, W. J., Jr. J. Am. Chem. Soc. 1944, 66, 964–967.
  - 57. Dehmlow, E. V.; Slopianka, M.; Pickardt, J. Liebigs Ann. 1979, 572-593.
  - 58. Pracejus, H.; Wallura, G. J. Prakt. Chem. 1963, 19, 33-36.
  - 59. Hodous, B. L.; Fu, G. C. J. Am. Chem. Soc. 2002, 124, 10006-10007.
  - 60. Hodous, B. L.; Ruble, J. C.; Fu, G. C. J. Am. Chem. Soc. 1999, 121, 2637-2638.
  - 61. Fu, G. C. Acc. Chem. Res. 2000, 33, 412-420.
  - 62. Kumar, K.; Kaur, S.; Ishar, M. P. S. Synlett 1999, 1237-1238.
  - 63. Lang, R. W.; Hansen, H.-J. Helv. Chim. Acta 1980, 63, 438-455.
  - 64. Snider, B. B.; Niwa, M. Tetrahedron Lett. 1988, 29, 3175-3178.
  - 65. Tanaka, K.; Otsubo, K.; Fuji, K. Synlett 1995, 933-934.
  - 66. Tomioka, K.; Shindo, M.; Koga, K. J. Org. Chem. 1990, 55, 2276-2277.

- 67. Allen, A. D.; Baigrie, L. M.; Gong, L.; Tidwell, T. T. Can. J. Chem. 1991, 69, 138-145.
- Danheiser, R. L.; Brisbois, R. G.; Kowalczyk, J. J.; Miller, R. F. J. Am. Chem. Soc. 1990, 112, 3093–3100.
- 69. Duckworth, A. C. J. Org. Chem. 1962, 27, 3146-3148.
- 70. Kresze, G.; Runge, W.; Ruch, E. Liebigs Ann. 1972, 756, 112-127.
- 71. Koo, S.; Liebeskind, L. S. J. Am. Chem. Soc. 1995, 117, 3389-3404.
- 72. Smith, L. I.; Hoehn, H. H. J. Am. Chem. Soc. 1939, 61, 2619-2624.
- 73. Brady W. T.; Parry, F. H., III; Stockton, J. D. J. Org. Chem. 1971, 36, 1486-1489.
- Padwa, A.; Chiacchio, U.; Fairfax, D. J.; Kassir, J. M.; Litrico, A.; Semones, M. A.; Xu, S. L. J. Org. Chem. 1993, 58, 6429–6437.
- 75. Baigrie, L. M.; Seikaly, H. R.; Tidwell, T. T. J. Am. Chem. Soc. 1985, 107, 5391-5396.
- 76. Schmittel, M.; von Seggern, H. Liebigs Ann. 1995, 1815-1821.
- Pitters, J. L.; Griffiths, K.; Kovar, M.; Norton, P. R.; Workentin, M. S. Angew. Chem., Int. Ed. 2000, 39, 2144–2147.
- 78. Kuhn, A.; Plüg, C.; Wentrup, C. J. Am. Chem. Soc. 2000, 122, 1945–1948.
- Acton, A. W.; Allen, A. D.; Antunes, L. M.; Federov, A. V.; Najafian, K.; Tidwell, T. T.; Wagner, B. D. J. Am. Chem. Soc. 2002, 124, 13790–13794.
- 80. Chapman, O. L.; Sheridan, R. S. J. Am. Chem. Soc. 1979, 101, 3690-3692.
- 81. Tomioka, H.; Ichikawa, N.; Komatsu, K. J. Am. Chem. Soc. 1993, 115, 8621-8626.
- 82. Plüg, C.; Ye, X.; Kuhn, A.; Wentrup, C. Eur. J. Org. Chem. 2002, 2683-2685.
- 83. Andersen, H. G.; Bednarek, P.; Wentrup, C. J. Phys. Org. Chem. 2003, 16, 519–524.
- Lisovenko, N. Y.; Maslivets, A. N.; Aliev, Z. G. Chem. Heterocyclic Comp. 2003, 39, 132–134.
- Maslivets, A. N.; Lisovenko, N. Y.; Krasnyk, O. P.; Tarasova, O. P.; Aliev, Z. G.; Atovmyan, L. O. *Russ. Chem. Bull. Int. Ed.* 2002, *51*, 850–853.
- Bonini, B. F.; Femoni, C.; Comes-Franchini, M.; Fochi, M.; Mazzanti, G.; Ricci, A.; Varchi, G. Synlett 2001, 1092–1096.
- 87. Aquilar-Aquilar, C.; Allen, A. D.; Cabrera, E.; Federov, A. V.; Fu, N.; Henry-Riyad, H.; Kobayashi, S.; Leuninger, J.; Tidwell, T. T. J. Org. Chem. 2005, in press.
- Hisatome, M.; Watanabe, J.; Yamashita, R.-I.; Yoshida, S.; Yamakawa, K. Bull. Chem. Soc. Jpn. 1994, 67, 490–494.
- 89. Dalton, A. M.; Zhang, Y.; Davie, C. P.; Danheiser, R. L. Org. Lett. 2002, 4, 2465-2468.

# 4.1.5 Cyclopropyl-, Cyclopropenyl-, and Oxiranylketenes

Cyclopropylketene is calculated by the HF/6-31G*//HF/6-31G* isodesmic energy comparison of equation 1 to be slightly less stable than methylketene.^{1–3} The size of the cyclopropyl group is intermediate between those of the ethyl and isopropyl groups, and this offers a modest degree of steric protection.⁴ The cyclopropyl group is highly strained, and when reaction pathways permit, ring opening may occur.

$$\begin{array}{c|c} & CH_3 & CH_3 \\ \hline \\ 1 & 2 \end{array} \xrightarrow{C=O} + \begin{array}{c} CH_3 \\ \hline \\ -1.8 \text{ kcal/mol} \end{array} \xrightarrow{CH_3} C=O + \begin{array}{c} CH_2 \\ \hline \\ CH_2 \end{array} \xrightarrow{CH_2} (1)$$

The original preparation of cyclopropylketene was by the Wolff rearrangement of diazo ketone **5** catalyzed by silver oxide in methanol, forming the methyl ester (equation 2).⁵ Generation of **1** by dehydrochlorination of the acyl chloride **6** resulted in isolation of the dimer **7** (equation 3).⁶ Pyrolysis of the dimer or of the ester *c*-PrCH₂CO₂Et at 550 and 600 °C, respectively, led to the formation of cyclopentenone **9**, evidently formed by rearrangement of **1** (equation 4).⁶ This rearrangement may be considered to occur either through a diradical intermediate **8** or, alternatively, through a concerted process analogous to the vinylcyclopropane rearrangement, which in the case of the ketene **1** would be a pseudopericyclic process.



2,2-Diphenylcyclopropylketene (10) was generated by photochemical Wolff rearrangement of the diazo ketone in  $H_2O/CH_3CN$  and trapped as the acid. Photolysis of 4,4-diphenylcyclopentenone (11) also led to 10 through a process proposed to involve the biradical 12 (equation 5).⁷ Photolysis of 4,4-dimethylcyclopentenone (13) in pentane gave rise to an IR absorption at 2110 cm⁻¹ assigned to the ketene 14, and trapping with methanol gave the ester 15 (equation 6).^{7,8} These and other cyclopentenone cleavages were proposed to involve formation of diradical intermediates in the cyclopentenone-cyclopropylketene interconversion.⁸



Cyclopropylketenes 1, 14, and 16 were generated by photochemical Wolff rearrangements in hydrocarbon solvents, and their IR spectra with ketenyl bands at  $2119-2120 \text{ cm}^{-1}$  were measured under ambient conditions (equation 7).⁹ The reaction of 16 with the aminoxyl radical TEMPO (TO•) gave the product 18 from ring opening to the intermediate radical 17 (equation 7).⁹



2-Phenyl-, 2-cyclopropyl-, and 2-*tert*-butyl-cyclopropylketenes **19a–c**, were prepared by dehydrochlorination of the acyl chlorides with triethylamine, and were obtained as stable and isolable products, with characteristic ketenyl IR bands at 2102, 2106, and 2097 cm⁻¹, respectively (equation 8).¹⁰ Addition to **19a** of *n*-BuLi or *t*-BuLi led to enolates that upon capture with Me₃SiCl gave the silyl enol ethers **20** with 79:21 and 9:91 preferences for addition of RLi *anti* to phenyl, respectively (equation 8).¹⁰



Cyclopropylketene (1) was more reactive than *n*-BuCH=C=O toward H₂O by a factor of only 3.1 (Section 5.5.1.2).¹¹ There is no evidence for major rate enhancements in other additions to this species.

Photochemical Wolff rearrangement of diazo ketone **21** in THF gave the strained cyclopropylketene **22**, which cyclized to bicyclooctadienone **23** by Cope rearrangement (equation 9).¹² Ketene **22** was reformed by photolysis of **23** and observed by IR at 2112 cm⁻¹ (neat film,  $-190 \,^{\circ}$ C).¹³ The ketene was trapped by methanol to give the ester in >90% yield, or in the absence of nucleophiles underwent thermal conversion back to **23**.¹³ Other similar vinylcyclopropylketenes were generated by ring opening of bicycloheptadienones.¹³



Dehydrochlorination of 24 gave the cycloheptatrienylketene 25, which was proposed to be in equilibrium with the norcaradienylketene 26 (equation 10).¹⁴

Evidence for the presence of 26 was the isolation of the possible rearrangement products 27 and 28 in 44 and 40% yields, respectively (equation 11).¹⁴



The ketene **30** formed by Wolff rearrangement of diazo ketone **29** using sonochemically assisted silver benzoate catalysis led to the cycloheptadienone **31** by ketene Cope rearrangement (equation 12).¹⁵ Modeling of the reaction by DFT calculations allowed the prediction of the effects of substituents in order to design substrates that favor particular products (see also Section 3.3).¹⁶



As noted in Section 3.4.2, photolysis of hydroxycyclopentenone **32** gave lactone **34**, which was proposed to form from (hydroxycyclopropyl)ketene **33** (equation 13).¹⁷ The ketene was not trapped by  $CH_3OH$ , and it was proposed that the cyclization rate was even faster than that of solvent attack.



Photolysis of **35** in an Ar matrix, as also noted in Section 3.4.2, led to the dimethylcyclopropenylketene **36**, as characterized by the IR band at  $2105 \text{ cm}^{-1}$ , and further photolysis resulted in decarbonylation and formation of the vinylallene **37** (equation 14).¹⁸



Ketene **39** was formed by thermolysis of the diazo ketone **38** in boiling *N*-methylaniline and was captured as the epimeric anilides **40** (equation 15).¹⁹ The IR of **39** generated in a matrix was observed at  $2140 \text{ cm}^{-1}$ .²⁰ Photolysis of the diazo ketone **41** in a matrix gave rise to an IR band at  $2117 \text{ cm}^{-1}$  and provided evidence of the formation of ketene **42** by ring contraction (equation 16).²⁰ However, photolysis in solution in the presence of trapping agents such as CH₃OH or Me₂NH gave products of trapping of cleavage products of an intermediate keto carbene.



Dicyclopropylcarbene (44) was generated in a matrix by photolysis of dicyclopropyldiazomethane (43), and reacted with CO, forming dicyclopropylketene (19b), IR 2112 cm⁻¹ (equation 17).²¹



Photolysis of the cyclopropyl cabene metal complex **45** gave ketene **46**, which was trapped by cyclohexadiene in a [2 + 2] cycloaddition, forming **47** (equation 18).²² Thermal Wolff rearrangement of the diazo ketone **48** proceeded with cyclopropyl group migration, forming the ketene **49**, which was isolated by distillation and reacted with ethanol to give the diethyl ester **50** (equation 19).^{23,24}





Cyclopropenylketene (52) was formed by Wolff rearrangement of the diazo ketone 51 in an argon matrix at 10 K and identified by the IR band at  $2108 \text{ cm}^{-1}$ .²⁵ Further photolysis of the ketene in the matrix led to decarbonylation and formation of cyclobutadiene (53, equation 20).²⁶

Upon photolysis at -78 °C the cyclopentadienones **54** reacted via the bicyclic ketones **55** and gave cyclopropenylketenes **56a,b**, as evidenced by the IR absorption at 2080 cm⁻¹ (equation 21).^{27,28} Upon treatment with CuCl₂ the ketenes reformed **54**.^{27,28}



Oxiranylketenes **58** formed by photochemical Wolff rearrangements of diazo ketones **57** gave stereoselective ring opening on reaction with nucleophiles, forming unsaturated hydroxyl esters **59** (equation 22).²⁹ In the absence of nucleophiles, rearrangement to butenolactones **60** took place (equation 23).³⁰



### **REFERENCES FOR SECTION 4.1.5**

- 1. Gong, L.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1991, 113, 6021–6028.
- Allen, A. D.; Andraos, J.; Kresge, A. J.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1992, 114, 1848–1879.
- 3. McAllister, M. A.; Tidwell, T. T. J. Org. Chem. 1994, 59, 4506-4515.
- Tidwell, T. T. In, 'The Chemistry of the Cyclopropyl Group'; Rappoport, Z., Ed.; Wiley, New York, 1987, pp. 565–632.
- 5. Turnbull, J. H.; Wallis, E. S. J. Org. Chem. 1956, 21, 663-666.
- 6. Berkowitz, W. F.; Ozorio, A. A. J. Org. Chem. 1975, 40, 527-528.
- Agosta, W. C.; Smith, A. B., III; Kende, A. S.; Eilerman, R. G.; Benham, J. *Tetrahedron Lett.* **1969**, 4517–4520.
- 8. Agosta, W. C.; Smith, A. B., III J. Am. Chem. Soc. 1971, 93, 5513-5520.
- Allen, A. D.; Fenwick, M. H.; Henry-Riyad, H.; Tidwell, T. T. J. Org. Chem. 2001, 66, 5759–5763.
- Allen, A. D.; Baigrie, L. M.; Gong, L.; Tidwell, T. T. Can. J. Chem. 1991, 69, 138– 145.
- Allen, A. D.; Andraos, J.; Kresge, A. J.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1992, 114, 1878–1879.
- 12. Freeman, P. K., Kuper, D. G. Chem. Ind. (London) 1965, 424-425.
- Chapman, O. L.; Kane, M.; Lassila, J. D.; Loeschen, R. L.; Wright, H. E. J. Am. Chem. Soc. 1969, 91, 6856–6858.
- 14. Goldstein, M. J.; Odell, B. G. J. Am. Chem. Soc. 1969, 91, 6356-6358.
- 15. Sarpong, R.; Su, J. T.; Stoltz, B. M. J. Am. Chem. Soc. 2003, 125, 13624-13625.
- 16. Su, J. T.; Sarpong, R.; Stoltz, B. M.; Goddard, W. A; III, J. Am. Chem. Soc. 2004, 126, 24–25.
- 17. Gowda, G.; McMurry, T. B. H. J. Chem. Soc., Perkin Trans. 1 1979, 274-278.
- 18. Maier, G.; Reisenauer, H. P. Chem. Ber. 1981, 114, 3916-3921.
- 19. Brook, P. R.; Brophy, B. V. J. Chem. Soc. Perkin Trans. 1 1985, 2509-2513.
- 20. Wiberg, K. B.; Snoonian, J. R. J. Org. Chem. 1998, 63, 1390-1401.
- 21. Ammann, J. R.; Subramanian, R.; Sheridan, R. S. J. Am. Chem. Soc. 1992, 114, 7592– 7594.
- 22. Söderberg, B. C.; Hegedus, L. S.; Sierra, M. A. J. Am. Chem. Soc. 1990, 112, 4364-4374.
- 23. Smith, L. I.; McKenzie, S., Jr. J. Org Chem. 1950, 15, 74-80.
- 24. Keating, J. T.; Skell, P. S. J. Am. Chem. Soc. 1969, 91, 695-699.
- 25. Maier, G.; Hoppe, M.; Lanz, K.; Reisenauer, H. P. Tetrhedron Lett. 1984, 25, 5645-5648.
- 26. Maier, G.; Franz, L. H.; Hartan, L.-G.; Lanz, K.; Reisenauer, H. P. *Chem. Ber.* **1985**, *118*, 3196–3204.
- 27. Maier, G.; Franz, L. H.; Boese, R. Liebigs Ann. 1995, 147-151.
- 28. Maier, G.; Rang, H.; Emrich, R. Liebigs Ann. 1995, 153-160.
- 29. Leemhuis, F. M. C.; Thijs, L.; Zwanenburg, B. J. Org. Chem. 1993, 58, 7170-7179.
- 30. van Haard, P. M. M.; Thijs, L.; Zwanenburg, B. Tetrahedron Lett. 1975, 803-806.
## 4.1.6 Acylketenes

It was originally proposed in 1908 that acetylketene (1) was the structure of the ketene dimer,¹ but this was disproved, and in 1909 carboethoxy(ethyl)ketene (4) was prepared by dehalogenation of **3** in an ethereal solution as the first authentic acylketene (equation 1).² This family has been studied ever since, and general reviews of acylketenes have appeared,^{3,4} as well as a brief review devoted to the chemistry of chlorocarbonylketenes.⁵



Formylketene (5) was calculated to be stabilized by 3.60 kcal/mol relative to methylketene by isodesmic energy comparison (equation 2).^{6–8} This has been attributed to the conjugative stabilization between the ketenyl group and the formyl group, as represented by the resonance structure **5a**. The conformer *syn-***5** was found^{6,7} to be more stable than *anti-***5** by 1.0 kcal/mol, and this was proposed to result from a favorable electrostatic interaction between the formyl oxygen and the ketenyl carbon.^{6,7}



Acylketenes have been prepared by a variety of routes. The parent formylketene **5** was formed by photochemical Wolff rearrangement of the diazo ketone **9** in an argon matrix at 10 K, and observed by IR as interconverting *syn* and *anti* forms, with IR bands at 2145 and 2142 cm⁻¹, respectively.⁹ Thermolysis of **9** in refluxing *n*-butyl vinyl ether gave **5**, which reacted with the vinyl ether by a [4 + 2] cycload-dition forming **10**, which gave **11** by elimination (equation 3).¹⁰



Chlorocarbonyl(phenyl)ketene (13) and related derivatives were formed by thermolysis of malonyl dichlorides 12 and are often isolable intermediates (equation 4).^{5,11–13} These ketenes tended to react preferentially at the ketenyl group with nucleophiles.¹¹ Reaction of **13** with imidazole gave a solid product identified as **14** (equation 4), and this formed **15** on reaction with  $Et_3N$  (equation 5).¹³



Acylketenes with bulky substituents are long-lived, and *tert*-butyl(carboethoxy) ketene **16**, prepared by dehydrochlorination (equation 6), was persistent,¹⁴ as were the corresponding methyl ester (IR  $2129 \text{ cm}^{-1}$ )¹⁵ and 1-adamantyl(carboethoxy) ketene (**17**), prepared as a solution in hexane by Wolff rearrangement (equation 7).¹⁶



Dipivalylketene **19** was prepared from pyrolysis of the furandione **18**, and while stable in solution at -20 °C, it reacted rapidly with MeOH to give the ester, and dimerized by a [4 + 2] pathway to give ketene **20** as a stable species, for which an X-ray structure was obtained (equation 8).¹⁷ The diverse chemistry found for **20** included differing reactions with amines¹⁸ and nucleophilic reactions leading to 2,4,6,8-tetraoxaadamantanes,¹⁹ and this has been reviewed.²⁰ The mechanism of thermolysis of furandiones has been studied experimentally, using IR detection, and by computation.²¹



An alternative mode of [4+2] dimerization of **19** across a C=O bond was induced by DMSO, *n*-Bu₃PO, or pyridine,²² and was proposed to result from an intermediate zwitterion forming **20a** (equation 9).²² Reactions of **19** with various RNH₂ reagents were observed.²³



Stereoisomeric diazo ketones **21** formed the corresponding acylketenes **22** on thermolysis, and the ketenes were isolated and characterized (equation 10).²⁴ The stereochemistry of the reactions of **22** with alcohols and amines was examined.²⁴ Acyclic diazo ketones with bulky aryl groups reacted with alcohols or amines to form enols whose molecular dynamics were studied by NMR.²⁵



Dibenzoylketene **24** was prepared from pyrolysis of the furandione **23** and trapped with amines,²⁶ and its IR absorption at 2140 cm⁻¹ was observed in a matrix at -196 °C (equation 11).²⁷ On warming, **24** gave the dimer resulting from [4 + 2] cycloaddition forming **25**, and this led to **26** and **27** (equation 12).²⁷



The diacylketene **29** was formed as a fairly persistent neat liquid by decarbonylation upon flash pyrolysis of furan-2,3-dione **28**, and gave [4 + 2] cycloaddition with a ketenimine involving reaction of the less hindered but normally less reactive ester carbonyl, forming **30** (equation 13).²⁸ When **29** was cogenerated with **31** the two ketenes underwent a [4 + 2] cycloaddition, forming **32** (equation 14).²⁹



Acylketenes have also been generated by the thermolysis or photolysis of dioxinones.^{30–35} For **33** with a side chain bearing a hydroxyl group forming the ketene **34**, this was a useful method for cyclization forming lactone **35** (equation 15).³¹



In other applications^{36–39} of the cyclization of acylketenes generated by the thermolysis of dioxinones bearing hydroxy or amino side chains, macrocyclic lactones or lactams were formed. The dioxinone **36** formed ketene **37**, which gave not only the 12-membered ring lactone **38** in 28% yield, but also the dimeric and trimeric 24- and 36-membered lactones in 41 and 12% yields, respectively (equation 16).³⁶ Efforts to form an analogous 11-membered ring lactone **40** by thermolysis of ketene **39** (equation 17) were unsuccessful, and only dimeric, trimeric, and higher molecular weight lactones were observed. Thus, thermolysis of **41** to form ketene **42** gave a series of cyclic oligolides **43**, and for n = 1-6 these were isolated and characterized as 22- to 77-membered rings by ¹H and ¹³C NMR and by mass

# spectroscopy (equation 18).³⁷



Flash vacuum thermolysis of  $\beta$ -keto esters **44** formed acylketenes **46** observed by IR spectrocopy at 12 K. It was proposed that these reactions involved initial enolization to **45**, as the more readily enolized substrates reacted more readily, and the residual starting material was enriched in the keto form (equation 19).^{40–44}



Pyrolysis of 2,4-diketo esters **47** at 170–190 °C took place with elimination and decarbonylation forming acylketenes **48**, which were captured by the alcohol forming esters **49** (equation 20).^{45,46} The reactions were proposed to occur by pericyclic elimination with decarbonylation, and the stretching frequencies of the ketenes in the region 2120–2140 cm⁻¹ were observed by GC-FTIR when the reactions were

carried out at 280 °C.



The addition of oxygen nucleophiles to acyl ketenes has the potential for inplane attack on the side of the acyl oxygen through pseudopericyclic transition states, as in the reaction of ketene **50**, for which the transition state **51** leading to the enol ester **52** was proposed (equation 21).^{25,47,48} A similar mechanism was put forward to account for the hydration of ketene **16**,⁴⁹ and this reaction was modeled with *ab initio* calculations.^{49,50}



The cyclization of formylketene **5** to **53** was calculated to be endothermic by 17 kcal/mol (equation 22),⁵¹ but cyclization of acetylketene **1** was proposed to occur to account for the products formed upon pyrolysis of **54** forming propyne via the intermediacy of **55** (equation 23).⁵² FVP of **56** was proposed to form carbomethoxyketene (**57**), which was captured with benzyl alcohol (equation 24).⁵³



Isomerization of cyclopropenone carbonyl oxide **58** to formylketene **5** was calculated at the MRDCI + Q level to have a barrier of 32.0 kcal/mol and to be exothermic by 117.5 kcal/mol (equation 25).⁵⁴ The geometry and vibrational spectra of **5** were calculated using improved methodology.⁵⁵



FVP of dioxinone **59** with matrix isolation gave IR spectra that exhibited bands due to two conformations of acetylketene (1), and these spectra were in good agreement with calculations (equation 26).⁵⁶ Photolysis gave new bands attributed to photochemically populated matrix sites.



Transesterification of  $\beta$ -keto esters **60** with alcohols in refluxing toluene over aluminosilicates occurred by formation of acetylketene (1) as an intermediate, which led to the ester **61** (equation 27).⁵⁷



Computations of the reaction of diformylketene (62) with formamide found a preference for a one-step asynchronous reaction through a pseudopericyclic transition state 63 with the formyl group of the formamide above the plane leading to the product 64 (equation 28).⁵⁸ Computations for the reaction in benzene solvent indicated that there was little change compared to the gas phase in the structures.⁵⁸



The formation of acylketenes **65** and **67** from the mesoionic species **64** and **66**, respectively, was calculated to be exothermic by about 2 kcal/mol in each case (equations 29, 30).⁵⁹ The formation of  $\alpha$ -amidoketenes R¹C(CONR)=C=O from the reaction of alkynes with CO₂ is discussed in Section 3.8.



FVT of mesoionic species **68** and matrix isolation resulted in formation of the unobserved acylketenes **69**, which interconverted with **68** (equation 31).⁶⁰ Wolff-type rearrangements of aryliodonium ylides **70** provided a route to diacylketenes **71**, which were trapped with amines, giving amides **72** (equation 32).^{61,62}



The rapid hydrolysis of 2-acetoxybenzoic acid **73** was proposed⁶³ to occur with intramolecular catalysis, with formation of the unobserved acylketene **74** (equation 33), which formed an observed acid enol whose rate of ketonization was measured.⁶³ Ketene **77** was generated by photochemical Wolff rearrangement of the diazo ketone **75** and also by photolysis of the dioxinone **76**, and upon hydration formed **78** (equation 34).⁶⁴ The hydration to the enol and ketonization of the enol were studied and compared to the reactions of the corresponding 5-membered

ring analogue.



The reaction of acetyl chloride with  $AlCl_3$  in  $CD_2Cl_2$  formed protonated diacetyl-ketene tetrachloroaluminate (79), which was characterized by ¹H- and ¹³C-NMR and by IR (equation 35).⁶⁵



The reaction of pivaloylethoxyacetylene (80) with Me₃SiCl led to the acylketene 81, and germanium halides reacted similarly, while Ph₂PCl formed the dimer 82 (equation 36).⁶⁶



FVP of the Meldrum's acid derivative **83** gave the carboxylketene **84**, as identified by FTIR, and this underwent decarboxylation to **85** (equation 37).^{67,67a} The corresponding ethoxy cumulated ketene was formed from **86** and then gave **87** (equation 38).^{67a} An *Organic Syntheses* procedure for preparation of 2-alkyl-4-pyrones **90** involves heating of the Meldrum's acid derivative **88**, which was proposed to lose acetone to generate the acylketene **89**, which was captured by *n*-butyl vinyl ether followed by decarboxylation and acid-catalyzed elimination (equation 39).^{68–70} A related generation of an acylketene from a hydroxymethylene Meldrum's acid derivative is shown in Section 3.4.6.⁷¹



Reaction of the methylthio derivative **91** was proposed to form **92** and then the ketene **93** (equation 40), which led to the thiophene derivative **94** (equation 40a).^{67a} Ketene **94a** from dioxinone **88** reacted with Boc(BocO)NH, leading to the  $\beta$ -keto hydroxamic acid **94b**, and was cyclized to a 3-isoxazole (equation 41).⁷²



Intramolecular rearrangements in acylketenes 95 have been examined by *ab initio* molecular orbital calculations. Particularly for groups such as  $Me_2N$  and

MeS with lone pair electrons available for donation, these were predicted to occur at reasonable temperatures through a transition state **96** (equation 42).^{73–75} 1,3-Shift of Cl in system **97** (equation 43) was also calculated⁷⁶ to be accelerated, as was a 1,5-Cl shift in O=C=CHCH=CHCCl=O (**99**) through the transition state **100** (equation 44). Measurements by ¹³C NMR show the 1,3-Cl shift in **97** has  $\Delta G^{\ddagger}$  of only 10.0 kcal/mol, in excellent agreement with the results of molecular orbital calculations (equation 43).⁷⁷ Calculations of the conformations and reactivities of a variety of acylketenes have also been reported.⁷⁸



The kinetics of the hydration of acetylketene **1** generated by flash photolysis of the dioxinone **59** (equation 26) were measured, and this ketene was remarkably reactive, with a rate constant of  $1.5 \times 10^6 \text{ s}^{-1}$  at 25 °C.⁷⁹ The initial product was acetoacetic acid enol (**101**), which formed acetoacetic acid (**102**) by a process whose mechanism was also elucidated (equation 45).⁷⁹ These studies did not provide evidence regarding the carbonyl-assisted hydration of acetylketene, and this question is still unanswered. The cyclic analogue **104** prepared by Wolff rearrangement of the diazo ketone **103** had  $k(\text{H}_2\text{O}) = 1.4 \times 10^6 \text{ s}^{-1}$ , almost identical to that of **1** (equation 46),⁸⁰ and the trifluoromethyl analogue **106** had a similar rate constant of  $1.75 \times 10^6 \text{ s}^{-1.81}$ 



Calculations of the cycloaddition of formylketene (3) with ethylene showed that planar and nonplanar transition states to form 107 had almost equal energy, whereas the reaction with acetylene to form 108 favored a nonplanar transition state (equations 47, 48).⁸²



Dehydrochlorination of adipyl dichloride was proposed to form the acylketene **104**, which underwent [4 + 2] cycloaddition with diphenylketene (**109**) to give the adduct **110** (equation 49).⁸³



The transition state **111** for [4 + 2] cycloaddition of acetylketene **1** plus acetone forming the dioxinone **59** has been calculated to have the bonding atoms essentially coplanar (equation 50).⁸⁴ This geometry was expected for in-plane attack on the ketenyl moiety. Computational studies of the reaction of **1** with aldehydes bearing chiral substituents have shown this to proceed with Felkin-Anh stereo-selectivity.⁸⁵



Photolysis of diazo diketone **112** formed phenylbenzoylketene (**113**), which reacted with amino acid derivatives (equation 51).⁸⁶ Photolysis of the diazo ketone **114** with an ionic side chain in the presence of DNA gave cleavage of the DNA at guanine.^{86–88} Further examples of DNA cleavage by ketenes

are given in Section 5.4.1.2.



Photochemical Wolff rearrangements of diazo ketone derivatives **115** gave carboxamide-substituted ketenes **116** (equation 52) that were trapped by nucleophiles, and it was proposed that the diazopyruvyl derivatives could be attached to thiol groups of proteins to give photoactivatable cross-linking agents.⁸⁹



Diazo ketone **117** gave a photochemical or thermal reaction leading to formation of diacylketene **119** by carbene capture by CO in matrices at 6 K of the carbene **118** competitive with Wolff rearrangement, forming **120** (equation 53).⁹⁰



Staudinger and Herzel prepared di(carboethoxy)ketene **122** in 1916 as a distillable liquid by the Wolff rearrangement of diazo ketone **121** (equation 54).⁹¹ This ketene reacted slowly with moisture and dimerized at room temperature,⁹² and underwent [4 + 2] cycloaddition with a methyl silylalkynyl ether, forming **122a** (equation 54).⁹³ 4-Dimethylaminopyridine (DMAP) reacted with **122** to form the zwitterion **123** as a solid that formed **124** on heating (equation 55).⁹⁴





The reaction of **97** with 3-alkenyl  $\delta$ -lactams **125** gave polycyclic lactones **126** that were thermally decarboxylated to **127** (equation 56).⁹⁵ The reaction of **97** with oximes formed heterocyclic mesomeric betaines.⁹⁶



Trifluoroacetyl Meldrum's acid, which exists in the enol form **128**, reacted with acetone at room temperature to give the corresponding dioxinone **129** in a process proposed to involve trifluoroacetylketene (**106**, equation 57).⁹⁷



Dibenzoyldiazomethane with rhodium acetate formed benzoylphenylketene (113), which reacted with nitriles by [4+2] cycloaddition to form 1,3-oxazine-4-ones (130, equation 58).⁹⁷ Trifluoroacetylketene (106) from dioxinone 129 reacted with the nitrile 131 to give the 1,3-oxazin-4-one 132 (equation 59).⁹⁸





The reactivity of aroylketenes **134** formed by thermolysis of 5-aryl-2,3-dihydrofuran-2,3-diones **133** with ethyl vinyl ether, forming **135**, was enhanced by electron-withdrawing aryl substituents (equation 60).⁹⁹ Bis(4-methoxybenzoyl) ketene (4-MeOC₆H₄CO)₂C=C=O) was generated by thermolysis of the 4-(4methoxybenzoyl)-5-(4-methoxyphenyl)furan-2,3-dione and reacted by [4 + 2] cycloadditions.¹⁰⁰ Reaction mechanisms of decomposition of **133** have been observed by IR study of the intermediates formed.¹⁰¹



Thermolysis of **136** formed ketene **137**, which cyclized to **138** (equation 61).¹⁰² Some related examples of acylketene formation from 2,3-dihydrofuran-2,3-diones were also examined.^{103,104} Camphorketene **140** formed by dehydrochlorination of **141** gave the dimer **142** (equation 62), whose stereochemistry was established by X-ray crystallography.¹⁰⁵ The dimerization was reported to occur through a planar pseudopericyclic pathway.¹⁰⁵





Pyrolysis of **144** with matrix trapping permitted the observation of IR of formylthioketene **145**, which rearranged through the hydrogen migration transition state **146** to thioformylketene **147** (equation 63).¹⁰⁶ Matrix photolysis of furandione **148** gave benzoylketene (**149**) by extrusion of CO (equation 64).¹⁰⁷



### **REFERENCES FOR SECTION 4.1.6**

- 1. Chick, F.; Wilsmore, N. T. M. J. Chem. Soc. 1908, 93, 946-950.
- 2. Staudinger, H.; Bereza, S. Chem. Ber. 1909, 42, 4908-4918.
- 3. Wentrup, C.; Heilmayer, W.; Kollenz, G. Synthesis 1994, 1219-1248.
- 4. Sato, M.; Iwamoto, K. J. Synth. Org. Chem. Jpn. 1999, 57, 76-83.
- Kappe, T. In *Encyclopedia of Reagents for Organic Synthesis*; Paquette, L. A., Ed.; Wiley, New York, 1995, pp. 1098–1100.
- 6. Gong, L.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1991, 113, 6021–6028.
- 7. McAllister, M. A.; Tidwell, T. T. J. Org. Chem. 1994, 59, 4506-4515.
- Allen, A. D.; Andraos, J.; Kresge, A. J.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1992, 114, 1848–1879.
- 9. Maier, G.; Reisenauer, H. P.; Sayrac, T. Chem. Ber. 1982, 115, 2192-2201.
- Wenkert, C.; Ananthanarayan, T. P.; Ferreira, V. F.; Hoffmann, M. G.; Kim, H. J. Org. Chem. 1990, 55, 4975–4976.
- 11. Nakanishi, S.; Butler, K. Org. Prep. Proc. Int. 1975, 7, 155-158.
- 12. Potts, K. T.; Kanemasa, S.; Zvilichovsky, G. J. Am. Chem. Soc. 1980, 102, 3971–3972.
- 13. Potts, K. T.; Murphy, P. M.; Kuehnling, W. R. J. Org. Chem. 1988, 53, 2889-2898.
- 14. Newman, M. S.; Zuech, E. A. J. Org. Chem. 1962, 27, 1436-1438.
- 15. Leung-Tong, R.; Wentrup, C. Tetrahedron 1992, 48, 7641-7654.
- 16. Ohno, M.; Itoh, M.; Ohashi, T.; Eguchi, S. Synthesis 1993, 793-796.
- Kappe, C. O.; Evans, R. A.; Kennard, C. H. L.; Wentrup, C. J. Am. Chem. Soc. 1991, 113, 4234–4237.

- Kappe, C. O.; F\u00e4rber, G.; Wentrup, C.; Kollenz, G. Tetrahedron Lett. 1992, 33, 4553– 4556.
- Heilmayer, W.; Dalvi, T. S.; Kappe, C. O.; Wentrup, C.; Gruber, K.; Sterk, H.; Kollenz, G. J. Chem. Soc., Chem. Commun. 1995, 797–798.
- 20. Kollenz, G.; Heilmayer, W.; Kappe, C. O.; Wallfisch, B.; Wentrup, C. Croatia Chem. Acta 2001, 74, 815–823.
- 21. Aycard, J.-P.; Bodot, H.; Hillebrand, M.; Hnach, M.; N'Guessan, T. Y.; Segal, E.; Purcea, I.; Volanschi, E. J. Chem. Res. (S), **1995**, 94–95.
- 22. Kappe, C. O.; Färber, G.; Wentrup, C.; Kollenz, G. J. Org. Chem. 1992, 57, 7078–7083.
- 23. Kollenz, G.; Kappe, C. O.; Dalvi, T. S.; Wentrup, C. Arkivoc 2001, (vi) 30-41.
- Nikolaev, V. A.; Korneev, S. M.; Terent'eva, S. M.; Korobitsyna, I. K. Zh. Org. Khim. 1991, 27, 2085-2100. Engl. Transl. 1991, 27, 1845–1858.
- 25. Buss, M.; Mayer, A.; Müller, K.; Meier, H. Liebigs Ann. 1996, 1223-1229.
- 26. Ziegler, E.; Kollenz, G.; Ott, W. Synthesis 1973, 679-680.
- 27. Wentrup, C.; Winter, H.-W.; Gross, G.; Netsch, K.-P.; Kollenz, G.; Ott, W., Biedermann, A. G. Angew. Chem. Int. Ed. Engl. 1984, 23, 800–802.
- 28. Stadler, A.; Zangger, K.; Belaj, F.; Kollenz, G. Tetrahedron 2001, 57, 6757–6763.
- 29. Wallfisch, B. C.; Belaj, F.; Wentrup, C.; Kappe, C. O.; Kollenz, G. J. Chem. Soc. Perkin Trans. 1 2002, 599–605.
- 30. Clemens, R. J.; Witzeman, J. S. J. Am. Chem. Soc. 1989, 111, 2186-2193.
- 31. Boeckman, R. K., Jr.; Pruitt, J. R. J. Am. Chem. Soc. 1989, 111, 8286-8288.
- 32. Coleman, R. S.; Grant, E. B. Tetrahedron Lett. 1990, 31, 3677-3680.
- 33. Coleman, R. S.; Fraser, R. S. J. Org. Chem. 1993, 58, 385-392.
- 34. Sato, M.; Sakaki, J.; Sugita, Y.; Yasuda, S.; Sakoda, H.; Kaneko, C. *Tetrahedron* **1991**, *47*, 5689–5708.
- 35. Sakaki, J.; Sugita, Y.; Sato, M.; Kaneko, C. Tetrahedron 1991, 47, 6197-6214.
- Chen, C.; Quinn, E. K.; Olmstead, M. M.; Kurth, M. J. J. Org. Chem. 1993, 58, 5011– 5014.
- 37. Eisenberg, S. W. E.; Chen, C.; Wu, J.; Lebrilla, C.; Kurth, M. J. *Tetrahedron Lett.* **1996**, *37*, 7683–7686.
- Boeckman, R. K., Jr.; Weidner, C. H.; Perni, R. B.; Napier, J. J. J. Am. Chem. Soc. 1989, 111, 8036–8037.
- Boeckman, R. K., Jr.; Barta, T. E.; Nelson, S. G. Tetrahedron Lett. 1991, 32, 4091– 4094.
- 40. Freiermuth, B.; Wentrup, C. J. Org. Chem. 1991, 56, 2286-2289.
- 41. Leung-Toung, R.; Wentrup, C. Tetrahedron 1992, 48, 7641-7654.
- 42. Campbell, D. S.; Lawrie, C. W. Chem. Commun. 1971, 355-356.
- 43. Witzeman, J. S. Tetrahedron Lett. 1990, 31, 1401-1404.
- 44. Witzeman, J. S.; Nottingham, W. D. J. Org. Chem. 1991, 56, 1713-1718.
- 45. Emerson, D. W.; Titus, R. L.; González, R. M. J. Org. Chem. 1990, 55, 3572-3576.
- 46. Emerson, D. W.; Titus, R. L.; González, R. M. J. Org. Chem. 1991, 56, 5301– 5307.

- 47. Meier, H.; Wengenroth, H.; Lauer, W.; Vogt, W. Chem. Ber. 1988, 121, 1643-1646.
- Meier, H.; Wengenroth, H.; Lauer, W.; Krause, V. *Tetrahedron Lett.* 1989, 30, 5253– 5256.
- 49. Allen, A. D.; Tidwell, T. T. Tetrahedron Lett. 1993, 34, 1095-1098.
- 50. Birney, D. M.; Wagenseller, P. E. J. Am. Chem. Soc. 1994, 116, 6262-6270.
- 51. Nguyen, M. T.; Ha, T.; More O'Ferrall, R. A. J. Org. Chem. 1990, 55, 3251-3256.
- 52. Biesiada, K. A.; Shevlin, P. B. J. Org. Chem. 1984, 49, 1151-1153.
- 53. Jullien, J.; Pechine, J. M.; Perez, F. Tetrahedron Lett. 1983, 24, 5525-5526.
- 54. Anglada, J. M.; Bofill, J. M. J. Org. Chem. 1997, 62, 2720-2726.
- 55. Badawi, H. M.; Förner, W.; Al-Saadi, A. J. Mol. Struct. (Theochem) 2000, 505, 19-30.
- 56. Zuhse, R. H.; Wong, M. W.; Wentrup, C. J. Phys. Chem. 1996, 100, 3917-3922.
- 57. Sasidharan, M.; Kumar, R. J. Mol. Catalysis A 2004, 210, 93-98.
- Montero-Campillo, M. M.; Rodríguez-Otero, J.; Cabaleiro-Lago, E. M. J. Phys. Chem A 2004, 108, 8373–8377.
- Böttcher, A.; Debaerdemaker, T.; Radziszewski, J. G.; Friedrichsen, W. Chem. Ber. 1988, 121, 895–907.
- 60. Fiksdahl, A.; Plüg, C.; Wentrup, C. J. Chem. Soc., Perkin Trans. 2 2000, 1841-1845.
- 61. Malamidou-Xenikaki, E.; Spyroudis, S.; Tsanakopoulou, M. J. Org. Chem. 2003, 68, 5627–5631.
- 62. Spyroudis, S.; Xanthopoulou, M. J. Org. Chem. 2002, 67, 4612.
- 63. Kirby, A. J.; Meyer, G. J. Chem. Soc., Perkin Trans. 2, 1972, 1446-1451.
- Chang, J. A.; Kresge, A. J.; Nikolaev, V. A.; Popik, V. V. J. Am. Chem. Soc. 2003, 125, 6478–6484.
- Csihony, S.; Bodor, A.; Rohonczy, J.; Horvath, I. T. J. Chem. Soc. Perkin Trans. 1, 2002, 2861–2865.
- Lukashev, N. V.; Fil'chikov, A. A.; Kazankova, M. A.; Beletskaya, I. P. *Heteroatom. Chem.* 1993, 4, 403–407.
- 67. Wentrup, C.; Lorencak, P. J. Am. Chem. Soc. 1988, 110, 1880-1883.
- 67a. Bibas, H.; Kappe, C. O.; Wong, M. W.; Wentrup, C. J. Chem. Soc., Perkin Trans. 2 1998, 493–498.
  - 68. Crimmins, M. T.; Washburn, D. G.; Zawacki, F. J. Org. Synth. 1999, 77, 114-120.
  - 69. Zawacki, F. J.; Crimmins, M. T. Tetrahedron Lett. 1996, 37, 6499-6502.
  - 70. Sato, M.; Ban, H.; Kaneko, C. Tetrahedron Lett. 1997, 38, 6689-6692.
  - Xu, F.; Armstrong, J. D., III; Zhou, G. X.; Simmons, B.; Hughes, D.; Ge, Z.; Grabowski, E. J. J. J. Am. Chem. Soc. 2004, 126, 13002–13009.
  - 72. Sørensen, U. S.; Falch, E.; Krogsgaard-Larsen, P. J. Org. Chem. 2000, 65, 1003-1007.
  - 73. Wong, M. W.; Wentrup, C. J. Org. Chem. 1994, 59, 5279-5285.
  - Wentrup, C.; Fulloon, B. E.; Moloney, D. W. J.; Bibas, H.; Wong, M. W. Pure Appl. Chem. 1996, 68, 891–894.
  - 75. Nguyen, M. T.; Landuyt, L.; Nguyen, H. M. T. Eur. J. Org. Chem. 1999, 401-407.
  - 76. Koch, R.; Wong, M. W.; Wentrup, C. J. Org. Chem. 1996, 61, 6809-6813.
  - Finnerty, J.; Andraos, J.; Yamamoto, Y.; Wong, M. W.; Wentrup, C. J. Am. Chem. Soc. 1998, 120, 1701–1704.

- Janoschek, R.; Fabian, W. M. F.; Kollenz, G.; Kappe, C. O. J. Comput. Chem. 1994, 15, 132–143.
- Chiang, Y.; Guo, H.-X.; Kresge, A. J.; Tee, O. S. J. Am. Chem. Soc. 1996, 118, 3386– 3391.
- Chiang, Y.; Kresge, A. J.; Nikolaev, V. A.; Popik, V. V. J. Am. Chem. Soc. 1997, 119, 11183–11190.
- Chiang, Y.; Kresge, A. J.; Meng, Q.; Morita, Y.; Yamamoto, Y. J. Am. Chem. Soc. 1999, 121, 8345–8351.
- 82. Wagenseller, P. E.; Birney, D. M; Roy, D. J. Org. Chem. 1995, 60, 2853-2859.
- 83. Jäger, G. Chem. Ber. 1972, 105, 137-149.
- 84. Eisenberg, S. W. E.; Kurth, M. J.; Fink, W. H. J. Org. Chem. 1995, 60, 3736-3742.
- Shumway, W.; Ham, S.; Moer, J.; Whittlesey, B. R.; Birney, D. M. J. Org. Chem. 2000, 65, 7731–7739.
- 86. Nakatani, K.; Shirai, J.; Tamaki, R.; Saito, I. Tetrahedron Lett. 1995, 36, 5363-5366.
- 87. Nakatani, K.; Shirai, J.; Sando, S.; Saito, I. J. Am. Chem. Soc. 1997, 119, 7626-7635.
- 88. Saito, I.; Nakatani, K. Bull. Chem. Soc. Jpn. 1996, 69, 3007-3019.
- 89. Mosier, G. G. J.; Lawton, R. G. J. Org. Chem. 1995, 60, 6953-6958.
- Visser, P.; Zuhse, R.; Wong, M. W.; Wentrup, C. J. Am. Chem. Soc. 1996, 118, 12598– 12602.
- 91. Staudinger, H.; Herzel, H. Chem. Ber. 1916, 49, 2522-2529.
- 92. Staudinger, H.; Herzel, H. Chem. Ber. 1917, 50, 1024-1035.
- 93. Himbert, G.; Henn, L. Liebigs Ann. 1987, 381-383, 771-776.
- 94. Gompper, R.; Wolf, U. Liebigs Ann. 1979, 1388-1405.
- Potts, K. T.; Rochanapruk, T.; Padwa, A.; Coats, S. J.; Hadjiarapoglou, L. J. Org. Chem. 1995, 60, 3795–3805.
- 96. Saidi, K.; Sheibani, H. Synth. Commun. 2001, 31, 1809-1814.
- 97. Yamagata, K.; Ohkubo, K.; Yamazaki, M. Liebigs Ann. 1995, 187-190.
- 98. Morita, Y.; Kamakura, R.; Takeda, M.; Yamamoto, Y. Chem. Commun. 1997, 359-360.
- 99. Toda, J.; Saitoh, T.; Oyama, T.; Horiguchi, Y.; Sano, T. *Heterocycles* **1996**, *43*, 2457–2464.
- 100. Saripinar, E.; Guzel, Y.; Onal, Z.; Ilhan, I. O.; Akcamur, Y. J. Chem. Soc. Pakistan 2000, 22, 308–317.
- 101. Aycard, J.-P.; Bodot, H.; Hillebrand, M.; Hnach, M.; N'Guessan, T. Y.; Segal, E.; Purcea, I.; Volanschi, E. J. Chem. Res. (S), 1995, 94–95.
- 102. Aliev, Z. G.; Krasnykh, O. P.; Maslivets, A. N.; Stepanov, O. S.; Andreichikov, Y. S.; Atovmyan, L. O. *Russ. Chem. Bull.* **1999**, *48*, 2127–2130.
- 103. Lisovenko, N. Y.; Maslivets, A. N.; Aliev, Z. G. Russ. J. Org. Chem. 2004, 40, 1053-1057.
- 104. Vostrov, E. S.; Leont'eva, E. V.; Tarasova, O. P.; Maslivets, A. N. Russ. J. Org. Chem. 2004, 40, 1058–1061.
- 105. Shumway, W. M.; Dalley, N. K.; Birney, D. M. J. Org. Chem. 2001, 66, 5832-5839.
- 106. Pedersen, C. Th.; Wong, M. W.; Flammang, R. J. Chem. Soc., Perkin Trans. 2 2001, 2047–2052.
- 107. Huben, K.; Zhu, Z.; Bally, T.; Gebicki, J. J. Am. Chem. Soc. 1997, 119, 2825-2831.

#### 4.1.7 Imidoylketenes

Imidoylketenes resemble acylketenes in structure, but are quite reactive and are not normally observable intermediates. The structure of imidoylketene **1** and of its cyclization product, the azetinone **2**, were calculated at the HF/6-31G*//HF/6-31G* level, and the ring closure was found to be endothermic by 12.0 kcal/mol (equation 1).¹ *Ab initio* calculations of the various conformations of imidoylketene (**1a–d**) have been reported.²



Calculations of the structures and energies for the hydration of 1 forming 4 (equation 2), and [4+2] cycloaddition with formaldehyde forming 6 (equation 3), have been reported.² The reactions of equations 2 and 3 involve pseudopericyclic transition states 3 and 5, respectively, which are essentially coplanar, but without continuous pi-electron delocalization, and these reactions are allowed for any number of electrons.



Imidoylketenes are isomers of acylketenimines. The 1,3-hydrogen shift of 1 to form ketenimine 8 is a pseudopericyclic process that has been studied

computationally and involves the 4-membered ring transition state 7 (equation 4).²



Experimentally, the interconversion of the imidoylketene **10** generated by FVPinduced decarbonylation of the pyrrolidione **9** forming the ketenimine **11** by methoxy group migration, was demonstrated by trapping of **11** in an argon matrix, with identification by low temperature IR spectroscopy and isolation from reaction at  $350 \,^{\circ}$ C (equation 5).^{3,4} Similar rearrangements have been studied by MINDO/3.⁵ The quinolone **12** was formed by Conrad-Limpach cyclization of imidoylketene **10** (equation 6).^{6,7}



Ketene **10** and other analogues were also formed from Meldrum's acid derivatives such as **13** above 200 °C (equation 7).^{4,8} The cumulenone **14** was also formed in this reaction, as identified by the IR bands at 2140 (sh) and 2222 cm⁻¹ (equation 7).^{4,8} Chemical generation of an imidoylketene was used in syntheses of pharmaceutically useful 4-arylquinolines.⁹



Pyrolysis of the analogous Meldrum's acid derivative **15** gave the imidoylketene **16**, which underwent reversible MeS group migration, forming the ketenimine

(equation 8).¹⁰ Imidoylketene **16** was also formed from the pyrrolidione analogous to  $9.^9$  Cyclization of an imidoylketene to an azetin-2-one (cf. **2**) was also observed.¹¹



Computations at the B3LYP/6-311+G**//B3LYP/6-31G* level indicated that rearrangement of the  $\alpha$ -imidoylketene **17** via the transition state **18** to the more stable  $\alpha$ -oxoketenimine **19** has a low activation barrier of 62 kJ/mol or 14.8 kcal/ mol (equation 9).¹³



FVT of **20a,b** formed the transient eneaminylketenes **21a,b** identifed by weak IR bands at 2130 and 2115 cm⁻¹ in a matrix (equation 10).¹⁴ The ketenes cyclized to pyrazolium oxides **22a,b** (equation 10), which underwent methyl migration to form **23a,b** (equation 11).¹⁴



Zhou and Birney have carried out experimental and theoretical studies of dimerizations of imidoylketenes (see Section 5.4.3).^{14,15}

Photolysis of 2-phenylisoxazol-5(2H)-one (24) gave the nitronate-substituted ketene 25, as evidenced by the capture of the ester 27, proposed to result from photochemical rearrangement of the methanol addition product 26

(equations 11, 12).¹⁶



Photolysis of oxazinone **28** and substituted derivatives gave ketenes **29** observed spectroscopically at temperatures below -160 °C (equation 14).^{17,18} These rearranged by hydrogen migration, forming isocyanates **30** (equation 14).^{17,18} Matrix photolysis of pyrimidones **31** gave imino analogues **32** (equation 15).¹⁹



Cycloaddition of acylisocyanate **33** with the alkynyl ether **34** was proposed to form imidoylketene **36** (equation 16), which was converted to the observed **37** (equation 17).²⁰ The interconversion of the *N*-formylazetinone **38** to the iminoketene **39**, and the further ring closure to the 1,3-oxazinon-6-one **28**, were studied computationally (equation 18).²¹





# **REFERENCES FOR SECTION 4.1.7**

- Nguyen, M. T.; Ha, T.-K.; More O'Ferrall, R. A. J. Org. Chem. 1990, 55, 3251– 3256.
- 2. Ham, S.; Birney, D. M. J. Org. Chem. 1996, 61, 3962-3968.
- Fulloon, B.; El-Nabi, H. A. A.; Kollenz, G.; Wentrup, C. Tetrahedron Lett. 1995, 36, 6547–6550.
- Wentrup, C.; Rao, V. V. R.; Franck, W.; Fulloon, B. E.; Moloney, D. W. J.; Mosandl, T. J. Org. Chem. 1999, 64, 3608–3619.
- 5. Li, Y.-H.; Hong, S.-G. Chin. J. Chem. 1996, 14, 105-108.
- 6. Fulloon, B. E.; Wentrup, C. J. Org. Chem. 1996, 61, 1363-1368.
- 7. Blatter, H. M.; Lukaszewski, H. Tetrahedron Lett. 1964, 855-861.
- Andrews, I. P.; Bannister, R.; Etridge, S. K.; Lewis, N. J.; Mullane, M. V.; Wells, A. S. *Tetrahedron Lett.* 1995, *36*, 7743–7746.
- 9. Walz, A. J.; Sundberg, R. J. J. Org. Chem. 2000, 65, 8001-8010.
- Kappe, C. O.; Kollenz, G.; Leung-Toung, R.; Wentrup, C. J. Chem. Soc. Chem. Commun. 1992, 487–488.
- Kappe, C. O.; Kollenz, G.; Netsch, K.-P.; Leung-Toung, R.; Wentrup, C. J. Chem. Soc. Chem. Commum. 1992, 488–490.
- Ebner, S.; Wallfisch, B.; Andraos, J.; Aitbaev, I.; Kislewsky, M.; Bernhardt, P. V; Kollenz, G.; Wentrup, C. Org. Biomol. Chem. 2003, 1, 2550–2555.
- 13. Finnerty, J. J.; Wentrup, C. J. Org. Chem. 2004, 69, 1909-1918.
- 14. Zhou, C.; Birney, D. M. J. Org. Chem. 2004, 69, 86-94.
- 15. Zhou, C.; Birney, D. M. J. Am. Chem. Soc. 2002, 124, 5231-5241.
- 16. Ang, K. H.; Prager, R. H. Tetrahedron 1992, 48, 9073-9084.
- 17. Maier, G.; Schäfer, U. Liebigs Ann. 1980, 798-813.
- Huang, B. S.; Pong, R. G. S.; Laureni, J.; Krantz, A. J. Am. Chem. Soc. 1977, 99, 4154– 4156.
- Lapinski, L.; Nowak, M. J.; Les, A.; Adamowicz, L. J. Am. Chem. Soc. 1994, 116, 1461– 1467.
- Pericás, M. A.; Serratosa, F.; Valenti, E.; Font-Altaba, M.; Solans, X. J. Chem. Soc. Perkin Trans. 2 1986, 961–967.

21. Alajarin, M.; Sánchez-Andrada, P.; Cossío, F. P.; Arrieta, A.; Lecea, B. J. Org. Chem. 2001, 66, 8470–8477.

#### 4.1.8 Cumulene-Substituted Ketenes

Cumulene-substituted ketenes include allenylketene 1, keteniminylketene 2, diazomethylketene 3, and isocyanatoketene 4. 1,2-Bisketenes are considered in Section 4.9.



The structures and energies of allenylketene **1** and the cyclization product 4methylenecyclobutenone **5** were calculated at the MP2/6-31G*//MP2/6-31G* level. The allenyl ketene *anti*-**1** was less stable than *syn*-**1** by 1.1 kcal/mol, which was less stable than **5** by 11.8 kcal/mol, with a barrier for ring closure to **5** of 33.7 kcal/mol (equation 1).^{1–3}



Photolysis of the diazo ketone **6** in an argon matrix was reported to form **1** as well as the methylenebicyclobutanone **7** (equation 2).⁴ Pyrolysis of the furan **8** gave **1**, which formed **5** as the observed product (equation 3).⁵



Photolysis of the acyl chloride **9** in an argon matrix gave the parent allenylketene **1**, identified by its IR band at 2127 cm⁻¹ (equation 4).⁶ Further photolysis gave the carbene **10**, and the structures of the ketene and carbene were calculated.⁶ The reaction of **9** with 1,8-bis(dimethylamino)naphthalene in toluene also gave

**1**, observed by IR, and reaction of **1** with the aminoxyl radical TEMPO (TO•) gave the product **11** of diaddition with allylic rearrangement (equation 5).⁷



FVP of **12** was proposed to form the isobenzofuran **13**, which was converted by successive rearrangements to **14**, giving ketene **15**, which led to the observed **16**, along with 33% of isomeric products (equations 6, 7).⁸ An analogous reaction gave **18** in 48% yield through the intermediacy of **17** (equation 8).⁹



Photolysis of the cyclopentadienone **19** resulted in rearrangement forming the allenylketene **20**, identified by its IR absorption at 1890 and 2080 cm⁻¹, and further photolysis gave the cumulene **21** (equation 9).¹⁰ Persistent allenylketenes **23** stabilized by Me₃Si groups were formed on photolysis of methylenecyclobutenones **22** (equation 10).¹ The X-ray structure of **23b** confirmed that these species prefer the coplanar *anti* conformation, as predicted by computations (Figure 4.1).¹ The use of the ketenyl and allenyl substituent stabilization parameters derived earlier allowed the prediction of the energy change for the reaction of equation 10, and



Figure 4.1 X-ray structure of 23b (reprinted from reference 1 with permission of the American Chemical Society).

experimentally measured equilibrium constants were in reasonable agreement with these predictions.¹



The phenyl-substituted allenylketene **25** was formed from **24**, and as expected was less stable relative to the cyclobutenone compared to **23** with R=H (equation 11).¹ The reactivities of **23b** and **25** in nucleophilic, electrophilic, and cycloaddition reactions have been examined.¹¹ Reaction of **23b** with NBS gave **26a** and reaction of **23b** with Me₃SiCHN₂ gave **26b** (equation 12), while reactions with





The bis(methylenecyclobutenone) **30** gave upon photolysis the bis(allenylketene) **31** as a 1:1 mixture of *dl* and *meso* isomers, which upon recrystallization gave a single isomer, with the ketenyl IR at 2080 cm⁻¹ (equation 13).¹



Photolysis of the cyclobutenone **32** with an arylisocyanide was proposed to result in formation of the keteniminylketene **33**, which led to **34** (equation 14).¹²⁻¹⁴



Diazoalkylketenes^{15–18} have been observed at 8 K in argon matrices from the photolysis of bis(diazo) ketones, as in the conversion of **35** to **36**, which was identified by the IR absorption at 2118 and 2120 cm⁻¹ and which gave the cyclopropenone **37** upon further photolysis (equation 15).¹⁵ Photolysis of the benzannulated compound **38** gave the diazo ketene **39**, IR 2106 cm⁻¹ (equation 16).¹⁶



Dehydrochlorination of 4-chloropyrazolin-5-one **40** was proposed to form 1,2diazacyclopentadienone **41**, which gave diazo ketene **42**, leading to bimane **43** (equation 17).^{19,20} Thermal rearrangement of **43** to the *anti*-isomer **44** was proposed to involve initial formation of ketenes **45** and **46** (equation 18).^{19,20}



Reaction of isocyanatoacetyl chloride **47** with triethylamine formed polymeric material, suggesting the intermediacy of isocyantoketene **48** (equation 19).²¹



# **REFERENCES FOR SECTION 4.1.8**

- 1. Huang, W.; Fang, D.; Temple, K.; Tidwell, T. T. J. Am. Chem. Soc. 1997, 119, 2832-2838.
- 2. Gong, L.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1991, 113, 6021-6028.
- 3. McAllister, M. A.; Tidwell, T. T. J. Org. Chem. 1994, 59, 4506-4515.
- Unpublished results of Chapman, O. L.; Chang, C.-C.; Rosenquist, N., cited in Chapman, O. L. Pure Appl. Chem. 1979, 51, 331–339.
- 5. Trahanovsky, W. S.; Park, M.-G. J. Am. Chem. Soc. 1973, 95, 5412.
- 6. Aycard, J.-P.; Allouche, A.; Cossu, M.; Hillebrand, M. J. Phys. Chem. A 1999, 103, 9013–9021.
- 7. Tidwell, T. T.; Fenwick, M. H. Eur. J. Org. Chem. 2001, 3415-3419.
- 8. Chen, P.-S.; Chou, C.-H. Tetrahedron 1997, 53, 17115-17126.
- 9. Chen, P.-S.; Tai, C.-L.; Chou, C.-H. Tetrahedron Lett. 1998, 39, 7381-7384.
- 10. Maier, G.; Lage, H. W.; Reisenauer, H. P. Angew. Chem. Int. Ed. Engl. 1981, 20, 976-977.
- 11. Huang, W.; Tidwell, T. T. Synthesis, 2000, 457-470.
- 12. Obata, N.; Takizawa, T. Chem. Commun. 1971, 587-588.
- 13. Obata, N.; Takizawa, T. Tetrahedron Lett. 1969, 3403-3406.
- 14. Obata, N.; Takizawa, T. Bull. Chem. Soc. Jpn. 1977, 50, 2017-2020.
- 15. Unpublished results of Chapman, O. L.; Roberts, D. C., cited in Chapman, O. L. Pure Appl. Chem. 1979, 51, 331–339.
- Chapman, O. L.; Gano, J.; West, P. R.; Regitz, M.; Maas, G. J. Am. Chem. Soc. 1981, 103, 7033–7036.
- 17. Murata, S.; Yamamoto, T.; Tomioka, H.; Lee, H.; Kim, H.-R.; Yabe, A. J. Chem. Soc. Chem. Commun. 1990, 1258–1260.
- 18. Trost, B. M.; Whitman, P. J. J. Am. Chem. Soc. 1974, 96, 7421-7429.
- 19. Kosower, E. M.; Pazhenchevsky, B. J. Am. Chem. Soc. 1980, 102, 4983-4993.
- 20. Kosower, E. M.; Pazhenchevsky, B.; Hershkowitz, E. J. Am. Chem. Soc. 1978, 100, 6516–6518.
- 21. Mormann, W.; Hoffman, S.; Hoffmann, W. Chem. Ber. 1987, 120, 285-290.

## 4.1.9 Ketenes with Charged, Radical, and Carbenic Side Chains

Ketenes with side chain carbocation (1a), radical (1b), or carbenic (1c) centers at the 3-position are known in a number of examples, consistent with a favorable electron donation from the ketenyl group to electron deficient centers, and display useful chemistry. Analogous carbanions (1d) are rare, as expected, due to an unfavorable interaction of ketenes with carbanionic centers. However, there has been no systematic study of these effects.



	$\begin{array}{c} R\\ C_2 = C_1\\ R^1 - C_3'\\ + \\ R^2 \end{array}$	=0			
	2a				
R	$R^1$	$R^2$	$C_1$	$C_2$	C ₃
Н	Н	Н	147.1	92.7	177.1
CH ₃	Н	Н	151.5	84.3	202.6
CH ₃	$CH_3$	Н	154.2	79.0	223.0
Н	Н	CH ₃	148.1	104.3	172.8
Н	CH ₃	CH ₃	151.3	94.3	193.5
Ph	Ph	Н	158.7	68.7	201.1
Ph	Н	Н	157.0	69.4	183.3
<i>c</i> -Pr	CH ₃	Н	160.4	71.2	227.0
COCI	Н	Н	146.7, 143.1	102.8	165.3, 162.0

TABLE 4.1 ¹³ C NMR	<b>Chemical Shifts</b>	of Acylium	Ions 2
-------------------------------	------------------------	------------	--------

Carbocationic derivatives were generated from a series of acrylyl fluorides with  $SbF_5$  in  $SO_2$  forming the acylium ions **2**, which were characterized by ¹H NMR and IR (equation 1).¹ These had IR bands at 2210–2250 cm⁻¹ for the ketenyl group and were reported to be at somewhat lower frequencies than for alkyl acylium ions, which were seen at 2205–2302 cm⁻¹.^{1,2} These IR bands and the ¹H NMR shifts of CH₃ groups at R¹ and R² were interpreted as supporting a contribution from ketene resonance structure **2a**.^{1,2} The observed ¹³C NMR spectra of cations **2** are given in Table 4.1 and were interpreted in terms of the substantial ketenic character of these ions.³



The cation *E*-**3** was generated from *E*-crotonic acid in  $H_2SO_4/SO_3$ , and the ¹³C NMR spectrum showed very low field absorption at C₁ and C₃, consistent with significant positive charge development at these positions (equation 2).⁴ Photolysis of *E*-**3** produced a spectrum attributed to *Z*-**3**.⁴ No isomerization of *E*-**3** to *Z*-**3** was observed after several hours at room temperature, and it was suggested that there was no ketene character **3a** as opposed to acylium ion character **3b** for this ion.⁴



Both maleic acid (4) and fumaric acid (5) reacted in  $H_2SO_4/SO_3$  to form *O*-protonated maleic anhydride 7 (equation 3).⁴ The isomerization was proposed to occur through an intermediate ketenyl cation **6**.⁴



The reactions of crotonyl chloride in DF-SbF₅ were followed by ¹H NMR, and the deuterium content of the product ester was determined after quenching with MeOH.⁵ It was concluded that vinylketene CH₂=CHCH=C=O was likely not to be an intermediate in the reaction, but the ketenic cation **8** was implicated (equation 4).⁵ Cation **8** was also formed from *c*-PrCOCl (**9**) in 1:1 HF-Sb₅ by protonation of the cyclopropane ring (equation 4).⁶



Acetyl chloride reacts with AlCl₃ in  $CD_2Cl_2$ , forming the cation **10** (equation 5), which was characterized by ¹H and ¹³C NMR and by IR.⁷



The solvolysis of naphthoyl chloride **11** has been proposed to occur through an acylium ion with a contribution of the ketenyl resonance structure **12** (equation 6).⁸ Distonic dehydrobenzoyl radical cation **13** in the gas phase was reported to have similar ketenic character.⁹



A computational study of the protonation of vinylketene **14** led to the prediction that attack at  $C_4$  leading to the carbocation **15** was strongly favored over protonation at  $C_2$  to form **16** (equation 7).¹⁰ This was confirmed by experiment for the reaction of **17**, which gave initial formation of the cation **18** leading to the acid **19** (equation 8).¹¹ The nonconjugated acid that would result from protonation at  $C_2$  was found to be more stable than **19**, so the latter did not form through equilibration.¹¹



It was proposed that the reaction of methylmalonic acid or its dimethyl ester in SbF₅-FSO₃F gave rise to the ketenyl acylium ion **20**, as evidenced by the ¹H NMR signal at  $\delta$  3.05 and an IR band at 2170 cm⁻¹ (equation 9).¹² Formation of the vinylogous analogue **22** from reaction of an azulene-1,3-dicarboxylic acid **21** was also proposed based on ¹H NMR evidence (equation 10).¹³



The reaction of acid 23 in  $FSO_3H/SO_2ClF$  gave rise to the acylium ion 24, whose rearrangement to 25 was observed by ¹H NMR (equation 11).¹⁴ Cation 25 was trapped with MeOH.



A theoretical study of the diazotization of guanine led to the proposal of a calculated structure 26 for the first ring-opened intermediate in the reaction (equation 12).¹⁵



Reaction of cyclobutenone **27** with bis(trimethylsilyl)acetylene catalyzed by Lewis acids was proposed to occur via formation of cation **28** with silyl migration and ring opening to cationic ketene **29** and cyclization leading to **30** (equations 13, 14).¹⁷ The reaction pathway was calculated with PM3 methods, and the ring closure step involved an in-plane p-orbital of the allenic moiety attacking the ketenyl group. Oxidation of cyclobutenones **31** with PhI(OAc)₂ was suggested to proceed with ring opening to the cationic ketenyl species **32**, which added methanol with ring closure (equation 15).¹⁸



The possible formation of radical **1b** or of the bent acyl radical **33** during photolysis of acrolein could occur by direct bond fission or by abstraction, and reaction with iodine atoms led to an estimate of the carbonyl C-H bond strength that was similar to that of acetaldehyde (equation 15a).^{18a,b} Based on this result and ESR data^{19,20} the structure **33** is favored.

Vinyl-substituted acyl radicals **34** and **35** were generated by hydrogen abstraction from aldehydes, but the ESR spectra of these species indicated that these were bent  $\sigma$  radicals, not ketenyl structures (equations 16, 17).^{19,20} Photolysis of hexa-1,5-dien-3-ones such as **36** was proposed to form the radical pair **37**, which recombined to form ketene **38**, but the acyl radical in **37** also presumably would have  $\sigma$  character (equation 18).^{21,22}





Photolysis of **39** was proposed to give cleavage to the radical pair **40**, and  $\pi$  character for the ketenyl radical in **40** was claimed, based on the expected conjugative ability of the phenyl groups (equation 19).²³ Recombination of the radical pair gave the ketene **41** observed by IR and captured with methanol, forming **42** (equation 19).²³



Reaction of selenyl esters 43 with *n*-Bu₃SnH-AIBN was proposed to give ketenyl radicals 44, which cyclized to cyclohexenones 45 (equation 20).²⁴ The ester 46 was suggested to give the cyclopropylacyl radical 47 in hot benzene which formed the aldehyde 49 and the dimer 51, while in MeOH the ester 50 was formed, although the timing of the dimerization and nucleophilic addition steps was not established (equations 21, 22).^{25,28a} In a further experiment the selenyl ester 52 gave the ketone 56 by a pathway proposed to involve the radical intermediates 53–55 (equations 23, 24).^{28b} An analogous reaction of an allenyl derivative gave a

cyclooctadienone.26



Precedent for the generation of vinyl acyl radicals was found in the photolysis of **57** in MeOH, which was proposed to form the diradicals **58** and **59**, leading to cyclization to the ketene **60**, which led to the observed product **61** (equations 25, 26).²⁷ The reactions of cyclopropylvinylacyl selenides with *n*-Bu₃SnH gave the corresponding acyl radicals, which underwent cyclopropyl ring opening and then radical cyclization of the ketenyl intermediates (equation 27).^{28,28a,b}




Reaction of the thioester **66** was proposed to give the ketenyl radical **67**, which reacted by an intramolecular 5-*exo* trig cyclization, forming the ketenyl radical **68**, which underwent a further intramolecular 5-*exo* trig cyclization leading to **69** and **70** (equations 28, 29).^{29,30} An analogous sequence was used in a formal synthesis of  $(\pm)$ -modhephene **71**.³¹ Tandem cyclization onto a cyclopentene was used in the preparation of pentalene.³⁰



The addition of CO to vinyl radicals gives acyl radicals,^{31a} and reaction of vinyl iodide **72** with *n*-Bu₃Sn• under pressure of CO gave vinyl radical **73**, which gave carbonylation to ketenyl radical **74**, which cyclized to thiolactone **75** (equation 30).³² As discussed in Section 3.4.1, oxidation of hydroxycyclobutenone **31** led to vinyl acyl/ketenyl radicals **76** (equation 31).³³





An unusual amine cyclization occurred with the ketenyl radicals **77** generated from  $\omega$ -amino terminal alkynes by free radical carbonylation (equation 32).³⁴ The reaction proceeded through addition of the radical *n*-Bu₃Sn• (G•) to the alkyne, forming a vinyl radical that combined with CO to form **77**. This cyclized, followed by partial destannylation, possibly with hydrogen atom transfer from an intermediate lactam enol, to give lactams **77a** (equation 33).



Ketene **78** with an oxyl radical side chain has been studied by computational means.^{34a} Density functional calculations indicated that the distonic ion **79** has a structure with most of the positive charge at the carbonyl carbon and the unpaired spin delocalized over the allyl group.³⁵ The reactivity of **79** was studied by Fourier transform ion cyclotron resonance.³⁵



Photolysis of fluoromaleic anhydride **80** in an argon matrix led to IR bands ascribed to fluoro(ketenyl)carbene **81** and fluorocyclopropenone **82** (equation 34), which underwent further slow photolysis to fluoroacetylene and CO.³⁶ The ketene **81** was assigned the *syn*-conformation shown, although this was calculated to be slightly higher in energy than the *anti*-conformation, because of the high calculated barrier for interconversion of the conformers and the low barrier for conversion of *syn*- **81** to **82**.³⁶ The calculations suggested that resonance structures such as **81a** gave rise to the high conformational barriers.³⁶ Photochemical and thermochemical decomposition of maleic and dichloromaleic anhyrides in matrices were also

suggested to form ketenylcarbenes and cyclopropenes.³⁷



Ketenyl carbanions are rare, but a possible example **83** was proposed to result from phenyllithium addition to diphenylcyclopropenone, which upon acidification gave a transient IR absorption at 2100 cm⁻¹ attributed to ketene **84**, and this added water to give **85** (equation 35).³⁸ Reaction of the vinyllithium compound **86** with CO was suggested to possibly involve a ketenyl carbanion **87** (equation 36).³⁹ Gas phase reaction of CO with the cyclopropenyl anion **88** was thought to involve the ketenyl carbanion **89** (equation 37).³⁹



Photolysis of bis(diazo) ketone **90** forms the  $\alpha$ -diazo ketene **91**, which on further photolysis gave ketenylcarbene **92** as the precursor to cyclopropenone **93** (equation 38).^{41,42} The photolysis and thermolysis of 1,2-bisketenes may also lead to ketenylcarbenes (Section 4.9). Computations of the effects of the substituents on the energies of ketenylcarbenes **94** and the corresponding cyclopropenones **95** formed by ring closure have been reported (equation 39).^{43,44}





## **REFERENCES FOR SECTION 4.1.9**

- 1. Olah, G. A.; Comisarow, M. B. J. Am. Chem. Soc. 1967, 89, 2694-2697.
- 2. Olah, G. A.; Comisarow, M. B. J. Am. Chem. Soc. 1968, 90, 4442-4447.
- 3. Olah, G. A.; Denis, J.-M.; Westerman, P. W. J. Org. Chem. 1974, 39, 1206-1210.
- Amat, A. M.; Asensio, G.; Castello, M. J.; Miranda, M. A.; Simon-Fuentes, A. *Tetra*hedron 1987, 43, 905–910.
- 5. Farcasiu, D.; Bologa, U. L. J. Org. Chem. 1996, 61, 8860-8865.
- Farcasiu, D.; Miller, G.; Bologu, U. L.; Katritzky, A. R.; Carpenter, B. K. J. Chem. Soc. Perkin Trans. 2 1998, 2659–2663.
- Csihony, S.; Bodor, A.; Rohonczy, J.; Horvath, I. J. Chem. Soc. Perkin Trans. 1 2002, 2861–2865.
- 8. Liu, K.-T.; Hwang, P. Y.-H.; Chen, H.-I. J. Phys. Org. Chem. 2002, 15, 750–757.
- 9. Moraes, L. A. B.; Eberlin, M. N. J. Am. Chem. Soc. 1998, 120, 11136-11143.
- Leung-Tuong, R.; Peterson, M. R.; Tidwell, T. T.; Czimadia, I. G. J. Mol. Struct. 1989, 183, 319–330.
- 11. Allen, A. D.; Stevenson, A.; Tidwell, T. T. J. Org. Chem. 1989, 54, 2843-2848.
- 12. Conrow, K.; Morris, D. L. J. Chem. Soc., Chem. Commun. 1973, 5-6.
- McDonald, R. N.; Morris, D. L.; Petty, H. E.; Hoskins, T. L. J. Chem. Soc. D 1971, 743–744.
- 14. Poulter, C. D.; Winstein, S. J. Am. Chem. Soc. 1972, 94, 2297-2305.
- 15. Glaser, R.; Son, M.-S. J. Am. Chem. Soc. 1996, 118, 10942-10943.
- 16. Rayat, S.; Glaser, R. J. Org. Chem. 2003, 68, 9882-9892.
- Yamamoto, Y.; Noda, M.; Ohno, M.; Eguchi, S. J. Org. Chem. 1997, 62, 1292– 1298.
- 18. Ohno, M.; Yamamoto, Y.; Eguchi, S. Synlett 1998, 1167-1174.
- 18a. Coomber, J. W.; Pitts, J. N., Jr. J. Am. Chem. Soc. 1969, 91, 547-550.
- 18b. Alfassi, Z. B.; Golden, D. M. J. Am. Chem. Soc. 1973, 95, 319-323.
- 19. Davies, A. G.; Sutcliffe, R. J. Chem. Soc. Perkin Trans. 2 1980, 819-824.
- Davies, A. G.; Hawari, J. A.-A.; Muggleton, B.; Tse, M.-W. J. Chem. Soc. Perkin Trans. 2 1981, 1132–1137.
- 21. Wolff, S.; Agosta, W. C. J. Am. Chem. Soc. 1983, 105, 1292-1299.
- Dauben, W. G.; Cogen, J. M.; Ganzer, G. A.; Behar, V. J. Am. Chem. Soc.. 1991, 113, 5817–5824.
- Adam, W.; Berkessel, A.; Peters, E.; Peters, K.; von Schnering, H. G. J. Org. Chem. 1985, 50, 2811–2814.
- 24. Ohno, M.; Oguri, I.; Eguchi, S. J. Org. Chem. 1999, 65, 8995-9000.
- 25. Hayes, C. J.; Pattenden, G. Tetrahedron Lett. 1996, 37, 271-274.

- 26. Harrington-Frost, N. M.; Pattenden, G. Synlett 1999, 1917-1918.
- 27. Erman, W. F.; Gibson, T. W. Tetrahedron 1969, 25, 2493-2515.
- 28. Herbert, N.; Pattenden, G. Synlett 1997, 69-70.
- 28a. Hayes, C. J.; Herbert, N. M. A.; Harrington-Frost, N. M.; Pattenden, G. Org. Biomol. Chem. 2005, 3, 316–327.
- 28b. De Boeck, B.; Herbert, N. M. A.; Harrington-Frost, N. M.; Pattenden, G. Org. Biomol. Chem. 2005, 3, 328–339.
- 29. De Boeck, B.; Pattenden, G. Tetrahedron Lett. 1998, 39, 6975-6978.
- 30. Harrington-Frost, N. M.; Pattenden, G. Tetrahedron Lett. 2000, 41, 403-405.
- 31. De Boeck, B.; Harrington-Frost, N. M.; Pattenden, G. Org. Biomol. Chem. 2005, 3, 340–347.
- 31a. Okuro, K.; Alper, H. J. Org. Chem. 1996, 61, 5312-5315.
- Ryu, I.; Okuda, T.; Nagahara, K.; Kambe, N.; Komatsu, M.; Sonoda, N. J. Org. Chem. 1997, 62, 7550–7551.
- 33. Yamamoto Y.; Ohno, M.; Eguchi, S. J. Am. Chem. Soc. 1995, 117, 9653-9661.
- Tojino, M.; Uenoyama, Y.; Fukuyama, T.; Ryu, I. Chem. Commun. 2004, 2482– 2483.
- 34a. Cooksy, A. L. J. Am. Chem. Soc. 2001, 123, 4003-4013.
- 35. Chou, P. K.; Thoen, K. K.; Kenttämaa, H. I. J. Org. Chem. 1998, 63, 4470-4474.
- 36. Dailey, W. P. J. Org. Chem. 1995, 60, 6737-6743.
- Ionescu, A.; Piétri, N.; Hillebrand, M.; Monnier, M.; Aycard, J. P. Can. J. Chem. 2002, 80, 455–461.
- 38. Ciabattoni, J.; Kocienski, P. J.; Melloni, G. Tetrahedron. Lett. 1969, 1883-1887.
- Ryu, I.; Hayama, Y.; Hirai, A.; Sonoda, N.; Orita, A.; Ohe, K.; Murai, S. J. Am. Chem. Soc. 1990, 112, 7061–7063.
- 40. Sachs, R. K.; Kass, S. R. J. Am. Chem. Soc. 1994, 116, 783-784.
- 41. Chapman, O. L.; Gano. J.; West, P. R.; Regitz, M.; Maas, G. J. Am. Chem. Soc. 1981, 103, 7033–7036.
- 42. Murata, S.; Yamamoto, T.; Tomioka, H.; Lee, H.-K.; Kim, H.-R.; Yabe, A. J. Chem. Soc., Chem. Commun. 1990, 1258–1260.
- Dogaru, D.; Pietri, N.; Aycard, J. P.; Hillebrand, M. J. Phys. Org. Chem. 2004, 17, 409– 417.
- 44. Sung, K.; Fang, D.-C.; Glenn, D.; Tidwell, T. T. J. Chem. Soc., Perkin Trans. 2 1998, 2073–2080.

# 4.1.10 Fulvenones and Cumulenones

Fulvenones are  $\alpha$ -oxo analogues of the well-studied fulvenes¹⁻³ and include triafulvenone (1), pentafulvenone (2), heptafulvenone (3), and nonafulvenone (4). Computational studies have evaluated the aromaticity of 1–4, and by the criteria of aromatic stabilization energy (ASE), bond alternation, nucleus independent chemical shifts (NICS), and magnetic susceptibility exaltation ( $\lambda$ ), 1 and 3 are antiaromatic, while 2 and 4 are aromatic.⁴



**4.1.10.1** Triafulvenones Triafulvenone **1** was predicted to be strongly antiaromatic^{1,4–6} and is unknown experimentally. Attempted preparation of the diazo ketone precursor **5** of the diphenyl analogue was unsuccessful and gave instead **6** (equation 1).⁷ The diazo ketone **5** was also expected to have  $4\pi$  electron antiaromatic character, which accounted for this result. Photolysis of **7** in the presence of 1-diethylaminopropyne gave a quantitative yield of N₂ and 30% of **9**, which may have formed from the ketene **8** (equation 2).⁷ A possible carbanionic derivative of **1** is noted in Section 4.1.9.⁸



**4.1.10.2 Pentafulvenones** Pentafulvenone (2) was prepared by Süs in 1944 by Wolff rearrangement of the diazo ketone **10** and found to undergo hydration to the carboxylic acid **12** in aqueous solution (equation 3).⁹ Kinetic investigations of this process led to the proposal that the reaction occurred with formation of intermediate enediol **11** (equation 3).¹⁰ Pentafulvenones **2**, **13**, and **14** were each generated in hydrocarbon solvents by photolysis of the corresponding diazo ketones at ambient temperatures, and their ketenyl IR absorptions were measured by conventional methods at 2129, 2117, and 2110 cm⁻¹, respectively.¹¹ Their reactions with the aminoxyl radical TEMPO were also examined (Section 5.7).¹¹





The yield of pentafulvenone 2 formed from photochemical Wolff rearrangement was diminished due to the formation of the dimeric product 15 formed by reaction of 2 with residual diazo ketone (equation 4).¹¹



Photoreactions of 2-chlorophenol, isomeric dichlorophenols, and 2,4,5-trichlorophenol in matrices formed substituted  $\alpha$ -ketocarbenes **16** that gave Wolff rearrangement to derivatives of **2** as detected by IR (equation 5).¹² Complexes of these pentafulvenones with HCL were also observed and their geometries derived from the vibrational shift.¹² The carbene **16** was also formed from photolysis of 2- bromophenol, and was observed by UV and rearrangement to **2**.¹³ Ketene **2** was also formed by carbonylation of the carbene formed from the diazoalkane **17** in a matrix containing CO (equation 5).¹⁴ Photolysis of chloro-*p*-hydroquinone in an argon matrix at 16 K gave formation of hydroxypentafulvenone **2a** as well as *p*-benzoquinone, proposed to form via the ketocarbene (equation 5a).^{14a} Deuterium labeling showed that *p*-benzoquinone formed by migration of the hydrogen indicated (equation 5a).



Theoretical studies of benzooxirene (18) and its benzo derivatives (19a-c) elucidated their conversion to ring-opened keto carbenes that rearranged to fulvenones

(equation 5).^{15,16}



Photolysis of tetrabromodiazocyclopentadiene in an argon matrix gave the carbene **20**, which in the presence of CO formed the ketene **21**, identified by its IR band at 2163 cm⁻¹ (equation 6).¹⁷ The pentafulvenone **23** was formed by photolysis of the diazo ketone **22** in an Ar matrix (equation 7) and identified by its IR band at  $2120 \text{ cm}^{-1}$ .¹⁸



Thermolysis of the iodonium zwitterions **24** in refluxing acetonitrile was reported to lead via a possible carbene intermediate to the unobserved diacylfulvenone **25**, which in undried solvent formed the dione **25a** in 55% yield, presumably by a hydration/decarboxylation route (equation 8).^{20,21} This reaction is also discussed in Section 4.1.6 with acylketenes.



FVP of **26** gave elimination reactions forming azafulvenone **27**, identified by its IR band at  $2148 \text{ cm}^{-1}$  in an Ar matrix at 12 K, and the formation of dimer **28** (equation 9).²² Pyrolysis of **26** at higher temperature gave reduced amounts of **27** and IR bands at 2142, 2139, and 2135.5 cm⁻¹, which were attributed to the cyanovinylketenes **30**, which were proposed to form by a retro-Wolff rearrangement to

the keto carbene **29** (equation 10).²³ The ketenes were trapped with methanol.²³



Low-temperature photochemical Wolff rearrangement of diazo ketone **31** in a matrix gave the azapentafulvenone **32** (equation 11).^{22,23} Diazo compound **33** on photolysis in a matrix formed carbene **34**, which underwent carbonylation, forming **35** (equation 12).²⁴



Hydration of fulvenone **13** to the acid **36** is the key reaction involved in its application in photolithography,^{25–28} and the mechanism of this reaction in solution has received much attention (equation 13).^{29–31} This transformation has also been achieved by two-photon absorption of light operating at 800 nm using an ultrafast pulse laser, which does not overlap with the absorption of the diazo ketone.³² Derivatives of **13** have been used in the growth of well-aligned carbon nanotubes.³³



Matrix photolysis of **37** gave the indenylketene **38**, with a characteristic ketenyl IR band at  $2142 \text{ cm}^{-1}$  (equation 14).³⁶ Derivatives of **38** are also important in

photoresists (Section 3.3.2).^{34–37}



Photolysis of **39** gave the diazo ketone **40**, which was converted to benzopentafulvenone **41** by Wolff rearrangement in a matrix, as identified by IR and trapping with methanol (equation 15).³⁸ Benzopentafulvenone **41** was also formed by pyrolysis at 750 °C of 3-acetoxy-2-naphthoic acid and trapped in a matrix at -196 °C, as shown by the IR absorption at 2127 cm⁻¹.³⁹



Wolff rearrangement of diazo ketone **42** gave azaindenylideneketene **43**, which formed the tetramer **44** (equation 16).^{40,41} Addition of CO to the carbene in the matrix also gave **43**.⁴¹



Photochemical Wolff rearrangement of 45 in a matrix gave 46, which was observed by IR and formed the dimer 47 (equation 17).⁴⁰ FVP of 48 with

elimination of methanol gave the ketene **49**, which was trapped in a matrix and observed by IR (equation 18).^{42,43}



Dibenzofulvenone (fluorenylideneketene, **14**), an analogue of diphenylketene, is an isolable material that was first prepared by zinc dechlorination of the  $\alpha$ -chloro acyl chloride **50** (equation 19).⁴⁴ Ketene **14** was also prepared by pyrolysis of the methyl trimethylsilyl ketene acetal **51** (equation 19),⁴⁵ and by Wolff rearrangement,^{46,47} and gave a ketenyl IR band at 2121 cm⁻¹ (CH₃CN)⁴⁶ and at 2114 and 2126 cm⁻¹ (hexane).¹¹



Preparation of 14 by dehydrochlorination of 9-fluorenylcarbonyl chloride (52) with triethylamine with in situ [2+2] cycloaddition with 1,3-cyclohexadiene gave 53 (equation 20).⁴⁸



Base-catalyzed hydrolysis of esters **54** of 9-fluorenylcarboxylic acid possessing aryl leaving groups with electron-withdrawing substituents was proposed to involve

ester enolate formation leading to ketene **14** (equation 21).^{49,50} The ketene **14** formed by photolysis of the diazo ketone showed high reactivity with amines and  $H_2O$  (Section 5.5.1).^{47,50}



Wolff rearrangement of **55** in methanol gave the unobserved azafluorenylideneketene **56**, which reacted by methanol addition and oxidation to radical **57**, which then formed the dimer (equation 22).⁵¹



The highly reactive fulvenone 2 formed zwitterionic adduct 58 with pyridine in a matrix between 15 and 50 K (equation 23).²² The fulvenones 13 and 14 behaved similarly.



Heptafulvenone **3** generated by dehydrochlorination of **59** gave in situ [2+2] cycloaddition with cyclopentadiene, forming **60** (equation 24).⁵²



Dehydrochlorination of **59** with Et₃N and 1,8-bis(dimethylamino)naphthalene (**61**) gave **3**, identified by its IR band at 2101 cm⁻¹ (equation 25).⁵³ The reactivity of **3** with the aminoxyl radical TEMPO was studied, and products from dimerization of an intermediate cycloheptatrienyl radical **62** were observed (equation 25).⁵³ Ketene **3** was also formed in a argon matrix by CO addition to the carbene and identified by the IR band at 2103 cm⁻¹.⁵⁴



Photolysis of the anhydride 63 gave the dehydronaphthalene, which upon photolysis added CO to give cumulenone 64 (equation 26).⁵⁵



# **REFERENCES FOR SECTION 4.1.10**

- Neuenschwander, N. In *The Chemistry of the Functional Groups. Supplement A. The Chemistry of Double-Bonded Functional Groups*, Vol. 2, Part 2; Patai, S.; Ed.; Wiley, New York, 1989, pp. 1131–1268.
- 2. Bergmann, E. D. Chem Revs., 1968, 68, 41-84.
- 3. Yates, P. Adv. Alicyclic Chem. 1968, 2, 59-184.
- 4. Najafian, K.; Schleyer, P. v. R.; Tidwell, T. T. Org. Biomol. Chem. 2003, 1, 3410-3417.
- 5. Trinquier, G.; Malrieu, J. J. Am. Chem. Soc. 1987, 109, 5303-5315.
- 6. McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1992, 114, 5362-5368.
- 7. Nickles, H.; Dürr, H.; Toda, F. Chem. Ber. 1986, 119, 2249-2260.
- 8. Sachs, R. K.; Kass, S. R. J. Am. Chem. Soc. 1994, 116, 783-784.
- 9. Süs, O. Liebigs Ann. Chem. 1944, 556, 65-84.
- 10. Urwyler, O. A.; Wirz, J. Angew. Chem. Int. Ed. Engl. 1990, 29, 790-791.
- 11. Allen, A. D.; Porter, J.; Tahmassebi, D.; Tidwell, T. T. J. Org. Chem. 2001, 66, 7420–7426.

- Akai, N.; Kudoh, S.; Takayanagi, M.; Nakata, M. J. Photochem. Photobiol. A Chem. 2001, 146, 49–57.
- 13. Bonnichon, F.; Richard, C.; Grabner, G. Chem. Commun. 2001, 73-74.
- Baird, M. S.; Dunkin, I. R.; Hacker, N.; Poliakoff, M.; Turner, J. J. J. Am. Chem. Soc. 1981, 103, 5190–5195.
- 14a. Akai, N, Kudoh, S.; Nakata, M. J. Photochem. Phtobiol. A Chem. 2005, 169, 47-55.
- 15. Lewars, E. Can J. Chem. 2000, 78, 297-306.
- Borisov, Y. A.; Garrett, B. C.; Feller, D.; *Izv. Akad. Nauk. Ser. Khim.* 1999, 1663–1667; *Russ. Chem. Bull.* 1999, 48, 1642–1646.
- 17. Dunkin, I. R.; McCluskey, A. Spectrochim. Acta 1993, 49A, 1179–1185.
- 18. Schweig, A.; Baumgartl, H.; Schulz, R. J. Mol. Struct. 1991, 247, 135-171.
- Sander, W.; Bucher, G.; Wandel, H.; Krafka, E.; Cremer, D.; Sheldrick, W. S. J. Am. Chem. Soc. 1997, 119, 10660–10672.
- 20. Papoutsis, I.; Spyroudis, S.; Varvoglis, A. Tetrahedron Lett. 1994 35, 8449-8452.
- 21. Malamidou-Xenikaki, E.; Spyroudis, S.; Tsanakopoulou, M. J. Org. Chem. 2003 68, 5627–5631.
- 22. Qiao, G. G.; Andraos, J.; Wentrup, C. J. Am. Chem. Soc. 1996, 118, 5634-5638.
- 23. Qiao, G. G.; Wong, M. W.; Wentrup, C. J. Org. Chem. 1996, 61, 8125-8131.
- 24. Maier, G.; Endres, J. Eur. J. Org. Chem. 2000, 2535-2539.
- 25. Reiser, A.; Shih, H.-Y.; Yeh, T. -F.; Huang, J.-P. Angew. Chem. Int. Ed. Engl. 1996, 35, 3428–3440.
- Reiser, A.; Huang, J. P.; He, X.; Yeh, T. F.; Jha, S.; Shih, H. Y.; Kim, M. S.; Han, Y. K.; Yan, K. *Eur. Polym. J.* **2002**, *38*, 619–629.
- 27. Han, Y. K.; Yan, K. Reiser, A. Macromolecules 1999, 32, 8421-8426.
- 28. Wallraff, G. M.; Hinsberg, W. D. Chem. Rev., 1999, 99, 1801-1821.
- Barra, M.; Fisher, T. A.; Cernigliaro, G. J.; Sinta, R.; Scaiano, J. C. J. Am. Chem. Soc., 1992, 114, 2630–2634.
- 30. Almstead, J.-I.; Urwyler, B.; Wirz, J. J. Am. Chem. Soc. 1994, 116, 954-960.
- 31. Andraos, J.; Kresge, A. J.; Popik, V. V. J. Am. Chem. Soc. 1994, 116, 961-967.
- 32. Urdabayev, N. K.; Popik, V. V. J. Am. Chem. Soc. 2004, 126, 4058-4059.
- 33. Yang, Y.; Huang, S.; He, H.; Mau, A. W. H.; Dai, L. J. Am. Chem. Soc. 1999, 122, 10832–10833.
- Nicolau, D. V.; Taguchi, T.; Taniguchi, H.; Yoshikawa, S. *Langmuir* 1998, 14, 1927– 1236.
- 35. Kappe, C. O.; Färber, G.; Wentrup, C.; Kappe, T. Chem. Ber. 1993, 126, 2357-2360.
- Moreau, W. M. Proc. SPIE-Int. Soc. Opt. Eng., 1997, 3049 (Advances in Resist Technology and Processing XIV), 618–627; Chem. Abs. 1997, 127, 255189.
- Leeson, M. J.; Pawloski, A.; Levering, V.; Yueh, W.; Willson, C. G. Proc. SPIE-Int. Soc. Opt. Eng. 1997, 3049, 861–870.
- 38. Blocher, A.; Zeller, K.-P. Chem. Ber. 1994, 127, 551-555.
- 39. Wentrup, C.; Gross, G. Angew. Chem. Int. Ed. Engl. 1983, 22, 543.
- Qiao, G. G.; Meutermans, W.; Wong, M. W.; Träubel, M.; Wentrup, C. J. Am. Chem. Soc. 1996, 118, 3852–3861.

- 41. Qiao, G. G.; Wentrup, C. Tetrahedron Lett. 1995, 36, 3913-3916.
- 42. Qiao, G. G.; Wong, M. W.; Wentrup, C. J. Org. Chem. 1996, 61, 8125-8131.
- 43. Gross, G.; Wentrup, C. J. Chem. Soc. Chem. Commun. 1982, 360-361.
- 44. Staudinger, H. Chem. Ber. 1906, 39, 3062-3067.
- 45. Ainsworth, C.; Chen. F.; Kuo, Y.-N. J. Organomet. Chem. 1972, 46, 59-71.
- Andraos, J.; Chiang, Y.; Kresge, A. J.; Popik, V. V. J. Am. Chem. Soc. 1997, 119, 8417– 8424.
- Camara de Lucas, N.; Netto-Ferreira, J. C.; Andraos, J.; Lusztyk, J.; Wagner, B. D.; Scaiano, J. C. *Tetrahedron Lett.* **1997**, *38*, 5147–5150.
- 48. Chung, Y.-S.; Kruk, H.; Barizo, O. M.; Katz, M.; Lee-Ruff, E. J. Org. Chem. **1987**, 52, 1284–1288.
- 49. Alborz, M.; Douglas, K. T. J. Chem. Soc., Perkin Trans. 2, 1982, 331-339.
- 50. Correia, V. R.; Cuccovia, I. M.; Stelmo, M.; Chaimovich, H. J. Am. Chem. Soc. 1992, 114, 2144–2146.
- Braven, J.; Hanson, R. W.; Smith, N. G. J. Heterocyclic Chem. 1995, 32, 1051– 1055.
- 52. Asao, T.; Morita, N.; Kitahara, Y. J. Am. Chem. Soc. 1972, 94, 3655-3657.
- 53. Tidwell, T. T.; Fenwick, M. H. Eur. J. Org. Chem. 2001, 3415-3419.
- 54. McMahon, R. J.; Chapman, O. L. J. Am. Chem. Soc. 1986, 108, 1713-1714.
- 55. Sato, T.; Niino, H.; Yabe, A. J. Phys. Chem. A 2001, 105, 7790-7798.

## 4.1.11 Oxoquinone Methides, Oxoxylylenes, and Related Species

Ketene analogues of quinone methides are oxoquinone methides 1 and 2, while the oxoxylylenes 3 and 4 are analogues of xylylenes. These species tend to be highly reactive, and are of particular interest because of the opportunities for reactions involving interactions of the conjugated functional groups.



Oxoquinone methide **1** was first observed from photolysis of the benzodioxinone **5** in a matrix at 77 K, and identified by its IR absorption at 2118 cm⁻¹ (equation 1),¹ and **1** was also formed by photolysis of phthalyl peroxide **6** or the furan-2,3-dione **7** (equation 1).² Photo interconversion of **1** and benzopropiolactone **8** was observed, with **8** favored by light of wavelength >340 nm and **1** favored by light of wavelength 315 nm (equation 2).³ Reaction of **1** with methanol formed methyl salicylate

(9).³ The chemistry of **1** has been reviewed.⁴



Photolysis of 1-benzofuran-2,3-dione (7) in an argon matrix at 12 K resulted in the loss of CO and formation of the  $\alpha$ -oxoketene **1**, which was identified by its IR bands at 1605 and 2089–2105 cm⁻¹ (equation 3).⁵ Initial loss of CO₂ from **7** to form the carbene **10**, leading to the cumulenone **11**, was also observed (equation 3).⁵ Photolysis of **5** in CH₃CN solution gave **1**, with a strong band at 2135 cm⁻¹ observed by TRIR spectroscopy (equation 4).⁶ Rate constants were measured for reaction of **1** with H₂O, MeOH, and Et₂NH at 22 °C of  $1.5 \times 10^7$ ,  $3.0 \times 10^7$ , and  $1.1 \times 10^9$  M⁻¹ s⁻¹, respectively.⁶ The pseudopericyclic transition state **12** for the reaction of **1** with H₂O leading to salicylic acid **13** was calculated at the MP2/6-31G* level, but no transition structure was located at the MP2/6-31G* level.⁶



Evidence for the 1,4-oxoquinonemethide **2** came from the kinetics of basecatalyzed hydrolysis of 2,4-dinitrophenyl 4-hydroxybenzoate (**14**), which reacted 340 times faster than expected for the  $B_{Ac}2$  mechanism, indicating the intervention of the E1cB pathway forming **2**, which was trapped by aniline, forming **15** (equation 5).⁷



Photolysis of diazo ketone **16** in an Ar matrix doped with 1% CO at 10 K gave the carbene **17**, and when this was warmed to 35 K, addition of CO gave the ketene **2**, as identified by its IR band at 2110 cm⁻¹ (equation 6).⁸ Similarly, photolysis of the tetrafluoro analogue **18** gave the carbene **19**, and carbonylation in the matrix formed the ketene **20** with IR absorption at 2172 cm⁻¹ (equation 7).⁹ The tetrachloro analogue **20a** was formed similarly, with an IR absorption at 2132 cm⁻¹.⁹



Photolysis of benzocyclobutenone **21** in methanol or isopropanol gave the oxo-1, 2-xylylene **3**, which was trapped as the ester **22** (equation 8).^{10,11} When generated by the photo reaction in an argon matrix at 20 K, **3** was identified by its IR absorption at 2120 cm⁻¹.^{12,13} Generation of **3** by flash photolysis of **21** in water permitted measurement of the kinetics of hydration forming 2-toluic acid.¹⁴ The rate of reaction of substituted derivative **3a** with nucleophiles were slowed due to hindrance to in-plane attack on the ketenyl group (equation 8a).^{14a}





Heating of neat benzocyclobutenone **21** with alcohols to form **3** gave esters **22**, as in equation 8, while heating of neat **21** gave the dimer **23** resulting from [4 + 2] cycloaddition (equation 9).¹⁵ Thermolysis of toluyl chloride **24** at 780 °C formed **3**, leading to **21** (equation 10).¹⁶



Reaction of 2-trimethylsilylmethylbenzoyl chloride (25) with anhydrous CsF gave 3, which was trapped with benzaldehyde by a [4 + 2] cycloaddition, forming 26 (equation 11).^{17–19} Substituted derivatives of 3 were formed similarly and gave [2 + 2] cycloadditions with alkenes.^{17–19} The reaction of 27 with Mo(CO)₆ formed 28, which gave benzocyclobutenones 29 (equation 12).²⁰



Photolysis of phthalaldehyde **30** formed **33** by a process proposed to involve reversible hydrogen atom transfer that formed **31**, which underwent bond rotation to **32**, which gave **33** (equation 13).^{21,22} The structures **32** and **33** may differ in both

electronic multiplicity and geometry.^{21,22}



Irradiation of 2-vinylbenzaldehyde (**34**) gave the ketene methide **35**, detected by UV absorption at 380 nm and by IR bands at 2086 and 2098 cm⁻¹ (equation 14).²³ In the presence of amines **35** gave amides **36**, and reacted slowly with methanol but did not react with dienophiles; this was attributed to rapid reversion of **35** back to **34**. Ionization of **34** by X-irradiation in argon matrices gave the radical cation of **35**.²³ Both **35** and its radical cation were also obtained from the corresponding methylbenzocyclobutenone.^{24,25}



Photolysis of **37** under high-intensity laser jet conditions gave the unobserved fulvenone **38**, which either cyclized to **39** (equation 15), underwent photochemical decarbonylation, or, in the presence of MeOH, was trapped as the ester **40** (equation 16).²⁶ Because **37** did not form **38** upon photolysis at 350 nm but gave other products instead, it was proposed that the conversion of **37** to **38** was not symmetry allowed from the first excited state but occurred from a higher state.²⁶ Hydrogen transfer in *o*-vinylbenzaldehyde giving an analogue of **38** was shown to be reversible, and upon prolonged photolysis an unusual dimerization of the aldehyde eventually occurred.²⁷ Photolysis of aryl ketones and esters with ortho methyl groups led to analogous hydrogen transfers and formation of ketene derivatives.^{28,29}



Photolysis of **41** under laser jet conditions was proposed to lead to the fulvenone **42**, which underwent photochemical decarbonylation to **43** and formation of **44a** and **44b** (equations 17, 18).³⁰ Under ordinary photolysis conditions no net reaction of **41** was observed, and this was attributed to rapid cyclization of **42** back to **41**.³⁰



Photolysis of **45a** formed isomeric ketene enols E/Z-**45b**, each detected by UV. The Z-isomer rapidly decays, and with H₂O was proposed to form **45c** (equation 18a).³¹



3-Oxomethylenene-6-methylene-1, 4-cyclohexadiene (4) was the subject of a computational study,³³ and was prepared by photolysis of the [2.2]cyclophane **46** in an Ar matrix at 10 K and identified by the IR band at 2108 cm⁻¹ (equation 19).³⁴ Photolysis of the unsaturated [2.2]cyclophane **47** gave **48**, which was trapped with EtOH, forming the ester **49** (equation 20).^{34,35}





Many heteroatom analogues of the oxoquinonemethides have been studied. Thus, flash photolysis of 2-nitrobenzaldehyde (**50**) in benzene gave a product with a strong UV absorption near 450 nm assigned as the ketene **51**, which formed 2-nitrosobenzoic acid either unimolecularly or with catalysis by H₂O (equation 21).³⁶ Irradiation of **50** with X-rays in an argon matrix also gave **51**, as identified by the IR absorption at 2108 and 2115 cm⁻¹.³⁷



Photolysis of (2-nitrophenyl)diazomethane in an argon matrix at 10 K formed 2nitrosobenzaldehyde (**52**), which upon further irradiation was interpreted as proceeding with formation of the ketene **53**, which rearranged to **54**, as assigned by the IR spectrum (equation 22).³⁸ Isomerization to *E*-**53** (IR 2114 cm⁻¹) was also observed.³⁸ Photolysis in an argon matrix of the azide **55** formed the nitrene **56**, leading to benzazetinone **58** and a geometrical mixture of ketenes **57**, identified by the IR absorption at 2088 and 2118 cm⁻¹ (equation 23).³⁹



Thermolysis of 1,2,3-benzotriazin-4(1H)-one **59** with trapping in an argon matrix gave benzazetinone **60**, which upon photolysis gave the imino ketene **61** in a reversible process, as identified by the IR absorption at 2125 cm⁻¹ (equation 24).⁴⁰ Formation of the NH derivative of **61** by pyrolytic routes has also been proposed, with formation of a dimer in 50% yield.^{41,42} Photolysis of **62** in *i*-PrOH was proposed to form **63** and the ketene **64**, which was observed by UV and captured, forming the ester **65** (equation 25).⁴³



Elimination induced by  $Et_3N$  from **66** was proposed to lead to the unobserved imino ketenes **67**, which cyclized to **68** (equation 26).^{44,45} It was proposed that photolysis of **68** gave **67** and that these were in thermal equilibrium, as indicated by the capture of **67** with alcohols or amines.⁴⁴



The mass spectrum of the esters 69 gave the ketene radical cation 70 as the base peak (equation 27).⁴⁶



Lithiation of **71** and carbonylation was proposed to form the ketene **72**, which cyclized to **73** (equation 28).⁴⁷



Pyrolysis of **74** at 575 °C was proposed to lead to the azafulvenone **75**, which at  $10^{-4}$  torr formed **76** (equation 29),⁴⁸ but at a higher pressure of  $10^{-2}$  torr and higher contact time only decarbonylation products were isolated. Photolysis of **76** in MeOH did not reform **75** but instead gave **77** quantitatively, resulting from MeOH addition at the bridgehead carbon adjacent to the nitrogen.⁴⁸



FVP or matrix photolysis of **78** gave the ketene **79**, identified by its IR bands at 2158, 2151, and 2147 cm⁻¹ (equation 30).⁴⁹ This ketene also dimerized and was captured by MeOH.⁴⁹



Nitrosative deamination of guanine (**80**) was reported to give the cationic ketene **82**, which was found by molecular orbital calculations to favor the resonance structure shown (equation 31; see also Section 4.1.9).^{50,51} Studies using ¹⁸O- labeled water for the nitrosative deamination of **80** showed the intermediacy of the diazonium ion **83**, which formed ketene **84** (equation 32), which upon hydration gave **85**, which cyclized to 2'-deoxyoxanosine **86** as the predominant reaction product (equation 33).^{52–54}





Photolysis of diazirene **88** gave the carbene **89** which abstracted hydrogen, forming the ketenes *syn/anti*-**90**, IR 2093, 2105 cm⁻¹ (equation 34).⁵⁵ Deuteration of the aldehydic hydrogen showed dramatically slowed conversion to the ketene in the matrix, a result attributed to quantum mechanical tunneling.⁵⁵



## **REFERENCES FOR SECTION 4.1.11**

- 1. Chapman, O. L.; McIntosh, C. L. J. Am. Chem. Soc. 1970, 92, 7001-7002.
- 2. Dvorák, V.; Kolc, J.; Michl, J. Tetrahedron Lett. 1972, 3443-3446.
- 3. Chapman, O. L.; McIntosh, C. L.; Pacansky, J.; Calder, G. V.; Orr, G. J. Am. Chem. Soc. **1973**, *93*, 4061–4062.
- 4. Wentrup, C.; Heilmeyer, W.; Kollenz, G. Synthesis 1994, 1219-1248.
- 5. Aycard, J.-P.; Volanschi, E.; Hnach, M.; Zineddine, H.; N'Guessan, T. Y. J. Chem. Res. (S) **1995**, 346.
- Liu, R. C.-Y.; Lusztyk, J.; McAllister, M. A.; Tidwell, T. T.; Wagner, B. D. J. Am. Chem. Soc. 1998, 120, 6247–6251.
- 7. Cevasco, G.; Guanti, G.; Hopkins, A. R.; Thea, S.; Williams, A. J. Org. Chem. 1985, 50, 479–484.
- 8. Sander, W.; Bucher, G.; Reichel, F.; Cremer, D. J. Am. Chem. Soc. 1991, 113, 5311-5322.
- 9. Sander, W.; Hübert, R.; Kraka, E.; Gräfenstein, J.; Cremer, D. Chem. Eur. J. 2000, 6, 4567–4579.
- 10. Cava, M. P.; Spangler, R. J. J. Am. Chem. Soc. 1967, 89, 4550-4551.

- Arnold, D. R.; Hedeya, E.; Merritt, V. Y.; Karnischky, L. A.; Kent, M. E. *Tetrahedron Lett.* 1972, 3917–3920.
- 12. Krantz, A. J. Am. Chem. Soc. 1974, 96, 4992-4993.
- 13. Hacker, N. P.; Turro, N. J. J. Photochem. 1983, 22, 131-135.
- 14. Chiang, Y.; Kresge, A. J.; Zhan, H.-Q. Can. J. Chem. 2003, 81, 607-611.
- 14a. Schiess, P.; Eberle, M.; Huys-Francotte, M.; Wirz, J. *Tetrahedron Lett.* 1984, 25, 2201– 2204.
- 15. Wang, Z. Y.; Suzzarini, L.; Gao, J. P. Tetrahedron Lett. 1997, 33, 5745-5746.
- 16. Suzzarini, L.; Lin, J.; Wang, Z. Y. Tetrahedron Lett. 1998, 39, 1695-1696.
- Kessar, S. V.; Singh, P.; Vohra, R.; Kaur, N. P.; Venugopal, D. J. Org. Chem. 1992, 57, 6716–6720.
- 18. Kessar, S. V.; Singh, P.; Venugopal, D. J. Chem. Soc., Chem. Commun. 1985, 1258–1259.
- 19. Aono, M.; Terao, Y.; Achiwa, K. Chem. Lett. 1985, 339-340.
- Kang, J.; Choi, Y. R.; Kim, B. J.; Jeong, J. U.; Lee, S.; Lee, J. H.; Pyun, C. *Tetrahedron Lett.* 1990, 31, 2713–2716.
- 21. Giebicki, J.; Kuberski, S.; Kaminski, R. J. Chem. Soc., Perkin Trans. 2 1990, 765–769.
- Tomioka, H.; Kobayashi, N.; Murata, S.; Ohtawa, Y. J. Am. Chem. Soc. 1991, 113, 8771–8778.
- 23. Kessar, S. V.; Mankotia, A. K. S.; Scaiano, J. C.; Barra, M.; Gebicki, J.; Huben, K. J. Am. Chem. Soc. **1996**, *118*, 4361–4365.
- 24. Huben, K.; Zhu, Z.; Bally, T.; Gebicki, J. J. Am. Chem. Soc. 1997, 119, 2825-2831.
- 25. Gebicki, J.; Bally, T. Acc. Chem. Res. 1997, 30, 477-485.
- Wilson, R. M.; Patterson, W. S.; Austen, S. C.; Ho, D. M.; Bauer, J. A. K. J. Am. Chem. Soc. 1995, 117, 7820–7821.
- 27. Kessar, S. V.; Mankotia, A. K. S. J. Chem. Soc. Chem. Commun. 1993, 1828-1829.
- Yoshioka, M.; Nishizawa, K.; Suzuki, J.; Iwata, Y.; Kumakura, S.; Hasegawa, T. J. Chem. Soc. Perkin Trans. 1 1995, 3097–3100.
- 29. Julliard, M.; Pfau, M. J. Chem. Soc., Chem. Commun. 1976, 184.
- Wilson, R. M.; Romanova, T. N.; Azadnia, A.; Bauer, J. A. K.; Johnson, R. P. *Tetrahe*dron Lett. **1994**, 35, 5401–5404.
- 31. Netto-Ferreira, J.; Scaiano, J. C. Can. J. Chem. 1993, 71, 1209-1215.
- 32. Kuuya, M.; Miyaki, F.; Okuda, T. J. Chem. Soc. Perkin Trans. 2 1984, 1471-1477.
- 33. Sander, W.; Hübert, R.; Kraka, E.; Gräfenstein, J.; Cremer, D. Chem. Eur. J. 2000, 6, 4567–4579.
- 34. Marquardt, R.; Sander, W.; Laue, T. Hopf, H. Liebigs Ann. 1995, 1643-1648.
- 35. Sander, W.; Marquardt, R.; Bucher, G.; Wandel, H. Pure Appl. Chem. 1996, 68, 353–356.
- 36. George, M. V.; Scaiano, J. C. J. Phys. Chem. 1980, 84, 492-495.
- 37. Michlak, J.; Gebicki, J.; Bally, T. J. Chem. Soc., Perkin Trans. 2 1993, 1321-1325.
- 38. Tomioka, H.; Ichikawa, N.; Komatsu, K. J. Am. Chem. Soc. 1992, 114, 8045-8053.
- 39. Tomioka, H.; Ichikawa, N.; Komatsu, K. J. Am. Chem. Soc. 1993, 115, 8621-8626.
- Dunkin, I. R.; Lynch, M. A.; Withnall, R.; Boulton, A. J.; Henderson, N. J. Chem. Soc., Chem. Commun. 1989, 1777–1778.
- 41. Chiu, S.-J.; Chou, C.-H. Tetrahedron Lett. 1999, 40, 9271-9272.

- 42. Hong, C.-M.; Tsai, F.-C.; Shiea, J. Anal. Chem. 2000, 72, 1175-1178.
- 43. Kraft, B. J.; Zaleski, J. M. New J. Chem. 2001, 25, 1281-1289.
- 44. Olofson, R. A.; Vander Meer, R. K.; Hoskin, D. H.; Bernheim, M. Y.; Stournas, S.; Morrison, D. S. J. Org. Chem. 1984, 49, 3367–3372.
- 45. Vander Meer, R. K.; Olofson, R. A. J. Org. Chem. 1984, 49, 3373-3379.
- 46. Black, M.; Cadogan, J. I. G.; McNab, H. J. Chem. Soc., Perkin Trans. 1 1994, 155–159.
- 47. Iwamoto, K.; Chatani, N.; Murai, S. J. Org. Chem. 2000, 65, 7944-7948.
- 48. Chou, C.-H.; Wu, C.-C.; Chen, W.-K. Tetrahedron Lett. 1995, 36, 5065-5068.
- 49. Kappe, C. O.; Färber, G.; Wentrup, C.; Kappe, T. Chem. Ber. 1993, 126, 2357-2360.
- 50. Glaser, R.; Rayat, S.; Lewis, M.; Son, M.-S.; Meyer, S. J. Am. Chem. Soc. 1999, 121, 6108–6119.
- 51. Rayat, R.; Glaser, R. J. Org. Chem. 2003, 68, 9882-9892.
- 52. Rayat, R.; Majumdar, P.; Tipton, P.; Glaser, R. J. Am. Chem. Soc. 2004, 9960-9969.
- 53. Qian, M.; Glaser, R. J. Am. Chem. Soc. 2004, 126, 2274-2275.
- 54. Qian, M.; Glaser, R. J. Am. Chem. Soc. 2005, 127, 880-887.
- 55. Nakane, N.; Enyo, T.; Tomioka, H. J. Org. Chem. 2004, 69, 3538-3545.

#### 4.2 NITROGEN-SUBSTITUTED KETENES

A diverse array of ketene substituents with nitrogen bonding directly to  $C_2$  of the ketene have been studied, including amino, azido, nitroso, nitro, and azo, but substituted derivatives of the NH₂ group are the most common. Nitrogen is strongly electronegative; consequently, nitrogen-substituted ketenes are destabilized relative to alkyl-substituted analogues, as shown by the isodesmic energy comparison for aminoketene (1) of equation 1.^{1,2} The effect of substituents on the stability of ketenes is correlated with the substituent group electronegativity, which accounts for the instability of aminoketene.^{1,2}

$$\begin{array}{cccc} H_2N & CH_3 & \Delta E = & CH_3 & H_2N \\ \hline C = O & + & & \\ 1 & 2 & & \\ \end{array} \begin{array}{c} & & CH_2 & \\ \hline (-7.2 \text{ kcal/mol}) & & \\ \end{array} \begin{array}{c} CH_3 & CH_2 & \\ \hline C = O & + & \\ \end{array} \begin{array}{c} H_2N \\ \hline CH_2 & CH_2 & \\ \end{array} \begin{array}{c} & (1) \\ \end{array}$$

Aminoketene was calculated to have the structure **1a**, with the lone pair of electrons in the ketene plane facing in the direction of the carbonyl carbon. This has been interpreted as indicating destabilization of conformations with the nitrogen lone pair perpendicular to the ketenyl plane interacting with the C=C pi bond due to  $n-\pi$  electron repulsion.^{1,2} An alternative interpretation, that there is a favorable interaction between the nitrogen lone pair and the carbonyl carbon, would suggest a shortening of the distance between the two atoms and a decrease in the N-C-C bond angle, which in the extreme would lead to the bridged structure **1b**. The HF/6-31G* calculated N-C-O bond angle of 118.6 deg was modestly less than the C-C-C bond angle for methylketene (**3**), suggesting that there may be some electrostatic attraction between nitrogen and the carbonyl carbon but not

significant bridging.^{1,2}



Benzoylated aminoketenes **5** have been suggested to be in equilibrium with oxazolium-5-oleate "münchnone" tautomers  $6^3$ , which are the structures usually detected spectroscopically (equation 2).⁴



Phthalimido-*tert*-butylketene (8) was formed by the dehydrochlorination of acyl chloride 7 and is a persistent ketene obtained as a crystalline solid, mp 96–98 °C, IR 2134 cm⁻¹ (equation 3).⁵



Diazoketene (9) was obtained by distillation of the product from diazoacetyl chloride (8) with DABCO (1,4-diazabicyclooctane) in ether, and was identified by the product obtained from reaction with EtOH followed by HBr (equation 4).⁶



Nitroketenes are rare, despite the fact that this strongly conjugating group should stabilize ketenes, as was found in the computed structure^{1,2} and as occurs for cyano and acyl groups. Pyrolysis of 4-nitropyrrolidione **12** was proposed to form the nitroketene **13**, which led to the observed indole **14** (equation 5).⁷ Possible evidence for the formation of nitroketene **15** was claimed by heating methyl or ethyl nitroacetate with  $P_2O_5$  and trapping the gas in  $H_2O$  to give the presumed  $O_2NCH_2CO_2H$  in 8–15% yield, as determined by titration (equation 6).⁸ No spectroscopic evidence for



the formation of 15 was reported, and further study of this reaction is needed.

Nitrosoketene **17** has been generated by the thermolysis of nitroso Meldrum's acid (**16**) in toluene, and trapping with carbonyl compounds yielded nitrones **19**, which were proposed to arise from initial [4+2] cycloadducts **18** (equation 7).^{9,10} The chiral [3 + 2] cycloadduct of **17** with *l*-menthone was used for the synthesis of the naturally occurring nonproteinogenic amino acid (2S,1'S)-(cyclopent-2-enyl)-glycine.¹⁰ Observation of **17** in the gas phase showed the ketene band at 2146 cm⁻¹ and a C=C band at 1314 cm⁻¹, in agreement with the calculated spectrum.¹¹ *Ab initio* molecular orbital studies of the cycloaddition of nitrosoketene (**17**) with carbonyl compounds favored a direct [3 + 2] reaction to form the nitrone **19** rather than an initial [4 + 2] formation of **18** followed by rearrangement to **19**.^{12,13}



Reaction of phenyl(trimethylstannyl)acetylene with  $N_2O_4$  results in addition of NO⁺ and formation of nitroso(phenyl)ketene (**19a**) which is converted to benzoyl cyanide (equation 7a).^{13a}



Photolysis of "syndnones" **20** formed *N*-nitoso-*N*-(3-aryl)aminoketenes **21** (equation 8).¹⁴ For **20** (Ar=Ph, R=alkenyl), photolysis in frozen solutions resulted in the formation of new absorption bands at 630 and 625 nm, respectively,

attributed to the formation of ketenes **21** for which INDO calculated structures were reported.¹⁵



Dehydrochlorination of azidoacetyl chloride (22) gave azidoketene (23),^{16–18} which reacted in situ with the chiral imine 24, forming the diastereomeric *cis*  $\beta$ -lactams 25 in a 51:49 ratio (equation 9).¹⁷



Azoketenes **27** were formed by dehydrochlorination of acyl chlorides **26** and trapped in [2+2] cycloadditions with imines, forming  $\beta$ -lactams **28** (equation 10), and by additions of alcohols and amines.^{19,20} Dehydrochlorination of **29** was proposed to form isocyanatoketene (**30**), leading to cross-linked and oligomeric products (equation 11).²¹



Isocyanoketene CNCH=C= $O^{22}$  (**30a**) and the iminoketene (MeS)₂C= NCH=C= $O^{23}$  (**30b**) were proposed as intermediates formed by dehydrochlorination of the corresponding acyl chlorides, as evidenced by their capture by [2 + 2] cycloaddition with Ph₂C=NPh as the  $\beta$ -lactams.



Nitrogen-substituted ketenes have found wide application, particularly in stereoselective [2 + 2] cycloadditions. For example, dehydrochlorination of **31** gave the chiral ketene **32**,^{24,25} which reacted in situ with imines, giving  $\beta$ -lactams **33** with high diastereoselectivity (equation 12).²⁵



Phthalimidoketene **35** was generated by dehydrochlorination of the *N*-phthalyl acyl chloride **34** with Et₃N at -78 °C and utilized in stereoselective [2+2] cycloaddition with imines,²⁶ and addition of *R*-pantolactone (ROH) gave the ester **36** of phenylglycine in 98% *de* (equation 13).²⁷ Similar reactions are considered in Section 5.9. *N*-Tetrachlorophthalimidoketene generated from the acyl chloride with Et₃N also reacted with imines to give  $\beta$ -lactams.²⁸



The reaction of the acid chloride of *N*-phthalyl valine (**37**) with various tertiary amines and *R*-pantolactone gave the unobserved valine-derived ketene **38**, which reacted with further amine to give the zwitterion **39**, which gave the ester **40** in 96% yield and 85% *de* with quinuclidine as the base (equation 14).²⁹



The reaction of chiral acyl chlorides **41** with Et₃N to give aminoketenes **42** followed by [2 + 2] cycloaddition with imines gave  $\beta$ -lactams **43** with high diastereoselectivities (equation 15).³⁰ Ketene equivalents **45** generated photochemically from chromium carbene complexes **44** reacted with alcohols to give esters **46** with high *de* (equation 16).³¹ Reaction of the proline derivative **47** with Et₃N led through imine [2 + 2] cycloaddition of ketene **48** to the spiro  $\gamma$ -lactam **49** (equation 17), which was used to prepare peptidominetics.³² The stereoselectivity of the reactions of the chiral derivative **50** is discussed in Section 5.4.1.7.³³



Ketene **51** formed by dehydration reacted with a dendrimer-bound imine, forming dendrimer-bound  $\beta$ -lactam **52** (equation 18).³⁴





Heating eneamino acids **53** with acetic anhydride containing  $Et_3N$  giving dihydroisoindoles **56** was proposed to occur through ketenes **54** tautomeric with 1,3-oxazolium-5-olates (münchnones) **55** (equations 19, 20).³⁵ The reaction of acyl chlorides with imines and of CO with palladium catalysis provided a four-component synthesis of münchnones **57** (equation 21).³⁶



As described in Section 2.5, the formation of aminoketene fragments from electrospray mass spectrometric peptide cleavage was proposed as a method for protein analysis.³⁷

## **REFERENCES FOR SECTION 4.2**

- 1. Gong, L.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1991, 113, 6021-6028.
- 2. McAllister, M. A.; Tidwell, T. T. J. Org. Chem. 1994, 59, 4506-4515.
- 3. Boyd, G. V. In *Comprhensive Heterocyclic Chemistry*, Vol. 6; Potts, K. T., Ed.; Pergamon, New York, **1984**, pp. 177–233.

#### 312 TYPES OF KETENES

- 4. Bayer, H. O.; Huisgen, R.; Knorr, R.; Schaefer, F. C. Chem. Ber. 1970, 103, 2581-2597.
- 5. Winter, S.; Pracejus, H. Chem. Ber. 1966, 99, 151-159.
- 6. Bestmann, H. J.; Soliman, F. M. Angew. Chem. Int. Ed. Engl. 1979, 18, 947-948.
- 7. Kappe, C. O.; Kollenz, G.; Wentrup, C. J. Chem. Soc. Chem. Commun. 1992, 485-486.
- Efremov, D. A.; Zavlin, P. M.; Essentseva, N. S.; Tebby, J. C. J. Chem. Soc., Perkin Trans. 1 1994, 3163–3168.
- Katagiri, N.; Sato, H.; Kurimoto, A.; Okada, M.; Yamada, A.; Kaneko, C. J. Org. Chem. 1994, 59, 8101–8106.
- 10. Katagiri, N.; Okada, M.; Morishita, Y.; Kaneko, C. Chem. Commun. 1996, 2137-2138.
- 11. Matsui, H.; Zückerman, E. J.; Katagiri, N.; Kaneko, C.; Ham, S.; Birney, D. M. J. Phys. Chem. A **1997**, 101, 3936–3941.
- 12. Ham, S.; Birney, D. M. Tetrahedron Lett. 1994, 35, 8113-8116.
- 13. Ham, S.; Birney, D. M. Tetrahedron Lett. 1997, 38, 5925-5928.
- 13a. Woltermann, C. J.; Shechter, H. Helv. Chim. Acta 2005, 88, 354–369.
- 14. Nespurek, S.; Böhm, S.; Kuthan, J. J. Mol. Struct. 1986, 136, 261-273.
- 15. Nespurek, S.; Obrda, J.; Lipinski, J. J. Photochem. Photobiol. A: Chem. 1993, 75, 49-59.
- Bose, A. K.; Anjaneyulu, B.; Bhattacharya, S. K.; Manhas, M. S. *Tetrahedron* 1967, 23, 4769–4776.
- Ojima, I.; Nakahashi, K.; Brandstadter, S. M.; Hatanaka, N. J. Am. Chem. Soc. 1987, 109, 1798–1805.
- Sullivan, D. F.; Scopes, D. I. C.; Kluge, A. F.; Edwards, J. A. J. Org. Chem. 1976, 41, 1112–1117.
- Bodnar, V. N.; Lozindskii, M. O. Ukr. Khim. Zh. 1983, 49, 301–303. Chem. Abs. 1983, 98, 197909f.
- Lozindskii, M. O.: Bodnar, V. N.; Pel'kis, P. S. Zh. Org. Khim 1980, 16, 228-229. Chem. Abs. 1980, 93, 46125t.
- 21. Mormann, W.; Hoffmann, S.; Hoffmann, W. Chem. Ber. 1987, 120, 285-290.
- 22. Hoppe, I.; Schöllkopf, U. Chem. Ber. 1976, 109, 482-487.
- 23. Hoppe, D.; Raude, E. Liebigs Ann. Chem. 1979, 2076-2088.
- 24. Evans, D. A.; Williams, J. M. Tetrahedron Lett. 1988, 29, 5065-5068.
- 25. Evans, D. A.; Sjogren, E. B. Tetrahedron Lett. 1985, 26, 3783-3786, 3787-3790.
- Palomo, C.; Cossio, F. P.; Cuevas, C.; Lecea, B.; Mielgo, A.; Roman, P.; Luque, A.; Martinez-Ripoll, M. J. Am. Chem. Soc. 1992, 114, 9360–9369.
- 27. Calmes, M.; Daunis, J.; Mai, N.; Natt, F. Tetrahedron Lett. 1996, 37, 379-380.
- Bose, A. K.; Jayaraman, M.; Okawa, A.; Bari, S. S.; Robb, E. W.; Manhas, M. S. *Tetra*hedron Lett. **1996**, 37, 6989–6992.
- 29. Calmes, M.; Daunis, J.; Mai, N. Tetrahedron 1997, 53, 13719–13726.
- Palomo, C.; Aizpurua, J. M.; Legido, M.; Mielgo, A.; Galarza, R. Chem. Eur. J. 1997, 3, 1432–1441.
- 31. Zhu, J.; Deur, C.; Hegedus, L. S. J. Org. Chem. 1997, 62, 7704-7710.
- Alonso, G.; López,-Ortiz, F.; del Pozo, C.; Peralta, E.; Macías, A.; González, J. J. Org. Chem. 2001, 66, 6333–6338.
- Macías, A.; Alonso, A.; del Pozo, C.; Venturini, A.; González, J. J. Org. Chem. 2004, 69, 7004–7012.

- 34. Díez-Barra, E.; García-Martinez, J. C.; Rodríquez-López, J. Synlett 2003, 1587-1590.
- 35. Gabbutt, C. D.; Hepworth, J. D.; Heron, B. M.; Pugh, S. L. J. Chem. Soc. Perkin Trans. *1* 2002, 2799–2808.
- 36. Dhawan, R.; Dghaym, R. D.; Arndtsen, B. A. J. Am. Chem. Soc. 2003, 125, 1474-1475.
- Masterson, D. S.; Yin, H.; Chacon, A.; Hachey, D. L.; Norris, J. L.; Porter, N. A. J. Am. Chem. Soc. 2004, 126, 720–721.

## 4.3 OXYGEN-SUBSTITUTED KETENES

Ketene **1** with the electronegative OH substituent was calculated by the isodesmic comparison of equation 1 at the MP2/6-31G*//MP2/6-31G* level to be destabilized by 14.5 kcal/mol relative to methylketene (**3**).^{1,2} Ketenes are stabilized by electropositive substituents that can donate electrons by  $\sigma$ - $\pi$  conjugation in the ketene plane to the carbonyl group, and are destabilized by  $\pi$  donors, and oxygen is disfavored by both mechanisms. The substituents OH, OCH₃, and O₂CCH₃ have calculated electronegativities of 3.55, 3.53, and 3.57,³ respectively, and so by this criterion are all predicted to be essentially equally destabilizing, although the lower  $\pi$ -donating ability of the acetoxy group may make acetoxyketene more stable. Ketenes with these substituents are quite reactive, and are typically generated and used in situ for further transformations.

Photolysis of formaldehyde in a CO matrix was reported to cause a hydrogen shift in formaldehyde leading to hydroxy carbene, which underwent carbonylation forming **1**, as detected by its IR absorption at 2148 cm⁻¹ (equation 2).⁴ Methoxy-ketene (**5**) was formed by pyrolysis of dimethyl malonate above 600 °C by decarbonylation of the intermediate carbomethoxyketene **6**, with trapping in an argon matrix at 12 K, or by photolysis of methyl diazoacetate in the matrix, and was identified by its IR absorption at 2136 cm⁻¹ (equation 3),⁵ while ethoxyketene (**7**) had an absorption at 2120 cm⁻¹.⁶

$$CH_2=0 \xrightarrow{hv} :CHOH \xrightarrow{CO} HO = C=0$$

$$1 (IR 2148 \text{ cm}^{-1})$$
(2)

$$\frac{\text{MeO}_2\text{C}}{\text{OMe}} \xrightarrow{\text{OO}_2\text{C}} \frac{\text{MeO}_2\text{C}}{\text{-MeOH}} \xrightarrow{\text{C}=\text{O}} \frac{\text{MeO}_2\text{C}}{\text{C}=\text{O}} \xrightarrow{\text{-CO}} \xrightarrow{\text{MeO}} C=\text{O} \xrightarrow{\text{hv}} \underbrace{\text{O}}_{\text{MeO}} \xrightarrow{\text{O}}_{\text{MeO}} N_2$$

$$6 \text{ (IR 2132, 2143 cm^{-1})} 5 \text{ (IR 2136 cm^{-1})}$$

(3)

Gas phase pyrolysis of dimethyl diazomalonate was proposed to involve formation of the ketene **8** (equation 4), which underwent decarbonylation leading to dimethoxyketene **9** which also lost CO, leading to methyl acetate (equation 5).⁷ Pyrolysis of diphenoxymalonic acid gave diphenoxyketene **10**, which reacted by [2 + 2] cycloaddition with Ph₂C=NPh, forming β-lactam **11** (equation 6).⁸



Photolysis of ethyl phenylglyoxylate **12** formed phenylhydroxyketene (**14**) to the extent of 80% by intramolecular cleavage via the diradical **13** (equation 7).⁹ The ketene **14** was not observed but was proposed to undergo further photolysis with decarbonylation to the carbene **15**, which rearranged to the observed product benzaldehyde (equation 8).^{9,10} This ketene was observed by TRIR spectroscopy at 2100 cm⁻¹,¹¹ and a band at 1651 cm⁻¹ was attributed to the triplet state of **12**.¹¹ Further investigations used step-scan FTIR for study of the reaction, with observation of the ketene band at 2105 cm⁻¹.¹²



Photolysis of the ester **16** substituted with a cyclopropyl clock gave products resulting from cyclopropylcarbinyl radical ring opening, as well as benzaldehyde resulting from the formation of ketene **14** in a 44:56 ratio, and led to the estimate

of a lifetime of 16 ns for the diradical 17 (equation 9).¹³



Photolysis of *n*-butyl and isobutyl phenylglyoxalates PhCOCO₂R (**18**) in aqueous solution gave **14** for kinetic studies of the hydration reaction, forming mandelic acid in a reaction proposed to proceed through formation of the acid enol **19** in water-, base-, and acid-induced reactions, which were interpreted as first forming the acid enol **19** (equation 10).¹⁴⁻¹⁶ The rate constants for the water and base reactions at 25 °C  $k(H_2O) = 5.51 \times 10^5 \text{ s}^{-1}$ ,  $k(OH^-) = 2.33 \times 10^9 \text{ M}^{-1} \text{ s}^{-1}$ , and  $k(H^+) =$  $5.45 \times 10^7 \text{ M}^{-1} \text{ s}^{-1}$  are large for this highly reactive ketene compared to some model compounds (Table 4.2; data from Table 1, Section 5.5.1.2).



Computational studies of the hydration of hydroxyketene (1) favored a pathway for protonation on  $C_2$  forming the cation **20**, stabilized by the acylium resonance structure **21** (equation 11).¹⁶ Addition of water dimer was favored to form the enediol **22** (equation 11),¹⁶ consistent with the experimental results for **14** (equation 10).



TABLE 4.2 Comparative Rate Data for Hydration of Ketenes PhCR=C=O

Ketene	$\begin{array}{c} k(\mathrm{H_2O}) \mathrm{\ s}^{-1} \\ (k_{\mathrm{rel}}) \end{array}$	$k(OH^{-}) M^{-1} s^{-1}$ $(k_{rel})$	$k(H^+) M^{-1} s^{-1} (k_{rel})$
PhCH=C=O	$4.77 \times 10^{3}$	$1.22 \times 10^{6}$	Not detected
PhCMe=C=O	55.7	$2.03 \times 10^4$	Not detected
PhC(OH)=C=O	$5.51 \times 10^{5}$ 116	$2.33 \times 10^9$ 1900	$5.45  imes 10^7$
Photolysis of the pyruvyl amide 23 in methanol was interpreted to proceed with formation of the zwitterionic intermediate 24, which underwent cleavage to the hydroxyketene 14, which was trapped by methanol, as well as undergoing cyclization to the oxazolidin-4-one 25 and the  $\beta$ -lactam 26 (equation 12).^{17,18} The formation of oxazolidin-4-ones was favored for substrates with small alkyl substituents.^{17,18}



Photolysis of **27** and **29** in CHCl₃ containing ethanol gave the ketenes **28** and **30**, respectively, which were captured as the ethyl esters (equations 13, 14).¹⁹ Another example of a bicyclo[2.2.2]octadiene cleavage giving an oxygen-substituted ketene is shown in Section 4.1.3.²⁰



Acetoxyketene **32**, generated in situ by dehydrochlorination of **31**, reacted with imines **33**, giving  $\beta$ -lactams **34** used to produce the taxotere side chain (equation 15),²¹ and also gave [2 + 2] cycloaddition with cyclopentadiene.²² Acetoxy(phenyl)ketene **35** was also prepared by dehydrochlorination, and gave [2 + 2] cycloaddition with the carbonyl group of biacetyl, forming **36** (equation 16).²³





Reaction of optically active acyl chloride **37** with the amino alcohol tropanol led to a racemic ester **39** by the intermediacy of ketene **38** (equation 17).²⁴ The use of tropanol hydrochloride prevented the racemization.



Dehydrochlorination of benzyloxyacetyl chloride (**40**) with Et₃N gave benzyloxyketene **41**, which was not observed but was trapped in a [2 + 2] cycloaddition with imine **42**, forming  $\beta$ -lactam **43** (equation 18).^{25,25a} Reaction of **41** generated in this way in CH₂Cl₂ with the chiral amine *t*-BuCH=NCH(CH₃)Ph also gave the *cis*-lactam, but with no diastereoselectivity.²⁶ Methanesulfonyloxyketene MsOCH= C=O (**44**) has also been generated by the dehydrochlorination route and trapped with imines, forming **45** (equation 19).²⁷ Triphosgene was used to generate ketene **46** containing a chiral auxiliary for synthesis of an equal mixture of the diastereomeric *cis*- $\beta$ -lactams **47**, which were readily separated into the optically pure forms by chromatography (equation 20).²⁸



Deposition of CO on La₂O₃ gave a UV band at 300 nm and IR bands at 2075 and 1367 cm⁻¹ ascribed to the anion **48** (equation 21).²⁹ Reaction of the potassium salt **49** with methyl iodide formed the ketene **50**, which gave the dimer **51** (equation 22).³⁰ Thermolysis of the alkynyl diether **52** yielded ketene **53**, which reacted with more of the alkyne **52** by [2+2] cycloaddition, forming the cyclobutenone **54** (equation 23).³⁰



Wolff rearrangement of diazo ketone **55** gave the ketene **56**, detected by UV spectroscopy, and this interconverted on photolysis with the cumulenone **57** (equation 24).³¹ The major product from acid-catalyzed hydration was **57a**, proposed to form by protonation at  $C_2$  of the cumuleneone.³¹



Photochemical generation of oxygen-substituted ketenes by Wolff rearrangement has generally been inefficient, as oxygen is poor in migration and keto carbenes are formed that undergo extensive capture by solvent before oxygen migrates. Thus, photolysis of **58** gave ketene-derived products in 33% yield in solution (equation 25).³² However, FVT in which the keto carbene was formed in the gas phase in the absence of solvent permitted enhanced Wolff rearrangement to ketene **59** and gave intramolecular [2 + 2] cyclization to **60** in 70% yield and

>99% de (equation 25).³²⁻³⁴



Computations gave the energetics for the reaction of acetylene with ozone, forming **61** and hydroperoxyketene **62** (equation 26).³⁵ Formation of the ketenyloxyl radical **63** has been considered.³⁶

$$H \longrightarrow H \xrightarrow{O_3} O \xrightarrow{O O} O \longrightarrow H \xrightarrow{HOO} H \xrightarrow{C=O} H \xrightarrow{O} C=O \xrightarrow{O} H \xrightarrow{C=O} (26)$$

Rhodium-catalyzed cyclization of the diazo ketone **64** gave cyclization to the transient bicyclo[3.3.0]octan-3-one-1-oxonium-2-ylide **65**, which underwent sigmatropic and stereospecific [3 + 2] cycloreversion to the 4-pentenyloxyketene **66**, which was trapped with MeOH in 88% yield (equation 27).³⁷ The effects of different ring sizes and substituents were examined, and the mechanism was analyzed using DFT computations.³⁷



The ethoxy(triphenylsilyl)ketene **68** was prepared from metal carbene complexes **67** and trapped in situ by ethyl vinyl ether by [2 + 2] cycloaddition (equation 28).³⁸ Trimethylsilyloxy(methyl)ketene **70** was formed by photochemical rearrangement of silyldiketone **69** in a Vaseline mull and detected by the IR absorption at 2106–2108 cm⁻¹ (equation 29).³⁹ Another silyloxy-substituted ketene is noted in Section 4.1.3.⁴⁰



Hydroxyl-substituted bisketenes are covered in Section 4.9.

## **REFERENCES FOR SECTION 4.3**

- 1. Gong, L.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1991, 113, 6021-6028.
- 2. McAllister, M. A.; Tidwell, T. T. J. Org. Chem. 1994, 59, 4506-4515.
- 3. Boyd, R. J.; Boyd, S. L. J. Am. Chem. Soc. 1992, 114, 1652-1655.
- 4. Lee, E. K. C.; Sodeau, J. R.; Diem, M.; Shibuyu, K. Proc. Yamada Conf. Free Radicals, 3rd, 1979, 98–112. Chem. Abstr. 1980, 93, 140857e.
- 5. Leung-Toung, R.; Wentrup, C. Tetrahedron 1992, 48, 7641-7654.
- 6. Krantz, A. J. Chem. Soc. Chem. Commun. 1973, 670-671.
- 7. Richardson, D. C.; Hendrick, M. E.; Jones, M., Jr. J. Am. Chem. Soc. 1971, 93, 3790-3791.
- 8. Staudinger, H.; Schneider, H. Helv. Chim. Acta 1923, 6, 304-315.
- 9. Hu, S.; Neckers, D. C. J. Org. Chem. 1996, 61, 6407-6415.
- 10. Hu, S.; Neckers, D. C. J. Org. Chem. 1997, 62, 7827-7831.
- Federov, A. V.; Danilov, E. O.; Rodgers, M. A. J.; Neckers, D. C. J. Am. Chem. Soc. 2001, 123, 5136–5137.
- Federov, A. V.; Danilov, E. O.; Merzlikine, A. G.; Rodgers, M. A. J.; Neckers, D. C. J. Phys. Chem. A. 2003, 107, 3208–3214.
- 13. Hu, S.; Neckers, D. C. J. Org. Chem. 1997, 62, 755-757.
- Chiang, Y.; Kresge, A. J.; Popik, V. V.; Schepp, N. P. J. Am. Chem. Soc. 1997, 119, 10203–10212.
- 15. Chiang, Y.; Kresge, A. J.; Pruszynski, P.; Wirz, J. Angew. Chem. Int. Ed. Engl. 1990, 792–794.
- 16. Andraos, J.; Kresge, A. J. J. Mol. Struct. 1991, 233, 165-184.
- Aoyama, H.; Sakamoto, M.; Kuwabara, K.; Yoshida, K.; Omote, Y. J. Am. Chem. Soc. 1983, 105, 1958–1964.
- Aoyama, H.; Hasegawa, T.; Watabe, M.; Shiraishi, H.; Omote, Y. J. Org Chem. 1978, 43, 419–422.
- 19. Becker, H.-D.; Ruge, B. Angew. Chem. Int. Engl. Ed. 1975, 14, 761-762.
- Fernández-Zertuche, M.; López-Cortina, S.; Meza-Aviña, M. E.; Ordóñez; M.; Ramírez-Solís, A. Arkivoc 2003 (xi) 89–99.
- 21. Lucatelli, C.; Viton, F.; Gimbert, Y.; Greene, A. E. J. Org. Chem. 2002, 67, 9468-9470.
- 22. Russell, G. A.; Schmitt, K. D.; Mattox, J. J. Am. Chem. Soc. 1975, 97, 1882-1891.
- 23. Domínquez, D.; Cava, M. P. Tetrahedron Lett. 1982, 23, 5513-5516.
- Ferorelli, S.; Loiodice, F.; Longo, A.; Molfetta, A.; Tortorella, V.; Amoroso, R. Chirality 2000, 12, 697–704.
- 25. Abouabdellah, A.; Bégue, J.-P.; Bonnet-Delpon, D. Synlett 1996, 399-400.
- 25a. Kuznetsova, L.; Ungureanu, I. M.; Pepe, A.; Zanardi, I.; Wu, X.; Ojima, I. *J. Fluorine Chem.* **2004**, *125*, 487–500.
  - Palomo, C.; Ganboa, I.; Odriozola, B.; Linden, A. *Tetrahedron Lett.* 1997, 38, 3093– 3096.
  - 27. Storz, T.; Bernet, B.; Vasella, A. Helv. Chim. Acta 1999, 82, 2380-2412.
  - Shinkre, B. A.; Puranik, V. G.; Bhawal, B. M.; Beshmukh, A. R. A. S. *Tetrahedron Asymm.* 2003, 14, 453–459.

- 29. Bailes, M.; Bordiga, S.; Stone, F. S.; Zecchina, A. J. Chem. Soc. Faraday Trans. 1996, 92, 4675–4682.
- 30. Serratosa, F. Acc. Chem. Res. 1983, 16, 170-176.
- Chiang, Y.; Gaplovsky, M.; Kresge, A. J.; Leung, K. H.; Ley, C.; Mac, M.; Persy, G.; Phillips, D. L.; Popik, V. V.; Rödig, C.; Wirz, J.; Zhu, Y. J. Am. Chem. Soc. 2003, 125, 12872–12880.
- 32. Sugimura, T.; Tei, T.; Mori, A.; Okuyama, T.; Tai, A. J. Am. Chem. Soc. 2000, 122, 2128–2129.
- 33. Sugimura, T.; Tei, T.; Okuyama, T. Tetrahedron Lett. 2003, 44, 3115-3117.
- 34. Sugimura, T. Eur. J. Org. Chem. 2004, 1185-1192.
- 35. Cremer, D.; Crehuet, R.; Anglada, J. J. Am. Chem. Soc. 2001, 123, 6127-6141.
- 36. Cooksy, A. L. J. Am. Chem. Soc. 2001, 123, 4003-4013.
- Oku, A.; Sawada, Y.; Schroeder, M.; Higashikubo, I.; Yoshida, T.; Ohki, S. J. Org. Chem. 2004, 69, 1331–1336.
- 38. Kron, J.; Schubert, U. J. Organomet. Chem. 1989, 373, 203-219.
- 39. Wright, B. B. J. Am. Chem. Soc. 1988, 110, 4456-4457.
- 40. Pollart, D. J.; Moore, H. W. J. Org. Chem. 1989, 54, 5444-5448.

## 4.4 HALOGEN-SUBSTITUTED KETENES

#### 4.4.1 Fluoroketenes and Fluoroalkylketenes

Fluoroketene (1) was calculated at the MP2/6-31G* level to be destabilized by 16.9 kcal/mol relative to methylketene by the isodesmic comparison of equation 1, and by this criterion is the least stable ketene of the group studied (Section 1.1.3).^{1–3} The effect of substituents on the stability of ketenes was correlated with the substituent group electronegativity, which accounts for the instability of fluoroketene.^{1,2}

$$F = C = O + CH_3 \xrightarrow{\Delta E =} CH_3 \xrightarrow{CH_3} C = O + F = CH_2 (1)$$

$$1 \qquad 2 \qquad 3 \qquad 4$$

The ketenyl IR bands of CFH=C=O  $(1, 2142 \text{ cm}^{-1})^4$  and CF₂=C=O  $(5, 2162 \text{ cm}^{-1})$  have been reported.^{5,6} Ketenyl IR frequencies measured in a single laboratory using laser pyrolysis elimination from acyl chlorides for CHHal=C=O C=O were 2145, 2151, 2148, and 2145 cm⁻¹ for Hal=F, Cl, Br, and I (1, 6, 7, 8, respectively) in matricies and 2148 and 2157 cm⁻¹ for Hal=F and Cl, respectively, in the gas phase (equation 2).⁷

Hal 
$$O$$
  $hv$  Hal  $C=O$  (2)  
5 Hal = F(1), Cl(6), Br (7), I (8)

Fluoroketenes are remarkably reactive, and CFH=C=O (1) and CF₂=C=O (5) have not been directly observed in solution. The latter also rapidly underwent spontaneous decarbonylation, forming CF₂, but as described below, there is evidence that this species does have a short existence, and it has been generated, observed in a low-temperature matrix, and trapped. Recent computations indicated that this decarbonylation was exothermic by 2.3 kcal/mol with a barrier of 9.9 kcal/mol to transition structure **5a**, forming the CF₂•CO complex **5b**, which was 1.2 kcal/mol more stable than separated CF₂ and CO (equation 3).⁸ Structural and thermodynamic data for **1** and **5** have also been obtained by computation.⁹

$$\begin{array}{c} F \\ F \\ F \\ 5 \end{array} \begin{array}{c} C = O \\ 9.9 \\ 5 \end{array} \begin{array}{c} \Delta E^{*} = \\ P \\ Sa \end{array} \end{array} \begin{array}{c} F \\ C - C \\ Sa \end{array} \end{array} \begin{array}{c} T \\ \Delta E^{*} = \\ -12.4 \\ kcal/mol \end{array} \begin{array}{c} F \\ F \\ C - C \\ Sb \end{array} \begin{array}{c} -CO \\ \Delta E = 1.2 \\ kcal/mol \end{array} CF_{2}$$
(3)

The reaction of hydrogen and fluorine atoms generated in a microwave discharge of  $H_2/CF_4$  with ICH₂COCl formed activated FCH₂COCl, which gave CHF=C=O by loss of HCl (equation 4).¹⁰



Photolysis of difluoroacetylene with 193 nm light in an Ar matrix at 10 K led to difluorovinylidene 9, and in the presence of CO this formed the nonlinear cumuleneketene 10 on warming to 35–42 K (equation 5).¹¹ Similar photolysis forming 9 at 7 K in a CO₂-doped Ar matrix that was annealed at 30 K gave difluoroketene (5), as identified by the IR bands at 2162, 1427, and 1274 cm⁻¹ (equation 6).^{5,6} Formation of 5 from 9 was proposed to involve electrophilic abstraction of an oxygen atom without formation of the lactone 11. Irradiation of CF₂=C=O with light >550 nm led to decarbonylation, but the resultant CF₂ was not observed to react with CO.



The B3LYP calculated C–C and C–O bond lengths of  $CF_2=C=O$  of 1.313 and 1.168 Å, respectively, are similar to those of  $CH_2=C=O$ , while the C–F bond length is 1.333 Å, and the F–C–F angle is 114.2°.^{5,6}

The mass spectrum of  $CF_2=CFOCF_3$  also indicated the formation of  $CF_2=C=O$ , which was estimated to dissociate with a barrier of 10.0 kcal/mol (42 kJ/mol) in a process that was exothermic by 6.0 kcal/mol (25 kJ/mol) (equation 7).¹²

$$\underset{F}{\overset{F}{\longrightarrow}} \underset{F}{\overset{OCH_{3}^{+^{*}}}{\longrightarrow}} \underset{F}{\overset{F}{\longrightarrow}} \underset{F}{\overset{C=0}{\longrightarrow}} \underset{F}{\overset{-co}{\longrightarrow}} \underset{F}{\overset{F}{\overset{C:}}}$$
(7)

Fluoroketene generated by dehydrochlorination of fluoroacetyl chloride underwent [2+2] cycloaddition with cyclopentadiene, with a 10:1 preference for the *endo* fluorine in the product **12** (equation 8).¹³

$$F \qquad \bigcirc O \qquad \xrightarrow{Et_{1}N} F \qquad \bigcirc C = O \qquad \xrightarrow{F} F \qquad \bigcirc O \qquad \xrightarrow{F} F \qquad (8)$$

$$1 \qquad 12 (32\% exo/endo 1:10)$$

Substituted fluoroketenes FCR=C=O, R = Me, Ph, and CF₃ (**13a–c**) have been prepared by dehydrohalogenation of the acyl chlorides and, while not observed directly, were trapped by cyclopentadiene by [2 + 2] cycloadditions, forming cyclobutanones **14** (equation 9).¹⁴ The products were formed with a preference for fluorine only in the *exo* position.¹⁴



Reaction of CBrF₂COCl in acetone with activated zinc, forming difluoroketene, took place with in situ trapping by [2+2] cycloaddition and formation of 2,2-difluoro-3,3-dimethylpropanolactone (**15**) in 50% yield (equation 10).¹⁵ Difluoroketene was also trapped by trimethylsilyl vinyl ethers, forming **16** and **17** (equations 11, 12).^{16,17}

$$Br \xrightarrow{F} Cl \xrightarrow{Zn} F F C=0 \xrightarrow{(CH_3)_2C=0} CH_3 \xrightarrow{F} O CH_3 \xrightarrow{CH_3} O CH_3 O$$



Reaction of the iridium complex **18a** with *tert*-butyllithium from -196 to -80 °C gave the complex **18b**, observed by NMR spectroscopy, which upon warming to 22 °C formed the diffuoroketene complex **19**, with ¹³C NMR signals at  $\delta$  207.2 (CO) and 178.0 (CF₂), and the ketenyl IR signal at 1725 cm⁻¹ (equation 12a).^{17a} The structure of **19** was confirmed by X-ray (Figure 4.2).^{17a}



Acylfluoroketenes 21 were generated by photolysis at room temperature or thermolysis in refluxing aromatic solvents of fluorodioxinones 20 and captured as the oxazinones 22 by reaction with cyanamide (equation 13).¹⁸ Photolysis of 20 (R=H)



**Figure 4.2** X-ray structure of diffuoroketene complex **19** (reprinted from ref. 17a with the permission of Wiley VCH).

in *t*-BuOH generated **21a**, which gave the ester **23**, isolated as the semicarbazone, in 57% yield.¹⁸



FVP of the hexafluorocyclohexadienone **24** together with some of the 2,5-dienone isomer was proposed to involve initial formation of the perfluoroketene **25**, which decarbonylated, forming the observed product perfluorocyclopentadiene **26**, trapped at  $-78 \,^{\circ}$ C in 53% yield and 95% purity (equation 14).¹⁹ The photochemical ring opening of cyclohexadienones is a well-known route to dienylketenes (Barton-Quinkert reaction; Section 3.4.3).



Reaction of the vinyl phosphate 27 with fluoride ion generated the unobserved fluoroketene 28, which was captured by EtOH, giving the ester 29 (equation 15).²⁰



Fluoroalkylketenes also have unique properties, and are often considered together with ketenes with fluorine directly bonded to the ketenyl carbon. The reaction of trifluoroacetic anhydride with diazomethane gave a simple preparation of the diazo ketone **30**, which was characterized by ¹H, ¹³C, ¹⁵N, ¹⁷O, and ¹⁹F NMR spectroscopy (equation 16).²¹ The *Z* conformation of **30** was calculated to be the most stable by 3.2 kcal/mol, and the EI mass spectrum obtained with minimal heating showed no formation of the ketene **31**, whereas with greater heating of **30** the mass spectrum of **31** was observed (equation 16).²¹ The diazo ketone **30** has also been made from the acyl chloride with diazomethane and used to generate the

34

ketene **31** in H₂O or alcohol for isolation of the acid or esters **32**.^{22–24} The ketene **31** was detected by UV when formed by flash photolysis for measurement of the reactivity with water, and was found to be six times more reactive than *n*-BuCH=C=O (Section 5.5.1.2).²⁵ A study of the Wolff rearrangement of ¹³C-labeled **30** showed no scrambling of the label, indicating that an oxirene intermediate was not formed.^{25a}



The structures and conformations of **31** and of CHF₂CH=C=O (**33**) have been calculated by *ab initio* methods.²⁶

Dehydration of the acid **34** by heating with  $P_2O_5$  gave **35**, bp 5 °C, with a characteristic IR absorption at 2203 cm⁻¹ (equation 17).^{27,28} The perfluoroalkylaldoketene **36** was prepared similarly.²⁹ The ketene **35** was also formed by photochemical Wolff rearrangement of the diazo ketone **37** (equation 18).³⁰ The diazine **37a** also formed in a reversible reaction upon photolysis of **37** (equation 18).³⁰

35 (94%)



36



A new preparation of  $(CF_3)_2C=C=O$  (**35**) beginning with hexafluoroacetone has been reported, which makes this species generally available (equation 19).³¹ Previous preparations relied upon toxic starting materials that are not convenient. Photolysis of hexafluoro-2-butyne in a matrix containing O₃ led to formation of **35** by reaction of oxygen atoms forming an observable keto carbene, followed by a Wolff-type rearrangement (equation 20).³² When CO was present in the matrix, addition to the keto carbene forming the acyl ketene **35a** was observed by IR (equation 20a).³²



The cycloaddition chemistry of bis(trifluoromethyl)ketene has been extensively investigated. Some of the reactions observed include [2 + 2] cycloaddition with norbornene forming **38** (equation 21),³³ norbornadiene forming **39** (equation 22),³⁴ and alkynes forming **40** (equation 23).³⁵ Pyrolysis of **40** proceeded with ring opening, forming the unobserved **41**, which reacted further to give products derived by fluoride migration with cyclization.³⁶



Bis(trifluoromethyl)ketene (**35**) undergoes [2 + 2 + 2] cyclization with quadricyclane (**41a**), forming **41b** in 92% yield under mild conditions (equation 23a).³⁷



Reaction of **35** with ethyl vinyl ether in hexane at 0 °C gave the oxetane **42**, as detected by NMR. This isomerized at 50 °C to the cyclobutanone **43** and the enone **44**, isolated in 67% and 13% yields, respectively, when the cycloaddition was performed at 100 °C (equation 24).³⁸ Reaction in CH₂Cl₂ at -80 °C permitted isolation of crystalline **42** (mp -43 °C), which at -20 °C gave a deep blue intermediate assigned as the zwitterion **45** (equation 25). Reaction of **42** in acetone gave the adduct **46** from **45** (equation 25).³⁸



Bis(trifluoromethyl)ketene **35** underwent ionization in SbF₅, forming the delocalized cation **47**, which was directly observed by NMR (equation 26).³⁹ The addition of fluoride ion to **35** gave the stable crystalline salt **48** (equation 27), which in the ¹⁹F NMR showed nonequivalent CF₃ groups even at 75 °C, indicating a barrier to rotation of at least 19 kcal/mol in the enolate.⁴⁰

$$\begin{array}{cccc} CF_3 & C=0 & \xrightarrow{SbF_5} & CF_3 & C=0 & \xleftarrow{CF_3} & C=0 \\ CF_3 & & & \downarrow CF_2 & C=0 & \longleftarrow & CF_2 & C=0 \\ 35 & & & 47 \end{array}$$
(26)



Phenyl(trifluoromethyl)ketene (49) was prepared by dehalogenation, and the addition of chiral alcohols gave stereoselectivity in the formation of the new chiral center in the ester 50 (equation 28).⁴¹



Hydrolysis of the perfluoroalkene **51** with aqueous NaOH solution formed the ketene **52**, which displayed characteristic IR absorption at  $2160 \text{ cm}^{-1}$  (equation 29).⁴²

$$\begin{array}{cccc} CF_{3}(C_{2}F_{5})_{2}C & & \stackrel{10\% \text{ NaOH}}{\text{diglyme}} & CF_{3}(C_{2}F_{5})_{2}C \\ C_{2}F_{5}(CF_{3})_{2}C & & \stackrel{80-90\ ^{\circ}C}{&} & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & \\ & & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & & \\ & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & \\ & & & & \\ & & & & \\ & & & &$$

Trifluoromethylmalonate esters **53** react with  $P_2O_5$ , forming ketenes  $CF_3C(CO_2R)=C=O$  (**54**, equation 30).⁴³ These ketenes are thermally stable but react with both electrophilic and nucleophilic reagents.⁴³

$$CF_{3} \xrightarrow{CO_{2}R} \xrightarrow{P_{2}O_{5}} \xrightarrow{CF_{3}} C=0$$

$$CO_{2}R \xrightarrow{P_{2}O_{5}} RO_{2}C =0$$

$$CF_{3} \xrightarrow{CO_{2}R} C=0$$

$$CF_{3} \xrightarrow{CO_{2}R}$$

Photolysis of matrix-isolated fluoromaleic anhydride **55** in an Ar matrix gave fluoro(ketenyl)carbene **56** and fluorocyclopropenone (**57**) (equation 31; see also Section 3.8).⁴⁴ Difluoromaleic anhydride (**58**) upon pyrolysis was proposed to form the carbene **59**, which gave the nonlinear cumulene **10** (equation 32).⁴⁵



# **REFERENCES FOR SECTION 4.4.1**

- 1. Gong, L.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1991, 113, 6021-6028.
- 2. McAllister, M. A.; Tidwell, T. T. J. Org. Chem. 1994, 59, 4506-4515.
- 3. Scott, A. P.; Platz, M. S.; Radom, L. J. Am. Chem. Soc. 2001, 123, 6069-6076.
- 4. Davidovics, G.; Monnier, M.; Allouche, A. Chem. Phys. 1991, 150, 395-403.
- 5. Kötting, C.; Sander, W.; Senzlober, M.; Bürger, H. Chem Eur. J. 1998, 4, 1611-1615.
- 6. Sander, W.; Kötting, C. Chem. Eur. J. 1999, 5, 24-28.

#### **330** TYPES OF KETENES

- 7. Russell, D. K.; Wesendrup, R. J. Chem. Phys. 2002, 117, 5252-5256.
- 8. Lewars, S. J. Mol. Struct. 2002, 579, 155-167.
- Zachariah, M. R.; Westmoreland, P. R.; Burgess, D. R., Jr.; Tsang, W.; Melius, C. F. J. Phys. Chem. 1996, 100, 8737–8747.
- Srivatsava, A.; Arunan, E.; Manke, G., II; Setser, D. W.; Sumathi, R. J. Phys. Chem. A 1998, 102, 6412–6420.
- Breidung, J.; Bürger, H.; Kötting, C.; Kopitzky, R.; Sander, W.; Senzlober, M.; Thiel, W.; Willner, H. Angew. Chem, Int. Ed. Engl. 1997, 36, 1983–1985.
- 12. Dawson, D. F.; Holmes, J. L. J. Phys. Chem. A 1999, 103, 5217-5220.
- Welch, J. T.; Arakai, K.; Kawecki, R.; Wichtowski, J. A. J. Org. Chem. 1993, 58, 2454– 2462.
- 14. Dolbier, W. R., Jr.; Lee, S. K.; Phanstiel, O., IV Tetrahdron 1991, 47, 2065–2072.
- 15. England, D. C.; Krespan, C. G. J. Org. Chem. 1968, 33, 816-819.
- 16. Habibi, M. H.; Saidi, K.; Sams, L. C. J. Fluorine Chem. 1987, 37, 177-181.
- 17. Saidi, K.; Habibi, M. H. J. Fluorine Chem. 1991, 51, 217-222.
- 17a. Cordaro, J. G.; van Halbeek, H.; Bergman, R. G. Angew. Chem. Int. Ed. 2004, 43, 6366–6369.
- 18. Iwaoka, T.; Murohashi, T.; Sato, M.; Kaneko, C. Synthesis 1992, 977-981.
- 19. Soelch, R. R.; Mauer, G. W.; Lemal, D. M. J. Org. Chem. 1985, 50, 5845-5852.
- 20. Ishihara, T.; Yamasaki, Y.; Ando, T. Tetrahedron Lett. 1986, 27, 2879-2880.
- Strazzolini, P.; Giumanini, A. G.; Gambi, A.; Verardo, G.; Cerioni, G. Can. J. Chem. 1996, 74, 1348–1359.
- Cheburkov, Y. A.; Bargamova, M. D. Izv. Akad. Nauk SSSR, 1967, 833–840. Engl. Transl. 1967, 801–806.
- 23. Brown, F.; Musgrave, W. K. R. J. Chem. Soc., 1953, 2087-2089.
- Park, J. D.; Larsen, E. R.; Holler, H. V.; Lacher, J. R. J. Org. Chem. 1958, 23, 1166– 1169.
- Allen, A. D.; Andraos, J.; Kresge, A. J.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1992, 114, 1848–1879.
- 25a. Haiss, P.; Keller, K.-P. Org. Biomol. Chem. 2003, 1, 2556-2558.
- Förner, W.; Badawi, H. M.; Al-Saadi, A. A.; Ali, S. K. J. Mol. Model. 2001, 7, 343– 353.
- 27. England, D. C; Krespan, C. G. J. Am. Chem. Soc. 1966 88, 5582-5587.
- Mal'tsev, A. K.; Zuev, P. S.; Nefedov, O. M. *Izv. Akad. Nauk SSSR* 1985, 957–958. *Engl. Transl.* 1985, 876.
- 29. Coe, P. L.; Ray, N. C. J. Fluorine Chem. 1989, 45, 90.
- Laganis, E. D.; Janik, D. S.; Curphey, T. J.; Lemal, D. M. J. Am. Chem. Soc. 1983, 105, 7457–7459.
- 31. Schwab, M.; Sundermeyer, W. J. Prakt. Chem./Chem. Ztg. 1997, 339, 479-481.
- Singmaster, K. A.; Jonnalagadda, S.; Chan, S. J. Am. Chem. Soc. 2000, 122, 9078– 9088.
- 33. England, D. C.; Krespan, C. G. J. Org. Chem. 1970, 35, 3300-3307.
- 34. England, D. C.; Krespan, C. G. J. Org. Chem. 1970, 35, 3308-3312.

- 35. England, D. C.; Krespan, C. G. J. Org. Chem. 1970, 35, 3312-3322.
- 36. England, D. C.; Krespan, C. G. J. Org. Chem. 1970, 35, 3322-3327.
- 37. Petrov, V. A.; Davidson, F.; Smart, B. E. J. Fluorine Chem. 2004, 125, 1543– 1552.
- 38. Machiguchi, T.; Okamoto, J.; Takachi, J.; Hasegawa, T.; Yamabe, S.; Minato, T. J. Am. Chem. Soc. 2003, 125, 14446–14448.
- 39. Snegirev, V. F.; Galakhov, M. V.; Petrov, V. A.; Makarov, K. N.; Bakhmuov, V. I. *Izv. Akad. Nauk SSSR* 1986, 1318–1325. *Engl. Transl.* 1986, 1194–1200.
- 40. Farnham, W. B.; Middleton, W. J.; Fultz, W. C.; Smart, B. E. J. Am. Chem. Soc. 1986, 108, 3125–3127.
- 41. Anders, E.; Ruch, E.; Ugi, I. Angew. Chem., Int. Ed. 1973, 12, 25-29.
- 42. Coe, P.L.; Owen, I. R.; Sellers, A. J. Chem. Soc. Perkin Trans. 1 1989, 1097–1103.
- 43. Kryukova, L. Y.; Kryukov, L. N.; Truskanova, T. D.; Isaev, V. L.; Sterlin, R. N.; Knunyants, I. L. Dokl. Akad. Nauk SSSR 1977, 232, 1311–1313. Engl. Transl. 1977, 232, 90–92.
- 44. Dailey, W. P. 211th National American Chemical Society meeting, Chicago, August 1995, Abstracts ORGN 096.
- 45. Ionescu, A.; Piétri, N.; Hillebrand, M.; Monnier, M.; Aycard, J. P. Can. J. Chem. 2002, 80, 455–461.

### 4.4.2 Chloro- and Bromoketenes

The study of haloketenes is a relatively recent development in ketene chemistry, even though attempts were made by early investigators to prepare these species, including dichloro-,¹ chloro(ethyl)-,² and bromo(methyl)ketenes **1**–**3**.³ However, none of these ketenes were isolated, and only ill-defined products were obtained. Finally, in 1965–1966, haloketenes were proven to exist by their generation and capture in independent studies in three different laboratories,^{4–6} but even then, these species were not directly observed. The haloketenes are highly reactive, and their synthetic utility quickly became apparent, particularly in [2 + 2] cycload-ditions with alkenes, alkynes, imines, and other reagents. Their chemistry has been actively pursued ever since and has been discussed in earlier reviews.^{7–9}

 $\begin{array}{ccc} Cl & Cl & Cl & Br \\ \searrow = C = O & Et & C = O \\ 1 & 2 & 3 \end{array}$ 

Chloroketenes are conveniently prepared by the dehydrochlorination of  $\alpha$ -chloroalkanoyl chlorides and trapped by reactive alkenes such as cyclopentadiene by [2+2] cycloadditions, forming bicyclo[3.2.0]pent-2-ene-6-ones **8** (equation 1).¹⁰ The adduct **8a** of dichloroketene was used in a simple and efficient preparation of tropolone **9** (equation 2).¹¹Dichloroketene (1) formed by dehydrochlorination of dichloroacetyl chloride (10) also gave an efficient reaction with the highly reactive *trans*-cyclooctene to form the [2 + 2] cycloaddition product **11** (equation 3).¹²



Chloroketene (4) was also obtained by the reaction of dichloroacetyl chloride (10) with Me₃SiPPh₂ at low temperature, forming 12, followed by heating and trapping of the ketene at -80 °C for measurement of the ¹³C NMR spectrum (equation 4).^{13,13a}

Dichloroketene (1) has been prepared both by the dehydrochlorination of dichloroacetyl chloride (equations 1, 3)^{10–12} and by the zinc dehalogenation of trichloroacetyl chloride, and in each case the ketene has been generated and used in situ.^{7–9} A variety of methods are used for activating the zinc used for the dehalogenation, including the use of ultrasound^{14.15} for the reaction with norbornene forming **13** (equation 5)¹⁴ or simple heating.¹⁶



Halogen-substituted ketenes are now widely used in synthesis, but in almost all cases the ketenes are not observed in preparative studies. Even derivatives with ketene-stabilizing silyl substitutents have not been isolated. The high reactivity of halogen-substitued ketenes is in agreement with the results of *ab initio* calculations,^{17,18} which indicate that chloroketene (**4**) is destabilized by 7.7 kcal/mol at the MP2/HF6-31G* level compared to methylketene (**14**) by isodesmic energy comparison (equation 6). This result was explained by the finding that ketenes are destabilized by electronegative substituents, and stabilized by electropositive groups, in agreement with the high reactivity of chloroketenes observed in practice.

$$\overset{\text{Cl}}{=} C = O + \overset{\text{CH}_3}{\stackrel{\frown}{=}} CH_2 \xrightarrow{\Delta E =} \overset{\text{CH}_3}{\stackrel{\frown}{=}} C = O + \overset{\text{Cl}}{\stackrel{\frown}{=}} CH_2 \qquad (6)$$
4

This high reactivity also has made these species quite useful, especially as they can be prepared using inexpensive starting materials, and the products can be used in a variety of further transformations. The Wolff rearrangement method, which is widely used in the preparation of many types of ketenes, has not, however, found application in the preparation of halogen-substituted ketenes. For direct observation of halogen-substituted ketenes, generation in the gas phase or in low-temperature matrices have been successful, as well as preparation at low temperature in solution.^{13,19}

Cycloaddition reactions of chloroketenes with alkenes gave chlorocyclobutanones that have been utilized in many synthetic transformations, as in the dehydrochlorination of the  $\alpha$ -chloroacyl chloride **15**, forming ketene **16**, which reacted in situ with cyclohexadiene by [2 + 2] cycloaddition, forming **17** (equation 7).²⁰ Reaction of **17** with *n*-Bu₃SnH resulted in cyclization to the unobserved intermediate **18**, which formed the product **19** by rearrangement and dechlorination (equation 8).²¹ Other related examples were observed.²²



Dichloroketene (1) generated by zinc reduction reacted with the alkene 20, forming the cyclobutanone 21, which underwent reductive cyclization to 22

(equation 9).²³ Dichloroketene reacted with cyclohexene, forming the [2+2] cycloadduct **23** (equation 10), and this was converted to the rearranged ketone **25** (equation 11).^{24,25}



The dichloroketene adduct with 1-methylcyclohexene was converted to cyclopropyl carboxylate ester **26**, which was a precursor to vinylketones **27** (equations 12, 13).²⁶ The [2 + 2] cycloaddition of dichloroketene generated by dehydrochlorination with the silyl alkenyl ether **28**, forming **29**, was evaluated in a pilot plant scale reaction (equation 14), but was unsatisfactory due to problems with the subsequent dehalogenative ring expansion.²⁷



In the synthesis of the CP compounds, which are complex antibiotics, [2+2] cycloaddition of dichloroketene with the alkene **30** was followed by dechlorination with zinc to give **31**, which was converted to the final target (equation 15).²⁸



Other examples of CCl₂=C=O cycloadditions included reactions with a vinyl ether forming **32** (equation 16),²⁹ addition to a carbohydrate derivative,³⁰ and reaction with 4-*m*-carboranyl-1-butene (Section 4.8.2).³¹ The reaction with allyl-substituted thioether proceeded by nucleophilic attack and rearrangement, forming **33** (equation 17).³² In [2 + 2] cycloadditions of **1**, ring opening could be suppressed by the substitution of DME for POCl₃ as the ZnCl₂ sequestering agent.³³ Allyl ethers react similarly. Reaction of a vinyl epoxide **34** was proposed to form **35**, which gave lactone **36** (equation 18).³³



Dichloroketene was used in reactions with bicyclo[3.3.0]octadiene **37**, forming **38** (equation 19),³⁵ and with dihydrothiophene **39**, forming **40** in the synthesis of a penicillin analogue (equation 20).³⁶



To elucidate the possible role of chloroketenes as intermediates in halo olefin bioactivation and possible DNA binding, the reactions of **1** and **4** with adenine, cytosine, guanine, and thymine were examined. Thymine did not react, but adenine and cytosine gave stable adducts with both **1** and **4**, as demonstrated by analysis, and guanine reacted with  $1.^{37}$  The adducts were identified as resulting from acylation of the exocyclic amino groups, as in the example of the formation of *N*-chloroacetyladenine (**42**) from adenine (**41**) (equation 21).³⁷ In H₂O at pH 7, **42** was cleaved back to **41** and also formed 1,*N*-6-acetylguanine **43** (equation 21).³⁷



Dimers of chloroketene and dichloroketene have apparently not been isolated and characterized, but in situ hydrolysis of the product of dehydrochlorination of ClCH₂COCl led to the isolation of **44** in 1.8% yield, and this implicated the formation and hydrolysis of the chloroketene dimer (equation 22).³⁸



Dehydrochlorination of dichloroacetyl chloride in ether at reflux over an extended time period led to the two oligomeric products **45b** and **45c**, proposed to form as shown in equations 22a–22c.³⁹



Experiments in which  $CH_2CICOCI$  and  $CHCl_2COCI$  were reacted with  $Et_3N$  and  $CH_3OD$  resulted in the formation of methyl esters that were monodeuterated to the extent of 32 and 41%, respectively, suggesting that the esterification took place partially through ketene intermediates.⁴⁰

Bromoketene (**46**) has been prepared by gas phase dehydrochlorination of bromoacetyl chloride at 700–800 °C and used for determination of the photoelectron spectrum.⁴¹ The reaction of bromoacetyl chloride with 1,8-bis(dimethylamino)naphthalene **47** as a strong stochiometric base and benzoylquinine as a shuttle base at -78 °C gave [2 + 2] cycloaddition with the imine EtO₂CCH=NTs forming the  $\beta$ -lactam product **48** in 61% yield, 96% *ee*, and 98 : 2 *dr* (equation 23a).⁴² However, it is not certain if bromoketene was a discrete intermediate in this reaction, as an enolate may have been involved.⁴² Bromoketene generation was, however, carried out using the insoluble polymeric base (BEMP), and stereoselective chlorination was achieved using hexachlorocyclohexadienone (HCC) as the chlorinating agent. In this case, formation of the ketene was unambiguous (equation 23b).⁴³





Dehydrochlorination of bromochloroacetyl chloride with triethylamine gave bromochloroketene (**50**), and this reacted in situ by [2 + 2] cycloaddition with cyclopentadiene, forming **51** in 60% yield, with selectivity for the bromo group in the *endo* position (equation 24).⁸ Bromophenylketene (**52**) was also formed by dehydrochlorination, and reacted with cyclopentadiene to give **53** with the bromo group preferentially in the *exo* position (equation 25).⁴⁴ Bromoarylketenes (**56**) were also formed from 2,2-dicyano-3-aryloxiranes (**54**) via intermediates **55** (equation 26) and gave stereoselective reactions with chiral amines **57**, forming **58** (equation 27).⁴⁵



Bromotrimethylsilylketene (59) was prepared by dehydrobromination and reacted with imine 60 to form the  $\beta$ -lactam 61 (equation 28).⁴⁶ Ketene 59 was long-lived in solution, as evidenced by observation of the ketenyl IR band at

2114 cm⁻¹, but polymerized upon removal of the Et₃NHCl salt.⁴⁶



Alkylbromo- and alkyliodoketenes **62** generated by dehydrochlorination of  $\alpha$ bromoalkanoyl chlorides reacted with *R*-pantolactone (**63**) to give esters **64** with diastereoselectivities of 75 to >95% (equation 29).⁴⁷



Palladium(II) complexes affect the *exo/endo* selectivity of [2 + 2] cycloadditions of bromo(alkyl)ketenes **66**, prepared by triethylamine dehydrochlorination of 2-bromoalkanoyl chlorides **65**, with cyclopentadiene forming bicyclo[3.2.0]hept-2-en-7-ones **67** (equation 30).^{48,49} The palladium catalyst increased the yields of cycloaddition products from 23–58% to 42–91%, with a modest increase in the preference for *exo* alkyl groups in the products, although in most cases *endo* alkyl stereochemistry was still predominant.



Dibromoketene (68) generated from zinc debromination of CBr₃COBr was captured by benzylamine, and when generated in the presence of alkenes such as *n*decene gave dibromocyclobutanone 69, which was cleaved by alkoxides to give  $\beta$ -dibromomethyl ester 70 (equation 31).⁵⁰ Reaction of trimethylsilyl tribromoacetate with triphenylphosphine formed dibromoketene, which reacted with cyclopentadiene, forming 71 (equation 32).⁵¹

$$CBr_{3} \xrightarrow{O}_{Br} \xrightarrow{Zn}_{POCl_{3}} \xrightarrow{Br}_{Br} C=O \xrightarrow{n-C_{8}H_{17}}_{n-C_{8}H_{17}} \xrightarrow{Br}_{r-BuOK} \xrightarrow{O}_{CH_{3}OH} \xrightarrow{CHBr_{2}}_{r-BuOK} \xrightarrow{CO_{2}CH_{3}}_{n-C_{8}H_{17}}$$

$$68 \qquad 69 (70\%) \qquad 70 (71\%)$$

$$(31)$$



α-Bromo-α-alkenylketenes **73** generated by dehydrochlorination of acyl chorides **72** reacted in situ with *R*-pantolactone (**63**), forming esters **74** with yields of 50–85% and diastereomeric ratios ranging from 72:28 to >95:5 (equation 33).⁵² Displacement of the bromine by benzylamine forming α-amino esters was followed by further synthetic elaboration of the resulting products. The alkenylketenes **73** also gave [4 + 2] cycloadditions with imines forming 3-bromo-5,6-dihydropyridin-2-ones.⁵³ Dehydrochlorination of 2-bromo-2-hexenoyl chloride (**75**) with triethylamine gave the ketene **76**, which was captured in situ by imines **77** giving the bromo-substituted β-lactams **78** (equation 34).⁵⁴



#### **REFERENCES FOR SECTION 4.4.2**

- 1. Staudinger, H.; Anthes, E.; Schneider, H. Chem. Ber. 1913, 46, 3539-3551.
- 2. Ott, E. Liebigs Ann. 1914, 401, 159-177.
- 3. Staudinger, H.; Schneider, H. Helv. Chim. Acta 1923, 6, 304-315.
- Stevens, H. C.; Reich, D. A.; Brandt, D. R.; Fountain, K. R.; Gaughan, E. J. J. Am. Chem. Soc. 1965, 87, 5257–5259.
- 5. Brady, W. T.; Liddell, H. G.; Vaughn, W. L. J. Org. Chem. Soc. 1966, 31, 626-628.
- 6. Ghosez, L.; Montaigne, R.; Mollet, P. Tetrahedron Lett. 1966, 135-139.
- 7. Brady, W. T. Tetrahedron 1981, 37, 2949-2966.
- 8. Brady, W. T. Synthesis 1971, 415-422.

- Leahy, J. W. In *Encyclopedia of Reagents for Organic Synthesis*; Paquette, L. A., Ed.; Wiley, New York, 1995, pp. 1714–1719.
- 10. Brook, P. R.; Duke, A. J.; Harrison, J. M.; Hunt, K. J. Chem. Soc. Perkin Trans. 1 1974, 927–932.
- 11. Minns, R. A. Organic Syntheses, Coll. Vol. VI, Noland, W. E., Ed.; Wiley, New York, 1988, pp. 1037–1040.
- 12. Montaigne, R.; Ghosez, L. Angew. Chem., Int. Ed. 1968, 7, 221.
- 13. Lindner, E.; Steinwand, M.; Hoehne, S. Angew. Chem. Int. Ed. 1982, 21, 355-356.
- 13a. Lindner, E.; Steinwand, M.; Hoehne, S. Chem. Ber. 1982, 115, 2181-2191.
- 14. Mehta, G.; Rao, H. S. P. Synth. Commun. 1985, 15, 991-1000.
- 15. Parker, M. S. A.; Rizzo, C. J. Synth. Commun. 1995, 25, 2781-2789.
- 16. Stenstrøm, Y. Synth. Commun. 1992, 22, 2801-2810.
- 17. Gong, L.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1991, 113, 6021-6028.
- 18. McAllister, M. A.; Tidwell, T. T. J. Org. Chem. 1994, 59 4506-4515.
- 19. Duddeck, H.; Praas, H-W. Magn. Res. Chem. 1993, 31, 182-184.
- 20. Dowd, P.; Zhang, W.; Mahmood, K. Tetrahedron Lett. 1994, 35, 5563-5566.
- 21. Dowd, P.; Zhang, W.; Mahmood, K. Tetrahedron 1995, 51, 39-50.
- 22. Zhang, W.; Dowd, P. Tetrahedron Lett. 1996, 37, 957-960.
- 23. Zhang, W.; Dowd, P. Tetrahedron Lett. 1994, 35, 5161-5164.
- 24. Zhang, W.; Hua, Y.; Hoge, G.; Dowd, P. Tetrahedron Lett. 1994, 35, 3865-3868.
- 25. Zhang, W.; Hua, Y.; Geib, S. J.; Hoge, G.; Dowd, P. Tetrahedron 1994, 50, 12579-12592.
- 26. Mehta, G.; Acharyulu, P. V. R. Tetrahedron Lett. 1993, 34, 8157-8158.
- Ragan, J. A.; Murry, J. A.; Castaldi, M. J.; Conrad, A. K.; Jones, B. P.; Li, B.; Makowski, T. W.; McDermott, R.; Sitter, B. J.; White, T. D.; Young, G. R. Org. Proc. Res. Dev. 2001, 5, 498–507.
- 28. Meng, D.; Danishefsky, S. J. Angew. Chem. Int. Ed. Engl. 1999, 38, 1485-1488.
- 29. Nebois, P.; Greene, A. E. J. Org. Chem. 1996, 61, 5210-5211.
- 30. Oh, J. Tetrahedron Lett. 1997, 38, 3249-3250.
- 31. Srivastava, R. R.; Singhaus, R. R.; Kabalka, G. W. J. Org. Chem. 1997, 62, 4476-4478.
- 32. Vedejs, E.; Buchanan, R. A. J. Org. Chem. 1984, 49, 1840-1841.
- 33. Johnston, B. D.; Slessor, K. N.; Oehlschlager, A. C. J. Org. Chem. 1985, 50, 114-117.
- 34. Ramesh, U.; Ward, D.; Reusch, W. J. Org. Chem. 1988, 53, 3469-3473.
- 35. Dehmlow, E. V.; Breyholz, H.-J.; Neumann, B.; Stammler, A.; Stammler, H.-G. *Eur. J. Org. Chem.* **2001**, *4*, 1413–1420.
- Martyres, D. H.; Baldwin, J. E.; Adlington, R. M.; Lee, V.; Probert, M. R.; Watkin, D. J. *Tetrahedron* 2001, *57*, 4999–5007.
- 37. Müller, M.; Birner, G.; Dekant, W. Chem. Res. Toxicol. 1998, 11, 454-463.
- 38. Bellus, D.; Fischer, H.-P.; Greuter, H.; Martin, P. Helv. Chim. Acta 1978, 61, 1784–1813.
- 39. Hopf, H.; Stamm, R.; Jones, P. G. Chem. Ber. 1991, 124, 1291-1294.
- 40. Truce, W. E.; Bailey, P. S., Jr. J. Org. Chem. 1969, 34, 1341-1345.
- 41. Bock, H.; Hirabayashi, T.; Mohmand, S. Chem. Ber. 1981, 114, 2595-2608.
- 42. Taggi, A. E.; Hafez, A. M.; Wack, H.; Young, B.; Ferraris, D.; Lectka, T. J. Am. Chem. Soc. 2002, 124, 6626–6635.

- France, S.; Wack, H.; Taggi, A. E.; Hafez, A. M.; Wagerle, T. R.; Shah, M. H.; Dusich, C. L.; Lectka, T. J. Am. Chem. Soc. 2004, 126, 4245–4255.
- 44. Brady, W. T.; Parry, F. H., III; Stockton, J. D. J. Org. Chem. 1971, 36, 1486-1489.
- 45. Roger, F.; Le Pironnec, M.-G.; Guerro, M.; Gougeon, P.; Gall, P.; le Grel, P.; Baudy-Floc'h, M., *Synthesis* 1999, 1341–1344.
- 46. Brady, W. T.; Owens, R. A. Tetrahedron Lett. 1976, 1553-1556.
- 47. Durst, T.; Koh, K. Tetrahedron Lett. 1992, 33, 6799-6802.
- 48. Fairlamb, I. J. S.; Dickinson, J.; Cristea, I. M. Tetrahedron 2001, 57, 2237-2246.
- 49. Fairlamb, I. J. S.; Dickinson, J.; Cristea, I. M. Tetrahedron Lett. 2000, 41, 3739–3742.
- 50. Chaumeil, H.; Le Drian, C. Helv. Chim. Acta 1996, 79, 1075-1084.
- 51. Okada, T.; Okawara, R. Tetrahedron Lett., 1971, 2801-2802.
- 52. Cardillo, G.; Fabbroni, S.; Gentilucci, L.; Perciaccante, R.; Tolomelli, A. *Tetrahedron* Asymm. 2004, 15, 593-601.
- 53. Cardillo, G.; Fabbroni, S.; Gentilucci, L.; Perciaccante, R.; Piccinelli, F.; Tolomelli, A. *Tetrahedron* **2004**, *60*, 5031–5040.
- 54. Cardillo, G.; Fabbroni, S.; Gentilucci, L.; Perciaccante, R.; Piccinelli, F.; Tolomelli, A. *Org. Lett.* **2005**, *7*, 533–536.

# 4.5 SILYL-, GERMYL-, AND STANNYLKETENES

Trimethylsilylketene (1) is a remarkably stable ketene that was first reported in 1965 from the pyrolysis of (trimethylsilyl)ethoxyacetylene (2) as an isolable oil with no apparent tendency toward dimerization (equation 1).^{1,2} It was found to react readily with nucleophiles^{1–3} and to undergo Wittig reactions forming allenes.³

$$\begin{array}{ccc} Me_{3}SiC \equiv COEt & \xrightarrow{120 \ \circ C} & & Me_{3}Si \\ 2 & & & C \equiv O \\ & & & 1 \ (90\%) \end{array}$$
(1)

A quantitative measure of the stabilization of ketenes provided by the silyl substituent was given by the MP2/6-31G*//MP2/6-31G* isodesmic energy change of 11.5 kcal/mol shown in equation  $2^{4.5}$ 

A review devoted to silylketenes has appeared,⁶ as well as concise descriptions of the chemistry of Me₃SiCH=C=O⁷ and CH₂=CHC(SiMe₃)=C=O (7).⁸ An optimized procedure for the preparation of 1 by the alkynyl ether pyrolysis of equation 1 has been presented,⁹ and pyrolysis of the *tert*-butoxy ether Me₃SiC=COBu-*t* 

forming **1** proceeded at 50 °C.¹⁰ Other preparations of **1** included the dehydrochlorination of the acyl chloride **8** (equation 3),¹¹ dehydration of trimethylsilylacetic acid **9** (equation 4),¹² and treatment of ketene with trimethylsilyl triflate, which reacted further with additional Me₃SiOTf to give (Me₃Si)₂C=C=O **10** (equation 5).¹³ Triethylsilylketene (**11**) was prepared by pyrolysis of the anhydride (equation 6)¹¹ and also by dehydrochlorination of the acyl chloride.¹⁴

$$\begin{array}{c} Me_{3}Si \\ OH \end{array} \xrightarrow{DCC} Me_{3}Si \\ = C=O \\ 9 \\ 1 (63\%) \end{array}$$

$$(4)$$

$$(Et_3SiCH_2C)_2O \xrightarrow{\Delta} Et_3Si = C=O$$

$$11 (80\%)$$
(6)

Bis(trimethylsilyl)ketene (10) has been prepared by treatment of the alkynyl ether 12 with Me₃SiBr (equation 7),^{15,16} by reaction of Me₃SiCl with the ynolate 13 formed by deprotonation of 1 (equation 8),¹⁷ by the E1cb elimination from the ester 14 (equation 9),¹⁸ and from photochemical Wolff rearrangement of the diazo ketone 15 as a by-product in a reaction that evidently proceeded through the silene intermediate 16 (equation 10).¹⁹





Bromotrimethylsilylketene **17** was formed by bromination/dehydrobromination of **1** and identified in solution by IR and trapped by [2 + 2] imine cycloaddition, forming the  $\beta$ -lactam **18** (equation 11),²⁰ Phenyltrimethylsilylketene **19** was prepared by dehalogenation of the  $\beta$ -bromoacyl chloride (equation 12),²¹ while *n*-butyltrimethylsilylketene **20** was obtained from silylation of the alkynyl ether (equation 13),²² and Me₃SiC(Pr-*i*)=C=O²³ and Me₃SiC(Hx-*n*)=C=O²⁴ were obtained similarly. Dehydrochlorination provided trimethylsilylketene **21** in 39–50% yield starting from the acid (equation 14).²⁵ Ethyltrimethylsilylketene **(22)** was also obtained by dehydrochlorination (equation 15).²⁶



Ketene	$\mathbb{R}^1$	R ₃ Si	Ketene	$\mathbb{R}^1$	R ₃ Si
25a	Ph	Et ₃ Si (87%)	25g	t-Bu	t-BuPh ₂ Si (26%)
25b	2-Thienyl	<i>i</i> -Pr ₃ Si (94%)	25h	<i>t</i> -Bu	<i>i</i> -Pr ₃ Si (38%)
25c	Me	Et ₃ Si (54%)	25i	Ph ₂ CH	<i>i</i> -Pr ₃ Si (71%)
25d	Me	<i>i</i> -Pr ₃ Si (49%)	25j	1-Ad	t-BuMe ₂ Si (72%)
25e	<i>t</i> -Bu	t-BuMe ₂ Si (50%)	25k	1-Ad	t-BuPh2Si (71%)
25f	<i>t</i> -Bu	t-BuMe ₂ Si (39%)	251	1-Ad	<i>i</i> -Pr ₃ Si (62%)

TABLE 4.3 Silylketenes from Photochemical Wolff Rearrangement (equation 16)

Silyl-substituted ketenes **25** were prepared from photochemical Wolff rearrangements of **24** obtained by silylation of diazo ketones **23** (equation 16).^{27,28} Examples are shown in Table 4.3. The rather high energy of the UV absorption of the product silylketenes allows them to undergo minimal photodecarbonylation by the light used, contributing to the success of this procedure.



Photolysis of ethyl (pentamethyldisilanyl)diazoacetate (**26**) in alcohol solvents gave the bis(silyl)-substituted ketene **28** by a process that could have involved successive migration of Me₃Si, addition of ROH to the silene **27**, and loss of EtOH (equation 17).^{29,30}



Silyl(alkenyl)ketene **30** was prepared in 20% yield from the chromium carbene complex **29** (equation 18), and in addition the  $Cr(CO)_3$  complex of **30** was obtained from the same reaction in 52% yield.^{31,32} Other preparations of silyketenes from metal carbene complexes are given in Section 3.5.

Dihydrofuran (**31**) reacted with *n*-BuLi, forming ynolate **32**, which was silvlated by  $R_3SiCl$  and converted to a silvl ynolate that gave bis(silvl)ketenes **10** and **33a**,**b** (equation 19).³³

$$(19)$$

$$\xrightarrow{n-\text{BuLi}}_{-\text{CH}_2=\text{CH}_2} \text{HC} \equiv \text{COLi} \xrightarrow{R_3\text{SiCl}}_{n-\text{BuLi}} \xrightarrow{R_3\text{Si}}_{R_3\text{Si}} = \text{C} = 0$$

$$(19)$$

$$31 \qquad 32 \qquad \text{R} = \text{Me}_3\text{Si} \text{ 10 (21\%), Me}_2\text{HSi}$$

$$33a (28\%), t-\text{BuMe}_2\text{Si} 33b (40\%)$$

The extraordinary resistance to dimerization of trimethylsilylketene (1) was attributed by Brady and Cheng³⁴ to ground state stabilization arising from hyperconjugative electron donation from the C-Si bond to the carbonyl group, as represented by resonance structure **1a**. Such donation is made possible by the coplanarity of the C-Si bond and the carbonyl pi system.



An alternative explanation is that the stabilization of silylketenes was due to back electron donation from the pi system of the carbonyl group to *d* orbitals on silicon,³⁵ However, while electron-donation from C-Si bonds to electron-deficient centers is well documented, there is little support for such d- $\pi$  bonding between carbon and silicon.^{36–39} The explanation of Brady and Cheng,³⁴ namely, that the stabilization of ketenes by silyl substituents is due to the  $\beta$ -silicon effect,^{36,38} is consistent with many aspects of silicon chemistry.

As noted in Section 5.5.1.1, trimethylsilyketene (1) is less reactive than ketene toward water by a factor of 160,⁴⁰ and this may also be attributed to the ground state stabilization of 1.

Silylketenes react with diazomethanes to form cyclopropanones and cyclobutanones, as in the formation of **34** by reaction of trimethylsilylketene **1** with trimethylsilyldiazomethane (equation 20).^{21,41}



Photolysis in a matrix at 77 K of the cyclopentadienone **35** gave the allenylketene **36**, which was characterized by IR bands at 1890 and  $2080 \text{ cm}^{-1}$  and which formed the tetra(trimethyl)silylbutatriene **37** on further photolysis (equation 21).⁴²



Photolysis of (trimethoxysilyl)diazomethane in an Ar matrix gave the carbene **38**, which was characterized by IR and UV spectroscopy, and when CO was included in the matrix, it reacted to give the ketene **39**, identified by its IR band at 2127 cm⁻¹ (equation 22).⁴³ Photolysis of the diazo ester **40** with 300 nm light in toluene gave the housane derivative **41**, isolated in 68% yield, which upon thermolysis gave the ketene **43** in 20% yield, with a characteristic IR absorption at 2089 cm⁻¹ (equation 23).⁴⁴ This transformation was proposed to occur by a [2 + 2] cycloreversion giving the silaethene **42**, which rearranged to **43** by a 1,3 OMe shift.⁴⁴



Carbonylation of the thorium complex **44** gave the silylated metalloxy ketene complex **45** with an X-ray crystal structure similar in its main dimensions to those of other ketenes (equation 24).⁴⁵ A complex of Me₃SiCH=C=O (1) with a tantalum  $\eta^2$ -(C,O) complex was also obtained.⁴⁶ Carbonylation of the lithiated silyldiazomethane **46** was proposed to occur through a sequence involving diazo ketone **47**, azo ketene **48**, and ynolate **13** leading to the isolated ketene **50** (equations 25, 26).⁴⁷ Bis(trimethylsilyl)ketene **10** reacted with *t*-BuOK, forming potassium ynolate **51**, which combined with Ph₃SiCl, forming the mixed bis(trialkylsilyl)

ketene **52** (equation 27).⁴⁸ Trimethylsilyldiazomethane was carbonylated with octacarbonyl dicobalt catalysis to trimethylsilylketene (1).^{48a}



Lithium trimethylsilyl ynolate **13** added to epoxides in a reaction promoted with Me₃Al, forming ketenes **54** that cyclize to the observed lactones **55** (equation 28).⁴⁷ Diethyl benzalmalonate reacted with **13**, giving ketene **57** in 62% isolated yield (equation 29).⁴⁷



Propylene imine **58** reacted with **13** to give lactam **60** through a ketene intermediate **59** (equation 30).⁴⁹ This reaction was proposed to proceed through enolate intermediate **61** which was also trapped with pivaldehyde forming **62** which gave

**63** by Peterson olefination (equation 31).⁴⁹ Aldimine **64** and **13** gave ketene intermediate **65**, which formed **66** (equation 32).⁴⁷



Following the procedure of equation 8,¹⁷ the reaction of *t*-BuMe₂SiCH=C=O (67) with *n*-BuLi at -100 °C gave ynolate 68, which reacted with cyclohexanone to give  $\beta$ -lactone 69 (equation 33).⁵⁰ Ynolate 68 reacted with Me₃SiCl at -70 °C to form ketene 70, but reaction with *t*-BuMe₂SiCl and workup at -50 °C gave the ynol ether 71 (equation 34).⁵⁰ Reaction with Ph₃PbCl gave *t*-BuMe₂Si(Ph₃Pb)C=C=O in 85% yield.⁵⁰



The [2+2] cycloaddition reaction of trimethylsilyketene with alkenes was usually unsuccessful,³ but did take place with aldehydes to give oxetanones^{3,51} and with tetralkoxyethylene to give cyclobutanone **72** (equation 35).⁵² The ketene **73** prepared in situ from a carbene complex was more reactive, and gave [2+2]

cycloaddition with reactive alkenes (equation 36).⁵³



Trimethylsilylketene **1** gave [2+2] cycloaddition with aliphatic aldehydes, forming desilylated oxetanones ( $\beta$ -lactones) that reacted with thiols, forming acids **74** (equation 37).⁵⁴ Hydrolysis of oxetanones formed from silylketenes **75** gave substituted acrylic acids **76** (equation 38; see also Section 5.4.5).^{55,56} Catalysis by EtAlCl₂ of oxetanone formation from Me₃SiC(Hx-*n*)=C=O and a chiral aldehyde in the synthesis of lipstatin has also been reported.⁵⁷ Reaction of Me₃SiCH=C=O with aldehydes catalyzed by methylaluminoimidazolines gave preferential formation of *cis*-oxetanones in 32–85% yields and 30–83% *ee*.^{58,59} Trimethylsilylketene gave [2+2] cycloaddition with cyclohexanecarboxaldehyde using the dichlorotitanium-TADDOL catalyst **77**, forming  $\beta$ -lactone **78** in 66% yield, 80% *ee* (equation 39).⁶⁰



Trimethylsilylketene gave stereoselective [2+2] cycloaddition with ethyl glyoxylate (79) catalyzed by chiral dirhodium(II) carboxamidates (80), forming  $\beta$ -lactones 81 (equation 40).⁶¹ A survey of chiral catalysts was carried out, and it

was found that the addition of quinine as a cocatalyst gave up to 99% ee and shorter reaction times.⁶¹



The combination of two equivalents of trimethylsilylketene with enamine **82** gave resorcinol derivatives **83** by the proposed mechanism shown, with silyl migration to oxygen (equation 41).⁶² 1,3-Diaza-1,3-diene **84** reacted with **1** by [4 + 2] cycloaddition, forming 4(*3H*)-pyrimidone **85** (equation 42).⁶³ Reactions of trialkyl-silylketenes **86** with Wittig reagents gave allenes **87** (equation 43).⁶⁴



Reaction of bis(silylated) ketenes 10, 70, and 88 with organolithium reagants formed Z- and E- $\alpha$ -silylenolates 89, which equilibrated under the reaction conditions and in the presence of HMPA underwent a *syn*-Peterson elimination to give good yields of silylated alkynes 90 (equation 44).⁶⁵ Organocerium reagents reacted with silylketenes 91, leading to silyl ketones 92 (equation 45).⁶⁶




[4+2] Cycloaddition of trimethylsilylketene with benzoylisocyanate was proposed to form 4-trimethylsilyloxy-1,3-oxazin-6-one **93**, which reacted in situ with electron-deficient alkynes to give 2-pyridone **94** (equation 46)⁶⁷ and with enamine **95** to give bicyclic 2-pyridone **96** (equation 47).⁶⁸ Reaction of trimethyl-silylketene (1) with  $\alpha$ ,*N*-diphenylnitrone gave oxindole **97** (equation 48), and reaction with *N*-benzylidenebenzylamine *N*-oxide gave **98** (equation 49).⁶⁹



Pyrolysis of ethoxyalkynes gave the disilanylketene **99**, the disilanylbisketene **100**, and the 1,6-bisketene **101** as isolable, long-lived species that showed characteristic ketenyl NMR and IR spectra (equations 50-52).⁷⁰ The relative rates of hydration of **99–101** compared to Me₃SiCH=C=O in 60% H₂O/CH₃CN at 25 °C were 1.9, 5.6, and 0.9, respectively.⁷⁰

$$Me_{5}Si_{2}C \equiv COEt \xrightarrow[-CH_{2}=CH_{2}]{} Me_{5}Si_{2} = C = O$$

$$99 (IR \ 2108 \ cm^{-1})$$
(50)



Isodesmic energy comparisons based on *ab initio* calculated energies at the RHF/6-31G*//RHF/6-31G* level indicated that the Si₂H₅ substituent has essentially the same ketene-stabilizing ability as SiH₃ (equation 53) and that each ketenyl group in (SiH₂CH=C=O)₂ had essentially the same stabilization as that in SiH₃CH=C=O (equation 54).⁷⁰



Isodesmic comparisons of  $(SiH_3)_2C=C=O(104)$  and  $SiH_3CH=C=O(3)$  (equations 55, 56) indicated that the stabilizing effects of the two silyl groups on this ketene are additive.⁷¹ Both of the silyl groups also contributed to the upfield ¹⁷O NMR chemical shifts seen in silylketenes, as  $(Me_3Si)_2C=C=O(10)$ ,  $Me_3SiCH=C=O(1)$ , and  $Me_2C=C=O$  have  $\delta$  211.2, 255, and 329, respectively.⁷⁰ There were also cumulative effects on the UV spectra of silylketenes, as  $(Me_3Si)_2C=C=O$ ,  $Me_3SiCH=C=O$ , and *t*-Bu₂C=C=O have their highest-wavelength UV  $\lambda_{max}$  at 278, 308, and 360 nm, respectively. This latter effect was attributed to an interaction of the ketenyl C-Si  $\sigma$  bond with the in-plane carbonyl  $\pi$  and  $\pi^*$  orbitals, lowering the energy of the former and raising the energy of the latter.⁷⁰ As there was little interaction with the *n* orbitals on oxygen, the net result was an increase in the  $n \to \pi^*$  energy on silyl substitution.⁷⁰

3

The hydration rate ratios  $k(CH_2=C=O)/k(Me_3SiCH=C=O)$  of 160 and  $k(CH_2=C=O)/k[(Me_3Si)_2C=C=O]$  of 1030 indicated that both silvl groups decreased the reactivity of ketenes. This has been attributed to an electronic ground state stabilization of the ketenes,⁴⁰ while the smaller rate effect by a factor of 6 for the second Me_3Si group was a measure of the steric effect.⁷⁰

Hydration of *t*-BuMe₂SiCH=C=O (**70**) and Et₃SiCH=C=O (**11**) has been used to prepare the corresponding acids.⁷² The reaction of silylketene **11** with Me₃SiOTf produced a mixture of silyloxyalkyne **105** and mixed bis(silyl)ketene **50** (equation 57).⁷³ Bis(trialkylsilyl), silylgermyl, and bis(germyl)ketenes **107** were prepared by metalation of alkynyl ethers (equation 58).^{74,75} A computational study of the thermal conversion of metallated alkyl alkynyl ethers  $R_3MC\equiv$ COCHRCH₃ to metallated ketenes  $R_3MCH=C=O$  has been reported.⁷⁶



Silylation of Meldrum's acid produced the dioxinone **108**, which was unstable in solution at 0 °C and reacted with alcohols, amines, and thiols to produce acetoacetate derivatives, indicating the intermediacy of ketene **109** (equation 59).¹³



 $\alpha$ -Silylated alkenylketenes **110** and some analogues gave [4 + 1] cycloadditions with various nucleophilic carbenes generated from oxadiazolines, forming cyclopentenones **111** (equation 60).⁷⁷ Other syntheses and reactions of some  $\alpha$ -silylated alkenylketenes are discussed in Section 4.1.2. Trimethylsilyl-substituted dienylketene **113** was generated by thermolysis of cyclobutenone **112** (equation 61), as discussed in Section 3.4.1.⁷⁸





Trialkylsilyl(aryl)ketenes (**115**) from Wolff rearrangements reacted with trimethylsilyldiazomethane in a [4 + 1] annulation process, forming 2-indanones (**116**) (equation 62).⁷⁹ The reactions were proposed to involve initial attack of Me₃SiCHN₂ from the side of the ketene opposite the bulky silyl substituent forming an intermediate **117**, which could form **116** directly (equation 63) or react through a cyclopropanone or oxyallylic intermediate.⁷⁹



Reaction of silvlated diazo ketones **118** with rhodium (II) octanoate gave silvl-ketenes **119** that were purified by chromatography and characterized (equation 64).⁸⁰

 $R \xrightarrow{N_{2}} SiEt_{3} \xrightarrow{Rh(O_{2}CC_{7}H_{15})_{4}}_{(12-80\%)} \xrightarrow{Et_{3}Si}_{R} = C = O$ 118
119
(12-80%)
(64)  $R = n-C_{7}H_{15} (a), PhCHCH_{3} (b),$ PhCH=CH (c),Ph (d), 4-Anis (e),
4-ClC_{6}H_{4} (f), furanyl (g),
3-(N-Boc)quinolyl (h), 1-(O-Bn)ribosyl (i)

Photolysis of silyl-subtituted ketenes 1 and 10 proceeded by decarbonylation to carbenes 120, which rearranged, providing a useful synthesis of transient silenes 121 that were captured by nucleophilies (equation 65).^{81,82} The disilane-substituted ketene 99 gave the silenes 123 and 124 resulting from Me₃Si and Me migration in a 9:1 ratio (equation 66), and the reactivities of the silenes with nucleophiles were

measured and compared.^{81,82}



Reaction of *n*-decyltrimethylsilylketene (125) with the cyclohexene complex of  $Ti(O-PrO-i)_2$  to form titanium complex 126 was proposed to insert added acetaldehyde forming 127, which gave 128 (equation 67).⁸³ Complex 127 also added another molecule of acetaldehyde forming intermediate 129, which underwent elimination, forming 130 (equation 68).⁸³



Reaction of **125** with the corresponding propene/tunsten complex was proposed to form the intermediate **129**, which reacted with acetaldehyde to form **130** (equation 69).⁸³



Trimethylsilylketene inserted into the silaaziridine **131** in a reaction proposed to involve attack of the ketenyl oxygen on silicon to form the zwitterionic intermediate **132**, which forms the product **133** (equation 70).⁸⁴ Examination of the product mixture by ¹H NMR showed complete conversion to **133** after 24 hours, but there was considerable decomposition upon chromatography.



Thermal Wolff rearrangement of Wang resin-bound, silvlated diazo ketone **134** was carried out in the absence of solvent by heating to  $180 \,^{\circ}$ C for 20 min to generate the polymer-bound ketene **135** (equation 71).⁸⁵ In the presence of benzyl alcohol the polymer-bound ester **136** was formed, and cleavage gave **137** in 63% yield (equation 72).⁸⁵ A resin with an aliphatic chain and ketene reactions with amines were also studied.⁸⁵



Thermal ring opening of trimethylsilylcyclobutenedione **138** formed bisketene **139**, which gave ketenyl lactone **141**, which could form through the intermediate [2 + 2] cycloaddition dimer **140** (equation 73).⁸⁶



 $\alpha$ -Substituted trimethylsilyl and phenyldimethylsilylketenes **142** were prepared by either dehalogenation or dehydrochlorination routes, and underwent stereoselective esterification with thiophenol catalyzed by cinchona alkaloids, forming  $\alpha$ -silylthioesters **143** in 41–93% yields and enantiomeric excess of 79–93% (equation 74).⁸⁷



Silylated vinylketenes **144** generated by either Wolff rearrangement or cyclobutenone ring opening reacted by [4 + 2] cycloaddition with carbonyl compounds and with imines, forming  $\alpha,\beta$ -unsaturated  $\delta$ -valerolactones **145** and lactams **146**, respectively (equation 75).⁸⁸



Lipase enzyme catalyzed acylation of alcohols in ether by trimethylsilylketene (1), triethylsilylketene (11), and dimethylphenylsilylketene gave esters in yields of 70–99% (equation 76).⁸⁹



# **REFERENCES FOR SECTION 4.5**

- Shchukovskaya, L. L.; Pal'chik, R. I.; Lazarev, A. N. Dokl. Akad. Nauk SSSR 1965, 164, 357–360. Engl. Transl. 1965, 164, 884–890.
- 2. Shchukovskaya, L. L.; Kol'tsov, A. I.; Lazarev, A. N.; Pal'chik, R. I. Dokl. Akad. Nauk SSSR 1968, 179, 892–895. Engl. Transl. 1969, 179, 318–320.
- 3. Ruden, R. A. J. Org. Chem. 1974, 36, 3607-3608.
- 4. Gong, L.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1991, 113, 6021-6028.
- 5. McAllister, M. A.; Tidwell, T. T. J. Org. Chem. 1994, 59, 4506-4515.

- Pommier, A.; Kocienski, P.; Pons, J. -M. J. Chem. Soc. Perkin Trans. 1 1998, 2105– 2118.
- 7. Loebach, J. L.; Danheiser, R. L. In *Encyclopedia of Reagents for Organic Synthesis*; Paquette, L. A., Ed.; Wiley, New York, 1995, pp. 5266–5268.
- 8. Loebach, J. L.; Danheiser, R. L. In *Encyclopedia of Reagents for Organic Synthesis*; Paquette, L. A., Ed.; Wiley, New York, 1995, pp. 5321–5322.
- Black, T. H.; Farrell, J. R.; Probst, D. A.; Zotz, M. C. Synth. Commun. 2002, 32, 2083– 2088.
- 10. Valentí, E.; Pericas, M. A.; Serratosa, F. J. Org. Chem. 1990, 55, 395-397.
- Kostyuk, A. S.; Boyadzhan, Zh. G.; Zaitseva, G. S.; Sergeev, V. N.; Savel'eva, N. I.; Baukov, Yu. I.; Lutsenko, I. F. *Zh. Obshch. Khim.* **1979**, *49*, 1543–1552. *Engl. Transl.* 1346–1353.
- 12. Olah, G. A.; Wu, A.; Farooq, O. Synthesis 1989, 568.
- 13. Uhlig, W.; Tzschach, A. Z. Chem. 1988, 28, 409-410. Chem. Abstr. 1989, 111, 194849n.
- Lutsenko, I. F.; Baukov, Yu. I.; Kostyuk, A. S.; Savelyeva, N. I.; Krysina, V. K. J. Organomet. Chem. 1969, 17, 241–262.
- Ponomarev, S. V.; Erman, M. B.; Lebedev, S. A.; Pechurina, S. Y.; Lutsenko, I. F. Zh. Obshch. Khim. 1971, 41, 127–133; J. Gen. Chem. (Engl. Transl.) 1971, 41, 122– 127.
- Pal'chik, R. I.; Shchukovskaya, L. L.; Kol'tsov, A. I. Zh. Obshch. Khim. 1969, 39, 1792– 1796; J. Gen. Chem. (Engl. Transl.) 1969, 39, 1756.
- 17. Woodbury, R. P.; Long, N. R.; Rathke, M. W. J. Org. Chem. 1978, 43, 376.
- 18. Sullivan, D. F.; Woodbury, R. P.; Rathke, M. W. J. Org. Chem. 1977, 42, 2038–2039.
- 19. Schneider, K.; Daucher, B.; Fronda, A.; Maas, G. Chem. Ber. 1990, 123, 589-594.
- 20. Brady, W. T.; Owens, R. A. Tetrahedron Lett. 1976, 1553-1556.
- 21. Brady, W. T.; Cheng, T. C. J. Organomet. Chem. 1977, 137, 287-292.
- 22. Sakurai, H.; Shirahata, A.; Sasaki, K.; Hosomi, A. Synthesis 1979, 740-741.
- 23. Efrimova, I. V.; Kazanokova, M. A.; Lutsenko, I. F. Zh. Obshch. Khim. 1985, 55, 1647–1649; J. Gen. Chem. (Engl. Transl.) 1985, 55, 1465–1466.
- 24. Pons, J.-M.; Kocienski, P. Tetrahedron Lett. 1989, 30, 1833-1836.
- 25. Danheiser, R. L.; Sard, H. J. Org. Chem. 1980, 45, 4810-4812.
- 26. Baigrie, L. M.; Seikaly, H. R.; Tidwell, T. T. J. Am. Chem. Soc. 1985, 107, 5391-5396.
- 27. Maas, G.; Brückmann, R. J. Org. Chem. 1985, 50, 2801-2802.
- 28. Brückmann, R.; Schneider, K.; Maas, G. Tetrahedron 1989, 45, 5517-5530.
- 29. Ando, W.; Sekiguchi, A.; Sato, T. J. Am. Chem. Soc. 1981, 103, 5573-5574.
- 30. Ando, W.; Sekiguchi, A.; Sato, T. J. Am. Chem. Soc. 1982, 104, 6830-6831.
- 31. Dötz, K. H.; Mühlemeier, J.; Trenkle, B. J. Organomet. Chem. 1985, 289, 257-262.
- 32. Dötz, K. H. Angew. Chem., Int. Ed. Engl. 1979, 18, 954-955.
- 33. Groh, B. L.; Magrum, G. R.; Barton, T. J. J. Am. Chem. Soc. 1987, 109, 7568-7569.
- 34. Brady, W. T.; Cheng, T. C. J. Org. Chem. 1977, 42, 732-734.
- 35. Runge, W. Prog. Phys. Org. Chem. 1981, 13, 315-484.
- Brook, M. A. Silicon in Organic, Organometallic, and Polymer Chemistry; Wiley, New York, 2000, pp. 485–496.

- 37. Apeloig, Y.; Schleyer, P. v. R. In *The Chemistry of Organic Silicon Compounds*, *Part* 2; Patai, S.; Rappoport, Z., Eds.; Wiley, New York, 1989.
- 38. Weber, W. P. Silicon Reagents for Organic Synthesis; Springer-Verlag, Berlin, 1983.
- Karni, M.; Apeloig, Y.; Kapp, J.; Schleyer, P. v. R. In *The Chemistry of Organic Silicon* Compounds; Rappoport, Z.; Apeloig, Y., Eds.; Wiley, New York, 2001, Vol. 3, pp. 1–163.
- 40. Allen, A. D.; Tidwell, T. T. Tetrahedron Lett. 1991, 32, 847-850.
- Fedorenko, E. N.; Zaitseva, G. S.; Baukov, Yu. I.; Lutsenko, I. F. Zh. Obshch. Khim. 1986, 56, 2431–2432. J. Gen Chem. (Engl. Transl.) 2150–2151.
- Maier, G.; Lage, H. W.; Reisenauer, H. P. Angew. Chem., Int. Ed. Engl. 1981, 20, 976– 977.
- 43. Trommer, M.; Sander, W. Organometallics 1996, 15, 736-740.
- 44. Maas, G.; Daucher, B.; Maier, A.; Gettwert, V. Chem. Commun. 2004, 238-239.
- Radu, N. S.; Engeler, M. P.; Gerlach, C. P.; Tilley, T. D.; Rheingold, A. J. Am. Chem. Soc. 1995, 117, 3621–3622.
- 46. Schmidt, J. A. R.; Arnold, J. Organometallics 2002, 21, 3426-3433.
- 47. Kai, H.; Iwamoto, K.; Chatani, N.; Murai, S. J. Am. Chem. Soc. 1996, 118, 7634–7635.
- 48. Ito, M.; Shirakawa, E.; Takaya, H. Synlett 2002, 45, 1329-1331.
- 48a. Ungvári; Kégl, T.; Ungváry, F. J. Mol. Catal. A: Chem. 2004, 219, 7-11.
  - 49. Iwamoto, K.; Kojima, M.; Chatani, N.; Murai, S. J. Org. Chem. 2001, 66, 169-174.
  - 50. Akai, S.; Kitagaki, S.; Naka, T.; Yamamoto, K.; Tsuzuki, Y.; Matsumoto, K.; Kita, Y. J. Chem. Soc. Perkin Trans. 1 **1996**, 1705–1709.
  - 51. Brady, W. T.; Saidi, K. J. Org. Chem. 1979, 44, 733-737.
  - 52. Brady, W. T.; Saidi, K. J. Org. Chem. 1980, 45, 727-729.
  - 53. Kron, J.; Schubert, U. J. Organomet. Chem. 1989, 373, 203-219.
  - 54. Rossé, G.; Gerber, F.; Specklin, J.-L.; Hubschwerlen, C. Synlett 2001, 538-540.
  - 55. Black, T. H.; Zhang, Y.; Huang, J.; Smith, D. C.; Yates, B. E. Synth. Commun. 1995, 25, 15–20.
  - 56. Concepcion, A. B.; Maruoka, K.; Yamamoto, H. Tetrahedron 1995, 51, 4011-4020.
  - 57. Pommier, A.; Pons, J.-M.; Kocienski, P. J. J. Org. Chem. 1995, 60, 7334-7339.
  - 58. Dymock, B. W.; Kocienski, P. J.; Pons, J.-M. J. Chem. Soc. Chem. Commun. 1996, 1053–1054.
  - 59. Dymock, B. W.; Kocienski, P. J.; Pons, J.-M. Synthesis 1998, 1655–1661.
  - 60. Yang, H. W.; Romo, D. Tetrahedron Lett. 1998, 39, 2877-2880.
  - 61. Forslund, R. E.; Cain, J.; Colyer, J.; Doyle, M. P. Adv. Synth. Catal. 2005, 347, 87-92.
  - 62. Takaoka, K.; Aoyama, T.; Shioiri, T. Synlett. 1994, 1005-1006.
  - Arai, S.; Sakurai, T.; Asakura, H.; Fuma, S.; Shioiri, T.; Aoyama, T. *Heterocycles* 2001, 55, 2283–2287.
  - 64. Kita, Y.; Tsuzuki, Y.; Kitagaki, S.; Akai, S. Chem. Pharm. Bull. 1994, 42, 233–236.
  - 65. Ito, M.; Shirakawa, E.; Takaya, H. Synlett. 1996, 635-636.
  - Akai, S.; Kitagaki, S.; Matsuda, S.; Tsuzuki, Y.; Naka, T.; Kita, Y. Chem. Pharm. Bull. 1997, 45, 1135–1139.
  - 67. Takaoka, K.; Aoyama, T.; Shiori, T. Tetrahedron Lett. 1996, 37, 4973–4976.
  - 68. Takaoka, K.; Aoyama, T.; Shiori, T. Tetrahedron Lett. 1996, 37, 4977-4978.

- 69. Takaoka, K.; Aoyama, T.; Shiori, T. Tetrahedron Lett. 1999, 40, 3017-3020.
- 70. Sung, K.; Tidwell, T. T. Organometallics 1997, 16, 78-85.
- 71. Sung, K.; Tidwell, T. T. J. Am. Chem. Soc, 1996, 118, 2768-2769.
- 72. Hudrlik, P. F.; Roberts, R. R.; Ma, D.; Hudrlik, A. M. Tetrahedron Lett. 1997, 38, 4029–4032.
- 73. Nikolaeva, S. N.; Ponomarev, S. V.; Petrosyan, V. S.; Lorberth, J. J. Organomet. Chem. 1997, 535, 213–215.
- Ponomarev, S. V.; Zolotareva, A. S.; Leont'ev, A. S.; Kuznetsov, Yu. V.; Petrosyan, V. S. Russ. Chem. Bull. Int. Ed. 2001, 50, 1088–1092.
- Ponomarev, S. V.; Zolotareva, A. S.; Ezhov, R. N.; Kuznetsov, Y. V.; Petrosyan, V. S. *Russ. Chem. Bull.* 2001, 50, 1093–1096.
- 76. Oblin, M.; Rajzmann, M.; Pons, J.-M. Tetrahedron 1997, 53, 8165-8172.
- 77. Rigby, J. H.; Wang, Z. Org. Lett. 2003, 5, 263-264.
- 78. Liu, F.; Liebeskind, L. S. J. Org. Chem. 1998, 63, 2835-2844.
- 79. Dalton, A. M.; Zhang, Y.; Davie, C. P.; Danheiser, R. L. Org. Lett. 2002, 4, 2465-2468.
- 80. Marsden, S. P.; Pang, W.-K. Chem. Commun. 1999, 1199-1200.
- Leigh, W. J.; Kerst, C.; Boukherroub, R.; Morkin, T. L.; Jenkins, S. I.; Sung, K.; Tidwell, T. T. J. Am. Chem. Soc. 1999, 121, 4744–4753.
- Morkin, T. L.; Leigh, W. J.; Tidwell, T. T. Allen, A. D. Organometallics 2001, 20, 5707– 5716.
- 83. Raponi, E.; Pons, J.-M. Tetrahedron Lett. 2003, 44, 9193-9196.
- 84. Brook, A. G.; Habtemariam, A. Can. J. Chem. 2003, 81, 1164-1167.
- 85. Iso, Y.; Shindo, H.; Hamana, H. Tetrahedron 2000, 56, 5353-5361.
- 86. Allen, A. D.; Tidwell, T. T. Can. J. Chem. 1999, 77, 802-805.
- Blake, A. J.; Friend, C. L.; Outram, R. J.; Simpkins, N. S.; Whitehead, A. J. *Tetrahedron Lett.* 2001, 42, 2877–2881.
- 88. Bennett, D. M.; Okamoto, I.; Danheiser, R. L. Org. Lett. 1999, 1, 641-644.
- 89. Yamamoto, Y.; Ozasa, N.; Sawada, S. Chem. Express 1993, 8, 305-308.

# 4.6 PHOSPHORUS-SUBSTITUTED KETENES

Phosphinyl-substituted ketenes are predicted to have substantial stabilization,^{1,2} as shown by isodesmic comparison based on MP2/6-31G*//MP2/6-31G* calculated energies (equation 1).² However, ketenes substituted with P(III) are rare. Among the first examples were **6** and **7**, prepared from reaction of alkynyl ethers with phosphinyl halides (equation 2).³ Ketene **6** was not stable even in solution, and on standing was converted to a complex mixture. However, the IR spectrum and the ³¹P NMR spectrum supported the assigned structure. Ketene **7** was quite stable and was characterized by IR, ³¹P NMR, and elemental analysis. Ketenes **9** and **10** were prepared similarly and are much more stable than **6** (equation 3).³

$$PH_{2} \qquad CH_{3} \qquad CH_{3} \qquad PH_{2} \qquad CH_{3} \qquad PH_{2} \qquad CH_{3} \qquad PH_{2} \qquad CH_{3} \qquad PH_{2} \qquad (1)$$

$$1 \qquad 2 \qquad AE = C=O + CH_{3} \qquad PH_{2} \qquad (1)$$



Ketenes substituted with P(V) include those with the Ph₂P(O) and (MeO)₂P(O) substituents formed by Wolff rearrangements (equations 4, 5).^{4–6} The kinetics of thermal decomposition of the diazo ketone **11** forming ketene **12** were measured, and **11** was found to be less reactive than **13** by a factor of 22 (equation 4).^{4,5} Ketene **12** was identified by the strong IR absorption at 2120 cm⁻¹ upon photolysis in a novolac resin.⁶ Photolysis of the diazo ketone **14** in hexane led to formation of the ketene **15**, as detected by its IR absorption at 2160 cm⁻¹, while photolysis in MeOH resulted in capture of the ketene, forming the symmetrical dimethyl malonate **16** in quantitative yield (equation 5).⁷



The phosphonate-substituted diazo ketone 17, upon photolysis in MeOH, resulted in the formation of the products 19 and 21 derived from hydrogen abstraction by an intermediate carbene 18 (equation 6) and Wolff rearrangement to the ketene 20, respectively (equation 7).⁸ When R was methyl the ratio of the products 19a and 21a was near 1:1, but when R was Na the Na salt 17 gave almost exclusively Wolff rearrangement to the ketene 20b.⁸ The ketene 20a was also generated in a matrix at 10 K by photolysis of 17a (R=Me) and observed by IR near 2120 cm⁻¹, and was also formed by carbonylation in a matrix of the carbene 23 generated by photolysis of the diazo ketone 22 (equation 8).⁹





Reactions of the phosphonate **24a** and the phosphine oxide **24b** in refluxing  $CH_2Cl_2$  containing rhodium(II) acetate gave mixtures of cyclized carbene insertion products **26** and ketenes **27**, as shown by reaction with ethanol forming **28** (equations 9, 10).¹⁰ The formation of ketenes in  $Rh_2(OAc)_4$ -catalyzed reactions was unusual, and was interpreted in terms of the low electron-withdrawing character of the phosphorus substituents diminishing the electrophilic character and the tendency for insertion of intermediate rhodium carbene species **25**.¹⁰



Thermal Wolff rearrangement of the diazo ketone **29** formed the phosphonatosubstituted ketenes **30**, which could be trapped by MeOH or allowed to undergo intramolecular benzannulation reactions providing a variety of heterocycles (equation 11).¹¹ Diazo ketone **32** with a pendant *tert*-amino substituent upon thermolysis formed ketene **33**, which gave rise to a mixture of 2-oxindolinium enolate **34** as well as the product **35** from capture of the carbene in a 51:49 ratio (equation 12).¹² The structure of **34** was determined by X-ray.¹² These reactions are also discussed in Section 3.3.1.





Thermal dehydrochlorination of the phosphoryl-substituted acyl chloride **36** by refluxing in CCl₄ provided a route to the diethylphosphonoketene **37** in 70% yield (equation 13).^{13–15} Phosphine **38** reacted with carbon tetrachloride to form the chloro-substituted phosphorus ylides **39**, which reacted with CO₂ to form the phosphoryl-substituted ketene **40**, characterized by IR and NMR and trapped with MeOH, forming the ester **41** (equation 14).^{14,15} The fluoro phosphorus ylide **42** reacted with CO₂ to form a fluorooxaphosphorane that on heating formed the ketene **43** (equation 15).^{16,17}



Heating of the vinyl ether **44** resulted in elimination of MeCl and formation of the ketene **45** in 92% yield, identified by the IR and NMR spectra (equation 16).¹⁸ Trimethylsilyl iodide reacted with the alkynyl ether **46** to give the thiophosphoryl-ketene **47** (equation 17).¹⁹

$$(i-Pr)_{2}P(S)CEt \xrightarrow{OMe} \xrightarrow{145 \circ C} \xrightarrow{(i-Pr)_{2}P'} C=0$$
(16)  
44 45 (IR 2100 cm⁻¹)

	01 30
4.2	1.0
	4.2

 TABLE 4.4
 Relative Rate Constants for Addition of 4-XC₆H₄OH to 48²⁰

 $t-Bu_2P(S)C \equiv COEt \xrightarrow{Me_3SiI} \begin{array}{c} t-Bu_2P' \\ \hline Me_3Si \end{array} C = C = O \\ Me_3Si \end{array} (17)$ 

Phosphoryl-substituted ketenes underwent typical additions of nucleophiles such as water, alcohols, and amines. Kinetics of the reaction of ketene **48** with substituted phenols in CCl₄ have been reported, and showed greater reactivity with substituted phenols for  $\pi$ -donors and electronegative substituents (Table 4.4). Ketene **48** was also 2000 times more reactive toward phenol than was Ph₂C=C=O,²⁰ suggesting a high degree of carbanion character that would be stabilized by the (EtO)₂P(O) substituent, so a mechanism involving a polar transition state **49** appeared to be involved (equation 18). This is consistent with the results of other studies of nucleophilic attack on ketenes (Section 5.5.1.2). The question of the preferred direction of in-plane attack on the ketene was unanswered.

$$\stackrel{O}{(\text{EtO})_2P'}_{Ph} C=O \xrightarrow{4 \cdot XC_6H_4OH}_{(ArOH)} \left[ \stackrel{O}{(\text{EtO})_2P'}_{Arod} \stackrel{O}{\rightarrow}_{H} \stackrel{O}{(\text{EtO})_2P'}_{Arod} \stackrel{O}{\rightarrow}_{Ph} \stackrel{O}{(\text{EtO})_2P'}_{OAr} \stackrel{O}{\rightarrow}_{Ph} \stackrel{O}{(\text{EtO})_2P'}_{OAr} \stackrel{O}{\rightarrow}_{Ph} \stackrel{O}{(\text{EtO})_{Ar}}_{OAr} \right]$$

$$\begin{array}{c} 48 \qquad 49 \qquad 50 (70\%, \text{Ar} = 4 \cdot \text{MeOC}_6H_4) \end{array}$$

Addition of the electrophilic NOCl to **51** formed by dehydrochlorination gave **52**, which was converted to the amide (equation 19).²¹ Ketene **51** also reacted by [2 + 2] cycloaddition with imines, forming  $\beta$ -lactams, and with cyclopentadiene formed **54**, which was assigned the configuration with the bulky phosphonate substituent in the *exo* position (equation 20).²² This stereochemistry was contrary to the usual result of ketene cycloadditions, in which the larger substituent occupied the more crowded *endo* position as a result of the perpendicular approach of the ketene to the reacting double bond, and formation of the bond to C₂ of the ketene before this can rotate to a less crowded conformation (Section 5.4.1.2). The formation of **54** may indicate that a long-lived zwitterionic intermediate **53** was formed that had the opportunity to form the thermodynamically most stable product (equation 20).





Ketenylidene phosphoranes (phosphaketene ylides) **56** are a unique family of phosphorus-substituted ketenes, which can be formed by elimination from the carbomethoxy phosphorane **55** (equation 21).^{23,24} The ketene **56** was a crystalline solid with a P-C₂-C₁ angle of 145.5°, as determined from the X-ray structure.²³ Alkoxyethynylphosphines **58** formed by reaction of phosphorus halides R₂PHal with lithium acetylides **57** rearranged thermally to ketenylidenephosphoranes **59** upon heating (equation 22).^{25,26}

$$Ph_{3}P = CHCO_{2}Me \xrightarrow{NaN[Si(Pr-i)_{3}]_{2}} Ph_{3}P = C = C = O \longrightarrow Ph_{3}\dot{P} - \bar{C} = C = O$$
(21)  

$$55 \qquad 56 (80\%, IR 2080 cm^{-1})$$

$$LiC \equiv COMe \xrightarrow{R_{2}PhHal} R_{2}PC \equiv COMe \xrightarrow{\Delta} MeR_{2}PC = C = O$$
(22)  

$$57 \qquad 58 \qquad 59 (90\%, R = t - Bu)$$

Synthetic applications of phosphaketene ylides included additions to carbonyl groups,²⁷ Wittig reactions,^{28–30} an addition-Wittig cyclization sequence of **56** converting benzoylpyrrole to **62** (equation 23),²⁹ and [4 + 2] cycloadditions with  $\alpha$ , $\beta$ -unsaturated carbonyl compounds (equation 24).³¹



Protonation of the ketenylidenephosphorane **56** with HCl gave the cationic species **64**, which immediately underwent cycloaddition with another molecule of **56** to form the cyclobutanedione **65** (equation 25).³² Oxidation of **65** with **66** gave the trione **67** (equation 25).³² Reaction of **67** with **56** yielded **68** (equation 26).³² The structures of **67** and **68** were determined by X-ray. The trimer **69** of **56** has also been prepared by reaction of **56** with HCl and has been described as a hybrid

between a ketone and an ylide.³³

56



Reaction of **56** with Pt(II) complexes gave products such as **70** (equation 27), and a large number of related ketene and bisketene analogues were obtained (see also Section 4.9),^{34–36} and their ¹³C and ³¹P NMR spectra were recorded.³⁷ Theoretical studies showed that the cycloaddition reactions of H₂P=C=C=O occurred at the C–P bond.³⁸ The reaction of **56** with H₂O yielded **72**, and this was proposed to take place by initial proton transfer forming **71**, followed by hydration and decarboxylation (equation 28).³⁹

$$Ph_{3}PC=C=O + [PtCl_{2}(PPh_{3})]_{2} \longrightarrow Ph_{3}P C=C=O$$

$$Ph_{3}PCl_{2}Pt C=C=O$$

$$Ph_{3}PCl_{2}Pt C=O \rightarrow Ph_{3}P C=C=O$$

$$Ph_{3}PC=C=O \longrightarrow Ph_{3}P C=C=O \rightarrow Ph_{3}P C=C=O$$

$$Ph_{3}PCH_{3}^{+}(HCO_{3})^{-} (28)$$

Cycloaddition of **56** with carbon suboxide led to the spiro tetraketoheptane derivative **73** (equation 29).⁴⁰ Reaction of **56** with esters bearing OH, NHR, or SH groups in the  $\alpha$ -,  $\beta$ -, or  $\gamma$ -position gave five-, six-, or seven-membered heterocycles (equation 30).^{41,42} Cyclization of **56** with **76** gave the spiro tetronate **77** (equation 31).⁴³

71



72 (73%)



A unique phosphorus-substituted bisketene **78** has been prepared and its structure established by X-ray (equation 32).⁴⁴



Arsenic-substituted ketenes have also been prepared.^{45–47} The acetylenic ether **79** reacted with Ph₂AsCl to give the bis(arsenic)-substituted ketene **80**, which was purified by distillation (equation 33).⁴⁵ The silyl alkynyl ether **81** reacted similarly, forming the silylated and arsenic-substituted ketene **82** (equation 34), and Ph₃GeCl reacted with the ethyl ether analogue of **79** to form Ph₂AsC (GePh₃)= C=O.⁴⁵

Ph₂AsC 
$$\equiv$$
 COMe  $\xrightarrow{Ph_2AsCl}$   $\xrightarrow{Ph_2As}$  C=O  
Ph₂As  $\xrightarrow{Ph_2As}$  C=O  
(33)  
79 80 (78%, IR 2078 cm⁻¹)

$$MePh_2SiC \equiv COMe \xrightarrow{Ph_2AsCl} Ph_2As \longrightarrow C=O$$

$$MePh_2Si \longrightarrow C=O$$

The tungsten complex **83** was converted to the ketene complex **84**, which reacted with  $Ph_3P$  to give the isolable free ketene **85** (equation 35).^{45,46}



Triphenylarsenylideneketene **87** was obtained as a crystalline solid by elimination from the ester **86** (equation 36) and reacted with methyl iodide to form the ketene **88**, which combined with **87** by [2+2] cycloaddition, forming **89** (equation 37).⁴⁸



# **REFERENCES FOR SECTION 4.6**

- 1. Gong, L.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1991, 113, 6021-6028.
- 2. McAllister, M. A.; Tidwell, T. T. J. Org. Chem. 1994, 59, 4506-4515.
- 3. Lukashev, N. V.; Artyushin, O. I.; Lazhko, É. V.; Luzikova, E. V.; Kazankova, M. A. Zh. Obshch. Khim. **1990**, 60, 1539–1549. J. Gen. Chem. (Engl. Transl.) 1374–1382.
- Regitz, M.; Anschütz, W.; Bartz, W.; Liedhegener, A. Tetrahedron Lett. 1968, 3171– 3174.
- 5. Regitz, M.; Bartz, W. Chem. Ber. 1970, 103, 1477-1485.
- Willson, C. G.; Miller, R. D.; McKean, D. R.; Pederson, L. A.; Regitz, M. Proc. SPIE-Int. Soc. Opt. Eng. 1987, 771, 2–10. Chem. Abstr. 1987, 107, 225797t.
- Polozov, A. M.; Pavlov, V. A.; Polezhaeva, N. A.; Liorber, B. G.; Tarzivolova, T. A.; Arbuzov, B. A. Zh. Obshch. Khim. 1986, 56, 1217–1220. J. Gen. Chem. (Engl. Transl.) 1986, 56, 1072–1074.
- 8. Tomioka, H.; Hirai, K. J. Chem. Soc. Chem. Commun. 1990, 1611-1613.
- 9. Tomioka, H.; Komatsu, K.; Shimizu, M. J. Org. Chem. 1992, 57, 6216-6222.
- Corbel, B.; Hernot, D.; Haelters, J.-P.; Sturtz, G. Tetrahedron Lett. 1987, 28, 6605–6608.

- 11. Chen, Y. P.; Chantegrel, B.; Deshayes, S. Heterocycles 1995, 41, 175-186.
- 12. Léost, F.; Chantegrel, B.; Deshayes, S. Tetrahedron 1997, 53, 7557-7576.
- Kolodiazhnyi, O. I. In Synth. Methods Organomet. Inorg. Chem. Karsch, H. H., Ed.; Thieme, Stuttgart, 1996, Vol. 3, pp. 96–97.
- Kolodiazhnyi, O. I. In Synth. Methods Organomet. Inorg. Chem. Karsch, H. H., Ed.; Thieme, Stuttgart, 1996, Vol. 3, pp. 94–95.
- 15. Kolodyazhnyi, O. I. Tetrahedron Lett. 21, 1980, 3983-3986.
- 16. Kolodiazhnyi, O. I.; Schmutzler, R. Synlett 2001, 1065-1078.
- Kolodyazhnyi, O. I. Zh. Obshch. Khim. 1987, 57, 821–827. J. Gen. Chem. (Engl. Transl.) 1987, 57, 724–730.
- Trostyanskaya, I. G.; Kazankova, M. A.; Beletskaya, I. P. Russ. J. Org. Chem. 1995, 31, 601–604.
- Lukashev, N. V.; Fil'chikov, A. A.; Luzikov, Yu. N.; Kazankova, M. A. Zh. Obshch. Khim. 1990, 60, 1672–1674. Engl. Trans. 1990, 60, 1492–1493.
- Vdovenko, S. I.; Yakovlev, V. I.; Kolodyazhnyi, O. I.; Kukhar, V. P. Zh. Obshch. Khim. 1982, 52, 2223–2227; J. Gen. Chem. (Engl. Transl.) 1982, 52, 1978–1981.
- Kashemirov, B. A.; Skoblikova, L. I.; Khokhlov, P. S. Zh. Obshch. Khim. 1988, 58, 1672– 1673; J. Gen. Chem. (Engl. Transl.) 1988, 58, 1492–1493.
- 22. Motoyoshiya, J.; Hirata, K. Chem. Lett. 1988, 211-214.
- 23. Bestmann, H. J. Angew. Chem. Int. Ed. Engl. 1977, 16, 349-364.
- 24. Bestmann, H. J.; Sandmeier, D. Chem. Ber. 1980, 113, 274-277.
- Lukashev, N. V.; Artyushin, O. I.; Luzikov, Yu. N.; Kazankova, M. A.; Luchenko, I. F. Zh. Obshch. Khim. 1987, 57, 2491–2499. J. Gen. Chem. (Engl. Transl.) 1987, 57, 2218– 2225.
- Lukashev, N. V.; Fil'chikov, A. A.; Kozlov, A. I.; Luzikov, Yu. N.; Kazankova, M. A. Zh. Obshch. Khim. 1991, 61, 1739–1743. J. Gen. Chem. (Engl. Transl.) 1991, 61, 1600– 1603.
- 27. Bestmann, H. J.; Schmid, G.; Sandmeier, D. Chem. Ber. 1980, 113, 912-918.
- 28. Bestmann, H. J.; Schobert, R. Angew. Chem. Int. Ed. Engl. 1983, 22, 780-782.
- 29. Nickisch, K.; Klose, W.; Nordhoff, E.; Bohlmann, F. Chem. Ber. 1980, 113, 3086–3088.
- 30. Klose, W.; Nickisch, K.; Bohlmann, F. D. Chem. Ber. 1980, 113, 2694-2698.
- 31. Bestmann, H. J.; Schmid, G. Tetrahedron Lett. 1984, 25, 1441-1444.
- 32. Bestmann, H. J.; Fürst, T. G.; Schier, A. Angew. Chem. Int. Ed. Engl. 1993, 32, 1746–1747.
- Bestmann, H. J.; Fürst, T. G.; Schier, A. Angew. Chem., Int. Ed. Engl. 1993, 32, 1747– 1749.
- 34. Bertani, R.; Meneghetti, F.; Pandolfo, L.; Scarmagnan, A.; Zanotto, L. J. Organomet. Chem. 1999, 583, 146–151.
- 35. Bertani, R.; Casarin, M.; Pandolfo, L. Coord. Chem. Rev. 2003, 236, 15-33.
- 36. Schobert, R.; Gordon, G. J. Curr. Org. Chem. 2002, 6, 1181-1196.
- 37. Pandolfo, L.; Sassi, A.; Zanotto, L. Inorg. Chem. Commun. 2001, 4, 145-149.
- 38. Salzner, U.; Bachrach, S. M. J. Organomet. Chem. 1997, 529, 15-22.
- 39. Benetollo, F.; Bertani, R.; Ganis, P.; Pace, G.; Pandolfo, L.; Zanotto, L. J. Organomet. Chem. 2002, 642, 64–70.

- 40. Pandolfo, L.; Facchin, G.; Bertani, R.; Ganis, P.; Valle, G. Angew. Chem. Int. Ed. Engl. 1996, 35, 83–85.
- 41. Löffler, J.; Schobert, R. J. Chem. Soc. Perkin Trans. 1 1996, 2799-2802.
- 42. Löffer, J.; Schobert, R. Liebigs Ann./Recueil 1997, 217-220.
- 43. Rath, J.-P.; Eipert, M.; Kinast, S.; Maier, M. E. Synlett 2005, 314-318.
- 44. Byrne, L. T.; Johnson, J. A.; Koutsantonis, G. A.; Skelton, B. W.; White, A. H. Chem. Commun. 1997, 391–392.
- 45. Himbert, G.; Henn, L. Tetrahedron Lett. 1984, 25, 1357-1358.
- Kreissl, F. R.; Wolfgruber, M.; Sieber, W. Angew. Chemie Int. Ed. Engl. 1983, 22, 1001– 1002.
- 47. Wolfgruber, M.; Sieber, W.; Kreissl, F. R. Chem. Ber. 1984, 117, 427-433.
- 48. Bestmann, H. J.; Bansal, R. K. Tetrahedron Lett. 1981, 22, 3839-3842.

# 4.7 SULFUR-SUBSTITUTED KETENES

Mercaptoketene (1) was predicted by an isodesmic exchange reaction based on MP2/6-31G*//MP2/6-31G* calculated energies to be stabilized to the same extent as methylketene 3 (equation 1).^{1,2} There has been relatively little study of this family, but there is considerable evidence that these ketenes are relatively easy to form and have appreciable stability. There is a rich chemistry of sulfur-substituted compounds, and further study of these ketenes appears warranted.

Methylthioketene **6** was prepared by Wolff rearrangement upon direct photolysis of *S*-methyl diazothioacetate **5** in methanol with 310–380 nm light, resulting in the formation of the methyl ester **7** in 86% yield as the only observable volatile product (equation 2).³ The ketene EtSCH=C=O (**8**) was also prepared by photochemical Wolff rearrangement and trapped with alcohols.⁴ These results are in contrast to those obtained with photolysis of  $\alpha$ -diazo esters and amides, which gave only 20–60% of Wolff rearrangement in solution,⁴ and indicate a high migratory aptitude for the thiol group in Wolff rearrangement. The failure to trap **6** from the corresponding sensitized reaction was taken as evidence that Wolff rearrangement did not occur from a triplet carbene.³

$$CH_{3S} \xrightarrow{O} N_{2} \xrightarrow{hv} CH_{3S} \xrightarrow{CH_{3}O} C=O \xrightarrow{CH_{3}OH} CH_{3S} \xrightarrow{O} OCH_{3}$$

$$5 \qquad 6 \qquad 7 (86\%)$$

$$(2)$$

R	$k(rel)(H^+)$	$k(rel)(H_2O)$	$k(rel)(OH^{-})$
ОН	$1.2 \times 10^7$	$5.4  imes 10^3$	$1.2 \times 10^5$
SCH ₃	1.0	1.0	1.0
CH ₃	—	0.55	1.1

TABLE 4.5 Relative Hydration Reactivitities of PhCR=C=O in H₂O at 25 °C (data from Section 5.5.1)⁵

Photolysis of methyl phenyldiazoacetate **8** in water formed (methylthio)phenylketene (**9**), which formed the acid **10** as the only detectable product by HPLC analysis (equation 3).⁵ The kinetics of hydration of **9** were measured,⁵ and the reactivity was found to be much less than that of PhC(OH)=C=O, and about the same as that of PhCMe=C=O, in acid-, neutral, and base-induced reactions (Section 5.5.1, Table 4.5).



Photolysis of the diazomalonate derivative **11** gave the methylthio-substituted ketene **12**, as characterized by the IR absorption at  $2120 \text{ cm}^{-1}$  in CDCl₃, as well as by ¹H NMR (equation 3a).⁶ No detectable product from methoxy group migration was observed, showing the much greater propensity for thio group migration, and the ketene was also trapped as the corresponding  $\beta$ -lactam **13** upon [2 + 2] cycloaddition with the imine PhN=CHPh (equation 3a).⁶ Thermolysis of the diazo ketone **14** formed the ketene **15**, detected by photoelectron spectroscopy, which upon decarbonylation and hydrogen abstraction gave benzothiete **16** (equation 4).⁷



The  $Rh_2(OAc)_4$ -catalyzed reaction of  $\alpha$ -diazo thiol esters 17 provided a general route to thio-substituted ketenes 18 that have been used in cycloadditions with

alkenes, alkynes, and imines (equation 5).8,9



Ketenes RSCH=C=O (21) generated by dehydrochlorination of the acyl chlorides (20) reacted with imines 22 to give observable azadienes 23 that upon thermolysis form  $\beta$ -lactams 24 (equation 6).¹⁰ For R¹ = PhCH=CH a six-membered ring dihydropyridone was formed from reaction of 21 and 22.¹⁰ Dehydrochlorination of 25 gave ketene 26, which was trapped by imines or with cyclopentadiene (equation 7).¹¹



ketenes **30** with two sulfur substituents, and these were sufficiently stable in MeOH that this was used as a solvent to measure their UV spectra (equation 8).¹² Dehydrochlorination of **31** gave ketene **32**, which was trapped in situ by [2 + 2] cycloaddition with cyclopentadiene (equation 9)^{11,13} or with imines.¹⁴



Ketene 32 also gave [2+2] cycloaddition with the silyl enol ether 34 and formed 35 (equation 10).¹⁵ Thermolysis of 36 formed ketene 32, which reacted

with the alkene **37**, forming **38** (equation 11).¹⁶



Wolff rearrangement of the diazo ketone 39 gave the phenylsulfonylketene 40 (equation 12).^{17, $\tilde{18}$} A series of isolable SF₅-substituted ketenes **41** were prepared by dehydration of the carboxylic acids and dimerized upon heating (equation 13).¹⁹



Dehydration of 43 gave 44 with strong UV absorption at 220 and 325 nm, and upon photolysis **44** was proposed to give the carbene **45** (equation 14).^{20,21} The selenyl analogue 46 was obtained by dehydration in 45% yield and was also prepared from  $Ag_2C=C=O$  (equation 15).²² The ketene 47 was obtained by the unique method of CrO₃ oxidation of 44 (equation 16).²²

$$\begin{array}{cccc} CF_{3}S & O & \xrightarrow{P_{2}O_{5}} & CF_{3}S \\ CF_{3}S & OH & & CF_{3}S \\ \end{array} \xrightarrow{cF_{3}S} & CF_{3}S & C=O & \xrightarrow{hv} & CF_{3}S \\ \hline & & CF_{3}S & CF_{3}S \\ \end{array} \xrightarrow{cF_{3}S} C=O & \xrightarrow{hv} & CF_{3}S \\ \end{array} \xrightarrow{cF_{3}S} C \xrightarrow{cF_{3}} C \xrightarrow{cF_{3}S} C \xrightarrow{cF_{3}}$$

**44** (78%, IR 2141cm⁻¹)

$$Ag_2C=C=O \xrightarrow{CF_3Secl} CF_3Se \xrightarrow{CF_3Se} C=O$$
(15)

**46** (45%, IR 2127 cm⁻¹)

$$\begin{array}{ccc} CF_{3}S \\ \hline CF_{3}S \end{array} \xrightarrow{C=0} & \xrightarrow{CrO_{3}} & \xrightarrow{C_{6}F_{5}SO_{2}} \\ H^{+} & \xrightarrow{C_{6}F_{5}SO_{2}} \\ \end{array} \xrightarrow{C=0} (16)$$

$$\begin{array}{ccc} 44 \\ 47 (5\%, IR \ 2139 \ cm^{-1}) \end{array}$$

The structure of the ketene  $(CF_3S)_2C=C=O$  (44) has been determined by gas phase electron diffraction and was found to have C₂ symmetry, with the S-CF₃ bonds perpendicular above and below the molecular plane,²³ as previously predicted by *ab initio* calculations.¹ This structure was in agreement with the interpretation that there is a repulsion between substituent lone pairs and the electron-rich ketenyl HOMO perpendicular to the ketene plane at C₂, so that the favored conformation minimizes this interaction.

Ketene **49** was formed either from  $P_4O_{10}$  dehydration of the carboxylic acid **48** (equation 17) or by reaction of  $C_6F_5SCl$  with  $Ag_2C=C=O.^{24}$  This ketene was a solid with a typical ketenyl IR band at 2135 cm⁻¹ and ¹³C NMR signals at  $\delta$  172.6 and 31.4 for  $C_1$  and  $C_2$  of the ketenyl group.



Reaction of the carbene complex **50** with thioethanol formed the unstable complex **50a**, which at 20 °C gave the ketene **51**, crystallized at -30 °C in 12% yield as a rather unstable material, characterized by the IR absorption at 2093 cm⁻¹ and by ¹H NMR (equation 18).^{25,26} The structure of the silyl-stabilized ketene **51** was determined by X-ray.²⁵



Heating of the mesoionic dithiole **52** was proposed to give rise to the unobserved ketene **53**, which was captured by PhNCO as the [2 + 2] cycloaddition product **54** (equation 19).²⁷



Dehydrochlorination of the seleno-substituted acyl chloride **55** with Et₃N in the presence of a phosphorus ylide gave ketene **56** that was converted to allene **57** by Wittig reaction (equation 20).²⁸ The phenylselenylketene **56** reacted with an imine by [2 + 2] cycloaddition (equation 21).²⁹



#### **REFERENCES FOR SECTION 4.7**

- 1. Gong, L.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1991, 113, 6021-6028.
- 2. McAllister, M. A.; Tidwell, T. T. J. Org. Chem. 1994, 59, 4506-4515.
- 3. Hixson, S. S.; Hixson, S. H. J. Org. Chem. 1972, 37, 1279-1280.
- Orphanides, G. G., Ph. D. Thesis, Ohio State University, 1972, quoted by Ando, W., *The Chemistry of Diazonium and Diazo Groups*; Patai, S., Ed; Wiley, New York, 1978, p. 473.
- 5. Jones, J., Jr.; Kresge, A. J. J. Org. Chem. 1992, 57, 6467-6469.
- 6. Georgian, V.; Boyer, S. K.; Edwards, B. J. Org. Chem. 1980, 45, 1686-1688.
- 7. Schulz, R.; Schweig, A. Tetrahedron Lett. 1980, 21 343-346.
- 8. Lawlor, M. D.; Lee, T. W.; Danheiser, R. L. J. Org. Chem. 2000, 65, 4375-4384.
- 9. Danheiser, R. L.; Okamoto, I.; Lawlor, M. D.; Lee, T. W. Org. Synth. 2003, 80, 160-171.
- Panunzio, M.; Bongini, A.; Tamanini, E.; Campana, E.; Martelli, G.; Vincennati, P.; Zanardi, I. *Tetrahedron* 2003, *59*, 9577–9582.
- Michel, P.; O'Donnell, M.; Biname, R.; Hesbain-Frisque, A. M.; Ghosez, L.; Declercq, J.P.; Germain, G.; Arte, E.; Van Meerssche, M. *Tetrahedron Lett.* **1980**, *21*, 2577–2580.
- 12. Inoue, S.; Hori, T. Bull. Chem. Soc. Jpn. 1983, 56, 171-174.
- 13. Cossement, E.; Binamé, R.; Ghosez, L. Tetrahedron Lett. 1974, 997-1000.
- 14. Bellus, D. Helv. Chim. Acta 1975, 58, 2509-2511.
- 15. Pak, C. S.; Kim, S. K. Bull. Korean Chem. Soc. 1994, 15, 337-338.
- Cagnon, J. R.; Bideau, F. L.; Marchand-Brynaert, J.; Ghosez, L. *Tetrahedron Lett.* 1997, 38, 2291–2294.
- 17. Regitz, M.; Bartz, W. Chem. Ber. 1970, 103, 1477-1485.
- 18. Capuano, L.; Mörsdorf, P.; Scheidt, H. Chem. Ber. 1983, 116, 741-750.
- 19. Bittner, J.; Seppelt, K. Chem. Ber. 1990, 123, 2187-2190.
- 20. Haas, A.; Lieb, M.; Praas, H.-W. J. Fluorine Chem. 1989, 44, 329-337.
- 21. Dorra, M.; Haas, A. J. Fluorine Chem. 1994, 66, 91-94.
- 22. Haas, A.; Praas, H.-W. Chem. Ber. 1992, 125, 571-579.
- Korn, M.; Mack, H.-G.; Praas, H.-W.; Vedova, C. O. D.; Oberhammer, H. J. Mol. Struct. 1995, 352/353, 145–151.

24. Haas, A.; Praas, H.-W.; Radau, G. J. Fluorine Chem. 1994, 67, 257-260.

- 25. Hörnig, H.; Walther, E.; Schubert, U. Organometallics 1985, 4, 1905–1906.
- 26. Schubert, U.; Kron, J.; Hörnig, H. J. Organomet. Chem. 1988, 355, 243-256.
- 27. Bssaibis, M.; Robert, A.; Souizi, A. J. Chem. Soc. Chem. Commun. 1993, 998-999.
- 28. Silveira, C. C.; Boeck, P.; Braga, A. L. Tetrahedron Lett. 2000, 41, 1867-1869.
- 29. Agawa, T.; Ishida, M.; Ohshiro, Y. Synthesis 1980, 933-935.

## 4.8 METAL-SUBSTITUTED KETENES

#### 4.8.1 Ynols and Ynolates

Ethynol 1 is a valence tautomer of ketene (2), and is the parent of the ynol, or hydroxyacetylene, family. Ynolates such as 3 are conjugate bases of both ketenes and ynols. Ethynol 1 has been generated in the gas phase by decarbonylation and neutralization of the radical cation of propiolic acid  $\text{HC}\equiv\text{CCO}_2\text{H}$  using tandem mass spectrometry.¹ Photolytic ring opening of hydroxycyclobutenedione 4 to the bisketene 5 in an argon matrix at 12 K and subsequent photochemical decarbonylation also formed 1, which was identified by its IR spectrum, and formed 2 upon further photolysis (equation 1).²



The identity reaction of proton exchange between ketene and the parent ynolate ion has been studied computationally and has a barrier of only 12.3 kcal/mol at the B3LYP level (equation 2).^{3,4} This indicated effective conjugation in the transition state. Computational studies predicted ethynol to be remarkably acidic,⁵ and the ethynolate ion **3** was readily produced in the gas phase by the collisional activation cleavage of enolates **7**⁶ and by proton abstraction from ethylene oxide **8** (equation 3).⁷ The gas phase acidity of ketene has been measured in the gas phase using the flowing afterglow method.⁸

$$\begin{array}{cccc}
& & & & & \\ & & & & \\ & & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\$$

Computations at the MP2/G-31G*//HF/6-31G* level indicated that in the absence of solvent the C-lithiated structure CHLi=C=O would be more stable than the O-lithiated isomer HC=COLi by 4 kcal/mol.⁹ For Li₂C₂O₂ the bis(ynolate) structure LiOC=COLi **9a** and the bicyclic structure **9b** were calculated to be of equal stability.¹⁰ "Potassium dicarbonyl," **10**, is a well-studies species for which the structure KOC=COK has been proposed.^{11,12}

Lio 
$$-$$
 OLi  $\stackrel{\text{Li}}{\circ} - \stackrel{\text{O}}{=} \stackrel{\text{O}}{=} \stackrel{\text{C}}{\underset{\text{Li}}{}}$  KO  $-$  OK  
9a 9b 10

Photolysis of arylhydroxycyclopropenones provided a general route to arylethynols,¹³ and irradiation of **11** gave phenylethynol (**12**), which was converted to phenylketene **13**. The kinetics of this conversion have been measured by TRIR spectroscopy (equation 4).^{13a} Further studies of this reaction in aqueous solution using UV spectroscopy have also been carried out.^{14,15}

$$\begin{array}{ccc} & & & \\ & & & \\ & & & \\ Ph & & \\ & & \\ 11 & & 12 & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & &$$

The chemistry of ynolate anions has been reviewed¹⁶ and these species have been generated by a number of routes, including the reaction of isoxazole 14 with *n*-BuLi, forming phenylethynolate 15, which reacted with Me₃SiCl, forming ketene 16 (equation 5).¹⁷ Reaction of two molecules of 15 with *t*-BuCOCl proceeded through ketene 17, forming pyrone 18 (equation 6).¹⁸ The *tert*-butyl ynolate 19 reacted with benzoyl chloride to form the isolable acyl ketene 20 (equation 7).¹⁸



*t*-Bu OLi 
$$\xrightarrow{PhCOCl}$$
  $\xrightarrow{Ph} \xrightarrow{O} C=O$  (7)  
**19 20** (53%, IR 2103 cm⁻¹)

Deprotonation of  $\alpha$ -haloenolate **21** led to dianion **22**, which formed ynolate **15** (equation 8), and this reacted with cyclohexanone to give  $\beta$ -lactones **23** (equation 9).¹⁹ Esters reacted with CH₂Br₂ and strong bases through an optimized route to form the ynolate **15**, and upon reaction with ethanol this gave the homologated ester **24** (equation 10).^{20–22}



The ynolate **25** formed from the procedure of equation 10 was suggested to combine with *n*-BuBr (used to prepare n-BuLi) to form the ketene **26**, which reacted with LiOEt (from the starting ester) to give **27** as a by-product after hydrolysis (equation 11).²² The use of *s*-BuLi avoided this problem.



Ynolates prepared by the Kowalski procedures were used to prepare a variety of homologated amino acid esters, without significant racemization, as in the conversion of **28** to **30** (equation 12).²³



40

Ynolates **31** were also formed by reaction of alkynyl tosylates or of triisopropylsilyl or alkynyl *tert*-butyldimethylsilyl ethers with MeLi, and further reaction with silyl chlorides proceeded with *O*-silylation, forming alkynyl silyl ethers **32** (equation 13).^{24,25} This is in contrast to the result of equation 5, where *C*-silylation occurred.

The reaction of  $OH^-$  with acyl chlorides **33** in negative ion mass spectroscopy formed both ketenes **34** and ynolates **35** by E2-type eliminations (equation 14).²⁶ Negative ion Wolff rearrangement of deprotonated diazo ketones **36** has been used to generate aryl ynolates in the gas phase (equation 15).²⁷

$$CH_{3}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{2}CH_{$$

The *n*-butyl ynolate anion  $39^{28-40}$  has been generated from ester dianions formed from the  $\alpha$ -bromo ester **38** (equation 16)²⁸ and from the  $\alpha,\alpha$ -dibromo ester **40** (equation 17).^{29,30} Capture of **39** with TIPSCI gave the alkynyl silyl ether in 60% yield.²⁸ The ynolate reacted by [2 + 2] cycloaddition with benzaldehyde, forming  $\beta$ -lactone **41** (equation 17), and with imine **42**, forming  $\beta$ -lactam **43** after protonation (equation 18).³⁵

$$\begin{array}{cccc} n-\mathrm{Bu} & O & & Ph & OH \\ \mathrm{Br} & & & & \\ \mathrm{Br} & & & OEt \end{array} & \begin{array}{c} r-\mathrm{BuLi} & & n-\mathrm{BuC} \equiv \mathrm{COLi} & \begin{array}{c} 1) \operatorname{PhCH=O} & & n-\mathrm{Bu} \\ \hline 2) \operatorname{H}^{+} & & & Ph & O \end{array} & (17)$$

41 (69%)

39

$$n-BuC \equiv COLi + PhCH=NSO_2 Tol \longrightarrow \qquad \begin{array}{c} n-Bu & O \\ Ph & SO_2 Tol \\ 39 & 42 & 43 (58\%) \end{array}$$
(18)

Ynolates were also generated by reaction of  $\alpha, \alpha$ -dibromoesters with lithium naphthalenide.³¹ The ynolates react with silylating agents to give silyl ynol ethers²⁸ or silylketenes and with aldehydes to give  $\beta$ -lactones (equation 17)²⁸ or acrylic acids.^{28,31} Reactions with ketones give stereoselective formation of trisubstituted acrylates,³⁰ while imines give  $\beta$ -lactams (equation 18)³⁵ and acrylamides,³⁶ in reactions that proceed through  $\beta$ -lactam enolates.

Keto ester **44** reacted with ynolate **39** through a Dieckmann-type reaction (equations 19, 20).^{32,36} This reaction has also been utilized for cyclopentenone formation.^{32,36,40} Reaction of ketones with ynolate **39** formed acrylic acids **49** proposed to form through  $\beta$ -lactone enolates **48** (equation 21).³⁰ The stereoselectivity of ring opening in this reaction was studied computationally.^{34a}



Ynolate **39** reacted with nitrones by a 1,3-dipolar cycloaddition forming 5-isoxazolidinones **50**, which were converted to  $\beta$ -amino acids **51** (equation 22).³⁸ Reaction of **39** with acylsilanes gave selective formation of *Z*-alkenes **52** (equation 23).³⁹



Phenyl ynolate **15** was generated by reaction of lithium phenylacetylide with *t*-BuOOLi, as evidenced by capture of the ketene dimer and PhCH₂CO₂Et on reaction with EtOH and capture as the silyl ethynyl ether **53** (equation 24).^{41,42}

$$PhC \equiv CLi + t - BuOOLi \longrightarrow PhC \equiv COLi \xrightarrow{i - Pr_3SiCl} PhC \equiv COSi(Pr - i)_3$$
(24)  
15 53 (60%)

In another route to ynolates, methyl benzoate was converted to the  $\alpha$ -chloro  $\alpha$ -sulfinyl ketone **54**, which reacted with potassium hydride followed by *tert*-butyllithium to form the phenyl ynolate **15**, which gave methyl phenylacetate on reaction with methanol (equation 25).^{43,44}

$$PhCO_{2}CH_{3} \xrightarrow{PhS(O)CH(Li)Cl} PhC \xrightarrow{O} \begin{array}{c} & 1 \\ PhC \\ & 54 \end{array} \xrightarrow{CHCIS(O)Ph} PhC \equiv COLi \xrightarrow{CH_{3}OH} PhCH_{2}CO_{2}CH_{3} \\ & 54 \\ & 84\% \end{array} \xrightarrow{CHCIS(O)Ph} \begin{array}{c} 15 \\ & 79\% \end{array}$$

$$(25)$$

Bis(trialkylsilyl)ketenes **55** were desilylated by treatment with *t*-BuOK in the presence of HMPA, forming potassiuum ynolates **56**, and these could be converted to other bis(trialkylsilyl)ketenes **57** (equation 26).⁴⁵ (See also Section 4.5.)

$$\begin{array}{ccc} R_{3}^{1}Si \\ R_{3}Si \\ R_{3}Si \end{array} \xrightarrow{t\text{-BuOK}} R_{3}SiC \equiv COK \xrightarrow{R_{3}^{2}SiX} R_{3}^{2}Si \\ \hline S5 & 56 & S7 \end{array} \xrightarrow{rag} C=0$$
(26)

Reaction of diethyl 1-hexynyl phosphate (**58**) with the enzyme phosphotriesterase resulted in inactivation of the enzyme. This was attributed to formation of *n*butylketene **59**, which acylated the enzyme (equation 27).⁴⁶



Generation of silylketenes from silyl ynolates is discussed in Sections 4.5 and 4.10.

#### **REFERENCES FOR SECTION 4.8.1**

- 1. von Baar, B.; Weiske, T.; Terlouw, J. K.; Schwarz, H. Angew. Chem. Int. Ed. Engl. 1986, 25, 282–284.
- 2. Hochstrasser, R.; Wirz, J. Angew. Chem. Int. Ed. Engl. 1989, 28, 181-183.

- 3. Bernasconi, C.; Wenzel, P. J. J. Am. Chem. Soc. 2001, 123, 2430-2431.
- 4. Bernasconi, C.; Wenzel, P. J. J. Am. Chem. Soc. 2001, 123, 7146-7153.
- 5. Smith, B. J.; Radom, L.; Kresge, A. J. J. Am. Chem. Soc. 1989, 111, 8297-8999.
- Bowie, J. H.; Stringer, M. B.; Currie, G. J. J. Chem. Soc. Perkin Trans. 2, 1986, 1821– 1825.
- 7. Lee, J.; Grabowski, J. J. Chem. Rev. 1992, 92, 1611-1647.
- Oakes, J. M.; Jones, M. E.; Bierbaum, V. M.; Ellison, G. B. J. Phys. Chem. 1983, 87, 4810–4815.
- 9. Gong, L.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1991, 113, 6021-6028.
- 10. Cioslowski, J. J. Am. Chem. Soc. 1990, 112, 6536-6538.
- 11. Büchner, W. Helv. Chim. Acta 1963, 46, 2111-2120.
- 12. Serratosa, F. Acc. Chem. Res. 1983, 16, 170-176.
- Chiang, Y.; Kresge, A. J.; Hochstrasser, R.; Wirz, J. J. Am. Chem. Soc. 1989, 111, 2355–2357.
- 13a. Wagner, B. D.; Zgierski, M. Z.; Lusztyk, J. J. Am. Chem. Soc. 1994, 116, 6433-6434.
- 14. Chiang, Y.; Kresge, A. J.; Popik, V. V. J. Am. Chem. Soc. 1995, 117, 9165-1971.
- 15. Chiang, Y.; Kresge, A. J.; Paine, S. W.; Popik, V. V. J. Phys. Org. Chem. 1996, 9, 361-370.
- 16. Shindo, M. Chem. Soc. Revs. 1998, 27, 367-374.
- 17. Hoppe, I.; Schöllkopf, U. Liebigs Ann. 1979, 219-226.
- 18. Zhdankin, V. K.; Stang, P. J. Tetrahedron Lett. 1993, 34, 1461-1462.
- 19. Kowalski, C. J.; Fields, K. W. J. Am. Chem. Soc. 1982, 104, 321-323.
- 20. Kowalski, C. J.; Haque, M. S.; Fields, K. W. J. Am. Chem. Soc. 1985, 107, 1429-1430.
- 21. Kowalski, C. J.; Reddy, R. E. J. Org. Chem. 1992, 57, 7194-7208.
- 22. Reddy, R. E.; Kowalski, C. J. Org. Synth. 1992, 71, 146-157.
- 23. Gray, D.; Concellón, C.; Gallagher, T. J. Org. Chem. 2004, 69, 4849-4851.
- 24. Stang, P. J.; Roberts, K. A. J. Am. Chem. Soc. 1986, 108, 7125-7127.
- 25. Kowalski, C. J.; Lal, G. S.; Haque, M. S. J. Am. Chem. Soc. 1986, 108, 7127-7128.
- 26. Lloyd, J. R.; Agosta, W. C.; Field, F. H. J. Org. Chem. 1980, 45, 1614-1619.
- Lebedev, A. T.; Hayes, R. N.; Bowie, J. H. J. Chem. Soc. Perkin Trans. 2, 1991, 1127– 1129.
- 28. Shindo, M. Tetrahedron Lett. 1997, 38, 4433-4436.
- 29. Shindo, M.; Sato, Y.; Shishido, K. Tetrahedron 1998, 54, 2411-2422.
- 30. Shindo, M.; Sato, Y.; Shishido, K. Tetrahedron Lett. 1998, 39, 4857-4860.
- Shindo, M.; Koretsune, R.; Yokota, W.; Itoh, K.; Shishido, K. *Tetrahedron* 2001, 42, 8357–8360.
- 32. Shindo, M.; Sato, Y.; Shishido, K. J. Am. Chem. Soc. 1999, 121, 6507-6508.
- 33. Shindo, M.; Sato, Y.; Shishido, K. J. Org. Chem. 2000, 65, 5443-5445.
- Shindo, M.; Sato, Y.; Yoshikawa, T.; Koretsune, R.; Shishido, K. J. Org. Chem. 2004, 69, 3912–3916.
- 34a. Mori, S.; Shindo, M. Org. Lett. 2004, 6, 3945-3948.
- 35. Shindo, M.; Oya, S.; Murakami, R.; Sato, Y.; Shishido, K. *Tetrahedron Lett.* **2000**, *41*, 5943–5946.

- 36. Shindo, M.; Oya, S.; Murakami, R.; Sato,Y.; Shishido, K. *Tetrahedron Lett.* **2000**, *41*, 5947–5950.
- 37. Shindo, M.; Sato, Y.; Shishido, K. J. Org. Chem. 2001, 6, 7818-7824.
- 38. Shindo, M.; Itoh, K.; Tsuchiya, C.; Shishido, K. Org. Lett. 2002, 4, 3119-3121.
- Shindo, M.; Matsumoto, K.; Mori, S.; Shishido, K. J. Am. Chem. Soc. 2002, 124, 6840– 6841.
- 40. Shindo, M.; Sato, Y.; Shishido, K. Tetrahedron Lett. 2002, 43, 5039-5041.
- 41. Julia, M.; Jalmes, V. P.-S.; Plé, K.; Verpeaux, J.-N.; Hollinsworth, G. Bull. Soc. Chim. Fr. 1996, 133, 15–24.
- 42. Julia, M.; Jalmes, V. P.-S.; Verpeaux, J.-N. Synlett 1993, 233-234.
- 43. Satoh, T.; Mizu, Y.; Hayashi, Y.; Yamakawa, K. Tetrahedron Lett. 1994, 35, 133– 134.
- 44. Satoh, T.; Unno, H.; Mizu, Y.; Hayashi, Y. Tetrahedron 1997, 53, 7843-7854.
- 45. Ito, M.; Shirakawa, E.; Takaya, H. Synlett 2002, 45, 1329-1331.
- 46. Blankenship, J. N.; Abu-Soud, H.; Francisco, W. A.; Raushel, F. M.; Fischer, D. R.; Stang, P. J. J. Am. Chem. Soc. **1991**, 113, 8560–8561.

## 4.8.2 Boron-Substituted Ketenes

Computations at the MP2/6-31G*//MP2/6-31G* level of the energies of BH₂substituted ketenes indicate that these species are stabilized by 17 kcal/mol relative to methylketene,^{1,2} as indicated by isodesmic energy comparison (equation 1).² This stabilization exceeded that predicted by a correlation with substituent electronegativity, which was interpreted as indicating a favorable  $\pi$ -acceptor ability of the boron substituent shown in **1a** adding extra stability.^{1,2}



Boron-substituted ketenes are known in only a limited number of examples and are sensitive materials. Thus, despite their thermodynamic stabilization, their reactivity has limited their study.

The reaction of alkynyl ethers **5** with BBr₃ gave ketenes **7** in a reaction proposed to involve the addition intermediate **6** (equation 2).^{3,4} Reaction of the ketenes with amines or dialkylaminostannanes displaced the halides and gave aminoborane ketenes **8** (equation 2).^{3,4} The substituents on boron in ketenes **7** were also replaced on reaction with alkoxystannanes to give ketenes **9** with alkoxyborane substituents (equation 3).³ Trimethylsilylethynyl ethyl ether (**10**) reacted directly with

(2)

(3)

(4)

OEt Br₂B  $(Me_2N)_2$ DD. R₃SiC≡COEt R₃Si Rr R₃Si 5 6 7 (62%, R = i-Pr, 8 (64%, R = i-Pr,  $IR = 2100 \text{ cm}^{-1}$ ) IR 2075 cm⁻¹) Br₂B (MeO)₂B MeOSnBu₃ C=0Me₃Si Me₃Si **9** (52%, IR 2080 cm⁻¹) 7  $(Me_2N)_2B$ (Me2N)2BBr  $Me_3SiC \equiv COEt$ 80 °C Me₃Si 10 **11** (43%, IR = 2070 cm⁻¹)

 $(Me_2N)_2BBr$ , forming **11** (equation 4).⁴

Aminoborane ether 12 was prepared and reacted with Me₃SiI, also forming ketene 11 (equation 5).⁵

$$LiC \equiv COEt \xrightarrow{(Me_2N)_2BBr} (Me_2N)_2BC \equiv COEt \xrightarrow{Me_3Si} (Me_2N)_2B \xrightarrow{(Me_2N)_2B} C=0$$
(5)  
12 11

Boron also replaced tin substituents on stannylketenes 13 with a silicon or germanium substituent forming boron-substituted ketenes 14 (equation 6).^{6,7} Methanol was added to 14 with cleavage of the boron-carbon bond (equation 6).⁷



Thermal Wolff rearrangement of 3-diazoacetyl-*o*-carborane (16) with Ag₂O in aqueous dioxane at 70 °C formed the boron-substutued ketene 17, which was hydrated to the acid 18 (equation 7).⁸



As discussed in Section 3.4.1, ring opening of carboranyl-substituted cyclobutenones **19** to carborane-substituted ketenes **20** with the ketenyl group bonded to carbon in the carborane led to formation of lactones **21** in "fair to good yields" (equation 8).⁹



Diphenylketene (22) reacted with 1,3-di-*tert*-butyl-2,3-dihydro-1*H*-1,3,2-diazaboroles 23 to form 1,3,2-oxaazaborolidines 24, in a reaction proposed to be initiated by boron attack at the ketenyl oxygen (equation 8).¹⁰



#### **REFERENCES FOR SECTION 4.8.2**

- 1. Gong, L.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1991, 113, 6021-6028.
- 2. McAllister, M. A.; Tidwell, T. T. J. Org. Chem. 1994, 59, 4506-4515.
- Nikolaeva, S. N.; Zolotareva, A. S.; Ponomarev, S. V.; Petrosyan, V. S. Russ. Chem. Bull. 1994, 43, 857–861.
- Ponomarev, S. V.; Nikolaeva, S. N.; Molchanova, G. N.; Kostyk, A. S.; Grishin, Yu. K.. Zh. Obshch. Khim. 1984, 54, 1817–1821. J. Gen. Chem. (Engl. Trans.) 1984, 54, 1620– 1623.
- Ponomarev, S. V.; Gromova, E. M.; Nikolaeva, S. N.; Zolotareva, A. S. Zh. Obshch. Khim. 1989, 59, 2277–2282. J. Gen. Chem. (Engl. Trans.) 1989, 59, 2040–2044.
- 6. Ponomarev, S. V. Angew. Chem. Int. Ed. Engl. 1973, 12, 675.
- Ponomarev, S. V.; Erman, M. B.; Gervits, L. L. Zh. Obshch. Khim. 1972, 42, 469–470. J. Gen.Chem. (Engl. Transl.) 1972, 42, 463.
- 8. Zakharkin, C. I.; Kalinin, V. N.; Gedymin, V. V. Tetrahedron 1971, 27, 1317–1322.
- 9. Goudgaon, N. M.; Shi, J.; Schinazi, R. F. Tetrahedron Lett. 1998, 39, 1869-1872.
- Weber, L.; Schneider, M.; Maciel, T. C.; Wartig, H. B.; Schimmel, M.; Boese, R.; Bläser, D. Organometallics 2000, 19, 5791–5794.

## 4.8.3 Other Metal-Substituted Ketenes and Metal Ketenides

Computations predict that ketenes with electropositive substituents will be stabilized, and this has been vividly demonstrated experimentally for the long-lived silicon, germanium, and tin-substituted ketenes (Section 4.5). Lithiumketene is predicted to be much more stabilized than the silylketenes,^{1,2} but in solution has strong ionic character in an ynolate structure (Section 4.8.1). Boron is also predicted to be highly stabilizing, with covalent bonds to carbon, but in the examples known experimentally, stabilizing Group 4 substituents are also present, and the effect of the boron is not well defined (Section 4.8.2).

Computations also predict strong stabilization of ketenes RCH=C=O with R = Na, BeH, MgH, and  $AlH_2$ ,^{1,2} but these species are not known experimentally.

Dilithiumketene (1) was prepared by the reaction of 2 with *n*-BuLi and gave bis(trimethylsilyl)ketene (3) by reaction with Me₃SiCl (equation 1).³



Metal ketenides represented as  $M_2C=C=O$  are known for M = Ag, Cu, Au, and Hg.⁴⁻⁹ Silver salts react with  $CH_2=C=O$  to give the ketenide  $Ag_2C=C=O$  (4), and this ketene was also generated in situ by reaction of acetic anhydride, silver acetate, and triethylamine (equation 2).^{7,8} The structure of crystalline silver ketenide and the pyridine complex have been determined by X-ray powder diffraction.⁸ The yellow monomer  $Ag_2C=C=O$  (4) was proposed to exist briefly in solution, and then to form a polymeric precipitate that could not be redissolved. Bis(trifluoromethyl-selenyl)ketene (5) was prepared by the use of this reagent (equation 3).¹⁰

$$\begin{array}{c} O \\ CH_{3} \\ CH_{3} \\ O \\ CH_{3} \\ CH_{3} \\ CH_{3} \\ CH_{3} \\ CH_{3} \\ H \\ C=0 \\ H \\ H \\ C=0 \\ CH_{3} \\ C=0 \\ CH_{3} \\$$

$$Ag_{2}C=C=O \xrightarrow{CF_{3}Sec1} CF_{3}Se} C=O$$

$$(3)$$

$$4 \qquad 5 (45\%)$$

The reaction of acetone with a silver surface that had been oxidized by heating with oxygen gave surface bound ketenylidene (C=C=O), as identified by its strong IR band at 2026 cm⁻¹.¹¹ This is characteristic of metal ketenides, which typically absorb between 2015 and 2070 cm⁻¹.^{4,5}

Copper ketenide **6** was prepared from ketene and  $CuO_2CCF_3$  (equation 4).⁹ The reaction of mercuric acetate and acetic anhydride gave the mercury derivative **7** (equation 5),⁴ and the gold compound **8** was obtained from (1,4-oxathian)gold(I),
2,6-dimethylpyridine, and  $Et_3N$  (equation 6).^{5,6}

$$\begin{array}{c}
H \\
H \\
H
\end{array} = C = O \xrightarrow{CuO_2CCF_3} Cu_2C = C = O \\
H
\end{array} \qquad (4)$$

$$\begin{array}{c} O \\ CH_{3} \end{array} \xrightarrow{O} O \\ CH_{3} \end{array} \xrightarrow{Hg(OAc)_{2}} (AcOHg)_{2}C=C=O \\ CH_{3}CH_{3}CN, 25 ^{\circ}C \end{array} \xrightarrow{(AcOHg)_{2}C=C=O} (5)$$

Zirconium and hafnium ketenylidene complexes have also been prepared, and the X-ray structure of a zirconium complex was determined.¹² Laser-ablated metal atoms (Nb,¹³ Ta,¹³ Zr,¹⁴ Hf,¹⁴ U,¹⁵ and Th¹⁶) in solid neon

Laser-ablated metal atoms (Nb,¹³ Ta,¹³ Zr,¹⁴ Hf,¹⁴ U,¹⁵ and Th¹⁶) in solid neon reacted with CO to form carbonyls  $M(CO)_n$  and insertion products (CMO) that reacted further under photolysis to form metal ketenides (OMCCO). The ketenyl IR absorptions were at 2065.5,¹³ 2087.5,¹³ 2046.4,¹⁴ 2063.5,¹⁴ and 2051.5¹⁵ cm⁻¹ for the Nb, Ta, Zr, Hf, and U ketenides, respectively. No such complex was reported for Th. Complexes isotopically labeled with ¹³C and ¹⁸O were also studied. Density functional calculations using BP86 and B3LYP functionals found  $C_s$  symmetric structures with bent (OMC) angles for these ketenes and reasonable agreement with the observed IR spectra.

Ketenylidenetriphenylphosphorane (9) reacted with platinum complexes, forming platinum-substituted ketenes 10 (equation 7).^{17,18} These compounds are stable solids, and a variety have been synthesized and characterized and their chemistry investigated (see also Section 4.6).^{17,18}

$$L_{2}Pt(Me)Cl \xrightarrow{1) Ag^{*}BF_{4}^{-}}_{L = cod} \left[ \begin{array}{c} Me \\ L_{2}Pt \\ Ph_{3}P \\ 9 \end{array} \right]^{+} BF_{4}^{-} BF_{4}^{-}$$

$$10 (93\%, \\ IR 2040 cm^{-1})$$
(7)

Reaction of the ester 11 in acid gave the metal acylium cluster 12, which was observed by NMR, and captured with MeOH, forming 13 (equation 8).^{19,20} Acetyl chloride reacted with a carbanionic iron carbonyl, forming  $[(OC)_3Fe]_3C=C=O]^{2-1}$  (14, equation 9),²¹ and analogous cobalt and osmium compounds were also prepared.²²

$$\begin{array}{cccc} Cp_{2}Co_{3}(CO)_{4}CO_{2}Pr\text{-}i & \xrightarrow{HPF_{6}} & [(OC)_{3}Co]_{3}C\text{-}\overset{+}{C}=O & PF_{6}^{-} & \xrightarrow{MeOH} & Cp_{2}Co_{3}(CO)_{4}CO_{2}Me & (8) \\ \hline 11 & 12 & (70\%, IR & 2040 \text{ cm}^{-1}) & 13 & (97\%) \end{array}$$



### **REFERENCES FOR SECTION 4.8.3**

- 1. Gong, L.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1991, 113, 6021-6028.
- 2. McAllister, M. A.; Tidwell, T. T. J. Org. Chem. 1994, 59, 4506-4515.
- 3. Hoppe, I.; Schöllkopf, U. Liebigs Ann. Chem. 1979, 219-226.
- 4. Blues, E. T.; Bryce-Smith, D.; Karimpour, H. J. Chem. Soc. Chem. Commun. 1979, 1043–1044.
- Blues, E. T.; Bryce-Smith, D.; Lawston, I. W.; Wall, G. D. J. Chem. Soc. Chem. Commun. 1974, 513–514.
- 6. Bryce-Smith, D.; Blues, E. T. Ger. Offen. DE 2,336,396. Chem. Abstr. 1974, 80, 108672m.
- 7. Bryce-Smith, D. Chem. Ind. (London) 1975, 155-158.
- Blues, E. T.; Bryce-Smith, D.; Shaoul, R.; Hirsch, H.; Simons, M. J. J. Chem., Perkin Trans. 2 1993, 1631–1642.
- 9. Blues, E. T.; Bryce-Smith, D.; Kettlewell, B.; Roy, M. J. Chem. Soc. Chem. Commun. 1973, 921.
- 10. Haas, A.; Praas, H.-W. Chem. Ber. 1992, 125, 571-579.
- 11. Sim, W. S.; King, D. A. J. Am. Chem. Soc. 1995, 117, 10583-10584.
- Calderazzo, F.; Englert, U.; Guarini, A.; Marchetti, F.; Pampaloni, G.; Segre, A.; Tripepi, G. Chem. Eur. J. 1996, 2, 412–419.
- 13. Zhou, M.; Andrews, L. J. Phys. Chem. A 1999, 103, 7785-7794.
- 14. Zhou, M.; Andrews, L. J. Am. Chem. Soc. 2000, 122, 1531-1539.
- 15. Zhou, M.; Andrews, L.; Li, J.; Bursten, B. E. J. Am. Chem. Soc. 1999, 121, 9712-9721.
- Zhou, M.; Andrews, L.; Li, J.; Bursten, B. E. J. Am. Chem. Soc. 1999, 121, 12188– 12189.
- 17. Bertani, R.; Pandolfo, L.; Zanotto, L. Inorg. Chim. Acta 2002, 330, 213-219.
- 18. Bertani, R.; Casarin, M.; Pandolfo, L. Coord. Chem. Rev. 2003, 236, 15-33.
- D'Agostino, M. F.; Mlekuz, M.; McGlinchey, M. J. J. Organomet. Chem. 1988, 345, 371–382.
- 20. Seyferth, D.; Hallgren, J. E.; Eschbach, C. S. J. Am. Chem. Soc. 1974, 96, 1730-1737.
- 21. Jensen, M. P.; Shriver, D. F. J. Mol. Catal. 1992, 74, 73-74.
- 22. Kollis, J. W.; Holt, E. M.; Shriver, D. F. J. Am. Chem. Soc. 1983, 105, 7307-7313.

#### 4.8.4 Metal-Complexed Ketenes

Metal complexes of ketenes have been prepared in a wide variety of structural forms, with metal coordination by either  $\sigma$  or  $\pi$  bonding.^{1,2} Ketenyl complexes

with a metal as one of the ketene substituents, and ketenylidene complexes in which the ketene carbon has only metals as substituents, are covered elsewhere. This section focuses on non-sigma bonded complexes, which may be formed by direct addition of the metal complex to the ketene (equation 1),³ by carbonylation of metal carbene complexes (equation 2),^{4–7} and by other routes. Often the bonding involves not only the ketenyl moiety but also the ketene substituents, as in the Fe(CO)₃ complex of a vinylketene (**6**). Reviews of the subject are available.^{1,8–11} This chapter is not intended to be comprehensive or systematic, but only illustrative of some of the known examples. Further information is given in Sections 3.5 and 4.8.3.

$$\overset{Ph}{\underset{Ph}{\longrightarrow}} C=O + Cp_2V \longrightarrow \overset{Ph}{\underset{Ph}{\longrightarrow}} \overset{O}{\underset{VCp_2}{\longrightarrow}} (1)$$

$$\begin{array}{cccc}
Ph \\
Ph \\
Ph \\
Ph \\
Mn(CO)_2Cp \\
\hline & Co \\
\hline & Fc = 0 \\
Ph \\
Mn(CO)_5Cp \\
\hline & S \\
\hline & Fe(CO)_3 \\
\hline & Fe(CO)_3 \\
\hline & Fe(CO)_3 \\
\hline & 6 \\
\end{array}$$
(2)

As described in Section 3.5, the Dötz reaction of photolysis of a metal carbene complex 7 in the presence of an alkyne has been proposed to form a metal-complexed alkenylketene 8, which reacts further, for example by intramolecular cyclization forming 9 (equation 3).^{5,6} The complexed ketenes have usually not been directly observed, but the products formed in some cases are still complexed with the metal.



A number of pathways for the course of these reactions have been presented.¹² For example, upon photolysis a chromium carbene complex **10** has been proposed to undergo direct insertion of CO forming ketene complex **11**, and this reacted by intramolecular [4 + 2] cycloaddition (equation 4). For the formation of **8** (equation 3), it was suggested that upon thermolysis of complex **7** in the presence of an alkyne

metallocyclobutene **13** could form by displacement of CO by the alkyne, followed by ring opening to carbene complex **14**, which inserts CO, forming a metal-complexed alkenylketene **8** (equation 5).^{6,12} Further studies of the mechanism of these reactions are presented in Section 3.5.



The structure of the complex of ketene with  $Cr(CO)_4OH_2$  was calculated using the B3LYP/6-31G* basis set for C, H, and O and a double  $\zeta$  basis set for Cr.¹³ The reaction pathway for reaction of the complex with  $CH_2=NH$  forming a  $\beta$ -lactam was also calculated.¹³ Molecular orbital studies of the complex of  $CH_2=C=O$ with Pd(PR₃)₂ indicated that bonding of Pd to C=C is favored, rather than to C=O, by 10–15 kcal/mol.¹⁴ Reaction of the chromium carbene complex **15** with diphenylacetylene was proposed to give alkenylketene complexes **16** (equation 6), that cyclized to give the observed product **17** (equation 7).^{15,16}



Photolysis of the alkenyl carbene complexes 18 in the presence of diphenylacetylene formed dienylketenes 19, which cyclized to the phenols 20 with an enhanced yield using photolysis with a Xenon lamp (equation 8).¹⁷ Cyclizations of metal complexes with two tethered alkynyl groups were carried out,¹⁸ and amino-tethered bisalkynyl chromium carbene complex **21** was proposed to react to give the ene-yne ketene chromium complex **22** (equation 9), which cyclized via diradical intermediate **23** to give indoline **24** (equation 10).¹⁹ The use of triethylsilane to capture the diradical helped to prevent demetallation.¹⁹



Photolysis of chiral hydroxycarbene complex **25** led to 4-hydroxy ketene **26** that cyclized to lactone **27**, and the major diastereomer was isolated (equation 11).²¹ Other chiral auxiliaries were also employed in these reactions.²¹



Alkenylketene complexes with iron tricarbonyl were formed by reaction of alkenylketones with alkyllithium reagents under an atmosphere of CO.²² Iron carbene complexes have been suggested as intermediates in these processes.²² Reaction of the enantiomerically enriched ketone complex **28** gave the ketene complex **30** with complete retention of stereochemistry, a result consistent with the intervention of a chiral carbene complex such as **29** (equation 12).²² Upon purification the product **30** was obtained in 57% yield, 96%  $ee.^{22}$ 



Reaction of vinylketene tricarbonyl iron (0) complex **31** with phosphoramidates gave complexed ketenimine **32** in an aza-Wittig reaction (equation 13).²³ The reactions of **31** with dienophiles have also been examined.²⁴ Reaction of complexed vinyl ketone **33** with methyllithium gave the complexed vinylketene **34** (equation 14).²⁵ Reaction of vinyllithium reagents generated from vinylstannanes **35** with Fe(CO)₅ followed by reaction with EtOTf gave vinylketene iron complexes **36** (equation 15).²⁶ The sequence was also successful with PPh₃ replacing one of the CO ligands.



Photolysis of the iridium complex 37 at -25 °C gave 38, which reformed 37 at ambient temperature, while photolysis of 38 at 28–30 °C gave the enol 39, which

isomerized to **40** on standing (equation 16).²⁷ Reaction of **37** with alkynes gave complexes **41**, which readily reacted further with C–C bond formation to give **42** (equation 17).²⁸ Iridium complex **43** was reversibly ionized to the carbene complex **44** by removal of chloride (equation 18).^{29,30} A number of structures were compared,³¹ and the chemistry of complex **43** has been modeled in a theoretical study.³²



Excess phenyl(methyl)ketene (**45**) reacted with the iridium-alkyne complex **46**, forming **47** in 78% isolated yield in a double C–H activation of the CH₃ group of **45** (equation 19).³³ There was some polymerization of **45**. Heating of **47** under a CO atmosphere at 60-80 °C formed **48**, also in 78% yield (equation 19).³³



Reaction of the tantalum complex **49** with CO gave the ketenyl complex **50**, whose structure was confirmed by X-ray determination (equation 20).³⁴ Niobium

ketene complexes **51** were prepared as *E* and *Z* isomers, but photoexcitation did not result in E/Z isomerization.³⁵ Vinyl complex **52** reacted on the double bond with cyclopentadiene in a Diels-Alder fashion and with free ketenes.³⁶ Other such niobium complexes have been prepared and studied.^{37–39}



The complex **54** was formed by a [2 + 2] cycloaddition of diphenylketene (**53**) with a titanaallene (equation 21).⁴⁰ The stable diarylstannylene **55** reacted with ketenes in [2 + 1] cycloadditions, forming **56** (equation 22).⁴¹ This reaction is also noted in Section 5.10.1. Reaction of acetylene with CrCl₂O₂ in an argon matrix at 7 K gave the end-on ketene complex **57**, identified by the IR band at 2095 cm⁻¹, as compared to the free ketene absorption at 2142 cm⁻¹ (equation 23).⁴²



The iron ketene complex **60** was prepared by the reaction of the metal complex **58** with trimethylsilylketene **59**, in 20% yield as determined by IR (equation 24).⁴³ Complex **60** was obtained independently in 95% yield and in better purity by the reaction of  $(OC)_2Fe(PEt_3)_2N_2$  with Me₃SiCH=C=O, and was assigned the *E* 

62

configuration on the basis of the upfield shift of the olefinic proton. The reaction of **60** with CO liberated  $Me_3SiCH=C=O$ .



Malonate esters were obtained in a low-yielding catalytic conversion of  $CH_2I_2$  that was proposed to involve **61** forming a ketene complex **62** (equations 25, 26).⁴⁴

 $RhI(CO)(PEt_{3})_{2} \xrightarrow{CH_{2}I_{2}} RhI_{2}(CO)(CH_{2}I)(PEt_{3})_{2} \xrightarrow{CO} RhI_{2}(CO)(OCCH_{2}I)(PEt_{3})_{2}$ (25) 61 61 61 61  $(RhI_{2}(CO)(O=C=CH_{2})]^{+}I^{-} \xrightarrow{EiOH, CO} CH_{2}(CO_{2}Et)_{2}$ (26)

Reaction of **63** with  $Ph_2C=C=O$  (**53**) gave the complex **64**, which was converted by potassium indenyl to **65**, which in turn was converted to **66**, the first example of a linkage-isomeric ketene complex (equation 27).⁴⁵ The ketenyl IR of **66** was at 1777 cm⁻¹, which is typical for such  $\eta^2(CO)$  complexes.⁴⁵ The structures of **65** and **66** were confirmed by X-ray crystallography. Reaction of  $Ph_2C=C=O$  (**53**) with a mixture of a phosphine and Rh-cyclooctene complexes **67** gave complexes **68**, which were converted to complexes **69** at rates determined by the nature of the phosphine complex (equation 28).⁴⁶



The manganese ketene complexes 70 and 71 were found by NMR studies to undergo both rotation around the ketene axis, interchanging the positions of the

aryl groups, and exchange of the ketene CO with those on manganese.⁴⁷

$$\begin{array}{c} \operatorname{Ar}_{\bullet} & \operatorname{C}_{\bullet} & \operatorname{C}_{\bullet} \\ \operatorname{Ph}_{\bullet} & \operatorname{C}_{\bullet} & \operatorname{PhCC}[\operatorname{CO}_2(\operatorname{CO})_6] \\ \operatorname{Mn}(\operatorname{CO})_2 & \operatorname{Ph}_{\bullet} & \operatorname{Cp} & \operatorname{Ph}_{\bullet} \\ \end{array}$$

Reaction of dimethylcyclobutenedione with ferrocenyl chromium carbene complex **72** has been proposed to form metal-complexed bisketene **73** as an intermediate leading to the lactone **74**, which was the only product observed from the cyclobutenedione (equation 29).⁴⁸ The unique organometallic-substituted ketene **76** was prepared from the dialkyne **75**, and the structure was confirmed by X-ray (equation 30).^{49,50}



Reaction of 2-ethynylphenyl epoxides **77** with the ruthenium catalyst TpRuPPh₃(CH₃CN)₂PF₆ [Tp = tris(1-pyrazolyl)borate] proceeded with high selectivity to give either 2-naphthols **81** (R = *i*-Pr, 81%) or 1-alkylidene-2-indanones **82** (R = R¹ = Me, 89%) (equations 31, 32).⁵¹ The process was shown to proceed through the intermediacy of the metal-complexed ketene **80** by deuterium labeling of the alkyne, resulting in formation of **81** and **82** with rearranged label, and by capture of the ketene **80b** with *i*-butanol, giving the ester **83** in 90% yield (equation 33).⁵¹ Ene-yne epoxides were similarly converted to complexed dienylketenes that formed cyclohexadienones (Section 4.1.2).⁵²



79

78

**77** (**a**, R = i-Pr,  $R^1 = H$ ; **b**,  $R = R^1 = Me$ )



Complexation of diphenylketene (53) with the lanthanum complexes 84 gave insertion into the metal-sulfur bond, forming complexes 85 (equation 34).⁵³



Intramolecular cyclization of a chromium carbene complex **86** gave a [2.2]-*meta*-cyclophane **88** through a ketene complex **87** (equation 35).⁵⁴



# **REFERENCES FOR SECTION 4.8.4**

- 1. Geoffroy, G. L.; Bassner, S. L. Adv. Organomet. Chem. 1988, 28, 1-83.
- 2. Jensen, M. P.; Shriver, D. F. J. Mol. Catal. 1992, 74, 73-84.
- Gambarotta, S.; Pasquali, M.; Floriani, C.; Chiesi-Villa, A.; Guastini, C. *Inorg. Chem.* 1981, 20, 1173–1178.
- Kelly, C.; Lugan, N.; Terry, M. R.; Geoffroy, G. L.; Haggerty, B. S.; Rheingold, A. L. J. Am. Chem. Soc. 1992, 114, 6735–6749.

- 5. Dötz, K. H. Angew. Chem. Int. Ed. Engl. 1975, 14, 644-645.
- 6. Fischer, H.; Mühlemeir, J.; Märkl, R.; Dötz, K. H. Chem. Ber. 1982, 115, 1355-1362.
- 7. Dötz, K. H.; Schäfer, T. O.; Harms, K. Synthesis 1992, 146-150.
- 8. Dötz, K. H. New J. Chem. 1990, 14, 433-445.
- 9. Dötz, K. H.; Tomuschut, P. Chem. Soc. Rev. 1999, 28, 187-198.
- 10. de Meijere, A.; Schirmer, H.; Duetsch, M. Angew. Chem. Int. Ed. Engl. 2000, 39, 3964.
- Wulff, W. D. In *Comprehensive Organometallic Chemistry II*; Abel, E. W.; Stone, F. G. A.; Wilkinson, G; Eds.; Pergamon Press, New York, 1995, Vol. 12, p. 469.
- 12. Merlic, C. A.; Xu, D. J. Am Chem. Soc. 1991, 113, 7418-7420.
- Arrieta, A.; Cossío, F. P.; Fernández, I.; Gómez-Gallego, M.; Lecea, B.; Mancheño, M. J.; Sierra, M. A. J. Am. Chem. Soc. 2000, 122, 11509–11510.
- 14. Liu, X. P.; Gong, J. K.; Seyler, J. W. J. Coord. Chem. 2000, 52, 47-56.
- Duetsch, M.; Vidoni, S.; Stein, F.; Funke, F.; Noltemeyer, M.; de Meijere, A. J. Chem. Soc. Chem. Commun. 1984, 1679–1680.
- 16. Flynn, B. L.; Silveira, C. C.; de Meijere, A. Synlett. 1995, 812-814.
- Choi, Y. H.; Rhee, K. S.; Kim, K. S.; Shin, G. C.; Shin, S. C. *Tetrahedron Lett.* 1995, 36, 1871–1874.
- 18. Rahm, A.; Wulff, W. D. J. Am. Chem. Soc. 1996, 118, 1807-1808.
- Bao, J.; Wulff, W. D.; Dragisich, V.; Wenglowsky, S.; Ball, R. G. J. Am Chem. Soc. 1994, 116, 7616–7630.
- 20. Schmeck, C.; Hegedus, L. S. J. Am. Chem. Soc. 1994, 116, 9927-9934.
- 21. Reed, A. D.; Hegedus, L. S. J. Org. Chem. 1995, 60, 3787-3794.
- 22. Benyunes, S. A.; Gibson, S. E. J. Chem. Soc. Chem. Commun. 1996, 43-44.
- 23. Benyunes, S. A.; Gibson, S. E.; Stern, J. A. J. Chem. Soc. Perkin Trans. 1 1995, 1333– 1334.
- Gibson, S. E.; Saberi, S. P.; Slawin, A. M. Z.; Stanley, P. D.; Ward, M. F.; Williams, D. J.; Worthington, P. J. Chem. Soc. Perkin Trans. 1 1995, 2147–2154.
- 25. Gibson, S. E.; Tustin, G. J. J. Chem. Soc. Perkin Trans. 1 1995 2427-2432.
- 26. Gibson, S. E.; Ward, M. F.; Kipps, M.; Stanley, P. D.; Worthington, P. A. J. Chem. Soc. Chem. Commun. 1996, 263–264.
- 27. Grotjahn, D. B.; Lo, H. C. J. Am. Chem. Soc. 1996, 118, 2097-2098.
- 28. Lo, H. C.; Grotjahn, D. B. J. Am. Chem. Soc. 1997, 119, 2958-2959.
- 29. Grotjahn, D. B.; Bikzhanova, G. A.; Collins, L. S. B.; Concolino, T.; Lam, K.-C.; Rheingold, A. L. J. Am. Chem. Soc. 2000, 122, 5222–5223.
- 30. Werner, H.; Blenel, E. Angew. Chem. Int. Ed 2001, 40, 145-146.
- Grotjahn, D. B.; Collins, L. S. B.; Wolpert, M.; Bikzhanova, G. A.; Lo, H. C.; Combs, D.; Hubbard, J. L. J. Am. Chem. Soc. 2001, 123, 8260–8270.
- Urtel, H.; Bikzhanova, G. A.; Grotjahn, D. B.; Hofmann, P. Organometallics 2001, 20, 3938–3949.
- 33. Grotjahn, D. B.; Hoerter, J. M.; Hubbard, J. L. J. Am. Chem. Soc. 2004, 126, 8866-8867.
- 34. Schmidt, J. A. R.; Arnold, J. Organometallics 2002, 21, 3426-3433.
- 35. Kerr, M. E.; Fermin, M. C.; Bruno, J. W. J. Chem. Soc. Chem. Commun. 1996, 1221– 1222.

- 36. Kerr, M. E.; Bruno, J. W. J. Am. Chem. Soc. 1997, 119, 3183-3184.
- 37. Kerr, M. E.; Sarker, N.; Hneihen, A. S.; Schulte, G. K.; Bruno, J. W. *Organometallics* **2000**, *19*, 901–912.
- Antiñolo, A.; Otero, A.; Fajardo, M.; Gil-Sanz, R.; Herranz, M. J.; López-Mardomingo, C.; Martín, A.; Gómez-Sal, P. J. Organomet. Chem. 1997, 533, 87–96.
- 39. Kerr, M. E.; Zhang, X.-M.; Bruno, J. W. Organometallics 1997, 16, 3249-3251.
- 40. Beckhaus, R.; Strauss, I.; Wagner, T. Z. Anorg. Allg. Chem. 1997, 623, 654-658.
- 41. Edelmann, F. T.; Belay, M. J. Fluorine Chem. 1997, 84, 29-33.
- 42. Wistuba, T.; Limberg, C. Eur. J. Inorg. Chem. 1999, 1335-1342.
- Kandler, H.; Bidell, W.; Jänicke, M.; Knickmeier, M.; Veghini, D.; Berke, H. Organometallics 1998, 17, 960–971.
- 44. Weston, W. S.; Lightfoot, P.; Cole-Hamilton, D. J. J. Organomet. Chem. 1998, 553, 473–476.
- 45. Bleuel, E.; Laubender, M.; Weberndörfer, B.; Werner, H. Angew. Chem. Int. Ed. 1999, 38, 156–159.
- 46. Grotjahn, D. B.; Bikzhanova, G. A.; Hubbard, J. L. Organometallics **1999**, *18*, 5614–5619.
- 47. Ortin, Y.; Coppel, Y.; Lugan, N.; Mathieu, R.; McGlinchey, M. J. Chem. Commun. 2001, 2636–2637.
- 48. Zora, M.; Yucel, B.; Peynircioglu, N. B. J. Organomet. Chem. 2002, 656, 11-17.
- 49. Akita, M.; Chung, M.-C.; Terada, M.; Miyauti, M.; Tanaka, Y.; Moro-oka, Y. J. Organomet. Chem. **1998**, 565, 49–62.
- 50. Akita, M.; Sakurai, A.; Chung, M. C.; Moro-oka, Y. J. Organomet. Chem. 2003, 670, 2–10.
- 51. Madhushaw, R. J.; Lin, M.-Y.; Abu Sohel, S. M.; Liu, R.-S. J. Am. Chem. Soc. 2004, 126, 6895–6899.
- 52. Ming-Yuan, L.; Madhushaw, R. J.; Liu, R.-S. J. Org. Chem. 2004, 69, 7700-7704.
- 53. Zhang, C.; Liu, R.; Zhou, X.; Chen, Z.; Weng, L.; Lin, Y. Organometallics 2004, 23, 3246–3251.
- 54. Dötz, K. H.; Mittenzwey, S. Eur. J. Org. Chem. 2002, 39-47.

#### 4.9 **BISKETENES**

The simplest bisketene,  $C_2O_2$ , or CO dimer (1), has been the object of continuing theoretical^{1,2} and experimental³ examination, and while an energy minimum structure for this molecule has been calculated,¹ it is still unknown experimentally. It was concluded that this species is an intrinsically short-lived molecule that was predicted to dissociate rapidly to CO within a few nanoseconds.³ Carbon suboxide,  $C_3O_2$  (2), was the first bisketene that was prepared, and was discovered by Diels and Wolf in 1906,⁴ only 1 year after the report of the first ketene,  $Ph_2C=C=O$ . Carbon suboxide, a stable, colorless gas with a linear structure, is discussed in Section 4.11. Ketene **3** has been described as a 1,2-bisketene, and **1** and **2** may correspondingly be classified as (0,0) and (1,1) bisketenes; **4–7** are the next higher

homologues of this series, respectively. Staudinger and Kreis in 1923 reported a systematic effort to prepare the linear  $\alpha,\omega$ -bisketenes, but were unsuccessful in their attempt to prepare O=C=CHCH=C=O (3) by the double decarboxylation of **8** and obtained the bis(anhydride) **9** instead (equation 1).⁵



The parent 1,2-bisketene **3** was generated by a double Wolff rearrangement of **10** in an argon matrix at 10 K, identified by its IR absorption at 2125 cm⁻¹, and captured by methanol as the succinate (equation 2).⁶ Pyrolytic decarbonylation of the trione **11** also gave **3**, as evidenced by the formation of cyclobutenedione **12** in 9% yield (equation 3).⁷ Photolysis of **12** in solution gave evidence for the formation of **3**, with IR absorption at 2120 cm⁻¹ and the ¹H NMR signal at  $\delta$  3.20 (equation 3).⁸ Reviews on cyclobutenedione chemistry have appeared.^{8a,b}



Initial computations at the RHF/3-21G level indicated that the bisketene 3a was an energy minimum, 4.1 kcal/mol more stable than the cyclobutenedione 12.9 However, further computations at the MP2/6-31G*//MP2/6-31G* level showed that

bisketene **3** was most stable in a twisted, almost perpendicular geometry, with energies relative to cyclobutenedione **12** for **3**, *cisoid* **3a**, and *transoid* **3b** of 2.9, 5.6, and 4.0 kcal/mol, respectively.^{9a} The planar geometry was proposed to be disfavored because of the repulsion between C₂ of the two ketenyl moieties arising from the high  $\pi$ -electron density at these atoms and the absence of any favorable conjugative interaction between the two ketenyl groups.^{9a} The preference for the twisted conformation of 1,2-bisketenes was confirmed by the X-ray structure of the silyl-substituted tetraketene **14** obtained from photolytic ring opening of the bis(cyclobutenedione) **13**, with dihedral angles of 119° between the adjacent ketenyl groups (equation 4).¹⁰ The tetraketene **14** was thermodynamically stable to reversion to **13**, confirming the predictions based on theoretical calculations.⁹ The measured dipole moment of 2.7 *D* of the bisketene **26** (vide infra)^{10a} and the photoelectron spectrum^{10a} also confirm that 1,2-bisketenes prefer twisted conformations.



The 1,4-bisketene **5** was formed by the double dehydrochlorination of adipyl chloride **15** using 1,8-bis(dimethylamino)naphthalene (**16**) as a strong stochiometric base and  $Et_3N$  as a kinetic shuttle base (Section 3.2.2), and was identified by the IR absorption at 2114 cm⁻¹ (equation 5) and by capture as the tetraadduct by the stable free radical TEMPO (Section 5.7).¹¹ The 1,5-bisketene **6** was prepared in a similar fashion.¹¹ The 1,6-bisketene **7** was prepared by a photochemical double Wolff rearrangement of bis(diazo ketone) **17**, identified by its IR absorption at 2120 cm⁻¹, and trapped with benzylamine as the diamide **18** (equations 6, 7).¹¹





The monophenyl bisketene **20** was first generated by refluxing of the cyclobutenedione **19** in MeOH, which led to the succinate ester **21** and the lactone **22**, although there was some question about whether the products could arise from attack on **19** (equation 8).¹² Photolysis of 3,4-diphenyl-1,2-cyclobutenedione **22** in THF at  $-78 \,^{\circ}$ C gave the bisketene **23**, as observed by the IR absorption at 2103 cm⁻¹ that persisted for several days at  $-78 \,^{\circ}$ C (equation 9).¹³⁻¹⁶ Upon generation of **23** in MeOH at 77 K, IR bands at 2100 and 2112 cm⁻¹ were observed, and at  $-80 \,^{\circ}$ C these decayed, with concomitant formation of **24** (equation 9).¹³



1,2-Bis(trimethylsilyl)cyclobutenedione **25** on thermal or photochemical ring opening gave the bis(trimethylsilyl)-1,2-bisketene **26** in a process that by MP2/6-31G*//MP2/6-31G* computations was exothermic by 5.0 kcal/mol (equation 10).^{9a,17,18} The thermal conversion at 100 °C was quantitative, and **26** appeared to be stable indefinitely in the absence of moisture or oxygen, but on continued photolysis underwent decarbonylation, forming bis(trimethylsilyl)acetylene (equation 10).¹⁷ Reaction of **26** with oxygen gave the maleic anhydride **28** in a reaction suggested to involve the peroxide intermediate **27** (equation 11).¹⁷





A series of aryl-substituted bisketenes **30** were generated photochemically, and the rates of their thermal conversion back to cyclobutenediones **29** were correlated with the  $\sigma_p^+$  constants of the aryl substituents, with  $\rho^+ = -1.10$  (equation 12).¹⁹ The ketenes were predicted to be stabilized by electropositive substituents,¹⁸ and the cyclobutenediones stabilized by electron donors. These effects were reflected in both the rates of ring closure and the amount of bisketene **30** present in thermal equilibrium with **29**.¹⁹



A larger group of 1,2-bisketenes **32** and others, including **33**, have been generated by photolysis of 1,2-cyclobutenediones **31**, and the kinetics of their ring closure (equation 13) were measured by UV or by TRIR spectroscopy, as reported in Table 4.6.^{20,21} All of these bisketenes were identified by their characteristic IR bands (Table 4.6), and in a number of cases the rate constants for ring closure were intermediate between the faster rates that could be measured by TRIR spectroscopy and the slower rates that were accessible by conventional UV spectroscopy.



A study of 1,2-bisketene **320** ( $R = R^1 = PhMe_2Si$ ) showed that this had a rate of ring closure similar to that of the corresponding substrate with Me₃Si substituents, and there was no evidence for an interaction of the phenyl group with the ketene functionalities that would affect the rate of ring closure or of reaction with H₂O.²² Similarly, there were no noticeable conjugative effects of phenyl on the ¹³C, ¹⁷O, or ²⁹Si NMR spectra of PhSiMe₂CH=C=O.²²

R, R ¹		$v_1, v_2 (cm^{-1})$	$k_{obs} \ (s^{-1})$
H, Ph	20	2103, 2127	$1.70 \times 10^{-4}$
Ph, Ph	23	2096, 2110	$3.89  imes 10^{-2}$
Me, Me	32a	2092, 2114	$3.60  imes 10^{-2}$
t-Bu, t-Bu	32b	2087, 2113	$8.19 imes10^{-4}$
Me ₃ Si, Me ₃ Si	26	2084	$(10^{-10})^b$
Me ₃ Si, Ph	30a	2093	$2.03  imes 10^{-6}$
Me ₃ Si, Me	32c	2101	$4.37 \times 10^{-6}$
Me, Ph	32d	2093, 2112	$3.50  imes 10^{-2}$
PhS, PhS	32e	2112	$5.62 \times 10^{-3}$
EtO, Me ₃ Si	32f	2090, 2104	$1.77 \times 10^{-3}$
CN, Ph	32g	2116, 2140	$8.51  imes 10^{-2}$
Cl, Cl	32h	2113, 2154	$3.20 \times 10^3$
Br, Ph	32i	2105, 2132	0.137
Cl, MeO	32j	2098, 2142	$2.97  imes 10^4$
t-BuO, t-BuO	32k	а	$2.5  imes 10^4$
t-Bu, i-PrO	321	2092, 2108	$3.2  imes 10^4$
$(t-BuMe_2SiC\equiv C)_2$	32m	2111, 2119	$4.23 \times 10^{-3}$
(PhC≡C) ₂	32n	a	$1.59  imes 10^{-2}$
	33	2072, 2128	$1.9  imes 10^4$

TABLE 4.6 Rate Constants for Ring Closure of Bisketenes 28 in CH₃CN at 25 °C^{20,21}

^aNot measured.

^bEstimated.

Photolysis of squaric acid (**34**) in an Ar matrix at 10 K gave IR bands at 2162, 2132, and 2092 cm⁻¹, attributed to the bisketene **35** (equation 14).²³ Further photolysis led successively to deltic acid **36** and dihydroxyacetylene (**37**). Attempted sublimation of **36** at 160 °C gave water and carbon suboxide  $O=C=C=C=O.^{23}$ 



Photolysis of the bis(dioxocyclobutenyl)benzene **38** gave the bisketene **39** (equation 15) and then the tetraketene **40** (equation 16), which was long-lived at room temperature but was converted thermally back to **39** and **38** in processes whose rates were measured using UV spectroscopy.²⁴





The reaction of bis(trimethylsilyl)-1,2-bisketene **26** with neat alcohols followed by rapid removal of the solvent permitted isolation of ketenyl esters **42**.¹⁷ The reaction was catalyzed by added lithium alkoxide in a process suggested to proceed with formation of the enolate **41** (equation 17).²⁵ Further addition of alcohols to **26** catalyzed by lithium alkoxides was rapid and gave succinate esters **44** by the proposed intermediacy of **43** (equation 18).²⁵ The use of the alkoxide catalyst minimized desilylation, which was a predominant pathway in the prolonged reaction of **26** with alcohols in the absence of base. The efficient addition of alkoxides to the ketenyl carbonyl was attributed to lithium coordination to the ketenyl oxygen.²⁵ The monoanion of catechol reacted with **26** to give ortho esters **45** under similar conditions (equation 19).²⁵



The reactions of alcohols with bisketene **26** catalyzed by Cu(I) and Cu(II) complexes gave butenolide **48** in a process that may conceivably involve reaction of complexes **46** or **47** with the alcohol (equation 20).^{26,27} Other bisketenes also formed butenolides.^{26,27} Chiral ligands on the catalyst did not give enantioselectivity

in the isolated **48**, a result attributed to racemization of the product.²⁷ An analogous ruthenium complex **50** was proposed to form in the reaction of diethylcyclopropenone (**49**) with  $Ru_3(CO)_{12}$  forming the pyranopyrandione **51** (equation 21).²⁸



The reactions of **26** with alcohols catalyzed by chiral amines gave 19–34% *ee* in formation of ketenyl esters **42** (equation 17).^{26,27} Catalysis of the addition of alcohols and other nucleophiles to **26** by lactams was effective but gave low % *ee*.^{26,27} Ketenyl esters **42** reacted with amines to give ester amides **52** (equation 22),^{26,27} with thiols to give mixed esters, and with cyclopentadiene to give the [2 + 2] cycloaddition product **53** (equation 22).^{26,27}



The bisketene **26** reacted rapidly with one equivalent of primary alkylamines to give selectively monoketenyl amides **54** that were isolated and fully characterized. These reacted with a second amine in a much slower process to give diamides **55** as *dl* and *meso* isomers that were separated by chromatography (equation 23).²⁹ The kinetics and mechanisms of these processes have been examined (Section 5.5).²⁹ Reaction of **26** with *R*-1-phenylethylamine gave a selectivity of 2:1 in formation of the new chiral center, and this product was purified by recrystallization to give **54** with a 98:2 *dr*.³⁰ There was high diastereoselectivity in the reaction of **54** with alcohols.^{27,30} Reaction of **26** with 3-amino-1-propanol gave a ketenyl amide **56** that cyclized to **57** on catalysis with 2-piperidone (equation 24).²⁷ A single

secondary amine reacted with 26 with cyclization to form aminodihydrofuranones 57a (equation 24a).³¹



57a (97–99%)

26

Reaction of **26** with CH₃CH=O catalyzed by BF₃ gave the  $\beta$ -lactone **58** as the only product detected, which on pyrolysis gave decarboxylation to the vinylketene **59** (equation 25).³² The reaction of **26** and of the unsymmetrical bisketene **30a** with CH₂N₂ gave mixtures of cyclopentenediones **60** and lactones **61** (R = H), while Me₃SiCHN₂, and PhCHN₂ gave only **60** (equation 26).³² The preference for formation of **60** with the bulkier substituted diazomethanes was attributed to in-plane attack on the ketenyl group from the side opposite the group R¹ forming the intermediate **62**. Dimethoxycarbene from **66** reacted with **26** to give cyclopentendione **67** (equation 27).³² Heating the cyclobutenedione **68** to form the bisketene **69** resulted in formation of the isolable ketene product **71**, and was interpreted to proceed by [2 + 2] cyclodimerization to **70**, which reacted by decarbonylation and rearrangement to give **71** (equation 28).⁸





The nucleophilic alkyne Me₃SiC $\equiv$ COEt reacted with **30a** to give spiro-cyclopropenylfuranone **72**, and also reacted by [4 + 2] cycloaddition to form benzoquinone **73** (equation 29).³² The metal complexed bisketene **74** analogous to **46** and **50** reacted with diarylalkynes to give **75** (equation 30).³³ The X-ray structure of complex **74** (Figure 4.3) showed C-C-O bond angles of 137.8 and 138.5°, indicating a major diminution of ketenyl character, as also manifested by the IR stretching frequency of 1810 cm⁻¹.³⁴





**Figure 4.3** X-ray structure of **74** (reprinted from ref. 34 with permission of the American Chemical Society).

Molecular orbital calculations predicted a preference for electrophilic attack on  $C_1$  of the 1,2-bisketene **3**. This was explained on the basis of the extended conjugation possible in the cation **76**, in contrast to the less extensive acylium conjugation possible in **77**, which would result from the usual electrophilic attack at  $C_2$  of ketenes (equation 31).³⁵ Nevertheless, CF₃CO₂H addition and hydration of **26** during photolysis gave the products **77** and **78**, respectively, arising from protonation of  $C_2$  (equation 32).³⁶ This result permitted the observation of the ketenyl acid **78**, which may also be formed as an unobserved intermediate during the neutral hydration of **26**.



Reaction of **26** with  $Br_2$  gave the fumaryl bromide **79** resulting from 1,4-addition, with the stereochemistry of **79** proven by X-ray crystallography (equation 33).³⁶ However, **79** was unstable and rearranged to the dibromofuranone **80** by an apparent ionization-recombination mechanism (equation 33).³⁶ The calculated transition structure for the reaction of **26** with  $Br_2$  showed initial coordination of  $Br_2$  to both carbonyl carbons, followed by rotation to the *trans* form.³⁶



Photolysis of alkynylcyclobutenedione **81** at  $\lambda > 338$  nm gave rise to ketenyl IR absorption at 2115 cm⁻¹, attributed to hexaketene **82** (equation 34).³⁷ Further irradiation at  $\lambda > 280$  nm resulted in decarbonylation and possible formation of *cyclo*-C₁₈ (**83**) (equation 35).³⁷ A molecule C₆₀H₆(CO)₁₂ containing six cyclobutenedione units on a bicyclic scaffold gave a negative-ion (ICR-LD) mass spectrum containing (among others) the C₆₀⁻⁻ ion, suggesting a decarbonylation pathway analogous to that of equation 34.³⁸



Photolysis of cyclobutenedione 84 in the presence of cyclopentadiene led to spiro adduct 88. This could form either by [2 + 2] cycloaddition of the intermediate

ketene **85**, forming **86** (equation 36), which rearranged with ring opening directly to **88**, or through an intermediate zwitterion **87** (equation 37).³⁹ Other cyclobutenediones and cyclobutanones⁴⁰ behaved similarly. Oxocarbene intermediates such as **89** isomeric with ketene **85** have been considered as reaction intermediates but have not been observed directly.



Photolysis of benzocyclobutenedione **90** formed the bisketene **33**,^{41–43} detected by UV absorption at 380 nm⁴³ and by IR absorption at 2077 and 2138 cm⁻¹ in a matrix at 11 K (equation 38).⁴² Photolysis of **90** in solution using TRIR spectroscopy also gave **33**, with IR bands at 2072 and 2128 cm⁻¹, and reformation of **90** with a rate constant (Table 4.6) for ring closure of **90** back to **33** of  $1.9 \times 10^4 \text{ s}^{-1}$ , as measured by IR (equation 38).^{20,21,43} In the presence of pyridine the ylide **91** with characteristic UV absorption was observed, and was attributed to capture of carbene **90a**, or reaction with **33** (equation 38).^{43,44} High-intensity (laser jet) photolysis of the trione **92** in benzene formed the bisketene **33**, leading to an 11% observed yield of the dione **90**.⁴⁵ Photochemical generation of **33** in MeOH led to formation of the diester **93** as well as the products **94** and **95** from addition of a single MeOH (equation 39).⁴⁴





Photolysis of **92** in alcohol solutions was interpreted as involving the intermediates **96** or **97**, which gave the observed product **98** (equation 40).⁴⁶ Photolysis of **90** led to the formation of bisketene **33**, as indicated by TRIR spectroscopy,^{21,47} but no absorption for the carbene **90a** was detected. The formation of **91** from **33** upon addition of pyridine argues against the intervention of the carbene **90a**, which has not been detected directly.⁴⁷ High-intensity laser photolysis of the diazo diketone **100** in CH₃OH was proposed to lead to **101** and then to **102** on the basis of capture of the corresponding esters (equation 41).⁴⁸



Photolysis of **90** forming **33** in the presence of the stable free radical NO gave **104**, proposed to form from nitroxyl radical **103** (equation 42).⁴⁹ Reaction of **33** with NO₂ by a [4 + 2] cycloaddition path was followed by loss of NO to give phthalic anhydride.⁴⁹ The bisketene **33** generated by photolysis reacted with C₆₀ to form the cycloaddition product **104a** (equation 42a).⁵⁰



Electrochemical reduction of terephthalyl chloride **105** was proposed to form the bisketene **106**, which led to the observed polymer (equation 43).⁵¹ The aza and diaza analogues **107** and **108** were suggested to form by similar routes.⁵¹



Photolysis of [2.2]paracyclophane-1,10-dione (**109**) in Ar at 10 K led to *p*-xylylene (**110**) and 1,4-dicarbonyl-2,5-cyclohexadiene (**106**), as identified by its IR band at 2082 cm⁻¹ (equation 44).^{52,52a} Further photolysis of **106** gave decarbonylation, but no further photo product could be identified. The RHF/6-31G (d,p) calculated structure of **106** gave the following bond distances (Å): C₁O (1.147), C₁C₂ (1.314), C₂C₃ (1.473), C₃C₄ (1.324), C₃H (1.075) and angles (degrees): C₁C₂C₃ (117.8°), C₂C₃C₄ (121.1°), C₃C₄H (120.1°).⁵² The calculated ketenyl IR band was at 2077 cm⁻¹. This was in contrast to the isomeric 1,2-bisketene **33**, which showed IR bands at 2077 and 2138 cm⁻¹, assigned to the coupled symmetric and antisymmetric vibrations of the ketene moieties, respectively.⁵² For **106** the coupled vibration showed zero intensity.⁵²



The bisketene **112** was obtained as an isolable solid by dehydrochlorination of the acyl chloride **111**,⁵³ and upon photolysis in argon at 10 K led to the carbene ketene **113**, identified by the strong IR absorption at  $2105.8 \text{ cm}^{-1}$  (equation 45).⁵⁴ Further photolysis gave the presumed diradical intermediate **114**, which formed **115** (equation 46).⁵⁴





Reaction of 1,4-bis(diethylaminoethynyl)benzene **116** with CO₂ was proposed to form unsaturated  $\beta$ -lactone **117** (equation 47), which gave formal ring opening to bis(acylketene) **118** (equation 48).⁵⁵ This was not observed but reacted further with the ynamine by [4 + 2] cycloaddition to give observed bis(pyrone) **119** (equation 49).⁵⁵ Stepwise generation of the ketenyl groups seemed likely.⁵⁵



Thermolysis of the bis- and tris-(ethoxyethynyl)silanes formed 1,3-bisketene **121** and trisketene **122**, respectively (equation 50).⁵⁶ Both **121** and **122** were isolated as long-lived species, and *ab initio* molecular orbital calculations indicated that the ketenyl groups in these molecules were each stabilized as much by the silicon as in Me₃SiCH=C=O.⁵⁶ Analogous ethoxyalkyne decompositions gave the 1,4-bisketene **123**,⁵⁷ the 1,6-bisketene **124**,⁵⁷ and the 1,8-bisketene **125**.²²

 $(EtOC \equiv C)_{2}SiMe_{2} \xrightarrow{\Delta} Me_{2}Si \xrightarrow{C=0} c=0$   $(EtOC \equiv C)_{3}SiMe \xrightarrow{Me}Si \xrightarrow{C=0} c=0$   $121 (75\%, IR 2129, 2108 cm^{-1})$   $122 (82\%, IR 216, 2108 cm^{-1})$  (50)



The reaction of diacyl chlorides with  $Et_3N$  followed by hydrolysis and decarboxylation under basic conditions (Blomquist reaction)⁵⁸ has been applied with a 1:1 mixture of suberyl chloride (**126**) and sebacyl chloride (**127**) to give the mixed diketone **130** arising from the mixed dimer **129** together those from the homodimerization of the individual diacyl chlorides (equations 51, 52).⁵⁹ These reactions can be formulated as involving bisketenes **7** and **128** but may proceed through a stepwise formation of monoketenes instead.



Dehydrochlorination of **133** with 1,8-bis(dimethylamino)naphthalene (**16**) as a stochiometric base and Et₃N as a catalytic shuttle base in toluene, as described above (equations 5–7),¹¹ gave 1,2-bis(ketenylidene)ethylene *E*-**134**, identified by IR absorption at 2118 cm⁻¹ (equation 53).⁶⁰



Photolyses of bis(diazo ketones) in the presence of nucleophiles may be represented as forming bisketenes. For example, the conversion of **135** could form **136**, which was trapped as the diester **137** (equation 54),⁶¹ and in the pagodane system photolysis of the bis(diazo ketone) **138** formally gave the bisketene **139**, which was trappped as **140** (equation 55),⁶² but these reactions were probably largely sequential. As noted above, bisketenes have been observed from photolysis of bis(diazo ketones) in the absence of trapping agents and directly observed in solution by IR (equation 6).¹¹



Thermal Wolff rearrangement of the bis(diazo ketones) **142** led to products **144** that are formally derived from bisketenes **143** (equation 56).⁶³



The aryl bisketenes **145** and **146** were generated by Wolff rearrangement and bis(dehydrochlorination), respectively, observed by IR, and trapped with TEMPO as the tetraaddition products.⁶⁴ Norbornadiene reacted by [4 + 2] cycloadditions with two molecules of **147** followed by loss of two N₂ molecules to form the isolable bisketene **148** (equation 57).⁶⁵



Reaction of diacyl chloride **149** with  $Et_3N$  gave ketene **150** as an isolable but sensitive yellow solid (equation 58).⁶⁶ The ketene polymerized to give a polymer containing cyclobutanedione and acid anhydride units that formed **151** upon reaction with CH₃OH (equation 59).⁶⁷



Other bisketenes are noted in Sections 3.4.3, 3.5, and 4.11.

Photolysis at  $\lambda > 300$  nm of benzoquinone (**152**) in an argon matrix at 10 K gave Z-bisketenylethylene (**134**), identified by its IR bands at 2105 and 2115 cm⁻¹, and further photolysis at 254 nm resulted in decarbonylation and formation of cyclopentadienone **156**, possibly by formation of the zwitterion **154** or the carbene **155** (equation 60).⁶⁸ Photolysis of the di-*tert*-butylquinone **157** gave the bisketene **158**, identified by its IR absorption at 2173 cm⁻¹, and this formed the isolable cyclopentadienone **159** (equation 61).⁶⁹



The bisdioxinones **160** and **163** are potential precursors to bisketenes **161** and **164**, respectively, but **160** upon heating with *tert*-butanol was shown to react sequentially to form **162** (equation 62).⁷⁰ Heating of bis(dioxinone) **163** in chlorobenzene gave products ascribed to intramolecular reactions of the bisketene **164** (equation 63).⁷¹ The bis(cyclohexadienone) **165** gave **167** upon photolysis in methanol. This may have involved bisketene **166** on photolysis or sequential

reaction as the major pathway (equations 64, 65).⁷²



As noted in Section 3.3.2, photolysis of **168** in an Ar matrix at 17 K gave the cyclic diacylbisketene **169**, with IR bands at 2101 and 1747 cm⁻¹ (equation 65a).^{72a}



Reaction of the Pt ketenyl complex **168** gave the unique bisketene **169** (equation 66; see also Section 3.8).⁷³ The electronic structure of **169** has been examined by quasi-relativistic density functional computations.⁷⁴

 $\begin{array}{c} (C_2H_4) \\ CI \\ Ph_3P \\ Ph_3P \\ \hline \\ 168 \\ \hline \\ 168 \\ \hline \\ 168 \\ \hline \\ 169 (59\%, \\ IR 2062 \text{ cm}^{-1}) \end{array}$ 

The bis(phenyliodonium) ylide **170** on thermolysis formally produced the bisketene **171**, which upon hydration and decarboxylation led to the tetraketone **172** (equation 67).⁷⁵



Dehydration of **173** with Mukaiyama's reagent **174** gave in a formal sense the bisketene **175** (equation 68), which reacted with the bisimine **176** to give the poly(2-azetidinone) **177** (equation 69).⁷⁶



Thermolysis of the bis(dioxinone) **178** gave the polycycle **180**, which could result from the bisketene **179** (equation 70).⁷⁷



As noted in Section 3.7, computational studies of possible products  $C_{60}O_n$  that conceivably could be formed by the oxidation of the fullerene  $C_{60}$  included pentaketene  $C_{60}O_5$ , as represented by the pentabenzocyclopentyl unit fragment of  $C_{60}$  **181** forming **182** (equation 71).⁷⁸



Bis(chromium carbene) complexes may also be represented as forming bisketene intermediates. Reaction of bis(chromium complex) **183** with the dialkyne **184** formally gave the bis(ketene) **185** (equation 72), which gave calixarene **185** (equation 73).⁷⁹ Other examples are given in Section 4.8.



The bis(chromium complex) **187** on heating reacted with the diyne **188**, formally giving the bis(ketene chromium complex) **189**, which formed **190** (equation 74).⁸⁰ As described in Section 3.5, an analogue of **187** was used to form a complexed bisketene that gave stereoselective bis[2 + 2] cycloaddition with a chiral imine. Other reactions of bis(ketene chromium complexes) are given

in Section 3.5, and reactions of bis(diazo ketones) are given in Section 3.3.2.



# **REFERENCES FOR SECTION 4.9**

- 1. Dixon, D. A.; Anduengo, A. J., III; Dobbs, K. D.; Khasnis, D. V. *Tetrahedron Lett.* **1995**, *36*, 645–648.
- 2. Glaser, R.; Murmann, R. K.; Barnes, C. L. J. Org. Chem. 1996, 61, 1047-1058.
- Schröder, D.; Heinemann, C.; Schwarz, H.; Harvey, J. N.; Dua, S.; Blanksby, S. J.; Bowie, J. H. Chem. Eur. J. 1998, 4, 2550–2557.
- 4. Diels, O.; Wolf, B. Chem. Ber. 1906, 39, 689-697.
- 5. Staudinger, H.; Kreis, W. Helv. Chim. Acta 1923, 39, 689-697.
- 6. Maier, G.; Reisenauer, H. P.; Sayrac, T. Chem. Ber. 1982, 115, 2192-2201.
- 7. Kasai, M.; Oda, M.; Kitahara, Y. Chem. Lett. 1978, 217-218.
- 8. Allen, A. D.; Tidwell, T. T. Can. J. Chem. 1999, 77, 802-805.
- de Meijere, A.; Kozhushkov, I. Methods of Organic Chemistry (Houben-Weyl); Thieme, Stuttgart, 1997, Vol. E 17f, pp. 999–126.
- Butenschön, H. Methods of Organic Chemistry (Houben-Weyl); Thieme, Stuttgart, 1997, Vol. E 17f, pp. 127–140.
- Budzelaar, P. H. M.; Kraka E.; Cremer, D.; Schleyer, P. v. R. J. Am. Chem. Soc. 1986, 108, 561–567.
- 9a. McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1994, 116, 7233-7238.
- 10. Allen, A. D.; Lough, A. J.; Tidwell, T. T. Chem. Commun. 1996, 2171-2172.
- 10a. Werstiuk, N. H.; Ma, J.; McAllister, M. A.; Tidwell, T. T.; Zhao, D.-c. J. Chem. Soc. Faraday Trans. 1994, 90, 3383–3390.
- Allen, A. D.; Fenwick, M. H.; Jabri, A.; Rangwala, H.; Saidi, K.; Tidwell, T. T. Org. Lett. 2001, 3, 4095–4098.
- 12. Mallory, F. B.; Roberts, J. D. J. Am. Chem. Soc. 1961, 83, 393-397.
- 13. Blomquist, A. T.; LaLancette, E. A. J. Am. Chem. Soc. 1961, 83 1387-1391.

- 14. Chapman, O. L.; McIntosh, C. L.; Barber, L. L. J. Chem. Commun. 1971, 1162–1163.
- 15. Obata, N.; Takizawa, T. J. Chem. Soc. D, Chem. Commun. 1971, 587-588.
- 16. Obata, N.; Takizawa, T. Bull. Chem. Soc. Jpn. 1977, 50, 2017-2020.
- 17. Zhao, D.; Allen, A, D.; Tidwell, T. T. J. Am. Chem. Soc. 1993, 115, 10097-10103.
- Allen, A, D.; McAllister, M. A.; Tidwell, T. T.; Zhao, D. Acc. Chem. Res. 1995, 115, 265–271.
- 19. Liu, R.; Tidwell, T. T. J. Chem. Soc. Perkin Trans. 2 1996, 2757-2762.
- Allen, A. D.; Colomvakos, J. D.; Egle, I.; Lusztyk, J.; McAllister, M. A.; Tidwell, T. T.; Wagner, B. D.; Zhao, D.-c. J. Am. Chem. Soc. 1995, 117, 7552–7553.
- Allen, A. D.; Colomvakos, J. D.; Diederich, F.; Egle, I.; Lusztyk, J.; McAllister, M. A.; Rubin, Y.; Tidwell, T. T.; Wagner, B. D.; Zhao, D.-c. J. Am. Chem. Soc. 1997, 119, 12045–12050.
- 22. Liu, R.; Marra, R. M.; Tidwell, T. T. J. Org. Chem. 1996, 61, 6227-6232.
- 23. Maier, G.; Rohr, C. Liebigs Ann. 1996, 307-309.
- 24. Liu, R.; Tidwell, T. T. Can. J. Chem. 1995, 73, 1818-1822.
- 25. Egle, I.; Lai, W.-Y.; Moore, P. A.; Renton, P.; Tidwell, T. T.; Zhao, D.-c. *J. Org. Chem.* **1997**, *62*, 18–25.
- 26. Dejmek, M. M.; Selke, R. Angew. Chem., Int. Ed. 1998, 37, 1540-1542.
- 27. Dejmek, M. M.; Selke, R. Synlett 2000, 13-21.
- Kondo, T.; Kaneko, Y.; Taguchi, Y.; Nakamura, A.; Okada, T.; Shiotsuki, M.; Ura, Y.; Wada, K.; Mitsudo, T. J. Am. Chem. Soc. 2002, 124, 6824–6825.
- Allen, A. D.; Moore, P. A.; Missiha, S.; Tidwell, T. T. J. Org. Chem. 1999, 64, 4690–4696.
- Huang, W.; Moore, P. A.; Rafai Far, A.; Tidwell, T. T. J. Org. Chem. 2000, 65, 3877– 3879.
- 31. Allen, A. D.; Huang, W.; Moore, P. A.; Rafai Far, A.; Tidwell, T. T. J. Org. Chem. 2000, 65, 5676–5679.
- 32. Colomvakos, J. D.; Egle, I.; Ma, J.; Pole, D. L.; Tidwell, T. T.; Warkentin, J. D. J. Org. Chem. **1996**, *61*, 9522–9527.
- 33. Yin, J.; Liebeskind, L. S. J. Org. Chem. 1998, 63, 5726-5727.
- Jewell, C. F., Jr.; Liebeskind, L.; Williamson, M. J. Am. Chem. Soc. 1985, 107, 6715– 6716.
- Allen, A. D.; Ma, J.; McAllister, M. A.; Tidwell, T. T.; Zhao, D.-c. J. Chem. Soc., Perkin Trans. 2 1995, 847–851.
- 36. Brown, R. S.; Christl, M.; Lough, A. J.; Ma, J.; Peters, E.-M.; Peters, K.; Sammtleben, F.; Sung, K.; Slebocka-Tilk, H.; Tidwell, T. T. J. Org. Chem. 1998, 63, 6000–6006.
- 37. Diederich, F.; Rubin, Y.; Chapman, O. L.; Goroff, N. S. *Helv. Chim. Acta* **1994**, 77, 1441–1457.
- 38. Rubin, Y.; Parker, T. C.; Pastor, S. J.; Jalisatgi, S.; Boulle, C.; Wilkins, C. L. Angew. Chem. Int. Ed. 1998, 37, 1226–1229.
- 39. Miller, R. D.; Kirchmeyer, S. J. Org. Chem. 1993, 58, 90-94.
- 40. Miller, R. D.; Theis, W.; Heilig, G.; Kirchmeyer, S. J. Org. Chem. 1991, 56, 1453-1463.
- 41. Staab, H. A.; Ipaktschi, J. Chem. Ber. 1968, 101, 1457-1472.
- 42. Chapman, O. L.; Mattes, K.; McIntosh, C. L.; Pacansky, J.; Calder, G. V.; Orr, G. J. Am. Chem. Soc. **1973**, 95, 6134–6135.
- Boate, D. R.; Johnston, L. J.; Kwomg, P. C.; Lee-Ruff, E.; Scaiano, J. C. J. Am. Chem. Soc. 1990, 112, 8858–8863.
- 44. Mosandl, T.; Wentrup, C. J. Org. Chem. 1993, 58, 747-749.
- 45. Adam, W.; Patterson, W. S. J. Org. Chem. 1995, 60, 7769-7773.
- 46. Tatsugi, J.; Hara, T.; Izawa, Y.; Chem. Lett. 1997, 177-178.
- 47. Oishi, S.; Ozaki, J. Chem. Lett. 1998, 1071.
- 48. Banks, J. T.; Scaiano, J. C. J. Photochem. Photobiol. A: Chemistry, 1996, 96, 31-33.
- 49. Adam, W.; Bottle, S. E.; Grice, I. D.; Pfeiler, D.; Wentrup, C. *Tetrahedron Lett.* **1996**, *37*, 2113–2116.
- 50. Tomioka, K.; Yamamoto, K. J. Chem. Soc. Chem. Commun. 1995, 1961-1962.
- 51. Uttley, J. H. P.; Gao, Y.; Lines, R. J. Chem. Soc. Chem. Commun. 1993, 1540-1542.
- 52. Marquardt, R.; Sander, W.; Laue, T.; Hopf, H. Liebigs Ann. 1995, 1643-1648.
- 52a. Sander, W.; Marquardt, R.; Bucher, G.; Wandel, H. Pure Appl. Chem. 1996, 68, 353-356.
  - 53. Blomquist, A. T.; Meinwald, Y. C. J. Am. Chem. Soc. 1957, 79, 2021-2022.
  - 54. Wenk, H. H.; Sander, W. Eur. J. Org. Chem. 1999, 57-60.
  - 55. Tsuda, T.; Hokazono, H.; Toyota, K. J. Chem. Soc. Chem. Commun. 1995, 2417-2418.
  - 56. Sung, K.; Tidwell, T. T. J. Am. Chem. Soc. 1996, 118, 2768-2769.
  - 57. Sung, K.; Tidwell, T. T. Organometallics 1997, 16, 78-85.
  - 58. Blomquist, A. T.; Prager, J.; Wolinsky, J. J. Am. Chem. Soc. 1955, 77, 1804–1806.
  - Gudmundsdottir, A. D.; Lewis, T. J.; Randall, L. H.; Scheffer, J. R.; Rettig, S. J.; Trotter, J.; Wu, C.-H. J. Am. Chem. Soc. 1996, 118, 6167–6184.
  - 60. Tidwell, T. T.; Fenwick, M. H. Eur. J. Org. Chem. 2001, 3414-3419.
  - 61. Gleiter, R.; Krennrich, G. Angew. Chem. Int. Ed. Engl. 1986, 25, 449-450.
  - 62. Melder, J.-P.; Fritz, H.; Prinzbach, H. Angew. Chem. Int. Ed. Engl. 1989, 28, 300-303.
  - 63. Rubin, M. B.; Bargurie, M.; Kosti, S.; Kaftory, M. J. Chem. Soc. Perkin Trans. 2 1980, 2670–2677.
  - 64. Allen, A. D.; Rangwala, H.; Saidi, K.; Tidwell, T. T.; Wang, J. *Russ. Chem. Bull.* 2001, 2130–2133.
  - 65. Christl, M.; Lanzendörfer, U.; Grötsch, M. M.; Ditterich, E.; Hegmann, J. *Chem. Ber.* **1990**, *123*, 2031–2037.
  - 66. Hatchard, W. R.; Schneider, A. K. J. Am. Chem. Soc. 1957, 79, 6261-6263.
  - 67. Naka, K.; Uemura, T.; Chujo, Y. Polymer Bull. 1999, 42, 367-372
  - Maier, G.; Franz, L. H.; Hartan, H.-G.; Lanz, K.; Reisenauer, H. P. Chem. Ber. 1985, 118, 3196–3204.
  - 69. Alkenings, B.; Betterman, H.; Dasting, I.; Schroers, H.-J. *Spectrochim Acta* **1993**, *49A*, 315–320.
  - 70. Kiegiel, J.; Józwik, J.; Wozniak, K.; Jurczak, J. Tetrahedron Lett. 2000, 41, 4959–4963.
  - 71. Kwon, T. W.; Song, S. J.; Kwon, Y. U.; Chung, S. K. Bull. Korean Chem. Soc. 2003, 24, 229–231.
  - 72. Quinkert, G.; Schmieder, K. R.; Dürner, G.; Hache, K.; Stegk, A.; Barton, D. H. R. *Chem. Ber.* **1977**, *110*, 3582–3614.

- 72a. Ohana, T.; Yabe, A. Chem. Lett. 1996, 325-326.
- Bertani, R.; Casarin, M.; Ganis, P.; Maccato, C.; Pandolfo, L.; Venzo, A.; Vittadini, A.; Zanotto, L. Organometallics 2000, 19, 1373–1383.
- 74. Casarin, M.; Pandolfo, L. J. Organomet. Chem. 2003, 682, 255-259.
- 75. Spyroudis, S.; Xanthopoulou, N. Arkivoc 2003 (N-630A), 95-105.
- 76. Sato, A.; Endo, T. Macromolecules 1998, 31, 7996-7998.
- 77. Sato, M.; Oda, T.; Iwamoto, K.-I.; Fujii, S. Heterocycles, 2003, 60, 899-908.
- 78. Kepert, D. L.; Clare, B. W. Inorg. Chim. Acta 2002, 327, 41-53.
- 79. Gopalsamuthiram, V.; Wulff, W. D. J. Am. Chem. Soc. 2004, 126, 13936-13937.
- Bao, J.; Wulff, W. D.; Fumo, M. J.; Grant, E. B.; Heller, D. P.; Whitcomb, M. C.; Yeung, S.-M. J. Am. Chem. Soc. 1996, 118, 2166–2181.

# 4.10 KETENYL RADICALS, ANIONS, AND CATIONS

Ketenyl radicals, anions, and cations are considered to be structures with charge or radical character at  $C_2$ , as in 1–3, respectively. Structure 2 is the ynolate ion and is also considered in Section 4.8.1. Electron transfer reactions of ketenes are considered in Section 5.1, and ketenes with charged, radical, or carbenic side chains are discussed in Section 4.1.9.

$$H\dot{C}=C=O$$
  $H\ddot{C}=C=O$   $H\ddot{C}=C=O$   
1 2 3

The ketenyl radical **1** was produced by the laser photolysis of ketene with 193 nm light,^{1,2} and was characterized by the measurement of the high-resolution IR spectrum¹ and examination by laser-induced fluorescence spectroscopy (equation 1).² The kinetics of reaction of **1** with NO were measured.¹ Computations indicated that the ketenyl radical **1** had a bent structure and that the ethynyloxy radical structure **1a** was not important.^{3,4} The chemistry of **1** is of major significance, as it is implicated in the combustion of acetylene and the formation of soot. The formation of **1** and **3** in the combustion of acetylene has been evaluated by DFT calculations.⁵ The gas phase acidity of **1** leading to the radical anion **1b** was also determined.⁶

The ketenyl radical **1** was generated in the gas phase by photolysis of ketene and reacted with NO to form CO and CO₂ (equation 2).⁷ The product branching ratio for formation of CO and CO₂ was 88:12, and the formation of CO was suggested to proceed via nitrosoketene (**4**) as a transient intermediate.⁷ Photolysis of the diazo ketone **5a** in a matrix at 77 K gave rise to an ESR spectrum attributed to the phenylketenyl radical **6** formed by phenyl radical expulsion from the keto

carbene 5 (equation 3). 8 

$$\overset{H}{\underset{H}{\longrightarrow}} C=0 \xrightarrow{h_{V}} \overset{h_{V}}{\underset{H}{\longrightarrow}} \overset{C}{\underset{H}{\longrightarrow}} C=0 \xrightarrow{NO} \overset{ON}{\underset{H}{\longrightarrow}} C=0 \xrightarrow{CO_{2}} + (HCN) \text{ or } CO + (HCNO)$$

$$1 \qquad 4 \qquad (2)$$

The ynolate ion **2** has been readily generated in the gas phase by the reaction of ketene with bases (equation 4),⁹ and the bond dissociation energy and electron affinity have been determined.^{9a} The ynolate anion was found to be highly stabilized, with a linear structure with the charge concentrated on oxygen.¹⁰ IR multiple-photon activation of enolate ions in the gas phase was interpreted as causing formation of an anion-ketene complex **7** that formed ynolate anion (equation 5; see also Section 4.8.1).¹¹

$$\overset{H}{\underset{H}{\rightarrowtail}} C=0 \xrightarrow{\text{base}} H-\overline{C}=C=0 \iff H-C\equiv C-O^{-}$$
(4)

$$\begin{array}{cccc} & & & & & \\ & & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & & \\ & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ &$$

The hydride affinity of the ketenyl cation **3** was found to be 308 kcal/mol (equation 6),¹² and the activation energy for formation of **3** from the radical cation **8** of ethynol has been estimated as 68 kcal/mol (equation 7).¹³ The cyano ketenyl cation **9** was proposed to form from  $C_3N^+$  (equation 8).¹⁴ Formation of the trimethylsilyl-substituted ketenyl cation **11** was proposed to occur upon photolysis of the diazo ester **10**, as evidenced by formation of the methyl ester **12** upon reaction with MeOH (equation 9).¹⁵

$$\stackrel{H}{\longrightarrow} C=O \xrightarrow{\Delta H^{\circ}_{298}=} H^{-}C=C=O + H^{-}$$

$$308 \text{ kcal/mol} \qquad 3$$
(6)

$$\begin{array}{cccc} H-C\equiv C-OH^{+\bullet} & \longrightarrow & H-\overset{+}{C}=C=O & + & H^{\bullet} \\ 8 & & 3 \end{array}$$
(7)

$$^{+}N=C=C=C: \xrightarrow{O_{2}} N\equiv C-\overset{+}{C}=C=O$$

$$9$$

$$(8)$$



The cationic species 13 and 15 were generated in the gas phase, and 13 underwent decarbonylation to 14, while 15 reacted with pyridine to form 16, which rapidly lost CO (equations 10, 11).¹⁶

$$0 = C = \stackrel{+}{N} = C = 0 \xrightarrow{-C0} \stackrel{+}{N} = C = 0 \qquad (10)$$

$$0 = \stackrel{+}{N} = C = C = 0 + \swarrow \stackrel{N}{N} \xrightarrow{-C} \stackrel{N+}{N} \xrightarrow{-C} = C = 0 \qquad (11)$$

$$15 \qquad 0 \stackrel{+}{N} \xrightarrow{-C} = C = 0 \qquad (11)$$

$$16 \qquad 0 \stackrel{+}{N} \xrightarrow{-C} = C = 0 \qquad (11)$$

The structures and electronic configurations of the radicals **17** and **18** have been determined by high-level computational studies.¹⁷



#### **REFERENCES FOR SECTION 4.10**

- 1. Unfried, K. G.; Glass, G. P.; Curl, R. F. Chem. Phys. Lett. 1991, 177, 33-38.
- Brock, L. R.; Mischler, B.; Rohlfing, E. A.; Bise, R. T.; Neumark, D. M. J. Chem. Phys. 1997, 107, 665–668.
- Hu, C.-H.; Schaeffer, H. F., III; Hou, Z.; Bayes, K. D. J. Am. Chem. Soc. 1993, 115, 6904–6907.
- 4. Harding, L. B. J. Phys. Chem. 1981, 85, 10-11.
- 5. Sheng, C.; Bozzelli, J. W. Int. J. Chem. Kinet. 2000, 32, 623-641.
- Van Doren, J. M.; Miller, T. M.; Miller, A. E. S.; Viggiano, A. A.; Morris, R. A.; Paulson, J. F. J. Am. Chem. Soc. 1993, 115, 7407–7414.
- 7. Rim, K. T.; Hershberger, J. F. J. Phys. Chem. A 2000, 104, 293-296.
- Murai, H.; Sarfarik, I.; Torres, M.; Strausz, O. P. J. Am. Chem. Soc. 1988, 110, 1025– 1032.
- Oakes, J. M.; Jones, M. E.; Bierbaum, V. M.; Ellison, G. B. J. Phys. Chem. 1983, 87, 4810–4815.
- 9a. Berkowitz, J.; Ellison, G. B.; Gutman, D. J. J. Phys. Chem. 1994, 98, 2744-2765.

- 10. Smith, B. J.; Radom, L.; Kresge, A. J. J. Am. Chem. Soc. 1989, 111, 8297-8999.
- 11. Foster, R. F.; Tumas, W.; Brauman, J. I. J. Phys. Chem. 1989, 93, 61-68.
- 12. Lossing, F. P.; Holmes, J. L. J. Am. Chem. Soc. 1984, 106, 6917-6920.
- 13. Aue, D. H.; Bowers, M. T. In *Gas Phase Ion Chemistry*; Bowers, M. T., Ed.; Academic Press, New York, Vol. 2, pp.
- 14. Petrie, S.; McGrath, K. M.; Freeman, C. G.; McEwan, M. J. J. Am. Chem. Soc. 1992, 114, 9130–9136.
- 15. Ando, W.; Sekiguchi, A.; Hagiwara, T.; Migita, T.; Chowdhry, V.; Westheimer, F. H.; Kammula, S. L.; Green, M.; Jones, M., Jr. *J. Am. Chem. Soc.* **1979**, *101*, 6393–6398.
- Carvalho, M. C.; Juliano, V. F.; Kascheres, C.; Eberlin, M. N. J. Chem. Soc. Perkin Trans. 2 1997, 2347–2352.
- 17. Cooksy, A. L. J. Am. Chem. Soc. 2001, 123, 4003-4013.

#### 4.11 CUMULENONES

Cumulenones are considered to include carbon suboxide (1), methyleneketenes such as propadienone (2), and carbonic species such as 3. The coverage in this section is not intended to be comprehensive, but rather to illustrate some of the diverse structural types that have been studied. Reviews of some of these compounds have appeared.^{1,2}

$$O=C=C=C=O$$
  $CH_2=C=C=O$   $C=C=O$   
1 2 3

Carbon suboxide was first prepared in 1906 by Diels and Wolf by the dehydration of malonic acid (equation 1)³ and in 1908 by Staudinger and Bereza by dehalogenation (equation 2).⁴ The preparation and properties of **1** have been extensively studied and reviewed,⁵ and only a few recent studies are noted here.

$$\begin{array}{ccc} HO_2C & \xrightarrow{P_2O_5} & O=C=C=C=O \\ H_2 & & \mathbf{1} \end{array}$$
 (1)

$$\begin{array}{cccc} \text{CIOC} & \xrightarrow{\text{COCI}} & \xrightarrow{\text{Zn}} & \text{O=C=C=C=O} \\ & & & \text{Br} & \text{Br} & & 1 \end{array}$$
(2)

The carbon dioxides  $C_nO_2$  with even numbers of carbons are less stable than those with odd numbers, and all attempts to prepare  $C_2O_2$ , the dimer of CO, have been unsuccessful.^{6–10,13} There is evidence for  $C_4O_2$  in an argon matrix¹¹ and in the gas phase,¹² and for the odd-numbered species  $C_5O_2^{14}$  and  $C_7O_2^{15}$  in matrices.

Carbon suboxide,  $C_3O_2$ , has a quasilinear structure with a very small energy barrier for bending around the central carbon, as discussed in Section 1.2. The structure of crystalline  $C_3O_2$  has been determined by X-ray.¹⁶

Carbon suboxide undergoes spontaneous polymerization, forming a red to black solid found to have a degree of polymerization  $n \approx 40$ , with the polypyronic structure **4** as determined by X-ray (equation 3).¹⁷



Ab initio calculations indicated that carbon suboxide is preferentially protonated on the central carbon to give **5**, which has a bent geometry (equation 4).^{18,19} The proton affinity of 200 kcal/mol (791 kJ mol⁻¹) was determined¹⁹ by Fourier transform mass spectrometry and was in good agreement with the calculated value of 200 kcal/mol (789 kJ/mol).¹⁹ The oxygen-protonated isomer (O=C=C=C= OH⁺) and the radical (O=C=CH–C=O•) have also been observed.¹⁹ The generation of the complex of **1** and HCl from photolysis of matrix-isolated O=C=CHCOCl is noted in Section 5.6.1. Complexation of pyridine with C₃O₂ in a matrix has been found computationally and experimentally to produce Tshaped complexes, but not a zwitterion.²⁰ The use of **1** in synthesis has been briefly reviewed in the context of malonate chemistry.²¹



Ion **5** was also formed in the gas phase by electron ionization of methyl trifluoroacetoacetate. One route proposed for this conversion involved successive loss of  $CF_{3^{\bullet}}$  and methanol (equation 5).²²

$$\begin{bmatrix} 0 & 0 \\ CF_3 & OMe \end{bmatrix}^{+\bullet} \xrightarrow{-CF_3 \bullet} 0 = \stackrel{+}{CCH} = C \xrightarrow{OH} \xrightarrow{-CH_3OH} O \xrightarrow{C} C + C \xrightarrow{C} O \xrightarrow{O} (5)$$

Upon photolysis in the gas phase carbon suboxide formed the triplet carbenic species :C=C=O (3), which reacted with ethylene, forming cyclopropylideneke-tene 6, which underwent decarbonylation, forming the carbene 7, which led to allene (equation 6).²³⁻²⁶ The triplet electronic state of 3 was confirmed by direct

observation in a matrix.²⁷

$$0=C=C=C=0 \xrightarrow{hv} : C=C=0 \xrightarrow{CH_2=CH_2} \searrow C=0 \xrightarrow{-CO} \bigcirc C: \longrightarrow CH_2=C=CH_2$$

$$1 \qquad 3 \qquad 6 \qquad 7 \qquad (6)$$

Upon photolysis of 1 in the gas phase generating 3 in the presence of tetramethylethylene, the ketene 8 was formed, as shown by the isolation of the acid 9 from hydration and the dimer 10 (equation 7).²⁸ The formation of 8 and 10 by a different route is discussed in Section 4.1.1. The use of 3 in the formation of tetrahedrane as a transient intermediate has also been reported.²⁹



Reaction of carbon suboxide with propene radical cation was proposed to proceed by initial addition forming the distonic ion **11**, followed by loss of two CO's and rearrangement to give the radical cation of 1-butyne (equation 8).³⁰

$$CH_{3}CH = CH_{2}^{+} \cdot \xrightarrow{C_{3}O_{2}} \xrightarrow{CH_{3}} \xrightarrow{C} \stackrel{O}{\leftarrow} \xrightarrow{-2CO} CH_{3}CH_{2}C \equiv CH^{+} \cdot$$
(8)

Computations of the structures and energies of the cumulenones 2, 12, and 13 have been reported.³¹ At the highest levels of calculation, all of these structures showed significant bending in the heavy-atom chain. Dimethylpropadienone 16, with an IR absorption at 2124 cm⁻¹, was obtained in a matrix by carbonylation of the carbene 15 obtained by photolysis of bisperester 14 (equation 9).³² This species was also obtained by a low-temperature dehalogenation of 17 (equation 10).³³

Propadienone (2) was formed by pyrolysis of diazo ketone 18 in a reaction suggested to give the acyl ketene 19, which lost  $CO_2$ , forming 2, which was trapped in an argon matrix and identified by IR absorption at 2125 cm⁻¹ (equation 11).³⁴ The methyleneketene 21 obtained by pyrolysis of Meldrum's acid derivative 20 was the first member of this class obtained as a stable compound in solution at room temperature and was characterized by the ¹³C NMR spectrum (equation 12).³⁵ Cumulenone 22 was prepared by a similar route and observed in solution at  $-50 \,^{\circ}C.^{36,37}$  Methylene derivatives 23 of Meldrum's acid on FVP formed surprisingly stable iminopropadienones 24 that were isolated and characterized (equation 13).³⁸ A variety of nucleophilic additions to 24 were examined.^{38,38a}



Pyrolyses of Meldrum's acids **25** were proposed to lead to thiosubstituted cumulenones **26**, which rearranged by hydrogen transfer to give ketenes **27**, which formed thiophene derivatives **28** (equation 14).^{39–41} An analogous transformation is mentioned in Section 4.1.6. Pyrolysis of **29** was proposed to form the ketene intermediate **30**, which reacted through a dihydropyridylmethyleneketene **31** (equation 15) to form the iminocumulenone **32**, identified by a strong IR band at 2247 cm⁻¹ (equation 16).⁴² Reactions of **32** with 2-methylaminopyridine gave the mesoionic derivative **33** (equation 16).⁴² Other reactions of **24** (R = neopentyl,

mesityl, 2-t-BuC₆H₄) were studied.⁴³



Pyrolysis or photolysis of phthalic anhydride gave benzyne as well as the cyclopentadienylideneketene **34**, characterized by its IR absorption at 2085 cm⁻¹ (equation 17).⁴⁴ This IR absorption had earlier been assigned to benzyne, and the structure of **34** was determined by *ab initio* molecular orbital calculations.^{45,46}



Photolysis of  $C_3O_2$  produced  $C_5O$ ,  $C_7O$ , and  $C_9O$ , and the structures of these species were determined from their rotational spectra with the assumption of linearity.⁴⁷ The radical HCCCO was calculated to have the alkynylacyl structure **35a**, 3.0 kcal/mol (12.5 kJ/mol) more stable than the ketenyl structure **35b**.^{48–50}

HC=C-C· 
$$\dot{C}$$
=C=C=C  
O H  
35a 35b

The oxime **36** of the unknown  $C_2O_2$  has been generated in an Ar matrix and identified by its IR spectrum.⁵¹ The analogue O=C=C=S has been prepared from CO and CS.⁵² The mass spectrum of  $C_3O_2$  produces  $C_2O_2^{+\bullet}$ , and the latter species has been examined by tandem mass spectroscopy.⁵³

# **REFERENCES FOR SECTION 4.11**

- 1. Brown, R. F. C.; Eastwood, F. W. In *The Chemistry of Ketenes, Allenes, and Related Compounds*; Patai, S., Ed.; Wiley, New York, 1980, pp. 757–778.
- 2. Kappe, T. Methoden der Organischen Chemie, Vol. E15. Thieme Verlag, Stuttgart, 1993.
- 3. Diels, O.; Wolf, B. Chem. Ber. 1906, 39, 689-697.
- 4. Staudinger, H.; Bereza, S. Chem. Ber. 1908, 41, 4461-4465.
- 5. Kappe, T. Encycl. Reagents Org. Synth., Wiley, New York, 1995, 996-997.
- Dixon, D. A.; Arduengo, A. J., III; Dobbs, K. D.; Khasnis, D. V. *Teterahedron Lett.* 1995, 36, 645–648.
- 7. Glaser, R.; Murmann, R. K.; Barnes, C. L. J. Org. Chem. 1996, 61, 1047-1058.
- Schröder, D.; Heinemann, C.; Schwarz, H.; Harvey, J. N.; Dua, S.; Blanksby, S. J.; Bowie, J. H. Chem. Eur. J. 1998, 4, 2550–2556.
- 9. Birney, D. M.; Berson, J. A. J. Am. Chem. Soc. 1985, 107, 4553-4554.
- 10. Berson, J. A.; Birney, D. M.; Dailey, W. P.; Liebman, J. F. In *Molecular Structure and Energetics*; VCH Publishers, New York, 1988, Vol. 6, pp.
- Maier, G.; Reisenauer, H. P.; Balli, H.; Brandt, W.; Janoschek, R. Angew. Chem. Int. Ed. Engl. 1990, 29, 905–908.
- 12. Sülze, D.; Schwarz, H. Angew. Chem. Int. Ed. Engl. 1990, 29, 908-909.
- 13. Sülze, D.; Weiske, T.; Schwarz, H. Int. J. Mass Spectrom. Ion Proc. 1993, 125, 75-79.
- 14. Maier, G.; Reisenauer, H. P.; Schäfer, U.; Balli, H. Angew. Chem. Int. Ed. Engl. 1988, 27, 566–568.
- 15. Maier, G.; Reisenauer, H. P.; Ulrich, A. Tetrahedron Lett. 1991, 32, 4469-4472.
- 16. Ellern, A.; Drews, T.; Seppelt, K. Z. Anorg. Allg. Chem. 2001, 627, 73-76.
- Ballauff, M.; Rosenfeldt, S.; Dingenouts, N.; Beck, J.; Krieger-Beck, P. Angew. Chem. Int. Ed. 2004, 43, 5843–5846.
- Allen, A. D.; McAllister, M. A.; Tidwell, T. T. J. Chem. Soc. Chem. Commun. 1995, 2547–2548.
- Tortajada, J.; Provot, G.; Morizur, J.-P.; Gal, J.-F.; Maria, P.-C.; Flammang, R.; Govaert, Y. Int. J. Mass Spectrom. Ion Proc. 1995, 141, 241–255.
- 20. Couturier-Tamburelli, I.; Aycard, J.-P.; Wong, M. W.; Wentrup, C. J. Phys. Chem. A. 2000, 104, 3466–3471.
- 21. Stadlbauer, W.; Badawey, E.-S.; Hojas, G.; Roschger, P.; Kappe, T. *Molecules* **2001**, *6*, 338–352.

- Tajima, S.; Watanable, D.; Ubukata, M.; Hiroi, Y.; Nakajima, S.; Nibbering, N. M. M. Int. J. Mass Spectrom. Ion Proc. 2002, 219, 475–483.
- 23. Bayes, K. J. Am. Chem. Soc. 1961, 83, 3712-3713.
- 24. Williamson, D. G.; Bayes, K. D. J. Am. Chem. Soc. 1968, 90, 1957-1965.
- Minato, T.; Osamura, Y.; Yamabe, S.; Fukui, K. J. Am. Chem. Soc. 1980, 102, 581– 589.
- McFarlane, J.; Polanyi, J. C.; Shapter, J. G.; Williamson, J. M. J. Photochem. Photobiol., A: Chem. 1989, 46, 139–158.
- 27. Milligan, D.; Jacox, M. J. Chem. Phys. 1965, 43, 3734-3746.
- Parker, G.; Wiseman, D.; Wintner, C.; MacKay, C. J. Org. Chem. 1984, 49, 4494– 4497.
- 29. Shevlin, P. B.; Wolf, A. P. J. Am. Chem. Soc. 1970, 92, 406-408.
- Pandolfo, L.; Somogyi, Á.; Catinella, S.; Traldi, P. J. Mass Spectrom. 1995, 30, 1049– 1050.
- 31. Scott, A. P.; Radom, L. J. Mol. Struct. 2000, 556, 253-261.
- Reed, S. C.; Capitosti, G. J.; Zhu, Z.; Modarelli, D. A. J. Org. Chem. 2001, 66, 287– 299.
- 33. Masters, A. P.; Sorensen, T. S.; Tran, P. M. Can. J. Chem. 1987, 65, 1499-1502.
- Chapman, O. L.; Miller, M. D.; Pitzenberger, S. M. J. Am. Chem. Soc. 1987, 109, 6867– 6868.
- Lorencak, P.; Pommelet, J. C.; Chuche, J.; Wentrup, C. J. Chem. Soc. Chem. Commun. 1986, 369–370.
- Wentrup, C.; Kambouris, P.; Evans, R. A.; Owen, D.; Macfarlane, G.; Chuche, J.; Pommelet, J. C.; Ben Cheikh, A.; Plisnier, M.; Flammang, R. J. Am. Chem. Soc. 1991, 113, 3130–3135.
- Ben Cheikh, A.; Pommelet, J. C.; Chuche, J. J. Chem. Soc. Chem. Commun. 1990, 615–617.
- Mosandl, T.; Kappe, C. O.; Flammang, R.; Wentrup, C. J. Chem. Soc. Chem. Commun. 1992, 1571–1573.
- 38a. Shtaiwi, M.; Wentrup, C. J. Org. Chem. 2002, 67, 8558-8565.
- 39. Hunter, G. A.; McNab, H. J. Chem. Soc. Perkin Trans. 1 1995, 1209-1214.
- 40. McNab, H.; Morrow, M. Arkivoc 2002 (viii), 125-133.
- 41. Plüg, C.; Frank, W.; Wentrup, C. J. Chem. Soc. Perkin Trans. 2 1999, 1087-1093.
- Andersen, H. G.; Mitschke, U.; Wentrup, C. J. Chem. Soc. Perkin Trans. 2 2001, 602– 607.
- Bibas, H.; Moloney, D. W. J.; Neumann, R.; Shtaiwi, M.; Bernhardt, P. V.; Wentrup, C. J. Org. Chem. 2002, 67, 2619–2631.
- Simon, J. G. G.; Münzel, N.; Schweig, A. Chem. Phys. Lett. 1990, 170, 187– 192.
- 45. Schreiner, A. C.; Schaefer, H. F., III. J. Am. Chem. Soc. 1992, 114, 4758-4762.
- 46. Scott, A. P.; Radom, L. Chem. Phys. Lett. 1992, 200, 15-20.
- 47. Ogata, T.; Ohshima, Y.; Endo, Y. J. Am. Chem. Soc. 1995, 117, 3593-3598.
- 48. Wang, H.; Cooksy, A. L. Chem. Phys. 1996, 213, 139-151.

- 49. Cooksy, A. L.; Tao, F.-M.; Klemperer, W.; Thaddeus, P. J. Phys. Chem. 1995, 99, 11095–11100.
- 50. Cooksy, A. L. J. Am. Chem. Soc. 1995, 117, 1098-1104.
- 51. Maier, G.; Reisenauer, H. P.; Röther, B.; Eckwert, J. Liebigs Ann. 1996, 303-306.
- 52. Maier, G.; Reisenauer, H. P.; Ruppel, R. Angew. Chem., Int. Ed. Engl. 1997, 36, 1862– 1864.
- 53. Dawson, D.; Chen, H. I.; Holmes, J. C. Eur. Mass Spectrom. 1996, 2, 373-375.

# **5** Reactions of Ketenes

# 5.1 OXIDATION AND REDUCTION OF KETENES

Ketenes are susceptible to oxidation forming the radical cation 1 (equation 1),¹ and the photoelectron spectra of  $CH_2=C=O$  and  $CD_2=C=O$  forming  $CH_2=C=O^{+\bullet}$ and  $CD_2=C=O^{+\bullet}$  have been measured and the vibrational energies of the ions derived.² This process was favored by the relative ease of electron removal from the ketene HOMO and the stabilization of the acylium ion resonance structure **1b**. Addition of an electron to the high-lying LUMO is, however, quite difficult. The oxirene radical cation **1d** has been prepared from several sources as a stable species in the gas phase, but did not yield neutral oxirene by charge neutralization.³



The ion  $CH_2=C=O^{+\bullet}$  was also generated from the gas phase reaction of protonated glycine (equation 2).⁴ The ion with m/z 118 in the 70 eV electron impact mass spectrum of (*N*-phenylacetyl)piperidin-2-one (3) has been identified as phenylketene radical cation 4 by its mass-analyzed kinetic energy spectrum (equation 3).⁵ The interconversion of HOCH=C=O^{+•} and HOCH=C⁻O⁻ is noted in Section 2.4.



Ketenes II, by Thomas T. Tidwell

Copyright © 2006 John Wiley & Sons, Inc.

Recent experimental and calculated values of the heat of formation  $(\Delta H_f^{298})$  of methylketene radical cation CH₃CH=C=O^{+•} and of (CH₃)₂C=C=O^{+•} (**5**) are presented in Section 1.3.^{6–8}

The electron impact mass spectrum of tetramethyl-1,3-cyclobutanone formed the dimethylketene radical cation **5**, which lost CO to form dimethylcarbene radical cation **6** (equation 4).⁹ The formation of ketene radical cations **8** by electron impact–induced cleavage of  $\beta$ -lactams **7** has been examined (equation 5).^{10,11} Other reports of the formation of ketene radical cations by mass spectrometry are given in Section 2.4.

The photoelectron spectra of arylketenes **10**, which were generated by thermal Wolff rearrangements of diazo ketones **9** (equation 6), gave the lowest vertical ionization potentials that were compared to calculated values (Table 5.1).¹² The measured and calculated values showed the same trends, with distinctly lower values for the first and third IP values for the 4-MeO groups, interpreted as showing the higher coefficient for the HOMO at the 4-aryl position.



Further ionization of the ketene radical cation **1**, generated by 70 eV electron impact of ketene or by loss of  $H_2O$  from the molecular ion of acetic acid, formed the dication **11** by charge stripping in the mass spectrometer (equation 7).¹³ The ion **11** could be generated in the gas phase, although it was very unstable and was expected to be highly oxidizing toward other molecules. Computations indicated

R	IP (ev)	IP $(eV)^a$	IP (eV)	IP $(eV)^a$	IP (eV)	IP $(eV)^a$
Н	8.23	7.93	9.34	9.1	10.41	10.1
CH ₃	8.22	7.68	9.36	9.0	10.17	9.8
CH ₃ O	7.91	7.36	9.22	9.1	9.56	9.4
Cl	8.29	7.94	9.25	9.3	9.74	9.8

TABLE 5.1 Measured and Calculated Ionization Potentials of Arylketenes 10

^aCalculated by B3LYP/6-31+G(d)//HF/6-31G(d).

a planar structure and at the MP2/6-31G*//4-31G level a 27.2 kcal/mol greater stability than the oxirene structure 12.¹³

$$\underset{H}{\overset{H}{\longrightarrow}} C = O^{+} \overset{-e^{-}}{\longrightarrow} \underset{H}{\overset{H}{\longrightarrow}} C = O^{2+} \underset{H}{\overset{O}{\longrightarrow}} \underset{H}{\overset{O}{\longrightarrow}} H$$
(7)

Electrochemical oxidation of diphenylketene forming the radical cation 13 in  $CH_3CN$  resulted in the formation of benzophenone, and could occur by attack of adventitious water (equation 8) or by a process including a Ritter-type reaction with the solvent.¹⁴

$$\stackrel{Ph}{\underset{Ph}{\longrightarrow}} C=O \xrightarrow{-e^{-}} \stackrel{Ph}{\underset{Ph}{\longrightarrow}} C=O^{+} \xrightarrow{H_2O} \stackrel{Ph}{\underset{-H^+}{\longrightarrow}} \stackrel{Ph}{\underset{OH}{\longrightarrow}} \stackrel{-CO}{\underset{OH}{\longrightarrow}} \stackrel{Ph}{\underset{OH}{\longrightarrow}} \stackrel{-CO}{\underset{OH}{\longrightarrow}} \stackrel{Ph}{\underset{OH}{\longrightarrow}} (8)$$

Oxidation potentials  $E_p$  (versus SCE) in CH₃CN were obtained by cyclic volltametry for 4-MeOC₆H₄CMe=C=O (14,  $E_p = 0.91$  V),^{15,16} 4-MeC₆H₄CMe=C=O (15,  $E_p = 1.11$  V),^{15,16} and *t*-Bu₂C=C=O (16,  $E_p = 1.44$  V)¹⁷ and indicated that the ease of oxidation depended upon the electron donor abilities of the ketene substituents. The radical cations 14 and 15 generated by one electron oxidation of the ketenes with 4-Tol₃N^{+•} SbF₅⁻ as catalyst reacted with pentamethylcyclopentadiene (17) by [2+2] cycloaddition, forming 18 (equation 9).^{15,16}



In contrast to the frequent observations of ketene oxidations, reports of their reduction have been scarce. Attempted electrochemical reduction of ketenes **16** and **19–22** was unsuccessful, as no reduction was observed up to -3 eV. A reduction for Ph₂C=C=O was observed at -2.06 eV, but no ESR spectrum for the radical anion could be observed.¹⁸



Evidence for the radical anion of ketene  $CH_2=C=O^{-\bullet}$  (25) was obtained by the reaction of vinylene carbonate radical anion 23, generated using low-energy electrons in the gas phase.^{19,20} Anion 23 was suggested to lose CO₂, forming 24,

but rearrangement to 25 has a high barrier¹⁹ and oxidation may be required to form 26 (equation 10).



Formation of haloketene radical anions **27** by carbene radical anion addition to dimethyl carbonate has been proposed (equation 11).²¹ The direct observation of some fulvenone radical cations was reported (Section 4.1.10).



# **REFERENCES FOR SECTION 5.1**

- 1. Hall, D.; Maier, J. P.; Rosmus, P. Chem. Phys. 1977, 24, 373-378.
- 2. Willitsch, S.; Haldi, A.; Merkt, F. Chem. Phys. Lett. 2002, 353, 167-177.
- 3. Hop, C. E. C. A.; Holmes, J. L.; Terlouw, J. K. J. Am. Chem. Soc. 1989, 111, 441-445.
- 4. Beranová, S.; Cai, J.; Wesdemiotis, C. J. Am. Chem. Soc. 1995, 117, 9492-9501.
- Shokhba, A. A. A.; Yamaoka, H.; Hori, K.; Fokkens, R. H.; Nibbering, N. M. M. Org. Mass Spectrom. 1994, 29, 155–156.
- 6. McKee, M. L.; Radom, L. Org. Mass Spectrom. 1993, 28, 1238-1244.
- 7. Aubry, C.; Holmes, J. L.; Terlouw, J. K. J. Phys. Chem. A 1997, 101, 5958-5961.
- 8. Scott, A. P.; Radom, L. Int. J. Mass Spectrom. Ion Proc. 1997, 160, 73-81.
- 9. Aubry, C.; Polce, M. J.; Holmes, J. L.; Mayer, P. M.; Radom, L. J. Am. Chem. Soc. 1997, 119, 9039–9041.
- 10. Alcaide, B.; Perez-Castells, J.; Polanco, C.; Sierra, M. A. J. Org. Chem. 1995, 60, 6012–6016.
- 11. Borgeois, G.; Picard, J. P.; Cossio, F. P.; Palomo, C. Adv. Mass. Spectrom. **1989**, 11A, 876.
- Werstiuk, N. H.; Muchall, H. M.; Ma, J.; Liu, M. T. H. Can. J. Chem. 1998, 76, 1162– 1173.
- 13. Koch, W.; Maquin, F.; Stahl, D.; Schwarz, H. J. Chem. Soc., Chem. Commun. 1984, 1679–1680.
- 14. Becker, J. Y.; Zinger, B. J. Am. Chem. Soc. 1982, 104, 2327-2329.
- 15. Schmittel, M.; von Seggern, H. Angew. Chem. Int. Ed. Engl. 1991, 30, 999-1001.

- 16. Schmittel, M.; von Seggern, H. J. Am. Chem. Soc. 1993, 115, 2165-2177.
- 17. D. Wayner, NRC Canada, unpublished results privately communicated.
- Klages, C.; Köhler, S.; Schaumann, E.; Schmüser, W.; Voss, J. J. Phys. Chem. 1979, 83, 738–741.
- 19. van Baar, B. L.; Heinrich, N.; Koch, W.; Postma, R.; Terlouw, J. K.; Schwarz, H. Angew. Chem. Int. Ed. Engl. 1987, 26, 140–142.
- 20. Compton, R. N.; Reinhardt, P. W.; Schweinler, H. C. Int. J. Mass Spectrom. Ion Phys. 1983, 49, 113–122.
- 21. Born, M.; Ingemann, S.; Nibbering, N. M. M. J. Chem. Soc. Perkin Trans. 2, 1996, 2537–2547.

#### 5.2 PHOTOCHEMICAL REACTIONS

The photodissociation of ketene (1) is a classic reaction that has been pursued continuously since 1934,¹ including an early study using the new technique of flash photolysis,^{1a} and remains a frequently utilized and convenient source of methylene for mechanistic studies (equation 1). This topic has been periodically reviewed.^{2–6}

$$\overset{H}{\underset{H}{\rightarrowtail}} C=0 \xrightarrow{hv} CH_2: + CO$$
(1)

Photolysis of  $CH_2=C=O$  with 193 nm light proceeded by four channels, including cleavage of a C–H bond rather than initial decarbonylation, followed by addition of H• to the resulting radical **2** (equation 2).⁷ Formation of  $CH_2$  occurred by two different channels and formation of :C=C=O and H₂ was also identified.⁷

$$\overset{H}{\underset{H}{\longrightarrow}} C=O \xrightarrow{hv}{\underset{-H^{*}}{\overset{193 \text{ nm}}{\longrightarrow}}} C=C=O \xrightarrow{H^{*}} CH_{2}: + CO$$

$$1 \qquad 2 \qquad (2)$$

Photolysis of ketene in an argon matrix at 12 K with a 308 nm XeCl laser resulted in the formation of ethynol (3) (equation 3).⁸ It was proposed that this isomerization involved excitation of the  $n \rightarrow \pi^*$  transition of ketene at 310 nm, followed by decarbonylation to CH₂ and CO, and then insertion of CO into the C–H bond of CH₂.

$$\stackrel{\text{H}}{\underset{\text{H}}{\longrightarrow}} C=0 \xrightarrow{hv} CH_2: + CO \longrightarrow HC \equiv C-OH$$
(3)

Photolysis of ¹⁴CH₂=C=O led to formation of both ¹²CO and ¹⁴CO, attributed to formation of oxirene **4** (equation 4).⁹ A slow increase in the yield of ¹⁴CO suggested that the degree of scrambling was increasing with time, indicating that some of the oxirene was forming rearranged ketene.⁹ Rearrangement of highly

0

vibrationally excited ketene labeled with  13 C in the CH₂ group followed by dissociation to methylene and CO was also detected.  ${}^{10-12}$ 

$$\overset{H}{\underset{H}{\longrightarrow}} C=0 \qquad \Longrightarrow \qquad \underset{H}{\overset{O}{\underset{H}{\longrightarrow}}} \underset{H}{\overset{O}{\underset{H}{\longrightarrow}}} \underset{H}{\overset{O}{\underset{H}{\longrightarrow}}} \underset{H}{\overset{O}{\underset{H}{\longrightarrow}}} \underset{H}{\overset{O}{\underset{H}{\longrightarrow}}} \underset{H}{\overset{O}{\underset{H}{\longrightarrow}}} (4)$$

Further studies of ketene photolysis have included consideration of internal conversion of photoexcited ketene,¹³ measurement of rate constants for the unimolecular dissociation of ketene and deuterated ketene by photofragmentation in a cold jet (equation 1),¹⁴ and study of the effects of K conservation.^{15,16} In accord with expectations, the rate constant increased in a stepwise manner as the energy increased.¹⁵ Other studies of ketene photodecomposition have appeared,^{17–21} including highlevel *ab initio* calculations of the reaction path and energetics for dissociation of singlet ketene.^{22–25} The translational energy distribution of rotational states of CO from this process has been examined.^{26,27} Other related studies have also been reported.^{28–31}

Photolysis of the higher ketenes MeCH=C=O ( $\mathbf{6}$ )^{32–34} and Me₂C=C=O ( $\mathbf{7}$ )^{35–37} formed the corresponding carbenes, and in the gas phase CH₂=CH₂ and MeCH=CH₂, respectively, are the major products.^{32,35}

Photolysis of **7** in cyclohexane solution also gave Me₂C=CMe₂, which was proposed to form via the cyclopropanone **9**, which was observed by IR at 1840 cm⁻¹ (equation 5).³⁶ The quantum yield for photodissociation of **6** was less than that for CH₂=C=O,³² and for **7** the quantum yield for dissociation was unity for irradiation at 254 nm. However, for photolysis at 366 nm the quantum yield was pressure dependent, indicating formation of a long-lived, excited state.³⁵



Dimethylketene (7) was formed in an argon matrix at 10 K by photolysis of dimethylpyrazolen-3,5-dione with 254 nm light or at greater than 570 nm (equation 6).³⁷ The ketene was essentially photostable in the matrix upon further photolysis at these wavelengths, but upon photolysis at 222 nm gave propene via dimethylcarbene (equation 6).³⁷

$$\underbrace{\stackrel{Me}{\underset{Me}{\overset{}}}_{N}}_{N} \underbrace{\stackrel{hv}{\underset{254 \text{ nm}}{\overset{}}}}_{Me} \underbrace{\stackrel{Me}{\underset{Me}{\overset{}}}_{C=O}}_{T} \underbrace{\stackrel{hv}{\underset{10 \text{ K}}{\overset{}}}}_{Me} \underbrace{\stackrel{Me}{\underset{222 \text{ nm}}{\overset{}}}}_{CH_3CH=CH_2} (6)$$

Photolysis of di-*tert*-butylketene (10) gave the carbene 11, which formed 12 and 13 (equation 7).^{38,39} The persistent ketene 14 upon photolysis in hydrocarbon solvents formed the carbene 15, which gave the cyclopropane 16 by intra-molecular insertion (equation 8).⁴⁰ Upon photolysis in CH₃OD at -60 °C, 16

and the deuterated aldehyde **18** were formed in a 1:1 ratio. The formation of **18** implicated photoprotonation of the ketene and rearrangement to the carbocation **17b** (equation 9).⁴⁰



Photolysis of diphenylketene  $Ph_2C=C=O(19)$  in several solvents gave products derived from diphenylcarbene, but in THF the diphenylacetyl derivative 21 was formed (equations 10, 11).⁴¹ This was proposed to form by hydrogen abstraction by the photoexcited ketene, forming a radical pair that recombined (equation 10,11).⁴¹ Multiphotonic reaction of dibenzoyldiazomethane (22) formed benzoylphenylketene (23), which underwent photochemical decarbonylation and Wolff rearrangement to diphenylketene, which in alcohol underwent decarbonylation to diphenylcarbene, which was protonated to the cation 24 (equation 12).⁴²



Photolysis of the acylketene 25 in an argon matrix resulted in decarbonylation and formation of trimethyleneketene 26 (equation 13).⁴³ Photolysis of the bis(diazo

ketone) **27** in an argon matrix at 10 K led to the bisketene **28**, which upon photochemical decarbonylation formed cyclopropenone **29** (equation 14).⁴⁴ Photolysis of formylketene (**30**) under the same conditions gave ketene, as well as isomerization to the hydroxypropadienone **30a** (equation 15).⁴⁴



Photolysis of dimesitylketene  $(31)^{45}$  and (4-*tert*-butyl-2,6-dimethylphenyl)mesityl-ketetene  $32^{46}$  led to decarbonylation to the carbenes, which dimerized to the alkenes 33 (equation 16). Photolysis of cyclopropylideneketene (34) in an argon matrix at 15 K gave reversible formation of CO and the carbene 35 and formation of allene (equation 17).⁴⁷

$$Ar^{1}_{Ar} = C = 0 \xrightarrow{hv}_{-CO} Ar^{1}_{Ar} C : \longrightarrow Ar^{1}_{Ar} Ar^{1}_{Ar} Ar^{1}_{Ar}$$

$$(16)$$
31 (Ar = Ar^{1} = 2,4,6-Me_{3}C_{6}H_{2})
32 (Ar = 2,4,6-Me_{3}C_{6}H_{2}, Ar^{1} = 2,6-Me_{2}-4-t-BuC_{6}H_{2})

$$\begin{array}{ccc} & & & & \\ & & & \\ \hline \end{array} \begin{array}{c} C = O & & & \\ \hline \end{array} \begin{array}{c} hv \\ \hline \end{array} \begin{array}{c} C : & & \\ \hline \end{array} \begin{array}{c} C : & \\ C H_2 = C = CH_2 \end{array} \end{array}$$

$$\begin{array}{c} (17) \\ (17) \end{array}$$

Irradiation of liquid ketene with positive muons gave muonated methyl radical.⁴⁸ The photolysis of ketene/acetylene mixtures has been used to prepare propargyl radical  $(C_3H_3)$ .⁴⁹

As described in Section 4.11, the photolysis of carbon suboxide (**36**) in the gas phase in the presence of ethylene, propene, and the butenes resulted in formation of the corresponding allenes, which arose from decarbonylation of  $C_3O_2$ , forming

:C=C=O (37), which added to the alkenes, forming transient cycloalkylidene ketenes such as 34 that lost CO, giving the allenes (equation 18).⁵⁰ Reaction in the presence of 2,3-dimethyl-2-butene formed the crystalline dimer 39 of the ketene 38 (equation 19).^{51,52}



#### **REFERENCES FOR SECTION 5.2**

- 1. Ross, W. F.; Kistiakowsky, G. B. J. Am. Chem. Soc. 1934, 56, 1112-1115.
- 1a. Knox, K.; Norrish, R. G. W.; Porter, G. J. Chem. Soc. 1952, 1477-1486.
- 2. Moore, C. B. Faraday Disc. 1995, 102, 1-15.
- 3. Kirmse, W. Carbene Chemistrty, 2nd ed.; Academic Press, New York, 1971, pp. 9-18.
- Johnson, R. P. In Organic Photochemistry; Padwa, A. Ed.; Dekker, New York, 1985, Vol. 7, pp. 135–136.
- 5. DeMore, W. B.; Benson, S. W. Adv. Photochem. 1964, 2, 219-261.
- 6. Lee, E. K. C.; Lewis, R. S. Adv. Photochem. 1980, 12, 1-96.
- 7. Glass, G. P.; Kumaran, S. S.; Michael, J. V. J. Phys. Chem. A 2000, 104, 8500-8507.
- 8. Hochstrasser, R.; Wirz, J. Angew. Chem. Int. Ed. Engl. 1990, 29, 411-413.
- 9. Russell, R. L.; Rowland, F. S. J. Am. Chem. Soc. 1970, 92, 7508-7510.
- Lovejoy, E. R.; Kim, S. K.; Alvarez, R. A.; Moore, C. B. J. Chem. Phys. 1991, 95, 4081–4093.
- 11. Lovejoy, E. R.; Kim, S. K.; Moore, C. B. Science 1992, 256, 1541-1544.
- 12. Lovejoy, E. R.; Moore, C. B. J. Chem. Phys. 1993, 98, 7846-7854.
- 13. Yarkony, D. R. J. Phys. Chem. A. 1999, 103, 6658-6668.
- 14. Kim, S. K.; Lovejoy, E. R.; Moore, C. B. J. Chem. Phys. 1995, 102, 3202-3219.
- Mellinger, A.; Hall, M. A.; Wade, E. A.; Moore, C. B. Ber. Bunsen-Ges. 1997, 101, 465–468.
- 16. Mellinger, A.; Ashikhmin, M. V.; Moore, C. B. J. Chem. Phys. 1998, 108, 8944-8949.
- 17. Wade, E. A.; Clauberg, H.; Kim, S. K.; Mellinger, A.; Moore, C. B. J. Phys. Chem. A, 1997, 101, 732–739.
- 18. Castillejo, M.; Martin, M.; de Nalda, R.; Solis, J. Chem. Phys. Lett. 1997, 268, 465–470.
- 19. Castillejo, M.; Couris, S.; Lane, E.; Martin, M.; Ruiz, J. Chem. Phys. 1998, 232, 353-360.
- 20. Cerezo, C.; Martin, M. J. Photochem. Photobiol. A 2000, 134, 127-132.
- 21. Arbilla, G.; de Oliveira, K. M. P. G. J. Braz. Chem. Soc. 1993, 4, 165-171.
- 22. Klippenstein, S. J.; East, A. L. L.; Allen, W. D. J. Chem. Phys. 1994, 101, 9198-9201.

- 22a. Forsythe, K. M.; Gray, S. K.; Klippenstein, S. J.; Hall, G. E. J. Chem. Phys. 2001, 115, 2134–2145.
  - 23. Klippenstein, S. J.; East, A. L. L.; Allen, W. D. J. Chem. Phys. 1996, 105, 118-140.
  - 24. Smith, S. C. Faraday Disc. 1995, 102, 17-29.
  - 25. Rasmussen, A. J.; Gates, K. E.; Smith, S. C. J. Chem. Phys. 1999, 110, 1354–1364.
  - 26. Morgan, C. G.; Drabbels, M.; Wodtke, A. M. J. Chem. Phys. 1996, 105, 4550-4555.
  - 27. Costen, M. L.; Katayanagi, H.; Hall, G. E. J. Phys. Chem. A 2000, 104, 10247-10258.
  - 28. Kaledin, A. L.; Seong, J.; Morokuma, K. J. Phys. Chem. A 2001, 105, 2731-2737.
  - 29. Blitz, M. A.; Pilling, M. J.; Seakins, P. W. Phys. Chem. Chem. Phys. 2001, 3, 2241-2244.
  - Blitz, M. A.; McKee, K. W.; Pilling, M. J.; Seakins, P. W. Chem. Phys. Lett. 2003, 372, 295–299
  - 31. Kobayashi, K.; Sears, T. J. Can. J. Phys. 2001, 79, 347-358.
  - 32. Chong, D. P.; Kistiakowsky, G. B. J. Chem. Phys. 1964, 97, 1793-1797.
  - 33. Umstead, M. E.; Shortridge, R. G.; Lin, M. C. J. Phys. Chem. 1978, 82, 1455-1460.
  - 34. Fujimoto, G. T.; Umstead, M. E.; Lin, M. C. J. Chem. Phys. 1985, 82, 3042-3044.
  - 35. Holroyd, R. A.; Blacet, F. E. J. Am. Chem. Soc. 1957, 79, 4830-4834.
  - 36. Haller, I.; Srinivasan, R. J. Am. Chem. Soc. 1965, 87, 1144-1145.
  - 37. Maier, G.; Heider, M.; Sierakowski, C. Tetrahedron Lett. 1991, 32, 1961–1962.
  - 38. Malatesta, V.; Forrest, D.; Ingold, K. U. J. Phys. Chem. 1978, 82, 2370-2373.
  - Marfisi, C.; Verlaque, P.; Davidovics, G.; Pourcin, J.; Pizzala, L.; Aycard, J.-P., Bodot, H. J. Org. Chem. 1983, 48, 533–537.
  - 40. Kirmse, W.; Spaleck, W. Angew. Chem. Int. Ed. Engl. 1981, 20, 776-777.
  - 41. Nozaki, H.; Nakano, M.; Kondo, K. Tetrahedron, 1966, 22, 477-481.
  - 42. Kirmse, W.; Krzossa, B.; Steenken, S. Tetrahedron Lett. 1996, 37, 1197-1200.
  - 43. Leung-Toung, R.; Wentrup, C. J. Org. Chem. 1992, 57, 4850-4858.
  - 44. Maier, G.; Reisenauer, H. P.; Sayrac, T. Chem. Ber. 1982, 115, 2192-2201.
  - 45. Zimmerman, H. E.; Paskovich, D. H. J. Am Chem. Soc. 1964, 86, 2149-2160.
  - Maeda, K.; Okamoto, Y.; Morlender, N.; Haddad, N.; Eventova, I.; Biali, S. E.; Rappoport, Z. J. Am Chem. Soc. 1995, 117, 9686–9689.
  - 47. Monnier, M.; Allouche, A.; Verlaque, P.; Aycard, J.-P. J. Phys. Chem. 1995, 99, 5977–5985.
  - McKenzie, I.; Addison-Jones, B.; Brodovitch, J.-C.; Ghandi, K.; Kecman, S.; Percival, P. W. J. Phys. Chem. A, 2002, 106, 7083–7085.
  - Blitz, M. A.; Beasley, M. S.; Pilling, M. J.; Robertson, S. H. Phys. Chem. Chem. Phys. 2000, 2, 805–812.
  - 50. Bayes, K. J. Am. Chem. Soc. 1961, 83, 3712-3713.
  - 51. Willis, C.; Bayes, K. D. J. Am. Chem. Soc. 1966, 88, 3203-3208.
  - 52. Parker, G.; Wiseman, D.; Wintner, C.; MacKay, C. J. Org. Chem. 1984, 49, 4494-4497.

# 5.3 THERMOLYSIS REACTIONS

Staudinger and Endle observed in 1913 that diphenylketene (1) underwent thermolysis with loss of CO and formation of diphenylcarbene (2), which cyclized to

fluorene (3) (equation 1).¹ It was also found that dimethylketene (4) on pyrolysis gave dimethylcarbene (5), which formed propene and tetramethylethylene (equation 2).¹ This work defined at an early date many of the fundamental reactions of carbene chemistry. Ketenes are available from a variety of sources that are suitable for carbene generation by thermal or photochemical processes and their full potential for carbene generation is yet to be exploited.



Thermal decarbonylation of ketenes is a reversible process, as has been shown in the gas phase thermolysis of azoethane for generation of methylcarbene in an atmosphere of CO, resulting in formation of methylketene **6** (equation 3).² There are also numerous examples of ketenes formed from carbenes in low-temperature matrices generated either in the gas phase and deposited in the matrix or by the photolysis of diazoalkanes in the matrix in the presence of CO, as in the formation dicyclopropylketene, and of **6a** (equation 3a).⁴

$$\stackrel{\text{Me}}{\underset{H}{\longrightarrow}} N_2 \xrightarrow[-N_2]{} \stackrel{\text{Me}}{\underset{H}{\longrightarrow}} C: \xrightarrow[C]{} \stackrel{\text{Co}}{\underset{H}{\longrightarrow}} \stackrel{\text{Me}}{\underset{H}{\longrightarrow}} C=0$$
(3)

$$(MeO)_2 \stackrel{P'}{\xrightarrow{Ph}} N_2 \xrightarrow{Ar} (MeO)_2 \stackrel{P'}{\xrightarrow{Ph}} C: \xrightarrow{CO} (MeO)_2 \stackrel{P'}{\xrightarrow{Ph}} C=O$$
(3a)  
$$6a (IR 2120 \text{ cm}^{-1})$$

A mechanism involving 36 reactions was developed for the high-temperature pyrolysis of ketene (7), including decarbonylation,^{5,6} and atom isomerization of vibrationally excited ketene indicated the intermediacy of the oxirene intermediate **8** (equation 4).⁷

$$H \xrightarrow{O}_{H} \xrightarrow{H}_{H} \xrightarrow{H}_{H} C=0 \xrightarrow{A}_{H} \xrightarrow{H}_{H} C: + CO$$
(4)  
8 7

Pyrolysis of methyl diazomalonate at 280–640 °C formed ketene 9 (equation 5), which itself underwent thermal decarbonylation under the reaction conditions 10 and led to methyl acetate (equation 6).⁸



The MP2-calculated activation energy for decarbonylation of ketene **11a** exceeded that of ketene **11b** by 14.3 kcal/mol.⁹ Ketene **11a** formed by thermal Wolff rearrangement was detected by photoelectron spectroscopy, but **11b** was not, suggesting that **11b** may have decarbonylated under the reaction conditions (equation 7).⁹



Pyrolysis of the Meldrum's acid derivative **12** was proposed to form ketene **13**, which lost CO and formed **14** (equation 8).¹⁰ The  $\beta$ -keto ester **15** lost MeOH at 560 °C, forming the acylketene **16**, which reacted by cyclization and ketene loss, leading to cyclopentenone **17** (equation 9).¹¹ Ketene **16** also underwent decarbony-lation to give **18** (equation 10) in a 4:1 **17/18** ratio.¹¹



FVP of (dimethylhydridosilyl)ketenes **19** resulted in formation of the alkyne **20** (equation 11).¹² Pyrolysis of the diazo ester **21** gave the ketene **22**, which underwent decarbonylation to form acetophenone as the most abundant product (equation 12).¹³ The thermal isomerization of ketene via an oxirene intermediate has been examined by computations.¹⁴



#### **REFERENCES FOR SECTION 5.3**

- 1. Staudinger, H.; Endle, R. Chem. Ber. 1913, 46, 1437-1442.
- 2. Seburg, R. A.; McMahon, R. J. Am. Chem. Soc. 1992, 114, 7183-7189.
- 3. Ammann, J. R.; Subramanian, R.; Sheridan, R. S. J. Am. Chem. Soc. 1992, 114, 7592–7594.
- 4. Tomioka, H.; Komatsu, K.; Shimizu, M. J. Org. Chem. 1992, 57, 6216-6222.
- 5. Hidaka, Y.; Kimura, K.; Kawano, H. Combust. Flame 1994, 99, 18-28.
- 6. Sato, K.; Hidaka, Y. Combust. Flame 2000, 122, 291-311.
- 7. Ranzi, E.; Faravelli, T.; Goldaniga, A.; Ferrari, F.; Lattuada, M. *Exp. Thermal Fluid Sci.* **2000**, *21*, 71–78.
- 8. Richardson, D. C.; Hendrick, M. E.; Jones, M., Jr. J. Am. Chem. Soc. 1971, 93, 3790-3791.
- 9. Werstiuk, N. H.; Ma, J.; Roy, C. D.; Kresge, A. J.; Jefferson, E. A. *Can. J. Chem.* **1996**, 74, 2536–2539.
- Brown, R. F. C.; Eastwood, F. W.; Lim, S. T.; McMullen, G. L. Aust. J. Chem. 1976, 29, 1705–1712.
- 11. Leyendecker, F. Tetrahedron 1976, 32, 349-353.
- 12. Barton, T. J.; Groh, B. L. J. Am. Chem. Soc. 1985, 107, 7221-7222.
- Tomioka, H.; Ohtawa, Y.; Murata, S. Nippon Kagaku Kaishi 1989, 1431–1439. Chem. Abst. 1989, 112, 118034g.
- 14. Gezelter, J. D.; Miller, W. H. J. Chem. Phys. 1995, 103, 7868-7876.

# 5.4 CYCLOADDITION REACTIONS OF KETENES

Cycloaddition reactions have been a characteristic feature of ketene chemistry from the very beginning, with the isolation of the ketene dimer (1) by Chick and

Wilsmore,^{1,2} which was eventually shown to have the methylene- $\beta$ -lactone structure **2** (equation 1). Dimethylketene (**3**) was found to form the symmetrical dimer **4** by Staudinger and Klever (equation 2).³ Staudinger also found, soon after the initial report of the preparation of diphenylketene (**5**), that it reacted with cyclopentadiene to form an addition compound **6** that was later shown to result from net [2+2] cycloaddition,⁴ and that it reacted with imines **7** to form the  $\beta$ -lactam **8** (equation 4) and with aldehydes to form  $\beta$ -lactone **9**, which, however, underwent decarboxylation, forming triphenylethylene (equation 5).^{4,5} Since that time, cycloaddition has remained the most distinctive, useful, and intellectually challenging aspect of ketene chemistry.



Ketenes have a unique propensity for giving facile [2+2] cycloaddition reactions, even when other pathways are available, and give [4+2] cycloadditions only in unusual circumstances. This feature of ketene chemistry is of major importance for preparative purposes in that the reaction with alkenes and dienes gives cyclobutanones (equation 3), usually with a high degree of stereoselectivity, in a process of considerable synthetic utility. The reaction with imines forming  $\beta$ -lactams (equation 4) is the most straightforward method for the synthesis of these compounds, which have valuable applications in the synthesis of antibiotics (Section 5.4.1.7). Ketenes also usually give [2+2] cycloadditions with dienes, and [4+2] cycloadditions to give cyclohexenones are so rare that "ketene equivalents" are used to mimic the behavior of hypothetical ketenes that would react by this pathway.^{6–8} Cycloaddition reactions of ketenes have been considered in a number of reviews.^{9–18d}

Ketene cycloadditions often proceed with a high degree of stereoselectivity, which increases their value in synthesis and heightens interest in their reaction mechanisms. The proposal of the Woodward-Hoffmann rules,¹⁹ and their eventual application to explain the distinctive [2+2] cycloaddition reaction mode and unusual stereoselectivity of ketenes, led to an intense period of investigation of many types of ketene cycloadditions. Ketene [2+2] cycloadditions became an important proving ground for testing theories of orbital symmetry control of organic reactivity, as their preferences for formation of the less stable products, often with complete selectivity, were predicted by theory.¹⁹

After several decades of intense study activity this field has reached maturity, but it still produces surprises. A notable example was the discovery that the canonical [2+2] cycloaddition of diphenylketene with cyclopentadiene did not proceed with initial cyclobutanone formation, but rather [4+2] cycloaddition to the C=O bond of the ketene (Sections 5.4.1.2, 5.4.1.3).

A great deal of the currently available experimental and computational evidence is consistent with many [2+2] cycloadditions of ketenes as proceeeding through concerted reactions governed by orbital symmetry. In particular, the reactions of ketenes with alkenes usually retain the stereochemistry of the alkene and often show stereoselectivity to form the least stable product. However, there are a number of cases where this does not occur and nonconcerted pathways are followed, and there is a suspicion that other seemingly well-established reactions may actually involve other transient intermediates, whose existence may be revealed by more careful examination.

The Woodward-Hoffmann analysis of ketene cycloadditions¹⁹ provided a theoretical basis for understanding ketene cycloaddition and greatly stimulated study in this field. It was suggested¹⁹ that ketene cycloaddition with alkenes proceeded by the perpendicular arrangement of the two reactants, as shown in Figure 5.1, with both bonds to the alkenes being formed from the same side (*supraficial*), while the bonds to the two carbons of the ketene were formed from opposite sides (*antarafacial*). The actual role of this process has been the subject of continuous debate and is still unsettled, as discussed below (Sec. 5.4.1.3). However, there is strong evidence that many ketene cycloadditions are not concerted, that is, that they proceed by a two-step process with formation of a zwitterionic intermediate. This latter process is also consistent with the stereospecific formation of the thermodynamically less stable products without loss of the stereochemistry of the reactants.



Figure 5.1 The [2+2] cycloaddition of ketene.

Thus, although the concerted pathway is consistent with the observed data in many cases, a stepwise process often provides at least an equally convincing explanation of the result.

Ketene cycloaddition is reversible as a useful process in only a few examples, including in the pyrolysis of the unsymmetrical dimer diketene to ketene (Section 3.1) and in the cleavage of some cyclobutanones to ketenes (Section 3.4.1).

Other important cycloaddition reactions of ketenes besides [2+2] cycloaddition to carbon-carbon double bonds are the [2+2] cycloaddition to imines (Section 5.4.1.7) and to carbonyl groups (Section 5.4.5), [3+2] and [4+2] cycloadditions (Sections 5.4.2 and 5.4.3), and [2+1] cycloadditions (Section 5.10).

#### **REFERENCES FOR SECTION 5.4**

- 1. Wilsmore, N. T. M. J. Chem. Soc. 1907, 91, 1938-1941.
- 2. Chick, F.; Wilsmore, N. T. M. J. Chem. Soc. 1908, 93, 946-950.
- 3. Staudinger, H.; Klever, H. W. Chem. Ber. 1906, 39, 968-971.
- 4. Staudinger, H. Chem. Ber. 1907, 40, 115-118.
- 5. Staudinger, H. Liebigs Ann. 1907, 356, 51-123.
- 6. Ranganathan, S; Ranganathan, D.; Mehrotra, A. K. Synthesis 1977, 289-296.
- 7. Williams, R. V.; Lin, X. J. Chem. Soc., Chem. Commun. 1989, 1872-1873.
- 8. Wenkert, E; Vial, C.; Näf, F. Chimia 1992, 46, 95-97.
- 9. Clemens, R. J. Chem. Rev. 1986, 86, 241-318.
- 10. Roberts, J. D.; Sharts, C. M. Org. React. 1962, 12, 1-56.
- Ghosez, L.; O'Donnell, M. J. In *Pericyclic Reactions* Marchand, A. P.; Lehr, R. E., Eds.; Academic Press, New York, 1977, pp. 79–140.
- Ulrich, H. Cycloaddition Reactions of Heterocumulenes; Academic Press, New York, 1967, pp. 38–109.
- 13. Luknitskii, F. I.; Vovsi, B. A. Russian Chem. Revs. 1969, 38, 487-494.
- Carruthers, W. Cycloaddition Reactions in Organic Synthesis; Pergamon Press, Oxford, 1990, pp. 332–367.
- 14. Brady, W. T. Tetrahedron 1981, 37, 2949-2966.
- 15. Hanford, W. E.; Sauer, J. C. Org. React. 1946, 3, 108-140.
- 16. Hyatt, J. A.; Raynolds, P. W. Org. React. 1994, 45, 159-646.
- Ghosez, L.; Marchand-Brynaert, J. In *Comprehensive Organic Synthesis*; Trost, B. M., Ed.; Pergamon, New York, 1991, Vol. 5, pp. 85–122.
- Snider, B. B. In *Methods of Organic Chemistry (Houben-Weyl)*, "Stereoselective Synthesis," 4th ed; Thieme, Stuttgart, 1995, Vol. E21c, pp. 3060–3084.
- 18a. Snider, B. B. Chem. Rev. 1988, 88, 793-811.
- 18b. Lee-Ruff, E. *Methods of Organic Chemistry (Houben-Weyl)*; Thieme, Stuttgart, 1997, Vol. E 17e, pp. 190–223.
- 18c. Chan, T. Methods of Organic Chemistry (Houben-Weyl); Thieme, Stuttgart, 1997, Vol. E 17f, pp 639–650.
- 18d. Schaumann, E.; Scheiblich, S. Methoden der Organischen Chemie (Houben-Weyl); Thieme, Stuttgart, 1993, Vol. E 15 3, pp. 2933–2957.

19. Woodward, R; B.; Hoffmann, R. Angew. Chem., Int. Ed. Engl. 1969, 8, 781–853. The Conservation of Orbital Symmetry; Verlag Chemie: Academic, New York, 1970.

#### 5.4.1 Intermolecular [2+2] Cycloaddition

5.4.1.1 Dimerization of Ketenes The tendency toward dimerization is perhaps the most characteristic feature of ketene chemistry, but [2+2] dimers have not been observed from silylketenes, ketenes with only halogen substituents (Section 4.4.2), or the highly substituted t-Bu₂C=C=O or (CF₃)₂C=C=O. As discussed in Section 4.1.1, ketene (1) has three possible dimers, 2–4, and dimerization in the absence of catalyst formed 1,3-cyclobutanedione 2 and the  $\beta$ -lactone dimer (3) in a 5:95 ratio (equation 6).¹ The most recent computations indicated that 3 is more stable by 1 kcal/mol.² The diketone dimer 2 was prepared selectively by hydrolysis of the vinyl ether 5 (equation 2),³ while the bis(vinyl ether) dimer 4 is unknown.



The structure of the dimer **3** was in doubt for almost 50 years and elicited the comment from workers in the field: "The extraordinary chemical behavior of the ketene dimers has lent exceptional interest to that class of substances, and the controversy which has raged for decades over the structures of the compounds is without parallel in the study of small molecules."⁴ The final structural proof that the major dimer had the methylene- $\beta$ -propiolactone structure, instead of one of a variety of others that had been considered (**2**, **4**, **6–8**), was based on chemical studies,⁵ electron diffraction,⁶ X-ray crystallography,^{7,8} and the microwave spectrum⁹ and there is an excellent review of the long controversy regarding this structure.¹⁰ The dimer **3** is commercially available, serves as a laboratory precursor for monomeric ketene,¹¹ and is a valuable synthetic reagent in its own right.^{10,12}



Previous computations on the dimers of ketene had indicated that the cyclobutanedione structure 2 was 1 kcal/mol more stable than the lactone structure 3, which

**Scheme 5.1** Calculated and experimental bond distances in diketene (2). Electron diffraction data⁶ in parentheses, X-ray results^{7,8} in brackets, and calculated values.^{2,14}

is 32 kcal/mol more stable than the unknown isomer **4**. The agreement of the calculated structures with experimental values was characterized as "only fair."¹³ Further calculations of the structure of **3** at higher levels of theory¹⁴ led to the conclusion that the theoretical bond lengths, which agree well with experimental microwave results,⁹ were more accurate than early electron diffraction⁶ and X-ray^{7,8} values. The comparative values (Å) are shown in Scheme 5.1, with electron diffraction data⁶ in parentheses, X-ray results^{7,8} in brackets and the calculated values.¹⁴

Dimethylketene **10** was prepared by zinc dehalogenation in 1906 and formed the cyclobutanedione dimer **11** (equation 3).¹⁵ Many monosubstituted ketenes ("aldo-ketenes") such as methylketene (**12**) tend to form both  $\beta$ -lactone dimers of type **13** and substituted derivatives of enolized cyclobutanedione analogous to **14**,^{16,17} whereas disubstituted ketenes ("ketoketenes") usually form cyclobutanedione dimers analogous to **11** (equation 3).



In the formation of dimers of monoalkylketenes such as methylketene (12) from the acyl chlorides and  $Et_3N$ , it was suggested that  $Et_3NH^+Cl^-$  promoted formation of lactone dimers such as 13,^{18–20} whereas in uncatalyzed reactions cyclobutanediones such as 11 (equation 3) or the enolized forms 14 are formed (equation 4).¹⁶ Lactone dimers prepared by dehydrochlorination of acyl chlorides derived from long-chain fatty acids have long been used in paper coatings (sizing).²¹

The lactone structure **13** for the liquid dimer of methylketene and the enolized 1,3-cyclobutanedione structure **14** for the crystalline, acidic dimer, were established in 1950 by Woodward and Small.⁴ In a prescient mechanistic analysis, these authors pointed out that the dimerization would be favored to proceed by formation of

zwitterionic intermediates such as **15**, forming **16**, which would be converted to the enol **14** (equation 5), and that attack on the C=C bond of ketenes would occur perpendicular to the ketene plane, while attack on the carbonyl would be in the plane.⁴



It has been suggested that, with the exception of  $CH_2=C=O$  itself, the primary uncatalyzed dimerization pathway for both aldo- and ketoketenes is formation of cyclobutanedione derivatives of type **11** or **14**.¹⁶ According to the Woodward-Hoffmann analysis, this would occur by a concerted  $[\pi 2_a + \pi 2_s]$  cycloaddition (vide infra).²² In the presence of acidic catalysts such as triethylammonium chloride or ZnCl₂, lactone dimers of type **13** or  $\gamma$ -pyrone type trimers (**17**) may be formed. Furthermore, it was demonstrated that in the presence of base the lactone dimers **13** revert to cyclobutanedione structures **14**.¹⁶



A rationalization of the effect of acid catalysts in promoting the formation of lactone structures was proposed in which the catalyst enhances the electrophilicity of the carbonyl carbon and assists nucleophilic attack by the carbonyl oxygen of a second ketene molecule (equation 6).¹⁶



Base-catalyzed isomerization of lactones **21** was suggested¹⁶ to occur by enolization and generation of ketene intermediates **23**, which cyclized to form **24** (equation 7). Trapping of **23** by a further ketene monomer could lead to pyrones **17**.



Lactone dimers **21** with long-chain alkyl groups were originally incorrectly formulated as acylketenes.^{18,19} On hydrolysis these dimers yielded  $\beta$ -keto acids that underwent decarboxylation in practical syntheses of long-chain ketones.^{18,19}

The dimerization of methylketene (12) catalyzed by chiral amines gave enantioselective dimerization to 13 (54 to 98% *ee*).^{23–28} For example, dehydrochlorination of propionyl chloride with diisopropylethylamine in CH₂Cl₂ at room temperature with 5 mol% trimethylsilylquinine as catalyst gave a 79% yield of the *R*-dimer 13, isolated as the β-ketoamide 25 after reaction with methoxy(methyl)amine (equation 8).²³ The dimerization was proposed to proceed by addition of the chiral catalyst to 12, forming a zwitterion 26 that reacted with a second molecule of 12, forming 13 (equation 9).²⁴ Kinetic studies showed that methylketene formation from propionyl chloride was rate determining and that this reacted rapidly to form the dimer. It was also shown that methylketene generated thermolytically from propionic anhydride gave the same selectivity for dimer formation, indicating that the dimerization involved two ketene molecules, not propionyl chloride.²⁸



Dimers **21** of ketenes RCH=C=O (**18**, R = Et, *i*-Pr, *t*-Bu, TIPSOCH₂ [TIPS = *i*-Pr₃Si], MeO₂CCH₂) were prepared similarly in 58–88% yields and 91–96% *ee* (equation 10).²³



Amination of the methylketene dimer **13** formed **26**, and selective reduction gave either isomer of **27** (equation 11).²⁵ The dimer was also used in the preparation of pamamycin 621A.²⁸ A modified procedure involved production of methylketene by pyrolysis of propionic anhydride, enantioselective dimerization and opening of the dimer with lithium methyl(methoxy)amide.²⁷ The product was used in further synthesis of polypropionates.



Reaction of the 2-bromo acyl chloride **28** with Zn/CuCl in CH₃CN with ultrasonication gave the unobserved ketene **29**, which formed isomeric dimers **30** and **31** together with the trimer **32** (equation 12).²⁹ Reaction of the free ketene gave only the dimer **30**,³⁰ and treatment of **30** with Lewis acids did not lead to isomerization to **31**.²⁹ Evidently, complexation with ZnCl₂ favored the formation of **31**.



Treatment of the dimer of pentamethyleneketene with NaOMe led to the trimeric 1,3,5-cyclohexanetrione **33** (equation 13).³¹ Treatment of Me₂C=C=O (**10**) or its dimer **11** with a catalytic amount of NaOMe in refluxing toluene also led to a trimer,³² and dimethyleneketene (**29**) and trimethyleneketene [(CH₂)₃C=C=O] (**34**) gave similar dimers and trimers (cf. equation 12).^{31,32}



Dehalogenation of  $\alpha$ -bromoacyl chlorides **28** and **28a** by the metal anions Mn(CO)₅⁻ or Cr(CO)₄NO⁻ formed the ketenes **29** and **34**, respectively

(equation 14), and led to the respective dimers **31** and **35**.³³ When the two ketenes were cogenerated, there was a strong kinetic preference for formation of the mixed dimer **36** (equation 15).³³



Mixed ketene dimers have also been formed by the generation of haloketenes in the presence of Me₂C=C=O (10), by the mixing of solutions of two different ketenes, and by the cogeneration of two different ketenes.^{34,35} When two unsymmetrical ketenes reacted, equal amounts of the stereoisomeric mixed dimers were formed.³⁵

Computational studies using several theoretical methods to analyze ketene dimerization predicted that the transition states **37** and **38** form the cyclobutanedione and  $\beta$ -lactone dimers **2** and **3**, respectively^{2,36–38} Barriers of 36 kcal/mol for the formation of **2** and 32 kcal/mol for the formation of the major product **3** were found.³⁶ Both of these transition structures were nonplanar, with ring dihedral angles ranging from 23.1 to 55.7°.^{36–38} The transition states were highly unsymmetrical and were interpreted as indicating nonsynchronous but concerted reactions.³⁶



The transition state **38** involved interaction of the HOMO of one ketene (the C=C  $\pi$  MO) and the LUMO of the other ketene (the C=O  $\pi^*$  MO), with negative charge development on oxygen.³⁶ This was lower in energy than **37**, in which both the positive and negative charge development was on carbon.³⁶ It was suggested that methyl substituents would tend to favor concerted dimerization, whereas electron-withdrawing groups would favor a stepwise process.³⁶ The



Scheme 5.2 Reaction pathways for [2+2] cyclodimerization of ketenes.

calculated^{37,38} deuterium isotope effects on the product distribution were in good agreement with experimental values measured using natural abundance ²H NMR spectroscopy of the dimer of CDH=C=O.³⁹

The stereochemistry of the products of dimerization of unsymmetrical ketenes **39** according to Scheme 5.2 are summarized in Table 5.2.^{40,41} These results were discussed in terms of the  $[\pi 2_s + \pi 2_a]$  mechanism with transition states **40** and **41** leading to the Z dimer **44**, and **42** and **43** leading to the E dimer **45**. It was proposed that as group R¹ became progressively larger, **41** and **43** were increasingly favored. However, the trends in the results were not consistent and particularly with neopentyl(methyl)ketene the *trans* isomer was favored. It was concluded that these results are not definitive for establishing a  $[\pi 2_s + \pi 2_a]$  mechanism.⁴¹

An alternative zwitterionic intermediate 46 with a strong interaction between the R groups could react with rotation of these groups away from one another, leading to the *trans* product 45. This would, however, increase the interaction between groups R and R¹, and the balancing of these effects may explain why the results do not follow a clear trend.

Ketene **47** underwent dimerization to form the *syn* and *anti* dimers **48** and **49**^{42–45} in a 1:9 ratio (equation 16).⁴⁵ This preference for the less crowded *anti* isomer was unusual and would not be predicted by the less hindered transition state **50** for a concerted  $[\pi 2_s + \pi 2_a]$  process. Possibly this behavior resulted from a zwitterionic intermediate **51** that underwent preferential formation of the less hindered

R	Ph	Ph	Ph	Ph	PhCH ₂	PhCH ₂	PhCH ₂	<i>i</i> -Pr	t-BuCH ₂
$\mathbb{R}^1$	Me	Et	<i>i</i> -Pr	PhCH ₂	Me	Et	<i>i</i> -Pr	Me	Me
44/45	20/80	62/38	100/0	73/27	60/40	58/42	64/36	76/24	23/77

TABLE 5.2 Dimerzation of Ketenes to 44 and 45

anti isomer 49.



The reaction of **52** gave the *anti* dimer **53** (equation 17), whose structure was established by X-ray crystallography.^{44,45} Just as in the reaction of **47**, this was not the product predicted by a concerted  $[\pi 2_a + \pi 2_s]$  cycloaddition and a two-step process via a zwitterionic intermediate analogous to equation 5 may be involved. The reason that **47** and **52** gave more of the *anti* dimer compared to **39** (Table 5.2) may result from stabilization of the acylium ion by the cyclopropane ring, leading to a longer lifetime for **51**. The compact size of the cyclopropyl group may also reduce the steric interactions in the path to the *anti* product.



Moore and Wilbur^{46,47} studied the reaction of *t*-BuC(CN)=C=O (**54**) with CH₂=C=O, MeCH=C=O, Me₂C=C=O, and MeEtC=C=O. In the first two examples  $\beta$ -propiolactone dimers **57** were formed, whereas the latter two formed cyclobutanediones **58** (equations 18, 19). These ketenes were proposed to react by a head-to-tail approach giving initially the zwitterionic intermediate **55**. When R = H this isomerized to **56**, and **55** and **56** formed **57** and **58**, respectively. The same intermediates were also generated from thermolysis of the corresponding 4-azido-5-*tert*-butylcyclopentene-1,3-diones **59** and gave the same products. Thus, these results were taken as showing unequivocally that for these highly polar ketenes, stepwise zwitterionic mechanisms were involved in the dimerizations.^{40,46,47} The preference for formation of  $\beta$ -propiolactone-type dimers **57** of aldoketenes (R = H) would appear to arise from the lower barrier of conversion of **55** to **57** or to a lower barrier to formation of **57** directly from the ketenes. The structures were represented^{46,47} with the ring-forming atoms in an essentially coplanar arrangement, but it would appear that steric interactions would favor


that shown in 60 (equation 20), with little decrease in bonding efficiency.

In the reaction of MeEtC=C=O there was very little selectivity in the formation of the dimer, with a 46:54 ratio of the E/Z isomers being observed. This lack of selectivity led Moore and Wilbur⁴⁶ to question if the products with high selectivity observed by Dehmlow et al. (Table 1)⁴⁰ were good evidence for a concerted pathway. As noted above, in later work Dehmlow et al. did not rule out stepwise processes.⁴¹

Bis(trifluoromethyl)ketene **61** did not dimerize thermally but was quite reactive toward other ketenes, reacting with Me₂C=C=O (**10**) to form both cyclobutanediones and  $\beta$ -propiolactone-type dimers **64** and **65**, and with CH₂=C=O and MeCH=C=O to form only  $\beta$ -propiolactone-type dimers.⁴⁸ The reactivity was explained in terms of steric hindrance in a zwitterionic intermediate **62** that allowed competitive formation of **63** leading to **65** (equations 21, 22).



Huisgen and Otto⁴⁹ measured the solvent dependence of the dimerization rate of  $Me_2C=C=O$  (10) by ¹H NMR and found good second-order kinetics, with a

variation in substrate at 35 °C from  $2.31 \times 10^{-5}$  M⁻¹s⁻¹ in CCl₄ to 67.4 × 10⁻⁵ M⁻¹ s⁻¹ in CH₃CN. Relative rates were 1 (CCl₄), 2.00 (C₆H₆), 2.81 (C₆H₅Cl), 10.3 (CDCl₃), 11.3 (CH₂Cl₂), 15.1 (PhCN), and 29.2 (CH₃CN). This variation in rate gave a moderately good correlation with the solvent polarity parameter  $E_{\rm T}$ . The reaction in PhCN gave  $\Delta H^{\ddagger} = 10.8$  kcal/mol and  $\Delta S^{\ddagger} = -42$  eu, suggesting an ordered transition state. These results were interpreted as showing less of an effect of solvent polarity on the rate than was expected for a zwitterionic transition state 66, so a one-step reaction proceeding through a dipolar transition state 67 was favored (equation 23). However, both transition states 66 and 67 are expected to be stabilized qualitatively in polar solvents; furthermore, the reactant ketenes are quite polar. Hence, the magnitude and even the direction of the solvent effect on the rates by either transition state are difficult to predict. As pointed out,⁴⁹ the dipole moment of the product dione was zero, and a product-like transition state would be expected to show an inverse dependence of rate on solvent polarity. Both processes are also expected to have large negative entropies of activation, so quantitative criteria for differentiating the two are not available.



In summary, the evidence appears compelling that at least some mixed ketene dimerizations involving ketenes with substituents of very different polarities proceed through a two-step process involving zwitterionic intermediates. For dimerization of ketenes with similar polarities, the dimerizations occur either through onestep processes with a high degree of polar character or through two-step processes with zwitterionic intermediates.

## **REFERENCES FOR SECTION 5.4.1.1**

- 1. Tenud, L.; Weilmann, M.; Dallwigk, E. Helv. Chim. Acta 1977, 60, 975-977.
- 2. Kelly, E.; Seth, M.; Ziegler, T. J. Phys. Chem. A, 2004, 108, 2167-2180.
- 3. Van Daalen, J. J.; Kraak, A.; Arens, J. F. Recl. Trav. Chim. Pays Bas, 1961, 80, 810-818.
- 4. Woodward, R. B.; Small, G., Jr J. Am. Chem. Soc. 1950, 72, 1297-1304.
- 5. Johnson, J. R.; Shiner, V. J., Jr. J. Am. Chem. Soc. 1953, 75, 1350-1355.
- 6. Bregman, J.; Bauer, S H. J. Am. Chem. Soc. 1955, 77, 1955-1965.
- 7. Katz, L.; Lipscomb, W. N. J. Org. Chem. 1952, 17, 515-517.
- 8. Kay, M. I.; Katz, L. Acta Crystallogr. 1958, 11, 897-898.
- 9. Mönnig, F.; Freizler, H.; Rudolph, H. D. Z. Naturforsch, A. 1967, 22, 1471-1473.
- 10. Clemens, R. J. Chem. Rev. 1986, 86, 241-318.
- 11. Andreades, S.; Carlson, H. D. *Organic Synthesis*; Baumgarten, H. E., Ed.; Wiley, New York, 1973, Coll. Vol. V, pp. 679–684.

- 12. Kato, T. Acc. Chem. Res. 1974, 7, 265-271.
- 13. Seidl, E. T.; Schaefer, H. F., III. J. Am. Chem. Soc. 1990, 112, 1493-1499.
- 14. Seidl, E. T.; Schaefer, H. F., III. J. Phys. Chem. 1992, 96, 657-661.
- 15. Staudinger, H.; Klever, H. W. Chem. Ber. 1906, 39, 968-971.
- Farnum, D. G.; Johnson, J. R.; Hess, R. E.; Marshall, T. B.; Webster, B. J. Am. Chem. Soc. 1965, 87, 5191–5197.
- 17. Baldwin, J. E.; Roberts, J. D. J. Am. Chem. Soc. 1963, 85, 2444-2445.
- 18. Sauer, J. C. J. Am. Chem. Soc. 1947, 69, 2444-2448.
- Sauer, J. C. Organic Synthesis; Rabjohn, N., Ed.; Wiley, New York, 1963, Coll. Vol. IV, pp. 560–563.
- 20. Dehmlow, E. V.; Fastabend, U. Synth. Commun. 1993, 23, 79-82.
- Abaecherli, C.; Miller, R. J. Kirk-Othmer Encyclopedia of Chemical Technology; Wiley, New York, 1995.
- 22. Woodward, R. B.; Hoffmann, R. Angew. Chem. Int. Ed. Engl. 1969, 8, 781–853. *The Conservation of Orbital Symmetry*; Verlag Chemie, Academic press, New York, 1970.
- 23. Calter, M. A.; Orr, R. K.; Song, W. Org. Lett. 2003, 5, 4745-4748.
- 24. Calter, M. A. J. Org. Chem. 1996, 61, 8006-8007.
- 25. Calter, M. A.; Guo, X. J. Org. Chem. 1998, 63, 5308-5309.
- 26. Calter, M. A.; Bi, F. C. Org. Lett. 2000, 2, 1529-1531.
- 27. Calter, M. A.; Guo, X.; Liao, W. Org. Lett. 2001, 3, 1499-1501.
- 28. Orr, R. K.; Calter, M. A. Tetrahedron 2003, 59, 3545-3565.
- 29. Wulferding, A.; Wartchow, R.; Hoffmann, H. M. R. Synlett 1992, 476-479.
- Baxter, G. J.; Brown, R. F. C; Eastwood, F. W.; Harrington, K. J. *Tetrahedron Lett.* 1975, 4283–4284.
- 31. Erickson, J. L. E; Collins, F. E., Jr.; Owen, B. L. J. Org. Chem. 1966, 31, 480-484.
- Hoffmann, H. M. R.; Eggert, U.; Walenta, A.; Weineck, E.; Schomburg, D.; Wartchow, R.; Allen, F. H. J. Org. Chem. 1989, 54, 6096–6100.
- 33. Masters, A. P.; Sorensen, T. S. Tetrahedron Lett. 1989, 30, 5869-5872.
- 34. Brady, W. T.; Ting, P. L. J. Org. Chem. 1976, 41, 2336-2339.
- 35. Brady, W. T.; Ting, P. L. J. Org. Chem. 1975, 40, 3417-3420.
- 36. Seidl, E. T.; Schaeffer, H. F., III J. Am. Chem. Soc. 1991, 113, 5195-5200.
- 37. Schaad, L. J.; Gutman, I.; Hess, B. A., Jr.; Hu, J. J. Am. Chem. Soc. 1991, 113, 5200-5203.
- 38. Salzner, U.; Bachrach, S. M. J. Am. Chem. Soc. 1994, 116, 6850-6855.
- Pascal, R. A., Jr.; Baum, M. W.; Wagner, C. K.; Rodgers, L. R.; Huang, D. J. Am. Chem. Soc. 1986, 108, 6477–6482.
- 40. Dehmlow, E. V.; Slopianka, M.; Pickardt, J. Liebigs Ann. Chem. 1979, 572-593.
- 41. Dehmlow, E. V.; Pickardt, J.; Slopianka, M.; Fastabend, U.; Dreschsler, K.; Soufi, J. *Liebigs Ann. Chem.* **1987**, 377–379.
- Hoffmann, H. M. R.; Wulff, J. M.; Kutz, A.; Wartchow, R. Angew. Chem. Int. Ed. Engl. 1982, 21, 83.
- 43. Wulff, J. M.; Hoffmann, H. M. R. Angew Chem. Int. Ed. Engl. 1985, 24, 605–606.
- 44. Hoffmann, H. M. R.; Walenta, A.; Eggert, U.; Schomburg, D. Angew. Chem. Int. Ed. Engl. 1985, 24, 607–608.

- 45. Hoffmann, H. M. R; Geschwinder, P. M.; Hollwege, H.-P.; Walenta, A. *Helv. Chem Acta* **1988**, *71*, 1930–1936.
- 46. Moore, H. W.; Wilbur, D. S. J. Org. Chem. 1980, 45, 4483-4491.
- 47. Moore, H. W.; Wilbur, D. S J. Am. Chem. Soc. 1978, 100, 6523-6525.
- 48. England, D. C.; Krespan, C. G. J. Org. Chem. 1970, 35, 3322-3327.
- 49. Huisgen, R.; Otto, P. J. Am. Chem. Soc. 1968, 90, 5342-5343.

**5.4.1.2** Cycloadditions with Alkenes and Dienes As noted in Section 5.4.1.1, the cycloaddition of diphenylketene (1) with cyclopentadiene to form cyclobutanone 2 was discovered very early^{1,2} and is a characteristic ketene reaction. In this section, some prominent examples of the reaction are given, followed by a fuller discussion of some of the mechanistic features in Section 5.4.1.3. This process is reversible in only a few examples, and this method for forming ketenes is discussed in Section 3.4.1.



Diphenylketene gave cyclobutanones on reaction with ethylene, propene, and 1-hexene at temperatures of 85–115 °C.^{3,4} Under these conditions *E*- and *Z*-2-butene reacted stereospecifically, with retention of the alkene stereochemistry, within the limits of ¹H NMR detection (4-6%).^{3,4} Cycloaddition of these alkenes with Me₂C=C=O (**3**) was carried out at 105 °C,⁵ while *n*-BuCEt=C=O gave [2+2] cycloaddition with cyclohexene at 180 °C.⁶ In the reaction of the 2-butenes with Me₂C=C=O the *Z*-isomer reacted about twice as fast and gave a stereospecific reaction, whereas the less reactive *E*-isomer gave significant loss of the alkene stereochemistry (equations 2, 3).⁵ The lower reactivity of *E*-2-butene provided evidence that the reaction occurred with a near-orthogonal approach of the ketene to the alkene double bond, as shown in **4** (equation 2), and the loss of stereochemistry was strong evidence for a stepwise process involving the zwitterion **5**, which underwent partial conversion to **6** to avoid steric crowding (equation 3).⁵



The cycloaddition of *t*-BuC(CN)=C=O (8) with 2-methyl-2-butene proceeded regio- and stereoselectively to form 10, as shown by X-ray crystallography.⁷ This reaction was interpreted⁷ as occurring by a concerted pathway, but the results can also be explained by a pathway analogous to equation 2 in which the initial bond rotation in the intermediate 9 that determines the stereochemistry of the final product involves rotation of the cyano group away from the *syn* methyl (equation 4).



Preparative cycloadditions require sufficiently reactive Ketene and alkene partners. Thus the highly reactive phenoxyketene (11) generated by dehydrochlorination gave [2+2] cycloaddition with [60]fullerene forming 12 in a sequence including acylation of the initially formed cyclobutanone enol (equation 5).⁸ Other aryloxyketenes reacted similarly, but there was no reaction with other less reactive ketenes.

Reactive unsaturated groupings were usually required to trap the more reactive ketenes, which otherwise were consumed by self-reaction or other pathways. Dienes have been particularly useful in capturing many of the most reactive ketenes, ^{9–28} such as FCH=C=O (**13**, equation 6).^{9,12,15} Electron-rich ketenophiles such as vinyl ethers^{3,4,29} and imines are discussed in later sections.



Palladium catalysis of the [2+2] cycloaddition to cyclopentadiene of ketenes RCBr=C=O (15) produced by the dehydrochlorination route forming 16 has been reported to give increased yields and selectivity (equation 7).³⁰ With

longer-chain alkyl groups ( $C_8$ – $C_{16}$ ) as ketene substituents the bromine was exclusively in the less hindered *exo* position.



The frequent preference of the larger substituent for the *endo* position in this cycloaddition is a hallmark of ketene chemistry and has been taken as diagnostic evidence of the  $[\pi 2_s + \pi 2_a]$  mode of addition (equation 8). The transition state 17 shows that the substituent in the least hindered position in the transition state will occupy the *endo* position in the final product, which is seemingly more hindered. Bartlett¹³ used the phrase "masochistic steric effect" to describe this situation, where the steric hindrance of a group was reversed between the transition state and the product.



This simple picture was not as clear-cut as it seemed, as the *endo* isomer of **18** is not necessarily the least stable isomer, and indeed in many cases is more stable than the *exo* isomer. This was demonstrated by equilibration of the two isomers of **18**, in which the equilibrium product ratios in Table 5.3 were found.⁹



 TABLE 5.3
 Product Percentages at Equilibrium for 7-Substituted

 Bicyclo[3.2.0]hept-2-en-6-ones 18⁹

R	endo	exo
F	88.7	11.3
Cl	86.7	13.3
CH ₃	76.3	23.7
Et	64.4	35.6
<i>i</i> -Pr	56.9	43.1
t-Bu	9.6	90.4
Ph	67	33

The explanation for this unanticipated result⁹ was that the cyclobutanone ring in **18** was not planar but puckered, and existed in the conformation depicted below for *endo*-**18**. This was established by an analysis of the ¹H NMR spectra, in particular the various H-H coupling constants. The preference for *endo*-**18** was explained by unfavorable steric interactions between the carbonyl group and H₁ and H₅ in the alternative conformation *exo*-**18**, and possibly by a favorable electronic interaction between H₅ and the carbonyl group in *endo*-**18** as well. The *endo* position in *endo*-**18** was more crowded, but this was offset for all but the *t*-butyl substituent by a preference of H₇ for the *exo* position so as to be orthogonal to the carbonyl group. ⁹ A stabilizing hyperconjugative interaction of the C-H₇ bond and the carbonyl may also be envisioned in this geometry.



There was a strong solvent effect on the *exo/endo* methyl product ratio of the reaction of ketenes  $CH_3CHal=C=O$  (Hal = Cl, Br) with cyclopentadiene, as shown in Table 5.4.¹⁷ This result suggested that the formation of the less stable product in these reactions was not good evidence of a concerted mechanism. Further, this significant effect of solvent polarity may be taken as evidence that a polar zwitterionic intermediate was formed in the reaction, and that the partitioning of this intermediate between *exo* and *endo* products was affected by solvent polarity. The more polar solvents favored the *exo*-methyl isomer, and this effect may derive from a longer lifetime of a polar intermediate, favoring a more stable product.

The *exo/endo* stereochemical preference for cycloaddition was greatest for cyclopentadiene compared to some other alkenes (Table 5.5).¹⁹ It has also been shown that cycloadditions of MeCCl=C=O and MeCBr=C=O with cyclopentadiene and cyclopentene gave similar selectivities.¹⁸

The fact that for many alkenes the *exo/endo* product ratio for ketene cycloadditions was insensitive to alkene structure, and rather close to 1, was evidence that

Hal	Solvent	exolendo (Me)	Hal	Solvent	exolendo (Me)
Cl	hexane	1:4.3	Br	hexane	1.0:0.71
Cl	Et ₃ N	1:2.2	Br	Et ₃ N	1:0.28
Cl	CHCl ₃	1:1.6	Br	CH ₃ CN	1:0.14
Cl	CH ₃ CN	1.0:0.59			

TABLE 5.4 Product Stereochemistry for [2+2] Cycloaddition of CH₃CHal=C=O with Cyclopentadiene in Different Solvents¹⁷

Alkene	exolendo (Me)
Cyclopentadiene	>20
EtOCH=CH ₂	2.3
Dihydropyran	1.7
Cyclohexene	2
Cyclooctene	1

 TABLE 5.5
 Exo/endo
 Product Ratios for [2+2]
 Cycloaddition of PhCMe=C=O in Hexane with Different Alkenes¹⁹

these cycloaddition transition states were not highly ordered. It was also notable that  $EtOCH=CH_2$  and cyclohexene, expected to involve extremes in their sensitivity to solvent polarity, gave similar results.

Rate data for the reaction of  $Ph_2C=C=O$  with some alkenes and dienes are given in Table 5.6.^{23–25} In the reaction with isoprene and chloroprene the ratios of addition to the substituted and unsubstituted double bonds were 1.5:1 and 1:4, respectively.²⁴ The reactions were interpreted in terms of "near-concerted" processes,²⁴ but there was a very large accelerating effect for cyclopentadiene  $(3 \times 10^4 \text{ relative to cyclopentene})$ , which was well suited for conjugative stabilization in a polar transition state. The regiochemistry of the reaction of isoprene also showed a significant polar effect, with a preference for reaction at the more hindered double bond.

In the reaction of *t*-BuC(CN)=C=O (8) with Me₂C=CHCH=CMe₂ the adducts **19** and **20** were formed, and on heating in benzene containing 7% EtOH the proportion of **20** in the mixture increased from 67 to 85% (equation 9).²⁶ The formation of two products, and the apparent isomerization of **19** to **20**, were interpreted as showing that a zwitterionic intermediate was involved in the reaction that was not trapped by EtOH. In the reaction of **8** with **21** the initial product showed a 5:1 preference for **22**, but on prolonged heating **23** was the only product observed

· /		
Alkene	<i>k</i> ₂ (40 °C, PhCN)	<i>k</i> ₂ (30 °C, THF)
Cyclopentene	1.25	
$CH_2 = CHCH = CH_2$		1.17
Cyclopentadiene	$3.73  imes 10^4$	280
Cyclohexadiene	53.0	
<i>E</i> -CH ₃ CH=CHCH=CH ₂	17.5	
Z-CH ₃ CH=CHCH=CH ₂	9.3	
CH ₂ =CMeCH=CH ₂	4.9	
Dihydropyran	23.5 ^{<i>a,b</i>}	16.5

TABLE 5.6 Rate Constants for Reaction of Ph_2C=C=O with Alkenes with Dienes  $(M^{-1}\ s^{-1}\times 10^6)^{23-25}$ 

^aIn THF; 34 (toluene), 16 (n-PrCN), and 9.4 (DMF).²⁵

 ${}^{b}\Delta H^{\ddagger} = 9.1$  kcal/mol,  $\Delta S^{\ddagger} = -43$  eu.



An illustration of the polar effect in ketene [2+2] cycloadditions was found in the reaction of dichloroketene (25) with the vinylcyclopropane 24 used in prostaglandin synthesis (equation 11).^{31,32} The selectivity for formation of the product 27 was attributed to the stabilization of the positively charged carbon in transition state 26 by the adjacent cyclopropyl group.



Cycloadditions of ketenes to 3,3-dimethylcyclopentene (**28**) show an interesting variation with ketene structure.³³ There was a reversal in the regiochemistry of the preferred product for  $CCl_2=C=O$  (**25**, equation 12) compared to *t*-BuC(CN)=C=O (**8**, equation 13). The results were interpreted in terms of nonparallel approach of the reactants and concerted [2+2] processes, but could also be explained by stepwise processes. Thus, formation of zwitterionic transition states (or intermediates) would be favored by both steric and polar factors (equation 12). In equation 13 the most stable product was formed, and in a stepwise process this would involve initial formation of a bond at the most crowded carbon. This arrangement, however, minimized the interaction of the geminal dimethyl groups with the *tert*-butyl and was evidently the most favorable arrangement. Diphenylketene at 100 °C adds to **28** as in equation 12, but at 200 °C dissociates to the reactants with a half-life of 15–20 min.³³



(equation 10).



Diphenylketene (1) underwent [2+2+2] cycloaddition with the 18-electron  $\eta^4$ arene complex  $[Mn(\eta^4C_6H_6)(CO)_3]^ [Ph_3PNPPh_3]^+$  (33), possibly through the monoadduct 34, to give the dihydroisochroman-3-one 35, with the structure confirmed by X-ray (equation 14).³⁴



Reaction of 1,3,5-tri-*t*-butylcyclopentadiene (**36**) with dichloroketene (**25**) gave the adducts **37–39** (equation 15).³⁵ The [2+2] cycloadduct **37** (54%) formed when **25** was generated from CHCH₂COCl and Et₃N, but **38** (60%) was formed when **25** was made from CCl₃COCl and Zn, and the latter reaction was proposed to involve ZnCl₂-promoted formation of a zwitterion that underwent rearrangement and cyclization (equation 16).³⁵ The formation of **39** was proposed to occur by several paths. Reaction of **36** with CHCl=C=O (**40**) gave the [2+2] cycloadduct **41**, which rearranged on heating to the [4+2] adduct **42** (55%, equation 17).³⁵ This was proposed to proceed through a diradical intermediate, but a zwitterionic intermediate as shown also appears to explain the results.





Cyclopropylalkenes **43** gave [2+2] cycloaddition with dichloroketene (**25**) without cyclopropane ring opening (equation 18).³⁶ Thus, while the regiochemistry for the reaction was consistent with positive charge development adjacent to the cyclopropyl in the transition state, it was not sufficient to induce ring opening.



Norbornene (**45**) and other bicyclo[2.2.1]heptenes^{37–41} gave [2+2] cycloaddition with *t*-BuC(CN)=C=O (**8**), with formation of adducts such as **46** with the *tert*-butyl group *syn* to the CH₂ bridge (equation 19).³⁷ This formation of the more crowded product was characteristic of ketene cycloadditions, and was consistent with a process in which the initial bond rotation that determined the product stereochemistry moved the cyano away from the congested norbornyl fragment (equation 19).



The reaction of *t*-BuC(CN)=C=O (8) with norbornadiene (47) led to both the [2+2] cycloaddition product 48 and the product of [2+2+2] cycloaddition to the ketene carbonyl 49 (equation 20).⁴⁰ These products arose from competing *exo* and

*endo* attack of the ketene, and the ratio fo the two products showed little dependence on the solvent polarity, indicating that the two transition states had similar polarities.⁴⁰ The ketene (CF₃)₂C=C=O (**50**) gave only a product analogous to **49** on reaction with norbornadiene,⁴² whereas CCl₂=C=O and *i*-PrCCl=C=O gave the product analogous to **48**,^{43,44} as did MeCBr=C=O, but the latter ketene showed a solvent-dependent stereochemistry of the substituents in the product.⁴⁵ Ketene **8** reacted with 3,3-dimethylcyclopropene to give [2+2] cycloadducts and ring-opened products, while **50** gave only ring-opened products.⁴⁵ Reaction of bis(trifluoromethyl)ketene (**50**) with nortricylane **51** gave the cycloadduct **52** under mild conditions in 92% yield (equation 21).⁴⁶



Norbornene (45) reacted with diphenylketene (1) in refluxing benzene to produce the 1:1 adduct 53 (84%), but reaction with excess  $Ph_2C=C=O$  led to a 5:1 ratio of 53 and the 2:1 adduct 54 (equation 22).⁴⁷



#### **REFERENCES FOR SECTION 5.4.1.2**

- 1. Staudinger, H. Chem. Ber. 1907, 40, 115-118.
- 2. Staudinger, H. Liebigs Ann. 1907, 356, 51-123.
- 3. Binsch, G.; Feiler, L. A.; Huisgen, R. Tetrahedron Lett. 1968, 4497-4501.
- 4. Huisgen, R.; Feiler, L. A. Chem. Ber. 1969 102, 3391-3404.
- 5. Frey, H. M.; Isaacs, N. J. J. Chem. Soc. B 1970, 830-832.

- Martin, J. C.; Gott, P. G.; Goodlett, V. W.; Hasek, R. H. J. Org. Chem. 1965, 30, 4175–4180.
- 7. Brook, P. R.; Eldeeb, A. F.; Hunt, K.; McDonald, W. S. J. Chem. Soc. Chem. Commun. **1978**, 10–11.
- 8. Matsui, S.; Kinbara, K.; Saigo, K. Tetrahedron Lett. 1999, 40, 899-902.
- Rey, M.; Roberts, S. M.; Dreiding, A. S.; Roussel, A.; Vanlierde, H.; Troppett, S; Ghosez, L. *Helv. Chim. Acta* 1982, 65, 703–720.
- Brook, P. R.; Duke A. J.; Harrison, J. M.; Hunt, K. J. Chem. Soc. Perkin Trans. 1 1974, 927–932.
- Brook, P. R.; Duke, A. J.; Griffiths, J. G.; Roberts, S. M.; Rey, M.; Dreiding, A. S. *Helv. Chim. Acta* 1977, 60, 1528–1544.
- 12. Jacobson, B. M.; Bartlett, P. D. J. Org. Chem. 1973, 38, 1030-1041.
- 13. Bartlett, P. D. Pure Appl. Chem. 1971, 27, 597-609.
- Rey, M.; Roberts, S.; Diffenbacher, A.; Dreiding, A. S. *Helv. Chim. Acta* 1970, 53, 417–430.
- 15. Brady, W. T.; Hoff, E. F., Jr. J. Am. Chem. Soc. 1968, 90, 6256.
- 16. Brady, W. T.; Roe, R., Jr. J. Am. Chem. Soc. 1970, 92, 4618-4621.
- Brady, W. T.; Roe, R., Jr.; Hoff, E. F.; Parry, F. H., III. J. Am. Chem. Soc. 90 92, 146– 148.
- 18. Brady, W. T.; Roe, R., Jr. J. Am. Chem. Soc. 1971, 93, 1662-1664.
- 19. Brady, W. T.; Parry, F. H., III; Stockton, J. D. J. Org. Chem. 1971, 36, 1486-1489.
- Brady, W. T.; Hoff, E. F.; Roe, R., Jr.; Parry, F. H., Jr. J. Am. Chem. Soc. 1969, 91, 5679– 5680.
- 21. Brady, W. T.; Holifield, B. M. Tetrahedron 1967, 23, 4251-4255.
- Brady, W. T.; Parry, F. H., III; Roe, R., Jr.; Hoff, E. F., Jr. *Tetrahedron Lett.* 1970, 819–822.
- 23. Huisgen, R.; Otto, P. Chem. Ber. 1969 102, 3475-3485.
- 24. Brady, W. T.; O'Neal, H. R. J. Org. Chem. 1967, 32, 2704-2707.
- 25. Brady, W. T.; O'Neal, H. R. J. Org. Chem. 1967, 32, 612-614.
- 26. Al-Husaini, A. H.; Khan, I.; Ali, S. A. Tetrahedron 1991, 47, 3845-3852.
- Holder, R. W.; Freiman, H. S.; Stefanchik, M. T. J. Org. Chem. 1976, 41, 3303– 3307.
- 28. Marvel, C. S.; Kohan, M. I. J. Org. Chem. 1951, 16, 741-756.
- 29. Martin, J. C.; Goodlett, V. W.; Burpitt, R. D. J. Org. Chem. 1965, 30, 4309-4311.
- Fairlamb, I. J. S.; Dickinson, J. M.; Cristea, I. M. *Tetrahedron Lett.* 2000, 41, 3739–3742. *Tetrahedron* 2001, 57, 2237–2246.
- 31. Corey, E. J.; Arnold, Z.; Hutton, J. J. Tetrahedron Lett. 1970, 307-310.
- 32. Kelly, R. C.; VanRheenen, V.; Schletter, I.; Pillai, M. D. J. Am. Chem. Soc. 1973, 95, 2746–2747.
- 33. Hassner, A.; Cory, R. M.; Sartoris, N. J. Am. Chem. Soc. 1976, 98, 7698-7704.
- 34. Lee, S.; Geib. S. J.; Cooper, N. J. J. Am. Chem. Soc. 1995, 117, 9572-9573.
- Dehmlow, E. V.; Bollmann, C.; Neumann, B.; Stammler, H,-G. *Liebigs Ann.* 1995, 1915– 1919.

- Bessmertnykh, A. G.; Bubnov, Yu, N.; Voevodskaya, T. L.; Donskaya, N. A.; Zykov, A. Yu. *Zh. Org. Khim.* 1990, 26, 2343–2355; *Engl. Transl.* 1990, 26, 2027–2033.
- 37. Moore, H. W.; Gheorghiu, M. D. Chem. Soc. Rev. 1981, 10, 289-328.
- 38. Gheorghiu, M. D.; Draghici, C.; Parvulescu, L. Tetrahedron, 1977, 33, 3295-3299.
- Gheorghiu, M. D.; Filip, P.; Draghici, C.; Parvulescu, L. J. Chem. Soc., Chem. Commun. 1975, 635–636.
- 40. Brook, P. R.; Hunt, K. J. Chem. Soc. Chem. Commun. 1974, 989-990.
- 41. Gheorghiu, M. D.; Parvulescu, L.; Turdibekov, K. M.; Struchkov, Y. T. *Rev. Roum. Chem.* **1990**, *35*, 427–435.
- 42. England, D. C.; Krespan, C. G. J. Org. Chem. 1970, 35, 3300-3307.
- 43. De Selms, R. C.; Delay, F. J. Am. Chem. Soc. 1973, 95, 274-276.
- 44. De Selms, R. C.; Delay, F. J. Org. Chem. 1972, 37, 2908-2110.
- 45. Aue, D.H.; Shellhamer, D. F.; Helwig, G. S. J. Chem. Soc. Chem. Commun. 1975, 603-604.
- 46. Petrov, V. A.; Davidson, F.; Smart, B. E. J. Fluorine Chem. 2004, 125, 1543-1552.
- 47. Feiler, L. A.; Huisgen, R.; Koppitz, P. J. Am. Chem. Soc. 1974, 96, 2270-2271.

5.4.1.3 Mechanism of Ketene [2+2] Cycloadditions with Alkenes The facile [2+2] cycloaddition reaction of ketenes is a centerpiece of one of the classic mechanistic problems of organic chemistry, namely, the readiness of many alkenes to undergo thermal concerted [4+2] cycloaddition with dienes, whereas thermal [2+2] cycloadditions of ordinary alkenes are clearly stepwise, if they occur at all. The appearance in the mid-1960s of the Woodward-Hoffmann rules¹⁻³ had a profound effect on the chemical community and provided the explanation that  $[\pi 4_s + \pi 2_s]$  cycloadditions were allowed by orbital symmetry, whereas  $[\pi 2_s + \pi 2_s]$  processes were forbidden.

Experimentally, it was known that unsymmetrically substituted ketenes 1 with one substituent  $R^L$  more sterically demanding than the other reacted by [2+2] cycloaddition with alkenes 2 with more sterically demanding substituents on one side than the other to produce preferentially cyclobutanones 4 with the larger substituents on one side of the ring (equation 1). In many cases there was no loss of the original stereochemistry of the alkene. This was explained as resulting from an almost orthogonal approach of the ketene to the alkene with the smaller ketene substituent  $R^S$  approaching the less hindered side of the alkene and a transition state 3 with antarafacial bonding, so that initially the smaller substituent  $R^S$  moved away from the alkene, but with the result that the larger substituent  $R^L$  was *cis* to the larger alkene substituents in the product.



Woodward and Hoffmann³ put forward the reasons for the tendency of ketenes to undergo concerted [2+2] cycloadditions as opposed to a [4+2] pathway. First, ketenes are powerful electrophiles; specifically, the  $\pi^*$  C=O orbital is quite low in energy and very susceptible to nucleophilic attack. This carbon is also only disubstituted, which facilitates its reactivity even more by the absence of steric interactions. Thus, nucleophilic attack is greatly facilitated and takes place rapidly, usually resulting in [2+2] cycloaddition. However, the reactivity of the C=C linkage of the ketene in a concerted [ $\pi^4_s + \pi^2_s$ ] process is not facilitated by the carbonyl group so in this process the ketene resembles a normal alkene and is not very reactive. Only in unusual cases in which the intermediates are long-lived does [4+2] cycloaddition result, but probably not by concerted processes (Section 5.4.3).

Ketene cycloadditions played a key role in the acceptance of these proposals, as the  $[\pi 2_a + \pi 2_s]$  cycloaddition is an allowed process. The existing data on ketene cycloadditions appeared to fit this proposal, and it was stated with authority regarding this mechanism that "the evidence is now conclusive that it is concerted."³ Interestingly, the first publication on the subject stated, "it must be concluded that those cycloadditions which do occur in cases prohibited by our selection rules for concerted reactions must proceed through multistep mechansisms, (e.g., dimerization of allenes and ketenes..."¹ However, after further analysis, it was realized that ketenes possessed structures permitting the  $[\pi 2_a + \pi 2_s]$  cycloaddition.

The evident readiness of ketenes to react in this way, whereas ordinary alkenes do not, was explained by the proposal that the ketene transition state **3** was much more suitably disposed to the  $[\pi 2_s + \pi 2_a]$  process than that for alkenes because of the lesser steric demands of the ketene, and also because of the favorable bonding interactions between both carbons of the alkene and the electron-deficient *p* orbital of the carbonyl carbon.

An alternative description suggested by Zimmerman in 1970 stated that this process was best described in Hückel-Möbius terms, and it allowed for six  $\pi$ -electron systems with no sign inversions or four  $\pi$  electrons with one sign inversion.^{4,4a} This was described by Baldwin and Kapecki as involving a p orbital on oxygen⁵ or as an (S+S+S) process, often given as an [ $\pi 2_s + (\pi 2_s + \pi 2_s)$ ] pathway using the C=C and C=O  $\pi$  bonds of the ketene and the  $\pi$  bond of the alkene.^{4,5} In this mechanism the major initial bond formation involves attack by the more nucleophilic of the alkene carbons on the carbonyl  $\pi$ -orbital at C₁ of the ketene. To a lesser extent the other carbon of the alkene carries out electrophilic attack on C₂ of the ketene. Thus, addition to the ketene involves nucleophilic attack at C₁ and electrophilic attack on the orthogonal orbital at C₂.

A counterproposal was made at the same time that instead of "a (more or less) synchronous cycloaddition," as proposed by Woodward and Hoffmann, cycloaddition of ketenes and alkenes occurred by a two-step reaction involving a dipolar intermediate.^{6,7} This intermediate **5** was proposed to result from initial bond formation from the more nucleophilic carbon of the alkene, to the carbonyl carbon with an interaction between the cationic carbon and the enolate system of the dipolar

intermediate (equation 2).^{4,5} This mechanism also involved an initial almost orthogonal approach of the ketene to the alkene from the least hindered direction, and a second step with initial bond rotation at  $C_2$  of the ketene so that the smaller substituent on the ketene moved away from the nearest alkene substituent. As the second bond formed, this brought the larger ketene and alkene substituents into a *cis* relationship (equation 2).

Many of the experimental criteria that have been applied to the differentiation of concerted and stepwise reaction pathways for ketene cycloaddition are ambiguous, as an unsymmetrical concerted process and a stepwise process have many of the same features. In cases where there is a loss of stereochemical homogeneity in the alkene,^{8,9} or where there is isolation, spectroscopic detection, or interception of an intermediate that lies on the cycloaddition pathway, there is unequivocal evidence for a stepwise process. Such intermediates have indeed been detected in a number of cycloaddition reactions of ketenes, as discussed below, and so these are known to be stepwise. The very high stereoselectivity observed in many ketene cycloadditions suggests that these reactions have all the features of concerted asynchronous reactions, but in some cases newer and better experimental methods could give more accurate measures of the degree of stereoselectivity and reexamination of some of these could be informative. Perhaps the best evidence for the occurrence of concerted processes is the failure to detect an intermediate in a number of highlevel theoretical studies. However, even in these cases there is still doubt that the theoretical methods provide a sufficiently accurate description of reaction in solution. Thus, proof that ketene cycloadditions proceed by concerted pathways remains an elusive goal.

The types of evidence available regarding the mechanism of ketene [2+2] cycloadditions include (a) theoretical analysis, (b) stereochemistry, (c) isotope effects, (d) substituent effects, (e) solvent effects, (f) pressure effects and (g) activation parameters. However, as described below, the effects of substituents, solvent effects, pressure effects and activation parameters are expected to be similar for the two alternative mechanisms under consideration, as both are bimolecular and lead to a polarized transition state. Similarly, both mechanisms have similar stereochemical outcomes, although it has often been argued that very high retention of configuration of the alkene stereochemistry is indicative of a concerted process.

There have been many theoretical studies of the mechanism of ketene cycloadditions, 10-22 and these studies are helping to define the possible variations in the transition states and the factors that influence them. The studies emphasize the



Scheme 5.3 Calculated MP2/ $6-31G^*$  transition state for [2+2] calculation of ketene and ethylene (reprinted from reference 14 with permission of Wiley VCH).

effects of substitutents on the ketenes and alkenes which can cause significant changes in the energetics of the reactions.

Wang and Houk¹³ calculated the geometry and energy of the transition state for the [2+2] cycloaddition of ketene and ethylene at the MP2/6-31G* level and found a highly asymmetric geometry (Scheme 5.3).¹⁴ The forming bond length of the carbonyl carbon is 1.78 Å, while that of the second forming bond is 2.43 Å.

This reaction was also studied by Bernardi et al.¹² at the 4-31G level. The authors reported one transition state resembling that shown in Scheme 5.3 formed by attack in the ketene plane. These authors also located a second transition state similar in energy to the first but with attack of the ethylene perpendicular to the ketene plane.

Both of these studies emphasized the highly asynchronous nature of the reaction.^{12,13} The carbonyl carbon of the ketene interacts strongly with both ethylene carbons.¹⁴ Mulliken population analysis shows an increase in negative charge on oxygen ( $\Delta q = -0.16$ ) and a concomitant increase in positive charge on the weakly bonding carbon from ethylene ( $\Delta q = +0.22$ ), showing the zwitterionic character of the structure.¹³

High-level *ab initio* molecular orbital calculations for the cycloaddition reaction between ketene and cyclopentadiene gave predicted reaction energies (MP4SDQ + ZPVE) for the products **8–12**, as shown (kcal/mol), with calculated activation energies (CCSD(T) + ZPVE) given in parentheses (Figure 5.2).²⁵ The [4+2] cycloaddition product **8** was predicted to be the most stable, but the barrier for formation of the [2+2] product **9** was less by 12.0 kcal/mol.²⁵ However, these studies did not include the [4+2] product **13** from addition to the C=O bond.

The transition state for the [4+2] reaction forming **8** had almost equal bond formation to the ketene carbonyl carbon  $C_1$  and to  $C_2$ , with bond distances of 2.187 and 2.276 Å, respectively and the reaction was described as a synchronous



Figure 5.2 Calculated MP4SDQ/ $6-31G^*//MP2/6-31G^*$  reaction energies (kcal/mol) for cycloaddition of ketene (6) with cyclopentadiene (7), with activation energies, (kcal/mol) in parentheses.

concerted process.²⁵ By contrast, the transition state for the [2+2] cycloaddition forming **9** had major bond formation to C₁, with a bond distance of 1.636 Å and no bond to C₂. There was a net charge transfer from the cyclopentadienyl moiety to the ketene of 0.4 electron.

However, a major reevaluation of ketene/alkene cycloaddition chemistry has been necessitated by the finding that the kinetically favored pathway in the prototypical reaction of diphenylketene (14) with cyclopentadiene at -20 °C in CH₂Cl₂ is [2+2] cycloaddition with the carbonyl group forming 15 (equation 3).^{26,27} Upon warming, 15 was converted by a [3,3] sigmatropic Claisen rearrangement to the previously observed cyclobutanone 16, as reported in the very first example of a ketene/alkene cycloaddition.^{28,29} Reactions of Ph₂C=C=O with acyclic 1,3-dienes also gave mixtures of [4+2] and [2+2] cycloadducts.²⁷ Theoretical analyses of these pathways have also been carried out.^{26,27,30}



A [4+2] cycloaddition to the C=O bond of a ketene has also been found for the reaction of  $(CF_3)_2C=C=O$  (17) and EtOCH= $CH_2$ .³¹ Reaction in hexane at 0 °C gave the oxetane 18 as detected by NMR, and this isomerized at 50 °C to the cyclobutanone 19 and the enone 20, isolated in 67 and 13% yields, respectively, when the cycloaddition was performed at 100 °C (equation 4).³¹ Reaction in  $CH_2Cl_2$  at -80 °C permitted isolation of 18, which at -20 °C gave a deep blue intermediate assigned as the zwitterion 21. Reaction of 18 in acetone gave the adduct 22 from 21 (equation 5).³¹



Evidence for nonconcerted mechanisms in some cases comes from studies of intramolecular [2+2] cycloadditions of ketenes with alkenes, which have revealed examples that proceed quite readily by net  $[\pi 2_s + \pi 2_s]$  stereochemistry (Section 5.4.4). For example, cyclization of the ketene **23** derived by dehydrochlorination forming **24** was evidently stepwise, for a concerted  $[\pi 2_a + \pi 2_a]$  *syn* addition was electronically forbidden, and an allowed  $[\pi 2_s + \pi 2_a]$  reaction was not geometrically feasible (equation 6).³²



The regiochemistry of ketene [2+2] cycloaddition with alkenes is almost invariably that expected for a species with dipolar charge, as shown by the examples of dichloroketene (**25**) addition to cyclopentadiene (equation 7) or to propene (equation 8).³³ Perturbation theory predicts this regiochemistry, in that bonding occurs between the carbonyl carbon of the ketene, the atom with the highest coefficient in the LUMO and the carbon of the alkene with the highest coefficient of the HOMO. Bonding between these atoms also produces the most stabilized zwitterionic intermediate, which leads to the products shown in equations 7,8.



Alkene	$k_2 \times 10^4$	Alkene	$k_2 \times 10^4$
1-Pyrolidinoisobutene	$2.4 \times 10^{5}$	$PhC(OMe) = CH_2$	12
2,3-Dihydrofuran	$1.04 \times 10^3$	E- $n$ -PrOCH=CHCH ₃	0.6
Cyclopentadiene	372	EtOCH=CMe ₂	0.29
2-Ethoxypropene	227	PhCH=CH ₂	0.23
1-Morpholinoisobutene	169	$PhCMe=CH_2$	0.038
Z-n-PrOCH=CHCH ₃	110	Cyclopentene	0.0125
EtOCH=CH ₂	45	$CH_2 = CHCN$	No reaction
Z-EtOCH=CHMe	109 ^a	Z-EtOCH=CHPr-i	$117^{a}$
<i>E</i> -EtOCH=CHMe	$1.29^{a}$	E-EtOCH=CHPr-i	$0.742^{a}$
Z-EtOCH=CHEt	$1.28^{a}$	Z-EtOCH=CHBu-t	3.6 ^{<i>a</i>}
E-EtOCH=CHEt	$1.20^{a}$	E-EtOCH=CHBu-t	$0.054^{a}$

TABLE 5.7 Rate Constants for Reaction of Alkenes with  $Ph_2C=C=O(k_2, M^{-1} s^{-1})$  in PhCN at 40.3 °C^{35,39}

^aReference 35.

The stereochemistry of addition is such that the largest substituents on the ketene and alkene are almost always in the *syn* position on the cyclobutanone.^{34–39} However, as discussed in Section 5.4.1.2, this product is also sometimes the thermodynamically most stable isomer. Z-Alkenes are also significantly more reactive than the *E*-isomers toward Ph₂C=C=O (Tables 5.7–5.9).^{34,35,39} and Me₂C=C=O (Table 5.10).⁴⁰ These results are explained by an intermediate or transition state **5** in which the more nucleophilic carbon C₁ attacks the carbonyl carbon in the ketene plane and the more bulky C_β terminus of the ketene is directed away from the alkene substituents (equation 2). For convenience this structure is often represented with an orthogonal alignment of the ketene and alkene, but theoretical studies indicate that the carbonyl carbon will tilt more closely toward the more nucleophilic carbon of the alkene, with C₂ of the ketene skewed toward the other alkene carbon as in Scheme 5.3, and also higher above the ketene plane than the carbonyl carbon. A [ $_{\pi}2_{s}+_{\pi}2_{a}$ ] or [ $_{\pi}2_{s}+(_{\pi}2_{s}+_{\pi}2_{s}$ ]] pathway from **5** leads to a *cis* disposition of all three R groups in the product, as is usually observed (equation 2).

The reaction of *t*-BuC(CN)=C=O (**29**) with *Z*- and *E*-cyclooctene forming **30a** and **30b**, respectively, provided an impressive example of the stereoselectivity of ketene cycloadditions.³⁶ In each case the reaction gave retention of the cyclooctene

TABLE 5.8 Rate Constants for [2+2] Cycloaddition of Enol Ethers with Ph_2C=C=O at 23  $^\circ C~(s^{-1}~M^{-1}\times 10^6)^{40}$ 

Enol Ether	$k_2$ (Z)(CCl ₄ )	$k_2$ (Z)(CDCl ₃ )	$k_2$ (E)(CCl ₄ )	$k_2(E)(\text{CDCl}_3)$
MeOCH=CHMe	48	880	0.303	5.6
MeOCH=CHEt	73	732	0.483	3.8
EtOCH=CHMe	258	2050	2.2	13
EtOCH=CHEt	173	2530	1.48	8.58

R Pł	$H_{C} Me$ $H_{C} OEt$ $H^{C} OEt$ $31a$	$\begin{array}{c} H \\ C \\ R \\ H \\ C \\ H \\ C \\ OEt \end{array}$	$H_{C}^{Me}$ $R_{H}^{R}_{H}^{H}_{H}^{H}_{H}^{H}_{H}^{H}_{H}^{H}_{H}^{H}_{H}^{H}$ $H$	$R^{\text{Ph}} \xrightarrow{\text{H}} C = 0$ $EtO^{C} H$ <b>31d</b>
Ete	$\begin{array}{c} Ph \\ R \\ H \\ H \\ \end{array} Me \\ H \\ 32a \end{array}$	$ \begin{array}{c}                                     $	Ph····· EtO H H Me 32c	$ \begin{array}{c}     Ph & O \\     R & H \\     EtO & H \\     H & Me \\     32d \end{array} $
R = Me Et <i>n</i> -Pr <i>i</i> -Pr <i>t</i> -Bu	985 77 65 —	130 65 62 26 5.1	61 3.5 2.4 0.022	0.40 0.35 0.13 0.028

TABLE 5.9 Partial Rate Constants ( $k_2$ ,  $s^{-1} M^{-1} \times 10^6$ ) for Reaction of Z- and *E*-Ethoxypropene with PhCR=C=O in PhCN at 40 °C³⁴

configuration and led to the stereoisomer with the *t*-butyl in the seemingly less stable position *cis* to the alkyl group (equations 9, 10). These results appear consistent with a concerted  $[\pi 2_s + \pi 2_a]$  cycloaddition whereas free zwitterionic intermediates would seem to easily convert to a more stable isomer if formed. Dichloroketene also adds stereospecifically to *Z*- and *E*-cyclooctenes.³⁷



The reactivity of  $Ph_2C=C=O$  with alkenes in PhCN decreased by a factor of  $10^7$  as the alkene structure became less electron rich (Table 5.7).³⁹ Comparable kinetic data were found for reactions of enol ethers with  $Ph_2C=C=O$  in  $CCl_4$  and  $CDCl_3$  (Table 5.8).⁴⁰ The span of reactivities was quite large and shows convincingly that very large polar effects are present in these reactions. The very high reactivity of the enamines in particular argues that zwitterionic intermediates are involved in these processes, as does the formation of noncyclized products in the reactions of some very nucleophilic alkenes. This evidence is discussed in Section 5.4.1.7.

The *Z/E* rate ratios in Table 5.8 varied from 115 to 160 in CCl₄ and from 150 to 295 in CDCl₃, confirming the greater reactivity of *cis* alkenes. The ratio  $k(CDCl_3)/k(CCl_4)$  varied from 5.7 to 18.5, showing a significant polar solvent effect on these reactions.

The reactions of the Z- and E-1-ethoxypropenes with PhCR=C=O proceed quantitatively to the cyclobutanones.³⁴ On the assumption that the products resulted exclusively from stereospecific  $[\pi^2_a + \pi^2_s]$  reactions through the complexes **31** shown, the rate and product data were quantitatively separated into the individual rate constants shown in Table 5.9.³⁴

The influence of R on the rates of the processes proceeding through the reaction complexes **31b** and **31d** was attributed both to the varying electronic properties of R and to a remote steric effect in which the more bulky groups R affected the conformation of the phenyl and increased steric repulsions in the transition states.³⁴ Because the electronic properties of these R groups varied much less than their steric bulk, the latter factor would appear to be most important.

The relative reactivities of Me₂C=C=O with alkenes determined by competitive techniques (Table 5.10)⁴¹ are generally consistent with the rate constants reported in Tables 5.7–5.9, although the latter sets cover a wider range of reactivities. The relative reactivities also show a large rate difference  $k(EtOCH=CH_2)/k(PhCH=CH_2) = 89$  that argues for highly polar transition states, at least for the vinyl ether.

Cycloaddition rate constants of  $Ph_2C=C=O$  with 1,1-diarylethylenes and substituted styrenes (Table 5.11) gave linear Hammett plots with the  $\sigma$  constants of the aryl substituents with  $\rho$  values of -0.78 and -0.73, repectively.⁵ These reactions were all first order in both [Ph₂C=C=O] and the alkene, but whereas the styrenes gave cyclobutanones, the diphenylethylenes gave adducts containing two molecules of Ph₂C=C=O and one of the alkene.⁴² It was proposed that the rate-limiting step for the latter reaction was a [2+2] cycloaddition to form a cyclobutanone followed by reaction with a second ketene. However, the structure of the final product and the

Alkene	<i>k</i> (rel)	Alkene	<i>k</i> (rel)
Z-2-butene	1.0	PhCH=CH ₂	0.1
Z-2-pentene	1.0	$4-MeOC_6H_4CH=CH_2$	2.6
1-butene	0.44	$4-MeC_6H_4CH=CH_2$	1.8
isobutene	0.124	$4-ClC_6H_4CH=CH_{28}$	0.36
E-2-butene	$8 imes 10^{-4}$	Cyclobutene	0.009
E-2-pentene	$10^{-3}$	Cyclopentene	0.19
1,3-butadiene	2.64	Cyclohexene	0.068
2-Me-2-butene	0.016	Cycloheptene	0.06
$Me_2C = CH_2$	No reaction	Cyclooctene	0.92
EtOCH=CH ₂	8.9	Methylenecyclopropane	1.81
<i>n</i> -BuOCH=CH ₂	9.0	EtOC≡CH	23.7

TABLE 5.10 Relative Reactivities in [2+2] Cycloaddition of Me_2C=C=O with Alkenes at 100  $^\circ C^{41}$ 

	$k(s^{-1} \times 10^{-4})$	$k(s^{-1} \times 10^{-4})$
Х	$Ar_2CH=CH_2$	ArCH=CH ₂
CF ₃	0.0167	2.85
Cl	0.0863	7.82
Н	0.113	$7.00^{a}$
CH ₃	0.212	9.43

TABLE 5.11 Rate Constants for Reaction of  $Ph_2C=C=O$  with  $(4-XC_6H_4)_2C=CH_2$  and  $4-XC_6H_4CH=CH_2$  at 120 °C in PhBr⁴⁵

 $^a6.66~(1,2\text{-}Cl_2C_6H_4),~6.50~([ClCH_2CH_2]_2O),~and~6.03~(DMF).$ 

details of the reaction have evidently not been settled. The [2+2] cycloaddition of Me₂C=C=O with substituted styrenes (Table 5.10) gave a  $\rho$  value of -1.4,⁴¹ indicative of modest charge development.

The large rate differences observed with the structural variations shown in Tables 5.7–5.10 are indicative of a high degree of positive charge buildup on the alkene. The substituent effects of the styrenes noted are more modest, and overall these trends are less than those observed in alkene protonations in water.⁴³ Thus, the EtOCH=CH₂/PhCH=CH₂ rate ratio for the latter reaction is 10⁵ and the  $\rho^+$  value for styrene protonation is -2.9.⁴³

The kinetics of the cycloaddition of *tert*-butylcyanoketene (**29**) with styrenes **33** gave a good correlation with  $\sigma_p^+$  parameters with  $\rho^+ = -2.48$  (r = 0.998), but not with  $\sigma_p$  values, and gave stereoselective formation of the less stable products **35** (equation 11).⁴⁴ Thus, the response of the reactivity of **29** to the styrene substituents, as measured by the  $\sigma_p^+$  value, was considerably more than for the corresponding reactions of dimethylketene at 100 °C and diphenylketene at 120 °C, which gave  $\rho$  values of -1.4 and -0.73, respectively. These results, and the failure to intercept any zwitterionic intermediates, were interpreted in terms of a one-step concerted but nonsynchronous cycloaddition involving an unsymmetrical transition state **34** with a high degree of zwitterionic character.⁴⁴ A stepwise process was proposed for the reaction of **29** with styrenes with 4-MeO, 4-F and 4-Cl substituents.^{45,46} For Ar = 9-anthryl **37** was the only observed product, and this was attributed to a stepwise process proceeding through intermediate **36** due to the bulk and electron-donor ability of the 9-anthryl group (equation 12).⁴⁶



X	$k_2$	Х	$k_2$
MeO	121	Н	1.59
<i>n</i> -Bu	5.92	4-Br	0.571
Me	9.78		

TABLE 5.12 Rates of Reaction of *t*-BuC(CN)=C=O with 4-XC₆H₄CH=CH₂ in C₆H₆ at 25 °C  $(M^{-1} s^{-1} \times 10^3)^{44}$ 

Sizable negative entropies of activation of [2+2] ketene cycloadditions^{39,42,44} are consistent with bimolecular processes proceeding by either concerted pathways or through zwitterionic intermediates.⁴⁷

Solvent effects on ketene cycloaddition reveal a rate ratio  $k(CH_3CN)/k(cyclohexane)$  of 160 for the reaction of Ph₂C=C=O with *n*-BuOCH=CH₂.³⁹ The comparable ratio for the same two solvents for the reaction of TCNE with *n*-BuOCH=CH₂ is about 2600.^{48–50} The latter reaction is established to be stepwise, and there is a qualitative resemblance between the magnitudes of the solvent effects of the two processes.

Isotope effects on ketene cycloadditions have been intensely studied but do not lead to unequivocal interpretations of the mechanism. For the reaction of Ph₂C=C=O with styrene, intermolecular competition studies indicated that  $k_{\rm H}/k_{\rm D}$  was 0.91 for the CH₂ carbon of styrene, consistent with a hybridization change from sp² to sp³, but for C₁  $k_{\rm H}/k_{\rm D}$  was 1.23.⁴⁵ For the reaction of Me₂C=C=O with PhCD=CH₂  $k_{\rm H}/k_{\rm D}$  at C₁ was found to be 0.80,⁴¹ in disagreement with the other results.^{5,51,53}

A further study of isotope effects involved all four possible  $Ph_2C=C=O$  and  $PhCH=CH_2$  structures monolabeled with ¹⁴C in the alkene carbons, as well as *E* and *Z* PhCH=CHD.⁵⁴ The  $k_H/k_D$  isotope effects were the same at 0884 ± 0.006, in agreement with the results of Baldwin and Kapecki.⁵ The  $k(^{12}C)/k(^{14}C)$  isotopes ranged from 1.0055 to 1.08, as shown in **34**. These isotope effects were interpreted as consistent with significant negative charge buildup on oxygen, greater bond breaking than formation at the carbonyl carbon, and essentially equal bond making and breaking at the other carbons.⁵⁴



The reaction of Ph₂C=C=O (14) with the selectively deuterated 5,5-dimethylcyclopentadienes 34 and 35 showed the ratio of formation of the product 37 with nondeuterated and deuterated double bonds  $k_{\rm H}/k_{\rm D}$  to be 0.84 ± 0.02 for 34 and 1.0 ± 0.01 for 35 (equation 13).⁵⁵ The large inverse isotope effect for 34 was interpreted as indicating that substantial bond formation occurred at C₁ in the transition state, with a change in hybridization from sp² to sp³, whereas the absence of a substantial isotope effect for 35 was interpreted as showing that there was little bond formation or change in hybridization at  $C_2$ .⁵⁵ Thus, these results favored a stepwise mechanism, and the transition state was depicted with the "usual" orthogonal geometry, **36**, with the diphenyl carbon of the ketene raised further above the diene plane than the carbonyl carbon to minimize steric interactions.⁵⁵



The effect of pressure on ketene cycloadditions also indicted a decrease in volume in the transition state, but did not provide a definitve differentiation between concerted and stepwise mechanisms.⁵⁶

Theoretical study of the ketene-imine cycloaddition supports a two-step process involving a zwitterionic intermediate for this reaction, as discussed in Section 5.4.1.7.^{17,18} The courses of the [2+2] cycloadditions of ketene with ethylene, methylenimine, and formaldehyde were compared by MP2/6-31G* calculations of the transition states and by intrinsic reaction coordinate examination of the approach to the transition states.^{19,30} All three reactions are characterized by an approach of the reactants that is skewed slightly from the orthogonal, and give transition states with skewed geometries with activation barriers of 26.9, 30.8, and 24.9 kcal/mol, for ethylene, methylenimine, and formaldehyde, respectively.

The ketene-ethylene complex has been directly observed and its structure determined by microwave spectroscopy (38).⁵⁷ This was reported to be "remarkably similar" to that calculated by Yamabe et al.¹⁹ for the early stages of the reaction.⁵⁷



# **REFERENCES FOR SECTION 5.4.1.3**

- 1. Hoffmann, R.; Woodward, R. B. J. Am. Chem. Soc. 1965, 87, 2046-2048.
- 2. Hoffmann, R.; Woodward, R. B. Acc. Chem. Res. 1968, 1, 17-22.
- Woodward, R. B.; Hoffmann, R. Angew. Chem. Int. Ed. Engl. 1969, 8, 781–853. The Conservation of Orbital Symmetry; Verlag Chemie, Academic Press, New York, 1970, pp.
- 4. Zimmerman, H. E. Acc. Chem. Res. 1971, 4, 272-280.

- 4a. Zimmerman, H. E. In *Pericyclic Reactions*; Marchand, A. P.; Lehr, R. E., Eds.; Academic Press, New York, 1977, Vol. 1, pp. 53–107.
  - 5. Baldwin, J. E.; Kapecki, J. A. J. Am. Chem. Soc. 1970, 92, 4874-4879.
  - 6. Wagner, H. U.; Gompper, R. Tetrahedron Lett. 1970, 2819-2822.
- 7. Gompper, R. Angew. Chem. Int. Ed. 1969, 8, 312-327.
- 8. Frey, H. M.; Isaacs, N. J. J. Chem. Soc. B 1970, 830-832.
- 9. Al-Hussaini, A. H.; Muqtar, M.; Ali, S. A. Tetrahedron 1991, 47, 7719-7726.
- 10. Burke, L. A. J. Org. Chem. 1985, 50, 3149-3155.
- Bernardi, F.; Bottoni, A.; Olivucci, M.; Robb, M. A.; Schlegel, H. B.; Tonachini, G. J. Am. Chem. Soc. 1988, 110, 5993–5995.
- Bernardi, F.; Bottoni, A.; Robb, M. A.; Venturini, A J. Am. Chem. Soc. 1990, 112, 2106–2114.
- 13. Wang, X.; Houk, K. N J. Am. Chem. Soc. 1990, 112, 1754-1756.
- 14. Houk, K. N.; Li, Y.; Evanseck, J. D. Angew. Chem., Int. Ed. Engl. 1992, 31, 682-708.
- Pericas, M. A.; Serratosa, F.; Valenti, E. J. Chem. Soc. Perkin Trans. 2 1987, 151– 158.
- 16. Valenti, E.; Pericas, M. A.; Moyano, A. J. Org. Chem. 1990, 55, 3582-3593.
- 17. Sordo, J. A.; Gonzalez, J.; Sordo, T. L. J. Am. Chem. Soc. 1992, 114, 6249-6251.
- Lopez, R.; Sordo, T. L.; Sordo, J. A.; Gonzales, J. J. Org. Chem. 1993, 58, 7036– 7037.
- 19. Yamabe, S.; Minato, T.; Osamura, Y. J. Chem. Soc. Chem. Commun. 1993, 450-452.
- 20. Rzepa, H. S.; Wylie, W. A. Int. J. Quantum Chem. 1992, 44, 469-476.
- Reguero, M.; Pappalardo, R. R.; Robb, M. A.; Rzepa, H. S. J. Chem. Soc. Perkin Trans. 2, 1993, 1499–1502.
- 22. Fang, D.; Fu, X. Int. J. Quantum Chem. 1992, 43, 669-676.
- 23. Rzepa, H. S.; Wylie, W. A. J. Chem. Soc. Perkin Trans. 2, 1991, 939-946.
- 24. Nguyen, M. T.; Ha, T.; More O'Ferrall, R. A. J. Org. Chem. 1990, 55, 3251-3256.
- 25. Salzner, U.; Bachrach, S. M. J. Org. Chem. 1996, 61, 237-242.
- 26. Yamabe, S.; Dai, T.; Minato, T.; Machiguchi, T.; Hasegawa, T. J. Am. Chem. Soc. **1996**, *118*, 6518–6519.
- 27. Machiguchi, T.; Hasegawa, T.; Ishiwata, A.; Terashima, S.; Yamabe, S.; Minato, T. J. Am. Chem. Soc. **1999**, *123*, 4771–4786.
- 28. Staudinger, H. Chem. Ber. 1907, 40, 1145-1148.
- 29. Staudinger, H. Liebigs Ann. 1907, 356, 51-123.
- 30. Yamabe, S.; Kuwata, K.; Minato, T. Theor. Chem. Acc. 1999, 102, 139-146.
- 31. Machiguchi, T.; Okamoto, J.; Takachi, J.; Hasegawa, T.; Yamabe, S.; Minato, T. J. Am. Chem. Soc. 2003, 125, 14446–14448.
- 32. Corey, E. J.; Desai, M. C.; Engler, T. A. J. Am. Chem. Soc. 1985, 107, 4339-4341.
- 33. Holder, R. W. J. Chem. Ed. 1976, 63, 81-85.
- 34. Huisgen, R.; Mayr, H. Tetrahedron Lett. 1975, 2969-2972.
- 35. Huisgen, R.; Mayr, H. Tetrahedron Lett. 1975, 2965-2968.
- Weyler, W., Jr.; Byrd, L R.; Caserio, M. C.; Moore, H. W. J. Am. Chem. Soc. 1972, 94, 1027–1029.

- 37. Montaigne, R.; Ghosez, L. Angew. Chem., Int. Ed. 1968, 7, 221.
- 38. Huisgen, R.; Feiler, L. A.; Otto, P. Chem. Ber. 1969, 102, 3405-3427.
- 39. Huisgen, R.; Feiler, L. A.; Otto, P. Chem. Ber. 1969, 102, 3444-3459.
- 40. Effenberer, F.; Prossel, G.; Fischer, P. Chem. Ber. 1971, 104, 2002-2012.
- 41. Isaacs, N. S.; Stanbury, P. J. Chem. Soc. Perkin Trans. 2 1973, 166–169.
- 42. Baldwin, J. E.; Kapecki, J. A. J. Am. Chem. Soc. 1970, 92, 4868-4872.
- 43. Oyama, K.; Tidwell, T. T. J. Am. Chem. Soc. 1976, 98, 947-951.
- 44. Ali, S. A.; Muqtar, M.; Al-Husaini, A. H. J. Chem. Soc. Perkin Trans. 2, 1995, 1001–1006.
- 45. Gheorghiu, M. D.; Parvulescu, L.; Draghici, C.; Mihai, M.; Popescu, A. *Rev. Roum. Chim.* **1996**, *41*, 283–289.
- 46. Gheorghiu, M. D.; Mihai, M.; Parvulescu, L. Rev. Roum. Chim. 1995, 40, 653-660.
- 47. Brady, W. T.; O'Neal, H. R. J. Org. Chem. 1967, 32, 612-614.
- 48. Steiner, G.; Huisgen, R. J. Am. Chem. Soc. 1973, 95, 5056-5068.
- 49. Huisgen, R. Acc. Chem. Res. 1977, 10, 117-124.
- 50. Huisgen, R. Pure Appl. Chem. 1980, 52, 2283-2302.
- 51. Isaacs, N. S.; Hatcher, B. G. J. Chem. Soc. Chem. Commun. 1974, 593-594.
- 52. Katz, T. J.; Dessau, R. J. Am. Chem. Soc. 1963, 85, 2172-2773.
- 53. Snyder, E. I. J. Org. Chem. 1970, 35, 4287-4288.
- Collins, C. J.; Benjamin, B. M.; Kabalka, G. W. J. Am. Chem. Soc. 1978, 100, 2570– 2571.
- 55. Holder, R. W.; Graf, N. A.; Duesler, E.; Moss, J. C. J. Am. Chem. Soc. **1983**, 105, 2929–2931.
- 56. Swieton, G.; von Jouanne, J.; Kelm, H.; Huisgen, R. J. Chem. Soc, Perkin Trans. 2, **1983**, 37–43.
- 57. Lovas, F. J.; Suenram, K. D.; Gillies, C. W.; Gillies, J. Z.; Fowler, P. W.; Kisel, Z. J. Am. Chem. Soc. **1994**, *116*, 5285–5294.

5.4.1.4 Cycloaddition of Ketenes with Nucleophilic Alkenes The cycloaddition of ketenes with strongly nucleophilic alkenes such as vinyl ethers are predisposed to favor formation of zwitterionic intermediates, and there is evidence for this process in different examples. Nucleophilic attack on the ketene leads to a transition state with a high degree of enolate character, but a freely rotating zwitterionic intermediate is usually not formed, as the stereochemistry of the alkene is often preserved and the addition is also usually stereospecific at  $C_2$  of the ketene to give in many cases the thermodynamically least stable cyclobutanone derivative.

Examples of [2+2] cycloadditions with nucleophilic alkenes include the reaction of dichloroketene (1) prepared by zinc dehalogenation with the silyl vinyl ether 2 forming the cyclobutanone 3 (equation 1).¹ Dichloroketene prepared by dehydrochlorination reacted similarly with the eneamide 4, forming 5 (equation 2).² The chloroalkylketene 6 prepared by dehydrochlorination reacted with 4 by [2+2] cycloaddition to give 7, which was converted to 8 (equation 3).³ Related examples have been observed.^{4–5a}



The eneamine **9** reacted with two molecules of trimethylsilylketene **10** to form **11** in a reaction proposed to involve an intermediate zwitterion that reacted with **10** (equation 4).⁶ The reaction of ketenes **12** with **13** gave mainly the regioisomer **14** (equation 5), a result attributed to steric and not electronic control.⁷ Other electrophiles preferentially attack **13** at C–5. Cycloadditions of dichloroketene (**1**) from dehydrochlorination with the vinyl ether **15** gave **16**, which was converted by dechlorination and Beckmann rearrangement, forming pyrrolidinone **17** (equation 6).⁸





The [2+2] cycloaddition of Me₂C=C=O (**18**) with *E*-MeOCH=CHMe gave the cyclobutanone **19** with retention of the alkene stereochemistry (equation 7).⁹ Similar results were found for a variety of enol ethers ROCH=CHR¹ with Ph₂C=C=O.¹⁰⁻¹² Kinetic data for some of these reactions are given in Tables 5.7–5.9 in Section 5.4.1.3.

$$\overset{Me}{\underset{Me}{\longrightarrow}} C=O + (\overset{Me}{\underset{OMe}{\longrightarrow}} \overset{Me}{\underset{MeO}{\longrightarrow}} \overset{Me}{\underset{MeO}{\longrightarrow}} \overset{Me}{\underset{Me}{\longrightarrow}} (7)$$

The [2+2] cycloaddition of *t*-BuC(CN)=C=O (**20**) with CH₂=CHOEt and CH₂=CHOAc, however, did not proceed with 100% stereoselectivity, but gave *cis/trans* ratios in the product cyclobutanones **2** of 3:1 for EtO and CN and 3.5:1 for AcO and CN (equation 8).¹³ The products from reaction of **20** with isopropenyl acetate and benzoate **22** formed from zwitterionic intermediates, as hydride transfer resulting in acyclic products **24** occurred, or acyl transfer, and not ring closure (equation 9).¹³



The [2+2] cycloaddition of the silyl enol ether Z-Me₃SiOCH=CHMe **25** with **20** showed higher reactivity than for the *E*-alkene and a high degree of stereoselectivity forming **26** (equation 10).¹⁴ The product from *E*-**25** was also formed stereoselectively and was stable upon heating, while on heating the product **26** was converted to **27** (equation 10)



#### 490 REACTIONS OF KETENES

By contrast, the *E*-isomer of *t*-BuMe₂OCH=CHMe (**28**) was more reactive toward **20** than the *Z*-isomer, although the cyclobutanone products from each were still formed with high stereoselectivity. It was proposed¹⁴ that the reason for the low reactivity of *Z*-**28** was that a zwitterionic intermediate **29** was being formed reversibly, and that ring closure of **29** was rate determining, implying that the formation of **29** was reversible (equation 11).¹⁴ Presumably the bulk of the *t*-Bu group made ring closure to **30** slow.



Confirmatory evidence that the reactions proceeded by zwitterionic intermediates was obtained from thermolysis of the azide *trans*-**31**, which gave the product **34**.¹⁴ It was proposed that the reaction of **31** gave **32**, which underwent bond rotation to **33** followed by closure to **34** (equation 12).¹⁴ Reaction of *Z*-**31** in refluxing benzene gave **26**, **27**, and **34** in relative yields of 40, 5, and 50% respectively. The same products were formed from the reaction of **20** with **25** under the same conditions.



The reaction of chloroketenes with silyl enol ethers, 1,1-dialkoxyalkenes and 1,1-disilyloxyalkenes gave [2+2] cycloaddition in some cases, acyclic products in others, or mixtures of the two.^{15–19} The reaction of phenylchloroketene (**35**) with 4,5-dihydrofuran gave a 75% yield of the cycloaddition product **37**, along with 15% of the acyclic product **38**, suggesting the formation of the zwitterionic intermediate **36** (equation 13).²⁰



Rearrangement also occurred in the reaction of  $Ph_2C=C=O$  (**39**) and the silvl vinyl ether **40** forming **42** and the product of [2+2] addition to the C=O bond **43** (equations 14, 15).²⁰ The formation of **43** was found to be reversible to the reactants and on prolonged heating was converted to **42**, and the zwitterion **41** was proposed as an intermediate (equations 14, 15).²⁰



Cycloaddition of  $CH_2=CH_2$  with  $Ph_2C=C=O$  (**39**) and other ketenes with tetramethoxyethylene gave variable amounts of oxetane products from addition to the C=O bond along with cyclobutanones.^{21–23}

The reaction of Me₂C=C=O (18) with *N*-isobutenyl pyrrolidine (44) gave the [2+2] cycloadduct 46 and the product 47 resulting from reaction of two molecules of ketene (equation 16).²⁴ The kinetics were dissected into two concurrent processes, one involving competition of intermediate 45 for ring closure to 46 and reaction with a second molecule of 18 to give 47, and a separate direct formation of 46 (equation 17).²⁴ The formation of mixed products from ketene reactions with eneamines has also been found in other examples.^{25,26}



### **REFERENCES FOR SECTION 5.4.1.4**

- 1. Kim, S. K.; Pak, C. S. J. Org. Chem. 1991, 56, 6829-6832.
- de Faria, A. R.; Matos, C. R. R.; Correia, C. R. D. Tetrahedron Lett. 1993, 34, 27– 30.
- Correia, C. R. D.; de Faria, A R.; Carvalho, E. S. *Tetrahedron Lett.* 1995, 36, 5109– 5112.
- de Faria, A. R.; Salvador, E. L.; Correia, C. R. D. J. Org. Chem. 2002, 67, 3651– 3661.
- 5. Carpes, M. J. S.; Miranda, P. C. M. L.; Correia, C. R. D. *Tetrahedron Lett.* **1997**, *38*, 1869–1872.
- 5a. de Oca, A. C. B. M.; Corriea, C. R. D. Arkivoc 2003 (x), 390-403.
- 6. Takaoka, K.; Aoyama, T.; Shioiri, T. Synlett 1994, 1005-1006.
- 7. Cagnon, J. R.; Le Bideau, F.; Marchand-Beynaert, J.; Ghosez, L. *Tetrahedron Lett.* **1997**, *38*, 2291–2294.
- Kanazawa, A.; Gillet, S.; Delair, P.; Greene, A. E. J. Org. Chem. 1998, 63, 4660– 4663.
- 9. Isaacs, N. S.; Stanbury, P. J. Chem. Soc. Perkin Trans. 2, 1973, 166-169.
- 10. Huisgen, R.; Mayr, H. Tetrahedron Lett. 1975, 2969-2972.
- 11. Huisgen, R.; Mayr, H. Tetrahedron Lett. 1975, 2965-2968.
- 12. Huisgen, R.; Feiler, L. A.; Otto, P. Chem. Ber. 1969, 102, 3444-3459.
- 13. Becker, D.; Brodsky, N. C. J. Chem. Soc. Chem. Commun. 1978, 237-238.
- 14. Al-Husaini, A. H.; Moore, H. W. J. Org. Chem. 1985, 50, 2595-2597.
- 15. Brady, W. T.; Lloyd, R. M. J. Org. Chem. 1979, 44, 2560-2564.
- 16. Brady, W. T.; Lloyd, R. M. J. Org. Chem. 1980, 45, 2025-2028.
- 17. Brady, W. T.; Lloyd, R. M. J. Org. Chem. 1981, 46, 1322-1326.
- 18. Brady, W. T.; Watts, R. D. J. Org. Chem. 1981, 46, 4047-4050.
- 19. Naidorf-Meir, S.; Hassner, A. J. Org. Chem. 1992, 57, 5102-5015.
- 20. Raynolds, P. W.; DeLoach, J. A. J. Am. Chem. Soc. 1984, 106, 4566-4570.
- Hoffmann, R. W.; Bressel, U.; Gehlhaus, J.; Häuser, H. Chem. Ber. 1971, 104, 873– 885.
- 22. Aben, R. W.; Scheeren, H. W. Chem. Soc. Perkin Trans. 1 1979, 3132-3138.
- 23. Bellus, D. J. Am. Chem. Soc. 1978, 100, 8026-8028.
- 24. Huisgen, R.; Otto, P. J. Am. Chem. Soc. 1969, 91, 5922-5923.
- 25. Feiler, L. A.; Huisgen, R. Chem. Ber. 1969, 102, 3428-3443.
- 26. Hoch, H.; Hünig, S. Chem. Ber. 1972, 105, 2660-2685.

**5.4.1.5** Cycloaddition of Ketenes with Allenes Ketenes react readily with allenes by [2+2] cycloaddition pathways, and these have some interesting stereochemical outcomes. The reactions of *tert*-butylcyanoketene (1) have been extensively examined; this reacted with dimethylallenes 2 and 4 to form the isomeric 2-methylenecyclobutanones 3 and 5, respectively (equations 1, 2).¹⁻³

Tetramethylallene (6) reacted with 1 to give a single product 7 (equation 3).¹⁻³



Reaction of optically active **4** gave optically active **5a,b**, but **5c,d** were racemic.² To explain these results, a mechanism involving initial encounter complexes **8** and **11** were proposed, with little preference because of the minimal steric effects involved.^{1–3} Bond formation from **8** gave the zwitterion **9**, which formed optically active **5a**, or by bond rotation formed achiral intermediate **10**, which led to to **5a,b** (equation 4). Encounter complex **11** formed **12** and **13**, which gave rise to optically active **5b** and racemic **5c,d**, respectively (equation 5).^{1–3}



Optically active 1,2-cyclononadiene 14 gave [2+2] cycloaddition with *t*-BuC(CN=C=O (1) to form a 2:3 mixture of the *cis/trans* stereoisomeric adducts 16 (equation 6).⁴ Both of the products were optically active, although the optical purities were not established. The results were interpreted as involving a zwitterion 15 (equation 6).^{1,4}



Optically active 1,3-diphenylallene (17) gave [2+2] cycloaddition with *t*-BuC(CN)=C=O to form only cyclobutanones 18 and 19, with the *E* configuration of the phenylvinyl groups (*anti* to the carbonyl) analogous to 5a,b, and both were optically active (equation 7).^{5,6} These results were interpreted¹ in terms of the formation of two encounter complexes resembling 8 and 11 and similar formation of chiral zwitterions. However, the greater steric interactions with the phenyl groups prevented the formation of the *Z*-stereoisomers corresponding to 5c,d.



The reactions of  $Me_2C=C=O^{7,8}$  with a large number of allenes have been studied, as well as reactions of other ketenes with allenes.⁹ The allenes included symmetrically and unsymmetrically substituted examples, and the reactions often formed all of the possible stereoisomeric products. The reaction pathways appear to be consistent with the processes shown above (equations 4, 5). The reaction rates and product distributions for various ketene/allene pairs were interpreted as involving formation of uncharged species by bonding between the sp carbons of the ketene and the allene.¹⁰

### **REFERENCES FOR SECTION 5.4.1.5**

- 1. Moore, H. W.; Gheorghiu, M. D. Chem. Soc. Rev. 1981, 10, 289-328.
- 2. Duncan, W. G.; Weyler, W., Jr.; Moore, H. W. Tetrahedron Lett. 1973, 4391-4394.
- 3. Bampfield, H. A.; Brook, P. R. J. Chem. Soc. Chem. Commun. 1974, 171-172.
- 4. Weyler, W., Jr.; Byrd, L R.; Caserio, M. C.; Moore, H. W. J. Am. Chem. Soc. 1972, 94, 1027–1029.

- 5. Bampfield, H. A.; Brook, P. R.; McDonald, W. S. J. Chem. Soc. Chem.Commun. 1975, 132–133.
- 6. Bampfield, H. A.; Brook, P. R.; Hunt, K. J. Chem. Soc. Chem. Commun. 1976, 146-147.
- 7. Bertrand, M.; Gras, J.-L.; Gore, J. Tetrahedron 1975, 31, 857-862.
- 8. Bertand, M.; Maurin, R.; Gras, J. L.; Gil, G. Tetrahedron 1975, 31, 840-855.
- 9. Brady, W. T.; Stockton, J. D.; Patel, A. D. J. Org. Chem. 1974, 39, 236-238.
- 10. Gras, J.-L.; Bertrand, M. Nouv. J. Chim. 1981, 5, 521-530.

**5.4.1.6** Cycloaddition of Ketenes with Alkynes Reactive ketenes substituted with electronegative groups gave [2+2] cycloaddition with nonactivated alkynes to form cyclobutenones.^{1,2} This reaction provides a major route for the preparation of cyclobutenones, which form vinylketenes upon ring opening (Section 3.4.1). For example, dichloroketene (1) generated by zinc dehalogenation reacted with 2-butyne, 3-hexyne, or 1-hexynes 2 to form adducts 3, which underwent isomerization in situ induced by the ZnCl₂ present to an equilibrium mixture with 4 (equation 1).^{2,3} Hydrolysis of 3 with acid gave cyclobutenediones 5 (equation 2),^{2,3} which may be used to prepare bisketenes (Section 4.9). Cyclobutenones 3 and 4 were reduced with zinc and acetic acid to cyclobutenones 6 (equation 2).^{2,3} *tert*-Butylcyanoketene reacted with a variety of alkyl- and arylalkynes to give cyclobutenones,⁴ and dimethylketene (7) gave [2+2] cycloaddition to alkynyl ether 8 forming cyclobutenone 9 (equation 3).⁵



A computational study of [2+2] cycloaddition reactions of ketene (10) with alkynes 11 examined the formation of cyclobutenones 12 and oxetes 13 and the dependence of the reactions on the alkynyl substituents R (equation 4).⁶ The cyclobutenones 12 can undergo ring opening to vinylketenes 14, which can rearrange to acylallenes 15, which can also be formed by ring opening of the oxetes



The reaction of trimethylsilylacetylene (16) with dichloroketene (1) gave mainly the product 17 resulting from electrophilic attack at the nonsilicon-substituted carbon of the alkyne, contrary to the usual direction of electrophilic substitution on this alkyne (equation 6).⁷ However, this carbon does have the highest coefficient of the HOMO of SiH₃C≡CH.⁷ With PhC≡CSiMe₃ (18) electrophilic attack occurs at the silylated carbon, in accord with the expected stability of an intermediate with positive charge adjacent to phenyl (equation 6).⁷



The reactive alkyne 1-triisopropylsilyloxyheptyne (20) reacted with ketene (10) to give 3-silyloxycyclobutenone 21 (equation 7).⁸ 1-Ethoxyheptyne was less reactive toward 10 than was 20. Reaction of 21 with MeLi followed by acid gave cyclobutenone 22 (equation 7).⁸



The reaction of *N*,*N*-bis(trimethylsilyl)ynamine **24** with ketene **23** led to the formation of cyclobutenone **25** (equation 8).⁹ However, reaction of the *n*-butyl analogue **26** with **23** gave 24% of **27**, together with the allenic amide **29** (35%) resulting from ring opening and rearrangement followed by hydrolysis of the initial product **28** (equation 9).⁹




Acylketene **30** (R = Cl) generated by dehydrochlorination reacted with 1-methoxybutyne, leading to azulenone **33** (equations 10, 11).¹⁰ With **30** (R = EtO) the [4+2] cycloadduct **37** was the major product (equation 12).¹⁰ On the basis of the observed solvent and substituent dependence of the product distribution, the zwitterionic intermediate **31** from **30** (R = Cl) was proposed to undergo cyclization on the phenyl ring leading to **33** (equations 10, 11), whereas with **30** (R = EtO) the zwitterion **34** cyclized to the cyclobutenone **35**, followed by ring opening to **36** leading to the pyrone **37** (equation 12).¹⁰ The reaction of **30** with other alkynyl ethers was also studied.¹⁰



Diarylketenes **38** and ethoxyacetylene also reacted to form azulenones in a process formulated as involving cycloaddition of the zwitterionic intermediate **39**,

which gave **40**, leading to **41** and the azulene derivatives **42** (equations 13, 14), as well as cyclobutenones from [2+2] cycloadditions.^{11,12} There was a preference for attack on the ring bearing an electron-donating substituent (MeO, Me) but not an electron-withdrawing group (Cl, Br, CO₂Et). The intermediate norcaradienes corresponding to **32** were isolated in the reaction of 1-naphthylphenylketene, di-(2-naphthyl)ketene, and di-(1-naphthyl)ketene with EtOC=CH.¹² The vinylketenes **43** and **44** gave only [2+2] cycloaddition products with HC=COEt.¹²



The [2+2] cycloaddition of hydroxyketene (**45**) with dihydroxyacetylene (**46**) was studied computationally using MNDO, and the reaction was predicted to proceed with formation of a zwitterionic intermediate **47** leading to the cyclobutenone **48** (equation 15).¹³ The structure of the complex of acetylene and ketene in the gas phase has been examined by microwave spectroscopy and found to be planar, with a 25° tilting of the axes of the molecules (Figure 5.3).¹⁴





**Figure 5.3** Ketene-acetylene complex as determined by microwave spectroscopy (reprinted form ref. 12 with permission of the American Chemical Society)

Carboethoxy and carboxamido ketenes (**49**, **52**, and **55**) reacted with alkynylamines to give cyclobutenones by [2+2] cycloaddition (equation 16), pyrones by [4+2] cycloaddition (equation 17), and allenes (equation 18), depending upon the particular substrates.^{15,16} Vinylketenes also gave allenes and other products on cycloaddition with alkynylamines, together with cyclobutenones.^{17–20}



#### **REFERENCES FOR SECTION 5.4.1.6**

- 1. Knoche, H. Liebigs Ann. Chem. 1969, 722, 232-233.
- 2. Danheiser, R. L.; Savariar, S. Tetrahedron Lett. 1987, 28, 3299-3302.
- 3. Ammann, A. A.; Rey, M.; Dreiding, A. S. Helv. Chim. Acta 1987, 70, 321-328.
- 4. Gheorghiu, M. D.; Draghici, C.; Stanescu, L.; Avram, M. Tetrahedron Lett. 1973, 9-12.
- 5. Haseck, R. M.; Gott, P. G.; Martin, J. G. J. Org. Chem. 1964, 29, 2510-2513.
- 6. Koch, R.; Wentrup, C. Org. Biomol. Chem. 2004, 2, 195-199.
- 7. Danheidser, R. L.; Sard, H. Tetrahedron Lett. 1983, 24, 23-26.
- 8. Kowalski, C. J.; Lal, G. S. J. Am. Chem. Soc. 1988, 110, 3693-3695.
- 9. Schulte, N.; Möller, M. H.; Rodewald, U.; Würthwein, E.-U. Chem. Ber. 1994, 127, 1287–1293.

- 10. Brown, D. G.; Hoye, T. R.; Brisbois, R. G. J. Org. Chem. 1998, 63, 1630-1636.
- 11. Teufel, H.; Jenny, E. F. Tetrahedron Lett. 1971, 1769-1772.
- 12. Wuest, J. D. Tetrahedron 1980, 36, 2291-2296.
- 13. Pericas, M. A.; Serratosa, F.; Valenti, E. J. Chem. Soc. Perkin Trans. 2 1987, 151-158.
- 14. Gillies, C. W.; Gillies, J. Z.; Lovas, F. J.; Suenram, R. D. J. Am. Chem. Soc. 1993, 15, 953-962.
- 15. Delaunois, M.; Ghosez, L. Angew. Chem. Int. Ed. Engl. 1969, 8, 72-73.
- 16. Ficini, J.; Pouliquen, J. Tetahedron Lett. 1972, 1135-1138.
- 17. Dötz, K. H.; Trenkle, B.; Schubert, U. Angew. Chem. Int. Ed. Engl. 1981, 20, 287.
- 18. Dötz, K. H.; Mühlemeier, J.; Trenkle, B. J. Organomet. Chem. 1985, 289, 257-262.
- 19. Ficini, J.; Falou, S.; d'Angelo, J. Tetrahedron Lett. 1977, 1931-1934.

5.4.1.7 Cycloaddition of Ketenes with Imines The reaction of diphenylketene (1) and benzanilide forming the  $\beta$ -lactam 2 was discovered by Staudinger in 1907 (equation 1).¹ This [2+2] cycloaddition of ketenes with imines (Staudinger reaction) provides a facile and versatile preparation of  $\beta$ -lactams, and because of the importance of these derivatives it has been of continuing and increasing interest and has been frequently reviewed.²⁻¹¹



The [2+2] cycloaddition reaction is usually carried out with in situ generation of the ketene from the acyl chloride in the presence of the imine. With no base such as triethylamine present, the reaction could take place without the formation of a ketene by attack of the imine on the acyl chloride.

Calculations at the B3LYP/6-31G* level indicated that when the ketene was preformed by adding the acyl chloride slowly to a tertiary amine/imine mixture, stepwise formation of zwitterion **3** followed by conrotatory ring closure gave the *cis* product **4**, in agreement with experimental findings (equation 2).¹² When the tertiary amine was added to the acyl chloride/imine mixture, an intermediate formed that led through an  $S_N2$  reaction giving *trans* product **5**, also in agreement with experimental results (equation 3).¹³



The mechanism of the reaction has been the subject of increasingly refined computational studies.^{13–15} A mechanistic scheme derived using a (CASPT2) and (CASSCF) approach identified two directions of approach of the imine to the ketene in the reaction (equation 4).¹³ This was proposed as a starting point for more detailed calculations, including substituent and solvent effects to predict the product stereochemistry.¹³ The thermodynamic stability of the cycloaddition of vinylimine (CH₂=C=NH) to ketene forming 3- and 4-methylene- $\beta$ -lactams was also examined by computational means.¹⁴

The enantioselective synthesis of  $\beta$ -lactams is an area of great interest,¹⁵ as also discussed in Section 5.9, and has been applied to the synthesis of  $\alpha$ - and  $\beta$ -amino acids.¹⁶ Thermally induced isomerization of the initial *cis*- $\beta$ -lactams **4** to form the *trans* isomers has been found, and has provided evidence for reversible formation of zwitterions **3** that form the thermodynamically more stable products.¹⁷ Large polycyclic aryl *N*-substituents on the imines also favored *trans* products.¹⁸

Benzyloxyketene (7) reacted with ketenimine **8**, forming the *cis*-lactam **9** as a single isomer (equation 5), and this was further converted to  $\beta$ -amino acids.¹⁹



Cycloaddition of ketenes 7, 11, and 12 with imines 10 gave  $\beta$ -lactams 13 that upon palladium-catalyzed cycloaddition gave polycyclic  $\beta$ -lactams 14 (equation 6).²⁰ Phenoxyketene (16) reacted with the chirally substituted imine 15 to give a 17:83 mixture of  $\beta$ -lactams 17 (equation 7).²¹





Solvent effects on the stereoselectivity of ketene-imine cycloaddition reactions have been studied by *ab initio* molecular orbital calculations,²² and this subject has been reviewed.²³ The reaction of fluoroketene with H₂NCH=NH was studied by computation.²⁴ Calculations have been interpreted as favoring a concerted pathway for the ketene-imine cycloaddition, but other interpretations favor stepwise processes, as noted above.²⁵ Ketene cyclizations with carbohydrate derived imines have been studied by (AM1) calculations.²⁶

Stereoselectivity in imine [2+2] cycloadditions with ketenes has been examined by the use of chiral substituents on both the ketene and the imine.^{27,28} The imine  $(Me_3Si)_2CHN=CH_2$  is a stable and isolable compound that reacted with chiral ketenes to provide 4-unsubstituted  $\beta$ -lactams in a single step.²⁹ The [2+2] cycloaddition of chiral ketene **19** formed by dehydrochlorination with the chiral imine **20** gave **21** (equation 8) in a two-step reaction with an observable intermediate (cf. **23**, equation 9), and a variety of related examples were observed.³⁰



Phthimidolketene (11) reacted with *N*-trimethylsilylimines 22 to give observable intermediates 23, which at reflux cyclized to *trans*- $\beta$ -lactams 24 that were suggested to form with silyl migration (equation 9).³¹ The silyl group was subsequently removed by hydrolysis. These reactions have been studied computationally.³² These two-step cycloadditions were also found to proceed with microwave assistance under solvent-free conditions,³³ and the stereochemistry changed when the ketene substituent was varied from benzyloxy to benzoyloxy.³⁴ Diastereoselective synthesis of  $\beta$ -lactams was also carried out with an *N*-( $\alpha$ -thiophenyl)benzyl (PhSCHAr) protecting group removable by oxidation.³⁵



Oxygen-substituted ketenes 7, 16, and 26 generated in situ by dehydrochlorination of 25 reacted with haloalkyl-substituted imines 27 forming azetidinones 28, which gave ring-opened products upon treatment with NaOMe (equation 10).³⁶

$$RO \xrightarrow{O}_{Cl} \xrightarrow{E_{13}N} \xrightarrow{RO}_{C=O} \xrightarrow{R^2 \xrightarrow{N}_{R^1}} \xrightarrow{RO}_{R^2 \xrightarrow{N}_{R^2}} \xrightarrow{RO}_{R^2 \xrightarrow$$

Benzyloxyketene (7) generated by dehydrochlorination reacted with 22 with a facile cyclization step forming the *cis* product 30, and the stereochemistry was proposed to arise from a two-step reaction with an electrostatic interaction between the oxygen and the imine carbon stabilizing the *E-anti*-azadiene 29 (equation 11).³⁷



The effects of the solvent and of the substituents on ketenes **31** and vinylimines **32** undergoing cycloaddition have been studied computationally and experimentally for selectivity in forming the [2+2] product **33** and the [4+2] product **34** (equation 12).³⁸ Reactions of trisubstituted imines with disubstituted ketenes gave  $\beta$ -lactams with quaternary carbons.³⁹ The reactivity of ketenes **31** to give both [2+2] and [4+2] cycloadditions with 1,3-diaza-1,3-butadienes **35**, forming **36** and **37** has been studied computationally (equation 13).⁴⁰



*N*-Imino- $\beta$ -lactams (40) were formed from the reaction of 2,3-diaza-1,3-dienes **39** with ketenes **38** from a variety of precursors (equation 14) and are useful in further synthetic transformations.⁴¹ The reaction of alkynylimines with ketenes **38**, forming  $\beta$ -lactams **41** followed by cyclization, led to unusual "inversely fused" bicyclic lactams **42** (equation 15).^{42,43} The ketene (EtS)₂C=C=O (**43**) was used to prepare **42a** (equation 15a).⁴³



Ketene **44** generated by Wolff rearrangement reacted with imines to give  $\beta$ -lactams **45**,^{44,45} and the use of an oxidatively removable chiral auxiliary on the imine **15** gave significant selectivity (equation 16).⁴⁶ Reactions of ketenes with peptidederived imines gave peptidic lactams.⁴⁷ Oxygen-substituted ketenes **7**, **16**, **25**, and **30** generated by dehydrochlorination reacted with diimines **46** to form bis- $\beta$ -lactams **47** (equation 17).⁴⁸ Carbomethoxyketene (**48**) gave [4+2] cycloaddition with the cationic triazapentadienium iodide **49** forming **50** (equation 18).⁴⁹





The diastereoselectivities in other ketene-imine cycloadditions have been studied,  ${}^{50-52}$  including reaction of dienylketene  $51^{51-53}$  with 52 to form 53 (equation 19),  52  and in [4+2] cycloadditions of  $51^{52,53}$  and other ketenes with 1,3-diazabutadienes forming 5-dienyl pyrimidones.  ${}^{53-55}$  Dehydration of the furanose-substituted carboxylic acid 54 by elimination from the mixed trifluoroacetic anhydride was proposed to generate the ketene 55, which reacted with 55a by a [4+2] cycloaddition to form the unobserved intermediate 55b (equation 20), which was converted to *C*-nucleosides 56 and 56a (equations 20, 20a).  56  The product 56a was suggested to arise from hydroysis and decarboxylation of 55b.  56 



Polymer-supported imine **58** gave high diastereoselectivities to form  $\beta$ -lactams **59** on reaction with ketenes **11**, **16**, **19**, and **57** generated by dehydrochlorination in solution, with removal of the  $\beta$ -lactams from the resin with 3% CF₃CO₂H (equation 21).⁵⁷ Polymer-bound aryl imines (polymer)NHCOC₆H₄CH=NR reacted similarly with AcOCH=C=O (**60**).⁵⁸ Imines bound to soluble polyethylene glycols gave  $\beta$ -lactams upon reaction with PhOCH=C=O (**16**).⁵⁹ Rink-resin-bound imines reacted

with ketenes 7, 16, and 60 to give  $\beta$ -lactams in 60–68% yields.⁶⁰



Benzyloxy ketene 7 underwent [2+2] cycloaddition with the (fluoroalkyl)acetaldimines  $R_fCH=N(C_6H_4OMe-4)$  where  $R_f=CF_3$ ,  $CF_2H$ , and  $CF_2Cl$  in yields 0f 55–72%.⁶¹ Diacylimines (RCO)(R¹CO)C=NPh reacted with a variety of ketenes to give  $\beta$ -lactams.⁶²

5-Phenylvaleryl chloride was heated with *n*-Bu₃N to generate ketene **61**, which reacted with phenanthridine (**62**) to give **63** as a mixture of *cis* and *trans* isomers in only 7% yield (equation 22),⁶³ but reaction with **64** gave the *trans*  $\beta$ -lactam in 47% yield.⁶³ Ketenes with electron-withdrawing substituents gave much better yields of  $\beta$ -lactams on reaction with **62**.⁶⁴ Chloroketene (**66**) reacted with 1,5-benzothiaze-pine **65** to give  $\beta$ -lactams **67** at room temperature but ring-opened products **68** at reflux (equation 23).⁶⁵ The structures of some ketene adducts with pyridine-*N*-oxides and dihydropyridines were determined by X-ray crystallography.⁶⁶



Ketenes generated by dehydrochlorination gave [2+2] cyclization with 1,4-benzodiazepines **71** and formed novel  $\beta$ -lactams **72** in yields of up to 95% (equation 24).⁶⁷



Reaction of the ketene **73** generated in situ with the imine equivalent **74** gave 85% selectivity for the desired  $\beta$ -lactam **75**, with a 60% yield of the two stereoisomers (equation 25).⁶⁸ The triazine **74** and BF₃ acted as an imine equivalent.



The diphenyloxazolidinyl-substituted ketene **70** reacted with imines **76** to give  $\beta$ -lactams **77** in 60–71% yield and 97% *de* (equation 26).^{69,70} Further reactions converted **77** to carbacephen intermediates. Phthalimidoketene (**11**) was also generated using Mukaiyama's salt (**78**, Section 3.2.1) for conversion of **76** to the cycloadduct **77** (R₂N = PhthN) in 89% yield.⁷⁰



Diphenylketene (1) and dimethylketene (79) reacted with the divinylimines 81 to form [2+2] adducts 82 that underwent [1,3] signatropic rearrangements forming the [4+2] adducts 83 on heating (equation 27).⁷¹ Dichloroketene (80) reacted with 81, giving the [4+2] adduct 83 directly.⁷¹



Reaction of the diacid **84** with Mukaiyama's reagent **78** generated the formal bisketene **85** (equation 28) that reacted with the imine **86** and gave the dilactam

**87** (equation 29).⁷² Reaction of **85** with a bis(amine) **88** formed a polymer with  $\beta$ -lactams in the amine chain, as shown in Section 4.9.⁷²



Diphenylketene (1) did not react with TsN=CHCO₂Et (90) at room temperature, but 5% Cp₂Co⁺Co(CO)₄⁻ catalyzed the rapid formation of 91 in a reaction proposed to involve the intermediate 89 (equation 30).⁷³



Chloroketene (66) generated by dehydrochlorination reacted with the imine 92, with formation of the spiro- $\beta$ -lactam 93 (equation 31).⁷⁴ This reaction has been employed in the total synthesis of spirocyclic  $\beta$ -lactam alkaloids isolated from marine sources.⁷⁵ Tetrahydrofuran-derived spiro- $\beta$ -lactams 95 were formed from ketene 94 (equation 32).⁷⁶ Isomeric ketene 96 behaved in the same manner.⁷⁶ Imines bearing chiral substituents gave high diastereoselectivities up to 95:5.⁷⁷ The reaction of chiral proline–derived ketenes 97 with imine 98 gave *cis*-stereoselectivity forming 99 (equation 33).⁷⁸ Computational modeling of the reaction using DFT methods was also carried out with ketene 100 reacting with the imine 101 forming 102 (equation 34), and also for the reaction of 100 with the chiral imine 103.⁷⁶





Acylketenes **105** generated from Meldrum's acid derivatives **104** reacted with  $\beta$ -thiazolines (**106**) to form 2-pyridinones **107** (equation 35).⁷⁹



N,N-Dialkylhydrazones **108** acted as the imine component in [2+2] cycloadditions with benzyloxyketene (7) forming *cis*-azetidinones **109** (equation 36).⁸⁰ However, the chiral *N*,*N*-dialkylhydrazones **111** reacted with the ketene **110** generated from the acid with Mukaiyama's salt (78) and *i*-Pr₂NEt to give the *trans*-azetidin-2-ones **112** as single diastereomers (equation 37).⁸¹ The formation of *trans* products was in contrast to the *cis*-selectivity observed with benzyloxyketene, and was interpreted as involving formation of a zwitterionic intermediate that, because of a steric barrier to conrotatory closure to the *cis*-adduct, instead underwent a C=N bond isomerization before conrotatory closing to the *trans* product (equation 38).





The transition metal-substituted imine *N*-rhenaimine **113** gave a remarkably fast [2+2] cycloaddition with diphenylketene (1) forming **114** even at -78 °C (equation 39).⁸² The structure of the product was confirmed by X-ray and examined by DFT computations.



Other studies of ketene-imine cycloadditions forming  $\beta$ -lactams include the use of imines derived from *D*-glucose,⁸³ and imines substituted with chiral chromium tricarbonyl–complexed arenes.⁸⁴

# **REFERENCES FOR SECTION 5.4.1.7**

- 1. Staudinger, H. Liebigs Ann. Chem. 1907, 356, 51-123.
- 2. Sheehan, J. C.; Corey, E. J. Org. React. 1957, 9, 388-408.
- Ulrich, H. Cycloaddition Reactions of Heterocumulenes; Academic Press, New York, 1967, pp. 38–109.
- Muller, L. L.; Hamer, J. 1,2-Cycloaddition Reactions. The Formation of Three- and Four-Membered Heterocycles; Wiley Interscience, New York, 1967.
- 5. van der Steen, F. H.; van Kouten, G. Tetrahedron 1991, 47, 7503-7524.
- Georg, G. I.; Ravikumar, V. T. In *The Organic Chemistry of* β-*Lactams*; Georg, G. I., Ed.; VCH, New York, 1992, pp. 295–368.
- Palomo, C.; Aizpurua, J. M.; Ganboa, I.; Oiarbide, M. Eur. J. Org. Chem. 1999, 3223– 3235.
- 8. Palomo, C.; Aizpurua, J. M.; Ganboa, I.; Oiarbide, M. Curr. Med. Chem. 2004, 11, 1837–1872.
- 9. Koppel, G. A. In *Small Ring Heterocycles*; Hassner, A., Ed.; Wiley, New York, 1983, Vol. 42, p. 219.
- Backes, J. In *Houben-Weyl, Methoden der Organischen Chemie*, Band E16B; Muller, E.; Bayer, O., Eds.; Thieme, Stuttgart, 1991, pp. 31–942.
- DeKimpe, N. In *Comprehensive Heterocyclic Chemistry II*, Katritzky, A. R.; Rees, C. W.; Scriven, E. F. V.; Padwa, A., Eds.; Pergamon, Oxford, 1996, Vol. 1B, pp. 507–568.
- 12. Arrieta, A.; Lecea, B.; Cossio, F. P. J. Org. Chem. 1998, 63, 5869-5876.
- 13. Venturini, A.; González, J. J. Org. Chem. 2002, 67, 9089-9092.
- Rode, J. E.; Dobrowolski, J. C.; Borowiak, M. A.; Mazurek, A. P. Phys. Chem. Chem. Phys. 2002, 4, 3948–3858.
- 15. Magriotis, P. A. Angew. Chem. Int. Ed. 2001, 40, 4377-4379.

- 16. Palomo, C.; Aizpurua, J. M.; Ganboa, I. Oiarbide, M. Synlett 2001, 1813-1826.
- 17. Alcaide, B.; Almendros, P.; Salgado, N. R.; Rodríquez-Vicente, A. J. Org. Chem. 2000, 65, 4453–4455.
- 18. Banik, B. K.; Becker, F. F. Tetrahedron Lett. 2000, 41, 6551-6554.
- Palomo, C.; Aizpurua, J. M.; Ganboa, I.; Odriozola, B.; Urchegui, R.; Görls, H. *Chem. Commun.* **1996**, 1269–1270.
- Burwood, M.; Davies, B.; Diaz, I.; Grigg, R.; Molina, P.; Sridharan, V.; Hughes, M. *Tetrahedron Lett.* **1995**, *36*, 9053–9056.
- 21. Hashimoto, Y.; Kai, A.; Saigo, K. Tetrahedron Lett. 1995, 36, 8821-8824.
- 22. Lopez, R.; Suárez, D.; Ruiz-López, M. F.; González, J.; Sordo, J. A.; Sordo, T. L. J. Chem. Soc. Chem. Commun. 1995, 1677–1678.
- 23. Fang, D.-C.; Fu, X.-Y. Int. J. Quant. Chem. 1996, 57, 1107-1114.
- 24. Troung, T. N. Int. Rev. Phys. Chem. 1998, 17, 526-546.
- 25. Fang, D.-C.; Fu, X.-Y. Chin. J. Chem. 1996, 14, 97-104.
- Barcza, M. V.; de M. Carneiro, J. W.; Serra, A. A.; Barboza, J. C. S. J. Mol. Struct. 1997, 394, 281–289.
- Palomo, C.; Aizpurua, J. M.; Legido, M.; Galarza, R.; Deya, P. M.; Dunogués, J.; Picard, J. P.; Ricci, A.; Seconi, G. Angew. Chem., Int. Ed. Engl. 1996, 35, 1239–1241.
- 28. Palomo, C.; Aizpurua, J. M.; Mielgo, A.; Linden, A. J. Org. Chem. 1996, 61, 9186–9195.
- 29. Palomo, C.; Aizpurua, J. M.; Legido, M.; Galarza, R. Chem. Commun. 1997, 233-234.
- Bongini, A.; Panunzio, M.; Piersanti, G.; Bandini, E.; Martelli, G.; Spunta, G.; Venturini, A. Eur. J. Org. Chem. 2000, 2379–2390.
- Bandini, E.; Martelli, G.; Spunta, G.; Bongini, A.; Panunzio, M. *Tetrahedron Lett.* 1996, 37, 4409–4412.
- 32. Arrieta, A.; Cossío, F. P.; Lecea, B. J. Org. Chem. 2000, 65, 8458-8464.
- 33. Martelli, G.; Spunta, G.; Panunzio, M. Tetrahedron Lett. 1998, 39, 6257-6260.
- Panunzio, M.; Bacchi, S.; Campana, E.; Fiume, L.; Vicennati, P. Tetrahedron Lett. 1999, 40, 8495–8498.
- 35. Karupaiyan, K.; Srirajan, V.; Deshmukh, A. R. A. S.; Bhawal, B. M. *Tetrahedron* **1998**, *54*, 4375–4386.
- 36. Dejaegher, Y.; De Kimpe, N. J. Org. Chem. 2004, 69, 5974-5985.
- 37. Bacchi, S.; Bongini, A.; Panunzio, M.; Villa, M. Synlett 1998, 843-844.
- 38. Lecea, B.; Arrastia, I.; Arrieta, A.; Roa, G.; Lopez, X.; Arriortua, M. I.; Ugalde, J. M.; Cossío, F. P. J. Org. Chem. 1996, 61, 3070–3079.
- Palomo, C.; Aizpurua, J. M.; García, J. M.; Galarza, R.; Legido, M.; Urchegui, R.; Roman, P.; Luque, A.; Server-Carrió, J.; Linden, A. J. Org. Chem. 1997, 62, 2070– 2079.
- 40. Bharatam, P. V.; Kumar, R. S.; Mahajan, M. P. Org. Lett. 2000, 2, 2725-2728.
- 41. Alcaide, B.; Miranda, M.; Perez-Castells, J.; Polanco, C.; Sierra, M. A. J. Org. Chem. **1994**, *59*, 8003–8010.
- 42. Ren, X.-F.; Konaklieva, M. I.; Turos, E. J. Org. Chem. 1995, 60, 4980-4981.
- 43. Ren, X.-F.; Konaklieva, M. I.; Shi, H.; Dickey, S.; Lim, D. V.; Gonzalez, J.; Turos, E. J. Org. Chem. **1998**, 63, 8898–8917.
- 44. Podlech, J. Synlett 1996, 582-584.

#### 512 REACTIONS OF KETENES

- 45. Podlech, J.; Linder, M. R. J. Org. Chem. 1997, 62, 5873-5883.
- 46. Podlech, J.; Steurer, S. Synthesis 1999, 650-654.
- 47. Linder, M. R.; Podlech, J. Org. Lett. 1999, 1, 869-871.
- 48. Alcaide, B.; Martín-Cantalejo, Y.; Perez-Castells, J.; Sierra, M. A.; Monge, A. J. Org. Chem. 1996, 61, 9156–9163.
- Landreau, C.; Deniaud, D.; Reliquet, A.; Reliquet, F.; Meslin, J. C. J. Heterocyclic Chem. 2001, 38, 93–98.
- 50. Sidhoum, D. A.; Benhaoua, H. Synth. Commun. 1997, 27, 1519-1525.
- 51. Sharma, A. K.; Jayakumar, S.; Mahajan, M. P. Tetrahedron Lett. 1998, 39, 7205–7208.
- 52. Sharma, A. K.; Kumar, R. S.; Mahajan, M. P. Heterocycles 2000, 52, 603-619.
- 53. Mukherjee, S.; Mazumdar, S. N.; Sharma, A. K.; Mahajan, M. P. *Hetreocycles* **1998**, *47*, 933–950.
- 54. Rossi, E.; Abbiati, G.; Pini, E. Tetrahedron 1997, 53, 14107-14114.
- 55. Rossi, E.; Abbiati, G.; Pini, E. Tetahedron 1999, 55, 6961-6970.
- 56. Katagiri, N.; Haneda, T.; Niwa, R.; Kato, T. Chem. Pharm. Bull. 1985, 33, 2671-2677.
- Ruhland, B.; Bhandari, A.; Gordon, E. M.; Gallop, M. A. J. Am. Chem. Soc. 1996, 118, 253–254.
- 58. Singh, R.; Nuss, J. M. Tetrahedron Lett. 1999, 40, 1249-1252.
- 59. Benaglia, M.; Cinquini, M.; Cozzi, F. Tetrahedron Lett. 1999, 40, 2019-2020.
- 60. Dasgupta, S. K.; Banik, B. K. Tetrahedron Lett. 2002, 43, 9445-9447.
- Abouabdellah, A.; Bégue, J.-P.; Bonnet-Delpon, D.; Nga, T. T. T. J. Org. Chem. 1997, 62, 8826–8833.
- 62. Bogdanowicz-Szwed, K.; Krasodomska, M. Monatsh. Chem. 1998, 129, 81-87.
- 63. Afonso, A.; Rosenblum, S. B.; Puar, M. S.; McPhail, A. T. *Tetrahedron Lett.* **1998**, *39*, 7431–7434.
- 64. Alcaide, B.; Rodríquez-Vicente, A. Tetrahedron Lett. 1999, 40, 2005-2006.
- Xing, Q.; Wang, H.; Zhou, X.; Jin, S.; Li, Y.; Chan, A. S. C. J. Heterocyclic Chem. 2001, 38, 561–567.
- 66. Oiso, S.; Eto, M.; Yoshitake, Y.; Harano, K. Chem. Pharm. Bull. 2003, 51, 1068–1074.
- del Pozo, C.; Marcías, A.; López-Ortiz, F.; Maestro, M. Á.; Alonso, E.; Gonzáles, J., Eur. J. Org. Chem. 2004, 535–545.
- 68. Cainelli, G.; Galletti, P.; Giacomini, D. Tetrahedron Lett. 1998, 39, 7779-7782.
- 69. Palomo, C.; Ganboa, I.; Kot, A.; Dembkowski, L. J. Org. Chem. 1998, 63, 6398-6400.
- 70. Matsui, S.; Hashimoto, Y.; Saigo, K. Synthesis 1998, 1161-1166.
- 71. Saito, T.; Kobayashi, S.; Ohgaki, M.; Wada, M.; Nagahiro, C. *Tetrahedron Lett.* **2002**, *43*, 2627–2631.
- 72. Suto, A.; Endo, T. Macromolecules 1998, 31, 7996-7998.
- 73. Wack, H.; Drury, W. J., III; Taggi, A. E.; Ferraris, D.; Lectka, T. Org. Lett. **1999**, *1*, 1985–1988.
- Barba, V.; Hernández, C.; Rojas-Lima, S.; Farfán, N.; Santillan, R. Can. J. Chem. 1999, 77, 2025–2032.
- 75. Lin, X.; Weinreb, S. M. Tetrahedron Lett. 2001, 42, 2631-2633.
- 76. Alonso, E.; del Pozo, C.; González, J. J. Chem. Soc. Perkin Trans. 1 2002, 571-576.

- Marcías, A.; Alonso, E.; del Pozo, C.; Venturini, A.; González, J. J. Org. Chem. 2004, 69, 7004–7012.
- Khasnov, A. B.; Ramirez-Weinhouse, M. M.; Webb, T. R.; Thiruvazhi, M. J. Org. Chem. 2004, 69, 5766–5769.
- 79. Emtenäs, H.; Alderin, L.; Almqvist, F. J. Org. Chem. 2001, 66, 6756-6761.
- 80. Fernández, R.; Ferrete, A.; Lassaletta, J. M.; Llera, J. M.; Martín-Zamora, E. Angew. Chem. Int. Ed. 2002, 41, 831–833.
- 81. Díez, E.; Fernândez, R.; Marquéz-López, E.; Martín-Zamora, E.; Lassaletta, J. M. Org. Lett. 2004, 6, 2749–2752.
- Hevia, E.; Perez, J.; Riera, V.; Miguel, D.; Campomanes, P.; Menéndez, M. I.; Sordo, T. L.; García-Granda, S. J. Am. Chem. Soc. 2003, 125, 3706–3707.
- Arun, M.; Joshi, S. N.; Puranik, V. G.; Bhawal, B. M.; Deshmukh, A. R. A. S. *Tetrahedron* 2003, *59*, 2309–2316.
- 84. Tanaka, Y.; Sakamoto, T.; Kamikawa, K.; Uemura, M. Synlett 2003, 519-521.

# 5.4.2 [3+2] Cycloadditions of Ketenes

[3+2] Ketene cycloadditions are 1,3-dipolar additions leading to 5-membered rings, and this area has been reviewed.¹

Dichloroketene (1) generated by dehydrochlorination reacted with the bridged cycloundecapentaenone 2 by a [3+2] cycloaddition to the ketenyl C=C bond, forming 3 (equation 1).²

$$Cl = C = 0 + 0 \longrightarrow 0 \longrightarrow 0$$

$$(1)$$

Diazoalkanes are 1,3-dipoles that give [3+2] cycloadditions with ketenes, as in the reaction of diphenylketene (4), which reacted with diphenyldiazomethane (5) to form 7 in a reaction that could be concerted or could proceed through a zwitterionic intermediate 6 (equation 2).³ Phenylketene (8) generated from diazoacetophenone (9) with silver benzoate and triethylamine formed the [3+2] cycloadduct 10 by reaction of 8 with 9, followed by tautomerization to 11 (equation 3).⁴



### 514 REACTIONS OF KETENES

*tert*-Butylcyanide *N*-oxide reacted with **4** to give a [3+2] adduct assigned as **12**, proposed to form with rearrangement by ring closure and ring opening (equation 4).^{5,6} Dimethylketene (**13**) reacted with 2,4,6-trimethylbenzonitrile *N*-oxide to give a product analogous to **12**.⁷ Nitrone **14** reacted with **4**, **15**, and **16** ( $\mathbb{R}^1 = t$ -Bu,  $\mathbb{R}^2 = CO_2Et$ ), forming mixtures of **17** and **18** in 52–78% yields (equation 5).^{8,9}



Thermolysis of dimethyl diazomalonate (19) formed a mixture of 21 and 22 in a process that could be envisioned to involve [3+2] cycloaddition of the ketene 20 with 19 in two regiochemical directions, with expulsion of nitrogen (equation 6).^{10–12} Other examples of ketene [3+2] cycloadditions with the precursor diazo ketones are shown in Sections 3.1.3, 3.3.1, and 4.1.10.2.



In further examples preformed ketene 4 gave [3+2] cycloadditions with diazo ester 19, forming 23 (equation 7),¹² and 24, forming 25 (equation 8),¹³ while the silylketene 28 reacted with the diazo ester precursor 27, forming 29 (equation 9).¹⁴

$$Ph = C = O + N = N = N = N = N = CO_2Me \xrightarrow{\Delta} CO_2Me \xrightarrow{A} MeO_2C \xrightarrow{O} OMe$$

$$4 \qquad 19 \qquad 23 \qquad (7)$$



A theoretical study of ketene cycloadditions with some metal oxides has appeared.¹⁵ Osmium tetroxide was reported to react by a [3+2] process (equation 10), while  $O_3Re-O^-$  gave [2+2] addition to the carbonyl group (equation 11) and MeReO₃ gave [2+2] addition to the C=C bond (equation 12).¹⁵

$$\begin{array}{c}
H \\
H \\
H \\
H
\end{array} = C = 0 \xrightarrow{O_{SO_4}} O_{O_{SO_5}} O_{O_{O_5}} O_{O_{O_{O_5}}} O_{O_{O_5}} O_{O_{O_5}}} O_{O_{O_5}} O_{O_{O_5}}} O_{$$

$$\begin{array}{cccc}
H & & & & \\
1 & & & & \\
\end{array} \xrightarrow{O_{3}Re-O^{-}} & H & & \\
H & & & & \\
O & ReO_{3}^{-} & & \\
\end{array} \tag{11}$$

### **REFERENCES FOR SECTION 5.4.2**

- 1. Huisgen, R. Angew. Chem. Int. Ed. 1963, 2, 565-598.
- 2. Mori, A.; Yan, Y. Z.; Takeshita, H.; Nozoe, T. Chem. Lett. 1997, 689-690.
- 3. Kirmse, W. Chem. Ber. 1960, 93, 2357-2360.
- 4. Takebayashi, M.; Ibata, T. Bull. Chem. Soc. Jpn. 1968, 41, 1700-1707.
- 5. Scarpati, R.; Sorrentino, P. Gazz. Chim. Ital. 1959, 89, 1525-1533.
- 6. Evans, A. R.; Taylor, G. A. J. Chem. Soc., Perkin Trans. 1 1983, 979-983.
- 7. Evans, A. R.; Taylor, G. A. J. Chem. Soc., Perkin Trans. 1 1987, 567-569.
- 8. Abou-Gharbia, M. A.; Jouillé, M. M. J. Org. Chem. 1979, 44, 2961-2966.

- Gettins, A. F.; Stokes, D. P.; Taylor, G. A.; Judge, C. B. J. Chem. Soc. Perkin Trans. 1 1977, 1849–1855.
- 10. Pomerantz, M.; Levanon, M. Tetrahedron Lett. 1991, 32, 995-998.
- 11. Maas, G.; Regitz, M. Chem. Ber. 1976, 109, 2039-2063.
- 12. Eichhorn, K.; Hoge, R.; Maas, G.; Regitz, M. Chem. Ber. 1977, 110, 3272-3278.
- 13. Kende, A. S. Chem. Ind. (London) 1956, 1053-1054.
- 14. Maas, G.; Gimmy, M.; Alt, M. Organometallics 1992, 11, 3813-3820.
- 15. Deubel, D. V.; Schlecht, S.; Frenking, G. J. Am. Chem. Soc. 2001, 123, 10085-10094.

## 5.4.3 [4+2] Cycloadditions

Preparative cycloaddition chemistry of ketenes is dominated by net [2+2] cycloadditions, and examples of [4+2] cycloadditions are rather limited. However, as noted above (Sec. 5.4.1.3), there are cases where [4+2] products have been formed as the initial intermediates, but then isomerize to [2+2] cycloaddition products.

In the reaction of diphenylketene (1) with the congested diene 2, [2+2] cycloaddition was hindered by steric crowding and the products **4a** and **4b** from net [4+2] cycloaddition to the C=C and C=O bonds of the ketene were observed. The formation of both could have occurred through the zwitterionic intermediate 3 (equation 1).¹



Cycloheptatriene reacted with diphenylketene (1) by a net [4+2] cycloaddition to the C=C bond of the ketene, forming **5** as the only observed product, but in only 11% isolated yield (equation 2).² The cyclohexadienes **6** and **7** reacted with diphenylketene, forming the [4+2] and [2+2] products **8** and **9** in 44:56 and 90:10 ratios, respectively (equation 3).^{3,4} Heating of **8** caused partial isomerization to **9**, suggesting that the [4+2] product **8** was the kinetic product.^{3,5} With R = Br only the product **9** was formed (equation 3).⁶

$$\xrightarrow{Ph}_{Ph} \xrightarrow{C=0} \xrightarrow{Ph}_{Ph} \xrightarrow{O}_{Ph}$$

$$\xrightarrow{(1)} \xrightarrow{Ph}_{Ph} \xrightarrow{O}_{Ph}$$

$$(2)$$



A variety of heteroatom-containing dienes underwent [4+2] cycloadditions with ketenes yielding heterocycles. Thus, formation of [4+2] cycloaddition products **12** occurred from ketenes **11** and arylimino-1,3,4-thiadiazoles **10** (equation 4).⁷ Lactone formation from reaction of ketene (**13**) with  $\alpha$ , $\beta$ -unsaturated aldehydes has long been known (equation 5),⁸ and reaction with 1,3-diazadiene **15** forms **16** after elimination of dimethylamine (equation 6).⁹ Reaction of the tautomer **16a** with ketene similarly gives **17** (equation 7).¹⁰



Reaction of the allylic amine **18** and RCH₂COCl with Me₃Al and K₂CO₃ occurred with aza-Claisen rearrangement, and may involve generation of ketenes **19** that form **20**, leading to **21** by a net [4+2] type cycloaddition (equation 8).¹¹ The reaction of diphenylketene (**1**) with ethoxy ethynyl ether proceeded by a [2+2+2] pathway, as shown by **22**, to form **23** (equation 9).¹² Naphthylketenes

reacted similarly (Section 5.4.1.6).¹²



Experimental and theoretical studies showed that imidoylketenes **25** formed by pyrolysis of acrylate esters **24** reacted by [4+2] cycloaddition pathways with planar, pseudopericyclic transition states, forming **26a** by reaction with **24** and **26b** by dimerization and rearrangement (equation 10).^{13,14}



The [4+2] and [2+2] cycloadditions of ketene (13) with 1,3-diazabutadiene forming 27a and 27b have been studied by DFT calculations (equation 11).¹⁵



1,2-Benzoquinones reacted by [4+2] cycloadditions with aryl- or alkylketenes generated in situ by thermal Wolff rearrangements, forming **28** (equation 12).^{16,17} The analogous bis(imines) **29** reacted similarly to form **30** (equation 13).¹⁸





Acylketenes often have given [4+2] cycloadditions, as in the reaction of acetylketene (**31**) with cyclohexanone to give dioxinone **32** (equation 14), and with *n*-butyl vinyl ether, forming **33**, used in the synthesis of sugars (equation 15).¹⁹ Benzonitrile reacted with the acylketene **34**, forming **35** (equation 16).²⁰



Ketenes **36** ordinarily react with pentamethylcyclopentadiene, giving [2+2] cycloaddition forming cyclobutanones, but in the presence of the oxidant **37** formed [4+2] adducts **38** via the ketene radical cation (equation 17).^{21,22} The cycloheptadieneone **39** was also formed by protonation of the diene, [2+2] cycloaddition, and rearrangement (Section 5.6.2).



As pointed out in Section 5.4.1.3, several reactions that result in net [2+2] cycloaddition between ketenes and dienes begin with an initial formation of a

transient [4+2] cycloaddition to the ketene carbonyl. Other examples of this reaction may be found.

# **REFERENCES FOR SECTION 5.4.3**

- 1. Mayr, H.; Heigl, U. W. J. Chem. Soc. Chem. Commun. 1987, 1804-1805.
- 2. Falshaw, C. P.; Lakoues, A.; Taylor, G. A. J. Chem. Res. S 1985, 106.
- Downing, W.; Latouche, R.; Pittol, C. A.; Pryce, R. J.; Roberts, S. M.; Ryback, G.; Williams, J. O. J. Chem. Soc. Perkin Trans. 1 1990, 2613–2615.
- 4. Maruya, R.; Pittol, C. A.; Pryce, R. J.; Roberts, S. M.; Thomas, R. J.; Williams, J. O. J. Chem. Soc. Perkin Trans. 1 1992, 1617–1621.
- 5. Davies, H. G.; Rahman, S. S.; Roberts, S. M.; Wakefield, B. J.; Winders, J. A. J. Chem. Soc. Perkin Trans. 1 1987, 85–89.
- 6. Roberts, S. M.; Sutton, P. W. J. Chem. Soc. Perkin Trans. 1 1995, 1499-1503.
- 7. Dutta, B. C.; Das, K. K.; Goswami, B. N. J. Chem. Res. (S) 1999, 36-37.
- 8. Young, F. G. J. Am. Chem. Soc. 1949, 71, 1346-1348.
- 9. Friot, C.; Reliquet, A.; Reliquet, F.; Meslin, J. C. Synthesis, 2000, 695-702.
- Landreau, C.; Deniaud, D.; Reliquet, F.; Reliquet, A.; Meslin, J. C. *Heterocycles* 2000, 53, 2667–2677.
- 11. Sudau, A.; Münch, W.; Nubbemeyer, U.; Bats, J. W. J. Org. Chem. 2000, 65, 1710–1720.
- 12. Wuest, J. D. Tetrahedron 1980, 36, 2291-2296.
- 13. Zhou, C.; Birney, D. M. J. Org. Chem. 2004, 69, 86-94.
- 14. Zhou, C.; Birney, D. M. J. Am. Chem. Soc. 2002, 124, 5231-5241.
- 15. Bharatam, P. V.; Kumar, R. S.; Mahajan, M. P. Org. Lett. 2000, 2, 2725-2728.
- 16. Reid, W.; Radt, W. Liebigs Ann. Chem. 1964, 676, 110-114.
- 17. Reid, W.; Radt, W. Liebigs Ann. Chem. 1965, 688, 170-173.
- 18. Friedrichsen, W.; Oeser, H.-G. Chem. Ber. 1975, 108, 31-47.
- 19. Coleman, R. S.; Fraser, J. R. J. Org. Chem. 1993, 58, 385-392.
- 20. Stetter, H.; Kiehs, K. Chem. Ber. 1965, 98, 2099-2102.
- 21. von Seggern, H.; Schmittel, M. Chem. Ber. 1994, 127, 1269-1274.
- 22. Schmittel, M.; von Seggern, H. J. Am. Chem. Soc. 1993, 115, 2165-2177.

# 5.4.4 Intramolecular Cycloaddition of Ketenes

Intramolecular cycloadditions of ketenes are often facile and have significant synthetic utility. The ring closures of vinyl-, imidoyl-, or acylketenes forming 4-membered rings were calculated at the MP4SDQ/6-31G**//HF/6-31G* level to be thermoneutral for formation of cyclobutanone (1) (equation 1) and endothermic by 10 and 17 kcal/mol, respectively, for formation of the azetone (2) and oxetone (3), respectively (equations 2, 3).¹ An example of preparative cyclobutenone

formation was cyclization of the vinylketene **5** formed by Wolff rearrangement of diazo ketone **4** to give **6** (equation 4).²



Cyclobutanone formation by intramolecular ketene cyclization has also been widely studied and was the subject of several reviews.^{3–5} For example, the ketene **7** generated by dehydrochlorination cyclized to the cyclobutanone in 80% yield (equation 5).⁶ In an application of intramolecular ketene [2+2] cycloaddition in synthesis of a precursor to (-)-salsolene oxide, the ketene **9** formed by dehydrochlorination of **8** gave intramolecular cycloaddition to form **10** (equation 6).⁷ Generation of allenylketenes **13** by dehydration of carboxylic acids **11** using Mukaiyama's reagent **12** resulted in [2+2] cycloaddition, forming **14** and **15** (equation 7).⁸





The reaction of 3-hydroxy-6-dimethyl-6-heptenoic acid (16) with potassium acetate and acetic anhydride (Perkin conditions)^{9–14} led to bicyclo[3.2.0]hept-3-en-6-ones 19 and 21, useful in synthesis,¹² and evidently involved formation of 17, which underwent competitive elimination to ketenes 18 and 20, which then cyclized (equations 8, 9).¹² Conversion of the acid 22 to the mixed anhydride and in situ elimination gave (2-alkenylphenoxy)ketene 23, which formed the tricyclic system 24 (equation 10).¹³



The reactivity in cyclizations forming cyclobutanones was influenced by both the reactivity of the ketene and the nucleophilicity of the alkene. Thus, the alkenyl group in **18** (equation 8) favors cyclization, as does the activating oxygen substituent in ketene **23** (equation 10). Cyclizations such as the conversion of the ketene **25** derived by dehydrochlorination, forming **26**, were evidently stepwise, for a

concerted  $[\pi 2_a + \pi 2_a]$  syn addition would be electronically forbidden and an allowed  $[\pi 2_s + \pi 2_a]$  reaction would not be geometrically feasible (equation 11).¹⁵



Dehydrochlorination of **27** formed the ketene **28**, which gave a six-electron electrocyclic ring closure to a 1:1 mixture of the isomers **29** in a useful synthesis of polycyclic rings (equation 12).^{16–19} This cyclization of dienylketenes to cyclohexadienones is the reverse of the Barton-Quinkert photochemical ring-opening reaction (Section 3.4.3).



Cyclobutenone **30** underwent thermal ring opening to form the  $\beta$ -phenylalkenylketene **31**, which cyclized to the naphthyl quinone **32** (equation 13).²⁰ Similar cyclization onto naphthyl groups gave phenanthrols, and these reactions are examples of the Smith-Hoehn reaction (Section 3.4.1).  $\beta$ -Arylalkenylketenes that underwent similar cyclization were also generated from the reaction of chromium carbene complexes with alkynes (Section 3.5).²¹ Other examples of such cyclizations were observed with ketenes generated by Wolff rearrangement (Section 3.3.1).



The ene-yne-substituted ketene **34** generated by thermolysis of the alkynylcyclobutenone **33** cyclized to form the quinone **35**, together with the cyclopentendione **35a** (equation 14).²² Photolysis of **33** gave an isomeric ene-yne ketene in which the vinyl OH group added to the ketene, forming a lactone.²² The dienylketene **37** generated by thermolysis of vinylcyclobutenone **36** formed the phenol **38** (equation 15).²³



Thermolysis or photolysis of the diazo ketones **39** gave indanols **42** proposed to form by intramolecular cyclization of the intermediate ene-yne ketenes **40**, giving diradicals **41** (equation 16).²⁴ Calculations at the HF/6-31G* level for the *s*-*trans* and *s*-*cis* conformers **40a** and **40b** showed that there was a preference for *s*-*trans* of 0.53 and 1.40 kcal/mol for R = Me and H, respectively. The lesser preference for **40a** in the case of R = Me was attributed to steric repulsion between the alkyne bond and the Me group, and the resulting higher proportion of **40b** for R = Me led to the higher yield of cyclization to indanol **42** in this case. Related  $\alpha$ -diazo ketones with DNA binding groups R gave improved DNA-cleaving activity compared to R = Me.²⁴ Further examples of DNA cleavage by ketenes are given in Section 4.1.6.



### **REFERENCES FOR SECTION 5.4.4**

- 1. Nguyen, M. T.; Ha, T.-K.; More O'Ferrall, R. A. J. Org. Chem. 1990, 55, 3251-3256.
- Andriamiadanarivo, R.; Pujol, B.; Chantegrel, B.; Deshayes, C.; Doutheau, A. Tetrahedron Lett. 1993, 34, 7923–7924.
- 3. Snider, B. Chem. Rev. 1988, 88, 793-811.
- 4. Snider, B. B. Chemtracts-Org. Chem. 1991, 4, 403-419.
- Snider, B. B. In Advances in Strain in Organic Chemistry; JAI Press, London, 1992, Vol. 2, pp. 95–142.
- Markó, I.; Ronsmans, B.; Hesbain-Frisque, A.-M.; Dumas, S.; Ghosez, L.; Ernst, B.; Greuter, H. J. Am. Chem. Soc. 1985, 107, 2192–2194.
- Paquette, L. A.; Sun, L.-Q.; Watson, T. J. N.; Friedrich, D.; Freeman, B. T. J. Am. Chem. Soc. 1997, 119, 2767–2768.
- 8. McCaleb, K. L.; Halcomb, R. L. Org. Lett. 2000, 2, 2631-2634.
- 9. Marotta, E.; Medici, M.; Righi, P.; Rosini, G. J. Org. Chem. 1994, 59, 7529-7531.
- 10. Rosini, G.; Laffi, F.; Marotta, E.; Pagani, I.; Righi, P. J. Org. Chem. 1998, 63, 2389-2391.
- Rosini, G.; Confalonieri, G.; Marotta, E.; Rama, F.; Righi, P. Org. Synth. 1996, 74, 158– 168.
- Confalonieri, G.; Marotta, E.; Rama, F.; Righi, P.; Rosini, G.; Serra, R.; Venturelli, F. *Tetrahedron* 1994, 50, 3235–3250.
- 13. Marotta, E.; Pagani, I.; Righi, P.; Rosini, G. Tetrahedron 1994, 50, 7645–7656.
- Donate, P. M.; da Silva R.; Valdo, G.; da Silva, J.; Alemán, C. Synth. Commun. 2001, 31, 141–150.
- 15. Corey, E. J.; Desai, M. C.; Engler, T. A. J. Am. Chem. Soc. 1985, 107, 4339-4341.
- Barron, C. A.; Khan, N.; Sutherland, J. K. J. Chem. Soc. Chem. Commun. 1987, 1728– 1730.
- Khodabocas, A.; Shing, T. K. M.; Sutherland, J. K.; Williams, J. G. J. Chem. Soc., Chem. Comm. 1989, 783–784.
- 18. Larsen, L.; Sutherland, J. K. J. Chem. Soc. Chem. Comm. 1989, 784-786.
- 19. Khodabocas, A. Arkivoc 2002, (i) 25-34.
- 20. Liebeskind, L. S.; Zhang, J. J. Org. Chem. 1991, 56, 6379-6385.
- Bao, J.; Wulff, W. D.; Dominy, J. B.; Fumo, M. J.; Grant, E. B.; Rob, A. C.; Whitcomb, M. C.; Yeung, S.-M.; Ostrander, R. L.; Rheingold, A. L. J. Am. Chem. Soc. 1996, 118, 3392–3405.
- Foland, L. D.; Karlsson, J. O.; Perri, S. T.; Schwabe, R.; Xu, S. L.; Patil, S.; Moore, H. W. J. Am. Chem. Soc. 1984, 111, 975–989.
- 23. Krysan, D. J.; Gurski, A.; Liebeskind, L. S. J. Am. Chem. Soc. 1992, 114, 1412-1418.
- Nakatani, K.; Maekawa, S.; Tanabe, K.; Saito, I. J. Am. Chem. Soc. 1995, 117, 10635– 10644.

#### 5.4.5 Cycloaddition of Ketenes with Carbonyl Groups

The [2+2] cycloaddition of ketenes with carbonyl groups forming  $\beta$ -lactones has long been known and is a valuable experimental tool because of the synthetic utility

of the products.^{1–5} This method has acquired increasing prominence because of the development of methods that give high stereoselectivity in the formation of new chiral centers. A striking early example was the addition of trichloroacetaldehyde to ketene catalyzed by a chiral tertiary amine to give the  $\beta$ -lactone **1** with high stereoselectivity (equation 1).^{6,7}

$$\overset{H}{\underset{H}{\longrightarrow}} C=O + CCl_{3}CH=O \xrightarrow{\text{quinidine}}_{\text{toluene}, -50 \, ^{\circ}C} O (1)$$

$$\overset{I}{\underset{M}{\longrightarrow}} 1 (89\%, 98\% \, ee)$$

Ketene underwent [2+2] cycloaddition with the electrophilic carbonyl group in hexafluoroacetone forming the  $\beta$ -lactone **2** (equation 2),⁸ and formed **3** with chlorodifluoroacetaldehyde in a reaction promoted with a Lewis acid catalyst (equation 3).⁹

The  $\beta$ -lactones formed by ketene cycloadditions are prone to decarboxylation, and the cycloaddition provides a means of olefination of carbonyl groups, as in the conversion of **4** to **5** using diphenylketene (equation 4)¹⁰ and the formation of **6** using dicyanoketene (equation 5).¹¹



Ketene [2+2] cycloaddition with carbonyl groups has been studied by computational methods.^{12,13} These studies included eight different gas phase and solution transition states for the reaction of chloroketene with acetaldehyde, forming 7 (equation 6).¹² It was found that the formation of *trans*-2-oxetanones was favored in the absence of catalyst, whereas the presence of catalyst favored the *cis* isomer, in agreement with experiment.¹² Theoretical study of BF₃ catalysis showed that coordination to the aldehyde carbonyl was favored, with a major reduction in the activation energy for C–C bond formation, while inclusion of solvent in the computation was shown to lead to an earlier transition state.¹⁴ The interaction with BH₃ favored C–C bond formation and hydride donation to carbon, forming **8** (equation 7).¹⁴

$$CI = C = 0 + CH_3 + H \longrightarrow CI = 0 + CH_3 + H \longrightarrow CH_3 + O = 0 + CH_2 = 0 + CH_2$$

Extensive theoretical studies of stereocontrol in [2+2] cycloadditions between ketenes and the aldehyde CH₃OCH(CH₃)CH=O forming **9** included the effects of lithium cation catalysis (equation 9).¹⁵ In the uncatalyzed reactions highly asynchronous concerted  $[_{2}\pi_{S}+(_{2}\pi_{S}+_{2}\pi_{S})]$  transition states were favored.

$$\overset{R}{\searrow} = C = O + \underset{CH_3}{\overset{O}{\longrightarrow}} \underset{CH_3}{\overset{H}{\longrightarrow}} \underset{CH_3}{\overset{R}{\longrightarrow}} \underset{CH_3}{\overset{O}{\longrightarrow}} \underset{CH_3}{\overset{R}{\longrightarrow}} \underset{CH_3}{\overset{O}{\longrightarrow}}$$
(8)

The development of catalytic, enantioselective methods has increased the utility of  $\beta$ -lactone formation by ketene cycloaddition with carbonyl groups.¹⁶ The stereoselectivity of the BF₃-catalyzed cycloaddition of cyclohexanecarboxaldehyde with ketene **10** (equation 9) has been studied both computationally and by measurement of the ¹³C kinetic isotope effects.¹⁷ The reaction gave >90% yields of **12** as a *cis/trans* mixture with a diastereoselectivity that varied from 2.7–3.1:1 at 23 °C to 1.3:1 at -21 °C.¹⁷ Distinctly different ¹³C kinetic isotope effects were observed for the two diastereomers, which were both interpreted as forming from the stepwise process of equation 9. The first step forming **11** was rate limiting for formation of the *trans* product, whereas the second step forming **12** from **11** was rate limiting for formation of the *cis* product.



The reaction of silylketenes with carbonyl compounds leading to  $\beta$ -lactones served as an alternative to the Wittig reaction¹⁸ and supplied precursors for studies of ring cleavage of the lactones.¹⁹ This reaction was also used to prepare five different panclicins **13**, which are pancreatic lipase inhibitors containing  $\beta$ -lactone rings.²⁰ Chiral catalysts such as the Ti-TADDOL derivatives **14** catalyzed the cycloaddition of aldehydes with Me₃SiCH=C=O with high stereoselectivity (equation 10).²¹



Bis(oxazoline)-Cu(II) complexes catalyzed stereoselective [4+2] cycloaddition of ethyl cinnamate with trimethylsilyketene, forming **15** (equation 11)²² (see also Section 4.5). Such reactions have been reviewed.²³ A new in situ ketene generation with an optically active catalyst has been developed for  $\beta$ -lactone formation (Section 5.9).²⁴ Palladium catalysts promoted formation of isolable  $\beta$ -lactones **17** from CH₂=C=O and 4-substituted vinyl aldehydes **16**, and these were isomerized to 3,6-dihdyro-2-(*H*)pyran-2-ones **18** (equation 12).^{25,26}



Reaction of dichloroketene generated by zinc dechlorination or dehydrochlorination with optically active  $\alpha$ -silyloxy or  $\alpha$ -amino aldehydes gave only single diastereomers of the  $\beta$ -lactones **19** or **20**, respectively (equations 13, 14).²⁷ The ketene

n-C₆H₁₃C(SiMe₃)=C=O reacted with these aldehydes to give the analogous products, with the stereochemistry established by X-ray analysis.²⁸



The [4+2] cycloaddition of acetylketene formed by FVP with trapping by chiral racemic aldehydes **21**, forming dioxinones **21a** and **21b**, proceeds stereoselectively, favoring the product **21a** by factors of 2.0–6.3:1.0, as predicted by the Felkin-Anh model (equation 15).²⁹ The pseudopericyclic transition state for this reaction has been calculated.²⁹



The reaction of acyl chlorides **22** with  $Et_3N$  in 5 M LiClO₄ in diethyl ether gave ketenes **23**, which reacted in situ with aldehydes at 0 °C, forming β-lactones **24** (equation 16).^{15,30} The reaction mixtures were stirred at room temperature, and the intermediate oxetanones **24** underwent decarboxylation to give alkenes or dienes (equation 16).³⁰ Both steps of this olefination sequence were proposed to be promoted by this highly polar aprotic medium.³⁰ The adduct **24a** ( $R^1 = R^2 = Me$ ,  $R^3 = 4-O_2NC_6H_4$ ) did not undergo decarboxylation under these conditions, and for  $R^1 = R^2 = H$  acids  $R^3CH=CHCO_2H$  were formed.

The reactions of acyl halides with aldehydes induced by diisopropylethylamine (DIPEA) in the presence of  $Al(SbF_6)_3$  catalyst gave  $\beta$ -lactones **25** with high

*cis/trans* product ratios (equation 17).^{31–36} Mechanistic studies implicated ketene intermediates in these reactions. The lactones gave  $\beta$ -hydroxy esters in catalyzed reactions with alcohols. Chiral catalyst **26** for cycloaddition of CH₂=C=O gave optically active  $\beta$ -lactones **27**, which were converted to **28** (equation 18).³⁵



Photochemical cycloaddition of diphenylketene to phenanthrenequinone gave **28** (equation 19).³⁷ Thermal reaction of the 1,5-azulenequinone **29** with Ph₂C=C=O formed the corresponding quinodimethane **30** through [2+2] cycloaddition and loss of CO₂ (equation 20).³⁸



Ketene **32** containing an aldehyde carbonyl group was generated from aldoacid **31** with Mukaiyama's reagent and gave intramolecular cycloaddition forming **33** in 86–92% *ee*, using 9-*O*-acetylquinine as catalyst (equation 21).^{39,40} There were earlier reports of such intramolecular cycloadditions.^{41–45}



2-Ketenylcyclobutanones **34** generated by Wolff rearrangement gave lactones **35** by carbonyl addition with rearrangement (equation 22).⁴⁶ The intermediate ketenes were observed by IR at 2120 cm⁻¹ and -50 °C and were trapped with alcohols. The aryl ketene **35** cyclized to pyrone **36** (equation 23).⁴⁷ Further examples of formation of  $\gamma$ -pyrones are given in Section 4.1.2.



## **REFERENCES FOR SECTION 5.4.5**

- Ghosez, L.; Marchand-Brynaert, J. In *Comprehensive Organic Synthesis*; Trost, B. M., Ed.; Pergamon, New York, 1991, Vol. 5, Chapter 2.2.
- 2. Muller, L. L.; Hamer, J. 1,2-Cycloaddition Reactions: The Formation of Three- and Four-Membered Heterocycles; Wiley Interscience, New York, 1967, Chapter 3.
- Ghosez, L.; O'Donnell, M. J. In *Pericyclic Reactions*; Marchand, A. P.; Lehr, R. E., Eds.; Academic Press, New York, 1977, pp. 79–140.
- 4. Ulrich, H. *Cycloaddition Reactions of Heterocumulenes*; Academic press, New York, 1967, pp. 38–109.
- 5. Pommier, A.; Pons, J.-M. Synthesis 1993, 441-459.
- 6. Wynberg, H.; Staring, E. G. J. J. Am. Chem. Soc. 1982, 104, 166-168.
- 7. Wynberg, H.; Staring, E. G. J. J. Org. Chem. 1985, 50, 1977-1979.
- 8. Knunyants, I. L.; Cheburkov, Yu. A. Izvest. Akad. Nauk SSSR, Otdel. Khim. 1960, 678–685. Chem. Abstr. 1960, 54 22349c.
- 9. Ito, T.; Shimizu, M.; Fujisawa, T. Tetrahedron 1998, 54, 5523-5530.
- Terpetschnig, E.; Penn, G.; Kollenz, G.; Peters, K.; Peters, E.-M.; von Schnering, G. Tetrahedron 1991, 47, 3045–3058.
- 11. Neidlein, R.; Bernhard, E. Angew. Chem. Int. Ed. Engl. 1978, 17, 369-370.
- Lecea, B.; Arrieta, A.; Lopez, X.; Ugalde, J. M.; Cossío, F. P. J. Am. Chem. Soc. 1995, 117, 12314–12321.
- 13. Pons, J.-M.; Pommier, A.; Rajzmann, M.; Liotard, D. J. Mol. Struct. (Theochem.) 1994, 313, 361–364.

- Pons, J.-M.; Oblin, M.; Pommier, A.; Rajzmann, M.; Liotard, D. J. Am. Chem. Soc. 1997, 119, 3333–3338.
- 15. Lecea, B.; Arrieta, A.; Arrastia, I.; Cossío, F. P. J. Org. Chem. 1998, 63, 5216-5227.
- 16. Schneider, C. Angew. Chem. Int. Ed. 2004, 41, 744-746.
- Singleton, D. A.; Wang, Y.; Yang, H. W.; Romo, D. Angew. Chem. Int. Ed. 2002, 42, 1572–1575.
- Black, T. H.; Zhang, Y.; Huang, J.; Smith, D. C.; Yates, B. E. Synth. Commun. 1995, 25, 15–20.
- 19. White, D.; Zemribo, R.; Mead, K. T. Tetrahedron Lett. 1997, 38, 2223-2226.
- Kocienski, P. J.; Pelotier, B.; Pons, J.-M.; Prideaux, H. J. Chem. Soc. Perkin Trans. 1 1998, 1373–1382.
- 21. Yang, H. W.; Romo, D. Tetrahedron Lett. 1998, 39, 2877-2880.
- 22. Evans, D. A.; Janey, J. M. Org. Lett. 2001, 3, 2125-2128.
- 23. Yang, H. W.; Romo, D. Tetrahedron 1999, 55, 6403-6434.
- 24. Tennyson, R.; Romo, D. J. Org. Chem. 2000, 65, 7248-7252.
- 25. Hattori, T.; Suzuki, Y.; Uesugi, O.; Oi, S.; Miyano, S. Chem. Commun. 2000, 73-74.
- 26. Hattori, T.; Suzuki, Y.; Ito, Y.; Hotta, D.; Miyano, S. Tetrahedron 2002, 58, 5215-5223.
- Palomo, C.; Miranda, J. I.; Cuevas, C.; Odriozola, J. M. J. Chem. Soc. Chem. Commun. 1995, 1735–1736.
- 28. Kocienski, P.; Pelotier, B.; Webster, M. Acta Crystallogr., Sect. C, Cryst. Struct. Commun. C52 1996, 3150–3152.
- 29. Shumway, W.; Ham, S.; Moer, J.; Whittlesey, B. R.; Birney, D. M. J. Org. Chem. 2000, 65, 7731–7739.
- 30. Arrastia, I.; Cossío, F. P. Tetrahedron Lett. 1996, 37, 7143-7146.
- 31. Nelson, S. G.; Wan, Z.; Peelen, T. J.; Spencer, K. L. Tetrahedron Lett. 1999, 40, 6535–6539.
- 32. Nelson, S. G.; Spencer, K. L. J. Org. Chem. 2000, 65, 1227-1230.
- 33. Nelson, S. G.; Peelen, T. J.; Wan, Z. Tetrahedron Lett. 1999, 40, 6541-6543.
- 34. Nelson, S. G.; Wan, Z. Org. Lett. 2000, 2, 1883-1886.
- 35. Zipp, G. G.; Hilfiker, M. A.; Nelson, S. G. Org. Lett. 2002, 4, 1823-1826.
- 36. Wan, Z.; Nelson, S. G. J. Am. Chem. Soc. 2000, 122, 10470-10471.
- Schnapp, K. A.; Motz, P. L.; Stoeckel, S. M.; Wilson, R. M.; Bauer, J. A. K.; Bohne, C. *Tetrahedron Lett.* **1996**, *37*, 2317–2320.
- 38. Mori, A.; Yan, Y. Z.; Takeshita, H.; Nozoe, T. Chem. Lett. 1997, 689-690.
- 39. Cortez, G. S.; Tennyson, R. L.; Romo, D. J. Am. Chem. Soc. 2001, 123, 7945-7946.
- 40. Coretz, G. S.; Oh, S. H.; Romo, D. Synthesis, 2001, 1731-1736.
- 41. Rull, T.; Ourisson, G. Bull. Soc. Chim. Fr. 1958, 1573-1581.
- 42. Kagan, H. B.; Jacques, J. Bull. Soc. Chim. Fr. 1958, 1600-1602.
- 43. Ayral-Kaloustian, S.; Wolff, S.; Agosta, W. C. J. Org. Chem. 1978, 43, 3314-3319.
- 44. Brady, W. T.; Marchand, A. P.; Giang, Y. F.; Wu, A.-H. Synthesis 1987, 395-396.
- 45. Brady, W. T.; Gu, Y.-Q. J. Heterocyclic Chem. 1988, 25, 969-971.
- 46. Miller, R. O.; Theis, W.; Helig, G.; Kirchmeyer, S. J. Org. Chem. 1991, 56, 1453-1463.
- 47. Fairfax, D. J.; Austin, D. J.; Xu, S. L.; Padwa, A. J. Chem. Soc. Perkin Trans. 1 1992, 2837–2844.
# 5.5 NUCLEOPHILIC ADDITION TO KETENES

## 5.5.1 Mechanisms of Hydration of Ketenes

The hydration of carbonyl compounds forming 1,1-diols has been found by experimental¹ and computational^{2–4} studies to involve multiple water molecules in which water clusters effect net hydroxide addition to the carbonyl carbon and proton addition to the carbonyl oxygen through hydrogen-bonded arrays (equation 1).

$$\underset{H}{\overset{O}{\longrightarrow}} \underset{H}{\overset{nH_{2}O}{\longrightarrow}} \underset{H}{\overset{HO}{\longrightarrow}} \underset{H}{\overset{OH}{\longrightarrow}}$$
(1)

Most ketenes are too reactive for measurements using conventional spectroscopy of the rate constants for addition of nucleophiles such as water to preformed ketenes, although early interpretations of hydration indicated that enediols were formed as the first intermediates.^{5,6} Studies using relatively unreactive ketenes in organic solvents containing small amounts of water permitted measurements of rate constants that were more than first order in water concentration^{7–14} but did not provide satisfactory interpretations of the details of the reactions.¹⁵

Initial quantitative measurements of the reactivity of arylketenes **1** generated by Wolff rearrangements using flash photolysis of diazo ketones in wholly aqueous media were carried out by measuring the change in conductivity due to the ionization of the substituted phenylacetic acids formed (equation 2).¹⁶ The arylketenes showed a dependence of the measured rate constants log*k*(H₂O) on the  $\sigma^n$  constants of the aryl substituents with a value of  $\rho = 1.19$ , indicating negative charge development in the transition state **2** near the aryl group. The reactivity of ketene (**5**) itself was then studied, and was found to be much less reactive than the arylketenes.¹⁷ This also indicated a significant conjugative stabilization by the aryl groups in the transition state for ketene hydration. Activation parameters  $\Delta H^{\ddagger} = 10.3$  kcal mol⁻¹ and  $\Delta S^{\ddagger} = -16$  cal deg⁻¹ mol⁻¹ and a solvent isotope effect  $k(H_2O)/k(D_2O) = 1.9$  were measured.¹⁷ The value of  $\Delta S^{\ddagger}$  is consistent with an associative mechanism, and the solvent isotope effect is consistent with nucleophilic water attack.



Computational studies of the hydration of  $CH_2=C=O(5)^{18-24}$  were generally in accord with the description for hydration of other carbonyl compounds,¹⁻⁴ and for addition of two water molecules a cyclic transition state **6** leading to the enol **7** of acetic acid was preferred (equation 3). A significantly higher barrier was found for addition of a single water molecule, as confirmed by the most recent computation.²² Structures with larger numbers of water molecules have also been considered,^{19,21}

and the enol **7** in solution would also be solvated by further water molecules. For the parent ketene **5**, addition using two water molecules to the C=C bond giving acetic acid directly was calculated to be close in energy to addition to the C=O bond, but for substituted ketenes in which stabilization of a polar transition state would be possible, as in equation 2, addition to the carbonyl group would be favored. The kinetics of ketene hydration have been examined computationally by "No Barrier Theory," and this gave a reasonable estimate of the known rate constant.²⁴ A theoretical study of the addition reactions of a single molecule of H₂O, H₂S, NH₃, HF, and HCN to ketene provided comparative information²⁵ but was not the best description of the reaction with multiple solvent molecules.



The ketene HOMO is perpendicular to the ketene plane with negative charge at C₂ and oxygen, whereas the LUMO is in plane with positive charge at C₂, the carbonyl carbon (Figure 5.4).^{26,27} Electrophilic addition to the C=C bond occurs on the HOMO and would require out-of-plane attack at the  $\pi$  bond perpendicular to the ketene plane in **5a**, and nucleophilic addition to the carbonyl group of ketene would involve attack at the LUMO in-plane  $\pi$  bond in **5b**.



Figure 5.4 Frontier molecular orbitals of ketene.

Experimental measurements of the gas phase hydration of ketene showed an uncatalyzed surface reaction, and a homogeneous reaction between ketene with one water molecule catalyzed by a single acetic acid molecule, but did not allow a definite conclusion as to the mechanism.²⁸ Experimental studies of the gas phase dehydration of acetic acid to ketene have also been carried out,^{22,29} and when the direct elimination to ketene and initial acid enol formation were compared, the latter process was found to be favored.²⁹

The hydration reactions of a number of ketenes have been studied by measuring the decrease in the UV absorption of the ketene, and the derived rate constants are collected in Table 5.13. For a few examples, the ketenes were sufficiently stable so that the reactions have been carried out by adding the prepared ketene to either a wholly aqueous or a partially aqueous solvent mixed with an organic component.^{30–35} Extrapolations permitted estimates of ketene reactivities in purely aqueous solutions from measurements in mixed solvents. Flash photolysis in water of suitable precursors, usually diazo ketones, has been used for the generation of more reactive ketenes in situ, and the kinetics have been monitored by UV spectroscopy.^{36–46} These include the kinetics of hydration of ketene (**5**) generated from diazoacetalde-hyde (N₂CHCH=O) in acid, neutral, and base solution.^{38,40}

The formation of intermediate ethene-1,1-diols (ketene hydrates) was first proposed by Staudinger and Meyer in 1922.⁵ Experimentally, acid enol intermediates have been observed for hydration of some ketenes by UV spectroscopy,^{41–45} including the reaction of phenylketene (**8**), forming the acid enol **8a**, and then phenylacetic acid (equation 4).^{45,46} Interestingly, when the absorption changes in the UV for this reaction were monitored at 260 nm, initial increase in absorbance followed by decay was observed, and it was concluded that the initial rise in absorption did not represent the hydration of the phenylketene, but rather that the subsequent decrease in absorption was due to conversion of **8** to **8a**.⁴⁵



The hydration with ¹⁸O-labeled H₂O of crowded arylketenes and diphenylketene resulted in incorporation of more than one ¹⁸O in the product diarylacetic acid, indicating that the hydration was reversible (equation 5).⁴⁷ This result was confirmed by the observation of ¹⁸O incorporation in the recovered ketene.⁴⁷ The hydration of bis(2,4,6-triisopropylphenyl)ketene (**9a**) has been shown to involve reversible formation of the 1,1-enediol **10**, which was directly observed as a long-lived species in solution by NMR (equation 5).^{48,49} Acid enols Ar₂C=C(OH)₂ (**10**, Ar = 2,4,6-Me₃C₆H₂ and C₆Me₅) from the hydration of the corresponding ketenes **9** have also been observed as reasonably long-lived intermediates in solution.⁵⁰ In the hydration of PhC(OH)=C=O (**11**) an acid-catalyzed term was detected that was interpreted as indicating the formation of the acid enol **13**, but the mechanism of this process

			ŀ	ŀ	$\frac{\mathbf{k}_{\mathrm{H}^+}}{\mathbf{k}_{\mathrm{H}^+}}$	k _{OH} -	
Ketene	$k_{\rm H}^+,$	$k_{\rm H_2O}, s^{-1}$	$M^{-1}s^{-1}$	$\frac{k_{H_2O}}{k_{D_2O}}$	$(M^{-1})$	$(M^{-1})$	Ref.
$CH_2 = C = O$	$1.01  imes 10^4$	44, 36.5	$5.26  imes 10^4$	1.59	276		17, 38
<i>n</i> -BuCH=C=O	$3.98 \times 10^3$	99.4	$3.29  imes 10^4$	1.88	40	330	36
ArCH=C=O, Ar=							
$4-O_2NC_6H_4$		$49.5 \times 10^{3}$					16
$4-NCC_6H_4$		$25.6 \times 10^{3}$					16
$4-ClC_6H_4$		$9.6 \times 10^{3}$					16
$4\text{-FC}_6\text{H}_4$		$5.9 \times 10^{3}$					16
$C_6H_5$		$4.77 \times 10^{3}$	$1.22 \times 10^{6}$	1.44		260	16, 36
$C_6F_5$		$2.76 \times 10^{3}$	$1.13 \times 10^{6}$	1.42		409	63
$4-CH_3OC_6H_4$		$2.9 \times 10^{3}$	$1.13 \times 10^{\circ}$			390	58
$4-CH_3C_6H_4$		$3.8 \times 10^{3}$	6				16
1-Naphthyl	d	$2.95 \times 10^{-3}$	$1.25 \times 10^{6}$	1.25		420	58
Mesityl	$4.7^{a}$	$1.10 \times 10^{2}$	$6.78 \times 10^{4}$	1.51	0.042	620	52, 58
$2,4,6-(MeO)_3C_6H_2$		8.28	$4.34 \times 10^{-5}$			520	58
2-Pyridyl		$1.1 \times 10^{4}$					46
3-Pyridyl		$2.5 \times 10^{4}$					46
4-Pyridyl		$5.0 \times 10^{-10}$					46
$Me_3SiCH=C=O$		0.254					33
$CF_3CH=C=0$		$6 \times 10^{-204}$	2 40 105	0.50		016	39
c-PrCH=C=O	1 (1 105	304 7.15 - 10 ⁴	$2.48 \times 10^{2}$	2.58	2.2	816	39
PhC=CCH=C=0	$1.61 \times 10^{-5}$	$7.15 \times 10$	$1.98 \times 10^{\circ}$	2.84	2.2	211	39
Ph C ⁼⁰	$2.09 \times 10^4$	$5.76  imes 10^3$	$2.31 \times 10^6$	2.09	3.6	401	39
0=C=C=C=0	0.195	$1.45  imes 10^{-2}$		2.20			64
$CH_3SCPh=C=O$	4.67	$1.02 \times 10^2$	$1.89  imes 10^4$	1.59	0.046	232	62
		$2.21 \times 10^5$	$1.66 \times 10^{7}$	1.20		75	71
∕s′		5					
HOCPh=C=O	$5.45 \times 10^{7}$	$5.51 \times 10^{-3}$	$2.33 \times 10^{9}$	1.49	9.9	4200	41
PhCOCPh=C=O	3	$6.29 \times 10^{-3}$					39, 71
t-BuCH=C=O	$4.77 \times 10^{3}$	14.6	$1.68 \times 10^{-4}$	2.01	327		31, 38
$Et_2C=C=O$		8.26 ^a					31
$\bigcirc = 0$	$3.13\times10^5$	$4.09\times10^2$			765		63
C=0	$2.14  imes 10^4$	$1.11 \times 10^2$	$2.14  imes 10^4$	3.39	193	350	63
C=0	$2.78  imes 10^3$	23.0	$6.24\times10^3$	2.04	121	271	31, 63
$\bigcirc$ =c=o		23.5	$6.08  imes 10^3$	1.91		259	63

# TABLE 5.13 Hydration Rate Constants for Ketenes at 25 $^\circ \text{C}$

					$\mathbf{k}_{\mathbf{H}^+}$	k _{OH} -	
			$k_{OH^{-}},$	$k_{H_2O}$	$k_{H_2O}$	$k_{H_2O}$	
Ketene	$k_{H}^{+},$	$k_{H_2O}, s^{-1}$	$M^{-1}s^{-1}$	$k_{D_2O}$	$(M^{-1})$ (	$(M^{-1})$	Ref.
						5	
t-Bu ₂ C=C=O	434	$1.57 \times 10^{-4}$		2	$2.8 \times 10^{\circ}$	J	30, 31
c-Pr ₂ C=C=O		13.4 ^{<i>a</i>}					34
$Ph_2C=C=O$		275	$6.11 \times 10^{4}$	1.27		220	36
MeCPh=C=O		55.7	$2.03 \times 10^{4}$	1.55	36		63
EtCPh=C=O		10.3	$6.05 \times 10^{3}$	1.48		587	63
c-PrCPh=C=O		11.1 ^a					34
i-PrCPh=C=O		$1.23^{a}$					32
t-BuCPh=C=O		$0.370^{a}$					32
		$1.96 \times 10^{3}$	$8.42 \times 10^{5}$	1 18		430	63
$\sum = c = 0$		1.90 × 10	0.42 × 10	1.10		-50	05
$\langle \rangle$							
$\succ$		103	$6.30 \times 10^{-4}$	1.60		612	63
<							
	7.03	$2.97 \times 10^{-4}$	0 104	2 32	$2.4 \times 10^{-10}$	0 ⁴ 350	) 32
	7.05	2.97 × 10	0.104	2.52	2.7 / 10	5 550	52
t-BuC(CO ₂ Et)=C=O		0.124		1.48			35
t-BuC(COBu- $t$ )=C=O		$2.35 \times 10^{-4}$	0	1.37			35
$CH_3COCH=C=O$		$1.54 \times 10^{6}$	$1.86 \times 10^{8}$	1.11		120	72
$CF_3COCH=C=O$		$1.75 \times 10^{6}$		0.88			51
-0							
C ^C		0.0 1084					-
		$8.2 \times 10^{60}$					/8
0							
$\mathcal{A}$		$1.4 \times 10^{6}$	$7.5 \times 10^{7}$		54		73
$\succ c = 0$							
$\checkmark$							
~~C ^{±0}			_				
ſŤ		$3.45 \times 10^{4}$	$1.08 \times 10^{7}$				77
0							
		0.0 105	4.0 107			50	27 (2
$\Box$		$9.0 \times 10^{-5}$	$4.8 \times 10^{-5}$			53	37,63
~							
U L		$1.2 \times 10^{6}$	$3.8 \times 10^{7c}$	1 32		36	55 63
$\swarrow$ $= c=0$		1.2 \ 10	5.0 A 10	1.52		50	55, 05

Ketene	k ₁ .	kн.о. s ⁻¹	$k_{OH^-}, \ M^{-1}s^{-1}$	$\frac{\mathbf{k}_{\mathrm{H}^{+}}}{\mathbf{k}_{\mathrm{H}_{2}\mathrm{O}}} \frac{\mathbf{k}_{\mathrm{H}^{+}}}{\mathbf{k}_{\mathrm{H}_{2}\mathrm{O}}} \frac{\mathbf{k}_{\mathrm{O}}}{\mathbf{k}_{\mathrm{H}}}$	$\frac{\underline{\mathrm{H}}^{-}}{\underline{\mathrm{I}}_{2}\mathrm{O}}$ Ref.
C=0		$1.07 \times 10^5$	$3.47 \times 10^{6}$	1.35	35 57
C ⁻⁰ CH ₂	$1.44  imes 10^5$	$1.35 \times 10^5$	$9.26  imes 10^6$	1.46	76
$\begin{array}{l} Mes_2C=C=O^a \\ (Me_5C_6)_2C=C=O^a \\ (Me_3Si)_2C=C=O^a \\ (Me_3SiC=C=O)_2^a \\ Me_2Si(CH=C=O)_2^a \\ Me_3SiCH=C=O \\ PhMe_2SiCH=C=O^a \\ t\text{-}BuMe_2SiCH=C=O^a \\ Me_5Si_2CH=C=O^a \\ (SiMe_2C=C=O)_2^a \\ (CH_2SiMe_2CH=C=O)_2^a \end{array}$	542 ^a )) ₂ ^a	$\begin{array}{c} 0.68^{a} \\ 0.16^{a} \\ 2.46 \times 10^{-4a} \\ 0.307^{a} \\ 3.34 \times 10^{-2} \\ 3.23 \times 10^{-3} \\ 5.40 \times 10^{-3} \\ 9.58 \times 10^{-4} \\ 9.31 \times 10^{-3} \\ 1.52 \times 10^{-2} \\ 5.19 \times 10^{-3} \end{array}$	70	$1.7 \times 10^5$ 2.1	$50 \\ 50 \\ 69 \\ 82 \\ 70 \\ \times 10^4 33 \\ 68 \\ 68 \\ 69 \\ 69 \\ 69 \\ 69 \\ 69 \\ 69$

The contraction	TA	BLE	5.13	(Continue	d
-----------------	----	-----	------	-----------	---

^aExtrapolated.

has not been explained (equation 6).⁴¹



The hydration of mesitylketene (**15**, 2,4,6-Me₃C₆H₂CH=C=O) generated by flash photolysis revealed acid-catalyzed, neutral, and OH⁻-catalyzed reactions.⁵² The arylketenes PhCR=C=O (**16**, R = *i*-Pr, *c*-Pr, *t*-Bu) previously were shown to undergo acid-catalyzed hydration by proton transfer to C₂ with formation of the intermediate acylium ions **17** (equation 7).^{32,34} The acid-catalyzed hydration requires rather high acid concentrations, a result attributed to ground state stabilization by aryl, as well as destabilization of the positively charged transition state by aryl.³²

The fulvenones **19**,³⁷ **20**,^{53–56} and **21**⁵⁷ were found to be very reactive in neutral hydration, a result attributable to the cyclopentadienyl anion character of the transition states for water addition. A rate constant of  $1.6 \times 10^8 \text{ M}^{-1} \text{ s}^{-1}$  for addition of cyanide ion to **19** was also measured.³⁷



Flash photolysis of arylhydroxycyclopropenones 22 in aqueous solution formed ynolates 23 that were converted to arylketenes 24, whose rates of hydration were measured (equation 8).⁵⁸ The identity of the ketenes was confirmed by comparison with the species formed by photochemical Wolff rearrangements.⁵⁸

$$\begin{array}{c} 0\\ Ar\\ Ar\\ OH\\ \hline \\ -H_{2}O\\ -H_{4}\end{array} Ar \xrightarrow{-CH_{2}O} O^{-} \xrightarrow{H_{3}O} Ar\\ H \\ \end{array} \xrightarrow{-CH_{2}O} O^{-} \xrightarrow{H_{3}O} Ar\\ H \\ -CH \\ H \\ -CH \\ -H_{4}O\\ OH\\ CH \\ -H_{4}O\\ -H_{4}O\\ CH \\ -H_{4}O$$

Alkylketenes were significantly less reactive in hydration than arylketenes (Table 5.13), with a rate ratio  $k_{rel}(PhCH=C=O/n-BuCH=C=O) = 500$ . Di-*tert*-butylketene (**26**) was less reactive than *t*-BuCH=C=O by a factor of 10⁵, a result attributed to steric hindrance on both sides of the carbonyl carbon to in-plane attack of water (equation 9).³¹

$$\overset{t-\mathrm{Bu}}{\underset{t-\mathrm{Bu}}{\longrightarrow}} C = O \xrightarrow[t-\mathrm{Bu}]{} \overset{t-\mathrm{Bu}}{\underset{t-\mathrm{Bu}}{\longrightarrow}} \overset{O}{\underset{OH}{\longrightarrow}} (9)$$

A collection of rate constants has been reported for the neutral hydration of a number of alkylketenes obtained from flash photolysis^{59,60} and dialkylketenes measured by stop-flow methods.⁶¹ These results were plausibly interpreted in terms of in-plane attack of water on the ketenes, but certain unusual features of the data, including the report of a rate constant measured by stop-flow for *t*-Bu₂C=C=O in 10% H₂O/CH₃CN with a half-life greater than 1 year suggest that confirmation of these results would be desirable.

The solvent isotope effects of the hydroxide ion–catalyzed hydration of a series of ketenes are all weakly inverse  $(k_{OH}^{-}/k_{OD}^{-} = 0.76-0.97)$ , and this was suggested as a mechanistic criterion for this reaction.⁶² There was a linerar correlation between  $\log k(H_2O)$  and  $\log k(OH^{-})$  for ketene hydration, which was also suggested to be diagnostic for ketene hydration.⁶³ A deviation from the correlation was noticed for PhC(OH)=C=O (11), suggesting that the mechanistic assignment for this reaction required revision.⁶³

The hydration of 2- and 4-pyridylketenes (2- and 4-27) generated by photochemical Wolff rearrangement, forming pyridylacetic acids **29**, have been found to involve intermediate carboxymethylenedihydropyridines 2- and 4-**28**, with a strong intramolecular hydrogen bond in 2-**28** (equations 10, 11).⁴⁶ 3-Pyridylketene (3-**27**) reacted with formation of the intermediate acid enol 3-**29** (equation 12).⁴⁶



The rate of hydration of carbon suboxide (O=C=C=C=O, **31**) forming malonic acid (**32**) has been measured (equation 13), and the reaction was also studied using *ab initio* calculations.⁶⁴ Complexes of C₃O₂ with H₂O (**33**) and NH₃ (**34**) in argon matrices were studied by FTIR spectroscopy, and calculations of the structures of these pre-reactive complexes suggested that they could lead to initial 1,4-addition.⁶⁵ A new theoretical study of the hydration of C₃O₂ has appeared,⁶⁶ and computed values for the proton affinity of 765 to 801 kJ/mol^{64,66} are in reasonable agreement with the experimental value of 791 kJ/mol.⁶⁷

Further studies of ketene hydration include those of acetylketene (Section 4.1.6, ref. 11),  $\alpha$ -oxoketenes (Section 4.1.10), and disilanylketenes (Section 4.5).

## **REFERENCES FOR SECTION 5.5.1**

- 1. Bell, R. P. Adv. Phys. Org. Chem. 1966, 3, 1-29.
- 2. Wolfe, S.; Kim, C.-K.; Yang, K.; Weinberg, N.; Shi, Z. J. Am. Chem. Soc. 1995, 117, 4240–4260.
- 3. Guthrie, J. P.; Pitchko, V. J. Am. Chem. Soc. 2000, 122, 5520-5528.
- 4. Guthrie, J. P. J. Am. Chem. Soc. 2000, 122, 5529-5538.
- 5. Staudinger, H.; Meyer, P. Helv. Chem. Acta 1922, 5, 656-678.
- Lacey, R. N. The Chemistry of Alkenes; Patai, S., Ed.; Wiley, New York, 1964, pp. 1161– 1227.
- 7. Satchell, D. P. N.; Satchell, R. S. Chem. Soc. Rev. 1975, 4, 231-250.
- Donohoe, G.; Satchell, D. P. N.; Satchell, R. S. J. Chem. Soc. Perkin Trans. 2 1990, 1671–1674.
- 9. Poon, N. L; Satchell, D. P. N. J. Chem. Soc. Perkin Trans. 2 1984, 1083-1087.
- 10. Poon, N. L; Satchell, D. P. N. J. Chem. Soc. Perkin Trans. 2 1985, 1551-1554.
- 11. Poon, N. L; Satchell, D. P. N. J. Chem. Soc. Perkin Trans. 2 1983, 1381-1383.
- 12. Poon, N. L; Satchell, D. P. N. J. Chem. Soc. Perkin Trans. 2 1986, 1485-1490.
- 13. Satchell, D. P. N.; Satchell, M. J. F. Z. Naturforsch. B 1991, 46, 391-392.
- 14. Lillford, P. J.; Satchell, D. P. N. J. Chem. Soc. B 1968, 889-897.
- 15. Seikaly, H. R.; Tidwell, T. T. Tetrahedron 1986, 42, 2587-2613.
- Bothe, E.; Meier, H.; Schulte-Frohlinde, D.; von Sonntag, C. Angew. Chem. Int. Ed. Engl. 1976, 15, 380.
- 17. Bothe, E.; Dessouki, A. M.; Schulte-Frohlinde, D. J. Phys. Chem. 1980, 84, 3270-3272.
- 18. Nguyen, M. T.; Hegarty, A. F. J. Am. Chem. Soc. 1984, 106, 1552-1557.
- 19. Skancke, P. N. J. Phys. Chem. 1992, 96, 8065-8069.
- 20. Duan, X.; Page, M. J. Am. Chem. Soc. 1995, 117, 5114-5119.
- Nguyen, M. T.; Sengupta, D.; Raspoet, G.; Vanquickenborne, L. G. J. Phys. Chem. 1995, 9, 11883–11888.
- 22. Cannizzaro, C. E.; Houk, K. N. J. Am. Chem. Soc. 2004, 126, 10992-11008.
- 23. Nguyen, M. T.; Raspoet, G. Can. J. Chem. 1999, 77, 817-829.
- 24. Guthrie, J. P. Can. J. Chem. 1999, 77, 934-942.
- 25. Sánchez-Andrada, P.; Alkorta, I.; Elguero, J. J. Mol. Struct. 2001, 544, 5-23.
- 26. Houk, K. N.; Storzier, R. W.; Hall, J. A. Tetrahedron Lett. 1974, 897-900.
- 27. Cossio, F. P.; Ugalde, J. M.; Lopez, X.; Lecea, B.; Palomo, C. J. Am. Chem. Soc. 1993, 115, 995–1004.
- 28. Blake, P. G.; Davies, H. H. J. Chem. Soc. Perkin Trans. 2 1972, 321-323.
- 29. Butkovskaya, N. I.; Manke, G., II.; Setser, D. W. J. Phys. Chem. 1995, 99, 11115-11121.
- 30. Kabir, H. K.; Seikaly, H. R.; Tidwell, T. T. J. Am. Chem. Soc. 1979, 101, 1059-1060.

- 31. Allen, A. D.; Tidwell, T. T. J. Am. Chem. Soc. 1987, 109, 2774-2781.
- 32. Allen, A. D.; Stevenson, A.; Tidwell, T. T. J. Org. Chem. 1989, 54, 2843-2848.
- 33. Allen, A. D.; Tidwell, T. T. Tetrahedron Lett. 1991, 32, 847-850.
- 34. Allen, A. D.; Baigrie, L. M.; Gong, L.; Tidwell, T. T. Can. J. Chem. 1991, 69, 138-145.
- 35. Allen, A. D.; McAllister, M. A.; Tidwell, T. T. Tetrahedron Lett. 1993, 34, 1095-1098.
- Allen, A. D.; Kresge, A. J.; Schepp, N. P.; Tidwell, T. T. Can. J. Chem. 1987, 65, 1719– 1723.
- 37. Urwyler, B.; Wirz, J. Angew. Chem. Int. Ed. Engl. 1990, 29, 790-791.
- 38. Andraos, J.; Kresge, A. J. J. Photochem. Photobiol. A: Chem. 1991, 57, 165-173.
- Allen, A. D.; Andraos, J.; Kresge, A. J.; McAllister, M. A.; Tidwell, T. T. J. Am. Chem. Soc. 1992, 114, 1878–1879.
- 40. Chiang, Y.; Kresge, A. J.; Popik, V. V. J. Chem. Soc. Perkin. Trans. 2 1999, 1107-1109.
- 41. Chiang, Y.; Kresge, A. J.; Popik, V. V.; Schepp, N. P. J. Am. Chem. Soc. 1997, 119, 10203–10212.
- 42. Chiang, H.; Jefferson, E. A.; Kresge, A. J.; Popik, V. V. J. Am. Chem. Soc. 1999, 121, 11330–11335.
- 43. Bakulev, V. A.; Chiang, H.; Kresge, A. J.; Meng, Q.; Morzherin, Y. Y.; Popik, V. V. J. Am. Chem. Soc. 2001, 123, 2681–2682.
- 44. Andraos, J.; Chiang, Y.; Kresge, A. J.; Pojarlieff, I. G.; Schepp, N. P.; Wirz, J. J. Am. Chem. Soc. 1994, 116, 73–81.
- 45. Chiang, Y.; Federov, A. V.; Kresge, A. J.; Onyido, I.; Tidwell, T. T. J. Am. Chem. Soc. **2004**, *126*, 9382–9386.
- 46. Allen, A. D.; Federov, A. V.; Tidwell, T. T.; Vukovic, S. J. Am. Chem. Soc. 2004, 126, 15777–15783.
- 47. Frey, J.; Rappoport, Z. J. Am. Chem. Soc. 1995, 117, 1161-1162.
- 48. Frey, J.; Rappoport, Z. J. Am. Chem. Soc. 1996, 118, 5169-5181.
- 49. Frey, J.; Rappoport, Z. J. Am. Chem. Soc. 1996, 118, 5182-5191.
- Allen, B. M.; Hegarty, A. F.; O'Neill, P.; Nguyen, M. T. J. Chem. Soc. Perkin. Trans. 2 1992, 927–934.
- 51. Chiang, Y.; Kresge, A. J.; Meng, Q.; Morita, Y.; Yamamoto, Y. J. Am. Chem. Soc. 1999, 121, 8345–8351.
- 52. Andraos, J.; Kresge, A. J.; Schepp, N. Can. J. Chem. 1995, 73, 539-543.
- 53. Barra, M.; Fisher, T. A.; Cernigliaro, G. J.; Sinta, R.; Scaiano, J. C. J. Am. Chem. Soc. 1992, 114, 2630–2634.
- 54. Andraos, J.; Chiang, Y.; Huang, C.-G.; Kresge, A. J.; Scaiano, J. C. *J. Am. Chem. Soc.* **1993**, *115*, 10605–10610.
- 55. Almstead, J.-I. K.; Urwyler, B.; Wirz. J. J. Am. Chem. Soc. 1994, 116, 954–960.
- 56. Andraos, J.; Kresge, A. J.; Popik, V. V. J. Am. Chem. Soc. 1994, 116, 961-967.
- 57. Andraos, J.; Chiang, Y.; Kresge, A. J.; Popik, V. V. J. Am. Chem. Soc. 1997, 119, 8417– 8424.
- 58. Chiang, Y.; Kresge, A. J.; Popik, V. V. J. Am. Chem. Soc. 1995, 117, 9165-9171.
- 59. Lyashchuk, S. N.; Skrypnik, Yu. G. Russ. J. Org. Chem. 1994, 30, 10-17.
- 60. Lyashchuk. S. N.; Skrypnik, Yu. G. React. Kinet. Catal. Lett. 1994, 52, 399-404.

- 61. Lyashchuk, S. N.; Skrypnik, Yu. G. React. Kinet. Catal. Lett. 1994, 52, 453-459.
- 62. Andraos, J.; Chiang, Y.; Eustace, S. J.; Kresge, A. J.; Paine, S. W.; Popik, V. V.; Sung, K. *Can. J. Chem.* **1999**, *77*, 459–462.
- 63. Andraos, J.; Kresge, A. J. Can. J. Chem. 2000, 78, 508-515.
- 64. Allen, A. D.; McAllister, M. A.; Tidwell, T. T. J. Chem. Soc. Chem. Commun. 1995, 2547–2548.
- Couturier-Tamburelli, I.; Chiavassa, T.; Aycard, J.-P. J. Am. Chem. Soc. 1999, 121, 3756– 3761.
- 66. Nguyen, M. T.; Raspoet, G.; Vanquickenborne, L. G. J. Phys. Org. Chem. 2000, 13, 46-56.
- Tortajada, J.; Provot, G.; Morizur, J.-P.; Gal, G.-F.; Maria, P.-C.; Flammang, R.; Govaert, Y. Int. J. Mass Spectrom. Ion Proc. 1995, 141, 241–255.

#### **ADDITIONAL REFERENCES FOR TABLE 5.13**

- 68. Liu, R.; Marra, R. M.; Tidwell, T. T. J. Org. Chem. 1996, 61, 6227-6232.
- 69. Sung, K.; Tidwell, T. T. Organometallics 1997, 16, 78-85.
- 70. Sung, K.; Tidwell, T. T. J. Am. Chem. Soc. 1996, 118, 2768-2769.
- 71. Jones, J., Jr.; Kresge, A. J. J. Org. Chem. 1992, 57, 6467-6469.
- 72. Chiang, Y.; Guo, H.-X.; Kresge, A. J.; Tee, O. S. J. Am. Chem. Soc. 1996, 118, 3386–3391.
- 73. Chiang, Y.; Kresge, A. J.; Nikolaev, V. A.; Popik, V. V. J. Am. Chem. Soc. **1997**, 119, 11183–11190.
- 74. Chiang, Y.; Kresge, A. J.; Popik, V. V. J. Am. Chem. Soc. 1999, 121, 5930-5932.
- 75. Chiang, Y.; Kresge, A. J.; Schepp, N. P.; Xie, R.-Q. J. Org. Chem. 2000, 65, 1175-1180.
- 76. Chiang, Y.; Kresge, A. J.; Zhan, H.-Q. Can. J. Chem. 2003, 81, 607-611.
- 77. Chiang, Y.; Kresge, A. J.; Nikolaev, V. A.; Popik, V. V. J. Am. Chem. Soc. 2003, 125, 6478–6484.
- 78. Liu, R. C.-Y.; Lusztyk, J.; McAllister, M. A.; Tidwell, T. T.; Wagner, B. D. J. Am. Chem. Soc. 1998, 120, 6247–6251.
- 79. Chiang, Y.; Kresge, A. J.; Pruszynski, P.; Schepp, N. P.; Wirz, J. Angew. Chem. Int. Ed. Engl. **1990**, 29, 792–794.
- Chiang, Y.; Gaplovsky, M.; Kresge, A. J.; Leung, K. H.; Ley, C.; Mac, M.; Persy, G.; Phillips, D. L.; Popik, V. V.; Rödig, C.; Wirz, J.; Zhu, Y. J. Am. Chem. Soc. 2003, 125, 12872–12880.
- 81. Allen, A. D.; Gong, L.; Tidwell, T. T. J. Am. Chem. Soc. 1990, 112, 6396-6397.
- 82. Zhao, D.-c.; Allen, A. D.; Tidwell, T. T. J. Am. Chem. Soc. 1993, 115, 10097-10103.

### 5.5.2 Mechanisms of Amination of Ketenes

Reactions of ketenes with amines are quite facile and have long been of preparative use (Section 5.5.3.3). Computational study of the reaction of ketene (1) revealed that reaction with two molecules of  $NH_3$  was favored over reaction with a single  $NH_3$  molecule and proceeded through a transition state **2a**, followed by addition to the C=O bond leading to a complexed amide enol **2b** (equation 1).^{1,2} Transition

		$k_{H}^{+}$ ,		k _{он} ,	k _{H2} C	$rac{k_{ m H}^+}{k_{ m H_2O}}$	$\frac{k_{OH}^-}{k_{H_2O}}$	
Ketene	$\% H_2O$	$M^{-1}s^{-1}$	$k_{\rm H_2O},s^{-1}$	$M^{-1}s^{-1}$	$k_{D_2O}$	$(M^{-1})$	$(M^{-1})$	Ref.
t-BuCH=C=O	20%	135	0.0873	$1.7 \times 10$	3			31
$Ph_2C=C=O$	20%		4.17					50
$Mes_2C=C=O$	50%		0.162					50
$(Me_5C_6)_2C=C=O$	50%		$4.26 \times 10^{-3}$	2				50
$Et_2C=C=O$	50%		0.357		1.59	$145^{b}$	790	31
c-Pr ₂ C=C=O	50%		0.406					34
c-PrCPh=C=O	20%		0.297					34
i-PrCPh=C=O	50%	0.120	$0.0424^{a}$	58.4	1.67	2.8	1400	32
t-BuCPh=C=O	50%	0.488	0.00997	46.7		49	4700	32
$t-Bu_2C=C=O$	50%	3.6						31
Me ₃ SiCH=C=O	50%	542	$3.23 \times 10^{-3}$	³ 70		$1.7 \times 10^{5}$	$2.1 \times 10^{2}$	4 33
$(Me_3SiC=C=O)_2$	50%		0.307					82
PhMe ₂ SiCH=C=O	50%		$5.40 \times 10^{-3}$	3				68
<i>t</i> -BuMe ₂ SiCH=C=O	50%		$9.58 \times 10^{-6}$	4				68
Me ₅ Si ₂ CH=C=O	50%		$9.31 \times 10^{-3}$	3				69
$(SiMe_2C=C=O)_2$	50%		$1.52 \times 10^{-2}$	2				69
(CH ₂ SiMe ₂ CH=C=O	)2 50%		$5.19 \times 10^{-3}$	3				69
$Me_2Si(CH=C=O)_2$	50%		$3.34 \times 10^{-1}$	2				70

TABLE 5.14 Hydration Rates of Ketenes in H₂O/CH₃CN at 25 °C

^{*a*}Extrapolated. ^{*b*}9.0 °C.

°5 °C.

^dRef. 7.

structure **3** was found for tautomerization of **2b** to the product acetamide. The pathway for reversal of the formation of the amide enol **2b** to reform ketene was lower in energy than conversion of **2b** to product, suggesting that the formation of **2b** could be reversible. An alternative pathway involving direct addition of NH₃ to the C=C bond forming acetamide directly was thermodynamically 22.6 kcal/mol more favorable than the formation of **2b** but had a barrier that was 12.8 kcal/mol higher than for formation of **2b**.

Early kinetic studies of ketene amination by Satchell et al.^{3–6} found that the rate of reaction of  $Ph_2C=C=O$  (4) with anilines in benzene and of dimethylketene in

ether was dependent on both [amine] and [amine]², and the results were interpreted as showing that the reaction occurred by an amine-assisted concerted addition to the C=C bond (equation 3). The reaction of  $Ph_2C=C=O$  (5) in H₂O with a variety of different amines followed by UV spectroscopy showed a much greater reactivity for the amines compared to reaction with H₂O but no systematic dependence of the reactivity on amine basicity.⁷ These results were interpreted as demonstrating rate-limiting addition of the amines to the ketene C=O bond forming an amide enol, with an attenuation of reactivity of the more basic amines due to hydrogen bonding to the H₂O (equation 4).⁷ For the ketenes Ar₂C=C=O (Ar = mesityl, Me₅C₆, and 2,4,6-*i*-Pr₃C₆H₂) amide enols have been observed by IR and UV as rather long-lived intermediates.^{2,8,9} The kinetics of the reactions were first order in [amine], indicating that attack of a single amine molecule was rate determining.²



Measurements of the kinetics of amination of a variety of ketenes **6** by TRIR spectroscopy in CH₃CN were carried out by flash photolysis of the diazo ketones **5**, generating the ketenes which were identified by their characteristic IR absorption near 2100 cm⁻¹.¹⁰ The kinetics of the reactions of the ketenes were monitored by the disappearance of the characteristic ketenyl IR absorption and showed a first-order dependence of the rates on [amine]. New transients were formed with the same rate constants as the ketene decay, and had IR absorption near 1750 cm⁻¹ that also decayed with a first-order dependence on [amine]. These second transient intermediates were assigned as having the structures either of amide enols **7** or of zwitterions **8** and decayed to amides **9**^{10–12} (equations 5, 6). The measured rate constants for amination are given in Table 5.15. The bimolecular rate constants  $k_2$  for the ketenes varied from  $4.6 \times 10^5$  M⁻¹ s⁻¹ for *t*-BuCH=C=O with Et₂NH to  $1.6 \times 10^9$  M⁻¹ s⁻¹ for 4-O₂NC₆H₄CH=C=O with piperidine and were approaching the diffusion controlled limit.

$$\begin{array}{c} O \\ R \\ \hline \end{array} \\ R \\ \hline \end{array} \\ \begin{array}{c} O \\ R \\ \hline \end{array} \\ \begin{array}{c} N_2 \\ CH_3CN \end{array} \\ \hline \end{array} \\ \begin{array}{c} R \\ \hline \end{array} \\ \begin{array}{c} C = O \\ CH_3CN \end{array} \\ \hline \end{array} \\ \begin{array}{c} R \\ \hline \end{array} \\ \begin{array}{c} C = O \\ CH_3CN \end{array} \\ \begin{array}{c} R \\ \hline \end{array} \\ \begin{array}{c} R \\ C = O \end{array} \\ \begin{array}{c} R^{1}NH_2 \\ CH_3CN \end{array} \\ \begin{array}{c} R \\ \hline \end{array} \\ \begin{array}{c} O \\ NHR \end{array} \\ \begin{array}{c} R \\ CH_3CN \end{array} \\ \begin{array}{c} O \\ NH_2R \end{array} \\ \begin{array}{c} R \\ CH_3CN \end{array} \\ \begin{array}{c} O \\ NH_2R \end{array} \\ \begin{array}{c} R \\ CH_3CN \end{array} \\ \begin{array}{c} O \\ NH_2R \end{array}$$
 (5)

Ketene	Et ₂ NH	$(CH_2)_5 NH^a$	<i>i</i> -PrNH ₂	<i>n</i> -BuNH ₂	BnNH ₂	Ref
t-BuCH=C=O	$4.6\times10^5$	$6.7  imes 10^6$	$7.7  imes 10^5$	$1.6  imes 10^6$	$7.5  imes 10^5$	10
PhMe ₂ SiCH=C=O	$4.2 \times 10^{4c}$			$2.3 \times 10^{5}$		15
PhCH=C=O	$4.8  imes 10^8$	$5.0  imes 10^8$	$1.2 \times 10^{8}$	$3.3 \times 10^{8}$	$1.4 \times 10^{8}$	10
$4-MeOC_6H_4CH=C=O$	$3.3 \times 10^{8}$	$5.1 \times 10^{8}$	$9.5 \times 10^{7}$	$2.2 \times 10^{8}$	$9.2 \times 10^{7}$	10
$4-NCC_6H_4CH=C=O$	$9.5  imes 10^{8}$	$1.4 \times 10^{9}$	$5.8  imes 10^8$	$8.7  imes 10^{8}$	$5.8 \times 10^{8}$	10
$4-O_2NC_6H_4CH=C=O$	$9.5  imes 10^{8}$	$1.6 \times 10^{9}$	$8.1 \times 10^{8}$	$1.1 \times 10^{9}$	$7.0 \times 10^{8}$	10
PhCH=CHCH=C=O	$4.9 \times 10^{8}$	$8.0  imes 10^8$	$1.6 \times 10^{8}$	$5.1 \times 10^{8}$	$1.7 \times 10^{8}$	10
PhC=CCH=C=O	$1.1 \times 10^{9}$					10
2-PyridylCH=C=O				$2.2 \times 10^{8}$		13
3-PyridylCH=C=O				$3.8 \times 10^{8}$		13
4-PyridylCH=C=O				$3.7 \times 10^{9}$		13
$Ph_2C=C=O$	$1.2 \times 10^{7}$	$1.5 \times 10^{7}$	$4.4 \times 10^{6}$	$1.1 \times 10^{8}$	$5.4 \times 10^{7}$	10
PhCOCPh=C=O	$2.1 \times 10^{7}$	$3.8 \times 10^{7}$	$9.4 \times 10^{6}$	$2.3 \times 10^{7}$	$1.1 \times 10^{7}$	10
C ⁻⁰ 0 21	$1.1 \times 10^{9}$					12
29	$7.6 \times 10^{7}$	1.8 × 10 ⁸			]	1, 21
25 C=0 28	$1.0 \times 10^{9}$	$1.9 \times 10^{9}$				21
	$1.1 \times 10^{9}$	$3.5  imes 10^9$				21
30						

TABLE 5.15 Bimolecular Rate Constants for the Reaction of Ketenes with Amines in CH_3CN at 23–25  $^\circ\text{C}$ 

^aPiperidine.

^bAverage value.

^cMorpholine.

Phenylketene (10a) and *n*-BuNH₂ reacted with formation of the amide enol 11a observed by IR, and this formed the amide 12a in an amine-catalyzed reaction (equation 7).¹⁰ 3- and 4-Pyridylketene (10b,c) reacted similarly, but 2-pyridylketene (10d) formed an intermediate that decayed with an amine-catalyzed rate

constant that was slower than those of **10a–c** by factors near 10⁷.¹³ This was determined to be the intramolecularly hydrogen-bonded amide 2-methylene-1,2-dihydropyridine **13** (equation 8) on the basis of the long, wavelength UV absorption similar to that of the model compound **14**,¹⁴ and not that of the amide enol **11d**,^{13,14} which would have a different UV spectrum. Calculations using DFT methods indicated the dihydropyridine **13** to be 3.5 kcal/mol more stable than the amide enol **11d**, which in turn was 17 kcal/mol more stable than the structure **11e** without the intramolecular hydrogen bond.¹⁴



Kinetic studies of the less reactive ketene PhMe₂SiCH=C=O (**15**) in CH₃CN showed that this followed the rate equation 9, with second- and third-order terms in [amine] and a very large dependence of rate on the amine basicity.¹⁵ Thus, for PhMe₂SiCH=C=O the rate ratio  $k_{obs}^{n-BuNH_2}/k_{obs}^{CF_3CH_2NH_2}$  at 0.02³ M amine was 10⁷ (p $K_{BH}^+$  values for *n*-BuNH₂and CF₃CH₂NH₂ are 10 and 5, respectively).¹⁵ The ketene *t*-BuC(Pr-*i*)=C=O (**16**) was very unreactive, and gave a rate constant for amination in neat *n*-BuNH₂.¹⁵

PhMe₂Si  

$$C=0$$
  
15  
 $t-Bu$   
 $i-Pr$   
 $C=0$   
 $i-Pr$   
(9)  
 $k_{obs} = a[amine]^2 + b[amine]^3$   
16

These kinetic results were interpreted in terms of a mechanism analogous to that found by the theoretical studies, with reversible formation of an amide enol from reaction of ketene with two amine molecules, followed by irreversible amide formation in steps catalyzed either by a third amine molecule (equation 10) or by a basic solvent (equation 11). The observation of only a first-order dependence on [amine] in earlier studies^{7,10} that used more reactive ketenes as substrates was attributed to irreversibility of the initial ketene-amine reaction, so that the subsequent steps involving additional amine molecules did not affect the rate of ketene

disappearance, although the amine-catalyzed conversion of long-lived but irreversibly formed intermediate was directly observed by TRIR spectroscopy.¹⁰

ketene + 2 amine 
$$\longrightarrow$$
 ketene • 2 amine  $\longrightarrow$  product (10)  
ketene • 2 amine  $\longrightarrow$  product (11)

The amination of dienylketene **18** formed by photolysis of cyclohexadienone **17** showed the consecutive formation of two distinct intermediates during the conversion of the ketene to the amide, as observed by UV, and this was interpreted as formation of an initial zwitterion **19**, which then formed an amide enol **20** (equation 12).¹⁶ Long-lived zwitterionic species have been observed to form from ketenes and tertiary amines (vide infra), although in calculations of gas-phase structures, such zwitterions with hydrogens on the nitrogen atoms are not energy minima and only amide enol structures have been found as minima.¹ In studies by IR only one intermediate between the ketene and the amide has been detected, and whether or not zwitterions are involved as observable intermediates in aminations with primary and secondary amines is a subtle and unanswered question.



Amination of ketene **21** with Et₂NH gave a rate constant in CH₃CN at 25 °C of  $k_2 = 1.2 \times 10^9 \text{ M}^{-1} \text{ s}^{-1}$  or near the diffusion limit.¹² By analogy with theoretical studies of the reaction of **5** with H₂O and of O=CHCH=C=O with NH₃, this reaction may proceed through the cyclic transition state **22** and lead to a stable aromatic product. However, the ketene **30** (Table 5.15) has a comparably large amination rate constant²¹ but would lead to an antiaromatic product by this mechanism.



The 1,2-bisketene **23** reacted rapidly with amines, with high selectivity for initial formation of isolable ketenylamides **24**, which reacted with a second amine, in a step slower than the first by factors of nearly  $10^4$ , to form the diasteriomeric succinamides **25** (equation 14).¹⁹ The kinetics of the first step were also fit by equation 10, with terms in [amine]² and [amine]³. Reaction of the ketenylamide **24** with a different second amine gave mixed diamides, while reaction of **24** with MeOH gave a mixed ester amide. The latter reaction showed evidence for a rate enhancement and selectivity in formation of the diastereomeric product, indicating assistance by the amido function in enhancing the attack by MeOH.¹⁹ For *n*-BuNH₂, addition to

24 (R = n-Bu) showed the absence of a higher-order term in n-BuNH₂, also suggesting assistance by the carboxamido group.¹⁹



Tertiary amines such as pyridine reacted with ketenes including pentafulvenone (26), generated by photolysis of the diazo ketone in a matrix, forming zwitterionic species 27 that was directly observed by its IR absorption at 1683 cm⁻¹ (equation 15).^{18,20} Ketenes 28 and 29 gave similar behavior.²¹ The kinetics of the reaction of ketenes with tertiary amines have also been measured, ^{10,11} with rate constants that are lower than those of comparable secondary amines by factors near  $10^{4}$ .^{10,11} These lower rate constants may reflect the absence of hydrogen-bonding interactions, as depicted in equation 1 for the tertiary amines, and also steric barriers with the more substituted nucleophiles. The kinetics of the reactions of 28–30 with both secondary and tertiary amines have been measured using laser flash photolysis with UV detection.²¹



Comparative reactivity data for different ketenes with 0.1 M *n*-BuNH₂ are summarized in Table 5.16. In a previous consideration of the reactivity of  $Ph_2C=C=O$  compared to  $Me_2C=C=O$ , it was suggested that the greater reactivity of the former arises from the ability of the aryl groups to stabilize developing negative charge in the (TS) for amination. This view has been criticized,² as the transition states leading to amide enols were evidently felt not to possess much charge separation. However, there is a difference in rate constants by a factor of 17 for 4-pyridylketene compared to 4-anisylketene, and PhCH=C=O is 200 times more reactive than *t*-BuCH=C=O (Table 5.16). These differences provide strong evidence for the importance of conjugative stabilization in the transition state for amination.

	/		
t-BuCH=C=O PhCH=C=O 4-MeOC ₆ H ₄ CH=C=O 4-O ₂ NC ₆ H ₄ CH=C=O PhC=CHCH=C=O 2-PyridylCH=C=O 4-PyridylCH=C=O	$\begin{array}{c} k \ ({\rm s}^{-1}) \\ 1.6 \times 10^5 \\ 3.3 \times 10^7 \\ 2.2 \times 10^7 \\ 1.1 \times 10^8 \\ 1.1 \times 10^{8b} \\ 2.2 \times 10^7 \\ 3.7 \times 10^8 \end{array}$	PhMe ₂ SiCH=C=O t-BuC(Pr- $i$ )=C=O (Me ₃ SiC=C=O) ₂ (7) NCC ₆ H ₄ CH=C=O Ph ₂ C=C=O 3-PyridylCH=C=O	$k (s^{-1})  6.06 \times 10^{3a}  4.57 \times 10^{-3a}  4.1 \times 10^{4b}  8.7 \times 10^7  1.1 \times 10^7  3.8 \times 10^7$
PhCH=CHCH=C=O	$54.1  imes 10^7$	Me ₃ Si C Me ₃ Si CONHBu- <i>n</i>	0.211 ^{<i>a</i>}
$\mathbf{C}^{\mathbf{C}^{\mathbf{D}^{0}}}$	$1.1  imes 10^{8b}$	C=0	$7.6 \times 10^{6c}$
28	$1.0 \times 10^{8}$	29 C 0 30	$1.1 \times 10^{8c}$

TABLE 5.16 Comparative Rate Constants (s⁻¹) for Reactions of Ketenes in CH₃CN with 0.1 M *n*-BuNH₂ (ca. 25 °C)^{*d*}

^{*a*}Ref. 15. ^{*b*}Ref. 10. ^{*c*}Et₂NH, ref. 21. ^{*d*}Data from Table 5.15

Ketene **29** was generated together with imidazole and a phenolic resin by photolysis of the diazo ketone on a thin film at 10 K and observed by the IR absorption at  $2110 \text{ cm}^{-1}$ .²² Upon warming to 140 K, an absorption at 1635 cm⁻¹ grew in and was assigned to the ylide **31**. At 175 K this began to be replaced by an absorption at 1750 cm⁻¹ assigned to the ester **32** (equation 16).²² In the absence of imidazole, reaction of the ketene with the phenolic resin was observed only at 250 K.²²



Molecular complexes of  $C_3O_2$  (26) with NH₃, (CH₃)₂NH, (CH₃)₃N, and 4-(dimethylamino)pyridine were observed, and were converted at 40 K to zwitterionic products.²³ Computations for the reaction with (CH₃)₂NH showed formation of the complex 29 with a barrier of 13.0 kJ/mol that formed the ketene 30 in a step that was exothermic by 83.6 kJ/mol and led to tetramethyl malonamide (31, equation 17).²³



# **REFERENCES FOR SECTION 5.5.2**

- 1. Sung, K.; Tidwell, T. T. J. Am. Chem. Soc. 1998, 120, 3043-3048.
- Raspoet, G.; Nguyen, M. T.; Kelly, S.; Hegarty, A. F. J. Org. Chem. 1998, 63, 9669– 9677.
- 3. Briody, J. M.; Satchell, D. P. N. Tetrahedron 1966, 22, 2649-2653.
- 4. Lillford, P. J.; Satchell, D. P. N. J. Chem. Soc. B 1967, 360-365.
- 5. Lillford, P. J.; Satchell, D. P. N. J. Chem. Soc. B 1968, 54-57.
- 6. Lillford, P. J.; Statchell, D. P. N. J. Chem. Soc. B 1970, 1016-1019.
- 7. Andraos, J.; Kresge, A. J. J. Am. Chem. Soc. 1992, 114, 5643-5646.
- 8. Frey, J.; Rappoport, Z. J. Am. Chem. Soc. 1996, 118, 3994-3995.
- 9. Rappoport, Z.; Frey, J.; Sigalov, M.; Rochlin, E. Pure Appl. Chem. 1997, 69, 1933-1940.
- Wagner, B. D.; Arnold, B. R.; Brown, G. S.; Lusztyk, J. J. Am. Chem. Soc. 1998, 120, 1827–1834.
- de Lucas, N. C.; Netto-Ferreira, J. C.; Andraos, J.; Lusztyk, J.; Wagner, B. D.; Scaiano, J. C. *Tetrahedron Lett.* **1997**, *38*, 5147–5150.
- Liu, R. C.-Y.; Lusztyk, J.; McAllister, M. A.; Tidwell, T. T.; Wagner, B. D. J. Am. Chem. Soc. 1998, 120, 6247–6251.
- Acton, A.; Allen, A. D.; Antunes, L.; Federov, A. V.; Najafian, K.; Tidwell, T. T.; Wagner, B. D. J. Am. Chem. Soc. 2002, 124, 13790–13794.
- Allen, A. D.; Fedorov, A. V.; Tidwell, T. T.; Vukovic, S. J. Am. Chem. Soc. 2004, 126, 15777–15783.
- 15. Allen, A. D.; Tidwell, T. T. J. Org. Chem. 1999, 64, 266-271.
- Quinkert, G.; Scherer, S.; Reichert, D.; Nestler, H.-P.; Wennemers, H.; Ebel, A.; Urbahns, K.; Wagner, K.; Michaelis, K.-P.; Wiech, G.; Prescher, G.; Bronstert, B.; Freitag, B.-J.; Wicke, I.; Lisch, D.; Belik, P.; Crecelius, T.; Hörstermann, D.; Zimmermann, G.; Bats, J. W.; Dürner, G.; Rehm, D. *Helv. Chim. Acta* 1997, *80*, 1683–1772.
- 17. Birney, D. M.; Xu, X.; Ham, S.; Huang, X. J. Org. Chem. 1997, 62, 7114-7120.
- 18. Qiao, G. G.; Andraos, J.; Wentrup, C. J. Am. Chem. Soc. 1996, 118, 5634-5638.

- Allen, A. D.; Moore, P. A.; Missiha, S.; Tidwell, T. T. J. Org. Chem. 1999 64, 4690– 4696.
- 20. Visser, P.; Zuhse, R.; Wong, M. W.; Wentrup, C. J. Am. Chem. Soc. 1996, 118, 12598–12602.
- 21. De Lucas, N. C.; Netto-Ferreira, J. C.; Andraos, J.; Scaiano, J. C. J. Org. Chem. 2001, 66, 5016–5021.
- 22. Pacansky, J.; Chang, J. S.; Brown, D. W.; Schwarz, W. J. Org. Chem. 1982, 47, 2233–2234.
- Sessouma, B.; Couturier-Tamburelli, I.; Monnier, M.; Wong, M. W.; Wentrup, C.; Aycard, J. P. J. Phys. Chem. A 2002, 106, 4489–4497.

#### 5.5.3 Nucleophilic Addition to Ketenes: Preparative Aspects

**5.5.3.1** Hydride Reduction Ketenes have been found to be reduced by hydrides, although this reaction has been studied only occasionally. The reaction of bis(2,4,6-triisopropylphenyl)ketene ("ditipylketene," 1) with LiAlH₄ gave the stable enol 2 (equation 1).¹ Reaction of diphenylketene (3) with LiAlH₄ formed the enolate 4, which was acetylated with acetyl chloride, forming the vinyl ester 5 (equation 2).² Reduction with LiBEt₃H of the ketene 7 formed in situ from 6 gave an enolate that was alkylated and reduced further to 8 (equation 3).^{3,4}

$$\begin{array}{c} \text{Tip} \\ \xrightarrow{\text{Tip}} C = O \end{array} \xrightarrow{1) \text{ LiAlH}_4} \begin{array}{c} \text{Tip} \\ \xrightarrow{2) \text{ H}_2 O} \end{array} \xrightarrow{\text{Tip}} H \end{array}$$
(1)



The reduction of ketenes by silanes,⁵ germanes,⁶ and stannanes⁷ gave silyl, germyl, and stannyl enol ethers, respectively (equations 4–6). Hydrides of platinum and cobalt gave similar reactions.⁷

$$Ph \xrightarrow{Ph} C=O +R_3^{1}SiH \xrightarrow{H_2PtCl_6} Ph \xrightarrow{OSiR_3^{1}} (4)$$
9 (R = Et, Ph, *n*-Pr, *n*-Bu) 10 (80%, R = Et, *E/Z* = 48:52)



*tert*-Butyllithium and Grignard reagents reacted with crowded aliphatic ketenes **15** both by hydride donation forming aldehydes **16** and by alkyl group transfer forming ketones **17** (equation 7).^{8,9} *tert*-Butyl and isopropyl Grignard reagents reacted with ketenes **18** containing bulky aryl groups by hydride transfer, giving stable enols **19** after hydrolysis (equation 8).^{10,11}



Intramolecular hydride transfer occurred upon thermolysis of ketene **20**, forming **21** (equation 9).¹² Other hydride reactions of ketenes are shown in Section 3.4.6.¹³



# **REFERENCES FOR SECTION 5.5.3.1**

- 1. Frey, J.; Rappoport, Z. J. Org. Chem. 1997, 62, 8372-8386.
- 2. Micovic, V.; Rogic, M. M.; Mihailovic, M. L. Tetrahedron 1957, 1, 340-342.
- 3. Tomioka, K.; Shindo, M.; Koga, K. Tetrahedron Lett. 1990, 31, 1739-1740.
- 4. Tomioka, K.; Shindo, M.; Koga, K. J. Org. Chem. 1990, 55, 2276-2277.
- 5. Frainnet, E.; Caussé, J. Bull. Soc. Chim. Fr. 1968, 3034.
- 6. Satgé, J.; Riviere, P. J. Organomet. Chem. 1969, 16, 71-82.

- 7. Hneihen, A. S.; Bruno, J. W.; Huffman, J. C. J. Organomet. Chem. 1990, 382, 361-373.
- 8. Allen, A. D.; Baigrie, L. M.; Gong, L.; Tidwell, T. T. Can. J. Chem. 1991, 69, 138-145.
- Baigrie, L. M.; Lenoir, D.; Seikaly, H. R.; Tidwell, T. T. J. Org. Chem. 1985, 50, 2105– 2109.
- 10. Nugiel, D. A.; Rappoport, Z. J. Am. Chem. Soc. 1985, 107, 3669-3676.
- 11. Fuson, R. C.; Foster, R. E.; Shenk, W. J., Jr.; Maynert, E. W. J. Am. Chem. Soc. **1945**, 67, 1937–1939.
- 12. Wuest, J. D.; Madonik, A. M.; Gordon, D. C. J. Org. Chem. 1977, 42, 2111-2113.
- 13. Naef, F.; Decorzant, R. Tetrahedron 1986, 42, 3245-3250.

**5.5.3.2 Reaction of Oxygen Nucleophiles with Ketenes** Addition of oxygen nucleophiles is a characteristic reaction of ketenes. It plays a major role in the Arndt-Eistert chain elongation of carboxylic acids (Section 3.3) and has been extensively utilized in stereoselective additions to ketenes by either chiral alcohols or achiral alcohols with chiral catalysts (Section 5.9). The mechanism of ketene hydration has been actively pursued (Section 5.5.1).

Tricyclic pyridine bases catalyzed the addition of alcohols to ketenes, and enhanced the selectivity for acylation of primary alcohols in comparison to secondary alcohols^{1–3} and for selective acylation of polyols.⁴ The selectivities of ketenes generated by Wolff rearrangements for polyfunctional nucleophiles are noted in Section 3.3.1.

Ketenes 2 bearing conjugatively electron-withdrawing substituents generated from the reaction of carboxylic acids 1 with carbodiimides gave facile acylation of crowded alcohols (equation 1).⁵ It was proposed that  $EtO_2CCH=C=O$  (2a),  $(EtO)_2(O)PCH=C=O$  (2b), and  $EtO_2CC(Me)=C=O$  (2e) reacted through a pseudopericyclic transition state, as in 4, which was insensitive to the steric bulk of the alcohols.⁵

$$\begin{array}{c} R \\ \hline CO_2H \\ \hline DCC \\ \hline 1 \\ 2 \\ R = EtO_2C (a); (EtO_2O(P) (b), \\ CN (c), 4-TolSO_2 (d) \end{array} \right) \begin{array}{c} O^{--H} \\ R \\ \hline CO_2Bu-t \\ O \\ \hline O \\ O \\ (1) \end{array}$$

Reaction of cyanoketenes 5 with dimethyloxirane (6) gave allyl esters 8, which were proposed to form by an ene-like reaction, (equation 2).⁶



Intramolecular additions of hydroxyl groups to ketenes have provided routes to a variety of lactones. Thus ring opening of dichlorocyclobutenone 9 formed vinyl-ketene 10, which cyclized to lactone 11 (equation 3).⁷ Meldrum's acid derivative 12 formed ketene 13 at 400–500 °C, and this cyclized to 14 (equation 4).⁸



Ring opening of cyclobutenone **15** gave the vinylketene **16**, which cyclized to the lactone **17** (equation 5).⁹ 4-Hydroxycyclobutenone **18** obtained from the corresponding cyclobutenedione by reduction with NaBH₄ was refluxed in *p*-xylene with 1.1 eq of CF₃CO₂H to form the intermediate ketene *E*-**19**, which cyclized to the lactone **20** (equation 6).¹⁰ The acid catalyst increased the rate of reaction, which was proposed to be slowed by the favored but reversible ring opening of **18** to the *Z*-alkenylketene *Z*-**19**.¹⁰ As noted in Section 3.4.1, electron donor substituents such as OH in **18** favor outward rotation of the substituent forming *Z*-**19**.¹¹



Thermolysis of the dioxinone **21** formed the acylketene **22**, which gave the lactone **23** (equation 7).¹² Photolysis of cyclobutanone **24** formed the ketene **25**, which cyclized to the lactone **26** (equation 8).¹³ Photolysis of the chromium carbene complex **27** proceeded through the ketene **28** to lactone **29** (equation 9).¹⁴



Arylketenes **30** reacted with enolates **31** with a kinetic preference for attack of the enolate oxygen, forming **32**, but on prolonged reaction these were then converted to the thermodynamically more stable products **33** of C-acylation which on hydration gave **34** and **35**, respectively (equations 10, 11).^{15,16}



Thermolysis of **36** took place with decarbonylation to form the ketene **37**, which cyclized to **38** (equation 12).¹⁷ Wolff rearrangement of diazo ketone **39** formed ketene **40**, which reacted with methanol or water, forming the enol **41**, which underwent protonation with a preference for forming **42** (equation 13).¹⁸ The preference

for proton delivery from the more hindered side with MeOH and  $H_2O$  was attributed to the smaller reagents assisting by hydrogen bonding to the pyridyl nitrogen.¹⁸



Dichloroketene **43** formed in situ by zinc dechlorination reacted with the allylic ether **44**, forming a zwitterionic intermediate that underwent [3,3] sigmatropic rearrangement to give the lactone **45** (equation 14).¹⁹ Such addition of allylic ethers to ketenes forming zwitterionic species that rearranged by [3.3] sigmatropic rearrangement have been reviewed, and are referred to as the Bellus-Claisen or Malherbe-Bellus rearrangement.²⁰ Optically active sulfoxides **46** reacted similarly to give butyrolactones **47** in high enantiomeric purity (equation 15).²¹



Diazine **48** reacted with styrene by a [4+2] cycloaddition followed by loss of nitrogen to give ketene **49**, and this was captured by methanol in a Dieckmann-like reaction giving **50** (equation 16).²² Photolysis of **51** was proposed to give the ketene **52**, which added methanol with cyclization, forming **53** (equation 17).²³



Dichloroketene reacted with the epoxide **54** via the zwitterion **55**, forming the caprolactone **56** (equation 18).²⁴ Photolysis of **57** formed the ketene **58**, identified by the IR band at 2120 cm⁻¹, and this cyclized to **59** without incorporation of deuterium from MeOD (equation 19).²⁵ The oxiranylketene **61** was similarly formed by photolysis of **60** and could be captured by MeOH, but in benzene gave the cyclization product **62** (equation 20).²⁶



As noted in Section 3.4.2, photolysis of hydroxycyclopentenone **63** gave lactone **65** in low yield, and this was proposed to form from (hydroxycyclopropyl)ketene **64** (equation 21).²⁷ The ketene could not be trapped by CH₃OH, and it was proposed that the cyclization rate was faster than that of solvent attack.



The reactivity of ketenes **66** with MeOH (1 M in ether) was decreased by more than a factor of  $10^3$  for R = Me compared to R = H, which demonstrated steric retardation of in-plane by the oxygen nucleophile (equation 22).²⁸



## **REFERENCES FOR SECTION 5.5.3.2**

- 1. Schyja, W.; Petersen, S.; Lüning, U. Liebigs Ann. 1996, 2099-2105.
- Lüning, U.; Petersen, S.; Schyja, W.; Hacker, W.; Marquardt, T.; Wagner, K.; Bolte, M. *Eur. J. Org. Chem.* 1998, 1077–1084.
- 3. Petersen, S.; Lüning, U. Eur. J. Org. Chem. 1999, 847-854.
- 4. Lüning, U.; Hacker, W. J. Prakt. Chem. 1999, 341, 662-667.
- 5. Shelkov, R.; Nahmany, M.; Melman, A. J. Org. Chem. 2002, 67, 8975-8982.
- 6. Danila, M.-G.; Gottlieb, H.; Bartos, A.; Filip, P. Rev. Roum. Chim. 1994, 39, 1223-1228.
- 7. Dillon, J. L.; Gao, Q.; Dillon, E. A.; Adams, N. Tetrahedron Lett. 1997, 38, 2231-2234.
- 8. Jourdain, F.; Pommelet, J. C. Tetrahedron Lett. 1994, 35, 1545-1548.
- 9. Naidorf-Meir, S.; Hassner, A. J. Org. Chem. 1992, 57, 5102-5105.
- 10. Wang, J.; Jiang, X.; Chen, M.; Ge, Z.; Hu, Y.; Hu, H. J. Chem. Soc. Perkin Trans. 1 2001, 66–71.
- 11. Niwayama, S.; Kallel, E. A.; Sheu, C.; Houk, K. N. J. Org. Chem. 1996, 61, 2517-2522.
- 12. Boeckman, R. K., Jr.; Pruitt, J. R. J. Am. Chem. Soc. 1989, 111, 8286-8288.
- Rahman, S. S.; Wakefield, B. J.; Roberts, S. M.; Dowle, M. D. J. Chem. Soc. Chem. Commun. 1989, 303–304.
- 14. Vernier, J.-M.; Hegedus, L. S.; Miller, D. B. J. Org. Chem. 1992, 57, 6914-6920.
- 15. Gong, L.; Leung-Toung, R.; Tidwell, T. T. J. Org. Chem. 1990, 55, 3534-3539.
- 16. Leung-Toung, R.; Tidwell, T. T. J. Am. Chem. Soc. 1990, 112, 1042-1048.
- Maslivets, A. N.; Golovnina, O. V.; Krasnykh, O. P.; Aliev, Z. G. Chem. Heterocyclic Comp. 2000, 36, 355–356.

- 18. Zimmerman, H. E.; Wang, P. J. Org. Chem. 2002, 67, 9216-9226.
- 19. Malherbe, R.; Rist, G.; Bellus, D. J. Org. Chem. 1983, 48, 860-869.
- 20. Gonda, J. Angew. Chem. Int. Ed. 2004, 43, 3516-3524.
- Bravo, P.; Arnone, A.; Bandiera, P.; Bruché, L.; Ohashi, Y.; Ono, T.; Sekine, A.; Znada, M. Eur. J. Org. Chem. 1999, 111–115.
- 22. Christl, M.; Lanzendörfer, U.; Grötsch, M. M.; Hegmann, J.; Ditterich, E.; Hüttner, G.; Peters, E.; Peters, K.; von Schnering, H. G. *Chem. Ber.* **1993**, *126*, 797–802.
- 23. Ayral-Kaloustian, S.; Wolff, S.; Agosta, W. C. J. Org. Chem. 1978, 43, 3314–3319.
- 24. Ishida, M.; Muramaru, H.; Kato, S. Synthesis 1989, 562-564.
- 25. Nitta, M.; Nakatani, H. Chem. Lett. 1978, 957-960.
- 26. Davies, H. G.; Rahman, S. S.; Roberts, S. M.; Wakefield, B. J.; Winders, J. A.; *J. Chem. Soc. Perkin Trans. 1* **1987**, 85–89.
- 27. Gowda, G.; McMurry, T. B. H. J. Chem. Soc. Perkin Trans. 1 1979, 274-278.
- 28. Schiess, P.; Eberle, M.; Huys-Francotte, M.; Wirz, J. Tetrahedron Lett. 1984, 25, 2201-2204.

5.5.3.3 Reaction of Nitrogen Nucleophiles and Electrophiles with Ketenes Reaction with nitrogen nucleophiles, especially with amines, is a characteristic reaction of ketenes, with nucleophilic attack in the ketene plane on the carbonyl carbon. Various mechanistic and preparative aspects of these reactions are presented in Sections 4.1.4, 5.5.2, and 5.9. Imines typically react with ketenes by [2+2] cycloadditions, forming  $\beta$ -lactams in a valuable synthetic reaction (Section 5.4.1.7). Amines are much more reactive with ketenes than is H₂O, as illustrated by the reaction of diphenylketene generated by Wolff rearrangement, which reacted more readily with the amines in aqueous solution, although H₂O was present in much higher concentration (equation 1).¹

$$Ph \rightarrow C = O \xrightarrow{RNH_2} Ph \rightarrow O \xrightarrow{Ph} NHR$$
(1)  
1 2

Difunctional amines reacted with ketenes by several pathways. 2-Aminophenol (3) reacted with ketenes 4, forming 2-substituted benzoxazoles 6, evidently through the intermediate amide enols 5 (equation 2).² 2-Amino-1-azaazulene (7) reacted with two molecules of CHCl=C=O or PhCH=C=O (8a,b), generated in situ from the acid chlorides and Et₃N and gave fused heterocycles 9 (equation 3).³ This reaction presumably involved initial acylation of the NH₂ group followed by reaction with a second molecule of 8. Pyrido[1,2-a]pyrazines 10 reacted with in situ-generated ketenes 11, forming 12, with replacement of the imino group of 10 (equation 4).⁴





Reaction of a chiral amino acid derivative 14 with a chromium-complexed ketene 13 formed 15 with some diastereoselectivity (equation 5).⁵ Other examples of stereoselective reactions of ketene with amines are given in Section 5.9.



Amides are acylated by reaction with ketene (equation 6).⁶ Silylated imine 16 reacted with diphenylketene 1, forming the O-silvlated product 17 (equation 7),⁷ and the silvlated amine 18 gave the C-silvlated product 19 with ketene (equation 8).⁸ Reactions of PhCMe=C=O (19a) with LiNMe₂ and capture of the enolate with Me₃SiCl gave the silvl enol ethers **19b** in an E/Z ratio of 1.8:1 (equation 8a).⁹

$$\overset{H}{\underset{H}{\rightarrowtail}} C=O + \overset{O}{\underset{R}{\longleftarrow}} \overset{O}{\underset{NH_2}{\longrightarrow}} \overset{O}{\underset{R}{\longleftarrow}} \overset{O}{\underset{H}{\longleftarrow}} \overset{O}{\underset{CH_3}{\bigcup}} (6)$$

$$\overset{H}{\longrightarrow} C=O \xrightarrow{Et_3SiNMe_2} \overset{H}{\longrightarrow} \overset{H}{\longrightarrow} \overset{OSiEt_3}{\longrightarrow} \overset{Et_3Si}{\longrightarrow} \overset{O}{\swarrow} (8)$$

$$\begin{array}{c} Ph \\ \searrow \\ CH_3 \end{array} C = O \xrightarrow{1) Me_2 NLi} \\ OK_3 \\ CH_3 \\ Me_2 \end{array} \xrightarrow{Ph} \\ CH_3 \\ OK_3 \\ NMe_2 \end{array}$$

$$(8a)$$



19b

The germylamine **20** reacted with two molecules of **1**, forming the *O*-germylated product **21** (equation 9),¹⁰ and the silyated azide **22** formed **23** (equation 10).¹¹



Intramolecular reactions of pendant amino groups with ketenes provided a route to lactams by the photolysis of the cyclobutanone **24**, forming ketene **25**, which cyclized to **26** (equation 11).¹² Photolysis of the cyclohexadienone **27** generated the ketene **28**, which formed the lactam **29** (equation 12).¹³ The 2-carboxamidophe-nylketene **31** generated by Wolff rearrangement of **30** cyclized to **32** (equation 13.¹⁴ Ketenes with pendant amino functions generated by thermolysis of alkynyl ethers gave efficient cyclization to macrocyclic lactams.¹⁵



Tertiary amines react with ketenes by diverse pathways. In the preparation of diphenylketene (1) by reaction of  $Ph_2CHCOCl$  and  $Et_3N$  under reflux a 6% yield



Scheme 5.4

of the enone **33** was formed, which was suggested to arise from an initial hydride transfer from  $Et_3N$  to  $Ph_2C=C=O$  (Scheme 5.4).³

Photolysis of chromium carbene complex **34** in the presence of tertiary allylic amines **36** was proposed to proceed by generation of the chromium complexed ketene **35**, which reacted with the amine, leading to a zwitterionic aza-Cope rearrangement to form the unsaturated lactam **37** (Scheme 5.5).¹⁶





Ketenes **38** generated in situ from acyl chlorides and *i*-Pr₂NEt gave a Lewis acid–catalyzed reaction with allyl amines **39**, forming **41** through a route envisioned to proceed through **40** and a Claisen rearrangement (equation 14).¹⁷ (cf. Section 5.5.3.2). The use of a chiral Lewis acid catalyst such as **42** in this process gave high enantioselectivity for formation of **41**.¹⁸ The use of allyl diamines **43** in this sequence gave highly selective formation of diamides **44** (equation 15).¹⁹





Pyridine reacted with ketenes including fulvenones **45a** and **45b** to form observable ylides **46** (cf. **91**, Section 4.9; equation 16, Section 5.5.2) (equation 16).²⁰ The structures and bonding in these species have been studied by theoretical means.²¹ Bimolecular kinetics of the addition of pyridine to fulvenones in matrices forming complex **46** have been described by the methods of Kohlrausch kinetics.²² 4-(Dimethylamino)pyridine (DMAP) was added to the acylketene **47**, forming zwitterion **48** as a yellow solid (equation 17).^{23,24} Pyrido pyrimidinylium olates **49** were prepared and characterized, and were proposed to react through their acyl-ketene forms **50** (equation 18).^{25,26}



Reaction of acyl chlorides with pyridine and diazonium salts was proposed to proceed by formation of ketenes **51** with formed ylides **52** by nucleophilic attack by pyridine (equation 19).²⁷ This was followed by electrophilic attack of

aryldiazonium ions with 52 to form hydrazones (equation 19).²⁷



Reaction of the tertiary allylamine **53** with dichloroketene gave a zwitterionic intermediate **54**, which then rearranged to the azacyclodecane **55** (equation 20).²⁸ In a similar reaction, diphenylketene **1** reacted with **56** through the proposed intermediate **57** to form **58** (equation 21).^{29,30}



Diazarane **59** reacted with diphenylketene through the intermediacy of **60** to form **61** (equation 22),³¹ while azirene **62** gave **63** (equation 23), leading to **64–67** (equation 24).³² Azirene **69** reacted with *tert*-butyl(isopropyl)ketene (**68**) to form **70**, which gave **71** (equation 25).³²





Dimerization of *tert*-butyl(cyano)ketene **72** did not occur thermally, but 0.1 equivalent of  $Et_3N$  catalyzed formation of the dimer **75** through the intermediacy of zwitterion **73** (equation 26).³³ The allene **77** was also formed, and could result from conversion of **74** to **76** followed by decarboxylation (equation 27).³³



An unusual amine cyclization occurred with the ketenyl radicals **79** generated from  $\omega$ -amino terminal alkynes **78** by free radical carbonylation (equation 28).³⁴ The reaction proceeded through addition of the radical *n*-Bu₃Sn• (G•) to the alkyne forming **79**, which combined with CO to form **80**. This cyclized to **81**, which

underwent partial destannylation, possibly with hydrogen atom transfer from an intermediate lactam enol (equation 29).



#### **REFERENCES FOR SECTION 5.5.3.3**

- 1. Andraos, J.; Kresge, A. J. J. Am. Chem. Soc. 1992, 114, 5643-5646.
- Olagbemiro, T. O.; Agho, M. O.; Abayeh, O. J.; Amupitan, J. O. Recl. Trav. Chim. Pays-Bas 1995, 115, 337–338.
- 3. Abe, N.; Osaki, I.; Kojima, S.; Matsuda, H.; Sugihara, Y.; Kakehi, A. J. Chem. Soc. Perkin Trans. 1 1996, 2351–2356.
- 4. Billert, T.; Beckert, R.; Döring, M.; Görls, H. J. Prakt. Chem. 1999, 341, 332-341.
- 5. Miller, J. R.; Pulley, S. R.; Hegedus, L. S.; DeLombaert, S. J. Am. Chem. Soc. **1992**, 114, 5602–5607.
- 6. Dunbar, R. E.; White, G. C. J. Org. Chem. 1958, 23, 915-916.
- 7. Birkofer, L.; Schramm, J. Liebigs Ann. Chem. 1977, 760-766.
- Lutsenko, I. F.; Baukov, Yu. I.; Kostyuk, A. S.; Salvelyeva, N. I.; Krysina, V. K. J. Organomet. Chem. 1969, 17, 241–262.
- Vedejs, E.; Kruger, A. W.; Lee, N.; Sakata, S. T.; Stec, M.; Suna, E. J. Am. Chem. Soc. 2000, 122, 4602–4607.
- 10. Satgé, J.; Riviere-Baudet, M. Bull. Soc. Chim. Fr. 1968, 4093-4096.
- 11. Tsuge, O.; Urano, S.; Oe, K. J. Org. Chem. 1980, 45, 5130-5136.
- 12. Rahman, S. S.; Wakefield, B. J.; Roberts, S. M.; Dowle, M. D. J. Chem. Soc. Chem. Commun. 1989, 303–304.
- 13. Quinkert, G.; Nestler, H. P.; Schumacher, B.; del Grosso, M.; Dürner, G.; Bats, J. W. *Tetrahedron Lett.* **1992**, *33*, 1977–1980.
- Benati, L.; Montevecchi, P. C.; Spagnolo, P.; Foresti, E. J. Chem. Soc. Perkin Trans 1 1992, 2845–2850.
- 15. MaGee, D. I.; Ramaseshan, M. Synlett 1994, 743-744.
- 16. Deur, C. J.; Miller, M. W.; Hegedus, L. S. J. Org. Chem. 1996, 61, 2871-2876.
- 17. Yoon, T. P.; Dong, V. M.; MacMillan, D. W. C. J. Am. Chem. Soc. 1999, 121, 9726–9727.
- 18. Yoon, T. P.; MacMillan, D. W. C. J. Am. Chem. Soc. 2001, 123, 2911-2912.
- 19. Dong, V. M.; MacMillan, D. W. C. J. Am. Chem. Soc. 2001, 123, 2448-2449.

- 20. Barra, M.; Fischer, T. A.; Cernigliaro, G. J.; Sinta, R.; Scaiano, J. C. J. Am. Chem. Soc. **1992**, *114*, 2630–2634.
- 21. Fang, D.-C.; Fu, X.-Y. J. Mol. Struct. (Theochem.) 1998, 455, 59-68.
- 22. Andraos, J. J. Phys. Chem. A, 2000, 104, 1532-1543.
- Kollenz, G.; Holzer, S.; Kappe, C. O.; Dalvi, T. S.; Fabian, W. M. F.; Sterk, H.; Wong, M. W.; Wentrup, C. *Eur. J. Org. Chem.* 2001, 1315–1322.
- 24. Gompper, R.; Wolf, U. Liebigs Ann. Chem. 1979, 1388-1405.
- 25. Fiksdahl, P.; Plüg, C.; Wentrup, C. J. Chem. Soc. Perkin Trans. 2 2000, 1841-1845.
- Plüg, C.; Wallfisch, B.; Andersen, H. G.; Bernhardt, P. V.; Baker, L.-J.; Clark, G. R.; Wong, M. W.; Wentrup, C. J. Chem. Soc. Perkin Trans. 2 2000, 2096–2108.
- 27. Atlan, V.; El Kaim, L.; Supiot, C. Chem. Commun. 2000, 1385-1386.
- 28. Edstrom, E. D. J. Am. Chem. Soc. 1991, 113, 6690-6692.
- 29. Maurya, R.; Pittol, C. A.; Pryce, R. J.; Roberts, S. M.; Thomas, R. J.; Williams, J. O. J. Chem. Soc. Perkin Trans. 1 1992, 1617–1621.
- 30. Roberts, S. M.; Smith, C.; Thomas, R. J. J. Chem. Soc. Perkin Trans. 1 1990, 1493-1495.
- 31. Carboni, B.; Toupet, L.; Carrie, R. Tetrahedron 1987, 43, 2293-2302.
- 32. Schaumann, E.; Grabley, S.; Henriet, M.; Ghosez, L.; Touillaux, R.; Declercq, J. P.; Germain, G.; Van Meerssche, M. J. Org. Chem. 1980, 45, 2951–2955.
- 33. Moore, H. W.; Duncan, W. G. J. Org. Chem. 1973, 38, 156-158.
- 34. Tojino, M.; Uenoyama, Y.; Fukuyama, T.; Ryu, I. Chem. Comm. 2004, 2482-2483.

**5.5.3.4** Carbon and Silicon Nucleophiles The addition of Grignard reagents to ketenes was demonstrated by Staudinger in 1907,¹ and it was then shown by Gilman and Heckert that the enolate **2** formed by addition of phenylmagnesium bromide to diphenylketene (**1**) could be trapped as the enol ester **3** (equation 1).²

Surveys of the reactions of ketenes with Grignard reagents³ and organolithium reagents⁴ have been presented, and reactions of ketenes with organolithium reagents are summarized in Table 5.17.^{5–11}

Di-*tert*-butylketene (4) is crowded and unreactive, but addition of *tert*-butyllithium formed the enolate 5, which reacted with Me₃SiCl and gave the silyl enol ether 6 (equation 2), and with water formed the ketone 7 (equation 3).¹² Efforts to enolize the ketone 7 were unsuccessful, so the ketene addition route was the only proven way to form 6.¹²


Ketene	RLi	Ref	
EtMeC=C=O	MeLi, PhCH ₂ Li	8	
<i>i</i> -PrMeC=C=O	MeLi, n-BuLi, PhCH ₂ Li	8	
t-BuMeC=C=O	MeLi, n-BuLi, PhCH ₂ Li	8	
$Me_2C=C=O$	<i>n</i> -BuLi, PhC≡CLi	8	
$Et_2C=C=O$	<i>n</i> -BuLi	8	
t-Bu ₂ C=C=O	PhLi	7	
t-Bu ₂ C=C=O	t-BuLi	12	
<b>∠</b> =0	<i>n</i> -BuLi, PhLi, PhCH ₂ Li	8	
C=0	PhCH ₂ Li	8	
C=0	PhCH ₂ Li	8	
C=0	PhCH ₂ Li	8	
$(Me_3M)_2C=C=O$	MeLi, n-BuLi, PhLi	9, 10	
M= Si, Ge, Sn			
$Mes_2C=C=O$	<i>i</i> -PrLi, <i>t</i> -BuLi	34	
PhMeC=C=O	<i>n</i> -BuLi	5	
PhEtC=C=O	<i>n</i> -BuLi	5	
Ph(i-Pr)C=C=O	n-BuLi, t-BuLi	6	
Ph(t-Bu)C=C=O	n-BuLi, t-BuLi	6	
C=0	PhLi	5	
C=0	PhLi	11	

TABLE 5.17 Ketene Reactions with Organolithium Reagents



Reactions of unsymmetrical disubstituted ketenes with organometallic reagents showed a preference for attack of the nucleophile from the least hindered side of the ketene, as summarized in Table 5.18.

In the addition of MesMgBr to mesitylphenylketene **8** there was a 3:1 preference for attack on the side of the mesityl group of the ketene forming **9**, a result ascribed to the adoption of a conformation **8a** in which the mesityl was nearly perpendicular to the ketene plane, so approach to the carbonyl carbon in the ketene plane from this

R ¹	$R^2$	RLi $R^1$	$\begin{array}{c c} OLi \\ R \\ R \\ R \\ \end{array} \xrightarrow{R^1} R^1 \\ R^2 \\ OLi \\ \end{array}$	Ref
t-Bu	Н	t-BuLi	>(95/5)	5
Et	Me	MeLi	1.7/1	7
<i>i</i> -Pr	Me	MeLi	7.0/1	7
<i>t</i> -Bu	Me	MeLi	>(99/1)	7
t-Bu	<i>i</i> -Pr	t-BuLi	>(95/5)	6
Ph	Me	n-BuLi	>(95/5)	5
Ph	Et	n-BuLi	>(95/5)	5
Ph	Et	Me ₃ SiLi	>(95/5)	6
Ph	<i>i</i> -Pr	t-BuLi	1/4	6
Ph	<i>t</i> -Bu	t-BuLi	>(5/95)	6
Ph	<i>c</i> -Pr	n-BuLi	79/21	6
Ph	<i>c</i> -Pr	t-BuLi	9/91	6
Ph	<i>c</i> -Pn	n-BuLi	88/12	6
Ph	<i>c</i> -Hx	n-BuLi	56/44	6
	:=0	CH ₂ =CHCH ₂ MgCl	9/1	11
Et	Me ₃ Si	n-BuLi	>(5/95)	5
Et	Me ₃ Si	CH ₂ =CHLi	>(5/95)	5
Et	Me ₃ Si	HC≡CLi	>(5/95)	5

TABLE 5.18 Stereoselectivity in Addition of Organometallic Reagents to Ketenes  $R^1R^2C=C=O$ 

direction was less hindered (equation 4).¹³ The transition state for this reaction was modeled by *ab initio* calculations.¹³



Reactions of diarylketenes with highly substituted benzene rings gave stable enols, as in the example of dimesitylketene (10) forming 11 (equation 5).^{14,15} Ditipylketene (12) reacted similarly with MeLi and *t*-BuLi to give the stable enols 13a and 13b, respectively (equation 6).¹⁴ The reaction of dimesitylketene 10 with

alkenyl- and alkynyllithium compounds formed dienols and yne-enols (equation 5) for studies of one-electron oxidation chemistry.¹⁵

Organocerium reagents were added to silylketenes **14** to generate enolate anions **15**, which reacted with NH₄Cl or alkyl halides to give  $\alpha$ -silylketones **16** (equation 7).^{16,17} Samarium iodide–promoted addition of allyl iodide to unsymmetrical disubstituted ketenes **17** gave enolates **18**, which were protonated enantiose-lectively with a chiral alcohol to ketones **19** (equation 8).^{18,19}



The reaction of Me₃SiCN with ketenes **1**, **20**, and **21** was proposed to take place by a  $[\pi_2+\pi_2+\sigma_2]$  cycloaddition forming silyl enol ethers **22**, which were converted to the corresponding enol sulfonates, carboxylates, or phosphates (equation 9).²⁰ *tert*-Butylisocyanide reacted with diphenylketene (**1**), forming the intermediate **23**, which gave **24** (equation 10).²¹



## 572 REACTIONS OF KETENES

The kinetics of addition of cyanide ion to pentafulvenone (**25**) in aqueous NaOH were measured, with a rate constant at 24 °C  $k(CN^-) = 1.6 \times 10^8 \text{ s}^{-1} \text{ M}^{-1}$ , which was three times greater than the corresponding rate constant for reaction with OH⁻ (equation 11).²² However, the product of the reaction could not be isolated.²² *Ab initio* computations at the MP2/6-31+G*//MP2/6-31+G* level for the attack of cyanide anion on the ketenes **20** and **28–30** show that in all cases the product enolates from in-plane attack leading are favored over those from perpendicular attack and also have lower barriers for reaction (equations 12–15).²³ In the case of **30** the barrier for perpendicular attack was only 2.8 kcal/mol higher than for in-plane attack, and this was suggested to be due to some incipient aromatic stabilization in the transition state.²³

$$\begin{array}{c} & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & &$$

$$\begin{array}{c} & & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ &$$

$$\bigcup_{0}^{C^{*O}} \xrightarrow{C^{N^{*}}}_{0} \bigcup_{0}^{C^{N^{*}}} (14)$$

 $C^{-1} \xrightarrow{C^{-1}}_{O} \xrightarrow{C^{-1$ 

Addition of the fluoro carbanion **32** to bis(trifluoromethyl)ketene **31** resulted in formation of the ketone **34** (equation 16).²⁴ Friedel-Crafts reaction of diphenylketene (1) was induced by aluminum chloride, forming **35** (equation 17),^{25,26} and **1** added to furan **36** without catalyst (equation 18).²⁷

28

29

$$\begin{array}{c} CF_{3} \\ \searrow \\ CF_{3} \end{array} C=0 \xrightarrow{CF(CF_{3})SO_{2}F} CF_{3} \xrightarrow{CF_{3}} O^{-} \xrightarrow{H^{+}} CF_{3} \xrightarrow{CF_{3}} O^{-} \xrightarrow{CF(CF_{3})SO_{2}F} O^{-} \xrightarrow{H^{+}} O^{-} \xrightarrow{CF_{3}} O^{-} \xrightarrow{CF(CF_{3})SO_{2}F} O^{-} \xrightarrow{H^{+}} O^{-} \xrightarrow{CF_{3}} O^{-} \xrightarrow{CF(CF_{3})SO_{2}F} O^{-} \xrightarrow{CF(CF_{3})SO_{2}F} O^{-} \xrightarrow{CF_{3}} O^{-} \xrightarrow{CF_{3}} O^{-} \xrightarrow{CF(CF_{3})SO_{2}F} O^{-} \xrightarrow{CF_{3}} O^{-} \xrightarrow{CF$$



Ketene acetal **38** added to dimethylketene (**21**) formed the vinyl silyl ether **39** (equation 19).²⁸ Ketene (**20**) generated by dehydrochlorination reacted with the ketene acetal **40** to form the silyl vinyl ether **41** in 98% yield (equation 20).²⁹ The product **41** was utilized for an oxonia-Cope Prins cyclization.²⁹ Enolates reacted with ketenes by initial attack on oxygen followed by rearrangement to the product of nucleophilic attack by carbon (equation 10, Section 5.5.3.2).



Phenylethylketene (42) reacted with aroyl chlorides in a reaction catalyzed by  $Pd(PPh_3)_4$  with decarbonylation and bond formation at  $C_1$  of the ketene, forming 43 (equation 21),³⁰ while diphenylketene (1) reacted with allylpalladium complex 44 with carbon-carbon bond formation at  $C_2$  and either decarbonylation in benzene, forming 45, or ester formation in DMF, forming 46 (equation 22).³¹



Allyl and propargyl iron complexes **47** and **50**, respectively, were interpreted as adding at  $C_1$  of diphenylketene (equations 23, 24).^{32,33} Reaction of  $Ph_2C=C=O$ 

with diazoborole 53 gave complexes 54 (equation 25).³⁴



Chromium-complexed ketene **56** generated by photolysis of the chromium-carbene complex **55** underwent attack by the electron-rich aromatic side chain in an intramolecular Friedel-Crafts acylation (equation 26).³⁵



Tris(trimethylsilyl)silyllithium (**59**) reacted with PhCMe=C=O (**58**) to give the enol ether **60**, proposed to form by silyl exchange (equation 27).³⁶ The formation of **60a** was demonstrated by quenching experiments.³⁶ Reaction of **59** with  $(Me_3Si)_2C=C=O$  gave the ynolate Me_3SiC=COLi.³⁷ Reaction of hexamethyldisilane with MeLi formed both Me_3SiLi, and Me_3SiMe_2SiLi, and addition of this mixture to dimesitylketene (**61**) gave the enols **62** and **63** (equation 28).^{37,38}



## **REFERENCES FOR SECTION 5.5.3.4**

- 1. Staudinger, H. Chem. Ber. 1907, 40, 1145-1148.
- 2. Gilman, H.; Heckert, L. C. J. Am. Chem. Soc. 1920, 42, 1010-1014.
- 3. Combret, J.-C. Ann. Chim. 1969, 481-496.
- 4. Seikaly, H. R.; Tidwell, T. T. Tetrahedron 1986, 42, 2587-2613.
- Baigrie, L. M.; Seikaly, H. R.; Gong, L.; Tidwell, T. T. J. Am. Chem. Soc. 1985, 107, 5391–5396.
- 6. Allen, A. D.; Baigrie, L. M.; Seikaly, H. R.; Tidwell, T. T. Can. J. Chem. 1991, 69, 138-145.
- 7. Schaumann, E.; Walter, W. Chem. Ber. 1974, 107, 3562-3573.
- 8. Häner, R.; Laube, T.; Seebach, D. J. Am. Chem. Soc. 1985, 107, 5396-5403.
- 9. Woodbury, R. P.; Long, N. R.; Rathke, M. W. J. Org. Chem. 1977, 43, 376.
- Lebedev, S. A.; Ponomarev, S. V.; Lutsenko, I. F. Zh. Org. Khim. 1972, 42, 647–651. J. Gen. Chem. 1972, 42, 643–647.
- 11. Fehr, C.; Galindo, J. J. Org. Chem. 1988, 53, 1828-1830.
- Baigrie, L. M.; Lenoir, D.; Seikaly, H. R.; Tidwell, T. T. J. Org. Chem. 1985, 50, 2105– 2109.
- Yamataka, H.; Aleksiuk, O.; Biali, S. E.; Rappoport, Z. J. Am. Chem. Soc. 1996, 118, 12580–12587.
- 14. Frey, J.; Rappoport, Z. J. Org. Chem. 1997, 62, 8372-8386.
- 15. Schmittel, M.; Langels, A. Liebigs Ann. 1996, 999-1004.
- 16. Kita, Y.; Matsuda, S.; Kitagaki, S.; Tsuki, Y.; Akai, S. Synlett 1991, 401-402.
- 17. Akai, S.; Kitagaki, S.; Matsuda, S.; Tsuzuki, Y.; Naka, T.; Kita, Y. *Chem. Pharm. Bull.* **1997**, *45*, 1135–1139.
- Takeuchi, S.; Ohira, A.; Miyoshi, N.; Mashio, H.; Ohgo, Y. *Tetrahedron: Asymm.* 1994, 5, 1763–1780.
- 19. Miyoshi, N.; Takeuchi, S.; Ohgo, Y. Bull. Chem. Soc. Jpn. 1994, 67, 445-451.
- 20. Hertenstein, U.; Hünig, S.; Reichelt, H.; Schaller, R. Chem. Ber. 1982, 115, 261-287.
- 21. ElGomati, T.; Firl, J.; Ugi, I. Chem. Ber. 1977, 110, 1603-1605.
- 22. Urwyler, B.; Wirz, J. Angew. Chem. Int. Ed. Engl. 1990, 29, 790-791.
- 23. Keefe, J. R. J. Phys. Org. Chem. 2004, 17, 1075-1083.
- 24. Ermolov, A. F.; Eleev, A. F. Zh. Org. Khim. **1983**, 19, 1340-1341. Engl. Transl. **1983**, 19, 1198–1199.
- 25. Fountain, K. R.; Heinze, P.; Maddex, D.; Gerhardt, G.; John, P. Can. J. Chem. **1980**, 58, 1939–1946.
- 26. Williams, J. W.; Osborn, J. M. J. Am. Chem. Soc. 1939, 61, 3438-3439.
- 27. Nenitzescu, C. Chem. Ber. 1931, 64, 1924-1931.
- Burlachenko, G. S.; Baukov, Yu. I.; Lutsenko, I. F. Zh. Obshch. Khim. 1965, 35, 933– 934; 1968, 38, 2815–2816. Engl. Transl. 1965, 35, 939; 1968, 38, 2716.
- 29. Dalgard, J. E.; Rychnovsky, S. D. J. Am. Chem. Soc. 2004, 126, 15662-15663.
- 30. Mitsudo, T.; Kadokura, M.; Watanabe, Y. J. Chem. Soc. Chem. Commun. 1986, 1539-1541.
- 31. Mitsudo, T.; Kadokura, M.; Watanabe, Y. J. Org. Chem. 1987, 52, 3186-3192.
- 32. Bucheister, A.; Klemarczyk, P.; Rosenblum, M. Organometallics 1982, 1, 1679-1684.

- Chen, L. S.; Lichtenberg, D. W.; Robinson, P. W.; Yamamoto, Y.; Wojcicki, A. *Inorg. Chem. Acta* 1977, 25, 165–172.
- Weber, L.; Schneider, M.; Maciel, T. C.; Wartig, H. B.; Schimmel, M.; Boese, R.; Bläser, D. Organometallics 2000, 19, 5791–5794.
- 35. Bueno, A. B.; Moser, W. H.; Hegedus, L. S. J. Org. Chem. 1998, 63, 1462-1466.
- 36. Naka, A.; Ohshita, J.; Kunai, A.; Lee, M. E.; Ishikawa, M. J. Organomet. Chem. 1999, 574, 50–57.
- 37. Nadler, E. B.; Rappoport, Z. Tetrahedron Lett. 1990, 31, 555-558.
- 38. Nugiel, D. A.; Rappoport, Z. J. Am. Chem. Soc. 1985, 107, 3669-3676.

# 5.5.4 Wittig-Type Reactions

Diphenylketene (1) was found to react with the phosphorus ylide 2 to form tetraphenylallene (3) by Lüscher in the laboratory of Staudinger in 1922 (equation 1).^{1,2} This reaction was later developed by Wittig and Haag,³ and the Horner-Emmons reaction of carboethoxyphosphonium ylides was applied to ketenes generated in situ from acyl chlorides^{4,5} for the reaction of phenylketene (4) forming 5 (equation 2), and *tert*-butylketene (6) forming 7 (equation 3).^{6,7}

$$Ph \xrightarrow{Ph} C = O + Ph_{3}P = CPh_{2} \xrightarrow{Ph} Ph \xrightarrow{Ph} C = \swarrow Ph$$

$$Ph \xrightarrow{Ph} Q \xrightarrow{FLN} Ph \xrightarrow{Ph} Ph \xrightarrow{Ph-P-CHCO_{2}Me} Ph \xrightarrow{Ph} H$$

$$(1)$$

$$t \xrightarrow{\text{Etg}AV} \xrightarrow{\text{Etg}AV} \xrightarrow{\text{T-Bu}} = C = 0 \xrightarrow{\text{Etg}AV} \xrightarrow{\text{T-Bu}} = C = 0 \xrightarrow{\text{CO}_2\text{Me}} (2)$$

$$t \xrightarrow{\text{Etg}AV} \xrightarrow{\text{T-Bu}} \xrightarrow{\text{C}} = 0 \xrightarrow{\text{Ph}_3\text{P}=\text{CHCO}_2\text{Et}} \xrightarrow{\text{T-Bu}} = C \xrightarrow{\text{C}} \xrightarrow{\text{CO}_2\text{Me}} (2)$$

$$t \xrightarrow{\text{C}} \xrightarrow{\xrightarrow{C}} \xrightarrow{\text{C}} \xrightarrow{\text{C}} \xrightarrow{\text{C}} \xrightarrow{\text{C}} \xrightarrow{\xrightarrow{C} \xrightarrow{C} \xrightarrow{\text{$$

Allenyl ketones **9** were also prepared from arylketenes **8** by this method (equation 4).⁸ Phosphorane **10** with a chiral auxiliary attached reacted with methylketene (**11**) for formation of chiral allene **12** (equation 5).⁹



Phosphonate anion **14** reacted with ketene **13** to form allene **15** (equation 6).^{10–15} Ketene (CH₂=C=O) reacted with phosphonate anions to give terminal allenes.^{17,18} Reaction of the optically active phosphinylacetate anion **17** with ketene **16** gave the enantiomerically enriched allene **18** (equation 7).¹⁴

Vinyl phosphorane **20** reacted with dimethylketene **19** to give the vinylallene **21** (equation 8), but the  $\alpha$ -phenyl analogue **22** reacted with two moles of **19** to form **24** after hydrolysis (equation 9).¹⁹



The chromium alkoxycarbene complex 25 upon photolysis gave the chromiumcomplexed ketene 26, which gave a Wittig reaction forming allene 27 (equation 10).²⁰ In some cases the allenes formed in this way were not isolated but were hydrolyzed to the corresponding ketones.²⁰



Wittig reaction of diphenylketene (1) with the phosphonioalkylidene rhenium (VII) complex 28 provided an expeditious route to the allenylidene rhenium (VII) complex 29 (equation 11).²¹ The structure of 29 was confirmed by X-ray analysis.



The reaction of ketenes with iminophosphoranes to give aza-Wittig reactions was discovered by Staudinger and Hauser²² and has been used to prepare ketenimines.²³ This procedure was applied to the reaction of trimethylsilylketene (**30**) with bistrimethyliminophosphorane (**31**) and gave the ketenimine **32** (equation 12).²⁴ Diphenylketene gave a similar reaction.²⁴

$$\underset{H}{\overset{Me_{3}Si}{\underset{H}{\longrightarrow}}} C=O + Ph_{3}P=NCH(SiMe_{3})_{2} \longrightarrow \underset{H}{\overset{Me_{3}Si}{\underset{H}{\longrightarrow}}} C=N \xrightarrow{CH(SiMe_{3})_{2}} (12)$$

Arylketenes **33** generated by dehydration of phenylacetic acids with DCC/Et₃N in THF at 0 °C reacted with iminophosphoranes to give ketenimines **34** that underwent cyclization to dihydroisoquinolines **35** (equations 13, 14).²⁵ Pyrimidone **38** was obtained by the aza-Wittig reaction of PhCH=C=O (**4**) with **36**, with cyclization of the intermediate **37** (equation 15).²⁶



Allyliminophosphorane **39** reacted with ketenes **40**, forming ketenimines **41**, which gave aza-Claisen rearrangements to **42** (equation 16).²⁷ Diphenylketene

(1) reacted with the acyliminophosphorane 43 to form the acylketenimine 44 (equation 17).²⁸



As discussed in Section 4.6, phosphaketene ylide  $Ph_3P=C=C=O$  reacts with nucleophiles at the carbonyl carbon, forming reactive acyl Wittig reagents.

# **REFERENCES FOR SECTION 5.5.4**

- 1. Lüscher, G. Dissertation, ETH Zürich, 1922, as cited in ref. 2.
- 2. Lang, R. W.; Hansen, H.-J. Helv. Chim. Acta 1980, 63, 438-455.
- 3. Wittig, G.; Haag, A. Chem. Ber. 1963, 96, 1535-1543.
- 4. Bestmann, H.; Hartung, H.-J. Chem. Ber. 1966, 99, 1198-1207.
- 5. Nader, F. W.; Brecht, A.; Kreisz, S. Chem. Ber. 1986, 119, 1196-1207.
- 6. Lang, R. W.; Hansen, H.-J. Helv. Chim. Acta 1979, 62, 1025-1039.
- 7. Marshall, J. A.; Wolf, M. A.; Wallace, E. M. J. Org. Chem. 1997, 62, 367-371.
- 8. Kumar, K.; Kaur, S.; Ishar, M. P. S. Synlett 1999, 1237-1238.
- Melo, T. M. V. D. P. e.; Cardoso, A. L.; Gonsalves, A. M. d'A. R.; Storr, R. C.; Pessoa, J. C.; Paixão, J. A.; Beja, A. M.; Silva, M. R. *Tetrahedron Lett.* 2003, 44, 6409–6412.
- 10. Runge, W.; Kresze, G.; Ruch, E. Liebigs Ann. Chem. 1975, 1361-1378.
- 11. Kresze, G.; Runge, W.; Ruch, E. Liebigs Ann. Chem. 1972, 112-127.
- 12. Lang, R. W.; Hansen, H.-J. Org. Synth. 1984, 62, 202-209.
- Schweizer, E. E.; Hsueh, W.; Rheingold, A. L.; Durney, R. L. J. Org. Chem. 1983, 48, 3889–3894.
- 14. Musierowicz, S.; Wróblewski, A. E. Tetrahedron 1980, 36, 1375-1380.
- Fillion, H.; Refouvelet, B.; Péra, M. H.; Dufaud, V.; Luche, J. L. Synth. Commun. 1989, 19, 3343–3348.
- 16. Aksnes, G.; Frøyen, P. Acta Chem. Scand. 1968, 22, 2347-2352.
- 17. Rafizadeh, K.; Yates, K. J. Org. Chem. 1984, 49, 1500-1506.
- 18. Hamlet, Z.; Barker, W. D. Synthesis 1970, 543-544.
- 19. Capuano, L.; Wamprecht, C.; Willmes, A. Chem. Ber. 1982, 115, 3904–3907.
- 20. Sestrick, M. R.; Miller, M.; Hegedus, L. S. J. Am. Chem. Soc. 1992, 114, 4079-4088.

- 21. Li, X.; Schopf, M.; Stephan, J.; Kippe, J.; Harms, K.; Sundermeyer, J. J. Am. Chem. Soc. **2004**, *126*, 8660–8661.
- 22. Staudinger, H.; Hauser, E. Helv. Chim. Acta 1921, 4, 887-896.
- 23. Lee, K.-W.; Singer, L. A. J. Org. Chem. 1974, 39, 3780-3781.
- 24. Barbaro, G.; Battaglia, A.; Giorgianni, P.; Guerrini, A.; Seconi, G. J. Org. Chem. 1995, 60, 6032–6039.
- 25. Rodriguez, J. A. R.; Leiva, G. C.; de Sousa, J. D. F. Tetrahedron Lett. 1995, 36, 59-62.
- 26. Jayakumar, S.; Kumar, V.; Mahajan, M. P. Tetrahedron Lett. 2001, 42, 2235-2237.
- 27. Molina, P.; Alajarin, M.; Lopez-Leonardo, C. Tetrahedron Lett. 1991, 32, 4041-4044.
- 28. Meier, S.; Würthwein, E.-U. Chem. Ber. 1990, 123, 2339-2347.

# 5.6 ELECTROPHILIC ADDITION TO KETONES

#### 5.6.1 Protonation of Ketenes

In agreement with the known electronic distribution of ketenes with negative charge at oxygen and C₂, complexes of HCl with  $CH_2=C=O(1a)$ ,  $(CH_2)_2C=C=O$  (ketenylidenecyclopropane, **1b**), and C₃O₂ (carbon suboxide, **1c**) have been characterized by FTIR matrix spectroscopy and *ab initio* calculations as T-shaped species involving electrophilic proton addition to C₂ of the ketene, as in **1a–c**.¹ For C₃O₂ a linear complex with hydrogen bonding to oxygen was calculated to be slightly more stable, but the observed spectra were more compatible with the T-structure **1c**.¹ Theoretical studies of the HCl and HF complexes were reported.² These complexes serve as models for proton addition to ketenes.



As noted in Section 3.2.2, photolyses of acetyl chloride and acetyl bromide in solid argon gave rise to IR spectra attributed to ketene complexes with HCl and HBr respectively.^{3–5} While the HCl and HBr IR absorptions differed significantly from those of the monomers, the ketene IR was essentially unchanged. Complexes of HCl and C₃O₂ were prepared directly,⁶ or by matrix photolysis of chlorocarbonyl-ketene **2** forming **1c** (equation 1).⁷ Studies of protonation of C₃O₂ are reported in Section 3.2.2.^{8,9}

$$CI \xrightarrow{0} C=0 \xrightarrow{hv} \xrightarrow{0 \leq C \leq C^{\leq O}} I \qquad (1)$$

$$2 \qquad Ic$$

The addition of HCl to trimethylsilylketene in a matrix has been studied by IR, and shows formation of an initial complex **1b** with the ketene IR band shifted to  $2110 \text{ cm}^{-1}$ . Upon warming, this forms the addition product **1c**, and then cleavage of the C–Si bond and formation of acetyl chloride (equation 1a).^{9a}



As discussed in Section 1.1.2, gas phase protonation of ketene (**3**) and methylketene (**5**) has been extensively examined experimentally and theoretically and occurs on C₂, forming the acylium ions **4** and **6**, respectively, which are stabilized by  $\pi$ -electron donation from oxygen (equations 2, 3).^{10–18} Protonation of ketenes RCH=C=O (R=NH₂, OH, and F) were also considered.^{18a} The proton affinity of **3** is -195 kcal/mol (equation 2),¹⁹ while that of **5** is -201 kcal/mol (equation 3).²⁰ The difference may be attributed to stabilization of **6** by the methyl group, but this difference is attenuated by the strong stabilization by oxygen.

6

The energetics of the addition of hydrogen halides to  $CH_2=C=O$  (3) and  $Me_2C=C=O$  (7) to give C- and O-protonated ketene ion pairs were examined, and it was concluded that C-protonation was more favorable.²³ Calculations of the protonation of vinylketene (8) indicated that protonation at  $C_4$  is favored by 15.8 kcal/mol relative to protonation at  $C_2$  (equation 4).²⁴ The resulting ion 9 has an allylic resonance structure with acylium ion character.

5

$$\xrightarrow{H^*} \xrightarrow{CH_3 - \overset{+}{CH}} \xrightarrow{CH_3 - \overset{+}{CH}} \xrightarrow{CH_3 - CH} \xrightarrow{(4)} \xrightarrow{8} 9$$

Protonation of diphenylketene (10) and di-*tert*-butylketene (11) in FSO₃H-SbF₅/SO₂ClF at -60 °C gave the corresponding acylium ions 12, as detected by their ¹³C NMR signals for CO⁺ at  $\delta$  154.7 and 155.3, respectively (equation 5).²⁵ Reaction of Ph₂C=C=O (11) with NOBF₄ or I₂ led to the formation of Ph₂C=CPh₂, which

may be attributed to decarbonylation of the acylium ion and reaction of the resulting cation 12a with 10 (equation 5a).²⁵

$$\begin{array}{c}
\stackrel{R}{\longrightarrow} C=O & \xrightarrow{H^{+}} & \stackrel{R}{\longrightarrow} \stackrel{+}{\longrightarrow} C=O \\
\stackrel{R}{\longrightarrow} & \stackrel{R}{\longrightarrow} C=O \\
10 (R=Ph), 11 (R=t-Bu) & 12
\end{array}$$
(5)

The  $\gamma$ -oxoketene **13** added HCl, leading to the  $\delta$ -chloro- $\gamma$ -lactone **15**, which resulted from protonation forming **14** or alternatively by protonation on oxygen followed by cyclization (equation 6).^{26,27}



Addition of anhydrous acids to ketenes permitted the formation of mixed anhydrides, including acetyl dihydrogen phosphate **16**, from ketene (equation 7).²⁸ Addition of  $CH_3SO_3H$  to ketenylidenecyclohexane (**17**) gave the mixed anhydride **18** (equation 8).²⁹ Carboxylic acids also added to ketenes to give anhydrides.^{30,31}



 $\alpha$ -Alkoxy carbocations **19** generated from CH₃CH(OAc)OEt and ZnCl₂ added to C₂ of ketene, forming **20** (equation 9).³² Reactions of dichloroketene (**21**) with acetals **22** catalyzed by ZnCl₂ gave  $\beta$ -alkoxy esters **23** and were proposed to involve initial nucleophilic attack by the acetal oxgyen with Lewis acid activation of the ketene to give an enolate that combined with the carboxonium ion (equation 10).³³ Chiral 1,3-dioxolanes **24** reacted similarly with phenylketene (**25**) to form 1,4-dioxepan-5-ones **26** with high stereoselectivity (equation 11).³³ These reactions may alternatively be depicted as involving initial attack of carboxonium ions at C₂

of the ketene, but the low reactivity of the electron-poor dichloroketene makes this unlikely.



Triphenylmethyl cation **27** added to  $C_2$  of ketene, forming **28** (equation 12).³⁴ The addition of **27** to diphenylketene (**11**) gave the acylium ion **29**, which formed the observed pentaphenylethyl cation **30** by decarbonylation (equation 13).²⁵



Carbocation **31** generated from protonation of pentamethylcyclopentadiene reacted with alkyl(aryl)ketenes **32** proposed to give a [2+2] cycloadduct **33** (equation 14), leading to formation of cycloheptenediones **36** (equation 15).³⁵





Reaction of acyl chlorides **37** with pyridine and trifluoroactic anhydride (TFAA) was proposed to involve formation of ketenes **38** that underwent electrophilic attack by TFAA to give  $\beta$ -trifluoroacetylacetic acid derivatives **39** that decarboxylated, forming trifluoromethyl ketones **40** (equation 16).³⁶ Addition of MeOH trapped **38** as the esters. Quenching of the reactions with dienophiles led to formation of heterocycles **42** that evidently result from cycloaddition reactions of intermediate acylketenes **41** formed by elimination from **39** (equation 17).^{36–38}



Halogens and analogous reagents add readily to ketenes; for example, the addition of bromine to trimethylsilylketene (1a) formed 43, which was dehydrobrominated to the bromoketene that was identified by IR but could not be isolated, but was trapped by imine [2+2] cycloaddition (equation 18).³⁹ Iodine added similarly to dimethylketene (7, equation 19).⁴⁰ Rate constants for the latter reaction and for bromination of Ph₂C=C=O (10) have been reported,⁴⁰ as well as for bromination of some silylated ketenes (Section 4.5).



Phenylketene (25) was generated in toluene by dehydrochlorination, and reacted with hexachlorocyclohexadienone 45 as a mild chlorinating agent and with benzoylquinine as a chiral catalyst to give stereoselective formation of the  $\alpha$ -chloro ester 46 (equation 20).^{41–43a}



Hydroboration of ketene in the gas phase produced a product with the mass of a 1:1 addition compound, and an adduct with the structure  $H_2BCH_2CH=O$  was proposed, possibly with some coordination between boron and oxygen.⁴⁴ Diphenyl-ketene (11) reacted with 1,3-di-*tert*-butyl-2,3-dihydro-1*H*-1,3,2-diazaboroles **47** to form 1,3,2-oxaazaborolidines **48** in a reaction proposed to be initiated by boron attack at the ketenyl oxygen (equation 21).⁴⁵



Transfer of a methyl cation derived from methanol to ketene has been proposed as a step in the zeolite-promoted methanol to gasoline (MTG) conversion (equation 22).⁴⁶

$$\overset{H}{\underset{H}{\longrightarrow}} C=0 \xrightarrow{Z-CH_3} \overset{CH_3}{\underset{C}{\longrightarrow}} \overset{CH_3}{\underset{C}{\longrightarrow}} \overset{+}{\underset{C}{\longrightarrow}} O \xrightarrow{-H^*} \overset{H}{\underset{H}{\longrightarrow}} \overset{H}{\underset{H}{\longrightarrow}} \overset{H}{\underset{H}{\longrightarrow}} (22)$$

The rate constant for ketene protonation in the gas phase has been calculated,⁴⁷ and the structure of the nitosonium adduct with ketene (**49**) was determined by computation.⁴⁸



# **REFERENCES FOR SECTION 5.6.1**

- 1. Pietri, N.; Chiavassa, T.; Allouche, A.; Aycard, J.-P. J. Phys. Chem. A 1997, 101, 1093–1098.
- 2. Sumathi, R.; Chandra, A. K. Chem. Phys. Lett. 1997, 271, 287-295.
- 3. Kogure, N.; Ono, T.; Suzuki, E.; Watari, F. J. Mol. Struct. 1993, 296, 1-4.
- 4. Kogure, N.; Hatakeyama, R.; Suzuki, E.; Watari, F. J. Mol. Struct. 1993, 299, 105-109.

- 5. Rowland, B.; Hess, W. P. Chem. Phys. Lett. 1996, 263, 574-580.
- 6. Piétri, N.; Tamburelli, I.; Aycard, J. P.; Chiavassa, T. J. Mol. Struct. 1997, 416, 187-195.
- 7. Piétri, N.; Chiavassa, T.; Allouche, A.; Rajzmann, M.; Aycard, J.-P. J. Phys. Chem. 1996, 100, 7034–7041.
- Tortajada, J.; Provot, G.; Morizur, J.-P.; Gal, J.-F.; Maria, P.-C.; Flammang, R.; Govaert, Y. Int. J. Mass. Spectrom. Ion Proc. 1995, 141, 241–255.
- 9. Allen, A. D.; McAllister, M. A.; Tidwell, T. T. J. Chem. Soc. Chem. Commun. 1995, 2547–2548.
- 9a. Piétri, N.; Gaudel-Siri, A.; Couturier-Tamburelli, I.; Pons, J.-M.; Aycard, J.-P. J. Phys. Chem. A 2005, 109, 2120–2126.
- 10. Eberlin, M. N.; Majumdar, T. K.; Cooks, R. G. J. Am. Chem. Soc. 1992, 114, 2884–2896.
- 11. Ausloss, P.; Lias, S. G. Chem. Phys. Lett. 1977, 51, 53-56.
- 12. Vogt, J.; Williamson, A. D.; Beauchamp, J. L.; J. Am. Chem. Soc. 1978, 100, 3478-3483.
- 13. Debrou, G. B.; Fulford, J. E.; Lewars, E. G.; March, R. E. Int. J. Mass Spectrom. Ion Phys. 1978, 26, 345–352.
- 14. Davidson, W. R.; Lau, Y. K.; Kebarle, P. Can. J. Chem. 1978, 56, 1016-1019.
- Armitage, M. A.; Higgins, M. J.; Lewars, E. G.; March, R. E. J. Am. Chem. Soc. 1980, 102, 5064–5068.
- Beach, D. B.; Eyermann, C. J.; Smit, S. P.; Xiang, S. F.; Jolly, W. L. J. Am. Chem. Soc. 1984, 106, 536–539.
- Traeger, J. C.; McLoughlin, R. G.; Nicholson. A. J. C. J. Am. Chem. Soc. 1982, 104, 5318–5322.
- 18. Nobes, R. H.; Bouma, W. J.; Radom, L. J. Am. Chem. Soc. 1983, 105, 309-314.
- 18a. Ervasti, H. K.; Burgers, P. C., Ruttink, P. J. A. Eur. J. Mass. Spectrom. 2004, 10, 791-799.
- 19. Bouchoux, G.; Salpin, J.-Y. J. Phys. Chem. 1996, 100, 16555-16560.
- 20. Bouchoux, G.; Salpin, J.-Y. Rapid Commun. Mass Spectrom. 1999, 13, 932-936.
- 21. McKee, M. L.; Radom, L. Org. Mass Spectrom. 1993, 28, 1238-1244.
- 22. Scott, A. P.; Radom, L. Int. J. Mass. Spectrom. Ion Proc. 1997, 160, 73-81.
- 23. Lyashchuk, S. N.; Skrypnik, Yu. G. Russ. J. Org. Chem. 1996, 32, 463-473.
- Leung-Toung, R.; Peterson, M. R.; Tidwell, T. T.; Csizmadia, I. G. J. Mol. Struct. (Theochem.) 1989, 183, 319–330.
- Olah, G. A.; Alemayehu, M.; Wu, A.; Farooq, O.; Prakash, G. K. S. J. Am. Chem. Soc. 1992, 114, 8042–8045.
- Hegmann, J.; Ditterich, E.; Hüttner, G.; Christl, M.; Peters, E.-M.; Peters, K; von Schnering, H. G. *Chem. Ber.* **1992**, *125*, 1913–1918.
- 27. Christl, M.; Lanzendörfer, U.; Grötsch, M. M.; Hegmann, J.; Ditterich, E.; Hüttner, G.; Peters, E.-M.; Peters, K; von Schnering, H. G. *Chem. Ber.* **1993**, *126*, 797–802.
- 28. Bentley, R. J. Am. Chem. Soc. 1948, 70, 2183-2185.
- 29. Tempesti, E.; Giuffre, L.; Sioli, G.; Fornaroli, M.; Airoldi, G. J. Chem. Soc. Perkin Trans. *1* 1974, 771–773.
- 30. Poon, N. L.; Satchell, D. P. N. J. Chem. Res. Synop. 1985, 260-261.
- 31. Poon, N. L.; Satchell, D. P. N. J. Chem. Res. Synop. 1983, 182-183.
- 32. Hurd, C. D.; Kimbrough, R. D., Jr. J. Am. Chem. Soc. 1961, 83, 236-240.

- 33. Mulzer, J.; Trauner, D.; Bats, J. W. Angew. Chem., Int. Ed. Engl. 1996, 35, 1970-1972.
- 34. Blomquist, A. T.; Holley, R. W.; Sweeting, O. J. J. Am. Chem. Soc. 1947, 69, 2356-2358.
- 35. von Seggern, H.; Schmittel, M. Chem. Ber. 1994, 127, 1269-1274.
- 36. Boivin, J.; El Kaim, L.; Zard, S. Z. Tetrahedron Lett. 1992, 33, 1285-1288.
- 37. Boivin, J.; El Kaim, L.; Zard, S. Z. Tetrahedron 1995, 51, 2573-2584.
- 38. Boivin, J.; El Kaim, L.; Zard, S. Z. Tetrahedron 1995, 51, 2585-2592.
- 39. Brady, W. T.; Owens, R. A. Tetrahedron Lett. 1976, 1553-1556.
- 40. Ogata, Y.; Adachi, K. J. Org. Chem. 1982, 47, 1182-1184.
- 41. Taggi, A. E.; Hafez, A. M.; Wack, H.; Young, B.; Ferraris, D.; Lectka, T. J. Am. Chem. Soc. 2002, 124, 6626–6635.
- 42. Taggi, A. E.; Hafez, A. M.; Wack, H.; Young, B.; Drury W. J., III; Lectka, T. J. Am. Chem. Soc. 2000, 122, 7831–7832.
- 43. Wack, H.; Taggi, A. E.; Hafez, A. M.; W. Drury, III, J.; Lectka, T. J. Am. Chem. Soc. 2001, 123, 1531–1532.
- 43a. France, S.; Wack, H.; Taggi, A. E.; Hafez, A. M.; Wagerle, T. R.; Shah, M. H.; Dusich, C. L.; Lectka, T. J. Am. Chem. Soc. 2004, 126, 4245–4255.
  - 44. Fehlner, T. P. J. Phys. Chem. 1972, 76, 3532-3538.
  - 45. Weber, Schnieder, M.; Maciel, T. C.; Wartig, H. B.; Schimmel, M. Organometallics 2000, 19. 5791–5794.
  - 46. Tajima, N.; Tsuneda, T.; Toyama, F.; Hirao, K. J. Am. Chem. Soc. 1998, 114, 8222-8229.
  - 47. Zhao, J.; Zhang, R. Atmos. Environ. 2004, 138, 2177-2185.
  - 48. Bernardi, F.; Cacace, F.; de Petris, G.; Pepi, F.; Rossi, I.; Troianc, A. *Chem. Eur. J.* **2000**, 537–544.

# 5.6.2 Oxygenation of Ketenes

Diphenylketene (1) was observed to react with oxygen when it was first prepared, and was found by Staudinger et al. in 1925 to form a polyester product **3**, which was proposed to arise from an initial cyclic peroxide **2** (equation 1).¹ Later study showed that the same product was formed from the bis(perester) **4**, which was proposed to form the radical **5**, which cyclized to the  $\alpha$ -lactone **6** (equation 2).² It was suggested that the conversion of **1** to **3** also proceeded through **6**, which formed as shown in equations 3 and 4.²





The reaction of **1** with ozone at -78 °C also gave **3**, and this reaction was also interpreted as proceeding through the  $\alpha$ -lactone **6**, which could be captured by reaction with MeOH as the  $\alpha$ -methoxy acid **9** (equation 5).³ Addition of ozone to di*tert*-butylketene (**10**) gave the  $\alpha$ -lactone **11**, which was identified by its ¹H NMR spectrum at -60 °C and which formed the polymer **12** on warming (equation 6).³



Upper limits for the rate constants of reactions of ozone with ketene (13), methylketene (14), ethylketene, and dimethylketene (15) gave relative rates of 1, 700,  $10^3$ , and  $4 \times 10^4$ , respectively.⁴ Wall and radical reactions complicated the analysis, but the reaction of dimethylketene (15) forming the trioxide 16 with zwitterionic and singlet diradical character leading to diradical 17 was proposed (equation 7). Transfer of oxygen between 15 and 17 led to the  $\alpha$ -lactone 18, which underwent decarbonylation to the ketone 19 (equation 8).⁴ Enhancement of the reaction by methyl substituents was contrary to what was found for nucleophilic attack at C₁, but was consistent with reactions involving a transition state with positive charge or free radical development at C₂.



Electron-transfer oxygenation of arylalkylketenes **20** initiated by aminium salts gave succinic anhydrides **22** in a process proposed to involve combination of **20** with  $\alpha$ -lactones **21** formed from the ketenes (equation 9).⁵



Computer simulation of an 85-step mechanism of the reaction of  $CH_2=C=O$  (23) with O₂ in the temperature range 1050–2050 K was used to derive rate expressions for steps including ketene oxygenations (equations 10, 11).⁶ Some of these reactions were also involved in acetylene combustion,⁷ and the mechanism of the reaction of C₂H with O₂ has been studied theoretically.^{8–10}

Diphenylketene (1) reacted with singlet oxygen generated from (PhO)₃PO₃, and in the presence of 9,10-bis(phenylethynyl)anthracene (24) there was chemiluminescent emission from 24, with formation of Ph₂C=O and CO₂ (equation 12).¹¹ It was proposed that Staundinger's peroxide 2 was formed. This was identified by the distinctive IR absorption at 1870 cm⁻¹ and was found to react with 24, forming the observed products and excited 24.¹¹

 $\alpha$ -Peroxylactones such as 27 were also formed from  ${}^{1}O_{2}$  reaction with ketenes (Me₂C=C=O (15), *t*-BuCH=C=O, *n*-BuCH=C=O, *n*-PrCMe=C=O, *n*-BuCPh=C=O, and (CF₃)₂=C=O) as well as with 1, and these were purified at  $-20 \,^{\circ}$ C and the structures confirmed by their  1 H NMR and IR spectra (equation 13).^{12,13} Kinetic measurements and reaction of the ketenes with  ${}^{1}O_{2}$  were consistent with the proposed reaction pathway (equation 13).¹² Reaction in the presence of MeOH resulted in the formation of  $\alpha$ -methoxyperacetic acid 28 (equation 14), so

the reactions were interpreted as involving either a perepoxide 25 or a zwitterion 26, and not direct formation of the  $\alpha$ -peroxylactone 25 (equation 14).¹⁴



Additional study^{14,15} of the reactions of ketenes with triplet oxygen confirmed and extended the work of Staudinger et al.¹ Reaction of Me₂C=C=O (**15**) with ³O₂ in ether at -20 °C gave the poly(peroxyester) **29** (96%) and the polyester **29a** (4%), while Ph₂C=C=O (**1**) gave no poly(peroxyester), 70% polyester **3**, and 30% Ph₂C=O and CO₂.¹⁴ Reaction of Me₂C=C=O (**15**) in MeOH gave HOOCMe₂. CO₂Me as the only major product, while Ph₂C=C=O (**1**) gave a significant amount of Ph₂C(OMe)CO₃H.¹⁴ These results and others were interpreted as showing that Me₂C=C=O (**15**) reacted with ³O₂ to form diradical **30**, which polymerized to **29** (equation 15), whereas ¹O₂ formed perepoxide **25** and/or zwitterion **26**, which were captured by MeOH (equations 13, 14). Diradicals such as **30** can undergo conversion to **25/26**, and the poly(peroxyesters) can decompose to ketones and CO₂. The diradical **31** from Ph₂C=C=O (**1**) formed a zwitterion analogous to **25/26**, which reacted with **1** to form the α-lactone **2**, which gave the polyester **3** (equation 16).



The conversion of other ketenes to ketones by oxygen has been observed,^{16–18} as well as the reactions of ketenes with ozone and peracids forming polyesters and ketones.^{19–21} The perfluorinated ketene **32** gave the stable  $\alpha$ -lactone **33** (equation 17).²²

$$(C_{2}F_{5})_{2}(CF_{3})C \longrightarrow C=0 \xrightarrow{NaOCl} (C_{2}F_{5})_{2}(CF_{3})C \longrightarrow O$$

$$C_{2}F_{5}(CF_{3})CF \longrightarrow O$$

$$C_{2}F_{5}(CF_{3})CF \longrightarrow O$$

$$(17)$$

The reaction of diphenylketene (1) with DMSO and aqueous acid was proposed to occur with nucleophilic attack at C₁ forming **34**, which gave benzilic acid (equation 18).²³ Iodosobenzene (PhIO) reacts with **1** to form polyester **3**,²⁴ and this presumably also occurs via nucleophilic attack at C₁, followed by formation of  $\alpha$ -lactone **2**.



#### **REFERENCES FOR SECTION 5.6.2**

- Staudinger, H.; Dyckerhoff, K.; Klever, H. W.; Ruzicka, L. Helv. Chem. Ber. 1925, 58, 1079–1087.
- 2. Bartlett, P. D.; Gortler, L. B. J. Am. Chem. Soc. 1963, 85, 1864-1869.
- 3. Wheland, R.; Bartlett, P. D. J. Am. Chem. Soc. 1970, 92, 6057-6058.
- 4. Hatakeyama, S.; Honda, S.; Akimoto, H. Bull. Chem. Soc. Jpn. 1985, 58, 2411-2412.
- 5. Schmittel, M.; von Seggern, H. Liebigs Ann. 1995, 1815-1821.
- 6. Hidaka, Y.; Kimura, K.; Hattori, K.; Okuno, T. Combust. Flame 1996, 106, 155-167.
- 7. Hidaka, Y.; Hattori, K.; Okuno, T.; Inami, K.; Abe, T.; Koike, T. *Combust. Flame* **1996**, *106*, 401–417.
- 8. Sumathi, R.; Peeters, J.; Nguyen, M. T. Chem. Phys. Lett. 1998, 287, 109-118.
- Hidaka, Y.; Nishimori, T.; Sato, K.; Henmi, Y.; Okuda, R.; Inami, K.; Higashihara, T. Combust. Flame 1999, 117, 755-776.
- Washida, N.; Hatekeyama, S.; Takagi, M.; Kyogoku, T.; Sato, S. J. Chem. Phys. 1983, 78, 4533–4540.
- 11. Bollyky, L. J. J. Am. Chem. Soc. 1970, 92, 3230-3232.
- Turro, N. J.; Ito, Y.; Chow, M.; Adam, W.; Rodriquez, O.; Yany, F. J. Am. Chem. Soc. 1977, 99, 5836–5838.
- 13. Turro, N. J.; Chow, M.-F. J. Am. Chem. Soc. 1980, 102, 5058-5064.
- 14. Turro, N. J.; Chow, M.-F.; Ito, Y. J. Am. Chem. Soc. 1978, 100, 5580-5582.
- 15. Bartlett, P. D.; McCluney, R. E. J. Org. Chem. 1983, 48, 4165-4168.
- 16. Majerski, Z.; Vinkovic, V. Synthesis 1989, 559-560.
- 17. Toda, F.; Todo, Y.; Todo, E. Bull. Chem. Soc. Jpn. 1976, 49, 2645-2646.
- Aoygi, R.; Tsuyki, T.; Takai, M.; Takahshi, T.; Kohen, F.; Stevenson, R. *Tetrahedron* 1973, 29, 4331–4340.
- 19. Crandall, J. K.; Sojka, S. A.; Komin, J. B. J. Org. Chem. 1974, 39, 2172-2175.
- 20. Moriarty, R. M.; White, K. B.; Chin, A. J. Am. Chem. Soc. 1978, 100, 5582-5584.
- 21. Brady, W. T.; Saidi, K. Tetrahedron Lett. 1978, 721-722.
- Coe, P. L.; Sellars, A.; Tatlow, J. C.; Whittaker, G.; Fielding, H. C. J. Chem. Soc. Chem. Commun. 1982, 362–363.

- 23. Lillien, I. J. Org. Chem. 1964, 29, 1631-1632.
- 24. Moriarty, R. M.; Gupta, S. C.; Hu, H.; Berenschot, D. R.; White, K. B. J. Am. Chem. Soc. **1981**, *103*, 686-688.

## 5.7 RADICAL REACTIONS OF KETENES

A theoretical study of the reaction of H, CH₃, OH, F, SiH₃, and Cl radicals with CH₂=C=O (1) forming the products 2a-c by addition to C₂, C₁, and oxygen of the ketene, respectively, gave good agreement with available data on the relative stabilities of the products and of the barriers for attack (Table 5.19).¹ The results for H• and CH₃• were similar, with little selectivity for attack at C₁ and C₂, and there was little difference between the relative energies of the isomeric products. For HO•, F•, and Cl• the product enolic radicals from attack at C₁ were significantly more stable than the products for attack at C₂, but small barriers for attack at C₁ were calculated, while none were found for attack at C₂, so selectivity for attack at C₂ was predicted, in agreement with experimental results. For SiH₃• a preference for attack at C₂ was predicted, and this radical was also predicted to be the most stable. These studies were confirmed by further calculations for H•,² Cl•,³ OH•,⁴ and F•⁵ radical addition to ketene and product radicals.



The results in Table 5.19 show barriers for rearrangement from the less stable enolic radical **2b** to acyl radical **2a** of 42.1 kcal/mol for R = H, and for rearrangement of the less stable acyl radicals **2a** to the more stable enolic radical **2b** of 27.6 and 4.6 kcal/mol for R = F and Cl, respectively.

R	$\operatorname{RCH}_{2}^{\bullet}\operatorname{C=O}(2\mathbf{a})$		$\begin{bmatrix} H & R \\ H & & \\ H & & \\ \end{bmatrix}^{\ddagger}$	•CH ₂	•CH ₂ C(O)R ( <b>2b</b> )		$CH_2 = COR (2c)$	
	$\Delta E_{\rm ts}$	$\Delta E^0$	$\Delta E_{ m ts}^{ m Rearr}$	$\Delta E_{\rm ts}^{\rm Perp}$	$\Delta E_{\rm ts}^{\rm in-pla}$	ne $\Delta E^0$	$\Delta E_{\rm ts}$	$\Delta E^0$
Н	0.5	-44.2	2.6	5.4	6.6	-39.5	13.4	-9.3
CH ₃	9.3	-29.7		11.9	8.5	-31.4	22.5	10.0
OH		-32.6		3.8		-54.7		
F		-50.1	-22.5			-69.2		
SiH ₃	3.3	-20.6		9.2	6.3	-4.2	11.2	-12.9
Cl		-20.5	-15.9	-13.4		-28.6		

TABLE 5.19 B3LYP/6-31G*//B3LYP/6-31G* Calculated Energy Changes (kcal/mol) for Additions of Radicals  $R \bullet$  to  $CH_2=C=O^a$ 

^{*a*}Energy differences relative to the separated reactants in each case.

The reactions of F, Cl, and OH radicals with ketene were monitored using a flow system with far IR detection of the reaction products.⁶ The predominant pathways observed with F• were radical addition to the CH₂ group, forming **3** followed by decarbonylation, and hydrogen abstraction forming the ketenyl radical **4** (equation 2).⁶ The radicals Cl• and HO• behaved similarly.^{7–10} Experimentally, reaction of ketene with H• occurred by addition to C₂, forming the radical **5**, and led to formation of CH₃• (equation 3).¹¹ The rate constants for addition of F•,⁷ Cl•,⁷ and H•¹¹ to ketene were measured. Reaction with O atoms with several ketenes occurred primarily at C₁, leading to formation of CO₂, and for **1** could involve an intermediate **6** with diradical or zwitterionic character (equation 4).¹²

$$\dot{c} = C = O \xrightarrow{F \cdot}_{HF} \stackrel{H}{\xrightarrow{}_{H}} C = O \xrightarrow{F \cdot}_{H} \stackrel{F}{\xrightarrow{}_{H}} \dot{c}_{z_{O}} \xrightarrow{-CO}_{H_{2}F} (2)$$

The kinetics of the reactions of the radicals  $\bullet$ CN and  $\bullet$ N=C=O with ketene in the gas phase were measured, and occurred predominantly by C-C bond formation followed by decarbonylation yielding NCCH₂ $\bullet$  and  $\bullet$ CH₂NCO, respectively.¹³ Hydrogen abstraction forming HCN and HNCO, respectively, together with the radical  $\bullet$ CH=C=O, was a minor or negligible reaction channel.¹³

The kinetics of addition of phenyl radicals to ketene were measured by monitoring the IR spectrum of the phenyl radical, and showed that the reaction occurred by addition and not hydrogen abstraction, as there was no measurable isotope effect for reaction of  $CD_2=C=O$ .¹⁴ Computational studies indicated that addition could occur at either C₁ or C₂, forming either **7** or **8**, respectively, and the radicals formed could interconvert with a low low barrier leading to decarbonylation with formation of PhCH₂• (equation 5).¹⁴

Hydrogen atoms reacted with methylketene, dimethylketene, and ethylketene (9) forming alkyl radicals and CO, and evidently led to radical intermediate 10

(equation 6).^{15,16}

Addition of H• radical to di-*tert*-butylketene (11) in solution gave both the acyl radical 12 and the enolic radical 13, as identified by ESR (equation 7).¹⁷ The acyl radical was assigned as having a  $\sigma$  structure and decayed rapidly by decarbonylation, forming *t*-Bu₂CH•. Addition of C₆F₅ or CF₃ radicals to di-*tert*-butylketene (11) gave the radicals 14, which were assigned as having perpendicular conformations from their ESR spectra (equation 8).¹⁷ This acyl radical conformation was favored because of the steric crowding in a planar conjugated structure.

Ethanethiol added to dimethylketene (**9b**) in solution in a radical initiated chain process with radical attack at  $C_1$  forming the intermediate **15**, which formed the thioester **16** (equation 9).¹⁸ Radical addition was proposed to occur to  $C_1$  of diphenylketene (**17**) formed by thermolysis of azabenzil, giving rise to radical **18**, detected by ESR (equation 10).¹⁹

The addition of the nitroxyl radical  $(CF_3)_2NO\bullet$  to  $Ph_2C=C=O$  (17) formed the 1.2-diaddition product 19 (equation 11).²⁰ The less reactive nitroxyl radical 2,2,6,6-tetramethylpiperidinyloxy (TEMPO, TO•) also reacted with diphenylketene (17) but did not add a second TEMPO (equation 12).²¹ On standing in air, reaction

with oxygen formed the peroxide **22**, whose structure was confirmed by X-ray determination.²¹ This result indicated that the nitroxyl radical added initially at  $C_1$  of the ketene, forming the intermediate radical **20**, which was presumably in equilibrium with the unsymmetrical dimer, and upon exposure to  $O_2$  formed the peroxide (equations 12, 13).²¹



Computations at the B3LYP/6-31G* level indicated that while addition of HO• to ketene to form HOCH₂C(•)=O and •CH₂CO₂H was predicted to be exothermic by 32.6 and 54.7 kcal/mol, respectively (Table 5.19),¹ the corresponding reactions of aminoxyl radical (H₂NO•) or of the 2,2,6,6-tetramethylpiperidinyloxy (TEMPO, TO•) were calculated to be much less exothermic, but still with a strong preference for addition at C₁ compared to C₂, with  $\Delta E$  of 4.8 and 23.6 kcal/mol for formation of **23** and **24**, respectively (equation 14).²² Thus, addition at C₁ was predicted to be favored, in agreement with with the experimental results for **17**.²¹

$$TO \xrightarrow{H}_{H} \dot{C}_{\approx O} \xrightarrow{TO.}_{AE = 23.6} \xrightarrow{H}_{H} C=O \xrightarrow{TO.}_{AE = 4.8} \xrightarrow{H}_{Kcal/mol} OT$$
(14)  
24 1 23

At elevated temperatures TEMPO reacted with the bisketene **25** to give the anhydride **27** in a reaction suggested to involve radical attack on the carbonyl carbon (equation 15).²¹ The reactions of a number of other ketenes with TEMPO have been studied, and a linear correlation (Figure 5.5) between  $\log k_2$ (TO•) and the measured rate constants for hydration  $\log k(H_2O)$  at 25 °C was found (equation 16).^{22–24}





**Figure 5.5** Log $k_2$ (TEMPO) versus log  $k(H_2O)$  for ketenes.

Phenylketene (28) generated in situ by Wolff rearrangement or dehydrochlorination of the acyl chloride in the presence of TEMPO gave 1,2-diaddition of two molecules of TEMPO to form 30 (equation 17).²³ Butadienylketene (31) reacted by 1,6-addition (equation 18),³² while dienylketene 33 was proposed to form the observed ester 35 by hydrogen abstraction from an intermediate radical 34 (equation 19).³⁰ Pentafulvenone (36) formed 37 and 38 by complex reaction pathways (equation 20).²⁸ Other ketenes whose reactions with TEMPO were examined include 1-naphthylketene,²⁵ cyclopropylketene,²⁷ 4-butenylketene,²⁶ allenylketene,³² and (phenylethynyl)ketene.²⁶



Cyclization of **39** with radical addition to the ketenyl group forming **40** was studied computationally for comparison to experimental results (equation 21).³³ Other intramolecular reactions of radicals with ketenes are shown in Section 4.1.9.

$$( \bigcirc C_{\leq 0} \longrightarrow (21)$$

Addition of the nitrate radical to ketene (1) was proposed to occur by formation of the acyl radical **41**, which fragmented to formaldehyde (equation 22).³⁴ Ketene (1) was formed during the reaction of 2-butyne with NO₃ and was proposed to arise from acetyl nitrate **42**, which eliminated nitric acid (equation 23).³⁴



Bisketene **43** reacted with NO by [4+1] cycloaddition, forming the nitroxyl radical **44**, which led to **45** (equation 24), and with NO₂, forming phthalic anhydride through the proposed intermediate **46** (equation 25).³⁵



Fluorine atoms reacted with carbon suboxide **46a** to form CO. The possible first steps may have involved F addition or CO abstraction (equation 26).³⁶

$$F \longrightarrow O^{\bullet} \xrightarrow{F^{\bullet}} O = C = C = C = O \xrightarrow{F^{\bullet}} F \xrightarrow{F^{\bullet}} C_{\geq O} + C = C = O$$
(26)  
46a

The distonic species **47** formed in the gas phase from cyclobutanone added to dideuteroketene  $(1-d_2)$ , forming **48**, which then lost CO (equation 27).³⁷ Reaction of CH with ketene to give CO, C₂H₂, and H• was proposed to proceed by insertion, forming intermediate **49** (equation 28).³⁸



It was proposed that during polymerization of *N*-vinylpyrrolidinone in 3-methylbutan-2-one, enolic radicals **50** were formed that cleaved, forming ketene and isopropyl radicals, and that the ketene subsequently was attacked by the polymerizing chain **P**•, forming acyl radical **51**, which abstracted hydrogen, giving aldehyde **52** (equation 29).³⁹ Anionic polymerization of ethyl(phenyl)ketene was initiated by the nitroxyl radical **53**, forming a polymer with a nitroxyl end group.⁴⁰ Heptafulvenone **54** reacted with TEMPO (TO•) to form the dimer **55** and the *o*-, *m*-, and *p*substituted benzaldehydes **56** by radical processes (equation 30).³²



#### **REFERENCES FOR SECTION 5.7**

- 1. Sung, K.; Tidwell, T. T. J. Org. Chem. 1998, 64, 9690-9697.
- 2. Lee, J.; Bozzelli, J. W. Int. J. Chem. Kinetics 2002, 35, 20-44.
- 3. Hou, H.; Wang, B.; Gu, Y. J. Phys. Chem. A 2000, 104, 320-328.

- 4. Hou, H.; Wang, B.; Gu, Y. Phys. Chem. Chem. Phys. 2000, 2, 2329-2334.
- Cao, D.-B.; Ding, Y.-H.; Li, Z.-S.; Huang, X.-R.; Sun, C.-C. J. Phys. Chem. A 2002, 106, 8917–8924.
- 6. Grussdorf, J.; Nolte, J.; Temps, F.; Wagner, H. Gg. Ber. Bunsenges. Phys. Chem. 1994, 98, 546–553.
- 7. Ebrecht, J.; Hack, W.; Wagner, H. Gg. Ber. Bunsenges. Phys. Chem. 1990, 94, 587-593.
- Hatakeyama, S.; Honda, S.; Washida, N.; Akimoto H. Bull. Chem. Soc. Jpn. 1985, 58, 2157–2162.
- 9. Brown, A. C.; Canosa-Mas, C. E.; Parr, A. D.; Wayne, R. P. Chem. Phys. Lett. 1989, 161, 491-496.
- Oehlers, C.; Temps, F.; Wagner, H. G.; Wolf, M. Ber. Bunsenges. Phys. Chem. 1992, 96, 171–175.
- 11. Michael, J. V.; Nava, D. F.; Payne, W. A.; Stief, L. J. J. Chem. Phys. 1979, 70, 5222-5227.
- 12. Washida, N.; Hatakeyama, S.; Takagi, H.; Kyogoku, T.; Sato, S. *J Chem. Phys.* **1983**, 78, 4533–4540.
- 13. Edwards, M. A.; Hershberger, J. F. Chem. Phys. 1998, 234, 231-237.
- 14. Choi, Y. M.; Lin, M. C. ChemPhysChem 2004, 5, 225-232.
- Umemoto, H.; Tsunashima, S.; Sato, S.; Washida, N.; Hatakeyama, S. Bull. Chem. Soc. Jpn. 1984, 57, 2578–2580.
- 16. Slemr, F.; Warneck, P. Ber. Bunsenges. Phys. Chem. 1975, 79, 152-156.
- 17. Malatesta, V.; Forrest, D.; Ingold, K. U. J. Phys. Chem. 1978, 82, 2370-2373.
- 18. Lillford, P. J.; Satchell, D. P. N. J. Chem. Soc. B 1970, 1303-1305.
- 19. Singer, L. S.; Lewis, I. C. J. Am. Chem. Soc. 1968, 90, 4212-4218.
- 20. Banks, R. E.; Haszeldine, R. N.; Murray, M. B. J. Fluorine Chem. 1981, 17, 561-564.
- 21. Huang, W.; Henry-Riyad, H.; Tidwell, T. T. J. Am. Chem. Soc. 1999, 121, 3939-3943.
- 22. Allen, A. D.; Henry-Riyad, H.; Tidwell, T. T. Arkivoc 2002 (xii), 63-74.
- Allen, A. D.; Cheng, B.; Fenwick, M. H.; Huang, W.; Missiha, S.; Tahmassebi, D.; Tidwell, T. T. Org. Lett. 1999, 1, 693–696.
- Allen, A. D.; Fedorov, A. V.; Najafian, K.; Tidwell, T. T.; Vukovic, S. Pure Appl. Chem. 2004, 76, 1953–1966.
- 25. Carter, J.; Fenwick, M. H.; Huang, W.; Popik, V. V.; Tidwell, T. T. *Can. J. Chem.* **1999**, 77, 806–809.
- Allen, A. D.; Fenwick, M. F.; Givehchi, B.; Henry-Riyad, H.; Nikolaev, V. A.; Shikova, E. A.; Tahmassebi, D.; Tidwell, T. T.; Wang, S. J. Org. Chem. 2001, 66, 5759–5765.
- 27. Allen, A. D.; Fenwick, M. H.; Henry-Riyad, H.; Tidwell, T. T. J. Org. Chem. 2001, 66, 5759–5765.
- 28. Allen, A. D.; Porter, J.; Tahmassebi, D.; Tidwell, T. T. J. Org. Chem. 2001, 66, 7420-7426.
- Allen, A. D.; Fenwick, M. H.; Jabri, A.; Rangwala, H.; Saidi, K.; Tidwell, T. T. Org. Lett. 2001, 3, 4095–4098.
- 30. Saidi, K.; Tidwell, T. T. Arkivoc 2001, (iii), 100-107.
- 31. Allen, A. D.; Rangwala, H.; Saidi, K.; Tidwell, T. T.; Wang, J. Russ. Chem. Bull. 2001, 2130–2133.
- 32. Tidwell, T. T.; Fenwick, M. H. Eur. J. Org. Chem. 2001, 3415-3419.
- 33. Yamamoto, Y.; Ohno, M.; Eguchi, S. J. Org. Chem. 1996, 61, 9264-9271.

- 34. Seland, J. G.; Noremsaune, I. M. W.; Nielsen, C. J. J. Chem. Soc. Faraday Trans. 1996, 92, 3459–3465.
- 35. Adam, W.; Bottle, S. E.; Grice, I. D.; Pfeiler, D.; Wentrup, C. *Tetrahedron Lett.* **1996**, *37*, 2113–2116.
- 36. Lange, W.; McCrumb, J. L.; Wagner, H. G.; Walther, C. D. Ber. Bunsen-Gesellsch. 1977, 81, 720–724.
- 37. Smith, R. L.; Chyall, L. J.; Chou, P. K.; Kenttämaa, H. I. J. Am. Chem. Soc. 1994, 116, 781–782.
- 38. Hancock, G.; Heal, M. R. J. Chem. Soc. Faraday Trans. 1992, 88, 2121-2128.
- 39. Liu, Z.; Rimmer, S. Macromolecules 2002, 35, 1200-1207.
- 40. Sudo, A.; Uchino, S.; Endo, T. J. Polym. Sci. A 2000, 38, 1073-1082.

# 5.8 POLYMERIZATION OF KETENES

Polymers derived from ketenes were first observed by Wedekind¹ and by Staudinger,^{2,3} and this provided an impetus for the further development of polymer chemistry. The formation of high polymers from ketenes was examined systematically by Natta, et al.,^{4,5} and this work has been reviewed.^{6–9} Types of ketene polymers include those formed by 1,2-additions of the C=C bonds (1) and of the C=C bonds (2), which are poly(ketones) and poly(acetals), respectively, and esters 3 that have both types of linkages. All three have been prepared from Me₂ C=C=O (4) by appropriate choice of catalyst and solvent (equations 1–3).⁴



The formation of **1** by polymerization of **4** with AlBr₃ in toluene was proposed to involve formation of enolate **5**, which added **4** (equation 1), while **2** was formed from **4** by *n*-BuLi in ether (equation 2). The formation of **3** took place when AlEt₃ in toluene was used as the catalyst (equation 3).^{4,7,10,11}



Ketene was polymerized by La(OPr-i)₃, probably in a process involving a lanthanum enolate.¹² This species served to initiate methyl methacrylate polymerization.¹² This catalyst also formed a copolymer of diethylketene and methyl methacrylate.¹³ Cationic polymerization of dimethylketene has been reported,¹⁴ as well as radiation cryopolymerization of ketene, which was proposed to proceed by an anionic mechanism.^{14a}

Arylalkylketenes **6** have also been polymerized,  $^{15-21}$  and anionic living polymerization initiated with alkoxy anion **7** containing a stable tetramethylpiperidinyl radical gave **8** tagged with a radical end group (equation 4).¹⁵



Polymerization of phenyl(ethyl)ketene 9 in THF at -20 °C initiated by *n*-BuLi gave a living polymer 10 that could be quenched by MeOH.¹⁶ Alternatively, the addition of further ketene to the living polymer gave renewed polymerization, or the addition of *tert*-butyl methacrylate (11) gave block polymer 12 (equation 5).¹⁹

$$\begin{array}{c} Ph \\ Et \end{array} \xrightarrow{R-BuLi} C = 0 \xrightarrow[-20\ \circ C]{hexane} \\ 9 \\ \end{array} \begin{array}{c} Et \\ -20\ \circ C \end{array} \xrightarrow{Ph} 0 \\ Et \\ Ph \\ Et \\ Ph \\ \end{array} \xrightarrow{Ph} 0 \\ Et \\ Ph \\ \end{array} \begin{array}{c} CO_2Bu-t \\ (11) \\ (11) \\ Et \\ Ph \\ (11) \\ Et \\ Ph \\ He \\ CO_2Bu-t \end{array} \begin{array}{c} CO_2Bu-t \\ O \\ Et \\ Ph \\ He \\ CO_2Bu-t \end{array}$$
(5)

Diphenylketene has also been polymerized using EtLi or *t*-BuLi.²¹ Ketenes also form copolymers with other carbonyl compounds^{22,23} or with isocyanates.²⁴

Polyketene (15), the polymer of ketene, was formed by the reaction of acetyl chloride with AlCl₃, and was proposed to result from deprotonation of the acylium ion 13 forming ketene, which then reacted with 13 forming 14, followed by addition of ketene monomers leading to 15 (equation 6).²⁵ Polyketene has also been formed by acid polymerization of the ketene acetal  $CH_2=C(OHx-n)_2$  followed by mild hydrolysis of the resulting acetal.²⁶ Polyketene was found to be extensively enolized, with interesting conducting properties.²⁷

$$\begin{array}{c} 0 \\ CH_{3} \\ \hline Cl \end{array} \xrightarrow{AlCl_{3}} CH_{3} \\ \hline CH_{3} \\ CH_{3} \\ \hline CH_{3}$$



Ketoketenes **17** formed by photolysis of 2-diazo-1,3-diketones **16** in Et₂O polymerize to polyesters **18** (equation 8).²⁸ A variety of polymers have been prepared from carbon suboxide ( $C_3O_2$ ).^{29,30} Carbon suboxide spontaneously forms a low-molecular-weight polymer, as discussed in Section 4.11.³¹ The formation of polymers from other bisketenes is mentioned in Section 4.9.



# **REFERENCES FOR SECTION 5.8**

- 1. Wedekind, E. Liebigs Ann. Chem. 1902, 323, 246-257.
- 2. Staudinger, H.; Ruzicka, L. Liebigs Ann. Chem. 1911, 380, 278-303.
- 3. Staudinger, H. Helv. Chim. Acta 1925, 8, 306-332.
- 4. Natta, G.; Mazzanti, G.; Pregaglia, G. F.; Binaghi, M.; Peraldo, M. J. Am. Chem. Soc. **1960**, 82, 4742–4743.
- Natta, G.; Mazzanti, G.; Pregaglia, G. F.; Binaghi, M. J. Am. Chem. Soc. 1960, 82, 5511– 5512.
- 6. Zarras, P.; Vogl, D. Prog. Polym. Sci. 1991, 16, 173-201.
- Pregaglia, G. F.; Binaghi, M. In *Encyclopedia of Polymer Science and Technology*; Wiley, New York, 1968, Vol 8, pp. 45–57.
- Pasquon, I.; Porri, L.; Giannani, U. In *Encyclopedia of Polymer Science and Technology*, 2nd ed.; Wiley, New York, 1989, Vol. 15, pp. 713–715.
- Pregaglia, G. F.; Binaghi, M. In *The Stereochemistry of Macromolecules*; Ketley, A. D., Ed.; Dekker, New York, 1967, Vol. 2, pp. 111–175.
- 10. Yamashita, Y.; Miura, S.; Nakamura, M. Makromol. Chem. 1963, 68, 31-47.
- 11. Cash, G. O., Jr.; Martin, J. C. U.S. Patent 3,321,441; Chem. Abstr. 1968, 68, 13684e.
- 12. Sugimoto, H.; Kato, M.; Inoue, S. Macromol. Chem. Phys. 1997, 198, 1605-1610.
- 13. Kato, M.; Sugimoto, H.; Inoue, S. Kidorui 1997, 30, 62–63. Chem. Abstr. 1997, 127, 66386.
- 14. Egret, H.; Couvercelle, J. P.; Belleney, J.; Bunel, C. Eur. Polym. J. 2002, 38, 1953-1961.
- 14a. Kichigina, G. A.; Kiryukhin, D. P.; Barkalov, I. M. Polym. Sci. Ser. 2002, 44, 163–166. Chem. Abstr. 2002, 138, 73652.
- 15. Sudo, A.; Uchino, S.; Endo, T. Macromolecules 1999, 32, 1711-1715.
- 16. Sudo, A.; Uchino, S.; Endo, T. J. Polym. Sci. A. 2001, 39, 2093-2102.
- 17. Sudo, A.; Uchino, S.; Endo, T. J. Polym. Sci. A. 2001, 39, 1596-1600.
- 18. Sudo, A.; Endo, T. J. Polymer Sci. A, Polymer Chem. 2002, 40, 3103-3111.
- Sudo, A.; Iinuma, R.; Endo, T. J. Polymer Sci. A, Polymer Chem. 2002, 40, 3455– 3459.

- 20. Sudo, A.; Uchino, S.; Endo, T. J. Polymer Sci. A, Polymer Chem. 2000, 38, 1073-1082.
- 21. Nadzhimutdinov, Sh.; Cherneva, E. P.; Kargin, V. A. Vysokmol. Soedin., Ser. B, 1967, 9, 480-485. Chem. Abstr. 1967, 67, 100431t.
- 22. Nunomoto, S.; Yamashita, Y. Kogyo Kagaku Zasshi 1968, 71, 2067-2072; Chem. Abstr. 1969, 70, 47947z.
- 23. Dyer, E.; Sinich, E. J. Polym. Sci., Part A-1 1973, 11, 1249-1260.
- 24. Higashi, H.; Harada, H. Kogyo Kagaku Zasshi 1966, 69, 2344–2345; Chem. Abstr. 1967, 66, 105236x.
- 25. Olah, G. A.; Zadok, E.; Edler, R.; Adamson, D. H.; Kasha, W.; Prakash, G. K. S. J. Am. Chem. Soc. **1989**, 111, 9123–9124.
- 26. Khemani, K. C.; Wudl, F. J. Am. Chem. Soc. 1989, 111, 9124-9125.
- 27. Cui, C. X.; Kertesz, M. J. Am. Chem. Soc. 1991, 113, 4404-4409.
- Nikolaev, V. A.; Frenkh, Yu.; Korobitsyna, I. K. Zh. Org. Khim. 1978, 14, 1147-1160. Engl. Transl. 1978, 14, 1069–1080.
- 29. Porejko, S.; Makauk, L.; Kepka, M. Polimery **1967**, *12*, 464-467. Chem. Abstr. **1968**, 69, 3211e.
- Sladkov, A. M.; Korshak, V. V.; Nepochatykh, V. P. *Izv. Akad. Nauk SSSR, Ser. Khim.* 1968, 196–197. *Engl. Transl.* 1968, 191–193.
- 31. Ballauff, M.; Rosenfeldt, S.; Dingenouts, N.; Beck, J.; Krieger-Beck, P. Angew. Chem. Int. Ed. 2004, 43, 5843–5846.

## 5.9 STEREOSELECTIVITY IN KETENE REACTIONS

Additions to ketenes often involve the generation of new chiral centers. The search for stereoselectivity in these reactions was initiated in 1919 by Weiss, who added the chiral alcohol menthol to phenyl-*p*-tolylketene and claimed to have observed diastereoselectivity in formation of the ester.¹ This was, however, improbable and was later disproved.^{2,3} Despite this initial setback, there has been continuing interest and rapidly increasing success in the search for methods to achieve stereoselectivity in ketene reactions, and the subject has been frequently reviewed.^{3–11}

Further studies of stereoselective addition of alcohols to ketenes have involved the use of chiral alcohols, chiral catalysts, and chiral auxiliaries. Appreciable stereoselectivity in the addition of methanol to phenylmethylketene (1) catalyzed by the chiral base brucine forming the ester **3** was achieved by Pracejus in 1960 (equation 1).¹² There was a 25% excess of the *S*-(+) ester at  $-110 \,^{\circ}$ C and a 10% excess of the *R*-(-) ester at 80 °C. Selective protonation of a zwitterionic intermediate **2** was proposed,^{5,12,13} but more recently, addition of the catalyst to the ketene forming an enolate that undergoes stereoselective protonation has been suggested for many but not all such reactions (equation 2).^{8,10,11} Optically active amines bound to polymers were also effective catalysts for this reaction.^{15,16} A chiral dendritic diamine gave enantioselective protonation in the reaction of PhCMe=C=O (**1**) with MeOH.¹⁷ The enantioselective protonation of enolates derived from ketenes has been reviewed.⁶ A chiral azaferrocene catalyst (vide infra) promoted the stereoselective addition of CH₃OH to PhCEt=C=O.^{10,11}



Addition of the chiral alcohol pantolactone (5) to the arylmethylketene **4** generated from the acyl chloride with a tertiary amine has been found to give diastereoselectivities as high as 99% in forming 2-arylpropionates **6** (equation 3).^{18,19} Many variations of the reaction of chiral alcohols with ketenes have been studied,^{20–33} and a detailed computational investigation has been carried out.^{30,31} Other unsymmetrical ketenes such as R₂NCR=C=O,^{24,25} arylhaloketenes RCHal=C=O,²⁸ and phenyltrifluoromethylketene (7) with chiral alcohol **8** forming **9** gave more modest stereoselectivity (equation 4).²⁹



The ketene 10 generated by dehydrochlorination reacted with (*R*)-pantolactone 5 to give the ester 11 in 85% yield, 94% *de* (equations 5, 6).³⁴



11 (85%, 94% de)


**Figure 5.6** Favored transition state calculated for proton transfer to carbon in the esterification of methyl(phenyl)ketene by methyl (*S*)-lactate catalyzed by trimethylamine (reprinted with permission of the American Chemical Society).

A computational study at the B3LYP/6-31G* level of the reaction of PhCMe=C=O (1) with methyl (*S*)-lactate lactate and Me₃N found initial complexes, TS for alcohol addition *syn* to the Me group of 1 forming a zwitterionic intermediates, and TS for proton transfer to the  $\alpha$ -carbon.³¹ The important final TS involves a favorable C–H–O hydrogen bond from the lactate carbonyl to a methyl C–H (Figure 5.6).³¹ This model accounts for the observed stereoselectivity of (*S*)-pantolactone to methyl(phenyl)ketene.

Additions of the polymer-bound alcohol **12** with ArCMe=C=O (**4**) gave esters **13** (equation 7), which were saponified to give the chiral acids in >99% yield and ee > 80%.³² The free chiral alcohol was also recovered from the reaction. Reaction of **12** with phthalimido ketenes **14** formed esters **15a**, which upon cleavage gave 2-aryl-3-amino propionic acids **15b** with 75–88% *ee* (equation 8).³³ Sugars reacted with aryl(methyl)ketenes **4**, forming esters that after hydrolysis gave  $\alpha$ -propionic acids with *dee* up to 78%.³⁵



Chiral ketenes **17** from Wolff rearrangements of amino acid–derived diazo ketones **16** reacted stereoselectively with  $CH_3OH$  in the generation of the new chiral center in **18** (equation 9).³⁶ Other chiral ketenes also gave stereoselective reaction with achiral alcohols.^{37,38}



In a pioneering study of the reaction of the chiral amines, *S*-(-)-1-phenylethylamine (**19**) with the ketene **1** in toluene at -100 °C gave the *S*,*S*-amide **20** with 60% *de* (equation 10).³⁹ The addition of the isopropyl ester of *R*-alanine (**22**) to ketene **21** gave the *RR*-amide **23** in 63% *de* (equation 11).³⁹ The dienylketene **25** generated by photolysis of **24** also gave stereoselectivity in reaction with *d*-ephedrine (equation 12).⁴⁰



The reaction of *R*-1-phenylethylamine (*R*-19) to bisketene **26** gave the ketenylamide **27** with significant selectivity (equation 13), and this was enriched to >90% *ee* with one recrystallization.⁴¹ The reaction of **27** with a second *R*-1-phenylethylamine (*R*-19) forming **28** also had 2/1 selectivity, and MeOH addition generating **29** proceeded with 96:4 *dr* (equation 14).⁴¹





Chiral amino acid esters added to ketenes ArCBr=C=O (**30**), generated in situ from 3-aryl-2,2-dicyanooxiranes, and gave amides with significant diastereoselectivity (equation 15).⁴² High diastereoselectivity (98:2) was also observed for the Et₂Zn-catalyzed reaction of (*S*)-(+)-3-hydroxytetrahydrofuran with phenyl(methy)-ketene (**1**), with 78% yield.⁴³



Phenyl(ethyl)ketene (**31**) and 2-cyanopyrrole in the presence of catalyst **32**^{10,11,44–46} gave stereoselective amination forming the amide (equation 16).⁴⁵ The stereoselectivity was proposed to arise from proton transfer to an amide enolate by the protonated catalyst.^{11,45} A number of ketenes ArCR=C=O and amines were examined, and the cyanopyrrole gave the highest stereoselectivity of the amines.⁴⁵



In studies of esterification of ketenes **31a** with phenols ArOH catalyst (-)-**32** gave esters **31b** with *ee* of 35–91%, with the highest *ee* with 2-*tert*-butylphenol (equation 16a).^{46a} A variety of ketenes with this phenol gave *ee* of 79–94% and yields of 66–97%. The mechanism proposed for this process involved deprotonation of the phenol, phenoxide attack on the ketene forming an enolate, and proton transfer from the resulting ion pair forming **38** (equation 16b).^{46a} Ketenes **31a** also reacts with aldehydes Ph₂CHCH=O with catalysis by (-)-**32** to give stereoselective formation of enol esters.^{46b}

$$Ar \xrightarrow{Ar} C=O + Ar^{1}OH \xrightarrow{3\% (-)-37} H \xrightarrow{Ar} Ar \xrightarrow{R} O \xrightarrow{O} O$$

$$31a \qquad 31b \qquad (16a)$$



The reaction of carboxylic acids with ketenes **33** catalyzed by chiral amines gave stereoselective formation of  $\alpha$ -aryl carboxylic acids **34** (equation 17).⁴⁷ Chiral alcohols were used to protonate enolate **36** formed from addition of arylthiophenoxides to ketene **35**, and gave stereoselectivity in the formation of **37** (equation 18).⁴⁸



Photolysis of the optically active amino-substituted chromium carbene complex **38** gave the chromium-complexed ketene **39**, which reacted with the chiral amino acid ester **40** to give the amide **41** with 98:2 *dr* (equation 19).⁴⁹ This procedure was adapted for solid-state peptide synthesis.⁵⁰ Stereoselective cyclization of the chromium-complexed ketene **42** formed by carbene complex photolysis gave the lactone **43** (equation 20).⁵¹ Other examples of stereoselective reactions of chromium-complexed ketenes have been reported.^{52,53}



Ketenes characteristically give high stereoselectivity in [2+2] cycloaddition reactions, so reactions with reagents containing preexisting chiral centers permit further stereochemical control. For example, dichloroketene **44** gave greater than

90% *ee* in cycloaddition with **45**, forming the cyclobutanone (equation 21).^{54,55} The selectivity was attributed to shielding of one face of the double bond by the phenyl.^{54,55} Enecarbamates **46** with chiral auxiliary groups R reacted with **44** with selectivity of up to 4:1 in the formation of diastereomeric products **47** (equation 22).⁵⁶ Dichloroketene also added selectively to the optically active carbohydrate enol ether **48**, forming **49** (equation 23).⁵⁷





Ketene **51** generated by photochemical Wolff rearrangement of diazo ketone **50** gave stereocontrolled intramolecular [2+2] cycloaddition (equation 24).⁵⁸ The cyclobutanone **52** was the only stereoisomer detected (>99% *de*) over the temperature range 195 to 423 K, and the lack of temperature dependence indicated the entropy control of the stereochemical outcome. The Wolff rearrangement of the ester was characteristically inefficient, with isolated yields of 10–33%, but generation of the ketene by an elimination route also gave **52** as the only stereoisomer.^{58–60} However, Wolff rearrangement by vacuum flash pyrolysis gave the product in 70% yield with undiminished stereoselectivity, as the intermediate carbene avoided intermolecular hydrogen atom transfer.^{59,60}



#### 610 REACTIONS OF KETENES

Methylketene **53** generated by dehydrochlorination reacted with allyl morpholines **54**, forming amides **55** with high selectivity for the isomer shown (equation 25).⁶¹ The reactions were envisioned as proceeding through a ketene aza-Claisen reaction reaction as shown, and were successful with catalysis by 5–10 mol%.  $Yb(OTf)_3$ ,  $AlCl_3$ ,  $Ti(OPr-i)_2Cl_2$ , or  $TiCl_4 \bullet (THF)_2$ . Nitrogen-, oxygen-, and sulfursubstituted ketenes were also successfully used in this procedure.



Allylic diamine **56** reacted successively with two molecules of methylketene (**53**), forming **57** in a cascade process depicted as occurring by two iterations of the aza-Claisen rearrangement (equation 26).⁶² The first reaction was considered to occur with the R₂N group *trans* to R¹. Reaction for R₂N = morpholine and R¹ = Me gave a 97% yield with a 98:2 preference for the product shown. The reaction was also successful for benzylketene. The catalysts **58** promoted this reaction with high stereoselectivity.⁶³ Use of the chiral catalysts **58** and **59** gave the product **57** in yields of up to 88% and up to 91% *ee*.⁶³



 $(R_2N = morpholine, pyrrolidine, piperidine; R^1 = Me, Cl, OBz, CN, SPh)$ 



Methylketene (**53**) generated by dehydrochlorination of propionyl chloride by diisopropylethylamine (DIPEA) underwent [2+2] cycloadditions with aldehydes catalyzed by Al(SbF₆)₃, forming  $\beta$ -lactones **60** with high *cis/trans* product ratios (equation 27).^{64–68} Stereoselectivity was achieved with the chiral catalyst **61**, giving

*ee* of 90–94%.^{65,66} This reaction was improved using benzotrifluoride as solvent, which resulted in precipitation of the ammonium salt.



High stereoselectivity in [2+2] cyclization was achieved with the chromiumcomplexed ketene **62** generated by photolysis of the carbene complex EtOC-Me=Cr(CO)₅, which reacted with the enamine **63** bearing a chiral substituent to form **64** (equation 28).⁶⁹ Asymmetric benzannulations of chromium carbene complexes with chiral auxiliaries have also been observed.⁷⁰



Ketenes gave effective [2+2] cycloaddition to aldehydes, and the reaction of  $CH_2=C=O$  (65) with chloral catalyzed by optically active tertiary amines proceeded with high stereoselectivity, forming the  $\beta$ -lactone 67 (equation 29).^{71–73} The stereoselectivity was attributed to formation of the chiral zwitterion 66 that reacts with chloral.



Reactions of ketene with aldehydes **68** to form oxetanones **69** catalyzed by chiral 1,1'-binaphthalene-2,2'-diol aluminum catalysts **70** gave up to 56% *ee*, while the catalyst **71** gave up to 92% *ee* (equation 30).^{74,75} Trimethylsilylketene (**73**) underwent [2+2] cycloaddition with aldehyde **72** promoted by a chiral catalyst  $Ar_2AlMe$ , forming **74** with high diastereoselectivity (equation 31).⁷⁶





Asymmetric synthesis using diethylketene (**75**) with catalyst (-)-**32** gave formation of the  $\beta$ -lactone with a quarternary carbon in 92% yield, 91% *ee* (equation 32).⁷⁷ The unsymmetrical ketene *i*-PrCMe=C=O (**76**) reacted with 91% *ee*, 48% yield, and 4.5:1 selectivity for the *cis*-lactone.⁷⁷

The asymmetric synthesis of  $\beta$ -lactams by [2+2] cycloaddition of ketenes with imines has received increasing attention.^{9,78} Phenylketene (**77**) underwent stereoselective [2+2] cycloaddition with the imine **78** catalyzed by the chiral base benzoylquinine, and with In(OTf)₃ as an electrophilic cocatalyst gave the  $\beta$ -lactam product **79** in 95% yield, 98% *ee*, and 60/1 *dr* (equation 33).⁷⁹ Mechanistic analysis led to the conclusion that the benzoylquinine (R₃N) formed the zwitterionic enolate **80**, which reacted with the metal-coordinated imine (equation 34).^{79,80} This reaction is discussed in more detail in Section 4.1.4. Molecular mechanics calculations have been used to predict the enantioselectivity of ketene [2+2] cycloadditions forming  $\beta$ -lactams.⁸¹



The chiral ketenes **81a–c** reacted with 1,3-diazabuta-1,3-diene by intermolecular [2+2] cycloadditions forming  $\beta$ -lactams with diasteriomeric excesses of 10, 10, and >95%, respectively (equation 35).⁸² Chiral ketenes **81a,c** were also used for stereo-selective [2+2] cycloadditions with imines RN=CR¹R² for the construction of chiral  $\beta$ -lactams with quarternary centers.^{82a}





Chiral imines derived from *D*-glucose have been used in asymmetric synthesis of  $\beta$ -lactams by [2+2] ketene cycloadditions.⁸³ A chiral imine derived from (-)-ephedrine was also used.⁸⁴ Imines **83** with chiral chromium tricarbonyl complexes as substituents reacted with phenoxyketene (**82**) generated by dehydrochlorination forming the chiral  $\beta$ -lactam **84** (equation 36).⁸⁵



Ketenes 1, 31, and 31a (Ar = Ph, R = *i*-Bu) reacted with imines and catalyst (-)-32 to give  $\beta$ -lactams with ~10:1 *cis/trans* preference and 89–98% ee with Ar CH = NTs,⁸⁶ and 80:20 to 98:2 *trans/cis* preference and 63–99% ee with Ar¹CH = NTF.⁸⁷

Arylimines **87** complexed with chromium tricarbonyl were used in completely stereoselective [2+2] cycloadditions forming  $\beta$ -lactams **88** with phthalimidoketene **86** generated from the acyl chloride **85** (equation 37).⁸⁸



Benzyloxyketene (90) gave [2+2] cycloaddition to chiral imine 89 with formation of  $\beta$ -lactams 91 that was converted to imines 92, which reacted again with 90, forming the further  $\beta$ -lactams 93, which were used in the synthesis of 1,3-polyols (equation 38).⁸⁹



Diphenylketene (94) gave [4+2] cycloaddition with chiral 2-alkenyloxazolines and 2-alkenylthiazolines (95) with complete stereoselectivity in the formation of 96 (equation 39).⁹⁰ Stereoselective [3,3] Claisen rearrangements of the zwitterionic intermediates from ketene reactions with allylic ethers, thioethers, and amines to ketenes have been reviewed⁹¹ and are discussed in Section 5.5.3.2.



The dimerization of methylketene (**53**) generated by dehydrochlorination of propionyl chloride with diisopropylethylamine in CH₂Cl₂ at room temperature with 5 mol% trimethylsilylquinine as catalyst gave a 79% yield of the *R*-dimer **97** with 97% *ee*, isolated as the  $\beta$ -ketoamide **98** after reaction with methoxymethylamine (equation 40).⁹² The dimerization was proposed to proceed by formation of a zwitterion **99** that reacted with a second molecule of **53**, forming **97** (equation 41).⁹² Kinetic studies showed that formation of **53** from propionyl chloride was rate determining, with rapid formation of the dimer. Generation of **53** by pyrolysis of propionic anhydride gave the same selectivity for dimer formation, indicating that the dimerization involved two ketene molecules, and not propionyl chloride.



Dimers **101** of ketenes RCH=C=O (**100**, R = Et, *i*-Pr, *t*-Bu, TIPSOCH₂, MeO₂CCH₂, respectively) were prepared similarly in 58–88% yields, 91–96% *ee* (equation 42).⁹²



### **REFERENCES FOR SECTION 5.9**

- 1. Weiss, R. Monatsch. Chem. 1919, 40, 391-402.
- 2. McKenzie, A.; Christie, E. W. J. Chem. Soc. 1934, 1070-1075.
- Morrison, J. D.; Mosher, H. S. Asymmetric Organic Reactions; Prentice Hall, New York, 1971.
- 4. Buschmann, H.; Scharf, H.; Hoffmann, N.; Esser, P. Angew. Chem. Int. Ed. Engl. 1991, 30, 447–515.
- 5. Pracejus, H.; Fortschr. Chem. Forsch. 1967, 8, 493-533.
- 6. Fehr, C. Angew. Chem., Int. Ed. Engl. 1996, 35, 2566-2587.
- 7. Orr, R. K.; Calter, M. A. Tetrahedron 2003, 59, 3545-3565.
- France, S.; Weatherwax, A.; Taggi, A. E.; Lectka, T. Acc. Chem. Res. 2004, 37, 592– 600.
- Palomo, C.; Aizpurua, J. M.; Ganboa, I.; Oiarbide, M. Curr. Med. Chem. 2004, 11, 1837– 1872.
- 10. Fu, G. C. Acc. Chem. Res. 2000, 33, 412-420.
- 11. Fu, G. C. Acc. Chem. Res. 2004, 37, 542-547.
- 12. Pracejus, H.; Liebigs Ann. Chem. 1960, 634, 9-22.
- 13. Pracejus, H.; Kohl, G. Liebigs Ann. Chem. 1969, 722, 1-11.
- 14. Weidert, P. J.; Geyer, E.; Hörner, L.; Liebigs Ann. Chem. 1989, 634, 533-538.
- 15. Yamashita, T.; Yasueda, H.; Nakamura, N. Chem. Lett. 1974, 585-588.
- 16. Yamashita, T.; Mitsui, H..; Watanabe, H.; Nakamura, N. Polym. J. 1981, 13, 179–181.
- 17. Butz, T.; Murer, P.; Seebach, D. Polym. Mater. Sci. Eng. 1997, 77, 132-133.
- Larsen, R. D.; Corley, E. G.; Davis, P.; Reider, P. J.; Grabowski, E. J. J. J. Am. Chem. Soc. 1989, 111, 7650–7651.
- Senanayake, C. H.; Larsen, R. D.; Bill, T. J.; Liu, J.; Corley, E. G.; Reider, P. J. Synlett 1994, 199–200.
- 20. Salz, U.; Rüchardt, C. Tetrahedron Lett. 1982, 23, 4017-4020.
- 21. Jähme, J.; Rüchardt, C. Angew. Chem. Int. Ed. Engl. 1981, 20, 885-887.
- 22. Calmes, M.; Daunis, J.; Jacquier, R.; Natt, F. Tetrahedron 1994, 50, 6875-6880.

- 23. Chen, C.-Y.; Dagneau, P.; Grabowski, E. J. J.; Oballa, R.; O'Shea, P.; Prasit, P.; Robichaud, J.; Tiller, R.; Wang, X. J. Org. Chem. 2003, 68, 2633–2638.
- 24. Calmes, M.; Daunis, J.; Mai, N. Tetrahedron 1997, 53, 13719-13726.
- 25. Calmes, M.; Daunis, J.; Mai, N. Tetrahedron Asymm. 1997, 8, 1641-1648.
- 26. Calmés, M.; Escale, F.; Glot, C.; Rolland, M.; Martinez, J. Eur. J. Org. Chem. 2000, 2459–2466.
- 27. Calmés, M.; Escale, F.; Juan, E.; Rolland, M.; Martinez, J. *Tetrahedron Asymm.* 2000, *11*, 1263–1266.
- 28. Durst, T.; Koh, K. Tetrahedron Lett. 1992, 33, 6799-6802.
- 29. Anders, E.; Ruch, E.; Ugi, I. Angew. Chem. Int. Ed. Engl. 1973, 12, 25-29.
- 30. Cannizzaro, D. E.; Strassner, T.; Houk, K. N. J. Am. Chem. Soc. 2001, 123, 2668-2669.
- 31. Cannizzaro, C. E.; Houk, K. N. J. Am. Chem. Soc. 2004, 126, 10992-11008.
- 32. Akkari, R.; Calmes, M.; Mai, N.; Rolland, M.; Martinez, J. J. Org. Chem. 2001, 66, 5859–5865.
- 33. Akkari, R.; Calmés, M.; Di Malta, D.; Escale, F.; Martinez, J. *Tetrahedron Asymm.* 2003, *14*, 1223–1228.
- Kanoh, N.; Tomatsu, A.; Nishikawa, T.; Ide, M.; Tsuchida, T.; Issiki, K.; Nakata, M. *Tet-rahedron Asymm.* 2003, 14, 1251–1253.
- Bellucci, G.; Berti, G.; Bianchini, R.; Vecchiani, S. Gazz. Chim. Ital. 1988, 118, 451– 456.
- 36. Yang, H.; Foster, K.; Stephenson, C. R. J.; Brown, W.; Roberts, E. Org. Lett. 2000, 2, 2177–2179.
- 37. López-Herrera, F. J.; Sarabia-Garcia, F. Tetrahedron Lett. 1994, 35, 2929-2932.
- 38. Winter, S.; Pracejus, H. Chem Ber. 1966, 99, 151-159.
- 39. Pracejus, H.; Tille, A. Chem Ber. 1963, 96, 854-865.
- 40. Schultz, A. G.; Kulkarni, Y. S. J. Org. Chem. 1984, 49, 5202-5206.
- 41. Huang, W.; Moore, P. A.; Rafai Far, A.; Tidwell, T. T. J. Org. Chem. 2000, 65, 3877-3879.
- 42. Roger, F.; Le Pironnec, M.-G.; Guerro, M.; Gougeon, P.; Gall, P.; le Grel, P.; Baudy-Floc'h, M. *Synthesis* **1999**, 1341–1344.
- 43. Tandon, V. K. Tetrahedron Lett. 2001, 42, 5985-5987.
- 44. Hodous, B. L.; Ruble, J. C.; Fu, G. C. J. Am. Chem. Soc. 1999, 121, 2637-2638.
- 45. Hodous, B. L.; Fu, G. C. J. Am. Chem. Soc. 2002, 124, 10006-10007.
- 46. Hodous, B. L.; Fu, G. C. J. Am. Chem. Soc. 2002, 124, 1578-1579.
- 46a. Wiskur, Sheryl L.; Fu, G. C. J. Am. Chem. Soc. 2005, 127, 6176-6177.
- 46b. Fu, G.; Schaefere, G. C. Angew. Chem. Int. Ed. 2005, 44, 4606-4608.
- 47. Stoutamire, D. W. U.S. Patent 4,570,017; Chem. Abstr. 1986, 105, 78661x.
- 48. Fehr, C.; Stempf, I.; Galindo, J. Angew. Chem. Int. Ed. Engl. 1993, 32, 1044-1046.
- Miller, J. R.; Pulley, S. R.; Hegedus, L. S.; DeLombaert, S. J. Am. Chem. Soc. 1992, 114, 5602–5607.
- 50. Pulley, S. R.; Hegedus, L. S. J. Am. Chem. Soc. 1993, 115, 9037-9047.
- 51. Hegedus, L. S.; Lastra, E.; Narukawa, Y.; Snustad, D. C. J. Am. Chem. Soc. 1992, 114, 2991–2994.
- 52. Hegedus, L. S.; de Weck, G.; D'Andrea, S. J. Am. Chem. Soc. 1988, 110, 2122-2126.

- 53. Hegedus, L. S.; Schwindt, M. A.; De Lombaert, S.; Imwinkelried, R. J. Am. Chem. Soc. 1990, 112, 2264–2273.
- 54. Greene, A. E.; Charbonnier, F.; Luche, M.-J.; Moyano, A. J. Am. Chem. Soc. 1987, 109, 4752–4753.
- 55. Greene, A. E.; Charbonnier, F. Tetrahedron Lett. 1985, 26, 5525-5528.
- 56. Miranda, P. C. M. L.; Correia, C. R. D. Tetrahedron Lett. 1999, 40, 7735-7738.
- 57. Redlich, H.; Lenfers, J. B.; Kopf, J. Angew. Chem. Int. Ed. Engl. 1989, 28, 777-778.
- 58. Sugimura, T.; Tei, T.; Mori, A.; Okuyama, T.; Tai, A. J. Am. Chem. Soc. 2000, 122, 2128–2129.
- 59. Sugimura, T.; Tei, T.; Okuyama, T. Tetrahedron Lett. 2003, 44, 3115-3117.
- 60. Sugimura, T. Eur. J. Org. Chem. 2004, 1185-1192.
- 61. Yoon, T. P.; Dong, V. M.; MacMillan, D. W. C. J. Am. Chem. Soc. 1999, 121, 9726–9727.
- 62. Dong, V. M.; MacMillan, D. W. C. J. Am. Chem. Soc., 2001, 123, 2448-2449.
- 63. Yoon, T. P.; MacMillan, D. W. C. J. Am. Chem. Soc., 2001, 123, 2911–2912.
- 64. Nelson, S. G.; Peelen, T. J.; Wan, Z. Tetrahedron Lett. 1999, 40, 6541-6543.
- 64a. Nelson, S. G.; Peelen, T. J.; Wan, Z. J. Am. Chem. Soc. 1999, 121, 9742-9743.
- 65. Nelson, S. G.; Wan, Z. Org Lett. 2000, 2, 1883-1886.
- 66. Wan, Z.; Nelson, S. G. J. Am. Chem. Soc. 2000, 122, 10470-10471.
- 67. Nelson, S. G.; Zhu, C.; Shen, X. J. Am. Chem. Soc. 2004, 126, 14-15.
- 68. Zhu, C.; Shen, X.; Nelson, S. G. J. Am. Chem. Soc. 2004, 126, 5352-5353.
- 69. Reed, A. D.; Hegedus, L. S. J. Org. Chem. 1995, 60, 3787-3794.
- Hsung, R. P.; Wulff, W.D.; Chamberlain, S.; Liu, Y.; Liu, R.-Y.; Wang, H.; Quinn, J. F.; Wang, S. L. B.; Rheingold, A. L. Synthesis 2001, 200–220.
- 71. Wynberg, H.; Staring, E. G. J. J. Am. Chem. Soc. 1982, 104, 166-168.
- 72. Wynberg, H.; Staring, E. G. J. J. Org. Chem. 1985, 50, 1977-1979.
- 73. Ketelaar, P. E. F.; Staring, E. G. J.; Wynberg, H. Tetrahedron Lett. 1985, 26, 4665-4668.
- 74. Tamai, Y.; Someya, M.; Fukumoto, J.; Miyano, S. J. Chem. Soc. Perk. Trans. 1 1994, 1549–1550.
- 75. Tamai, Y.; Yoshiwara, H.; Someya, M.; Fukumoto, J.; Miyano, S. J. Chem. Soc. Chem. Commun. 1994, 2281–2282.
- 76. Concepcion, A. B.; Maruoka, K.; Yamamoto, H. Tetrahedron 1995, 51, 4011-4020.
- 77. Wilson, J. E.; Fu, G. C. Angew. Chem. Int. Ed. 2004, 43, 6358-6360.
- 78. Palomo, C.; Aizpurua, J. M.; Ganboa, I.; Oiarbide, M. Eur. J. Org. Chem. 1999, 3223–3235.
- 79. Taggi, A. E.; Hafez, A. M.; Wack, H.; Young, B.; Ferraris, D.; Lectka, T. J. Am. Chem. Soc. 2002, 124, 6626–6635.
- France, S.; Shah, M. H.; Weatherwax, A.; Wack, H.; Roth, J. P.; Lectka, T. J. Am. Chem. Soc. 2005, 127, 1208–1215.
- 81. Taggi, A. E.; Hafez, A. M.; Dudding, J.; Lectka, T. Tetrahedron 2002, 58, 8351-8356.
- 82. Abbiati, G.; Rossi, E. Tetrahedron 2001, 57, 7205-7212.
- 82a. Palomo, C.; Aizpurua, J. M.; García, J. M.; Galarza, R.; Legido, M.; Urchegui, R.; Román, P.; Luque, A.; Server-Carrío, Linden, A. J. Org. Chem. 1997, 62, 2070–2079.
- Arun, M.; Joshi, S. N.; Puranik, V. G.; Bhawal, B. M.; Deshmukh, A. R. A. S. *Tetrahedron* 2003, *59*, 2309–2316.

- 84. Shinkre, B. A.; Puranik, V. G.; Bhawal, B. M.; Deshmukh A. R. A. S. *Tetrahedron Asymm.* 2003, 14, 453–459.
- 85. Tanaka, Y.; Sakamoto, T.; Kamikawa, K.; Uemura, M. Synlett 2003, 519-521.
- 86. Hodous. B. L.; Fu, G. C. J. Am. Chem. Soc. 2002, 124, 1578-1579.
- 87. Lee, E. C.; Hodous, B. L.; Bekgin, E.; Shih, C.; Fu, G. C. J. Am. Chem. Soc. 2005, 127, 11586–11587.
- 88. Del Buttero, P.; Molteni, G.; Papagni, A. Tetrahedron Asymm. 2003, 14, 3949-3953.
- Palomo, C.; Aizipurua, J. M.; Urchegui, R.; Garcia, J. M. J. Org. Chem. 1993, 58, 1646– 1648.
- Elliott, M. C.; Monk, A. E.; Kruiswijk, E.; Hibbs, D. E.; Jenkins, R. L.; Jones, D. V. Synlett 1999, 1379–1382.
- 91. Gonda, J. Angew. Chem. Int. Ed. 2004, 43, 3516-3524.
- 92. Calter, M. A.; Orr, R. K.; Song, W. Org. Lett. 2003, 5, 4745-4748.

#### 5.10 OTHER ADDITIONS TO KETENES

#### 5.10.1 Reaction with Diazomethanes

The reaction of diphenylketene (1) with diphenyldiazomethane was found by Staudinger and Reber to form tetraphenylethylene (3), evidently through the formation of tetraphenylcyclopropanone (2) followed by decarbonylation (equation 1).¹ With less highly substitututed ketenes and diazoalkanes, a variety of more stable cyclopropanones have been prepared from the reactions of ketenes and diazoalkanes,^{2–12} including the parent **5** from ketene, obtained in solution, and trapped with MeOH, giving a solution of the hemiacetal (equation 2).^{4,5} Tetramethylcyclopropanone (7) was obtained in a similar fashion from dimethylketene (**6**, equation 3),⁶ and methyl-,⁶ 2,2-dimethyl-,⁶ and trimethylcyclopropanone⁸ were also prepared.



Cyclobutanones may be are formed in these reactions from insertion reactions of the diazoalkanes with the cyclopropanones, as in the conversion of **8** to **9** (equation 4).^{8,9} Trimethylsilylketene (**10a**) and trimethylgermylketene (**10b**) gave the cyclopropanones **11** with diazomethane (equation 5),¹⁰ and **10a** reacted with trimethyl-silyldiazomethane to form **12**¹³ (equation 6).^{10–14}



The reactions of *t*-BuCH=C=O (**8a**) and CH₃CH=C=O (**13**) with *t*-BuCHN₂ and CH₃CHN₂ to give cyclopropanones **14** have been studied experimentally (equation 7),¹⁵ and the mechanisms of these processes have been elucidated with *ab initio* molecular orbital calculations.¹⁵ Arylketenes **15** prepared by Wolff rearrangements reacted with Me₃SiCHN₂ through intermediate cyclopropanones **16** that underwent decarbonylation to give trimethylsilyl-substituted styrenes **17** (equation 8).¹⁶



Norbornylideneketene **19** was generated by pyrolysis of the dimer **18** (Section 3.1) and collected at 196 K (equation 9).¹⁷ Reaction of **19** with diazomethane at 195 K gave the products **21a** and **21b** in a 1.6:1 ratio (equation 10).¹⁷ The low preference for *exo* attack was not consistent with initial attack at  $C_2$  of the norbornyl skeleton, but could have arisen from in-plane nucleophilic attack of the

diazomethane at the carbonyl carbon to give oxyallyls **20a** and **20b**, which closed to **21** (equation 10). At higher temperatures **21a** and **21b** interconverted, and eventually reached an 0.8:1 equilibrium ratio.¹⁷



Reaction of bis(2,4,6-triisopropylphenyl)ketene 22 with diazomethane formed the unobserved cyclopropanone 23, which rearranged to the isolated product 24 (equation 11).¹⁸



Thermolysis of the bis(diazo ketone) **25** was proposed to involve formation of ketene **26**, which reacted with the remaining diazo group, forming cyclopropanone **27**, followed by decarbonylation to give the enone product **28** (equation 12).¹⁹



Reaction of the stable diarylstannylene **30** with ketenes **29** gave three-membered stannocycles **31** (equation 13).²⁰ This reaction is also noted in Section 4.8.4.



#### 5.10.2 Reaction with Sulfur Dioxide

Sulfur dioxide reacted with ketene (1) at -78 °C to form a white solid that was identified as the cyclic adduct 2 on the basis of the ¹H NMR singlet observed at  $\delta 2.30$  at -67 °C in SO₂ and the formation of 3 upon reaction with aniline (equation 1).²¹ Reaction of 1 with azines was also reported.²² Dichloroketene (4) reacted with SO₂ and benzanilide to form 5 (equation 2).²³

In contrast to the thermal [2 + 1] cycloaddition of ketenes with SO₂, the photochemical reaction in matrices at 12 K with **6** was found to give [2+2] cycloadducts **7** (equation 3).²⁴

$$\begin{array}{c} R \\ H \\ H \\ 6 \end{array} \xrightarrow{SO_2} \\ 12 K \\ 0 \\ S \\ 7 \end{array} \xrightarrow{R} \\ S \\ 0 \\ 7 \end{array}$$

$$(3)$$

# 5.10.3 Hydrogenation of Ketenes

Ketene (1) is hydrogenated to acetaldehyde (2) using Group 9 (Co, Rh, Ir) or Group 18 (Ni, Pd, Pt) metal catalysts.²⁵ Reaction of acetaldehyde prepared in this way with further ketene provides a route to vinyl acetate (3) (equation 1).²⁶



# **REFERENCES FOR SECTIONS 5.10.1–5.10.3**

- 1. Staudinger, H.; Reber, T. Helv. Chim. Acta 1921, 4, 3-23.
- 2. Wasserman, H. H.; Berdahl, D. R.; Lu, T. In *The Chemistry of the Cyclopropyl Group*; Rappoport, Z., Ed.; Wiley, New York, 1987, pp.1455–1532.

- 3. Turro, N. J. Acc. Chem. Res. 1969, 2, 25-32.
- 4. Turro, N. J.; Hammond, W. B. J. Am. Chem. Soc. 1966, 88, 3672-3673.
- Schaafsma, H.; Steinberg, H.; de Boer, T. J. Recl. Trav. Chim. Pays-Bas 1966, 85, 1170– 1172.
- 6. Turro, N. J.; Hammond, W. B. Tetrahedron 1968, 24, 6017-6028.
- 7. Hammond, W. B.: Turro, N. J. J. Am. Chem. Soc. 1966, 88, 2880-2881.
- 8. Turro, N. J.; Gagosian, R. B. J. Am. Chem. Soc. 1970, 92, 2036-2041.
- 9. Audier, H.; Conia, J. M.; Fetizon, M.; Gore, J. Bull. Chim. Soc. Fr. 1967, 787-791.
- Zaitseva, G. S.; Bogdanova, G. S.; Baukov, Yu. I.; Lutsenko, I. F. Zh. Obshch. Chem. 1978, 48, 131–137. J. Gen. Chem. (Engl. Transl.) 1978, 48, 111–117.
- Zaitseva, G. S.; Novikova, O. P.; Livantsova, L. I.; Kisin, A. V.; Baukov, Yu. I. Zh. Obshch. Chem. 1990, 60, 1073–1077. J. Gen. Chem. (Engl. Transl.) 1990, 60, 947–951.
- Zaitseva, G. S.; Krylova, G. S.; Perelygina, O. P.; Baukov, Yu. I.; Lutsenko, I. F. Zh. Obshch. Chem. 1981, 51, 2252–2266. J. Gen. Chem. (Engl. Transl.) 1981, 51, 1935–1947.
- Zaitseva, G. S.; Kisin, A. V.; Fedorenko, E. N.; Nosova, V. M.; Livantsova, L. I.; Baukov, Yu. I. Zh. Obshch. Chem. 1987, 57, 2049–2060. J. Gen. Chem. (Engl. Transl.) 1987, 57, 1836–1845.
- Zaitseva, G. S.; Bogdanova, G. S.; Baukov, Yu. I.; Lutsenko, I. F. J. Organomet. Chem. 1976, 121, C21–C22.
- 15. Sorensen, T. S.; Sun, F. J. Am. Chem. Soc. 1997, 119, 11327-11328.
- 16. Léost, F.; Doutheau, A. Tetrahedron Lett. 1999, 40, 847-848.
- 17. Cordes, M. H. J.; Berson, J. A. J. Am. Chem. Soc. 1996, 118, 6224-6251.
- 18. Frey, J.; Rappoport, Z. J. Am. Chem. Soc. 1996, 118, 5182-5191.
- Nakatani, K.; Takada, K.; Odagaki, Y.; Isoe, S. J. Chem. Soc., Chem. Commun. 1993, 556–557.
- 20. Edelmann, F. T.; Belay, M. J. Fluorine Chem. 1997, 84, 29-33.
- 21. Bohen, J. M.; Joullié, M. M. J. Org. Chem. 1973, 38, 2652-2657.
- 22. Lysenko, Z.; Joullié, M. M. J. Org. Chem. 1976, 41, 3925-3927.
- 23. Bellus, D. Helv. Chim. Acta 1975, 58, 2509-2511.
- 24. Dunkin, I. R.; MacDonald, J. G. J. Chem. Soc. Perkin Trans. 2, 1984, 2079-2084.
- 25. Tustin, G. C.; Depew, L. S. PCT Int. Appl. WO 9734858. Chem. Abs. 1997, 127, 279838.
- 26. Tustin, G. C.; Colberg, R. D.; Zoeller, J. R. Catalysis Today 2000, 58, 281-291.

# 5.11 KETENE REACTIONS USING POLYMER SUPPORTS

Ketene reactions using polymer-supported reagents as well as reactions of polymersupported ketenes have proven to be useful in synthetic and mechanistic studies, and a review of ketenes in polymer-assisted synthesis has appeared.¹

Solid phase–supported imines **1** derived from amino acids gave high diastereoselectivities in reactions with ketenes generated in solution from acyl chlorides **2** forming  $\beta$ -lactams **3** with *dr* ranging from 55:45 to 100:0 after cleavage from the resin (equation 1).²  $\beta$ -Lactams formed similarly in the solid phase have been converted to quinolones³ and subjected to Suzuki and Heck cross-coupling reactions.⁴ Imines attached to soluble polyethylene glycol supports also formed  $\beta$ -lactams upon reaction with ketenes.⁵ Resin-attached peptides **6** are acylated by ketenes **5** generated from Fmoc-protected, amino acid–derived diazo ketones **4**, forming **7** with  $\beta$ -amino acids added to the peptide chains in purities >90% (equation 2).⁶ See also Section 3.3.1.⁷ Soluble polymer supports have also been utilized in  $\beta$ -lactam formation.⁸ Ketenes **8** formed by dehydrohalogenation reacted with Rink resin–supported imines **9**, giving resin-bound  $\beta$ -lactams **10** that were cleaved at nitrogen, yielding **11** (equation 3).⁹



Solid-supported cyclobutenones **12** upon ring opening formed supported alkenylketenes **13** (equation 4).¹⁰ The ketenes were not observed directly but they participated in the formation of a variety of products, including cyclization to naphthoquinones **14** in 30–50% yields after cleavage from the resin (equation 4).¹⁰ Ketene **16** prepared from bisketene **15** and monomethyl poly(ethylene glycol) was used in the preparation of succinamides **18** and 4-pyridones (equation 5).¹¹





The chiral solid–supported imine **19** reacted with phenoxyketene (**20**) generated by dehydrochlorination of the acyl chloride to form the supported  $\beta$ -lactam **21** with high stereoselectivity (equation 6).¹² Cleavage from the resin gave the free  $\beta$ -lactam **22** (equation 7).¹²



Ketenes ArCMe=C=O (24) generated in solution reacted with polymersupported alcohols 23 to give stereoselective formation of esters 25, which upon saponification gave optically active 2-arylpropionic acids 26 (equation 8).¹³ Additions of polymer-supported chiral alcohols to aryl(phthalimidomethyl)ketenes provided a route to  $\beta$ -amino acid derivatives.¹⁴



Anionic polymerization of ethylphenylketene on polystyrene beads has been used in solid-supported synthesis of poly(methylmethacrylate).¹⁵ Imine-substituted Wang resin **27** gave [2+2] cycloaddition with ketenes **28** formed by dehydrochlor-

ination, yielding polymer-bound  $\beta$ -lactams **29** that were cleaved to yield the free  $\beta$ -lactams **30** (equation 9).¹⁶



# **REFERENCES FOR SECTION 5.10.4**

- 1. Rafai Far, A. Angew. Chem. Int. Ed. 2003, 42, 2340-2348.
- Ruhland, B.; Bhandari, A.; Gordon, E. M.; Gallop, M. A. J. Am. Chem. Soc. 1996, 118, 253–254.
- 3. Pei, Y.; Houghten, R. A.; Kiely, J. S. Tetrahedron Lett. 1997, 38, 3349-3352.
- 4. Ruhland, B.; Bombrun, A.; Gallop, M. A. J. Org. Chem. 1997, 62, 7820-7826.
- 5. Molteni, V.; Annunziata, R.; Cinquini, M.; Cozzi, F.; Benaglia, M. *Tetrahedron Lett.* **1998**, *39*, 1257–1260.
- 6. Marti, R. E.; Bleicher, K. H.; Bair, K. W. Tetrahedron Lett. 1997, 38, 6145-6148.
- 7. Guichard, G.; Abele, S.; Seebach, D. Helv. Chim. Acta 1998, 81, 187-206.
- 8. Annunziata, R.; Benaglia, M.; Cinquini, M.; Cozzi, F. Chem Eur. J. 2000, 6, 133-138.
- 9. Dasgupta, S. K.; Banik, B. K. Tetrahedron Lett. 2002, 43, 9445-9447.
- 10. Tempest, P. A.; Armstrong, R. W. J. Am. Chem. Soc. 1997, 119, 7607-7608.
- 11. Rafai Far, A.; Tidwell, T. T. J. Org. Chem. 1998, 63, 8636-8637.
- 12. Wills, A. J.; Krishnan-Ghosh, Y.; Balasubramanian, S. J. Org. Chem. 2002, 67, 6646–6652.
- 13. Akkari, R.; Calmes, M.; Mai, N.; Rolland, M.; Martinez, J. J. Org. Chem. 2001, 66, 5859–5865.
- Akkari, R.; Calmes, M.; Di Malta, D.; Escale, F. O.; Martinez, J. *Tetrahedron Asymm.* 2003, 14, 1223–1228.
- 15. Sudo, A.; Iinuma, R.; Endo, T. J. Poly. Sci, Part A, Polymer Chem. 2002, 40, 3455–3459.
- 16. Delpiccolo, C. M. L.; Mata, E. G. Tetrahedron Asymm. 2002, 13, 905-910.

# Index

Ab initio calculations carbon suboxide, cumulenones, 429-433 chloroketenes, 333 fluoroalkylketene formation, 326 imidoylketenes, 263-267 ketene electronic states, 6-8 dimerization reactions, 9–12 substituent effects, 13-17 Absorption band assignments, ketene infrared spectra, 39-45 Acetic acid pyrolysis, ketenes from, 57-60 Acetone pyrolysis, alkylketenes, 195 6-Acetoxycyclohexa-2,4-dienone photosolvolysis, ketene formation, 151 Acetoxylketene, dehydrochlorination, 316 Acetyl chloride, ketene formation from cations and free radicals, 182 Acetylketenes alkynyl ether thermolysis, 157–158 cycloaddition reactions, 255 [4+2] cycloadditions, carbonyl groups, 529 dioxinone precursor, 153-155 from ester enolates, 76 nucleophilic additions, hydration mechanisms, 541 N-Acetylpropanamide, ketene formation, 168 Acid anhydrides, ketenes from, 61-64 Acid catalysts, intermolecular [2+2] cycloaddition and ketene dimerization, 455 Acrolein photochemical isomerization, ethylketene formation, 167 Activated acids, ketene formation from, 65 - 74

Acylallenes, vinylketene formationis, 209 - 2103-Acyl-4-aminocyclobutenone, ring opening reaction, 127-128 Acyl chloride ketenes acrylic acid reactions, 71 alkynylketene formation, 215 aminoketenes, 309-310 carbon substitution, 227-228 cyclopropylketene formation, 239 dehydrochlorination of, 70 diazo ketone formation, 102-104 electron impact mass spectra, 49-50 imine catalysis, 67 Acyl diazo ketones, photolysis, 105 Acylfluoroketene, photolysis, 325 Acyl halides, ketene formation from, 65-74 Acylisocyanate, imidoylketene formation, 266-267 Acylium clusters, ketene formation, 388-389 Acylketenes carbon substitution, 244-259 cycloaddition reactions alkynes, 497 imines, 509 intramolecular cycloadditions, 520-524 diazo ketone formation, ketocarbene rearrangements, 110 ester enolate formation of, 81 Meldrum acid thermolysis, 161 oxygen nucleophile reactions, 556 photolysis, 443-444 protonation, ketone electrophilic reactions, 584 thermolysis of, 448-449 titanium isopropoxide decomposition, 83 X-ray studies of, 20-21

Ketenes II, by Thomas T. Tidwell Copyright © 2006 John Wiley & Sons, Inc. Acylketenes (continued) ynamine reaction, 187 ynol/ynolates, 378-379 Acyl radicals, ketene reactions, 592-598 1-Adamantyl(carboethoxy)ketene, dehydrochlorination, 245 Adamantylketenes, dehydrochlorination, 201 Addition-elimination sequence, ketene formation, 73 Alcohols, bisketene formation and, 406–407, 413 Aldehyde reactions, ketene formation, 188 Aldoketenes intermolecular [2+2] cycloaddition and dimerization, 455 β-propiolactone dimers, 460 Aliphatic ketenes nucleophilic additions, hydride reduction, 553 X-ray studies of, 20 Alkenes chloroketene reactions, 333 cycloaddition reactions of ketenes, 451-452 intermolecular [2+2] cycloaddition, 453-510 basic mechanisms, 464-472 nucleophilic, 487-491 rapid reactions, 474-485 Alkenyl carbene complexes, dienylketenes, 391-392 Alkenylketenes carbon substitution, 206 Dötz reaction and, 171-172 metal substitution, 392-393 polymer supports for, 623 verbinone photolysis, 132-133 X-ray studies of, 20 Alkoxide catalyst, bisketene formation, 406 Alkyl acetoacetate pyrolysis, ketene formation, 82 Alkyl(aryl)ketenes, protonation, ketone electrophilic reactions, 583-584 Alkylbromoketene, 339 Alkyliodoketene, 339 Alkylketenes carbon-substituted structures, 193-203 [4+2] cycloadditions, 518-519

flash vacuum thermolysis, 49-50 fragmentation pathways, 49–50 nucleophilic additions, hydration mechanisms, 539 Alkyl substituents, chiroptical properties of ketenes, 38 Alkyne oxidation, ketene formation, 183-185 Alkynes, ketene cycloaddition, 495-499 Alkynylcyanoketenes, carbon substitution, 216 Alkynylcyclobutenones, ubiquinone formation. 124 Alkynyl ethers boron-substituted ketenes, 384-385 oxygen-substituted ketenes, 318 thermolysis, ketene formation, 156-159 ynolate ketenes, 380 Alkynyl ketenes, carbon substitution, 214-219 Alkynyl-substituted cyclohexadienone, diene-yne ketene formation, 150 Alkynyl(trimethylsilyloxy)ketene, carbon substitution, 216 Allenes carbon suboxide photolysis, 444-445 ketene cycloaddition, 492-494 nucleophilic ketene additions, Wittig reactions, 576-579 Allenylcyclobutenone, torquoselective ring opening, 127 Allenylketenes cumulene substitution, 268-272 intramolecular cycloadditions, 522-523 matrix p346-347hotolysis radical reactions, 596 4-Allyl-4-ethoxycyclobutenone, thermolytic ketene formation, 128-129 Allyl morpholines, methylketene formation, 202  $\alpha$ -substituted trimethylsilylketene, 358 α-Amidoketenes, carbon substitution, 251 Amines chiral ketene stereoselectivity, 606 cyclization bisketene formation and, 407-408 ketenyl radicals, 280 cyclohexadienone photolysis, 146

nitrogen nucleophile/electrophile reactions with ketenes, 560-567 nucleophilic ketene additions, 543-551 Aminocarbene complex, ketene formation, 172 Aminoketene nitrogen substitution, 305-306 nonplanar geometry, 16 Anionic ketenes, radical ketenyl structures, 425-427 Anisole protonation, cyclohexadienone formation, 148 Anti conformation allenylketenes, 269-270 ketene dimerization, 459-460 vinylketene, 206 Argon matrix benzazetinone thermolysis, 301 bisketene formation, 405 carbon-substituted ketene formation. 226-227 oxoquinone methides, 295-296 chromium carbene complex, ketene formation, 175 cumulenone formation, 433 cyclopropenylketene formation, 138 carbon substitution, 240 cyclopropylideneketene photolysis, 444 diazo ketone ketene formation, photochemical Wolff rearrangement, 107-110 dimethylketene photolysis, 442 fluoroketene formation, 329 6-imino-2,4-cyclohexadien-1-one photolysis, 149 ketene IR spectra, 41-45, 71 ketene photolysis, 441-445 ketenyl radicals, 281-282 (2-nitrophenyl)diazomethane photolysis, 300 pentafulvenone formation, 287 silylketene formation, 347 Z-bisketenyletehylene, 418 Arndt-Eistert reaction diazo ketone formation, catalyzed Wolff rearrangement, 100-104 oxygen nucleophile-ketene reactions, 554-559

Aromatic stabilization energy (ASE), fulvenones, 284 Aroylketenes, thermolytic formation, 258 Arsenic-substituted ketenes, 368 Arylalkenylketenes, diazo ketone formation, catalyzed Wolff rearrangement, 100-104 Arylalkylketenes carbon substitution, 228 oxygenation, 589 polymerization, 601-602 Arylalkynes, peracid oxidation, ketenes from. 183-185 Aryl bisketenes, metal substitution, 417 α-Aryl-α-cyanomethylketenes, dehydrochlorination of, 69 Arylhaloketenes, stereoselectivity, 604 Aryliodonium ylides, diazo ketones, ketene formation, ketocarbene rearrangements, 110 Arylketenes arylacetyl benzotriazole reactions and, 74 carbon substitution, 221-234 [4+2] cycloadditions, 518-519 diazomethane reactions, 619 from ester enolates, 76-77 nucleophilic additions hydration mechanisms, 533-534, 535, 538-539 oxygen nucleophile reactions, 556 Wittig reactions, 576-577, 578-579 photoelectron spectra of, 37-38, 438 Arylmethylketenes electron transfer oxygenation, 232 stereoselectivity, 604-6-5 Arylpropynes, oxidation, ketenes from, 184 Aryl-substituted bisketenes, 404 Aryl(trialkyl)silylketenes, Wolff rearrangements, 234 Arylvinylketene, carbon-substituted reactions, 230-231 Asymmetric geometry, intermolecular [2+2] cycloaddition of ketenes, alkene reactions, 477 Azafulvenone, pyrolysis, 302 Azaindenylideneketene, Wolff rearrangement of, 289-292 1-Adizocycloheptatrienone, (4-cyanobutadienyl)ketene, 145

Azidoalkylcyclohexendione, dienylketene formation, 148 Azidoketene, dehydrochlorination, 308 Azidoquinones, cyanoketene formation, 219 Azoketenes, dehydrochlorination, 308 B3LYP computational technique, ketene studies, 12 Barton-Quinkert reaction, cyclohexadienone ring opening, 142-151 Base catalysts, intermolecular [2+2] cycloaddition and ketene dimerization. 455 Base-induced formation, acid anhydride pyrolysis of ketenes, 62-64 Beckmann rearrangement, cycloaddition of ketenes, nucleophilic alkenes, 488-489 Bellus-Claisen rearrangement, oxygen nucleophile ketene reactions, 557-559 BEMP polymer base, dehydrochlorination of ketenes, 69 Benzannulation reactions chromium carbene photolysis, 175-176 ketene stereoselectivity, 611 Benzazetinone, thermolysis of, 301 Benzocvclobutenone metal-substituted bisketenes, 412 oxo-1,2-xylylene formation, 296-297 Benzooxirene, pentafulvenone formation, 286-287 Benzopentafulvenone, diazo ketone photolysis, 289 Benzoylated aminoketenes, "münchnone" tautomers, 306 Benzoylphenylketene cycloaddition reactions, 257-258 multiphotonic dibenzoyldiazomethane reaction, 443 Benzoylquinine (BQ), ketene catalytic asymmetric reactions, 70 Benzylketenes, stereoselectivity, 610 Benzyloxyketenes cycloaddition reactions, 320 imines, 501-506 dehydrochlorination, 317 stereoselectivity, 613-614 Bicyclic ketenes, carbon substitution, 198

Bicyclic ketones, hydrogen transfer, 132 Bicyclo[2.2.1]heptan-7-one, ketene aldehyde formation, 134 Bicyclo[2.2.1]heptene, cycloaddition reactions, 471 Bicyclo[3.2.0]hept-2-en-6-ones, equilibrium percentages, 466 Bicyclo[3.2.0]heptene, trapping agents, 207 Bicyclo[2.2.2]hexadienone, dimethylketene formation, 133 Bicyclo[2.1.1]hexan-2-one, 3-butenylketene formation, 136 Bicyclo[2.2.0]-hexanone, photochemical ketene formation, 115 Bicyclo[2.2.2]octadiene alkynyl(trimethylsilyloxy)ketene formation, 216 oxvgen-substituted ketenes, 316 Bicyclo[3.3.0]octadiene, dichloroketene reactions, 336 Bicyclo [1.1.1]pentan-2-one, photochemical ketene formation, 115 Bis(2,4,6-triisopropylphenyl)ketene ("Ditipylketene") diazomethane reactions, 620 nucleophilic additions, hydride reduction, 552-553 Bis(3,5-dibromomesityl)ketene, X-ray studies of, 20 Bis(acylketene), metal substitution, 415 Bis(alkenyl)ketene, carbon substitution, 206 Bis(chromium carbene) complexes, bisketene formation, 421 Bis(chromium) complex, bisketene formation, 421-422 Bis(diazo) diketone, ketene formation, photochemical Wolff rearrangement, 107 - 108Bis(diazo ketones), metal-substituted bisketenes, 416-417 Bisdioxinone bisketene precursors, 418-419 bsketene formation, 420 Bis(germyl)ketenes, 354 Bisketenes aryl-substituted complexes, 404 3,5-di-tert-butyl-1,2-b3enzoquinone photolysis, 148 infrared spectra of, 41-45

metal-complexed preparation, 397-398 metal substitution, 400-422 phosphorus-substituted complexes, 367-368 photochemical decarbonylation, 444 photoelectron spectra of, 35-38 polymer supports for, 623-624 radical reactions, 595-598 stereoselectivity, 606-607 thermal ring opening formation, 357 thermochemistry of, 24-25 1.2-Bisketenes ketenvl radical formation, 282 metal substitution, 400-422 nucleophilic additions, amination mechanisms, 548-549 1,2-Bis(ketenylidene)ethylene, 416 1,3-Bisketenes, metal substitution, 415 1.4-Bisketenes formation, 402-403 metal substitution, 415-416 1,6-Bisketenes, 352-353 metal substitution, 402-403, 415-416 1,8-Bisketenes, metal substitution, 415-416 Bis(methylenecyclobutenone), allenylketenes, 271-272 Bis(pentafluorophenyl)ketene, direct acid dehydration, 222 Bis(phenyliodonium) ylide, bisketene formation, 420 Bis(silylated) ketenes, lithium reagents, 351-352 Bis(silyl)-substituted ketene, 345-346 Bis(trialkylsilyl) ketenes, 354 ynol/ynolates, 382 Bis(trifluoromethyl)ketene carbon/silicon nucleophilic additions, 572-573 cycloaddition chemistry, 327-328 dimerization, 461 Bis(trifluoromethyselenyl)ketene, silver salt reaction, 387 Bis(trimethylsilyl)-1,2-bisketene, 403 neat alcohols and, 406 Bis(trimethylsilyl)ketene carbonylation, 347-348 classification and formation, 343-344

ester enolate formation of, 78 Blomquist reaction, metal-substituted ketenes, 416 Bond alternation, fulvenones, 284 Bond angles, molecular structure of ketenes, 18 - 21Bond distances intermolecular [2+2] cycloaddition of ketenes, alkene reactions, 478-484 molecular structure of ketenes, 18-21 Boron-substituted ketenes, classification and structure, 384-386 Bridged cyclohept-2,4-dien-1-one photolysis, cyclopropylketene formation, 151 2-Bromo acyl chloride, intermolecular [2+2] cycloaddition and ketene dimerization, 457-458 α-Bromo-α-alkenylketenes, 340 Bromoarylketenes, stereoselective reactions, 338 Bromochloroketene, cycloaddition reactions, 338 α-Bromodiphenylacetyl chloride, diphenylketene formation, 222  $\alpha$ -Bromoenol phosphates, ketene formation, 187 Bromoketenes, halogen substitution, 331, 337-340 Bromophenylketene, dehydrochlorination, 338 (bromopy)5-Bromo pyrones, thermolytic ketene formation, 163 Bromotrimethylsilylketene classification and formation, 344 halogen substitution, 338-339 Butadienylketenes, radical reactions, 596 Butenolides, bisketene formation of, 406-407 4-Butenylketenes, radical reactions, 596 1-tert-Butoxy-1,3-butadiyne, alkynylketene formation, 215-216 tert-Butyl alkynyl ethers, ketene formation, 156-157 3,5-di-tert-Butyl-1,2-benzoquinone photolysis, bisketene formation, 148 tert-Butyl(carboethoxy)ketene, dehydrochlorination, 245

tert-Butylcyanoketene carbon substitution, 219 cycloaddition kinetics, 483 alkynes, 495-499 allenes, 492-494 nitrogen nucleophile/electrophile ketene reactions, 566 (4-tert-Butyl-2,6-dimethylphenyl)mesitylketene, photolysis of, 444 di-tert-Butylketene carbon/silicon nucleophilic additions, 568 - 574diazo ketone thermolysis, 199 ketene radical reactions, 594 ketone electrophilic reactions oxygenation, 588 protonation, 581-582 photolysis, 442-443 stable properties, 197-198 tert-butylketene, photochemical Wolff rearrangement, 104-105 2,6-di-tert-Butyl-4-methylphenyl (BHT) esters aryloxide anion-forming ketenes, 229 ketene formation from, 78-79 2,6-di-tert-Butyl-4-methyoxyphenyl (BHA) esters, carbon-substituted ketenes, 229 n-Butyltrimethylsilylketene, classification and formation, 344 Camphorketene dehalogenation, 198-199 thermolytic formation, 258 Carbamates, chromium carbene complexes, 178 Carbanion decay carbon-substituted ketenes, 273-283 ketenyl radicals, 282 4-nitrophenylketene formation, 77 Carbanion intermediates, ester enolate formation of ketenes, 77-81 Carbenes cumulenones, 429-433 diazo ketones, 92-93 ketene formation, photochemical Wolff rearrangement, 105-110 dimesitylketene photolysis, 444 norbornadiene photolysis, 187

Carbenic centers, carbon-substituted ketenes, 273-283 Carbocationic derivatives carbon-substituted ketenes, 274-283 ketene photolysis, 443 Carboethoxyketenes alkyne cycloaddition reactions, 499 from ester enolates, 76 Carbohydrates, chloroketene cycloadditions, 335 Carbomethoxyketene [4+2] cycloaddition with imines, 504 oxygen substitution, 313-314 ¹³C nuclear magnetic resonance acylium ion shifts, 273-274 alkylketenes, 194 chloroketenes, 332 ketene physical properties, 27-33 Carbon atoms, thermolytic ketene formation, 160 Carbon-carbon double bonds ketene cycloadditions, 452 ketene polymerization, 600-602 Carbon dioxide cumulenones, 428-433 ketenyl radicals, anions, and cations, 425-427 Carbon monoxide bisketene formation, 400 cumulenones, 430-433 ketene intermediates, 189 ketene photolysis, 441-442 ketenyl anions, and cations, 425-427 ketenyl radicals, 279-280, 425-427 metal-complexed ketenes, 390-391 Carbon nucleophiles, ketene additions, 568-574 Carbon suboxide bisketene formation, 400-401 cumulenones, 428-433 ketene polymerization, 602 nucleophilic additions to ketenes, hydration mechanisms, 540 photolysis, allene formation, 444-445 X-ray studies of, 21 Carbon-substituted ketenes acylketenes, 244-259 alkenylketenes, 206-213 alkylketenes, 193-203

alkynyl and cyano ketenes, 214-219 aryl- and heteroarylketenes, 221-234 charged, radical, and carbenic side chains, 273 - 283cumulene-substituted ketenes, 268-272 cyclopropyl-, cyclopropenyl-, and oxiranylketenes, 237-242 fulvenones and cumulenones, 284-292 imidoylketenes, 263-267 oxoquinone methides, oxoxylylenes, and related species, 294-303 Carbonyl carbene complexes alkenylketene chromium complex, 171 - 179nitrogen-substituted ketenes, 305-306 Carbonyl groups [2+2] cycloadditions of ketenes, 452 cycloadditions of ketenes, 525-530 nucleophilic additions to ketenes, hydration mechanisms, 533-541 Carboxamide-substituted ketenes alkyne cycloaddition reactions, 499 photochemical Wolff rearrangement, 256 Carboxylic acids and derivatives acid anhydride pyrolysis of ketenes, 62 - 64carbon-substituted arylketenes, 224 dehalogenation of  $\alpha$ -halo derivatives, ketene formation, 86-89 diazo ketone formation of ketenes, 102 - 104ester-derived ketenes, 76-84 ester enolates, 76-81 pyrolysis, 81-84 ester enolate formation of ketenes, 78 ketenes from, 57-60 oxygen nucleophile-ketene reactions, 554-559 stereoselectivity of ketenes, 608 Carboxylketenes, Meldrum's acid derivative, 252 Catalytic dehydration, acetic acid-ketene formation, 57-60 Catalyzed Wolff rearrangement, diazo ketones, ketene formation, 97-104 Cationic ketenes formation mechanisms, 181-182 nitrosative guanine deamination, 303 radical ketenyl structures, 425-427

CBS-Q calculations, ketene thermochemistry, 24-25 CD spectra, ketene conformational behavior, 38 Cerium reagents carbon/silicon nucleophilic additions to ketenes, 571 silylketenes, 351-352 Cheleotropic fragmentation, ketene formation, 165-166 Chiral alkaloid bases, ketene catalytic asymmetric reactions, 70 Chiral centers, ketene stereoselectivity, 603-615 Chiral cyclohexadienone photolysis, dienylketene formation, 146 Chiral tertiary amines, ketene catalytic asymmetric reactions, 69 Chiroptical properties of ketenes alkenylketenes, 208 cycloaddition kinetics, allenes, 493-494 metal-complexed ketenes, 392-393 ultraviolet and photoelectron spectra, 35 - 38Chloroalkylketene, cycloaddition reactions, nucleophilic alkenes, 487-488 Chlorocarbonyl(phenyl)ketene, malonyl dichloride thermolysis, 244-245 Chlorocyanoketene, carbon substitution, 218 Chloroformylketene, malonyl dichloride thermolysis, 72 Chloroketenes cycloadditions carbonyl groups, 527-530 imines, 506-510 nucleophilic alkenes, 490-491 α-halo carboxylic acid derivatives, 87 halogen substitution, 331-340 4-Chloropyrazolin-5-one, dehydrochlorination, diazo ketene formation, 164 5-Chlorovalerylketene, dehydrochlorination, 67-68 Chromium carbene complexes aminoketenes, nitrogen substitution, 309-310 dienylkettene formation, 172-173 ketene stereoselectivity, 611

Chromium carbene complexes (continued) metal-complexed ketenes, 390-391, 398 stereoselectivity of ketenes, 608 Chromium complexed-ketenes, nucleophilic additions carbon/silicon nucleophilic additions, 574 nitrogen nucleophile/electrophile reactions, 561-564 Wittig reactions, 577-579 Claisen rearrangement, cycloaddition of ketenes alkene reactions, 478-484 [4+2] cycloadditions, 517 Cobalt complex, ketene formation, 172 Collisionally activated dissociation (CAD) deprotonated peptides, ketene formation, 168 methylketene formation, 188 resorcinol derivatives, alkenylketene intermediates, 150 Computational studies cycloadditions of ketenes, imine reactions, 501-510 ketene dipole moments, 48-49 ketene elucidation, 5-17 molecular and electronic structure and energy, 5-8 reaction mechanisms and energetics, 9 - 12substituent effects, 13-17 Concerted pathways, intermolecular [2+2] cycloaddition of ketenes, alkene reactions, 476-484 Cope rearrangement, cyclopropylketenes, 240Copper catalysis, diazo ketones, catalyzed Wolff rearrangement, 99-104 Copper ketenide, 387-388 Cotton effect, chiroptical properties of ketenes, 38 Cumulene structure cyanoketene formation, 217 imidoylketenes, 264 Cumulene-substituted ketenes, 5-8, 268-272 fluoroketenes, 322 Cumulenones carbon-substituted ketenes, 284, 292 α-oxoketene, 295 structure and classification, 428-433

(4-Cyanobutadienyl)ketene, 1-adizocycloheptatrienone formation, 145 Cyanoacetyl chloride, cyanoketene formation, 217 Cyanoketenes carbon substitution, 214-219 from ester enolates, 76 isodesmic comparisons, 217 oxygen nucleophile-ketene reactions, 554-559 Cyano ketenyl cation, 426-427 Cyanuric chloride, carboxylic acid-ketene formation, 59-60 Cyclic diacylbisketene argon matrix, 419 diazo ketone formation, photochemical Wolff rearrangement, 107-110 [2+2] Cycloadditions of ketenes, 450-451 carbonyl groups, 525-530 intermolecular interactions, 453-510 alkenes basic mechanisms, 464-472 nucleophilic, 487-491 rapid reactions, 474-485 alkynes, 495-499 allenes, 492-494 dienes, 464-472 dimerization. 453-462 imines, 500-510 Woodward-Hoffmann rules, 451 [3+2] Cycloadditions of ketenes, basic principles, 513-515 [4+2] Cycloadditions of ketenes, 450-451, 474 alkene reactions, 474-485 basic principles, 516-520 imine reactions, 504-505 Cycloadditions [2+2] cycloadditions intermolecular reaction, 453-510 alkenes basic mechanisms, 464-472 nucleophilic, 487-491 rapid reactions, 474-485 alkynes, 495-499 allenes, 492-494 dienes, 464-472 dimerization, 453-462

imines, 500-510 stereoselectivity of ketenes, 608-615 [3+2] cycloadditions, ketene formation, 513-515 [4+2] cycloadditions, ketene formation, 516-520 alkynyl ether thermolysis, 156-157 basic reaction mechanisms for ketenes, 449-452 carbonyl groups, ketene formation, 525-531 chloroketenes, 331-340 chromium carbene complexes, 178 dehydrochlorination of ketenes, 67-74 diazo ketone formation, photochemical Wolff rearrangement, 102-104 dioxinone formation of ketones, 153-155 fluoroalkylketene formation, 327 fluoroketenes, 323 intramolecular cycloadditions, ketene formation, 520-542 stereoselectivity, 609 nitrogen-substituted ketenes, 309 oxo-1,2-xylylene formation, 297 phosphorus-substituted ketenes, 365-368 photochemical Wolff rearrangement, ketene formation, 122-123 sulfur-substituted ketenes, 372-373 trimethylsilylketene, 350, 352 vinylketene formation, 10 carbon substitution, 211-212 Cycloalkanones, photolysis, ketene formation, 132-140 Cycloalkenones, ketene formation, 142-151 Cyclobutanediones, ketene dimeriziation, 460 1,3-Cyclobutanedione, ketene dimerization, 453 Cyclobutanones aryl ketene generation, 228-229 diphenylketenes, 464 intramolecular ketene cycloadditions, 521-522 ketene formation, photochemical and thermolytic methods, 114-129 ring opening reactions, thermolytic ketene formation, 160

Cyclobutenedione chemistry bisketenes, 401-402 ketene dimers, 458 Cyclobutenones allenyl thio ester thermolysis via vinylketenes, 161 carbon substituted diphenylketene, 225-226 ketene formation, photochemical and thermolytic methods, 114-129 ketene molecular dynamics, 9 solvent effects on ring opening, 119 substituent effects on ring opening, 120 thermal rearrangement from cyclobutenedione, 121 vinylketene formation, 210-211 Cycloheptatrienylketene, dehydrochlorination, 239-240 Cyclohexadienones, ketene formation, 142 - 151Cyclohexanone enamines, ester enolate formation of ketenes, 80-81 Cyclohexylamine, dienylketene formation, 147 Cyclohexylketene, alkynyl ether thermolysis, 158 2,4,6-Cyclooctatrienone, hexatrienylketene formation, 150 Cyclopentadienones, cyclopropenylketene formation, 242 Cyclopentadienylideneketene, cumulenone formation, 432 Cyclopentenone photolysis cyclopropylketene interconversion, 238 ketene formation, 132 Cyclopenten-4-one, ketene intermediates, 134 Cyclopropanocyclopentenone, ketene formation, 135-136 Cyclopropenium cation, sulfur-substituted ketenes, 372 Cyclopropenylketenes argon photolysis, 138 carbon substitution, 237-242 Wolff rearrangement, 242 Cyclopropyl carboxylate ester, dichloroketenes, 334 Cyclopropylideneketenes carbon substitution, 196

Cyclopropylideneketenes (continued)  $\alpha$ -halo carboxylic acid derivatives, ketene formation, 87 photolysis, 444 Cyclopropylketenes carbon substitution, 237-242 diazo ketone formation, catalyzed Wolff rearrangement, 100-104 radical reactions, 596 Cyclopropylvinylacyl selenides, ketenyl radicals, 278-279 Decarbonylation benzoylphenylketene photolysis, 443 carbon-substituted acylketenes, 248-249 fluoroketenes, 322 ketene thermolysis, 447-449 metal-substituted bisketenes, 411-412 oxygen-substituted ketenes, 314-320 silylketenes, 355-356 Decarboxylation, ketene cycloadditions, 526-530 n-Decyltrimethylsilylketene, 356 Decomposition rates diazo ketone ketene formation photochemical Wolff rearrangement, 104-110 thermal Wolff rearrangement, 94-97 phosphorus-substituted ketenes, 362 photochemical ketene formation, cyclobutanones, 116 Dehalogenation de-(2-naphthyl)ketene, 222  $\alpha$ -halo carboxylic acid derivatives, ketene formation, 86-89 phenyl(trifluoromethyl)ketene, 328-329 Dehydrobromination, ketene generation, 70-71 Dehydrochlorination acetoxylketene, 316 adamantylketenes, 201 alkynylketene formation, 215 azidoketene, 308 benzyloxyketene, 317 BHT ester formation of ketene, 342-343 1,4-bisketenes, 402-403 bromoketenes, 337-340 carbon-substituted acylketenes, 255 chloroketenes, 331-332

4-chloropyrazolin-5-one, diazo ketene formation, 164 cycloaddition of ketenes imine reactions, 505-506 intramolecular cycloadditions, 523 nucleophilic alkenes, 488-489 cycloheptatrienylketene, 239-240 cyclopropylketene formation, 239 diazoalkylketenes, 272 diethylphosphonoketene, 364 di-tert-butylketene, 197 fluoroketenes, 323 ketene preparation and isolation, 65-74 ketene stereoselectivity, 610-611 metal-substituted bisketenes, 414-415 stereoselectivity of ketenes, 604-605 sulfur-substituted ketenes, 376 trimethylsilylvinylketene, 344 Dehydrohalogenation alkenylketenes, 208 cyclopropylideneketene, 196 fluoroketenes, 323 ketene preparation and isolation, 65-74 Density functional theory (DFT) alkyne oxidation, ketenes from, 183-185 aryl chromium carbene complex, ketene formation, 173-175 [4+2] cycloadditions of ketenes, 518-519 cyclopropylketene substituent effects, 240ketene molecular structure, 6 ketenide formation, 388 ketenyl radicals, 281 vinylketene cycloadditions, 10 Deuterated ketenes, phhotofragmentation, 442 Deuterium isotope studies, alkynyl ether thermolysis, 156-159 Dewar bicyclo isomer, photochemical ketene formation, 163 Diacetylketenes carbon-substituted acylketenes, 252 dioxinone thermolysis, 155 Diacyl chloride, metal substituted bisketenes, 418 Diacylketenes carbene capture, 256 carbon substitution, 246-247 spectroscopic analysis, 33

Dialkylketenes dimer photolysis, 56 nucleophilic additions, hydration mechanisms, 539-540 Diarylalkynes, metal-complexed bisketenes, 409-410 7,7-Diarylbicyclo[2.2.1]hept-2-enes, kettene formation, 200 Diarylketenes carbon/silicon nucleophilic additions, 570-571 cycloaddition reactions, alkynes, 497-498 dehydrochlorination and formation of, 66 diazo ketone formation, thermal Wolff rearrangement, 94-97 NRM spectroscopy for, 27-33 1,4-Diazabicyclooctane (DABCO), nitrogen-substituted ketenes, 306 Diazoalkanes, [3+2] cycloadditions of ketenes, 513 Diazoalkylketenes, cumulene substitution, 272 Diazoketene, nitrogen substitution, 306 Diazo ketones arylketene photelectron spectra, 438-440 azafulvenone formation, 288 carbon-substituted ketenes, 226-234 acylketenes, 246 oxoquinone methides, 296 chromium carbene complex, 177 cumulene-substituted ketenes, 268-272 cumulenone formation, 431 [3+2] cycloadditions of ketenes, 514–515 cyclohexylketene formation, 202 di-tert-butylketene formation, 199 fluoroalkylketene formation, 325-326 intramolecular ketene cycloadditions, 524 ketene formation, 90-110 catalyzed Wolff rearrangement, 97-104 cyclopropylketene, 239 diazomethane reactions, 620 epimeric anilide capture, 241 phenylsulfonylketene, 374 phosphorus-substituted ketenes, 362 photochemical ketene formation thermolytic reactions, 117-118  $\alpha$ -trialkylsilylalkenylketenes, 208 photochemical Wolff rearrangement, 104 - 110

photolysis, 201 reaction mechanisms, 90-93 thermal Wolff rearrangement diphenylketene formation, 221-222 ketene formation, 93-97 Wolff rearrangement, overview, 92-93 Diazomethanes, ketene reactions with, 618 - 620Diazomethylketene, cumulene substitution, 268 3-Diazo-4-oxocoumarin, carbon-substituted ketenes, 227 Diazulenylketene, thermal Wolff rearrangement, 226 Dibenzofulvenone carbon substituted diphenylketene, 225 zinc dechlorination, 290 Dibenzoylketene furandione pyrolysis, 246 furandione thermolysis, 164 Dibromofuranone, metal-complexed bisketenes, 411 Dibromoketene α-halo carboxylic acid derivatives, ketene formation. 87-88 zinc generation, 339 Di(carboethoxy)ketene, Wolff rearrangement, diazo ketone, 256 - 257Dichloroketenes cycloadditions, 333-335 [2+2] cycloadditions alkenes and dienes, 469-471 alkynes, 496 imines, 507-508 nucleophilic alkenes, 487-489 [3+2] cycloadditions, 513-515 carbonyl groups, 528-529  $\alpha$ -halo carboxylic acid derivatives, 87 halogen substitution, 331-340 nitrogen nucleophile/electrophile ketene reactions, 565-566 oxygen nucleophile reactions, 557-559 protonation, ketone electrophilic reactions, 582-583 stereoselectivity, 608-609 sulfur dioxide reactions, 621 Dicyanoketene, carbon substitution, 219

Dicyclohexyl carbodimid (DCC), carboxylic acid-ketene formation, 58 Dicyclopropylketene, carbon substitution, 241 Dideutroketenes, radical reactions, 598 Dieckmann-type reactions oxygen nucleophile ketenes, 558 ynol/ynolates, 381-382 Dienes [2+2] cycloaddition of ketenes, 453-510 basic mechanisms, 464-472 nucleophilic, 487-491 rapid reactions, 474-485 [4+2] cycloaddition of ketenes, 520 Diene-yne ketene, alkynyl-substituted cyclohexadienone cyclization, 150 Dienylketenes acid anhydride pyrolysis of, 63 alkenyl carbene complexes, 391-392 chromium complexes, 172-173 cycloaddition reactions with imines, 505 cyclohexadienone interconversion, 143-151 ene-yne epoxide conversion, 397-398 nucleophilic additions, amination mechanisms, 548-549 radical reactions, 596 ring-opening mechanism, 325 stereoselectivity, 606 Diethylketene, stereoselectivity, 612 Diethylphosphonoketene, dehydrochlorination, 364 1,1-Diethynyldiene hydration, deprotonated peptides, ketene formation, 168 Difluoroketenes photolysis, 322 X-ray structure, 324 zinc activation, 323-324 Diformylketene, formamide reactions, 250 Dihydrobenzophenanthridine, dihydroisoquinoline synthesis, 125-126 Dihydrofuran, bis(silyl)kettenes, 346 Dihydroisoquinoline synthesis, ene-yne ketene formation, 125-126 Dihydropyridine-substituted ketene, eyectrocyclication reaction, 126 Dihydropyridone, intramolecular nucleophilic addition reaction, 126

Diisopropylethylamine (DIPEA) ketene cycloadditions, 529-530 ketene stereoselectivity, 610-611 Dimerization alkvlketenes, 195 chloroketenes/dichloroketenes, 336-337 dehydrohalogenation of ketenes and, 66 intermolecular [2+2] cycloadditions in ketenes, 453-462 ketene molecular dynamics, 9-12 ketenes, preparation from, 55-56 stereoselectivity of ketenes, 613-615 Dimesitylketenes carbon/silicon nucleophilic additions, 570-571 photolysis of, 444 X-ray studies of, 20 (2,4-Dimethylbenzoyl)formate esters, ketene formation, 168 3,3-Dimethylcyclopentene, ketene cycloadditions, 469 (Dimethylhydridosilyl)ketenes, flash vacuum pyrolysis, 449 Dimethylketenes alkylketene preparation, 193 bicyclo[2.2.2]jexadienone formation of, 133 cycloaddition reactions [2+2] cycloadditions dimerization, 454 imine reactions, 507 symmetrical dimers, 450 sytrenes, 483-484 [3+2] cycloadditions, 514 dimer pyrolysis, 55-56 α-halo carboxylic acid derivatives, ketene formation, 88-89 hydrogen radical reaactions, 593-594 ketone electrophilic reactions, oxygenation, 588 nucleophilic additions amination mechanisms, 544-545 carbon/silicon nucleophilic additions, 573 Wittig reactions, 577-579 photolysis in argon matrix, 442 pyrolysis, 447 thermochemistry, 22-25

Dimethylpropadienone, cumulenone formation, 430-431 Dinaphthylketenes, α-halo carboxylic acid derivatives, 86-89 Dioxinones carbon-substituted acylketenes, 247-248 carbon substitution, 247-248 ketene formation, 153-155 4,4-Diphenylcyclopentenone, keetene formation, 134 Diphenylcyclopropenone ketene formation, 166 thermolysis, ketene formation, 190 2,2-Diphenylcyclopropylketene, photochemical Wolff rearrangement, 238 Diphenylketenes boron substitution, 386 carbon substitution, 225-226 cycloaddition reactions [2+2] cyclyadditions, 451 alkene reactions, 478-484 with alkenes and dienes, 464-472 imine reactions, 500-510 net [2+2] cycloaddition, 450 norbornene, 472 sytrenes, 483-484 titanaalene complexes, 395 [3+2] cycloadditions, 513-514 [4+2] cycloadditions, 451, 516-518 carbonyl groups, 530 dehydrochlorination preparation of, 66-67 diazo ketone formation, thermal Wolff rearrangement, 93-97 diazomethane reactions, 618 dimeric structures, 222-223 diphenyldiazomethane reaction, 169 electrochemical oxidation, 439 α-halo carboxylic acid derivatives, 86-89 history of, 1 ketone electrophilic reactions oxygenation, 587-591 protonation, 581-585 lanthanum complexes, 398 nucleophilic additions carbon/silicon nucleophiles, 568-574 carbon/silicon nucleophilic additions, 571-574 hydration mechanisms, 535

hydride reduction, 552-553 nitrogen nucleophile/electrophile ketene reactions, 565-566 Wittig reactions, 576-579, 578-579 photolysis of, 443 polymerization, 601 stereoselectivity, 614 thermolytic reactions, 446-449 zinc reduction characterization, 221 Diphenyloxazolidinyl-substituted ketene, cycloadditions with imines, 507 Dipivaloylketene, X-ray studies of, 20 Dipivalylketene, furandione pyrolysis, 245 Dipole moments bisketenes, 402-403 ketene measurements. IR and Raman spectroscopy, 39-45 ketene physical properties, 48-49 molecular structure of ketenes, 18-21 Disilanylbisketene, 352-353 Disilanylketene, 352-353 Distonic dehydrobenzoyl radical cation, carbon-substituted ketenes, 275-276 Divinylketenes carbon substitution, 213 ethenylcyclobutenone formation, 123-124 DNA binding reactions, chloroketenes, 336 Dötz reaction alkenylketene chromium complex, 171-172 chromium carbene photolysis, benzannulation, 175-176 metal-complexed ketenes, 390-398 vinylcyclopropyl carbene complex, 179 E-(1,3-butadienyl)ketene, dehydrochlorination, 206 E-7 ketene, ozone photolysis, 143 Electron diffraction, molecular structure of ketenes, 18-21 Electronegativities chloroketenes, 333 ketene substituent effects, 14-17 Electronic structure chiroptical properties of ketenes, 35-38 diazo ketones, 92-93 ketenes, 5-8 Electron impact mass spectra acyl chloride ketenes, 49-50

Electron impact mass spectra (continued) ketene oxidation and reduction, 438-440 Electrophilic reactions [2+2] cycloaddition reactions of ketenes, 475 ketones, 580-591 oxygenation of ketenes, 587-591 protonation of ketenes, 580-585 nitrogen reactions with ketenes, 560-567 Electrospray mass spectrometry diazo ketones, catalyzed Wolff rearrangement, 98-104 ketene radical cations, 50-51 Enaminylketenes, flash vaccuum thermolysis, 265 Enantioselectivity of lactams, cycloadditions of ketenes, imine reactions, 501-510 Endo positions, ketene cycloadditions, 466-468 Enediynyl ethyl ether thermolysis, ene-yne ketene, 158-159 Ene reactions, alkenylketene formation, 209 Energy changes, ketene radical reactions, 592-598 Ene-yne ketenes bisalkynyl chromium carbene complex, 392 carbon substitution, ruthenium catalyst, 212 - 213diazo ketone ketene formation matrix photolysis, 211 photochemical Wolff rearrangement, 109 - 110dihydroisoquinoline synthesis, 125-126 enediynyl ethyl ether thermolysis, 158 - 159intramolecular cycloadditions, 523-524 vinylcyclobutenone formation, 123-125 Enolic radicals, ketene reactions, 592-598 Enone photolysis, ketene formation, 132-140 Enthalpy of formation, ketene thermochemistry, 25 Ester enolates bisketene formation and, 406-407 fulvenone formation, 291-292 ketenes from, 76-81 Esterification acyl chloride ketenes, 71

bicyclic ketone hydrogen transfer, 132-133 diazo ketone formation of ketenes, 104 photochemical Wolff rearrangement, 105 - 110ketene preparation, 67 photochemical ketene formation, cyclobutanones, 116 stereoselectivity of ketenes, 607-608 Esterolysis melt polymerization, ketene formation, 84 Esters, ketenes from, 76-84 ester enolates, 76-81 pyrolysis, 81-84 Et₃N ketene formation, esterification and, 67 Ethanol, oxygen-substituted ketenes, 316 Ethenylcyclobutenone, divinylketene formation, 123-124 Ethoxyalkynylsilanes, silylated ketenes and polyketenes, 158 Ethoxyketene, oxygen substitution, 313 Ethoxy(triphenylsilyl)ketene, preparation, 319 Ethylene, ketene cycloadditions, 485 Ethylketenes acrolein photochemical isomerization, 167  $\alpha$ -halo carboxylic acid derivatives, 86-89 hydrogen radical reaactions, 593-594 ketone electrophilic reactions, oxygenation, 588 quinoline derivatives and formation of, 73 vinyl and saturated ester formation, 83 Ethylphenylketene polymer supports for, 624 radical reactions, 598 Ethynyl ethers, mass spectrometric analysis, 159 Exo positions ketene cycloadditions, 467-468 Femtosecond time-resolved mass spectrometry, photochemical ketene formation, ring opening mechanism,

115

Fermi resonance, ketenylidenecycloalkane infrared spectra, 41–45

Ferrocenylketene, carbon substitution, 234

Flash vacuum pyrolysis (FVP) allenylketenes, 269-270 azafulvenone formation, 287-288 carbon-substituted acylketenes, 250 - 251carbon-substituted alkenylketenes, 206-207.211 cumulenone formation, 431 diazo ketone ketene formation, 96-97 (dimethylhydridosilyl)ketenes, 449 imidoylketenes, 264 Meldrum's acid derivative. carboxylketene, 252 oxoquinonemethides, 302 perfluoroketene formation, 325 Flash vacuum thermolysis (FVT) carbon-substituted acylketenes, 248 cyanoketene formation, 217 enaminylketenes, 265 oxygen-substituted ketenes, 318-319 2-pyridylcyanoketene, 233 Flowing afterglow analysis, ynol/ynolate ketenes, 377-382 Fluorenylideneketene diazo ketone formation, thermal Wolff rearrangement, 93-97 zinc dechlorination, 290 Fluoroalkylketenes, halogen substitution, 321, 325-329 Fluoroketenes, halogen substitution, 321-329 Fluorooxirane irradiation, ketene formation, 166 Formaldehyde photolysis, oxygensubstituted ketenes, 313 Formylketene carbon substitution, 244 computational studies, 11 cyclization reactions, 249-250 cycloaddition reaactions, 255 Formylthioketene, pyrolysis, 259 Fourier transform mass spectrometry, carbon suboxide, cumulenones, 429-433 Free radicals ketene formation from, 181–182 ketene radical cation spectra, 50-51 Frequency parameters, ketene IR frequencies, 40-45

Friedel-Crafts reaction, carbon/silicon nucleophilic additions to ketenes, 572-573 Fullerenes bisketene formation and, 413 oxidation, ketenes from, 185 pentaketene formation, 421 Fulvenones carbon-substituted ketenes, 284-292 nitrogen nucleophile/electrophile ketene reactions, 564 nucleophilic additions to ketenes. hydration mechanisms, 539 oxyquinone methides and oxoxylylenes, 298-299 Furandione acylketene formation, 246 carbon-substituted arylketenes, 233-234 dipivalylketene formation, 245 pseudopericuclic decarbonylation, ketene formation, 165 thermolytic ketene formation, 164 Furanone, cobalt complex, ketene formation, 172 Furan photolysis/pyrolysis, ketene formation, 167 Furaquinocin, acid anhydride pyrolysis of ketenes, 62-64 Furaquinocin E, cyclobutenone formation of vinylketene, 122 G3(MP2)//B3LYP, ketene computational studies, 16-17 Gas phase ion chemistry bromoketenes, 337-340 cumulenones, 429-433 ketene oxidation reactions, 437-440 ketene physical properties, 49-52 ketene thermolysis, 447 ketenyl radicals, anions, and cations, 4 26 - 427nucleophilic additions to ketenes, hydration mechanisms, 535 oxygen-substituted ketenes, 314 vinylketenes, 206 ynol/ynolate ketenes, 377-382

Gas phase IR laser photolysis, chloroformylketene formation, 72
Gas phase pyrolysis, cyanoketene molecular structure, 18-21 Gas phase thermolysis, ethoxyalkynylsilanes, silylated ketenes and polyketenes, 158 Gas phase UV spectra, ketenes, 35 Geometric properties, bisketenes, 401-402 Germylketenes classification and formation, 342-357 nitrogen nucleophile/electrophile reactions, 562 nucleophilic additions, hydride reduction, 552-553 Guanine, notrosative deamination, 303 Hafnium ketenylidene complexes, 388 α-Halo carboxylic acid derivatives, dehalogenation, ketene formation, 86-89 Halo derivatives chloroketenes, olefin bioactivation, 336 dioxinone formation of ketones, 154 ketene reduction, 440 Halogen-substituted ketenes (haloketenes) chloro- and bromoketenes, 331-340 fluoroketenes and fluoroalkyletenes, 321-329 intermolecular [2+2] cycloaddition and ketene dimerization, 458 Heats of formation, ketene thermochemistry, 22-25 Heavy-atom chain deformation, cumulenones, 430-433 Heck cross-coupling reactions, polymer supports for ketenes, 622-625 Heptafulvenone, 284, 291-292 Heteroarylketenes, carbon-substituted alkenylketenes, 221-234 Heteroatomic analogues [4+2] cycloadditions of ketenes, 517 oxoquinonemethides, 300 Hexachlorocyclohexadienone, bromoketenes, 337-338 Hexafluoroacetone, fluoroalkylketene formation, 326-327 Hexaketene, metal substitution, 411 Hexatrienylketene, 2,4,6-cyclooctatrienone photolysis, 150

Highest occupied molecular orbital (HOMO) ketenes, 5-8 [2+2] cycloaddition, 479-480 dimerization, 458-459 oxidation reactions, 437-440 photoelectron spectra, 37-38 nucleophilic additions to ketenes, hydration mechanisms, 533-534 sulfur-substituted ketenes, 375 Homer-Wadsworth-Emmons reactions, carbon-substituted ketenes, 229 Hückel-Möbius systems, [2+2] cycloaddition reactions of ketenes, 475 Hydration carbon-substituted acylketenes, 254 ketene computational studies, 11 nucleophilic ketene additions, mechanisms for, 533-541 oxygen-substituted ketenes, 315 silylketenes, 354 sulfur-substituted ketenes, 372 Hydride reduction, nucleophilic additions to ketenes, 552-553 Hydrogenation, of ketenes, 621 ¹H nuclear magnetic resonance alkvlketenes, 194 carbon-substituted ketenes, 274-283 ketene physical properties, 27-28 Hydrogen transfer alkynyl ether thermolysis, 156-159 cycloalkanone photolysis, 132 diazo ketone formation, 96-97 fluoroketenes, 322 phthalaldehyde photolysis, 297-298 Hydroperoxyketene, ozone photolysis, 319 Hydroxyalkynylalkane, chromium carbene complex, lactone formation, 176 Hydroxycubane, photochemical ketene formation, 118 Hydroxycumulenones, molecular structure, 7 Hydroxycyclobutenone oxidation, furanone formation, 123 Hydroxycyclopentenone photolysis (hydroxycyclopropyl)ketene, 140

lactone formation, 240 Hydroxyketene hydration of, 315 nonplanar geometry, 16 zwitterionic intermediates, 316 4-Hydroxyketene, 392 Hydroxylketene, alkyne cycloaddition reactions, 498-499 Hyperconjugative electron donation, silvlketene stabilization, 16 H-ZSM-5 zeolite-catalyzed methanol conversion to gasoline, ketene intermediates, 186 IGLO method, ketene physical properties, spectroscopic analysis, 29, 33 Imidoylketenes carbon substitution, 263-267 [4+2] cycloadditions, 518 intramolecular cycloadditions, 520-524 Imines [2+2] cycloadditions of ketenes, 452, 500 - 510zwitterionic intermediates, 485 nitrogen nucleophile/electrophile reactions with ketenes, 560-567 6-Imino-2,4-cyclohexadien-1-one, ketene formation, 149 Iminoketenes elimination reactions, 301 nitrogen substitution, 308 N-Imino-β-lactams, ketene cycloadditions, 504 Indenylketene, fulvenone formation, 288 - 289Infrared (IR) dissociation, acetic acid-ketene formation, 57-60 Infrared (IR) frequency, fluoroketenes, 322 Infrared (IR) spectroscopy ketene physical properties, 39-45 phosphorus-substituted ketenes, 361-362 Infrared laser-powered pyrolysis (IRLPP), ketenes from acid anhydrides, 61-64 Intermolecular [2+2] cycloadditions of ketenes, 453-510 alkenes basic mechanisms, 464-472 nucleophilic, 487-491 rapid reactions, 474-485

alkynes, 495-499 allenes, 492-494 dienes, 464-472 dimerization, 453-462 imines, 500-510 Intramolecular cycloadditions of ketenes [2+2] cycloadditions, 479 basic principles, 520-524 Intramolecular rearrangements carbon-substituted acylketenes, 253-254 oxygen-substituted ketenes, 318-319 "Inversely fused" bicyclic lactams, ketene cycloadditions with imines, 504 Ionization potentials (IPs) arylketenes, 438-440 photoelectron spectra of ketenes, 37-38 Iridium complex difluoroketene formation, 324 metal-substituted ketenes, 393-394 Iron ketene complex, trimethylsilylketene, 395-396 Irradiation chromium complexes, ketene formation, 173 - 174diazo ketone ketene formation, photochemical Wolff rearrangement, 107-110 IR-UV double resonance photolysis, ketene measurements, 39-45 Isocyanatoketene, cumulene substitution, 268, 272 Isocyanoketene, nitrogen substitution, 308 Isodesmic reaction alkylketenes, 193 carbon-substituted alkenylketenes, 207 diphenylketene preparation, 221 fluoroketene destabilization, 321-322 ketene substituent effects, 13-17 mercaptoketene, 371 oxygen-substituted ketenes, 313 phosphorus-substituted ketenes, 361-362 silylketenes, 353 Isolable acyl ketene, ynol/ynolates, 378-379 Isomeric ketenes cycloaddition with imines, 508 photolysis, 299 structure, 6-7 Isopropenyl ester pyrolysis, ketene formation. 81-82

Isopropenyl ether, cycloadditions of ketenes, nucleophilic alkenes, 489-490 Isotope effects ketene cycloadditions, 484-485 nucleophilic additions to ketenes, hydration mechanisms, 540 Isoxazol-5(2H)-ones, ketene formation, alternate routes, 189 Isoxazole-substituted ketenes, diazo ketone formation, 97 Ketene-Cope reaction, diazo ketone formation, catalyzed Wolff rearrangement, 100-104 Ketene dimers, ketene preparation from, 55-56 "Ketene equivalents," cycloaddition reactions, 450-451 Ketene radical cations, mass spectra of, 50 - 51Ketenes historical background, 1-3 theoretical and computational studies, 5 - 17molecular and electronic structure and energy, 5-8 reaction mechanisms and energetics, 9 - 12substituent effects, 13-17 Ketenimes, nucleophilic additions, Wittig reactions, 578-579 Keteniminylketene, cumulene substitution, 268, 271–272 Ketenyl acylium ions, carbon-substituted ketenes, 276-283 Ketenylidenecycloalkanes, infrared spectra, 41-45 Ketenylidene phosphoranes, 366-367 Ketenyl infrared (IR) stretching frequencies, ketene physical properties, 27-33 Ketenyloxyl radical, possible formation of, 319 Ketenyl radicals, anions, and cations, 425-427  $\beta$ -Keto acids, intermolecular [2+2] cycloaddition and ketene dimerization, 456 Ketocarbenes, rearrangements, routes for, 110

 $\alpha$ -Ketocarbenes, pentafulvenone formation, 286 β-Keto ester, ketene pyrolysis, 448 Ketoketenes decarbonylation, 200 intermolecular [2+2] cycloaddition and dimerization, 454-455 polymerization, 602 Ketones. See also Diazo ketones electrophilic reactions, 580-591 oxygenation of ketenes, 587-591 protonation of ketenes, 580-585 Kinetic energy release, ketene mass spectra, 51 Kohlrausch kinetics, nitrogen nucleophile/ electrophile ketene reactions, 564-565 Kowalski procedures, ynolate ketenes, 379-380 **B**-Lactams aminoketenes, nitrogen substitution, 391 azoketene dehydrochlorination, 308  $\alpha$ -bromo- $\alpha$ -alkenylketenes, 340 di-tert-butylketene and formation of, 197-198 diphenylketene cycloaddition, 450 ketene cycloaddition facility, 450-451 imine reactions, 500-510 ketene formation, 71-74 nitrogen-substituted ketenes, 310 phenylketene generation, 223-224 polymer supports for ketene reactions, 622-625 stereoselectivity of ketenes, 612-615 sulfur-substituted ketenes, 372 ynol/ynolates, 380  $\alpha$ -Lactones, arylmethylketenes, 232 β-Lactones acid anhydride pyrolysis of ketenes, 62-64 alkylketenes, 194-195 aryl chromium carbene complex, ketene formation, 175 bisketene formation and, 408-409 chromium carbene complex, hydroxyalkynylalkane reaction, 176-177 [2+2] cycloaddition of ketenes, 453-462 carbonyl groups, 525-530

diphenylketene cycloaddition reactions, 450 ketene molecular dynamics, 9-12 trimethylsilylketene formation, 350-351 ynol/ynolates, 379-380 Lactone formation difluoroketenes, 322 oxygen nucleophile reactions, 555-559 photochemical ketene formation, 139 stereoselectivity of ketenes, 610-615 Lanthanum enolate, ketene polymerization, 601 Laser-ablated metal atoms, ketenide formation, 388 Laser flash photolysis cyclohexadienones, ketene formation, 145 ketene/bisketene infrared spectra, 41-45 Lewis acids carbon-substituted ketenes, 277 ketene cycloadditions, 526 nitrogen nucleophile/electrophile ketene reactions, 563-564 Linkage-isomeric ketene complex, 396 Lipase enzyme, silylketene formation, 358 Liquid ketene irradiation, 444 Lithium compounds bis(silylated) ketenes, 351-352 carbon/silicon nucleophilic reactions, 569 ester enolate formation of ketenes, 79 ketene computational studies, 11-12 ketene cycloadditions, 527-530 nucleophilic additions to ketenes, hydride reduction, 552-553 silylketene formation, 348 ynol/ynolates, 381-382 Lithiumketene, stabilization properties, 387 Lowest unoccupied molecular orbital (LUMO) ketenes, 5-8 [2+2] cycloaddition, 479–480 dimerization, 458-459 oxidation reactions, 437-440 nucleophilic additions to ketenes, hydration mechanisms, 533-534 Lumiphotosantonin, photochemical rearrangement to photosantonic acid, 139-140

Macrocyclic lactams, nitrogen nucleophile/ electrophile reactions, 562-565 Maganese ketene complexes, 396-397 Magnetic susceptibility exaltation, fulvenones, 284 Malherbe-Bellus rearrangement, oxygen nucleophile ketene reactions, 557-559 Malonic acids, acid anhydride pyrolysis of ketenes, 63 Masochistic steric effect, cycloaddition reactions of ketenes, 466 Mass spectrometry ethynyl ethers, 159 fluoroketenes, 323 ketene physical properties, 49-52 Meldrum acid derivatives alkenylketene formation, 209 cumulenone formation, 431 cyanoketene formation, 217 diazo ketone ketene formation, photochemical Wolff rearrangement, 107 - 108dioxinone formation of ketones. 154-155 flash vacuum pyrolysis, carboxylketene, 252 imidoylketenes, 264-265 ketene cycloadditions, imines, 509 nitrosoketene formation, 307 pyrolysis, ketene formation, 448 silylketenes, 354 thermolysis, acylketene formation, 161 trifluoroacetylketene, 257 Mercaptoketene, 371 Mesitylketene, nucleophilic additions, hydration mechanisms, 538 Mesitylphenylketene carbon/silicon nucleophilic reactions, 569-570 computational studies, 11-12 X-ray studies of, 20 Mesoionic species, carbon-substituted acylketenes, 251 Metal carbene complexes, ketenes from, 171 - 179Metal ketenides, classification, 387-389 Metal oxides, [3+2] cycloadditions of ketenes, 515

Metal-substituted ketenes bisketenes, 397-398 boron-substituted ketenes, 384-386 electropositive substituents, 387-389 metal-complexed ketenes, 389-398 stabilization of, 387-389 ynols and ynolates, 377-382 Methane, ketene substituent effects, 13 - 17Methoxymethyl esters, ketene formation, 80 Methyl benzoate, ynol/ynolates, 382 Methyl diazo malonate, pyrolysis and ketene formation. 448 5-Methyl-1,5-hexadiene-3-one, 3-butenylketene formation, 136 Methylketenes acid anhydride pyrolysis of, 62 dehydrochlorination, 202 fluoroketene destabilization, 321-322  $\alpha$ -halo carboxylic acid derivatives, 86-89 hydrogen radical reaactions, 593-594 infrared spectra of, 40 intermolecular [2+2] cycloaddition and dimerization, 454-457 ketone electrophilic reactions oxygenation, 588 protonation, 581 nucleophilic additions, Wittig reactions, 576-577 stereoselectivity, 610, 614-615 thermochemistry, 22-25 vinyl and saturated ester formation, 83 Methyl(phenyl)ketene, stereoselectivity, 605 5-Methyl-pyrones, thermolytic ketene formation, 163 Methylrhenium trioxide, alkyne oxidation, ketenes from, 183-184 Methvlthioketene substituent effects, 372 Wolff rearrangement, 371 (Methylthio)phenylketene, 372 Micelle reactivity, ester enolate formation of ketenes. 78 Michael addition reactions, ester enolate formation of ketenes, 79 Microwave-assisted Wolff rearrangement, diazo ketone ketene formation, 95-97

Microwave spectroscopy alkyl- and arylketenes, thermal Wolff rearrangement, 200 cyanoketene formation, 217 fluoroketenes, 322 ketene dipole moments, 48-49 molecular structure of ketenes, 18-21 Mixed ketenes, carbon substitution. 227 - 228Molecular beam electronic resonance. ketene dipole moments, 48-49 Molecular ion peaks, ketene mass spectra, 50 Molecular orbital calculations. See also Highest occupied molecular orbital; Lowest unoccupied molecular orbital diazo ketones, 92-93 intermolecular [2+2] cycloaddition of ketenes, alkene reactions, 477-484 metal-complexed bisketenes, 410 metal-complexed ketenes, 391-398 Molecular structure of ketenes determination techniques, 18-21 molecular orbital calculations, 5-8 Monoalkylketenes dimerization, 196 intermolecular [2+2] cycloaddition and dimerization, 454-455 Monoketenes metal substitution and, 416 thermochemistry of, 24-25 Monophenyl bisketene, metal substitution, 403 Mukaiyama's reagent bisketene formation, 420 carboxylic acid-ketene formation, 59 intramolecular ketene cycloadditions, 522-523 ketene cycloadditions carbonyl groups, 530-531 imine reactions, 507-510 Mulliken population analysis, intermolecular [2+2] cycloaddition of ketenes, alkene reactions, 477-484 "Münchnone" tautomers benzoylated aminoketenes, 306 nitrogen-substituted ketenes, 311 Muonated methyl radicals, liquid ketene irradiation. 444

Naphthol chromium carbene complex, ketene formation, 175 ketene formation and, 72 photochemical ketene formation. cyclobutenone mechanisms, 118 Naphthoquinone-substitued porphyrins, cyclobutenone formation of vinylketene, 121 Naphthoquinone synthesis, ketene formation, 120 Naphthylketenes cycloaddition reactions alkynes, 498 [4+2] cycloadditions, 517-518 radical reactions, 596 "Near-concerted" processes, ketene cycloadditions, 468 Negative ion mass spectroscopy, ynolate ketenes, 380 Neopentyl(methyl)ketene, dimerization, 459 "Neutral hyperconjugation," ketene physical properties, spectroscopic analysis, 29-33 Niobium complex, ketene formation, 394-395 Nitrate radicals, ketene reactions, 597 Nitric oxide bisketene formation and, 413 ketenyl radicals, anions, and cations, 425-427 Nitrogen matrix, chromium carbene complex, ketene formation, 175 Nitrogen nucleophiles/electrophiles, ketene additions, 560-567 Nitrogen-substituted ketenes basic properties, 305-311 stereoselectivity, 610 Nitroketenes, formation of, 306-307 Nitronate-substituted ketenes, photochemical rearrangement, 265-266 Nitrones, ynol/ynolates, 381-382 (2-Nitrophenyl)diazomethane, photolysis in argon matrix, 300 4-Nitrophenylketene, from ester enolates, 77 Nitrosoketene, formation of, 307 Nitroso(phenyl)ketene, nitric oxide formation, 307

Nitroxyl radicals, ketene reactions, 594-598 "No Barrier Theory," hydration mechanisms, nucleophilic additions to ketenes. 533 Nonafulvenone, 284 Norbornadiene ketene cycloadditions, 471-472 metal substituted bisketenes, 417 thermolytic ketene formation, 162 Norbornene, ketene cycloadditions, 471 Norbornylideneketene diazomethane reactions, 619-620 dimeric structure, 55-56 ring-opening polymerization formation, 187 Norrish type 1 photocleavage bicyclic keto sulfide photolysis, 133 cycloalkanone photolysis, 132 photochemical ketene formation, ring opening mechanism, 114-129 Novolac-diazonaphthoquinone resists, carbon-substituted ketenes, 227 Nuclear magnetic resonance spectroscopy (NMRS), ketene physical properties, 27 - 33Nucleophilic alkenes, ketene cycloaddition, 487-491 intramolecular cycloadditions, 522-523 Nucleophilic ketene additions amination mechanisms, 543-551 hydration mechanisms, 533-541 preparative aspects, 551-574 carbon and silicon nucleophiles, 568-574 hydride reduction, 552-553 nitrogen nucleophiles and electrophiles, 560-567 oxygen nucleophiles, 554-559 Wittig reactions, 576-579 Nucleus independent chemical shifts, fulvenones, 284 Oligomeric products, chloroketenes, 337 Oligonucleotides, diazo ketone formation, photochemical Wolff rearrangement,

101–104

Organolithium reagents, keten reactions with, 569

Organometallic reagents, ketene additions, 570 Osmium tetroxide, [3+2] cycloadditions of ketenes, 515 Oxazinone imidoylketene formation, 266 photochemical ketene formation, 163-164 Oxetanones, trimethylsilylketene formation, 349-350 Oxidation reactions alkynes, ketenes from, 183-185 E-7 ketene production, 143 ketenes alternate formation routes, 189 mechanisms of, 437-440 Oxiranylketenes carbon substitution, 237-242 oxygen nucleophile reactions, 558-559 photochemical Wolff rearrangements, 242 Oxirene intermediate diazo ketones, 92-93 ketene thermolysis, 447-448 Oxirene intermediates, ketene photolysis, 441-442  $\alpha$ -Oxoketene carbon substitution, 295 nucleophilic addition, hydration mechanism, 541  $\gamma$ -Oxoketene, protonation, ketone electrophilic reactions, 582 3-Oxomethylenene-6-methylene-1, 4-cyclohexadiene, photolysis of, 299-300 5-Oxo-2,4-pentadienes, molecular structure, 7 Oxoquinone methide carbon-substituted ketenes, 294-303 ester enolate formation, 77-78 Oxoxylylenes, carbon-substituted ketenes, 294-303 Oxygenation of ketenes, electrophilic reactions, 587-591 ¹⁷O nuclear magnetic resonance, ketene physical properties, 27-33 Oxygen nucleophiles carbon-substituted acylketenes, 249 ketene additions, 554-559 Oxygen-substituted ketenes cycloadditions with imines, 503-504

stereoselectivity, 610 structure and classification, 313–320 Oxyl radical side chains, ketene studies, 281 Ozone photolysis di-tert-butylketene, 198 cyclohexadienone formation, 143 ketone electrophilic reactions, 588 oxygen-substituted ketenes, 319 Palladium catalysts bromoketene formation, 339 ketene [2+2] cycloaddition, 465-466 Parent ketene ( $CH_2=C=O$ ), historical development of, 1 Pentafulvenone, structural properties, 284-292 Pentamethyleneketene intermolecular [2+2] cycloaddition and dimerization, 457 photolysis, 56 Peracid oxidation, arylalkynes, ketenes from, 183-185 Perfluorinated ketene, ketone electrophilic reactions, 590-591 Perfluoroalkylaldoketene, acid dehydration, 326 Perfluoroketene, flash vacuum pyrolysis, 325 Pericyclic mechanisms, ketene formation from ester pyrolysis, 81-84 Perkin conditions acid anhydride pyrolysis of ketenes, 62-64 intramolecular ketene cycloadditions, 522 Perpendicular, carbon/silicon nucleophilic additions to ketenes, 572 Phenol photoisomers, cyclohexadienones, 143 Phenoxyketenes cycloaddition reactions, 465 polymer supports for, 624 stereoselectivity, 613 Phenyl acetate, ester pyrolysis and ketene formation, 83 β-Phenylalkenylketene, intramolecular cycloadditions, 523 Phenylalkylketenes, carbon-substituted reactions, 230

Phenylbenzoylketene, diazo diketone photolysis, 55-256 Phenylchloroketene, cycloadditions, nucleophilic alkenes, 490-491 Phenyldiazomethane photolysis, phenylketene formation, 226 Phenyldimethylsilylketene, 358 Phenylethylketenes n-butyllithium addition, 231 carbon/silicon nucleophilic additions, 573 polymerization, 601-602 radical reactions, 596 stereoselectivity, 607 Phenylethynol, phenylketene generation, 224 Phenylethynylketene, photochemical Wolff rearrangement, 214-215 Phenylhydroxyketene, photolysis of, 314 Phenylketenes carbon substitution, 221-234 [3+2] cycloadditions, 513 dehydrohalogenation reaction, 69 ester enolate formation, 79 nucleophilic additions amination mechanisms, 546-547 hydration mechanisms, 535 protocol of Lectka, 223 protonation, ketone electrophilic reactions, 585 radical reactions, 596-598 stereoselectivity, 612 1,3-thiazolidine reaction, 186-187 Wolff rearrangement, sulfur ylides, 190 ynol/ynolate structures, 378 zinc dehalogenation, 221 Phenylmethylketenes carbon substitution, 231 iridium-alkyne complex, 394 stereoselectivity, 603-615 Phenyl radicals, ketene radical reactions. 593 Phenylsulfonylketene, 374 Phenyl(trifluoromethyl)ketene, dehalogenation, 328-329 Phenyltrimethylsilylketene, classification and formation. 344

Phenylvinylketene, cyclohexadienone interconversion, 143 Phosphaketene ylides, 366-367 Phosphine complexes ketene cycloaddition reactivities, 481-482 nucleophilic alkenes, 491 linkage-isomeric ketenes, 396 Phosphonate-substituted diazo ketones, thermal decomposition, 96-97 Phosphonium ions, gas phase reaction, ketene formation, 83-84 ³¹P nuclear magnetic resonance, phosphorus-substituted ketenes, 361-362 Phosphorus-substituted ketenes, classification and formation, 361-369 Phosphoryl chlorides, ketene formation, 64 Phosphoryl-substituted ketenes, rate constants, 365 Photochemical ketene reactions alternate routes for, 160-169 cycloalkanone and enone photolysis, 132-140 cyclobutanones/cyclobutenones, 114-129 cyclohexadienones and cycloalkenones, 142 - 151dioxinone-based ketenes, 153-155 Photochemical Wolff rearrangement cyclopropylketene formation, 239 diazo ketone formation carbon-substituted acylketenes, 251-252 carbon-substituted arylketenes, 231-232 carboxamide-substituted ketenes, 256 catalyzed Wolff rearrangement, 101 - 104decomposition problems, 104-110 diphenylketene formation, 225-226 gold surface adsorption, 203 α-trialkylsilylalkenylketenes, 208 2,2-diphenylcyclopropylketene, 238 formylketene, 244 oxiranylketenes, 242 pentafulvenone formation, 285-292 phenylethynylketene formation, 214-215 Photochemical Wolff (continued) silvlketenes, 345 stereoselectivity of ketenes, 609 Photodissociation of ketenes, 441-445 Photoelectron spectroscopy (PES) bisketenes, 402-403 ketene chiroptical properties, 35-38 Photoexcited ketenes, internal conversion, 442 Photofragmentation, deuterated ketenes, 442 Photoionization mass spectrometry, ketene physical properties, 51 Photolithographic applications carbon-substituted ketenes, 227 fulvenone transformation, 288 Photolysis chromium complexes, ketene formation, 173 ketene dimers, 56 ketenyl radicals, 278 Phthalaldehyde photolysis, hydrogen transfer, 297-298 Phthalidimido-tert-butylketene, nitrogen substitution, 306 Phthalimido ketenes, stereoselectivity, 605, 613 Phthalylketene, nitrogen substitution, 309 Picosecond infrared (IR) detection, diazo ketone ketene formation, photochemical Wolff rearrangement, 107-110  $\pi$ -bonding [2+2] cycloaddition reactions of ketenes, 475 metal-complexed ketenes, 389-398 Platinum(II) catalysts bisketene formation, 419 phosphorus-substituted ketenes, 367 Platinum-substituted ketenes, 388 PM3 calculated IR frequencies carbon-substituted ketenes, 277 diazo ketone ketene formation, photochemical Wolff rearrangement, 107-108 Polar solvents, diazo ketone ketene formation, 95-97 Polyketenes from cations and free radicals, 182

ethoxyalkynylsilane thermolysis, 158-159 polymerization reactions, 601-602 UV spectra of, 35 Polymerization of ketenes, 600-602 Polymer supports, ketene reactions using, 622-625 Positive charge buildup, ketene cycloaddition kinetics with alkenes, 482 Post-Hartree-Fock calculations, ketene molecular structure, 6 Potassium dicarbonyl, ynol/ynolate ketenes, 378 Proline-derived ketenes, cycloaddition with imines, 508 Propandienone, cumulenone formation, 430-431 β-Propiolactone dimers, ketene cycloadditions, 460-461 Propylene imine, silylketene formation, 348-349 Proton abstraction, ketene computational studies, 10-11 Proton affinities, ketene thermochemistry, 22 - 25Protonation ketene molecular dynamics, 10 ketene photolysis, 443 ketone electrophilic reactions, 580-585 Pseudopericyclic decarbonylation [4+2] cycloadditions of ketenes, 518 furandione, ketene formation, 165 imidoylketenes, 263-267 2-Pyranone photolysis, alkenylketene formation, 210 Pyran ring opening, ketene formation, 125 Pyridine ring electrocyclization, ketene formation, 126 2-Pyridylcyanoketene, flash vacuum thermolysis, 233 Pyridylketenes nucleophilic additions amination mechanisms, 546-547 hydration mechanisms, 540 Wolff rearrangement, 232 3-Pyridyl(methyl)ketenee, Wolff rearrangement, 232 Pyrimidones, photochemical ketene formation, 163-164

Pyrolysis acid anhydride-ketene formation, 61-64 alkylketene preparation, 195 dimethylketene thermolysis, 447 esters, ketene formation from, 81-84 ketene dimers. 55-56 ketene formation, alternate routes, 189 2-Pyrone photolysis, photochemical ketene formation, 162-163 Pyrrole substituents, chromium carbene complex, 177 Quaternary carbon structure, ketene formation, 175 o-Quinone thermolysis, ketene formation, 149 Radical anions and cations carbon-substituted ketenes, 273-283 cumulenones, 430-433 cyclohexadienones and dienylketenes, 148 di-tert-butylketene, 198 ketene oxidation, 437-440 ketene reduction, 439-440 ketenyl structures, 425-427 Radical ketene reactions, 592-598 Raman spectroscopy, ketene physical properties, 39-45 Rate constants bisketene ring closure, 404-405 ketene cycloadditions alkenes, 468, 480-482 enol ethers, 480-481 ethoxypropene, 481 phosphines, 481-482 nucleophilic additions to ketenes amination mechanisms, 545-546 hydration mechanisms, 535-538 phosphoryl-substituted ketenes, 365 Reaction mechanisms halogen-substituted ketenes, 333 ketene computational studies, 9-12 nucleophilic additions to ketenes, amination mechanisms, 549-550 Reactive ketenes, dehydrochlorination, 67-68 Reduction of ketenes mechanisms for, 439-440 nucleophilic additions, hydride reduction, 552-553

Regiochemistry, ketene [2+2] cycloaddition, 479-480 Regioselectivity, Z-dibenzoylethylenes, 137 Resonance structures diazo ketones, ketene formation, 90-93 ketene dipole moments, 48-49 ketene IR frequencies, 40-45 ketene molecular orbitals, 5-6 substituent effects, 15-17 Resorcinol derivatives, trimethylsilylketene formation, 351 Reversibility, ketene cycloadditions, 452 Rhodium catalysts diazo ketones, catalyzed Wolff rearrangement, 98-104 oxygen-substituted ketenes, 319 phosphorus-substituted ketenes, 363 silylketenes, 355-356 thio-substituted ketenes, 372-373 Ring closure bisketenes, 404-405 cycloadditions of ketenes, nucleophilic alkenes, 489-491 Ring opening mechanism boron-substituted ketenes, 386 cyclohexadienone ketene formation, 142 - 151guanine diazotization, 277 oxygen-substituted ketenes, 314-315 photochemical ketene formation, 114-129 ynol/ynolate classification, 377-382 Rotational spectra, ketene measurements, 39-45 Ruthenium catalyst bisketene formation and, 407 metal-substituted ketenes, 397-398 Rydberg states, chiroptical properties of ketenes, 35-38 Selected ion flow tube (SIFT) reaction, ketene radicals, 51-52 Selenoxide derivative photolysis, ketene formation, 167 Selenyl esters, ketenyl radicals, 278-279 Short-lived ketenes, spectroscopic analysis,

33

Shuttle mechanism phenyketene formation by dehydrochlorination, 69-70 phenylketene generation, 223 Side chain reactions carbon-substituted ketenes, 273-283 dioxinone formation of ketones, 154 σ-bonding, metal-complexed ketenes, 389-398  $\beta$ -Silicon effect, silvlketene stabilization, 16, 346 ²⁹Si nuclear magnetic resonance, ketene physical properties, 29-33 Silicon nucleophiles, ketene additions, 568-574 Silicon-substituted ketenes, 387 Silver salts diazo ketones, catalyzed Wolff rearrangement, 97-104 metal ketenides, 387 Silyl(alkenyl)ketene, 345 α-Silylated alkenylketenes, 354 Silylated para-benzoquinone, vinylketene formation, 212 Silylated vinylketenes, 358 Silyl enol ether, cycloadditions of ketenes, nucleophilic alkenes, 489-490 Silvlgermyl ketenes, 354 Silylketenes classification and formation, 342-357 cycloadditions, carbonyl groups, 528 ethoxyalkynylsilane thermolysis, 158-159 mass spectra of, 50 nucleophilic additions, hydride reduction, 552-553 photochemical Wolff rearrangement, 345 physical properties, spectroscopic analysis, 29-33 stabilization energy, 16, 346-347 X-ray studies of, 20-21 ynol/ynolates, 382 Silyloxy-substituted ketenes, 214-219, 319 β-Silyl substituents, Norrish Type I cleavage of ketones, 132 Simple ketenes, thermochemistry, 22-25 Smith-Hoehn reaction, intramolecular ketene cycloadditions, 523 Sodioketene, stabilization energy, 15-17

Solvent effects cyclobutenone ring opening, 119 ketene cycloadditions, imine reactions, 502-503 nucleophilic additions to ketenes, hydration mechanisms, 540 ynol/ynolate ketenes, 378-382 Solvent polarity rate exo/endo cycloaddition stsereochemistry, 467-468 ketene dimerization, 461-462 norbornadiene, 472 Spirocyclopentenone, photochemical ketene formation, 139 Squaric acid, bisketene formation, 405 Stabilization energy (SE) ketene substituent effects, 13-17 metal-substituted ketenes, 387-389 Stannylketenes, 342-357 boron substituents, 385 nucleophilic additions, hydride reduction, 552-553 Stepwise reaction pathways, intermolecular [2+2] cycloaddition of ketenes, alkene reactions, 476-484 Stereoselectivity diazo ketone formation catalyzed Wolff rearrangement, 101-104 photochemical Wolff rearrangement, 105 - 110dioxinone formation of ketones, 153-155 ketene chiral centers, 603-615 ketene cycloadditions, 451 alkenes, 479-484 carbonyl groups, 527-530 imine reactions, 502-510 phosphorus-substituted ketenes, 365-366 photochemical ketene formation, cyclobutanone thermolysis, 115 Steric interactions bicyclic ketone hydrogen transfer, 132 cycloaddition reactions of ketenes, 466 alkene reactions, 474-485 ketene physical properties, spectroscopic analysis, 29-33 Stretching frequencies HCl/HBr, ketene formation, 71-72 ketene IR spectra, 41-45

Styrenes, ketene cycloaddition kinetics, 483-484 Substituent effects cyclobutenone ring opening, 120 diazo ketones, ketene formation, 90-93 fluoroketene destabilization. 321-322 on ketenes, 13-17 infrared spectroscopic analysis, 40-45 metal-complexed ketenes, 389-398 Sulfoxide derivative photolysis, ketene formation, 167 Sulfur dioxide, ketene reactions with, 621 Sulfur-substituted ketenes classification, 371-375 stereoselectivity, 610 Suzuki cross-coupling reactions, polymer supports for ketenes, 622-625 Syn conformation ketene dimerization, 459-460 vinylketene, 206 "Syndnones," photolysis of, 307-308 Tantalum complex, ketene formation, 394-395 TEMPO aminoxyl radical 1,4-bisketene formation, 402-403 ketene radical reactions, 594-598 pentafulvenone formation, 287-292 Terephthalyl chloride, bisketene formation and. 414 Tetraketene metal-substituted bisketenes and, 405 X-ray studies of, 20 Tetraketone, ketene formation, 166 Tetramethylcyclopropylideneketene, dehalogenation, 196-197 Tetramethylketene cyclopentenyl ketone formation of, 134 photolysis, 56 2,3,5,6-Tetra(trimethylsilyl)-1,4benzoquinone, ketene formation, 149 Thermal intramolecular cyclization, acyl chloride ketenes, 201 Thermal Wolff rearrangement alkyl- and arylketenes, microwave irradiation, 200 arylketene photelectron spectra, 438-440 boron-substituted ketenes, 385-386 [4+2] cycloadditions of ketenes, 518-519

diazo ketones, 93-97 cyclopropylketenes, 241-242 diphenyl ketene formation, 221-222 diazulenylketene, 226 ketene decarbonvlation, 448 metal-substituted bisketenes, 417 oxygen-substituted ketene formation, 106 phosphorus-substituted ketenes, 363-365 polymer-bound ketene, 357 thiadiazocycloheptanone, ketene formation, 199 Thermochemistry of ketenes, 22-25 Thermolytic ketene formation alkynyl ether thermolysis, 156-159 alternate routes for, 160-169 cycloalkanone and enone photolysis, 132-140 cyclobutanones/cyclobutenones, 114-129 cyclohexadienones and cycloalkenones, 142 - 151dioxinone-based ketenes, 153-155 reaction mechanisms, 446-449 Thiadiazocycloheptanone, thermal Wolff rearrangement, 199 1,3-Thiazolidine reaction, phenylketenes, 186 Thioesters, ketenyl radicals, 279 Thioketene, molecular structure determination of, 18-21 Thiophene derivatives, cumulenone formation, 431-432 Thiophosphorylketene, 364 Thorium complex, silvlated metalloxy ketenes, 347 Three-phase test of reaction intermediates, ester enolate formation of ketenes, 80 Thulium acetate hydrate, acetic acid-ketene formation, 58 Thymine hydroxylase, covalent modification, ketene formation, 184 Time-resolved infrared (TRIR) analysis carbon-substituted phenylketenes, 224 diazo ketone ketene formation, photochemical Wolff rearrangement, 106-107 ketene/bisketene infrared spectra, 41-45 phenylethynyllketene, 214-215

Time-resolved infrared (continued) phenylhydroxyketene, 314 phenylketenes, ynol/ynolate ketenes, 378 short-lived ketenes, 33 Titanium oxide surfaces, acetic acid-ketene formation, 58 Torquoselectivity allenylcyclobutenone, 127 ring opening of cyclobutenones, photochemical ketene formation, 119 Tosylate anhydrides, ketene formation, 63-64 Transesterification, dioxinone, 250 2,5-(Trans-4-octenyl)-bridged pbenzoquinone, ketene formation, 138-139 Triafulvenone, structural properties, 284-285 Trialkylamine, dehydrochlorination of ketenes. 69 α-Trialkylsilylalkenylketenes, photochemical formation, 208 Trialkylsilyl(aryl)ketenes, Wolff rearrangements, 355 Trialkylsilylketenes, Wittig reagents, 351 Triazoline photolysis, tricyclic ketone formation, 165 Tricyclic ketones, thermolytic ketene formation, 160 Triflate carbonylation, alkenylketene formation, 212 Trifluoroacetylketene, Meldrum acid derivatives. 257 Trifluoromethylmalonate esters, fluoroketene formation, 329 Trimethylgermylketene, diazomethane reactions, 619 Trimethylsilyl ester pyrolysis, ketene formation, 82-83 Trimethylsilylketenes alkynyl ether thermolysis, 156-157 carbonylation, 347-348 cycloaddition reactions, 350, 352 carbonyl groups, 528 nucleophilic alkenes, 488-489 dehydrochlorination and formation of, 66 diazomethane reactions, 346, 619 dimerization resistance, 346

iron ketene complex, 395-396 ketone electrophilic reactions, protonation, 581, 584 oxetanone formations, 349-350 pyrolysis of, 342 stereoselectivity, 611 Zwitterionic intermediate, 357 Trimethylsilyl-substituted dienylketene, thermolysis, 354-355 Trimethylsilylvinylketene, dehydrochlorination, 344 Triphenylarsenylideneketene, 369 2,2,3-Triphenylcyclobutanone, photochemical ketene formation, 116 2,2,4-Triphenylcyclobutenone, photochemical ketene formation, 118 Triphosgene, carboxylic acid-ketene formation, 59 Triplet ketocarbenes, photochemical Wolff rearrangement, 105-110 Triplet oxygen, ketene electrophilic additions, 590 Tungsten-substituted ketenes, 369 Two-photon absorption, diazo ketone ketene formation, photochemical Wolff rearrangement, 109-110

Ultraviolet (UV) spectroscopic analysis ketene chiroptical properties, 35–38 nucleophilic additions to ketenes, hydration mechanisms, 535–538 Unimolecular dissociation, ketene photolysis, 442 Unsaturated groupings, ketene cycloadditions, 465 Unsymmetrical ketenes, dimerization, 459 Uranium oxide, acetic acid-ketene formation, 58

Valine-derived ketenes, nitrogen substitution, 309
Verbinone photolysis, alkenylketene formation, 132–133
Vibrational spectra, ketene measurements, 39–45
Vinylcyclobutenone ketene formation, 120 vinyl ene-yne ketone formation, 123–125

Vinylcyclopropyl carbene complex, ketene formation, 179 Vinylcyclopropylketenes, photochemical Wolff rearrangement, 239 Vinyl esters, ketene formation, 83 Vinyl ethers, ketene cycloaddition, nucleophilic alkenes, 487-491 3-(2-Vinyl)indolylcarbene complexes, ketene formation, 178 Vinylketenes as alkenylketene parent, 206-213 alkynyl ether thermolysis, 157-158 allenyl thio ester thermolysis via, 161 carbocation derivatives, 276 chromium carbene complex, 176 collisionally activated dissociation of resorcinol derivatives, alkenylketene intermediates, 150 cyanoketene formation, 218 cycloaddition reactions, 10 alkynes, 495-499 cyclobutanone photolysis, 117 cyclobutenone ring opening, 118-129 cyclobutenone thermal rearrangement, 121 intramolecular cycloadditions, 520-524 ketone electrophilic reactions, protonation, 581 mass spectra of, 51 metal-complexed ketenes, 390-398 microwave spectrum, 18 molecular structure, 7 oxygen nucleophile reactions, 555 tricarbonyl iron complex, 393 Vinylogous Wolff rearrangement 3-butenylketene formation, 136 photochemical ketene formation, cyclobutanones, 116-129 Vinyl phosphate, fluoroketene formation, 325 Vinyl-substituted acyl radicals, carbonsubstituted ketenes, 278-279 Water-soluble cyclohexadienones, amino acids and potolysis of, 146 Wittig reaction BHT ester formation of ketene, 78-79 nucleophilic ketene additions, 576-579 silylketene cycloadditions, carbonyl groups, 528

trimethylsilylketene, 342 Wolff rearrangement 1-adamantyl(carboethoxy)ketene, 245 arylalkylketene formation, 230 aryl(trialkyl)silylketenes, 234 benzopentafulvenone, 289 bisketene formation, 401 bis(trimethylsilyl)ketene, 343-344 di-tert-butylketene formation, 199 carbon-substituted phenylketenes, 224-22 chiral ketene stereoselectivity, 606 cyclopropenylketene, 242 cyclopropylketene preparation, 238 diazo ketones, ketene formation alternative routes, 110 catalyzed rearrangement, 97-104 overview, 92-93 oxygen-substituted ketenes, 318-319 photochemical, 104-110 3-pyridyl(methyl)ketenee, 232 thermal rearrangement, 93-97 fluoroalkylketene formation, 326-327 ketene cycloadditions with carbonyls, 531 with imines, 504 ketene mass spectra, 51 ketene structure, 2-3 computational studies, 6-7 methylthioketene, 371 pentafulvenone formation, 284-292 phenylsulfonylketene, 374 phosphorus-substituted ketenes, 362 photochemical ketene formation, vinylogous catalyst, 116-129 trialkylsilyl(aryl)ketenes, 355 vinylketenes, intramolecular cycloadditions, 521 Woodward-Hoffmann rules intermolecular [2+2] cycloaddition of ketenes alkene reactions, 474-485 dimerization, 455 ketene cycloadditions, 451

Xenon matrix, ketene IR spectra, 41–45 X-ray diffraction allenylketenes, 269–270 molecular structure of ketenes, 20 Ylides, nitrogen nucleophile/electrophile ketene reactions, 564-565 Ynamine reaction, acylketene formation, 187 **Ynolates** collisionally activated dissociation of resorcinol derivatives, alkenylketene intermediates, 150 ketenyl radicals, anions, and cations, 425-427 silvlketene formation, 349 structure and classification, 377-382 Ynols, structure and classification, 377–382 Z-dibenzoylethylene, ketene formation, 136-137 Zinc activation arylalkylketene formation, 230 chloroketenes, 332-333 cycloaddition of ketenes, alkynes, 495-499 dibromoketenes, 339 difluoroketenes, 323-324 fluorenylideneketene, 290  $\alpha$ -halo carboxylic acid derivatives, ketene formation, 87 Zirconium ketenylidene complexes, 388

Zwitterionic intermediates aminocarbene complex, 172 bis(trifluoromethyl)ketenes, 328 carbon-substituted acylketenes, 246 cycloadditions of ketenes, 469-470 alkenes, 480-484 allenes, 493-494 imine reactions, 500-510 nucleophilic alkenes, 488-491 cyclohexylamine reaction, 147 diphenylketenes, 464-465 ketene dimerization, 459-460 metal-substituted bisketenes, 412 methylketene dimerization, 456 nucleophilic additions dienylketenes, amination mechanisms, 548 oxygen nucleophile reactions, 557-559 oxygen-substituted ketenes, 316 pentafulvenone formation, 287, 291-292 phosphorus-substituted ketenes, 365-366 photochemical ketene formation, 116 stereoselectivity of ketenes, 603–615 trimethylsilylketene, 357